Contents lists available at ScienceDirect

Physics Reports

journal homepage: www.elsevier.com/locate/physrep

The gravitational eikonal: From particle, string and brane collisions to black-hole encounters



HYSICS REPORTS

Paolo Di Vecchia ^{a,b}, Carlo Heissenberg ^{c,b}, Rodolfo Russo ^{d,*}, Gabriele Veneziano ^{e,f}

^a The Niels Bohr Institute, Blegdamsvej 17, DK-2100, Copenhagen, Denmark

^b NORDITA, KTH Royal Institute of Technology and Stockholm University, Hannes Alfvéns väg 12, SE-11419, Stockholm, Sweden

^c Department of Physics and Astronomy, Uppsala University, Box 516, SE-75120, Uppsala, Sweden

^d Centre for Theoretical Physics, Department of Physics and Astronomy, Queen Mary University of London, Mile End Road, E1

4NS London, United Kingdom

^e Theory Department, CERN, CH-1211 Geneva 23, Switzerland

^f Collège de France, 11 place M. Berthelot, 75005 Paris, France

ARTICLE INFO

Article history: Received 22 June 2023 Received in revised form 16 April 2024 Accepted 10 June 2024 Available online xxxx Editor: Stephan Stieberger

Keywords: Eikonal exponentiation Scattering amplitudes Black holes Gravitational waves

ABSTRACT

Motivated by conceptual problems in quantum theories of gravity, the gravitational eikonal approach, inspired by its electromagnetic predecessor, has been successfully applied to the transplanckian energy collisions of elementary particles and strings since the late eighties, and to string-brane collisions in the past decade. After the direct detection of gravitational waves from black-hole mergers, most of the attention has shifted towards adapting these methods to the physics of black-hole encounters. For such systems, the eikonal exponentiation provides an amplitude-based approach to calculate classical gravitational observables, thus complementing more traditional analytic methods such as the Post-Newtonian expansion, the worldline formalism, or the Effective-One-Body approach. In this review we summarize the main ideas and techniques behind the gravitational eikonal formalism. We discuss how it can be applied in various different physical setups involving particles, strings and branes and then we mainly concentrate on the most recent developments, focusing on massive scalars minimally coupled to gravity, for which we aim at being as self-contained and comprehensive as possible.

© 2024 The Author(s). Published by Elsevier B.V. This is an open access article under the CC BY license (http://creativecommons.org/licenses/by/4.0/).

Contents

1.	Introdu	uction	3
	1.1.	Aims and outline	3
	1.2.	Kinematics and conventions	6
	1.3.	Partial waves, unitarity, phase shifts and the eikonal phase	9
	1.4.	Regimes of $2 \rightarrow 2$ scattering	11
2.	Leading exponentiation: an appetizer		12
	2.1.	Minimally coupled massless scalars: from amplitudes to geometry	12
	2.2.	Eikonal phase and phase shift	17
	2.3.	Minimally coupled massless scalars: from geometry to amplitudes	19

* Corresponding author.

E-mail addresses: divecchi@nbi.dk (P. Di Vecchia), carlo.heissenberg@physics.uu.se (C. Heissenberg), r.russo@qmul.ac.uk (R. Russo), gabriele.veneziano@cern.ch (G. Veneziano).

https://doi.org/10.1016/j.physrep.2024.06.002

0370-1573/© 2024 The Author(s). Published by Elsevier B.V. This is an open access article under the CC BY license (http://creativecommons.org/ licenses/by/4.0/).



	2.4.	Collapse criteria in GR	20
3.	Leadin	g eikonal for generic gravitational 2-body scattering	21
	3.1.	Field theory at tree-level	21
		3.1.1. Minimally coupled massive scalars	21
		3.1.2. Graviton scattering off a massive scalar	23
		3.1.3. The effect of the dilaton on the eikonal scattering	25
		3.1.4. Maximally supersymmetric gravity	26
		3.1.5. Effective field theories beyond GR.	29
	2.2	3.1.6. Including classical spin	30
	3.2.	String theory	33
		3.2.1. String-orane scattering at tree level.	34
		5.2.2. The Closed/open string transition	38
		5.2.5. The Regeon vertex formation	40
		3.2.4. The string enhoused operator, the closed string sector	42
		3.2.6 String cattering at tree level	44
		327 Brane-brane scattering at leading order	49
4	Expon	Sign and the subleading 2-body eikonal	50
	4.1.	1-loop (2PM) in OFT	51
		4.1.1. Massive $\mathcal{N} = 8$ supergravity	51
		4.1.2. Real-analytic, crossing-symmetric reformulation	54
		4.1.3. General relativity	55
		4.1.4. Real-analytic, crossing-symmetric reformulation	58
		4.1.5. The probe limit	59
		4.1.6. Tidal effects in field theory	60
	4.2.	Eikonal exponentiation in string theory at one loop	60
		4.2.1. 1-loop in string theory	60
		4.2.2. Tidal forces in string theory	63
		4.2.3. A geometric description of the eikonal scattering.	64
		4.2.4. String elkonal and classical causality	66
F	Unitar	4.2.5. String-string scattering beyond tree level	6/
э.		Ity and radiation-reaction.	72
	5.1.	Ontarity in momentum space and in impact-parameter space	72
		5.1.1. Editor unitarity	75
	52	Single model and the second se	79
	5.3	Radiation reaction from $\text{Im} 2\delta_2$ by real analyticity.	80
6.	The 2-	-body eikonal at 3PM	81
	6.1.	Partial wave decomposition: from b to b ₁	81
	6.2.	Massive $\mathcal{N} = 8$ supergravity	83
		6.2.1. Eikonal and deflection angle to 3PM order	85
		6.2.2. Real-analytic, crossing-symmetric reformulation at two loops	87
	6.3.	General relativity	88
		6.3.1. Two-particle and three-particle cuts	89
		6.3.2. Real-analytic, crossing-symmetric ansatz in GR	91
		6.3.3. The 3PM eikonal $2\delta_2$.	93
		6.3.4. Extracting the radial action from the amplitude	94
7	The of	6.3.5. Radial action, effective potential, PN limit and bound orbits	98
7.	7 1	Kona operator in the soft mint	102
	7.1.	Silvenal operator including static modes	105
	7.2.	The ultrarelativitic and massless limits	117
	7.3.	The $\mathcal{M} = 8$ eikonal operator in the soft limit	112
8.	The ei	hold operator beyond the soft limit.	114
	8.1.	The elastic eikonal revisited	114
		8.1.1. Saddle points	115
		8.1.2. Impulse and angular momentum	116
	8.2.	Coherent state approximation beyond the soft limit	117
		8.2.1. Eikonal operator without static modes	118
		8.2.2. Eikonal operator including static modes	120
		8.2.3. The $N = 8$ eikonal operator	120
	8.3.	Computing observables from the eikonal operator	121
		8.3.1. Waveforms	121
		8.3.2. Linear momentum	125
0	Current of	8.3.3. Angular momentum	128
9.	Summ	ary and outlook	132

Declaration of competing interest.138Acknowledgments139Appendix A. Field theory details and conventions139A.1. Permutation identities139A.2. Feynman rules140A.3. Useful Fourier transforms to impact parameter space141Appendix B. The deflection angle in the probe limit142B.1. Geodesic motion in Schwarzschild143B.2. Geodesic motion in D-brane metric145B.3. Probe limit of the $N = 8$ case146Appendix C. Some relations satisfied by the Levi-Civita tensor in $D = 4$.147Appendix D. String theory background147D.1. String theory conventions.147D.1.1. Scales and coupling constants.149D.1.2. String mode expansion.149D.1.4. The boundary state149D.1.4. The boundary state149D.2. String-brane scattering: the bosonic theory at tree-level150D.2.1. The Reggeon vertex formalism for bosonic strings152D.3. String-brane scattering: the bosonic theory at tree-level150D.2.1. The Regreen vertex formalism for bosonic strings152D.3. String-brane scattering: the bosonic theory at one-loop154Appendix F. From the field to the asymptotic waveform158References159	CRedit authorship contribution statement 138				
Acknowledgments139Appendix A. Field theory details and conventions139A.1. Permutation identities139A.2. Feynman rules140A.3. Useful Fourier transforms to impact parameter space141Appendix B. The deflection angle in the probe limit142B.1. Geodesic motion in Schwarzschild143B.2. Geodesic motion in D-brane metric145B.3. Probe limit of the $\mathcal{N} = 8$ case146Appendix C. Some relations satisfied by the Levi-Civita tensor in $D = 4$.147D.1. String theory conventions.147D.1.1. Scales and coupling constants.147D.1.2. String mode expansion.149D.1.3. Normalizations for string amplitudes.149D.1.4. The boundary state149D.2. String-brane scattering; the bosonic theory at tree-level150D.2.1. The Reggeon vertex formalism for bosonic strings152D.3. String-brane scattering; the bosonic theory at one-loop.154Appendix F. From the field to the asymptotic waveform158References159	Declaration of competing interest				
Appendix A. Field theory details and conventions139A.1.Permutation identities139A.2.Feynman rules140A.3.Useful Fourier transforms to impact parameter space141Appendix B. The deflection angle in the probe limit142B.1.Geodesic motion in Schwarzschild143B.2.Geodesic motion in D-brane metric143B.3.Probe limit of the $\mathcal{N} = 8$ case146Appendix D. String theory background147D.1.String theory background147D.1.1.Scales and coupling constants147D.1.2.String mode expansion149D.1.3.Normalizations for string amplitudes149D.1.4.The boundary state149D.2.String-brane scattering: the bosonic theory at tree-level150D.3.String-brane scattering: the bosonic theory at one-loop.154Appendix E. More on the $2 \rightarrow 3$ kinematics and on the \times , + waveforms.158References159159	Acknowledgments				
A.1.Permutation identities.139A.2.Feynman rules.140A.3.Useful Fourier transforms to impact parameter space141Appendix B. The deflection angle in the probe limit142B.1.Geodesic motion in Schwarzschild143B.2.Geodesic motion in D-brane metric145B.3.Probe limit of the $\mathcal{N} = 8$ case146Appendix C. Some relations satisfied by the Levi-Civita tensor in $D = 4$.147D.1.String theory background147D.1.String theory conventions147D.1.3.Normalizations for string amplitudes.149D.1.4.The boundary state149D.1.5.String-brane scattering: the bosonic theory at tree-level152D.3.String-brane scattering: the bosonic theory at one-loop.154Appendix E. More on the $2 \rightarrow 3$ kinematics and on the \times , + waveforms.159References159	Appendix A. Field theory details and conventions				
A.2.Feynman rules140A.3.Useful Fourier transforms to impact parameter space141Appendix B. The deflection angle in the probe limit142B.1.Geodesic motion in Schwarzschild143B.2.Geodesic motion in D-brane metric145B.3.Probe limit of the $\mathcal{N} = 8$ case146Appendix C. Some relations satisfied by the Levi-Civita tensor in $D = 4$.147Appendix D. String theory background147D.1.String theory conventions.147D.1.1.Scales and coupling constants147D.1.2.String mode expansion149D.1.3.Normalizations for string amplitudes149D.1.4.The boundary state149D.2.String-brane scattering: the bosonic theory at tree-level152D.3.String-brane scattering: the bosonic theory at one-loop.154Appendix E. More on the $2 \rightarrow 3$ kinematics and on the \times , + waveforms.158References159	A.1. Permutation identities				
A.3.Useful Fourier transforms to impact parameter space141Appendix B. The deflection angle in the probe limit142B.1.Geodesic motion in Schwarzschild143B.2.Geodesic motion in D-brane metric145B.3.Probe limit of the $\mathcal{N} = 8$ case146Appendix C. Some relations satisfied by the Levi-Civita tensor in $D = 4$.147Appendix D. String theory background147D.1.String theory conventions.147D.1.1.Scales and coupling constants147D.1.2.String mode expansion.149D.1.3.Normalizations for string amplitudes.149D.1.4.The boundary state149D.2.String-brane scattering: the bosonic theory at tree-level150D.3.String-brane scattering: the bosonic theory at one-loop.154Appendix E. More on the $2 \rightarrow 3$ kinematics and on the \times , + waveforms.156Appendix F. From the field to the asymptotic waveform158References159	A.2. Feynman rules	. 140			
Appendix B. The deflection angle in the probe limit142B.1.Geodesic motion in Schwarzschild143B.2.Geodesic motion in D-brane metric145B.3.Probe limit of the $\mathcal{N} = 8$ case146Appendix C. Some relations satisfied by the Levi-Civita tensor in $D = 4$.147Appendix D. String theory background147D.1.String theory conventions.147D.1.1.Scales and coupling constants147D.1.2.String mode expansion.149D.1.3.Normalizations for string amplitudes149D.2.String-brane scattering: the bosonic theory at tree-level150D.3.String-brane scattering: the bosonic theory at one-loop.154Appendix E. More on the $2 \rightarrow 3$ kinematics and on the \times , + waveforms.158References159	A.3. Useful Fourier transforms to impact parameter space	. 141			
B.1.Geodesic motion in Schwarzschild143B.2.Geodesic motion in D-brane metric145B.3.Probe limit of the $\mathcal{N} = 8$ case146Appendix C. Some relations satisfied by the Levi-Civita tensor in $D = 4$.147Appendix D. String theory background147D.1.String theory conventions.147D.1.1.Scales and coupling constants147D.1.2.String mode expansion149D.1.3.Normalizations for string amplitudes149D.1.4.The boundary state149D.2.String-brane scattering: the bosonic theory at tree-level150D.3.String-brane scattering: the bosonic theory at one-loop154Appendix E. More on the $2 \rightarrow 3$ kinematics and on the \times , + waveforms158References159	Appendix B. The deflection angle in the probe limit	. 142			
B.2.Geodesic motion in D-brane metric145B.3.Probe limit of the $\mathcal{N} = 8$ case146Appendix C. Some relations satisfied by the Levi-Civita tensor in $D = 4$.147Appendix D. String theory background147D.1.String theory conventions.147D.1.1.Scales and coupling constants147D.1.2.String mode expansion.149D.1.3.Normalizations for string amplitudes149D.1.4.The boundary state149D.2.String-brane scattering: the bosonic theory at tree-level150D.3.String-brane scattering: the bosonic theory at one-loop.154Appendix E. More on the $2 \rightarrow 3$ kinematics and on the \times , + waveforms.158References159	B.1. Geodesic motion in Schwarzschild	. 143			
B.3. Probe limit of the $\mathcal{N} = 8$ case146Appendix C. Some relations satisfied by the Levi-Civita tensor in $D = 4$.147Appendix D. String theory background147D.1. String theory conventions.147D.1.1. Scales and coupling constants147D.1.2. String mode expansion.149D.1.3. Normalizations for string amplitudes149D.1.4. The boundary state149D.2. String-brane scattering: the bosonic theory at tree-level150D.3. String-brane scattering: the bosonic theory at one-loop.154Appendix E. More on the $2 \rightarrow 3$ kinematics and on the \times , + waveforms.158References159	B.2. Geodesic motion in D-brane metric	. 145			
Appendix C. Some relations satisfied by the Levi-Civita tensor in $D = 4$	B.3. Probe limit of the $N = 8$ case	. 146			
Appendix D. String theory background147D.1. String theory conventions.147D.1.1. Scales and coupling constants147D.1.2. String mode expansion.149D.1.3. Normalizations for string amplitudes149D.1.4. The boundary state149D.2. String-brane scattering: the bosonic theory at tree-level150D.2.1. The Reggeon vertex formalism for bosonic strings152D.3. String-brane scattering: the bosonic theory at one-loop.154Appendix E. More on the $2 \rightarrow 3$ kinematics and on the \times , + waveforms.156Appendix F. From the field to the asymptotic waveform158References159	Appendix C. Some relations satisfied by the Levi-Civita tensor in $D = 4$				
D.1.String theory conventions	Appendix D. String theory background	. 147			
D.1.1.Scales and coupling constants147D.1.2.String mode expansion149D.1.3.Normalizations for string amplitudes149D.1.4.The boundary state149D.2.String-brane scattering: the bosonic theory at tree-level150D.2.1.The Reggeon vertex formalism for bosonic strings152D.3.String-brane scattering: the bosonic theory at one-loop154Appendix E. More on the $2 \rightarrow 3$ kinematics and on the \times , + waveforms156Appendix F. From the field to the asymptotic waveform158References159	D.1. String theory conventions	. 147			
D.1.2.String mode expansion	D.1.1. Scales and coupling constants	. 147			
D.1.3. Normalizations for string amplitudes149D.1.4. The boundary state149D.2. String-brane scattering: the bosonic theory at tree-level150D.2.1. The Reggeon vertex formalism for bosonic strings152D.3. String-brane scattering: the bosonic theory at one-loop154Appendix E. More on the $2 \rightarrow 3$ kinematics and on the \times , + waveforms156Appendix F. From the field to the asymptotic waveform158References159	D.1.2. String mode expansion	. 149			
D.1.4. The boundary state149D.2. String-brane scattering: the bosonic theory at tree-level150D.2.1. The Reggeon vertex formalism for bosonic strings152D.3. String-brane scattering: the bosonic theory at one-loop154Appendix E. More on the $2 \rightarrow 3$ kinematics and on the \times , + waveforms156Appendix F. From the field to the asymptotic waveform158References159	D.1.3. Normalizations for string amplitudes	. 149			
D.2. String-brane scattering: the bosonic theory at tree-level150D.2.1. The Reggeon vertex formalism for bosonic strings152D.3. String-brane scattering: the bosonic theory at one-loop154Appendix E. More on the $2 \rightarrow 3$ kinematics and on the \times , + waveforms156Appendix F. From the field to the asymptotic waveform158References159	D.1.4. The boundary state	. 149			
D.2.1. The Reggeon vertex formalism for bosonic strings152D.3. String-brane scattering: the bosonic theory at one-loop.154Appendix E. More on the $2 \rightarrow 3$ kinematics and on the \times , + waveforms.156Appendix F. From the field to the asymptotic waveform.158References159	D.2. String-brane scattering: the bosonic theory at tree-level	. 150			
D.3. String-brane scattering: the bosonic theory at one-loop.154Appendix E. More on the $2 \rightarrow 3$ kinematics and on the \times , + waveforms.156Appendix F. From the field to the asymptotic waveform.158References159	D.2.1. The Reggeon vertex formalism for bosonic strings	. 152			
Appendix E. More on the $2 \rightarrow 3$ kinematics and on the \times , + waveforms	D.3. String-brane scattering: the bosonic theory at one-loop	. 154			
Appendix F. From the field to the asymptotic waveform 158 References 159	Appendix E. More on the 2 \rightarrow 3 kinematics and on the \times , + waveforms				
References	Appendix F. From the field to the asymptotic waveform				
	. 159				

1. Introduction

1.1. Aims and outline

Gravitational scattering, as seen from a Quantum-Field-Theory (QFT) perspective, was not a popular subject among particle theorists until the mid eighties. When in 1965 Steven Weinberg studied infrared gravitons in his now classic paper [1], he felt obliged to provide some "reasons for now attacking this question". One, he said, is "because I can", the second was that "something might go wrong" …but, he immediately added, "does not". For 20 more years or so particle theory was so much focused on the newly-formulated Standard Model of non-gravitational interactions—and of the elementary particles affected by them—that little attention was paid to the gravitational force in spite of the daring 1974 Scherk–Schwarz proposal [2,3] that the old (and unsuccessful) hadronic string theory should be recycled as a viable way to reconcile General Relativity (GR) with the quantum theory.

The situation changed dramatically in 1984 following the so-called first string revolution triggered by the breakthrough work by Green and Schwarz [4] making it at least plausible that some fully consistent (i.e. anomaly free) superstring theories could actually be candidate theories for all known forces and elementary particles. Although that idea still belongs to the dream category, the Green–Schwarz development gave a strong motivation for studying its implications as a full-fledged theory of quantum gravity. Starting in 1987 several groups [5–17] started to analyze, in the spirit of the "thought (gedanken) experiments" of the old quantum mechanics days, quantum gravitational scattering at transplanckian energies¹

$$\sqrt{s} \gg M_P \equiv \sqrt{\frac{\hbar}{G}}, \quad \text{i.e.} \quad \frac{Gs}{\hbar} \gg 1.$$
 (1.1)

Among the initial aims of those investigations we would like to mention:

- Understanding how unitarity bounds on partial waves, which are violated at tree level, get restored by loop corrections. This question can be asked both in QFT and in a string-theory context.
- Connecting the high-energy, fixed-angle behavior of string scattering amplitudes to modifications of gravity at short distance.
- Studying regimes in which the process is expected, classically, to lead to black-hole formation and, quantum mechanically, to subsequent black-hole evaporation. The construction of a unitary *S*-matrix in such a context would guarantee that information is preserved thus solving Hawking's famous paradox [18,19]. Alternatively, find a breakdown of unitarity.

¹ For simplicity in this introduction we work in four spacetime dimension. We remind the reader that the very concept of a transplanckian energy is quantum-mechanical: classical GR has no intrinsic mass or length scale (G^{-1} is a mass per unit length in units where c = 1 which we use throughout unless explicitly stated).

For scattering of elementary particles or strings which are much lighter than the Planck mass, the transplanckian regime (1.1) constitutes an ultra-relativistic limit, and is essential for dealing with the above questions. First, because gravity becomes the dominant force only at sufficiently high energies and this allows one to obtain (almost) theory-independent results. Second, because high energies are necessary (although, as we shall see, not always sufficient) in order to probe short distances. And, finally, because one would like to deal with black holes whose radius is (much) larger than the Planck length $\ell_P = \sqrt{G\hbar}$ in order to apply to them the semiclassical approximations used in [18].

A high-energy limit naturally leads to a semiclassical approximation to gravitational scattering, simplifying considerably all the calculations. In non-relativistic quantum mechanics the semiclassical limit goes under the name of WKB approximation (see e.g. [20,21]). In a more general relativistic framework, it is associated with the so-called eikonal approximation, the basic tool to be described in this review article. The early applications of the eikonal approximation have met with considerable success in some regimes, including those in which point-particles and extended objects (strings, branes) behave differently [8,10,13,15–17,22–24]. Parts of this review will be dedicated to these aspects of gravitational scattering. By comparison, less progress was made on the regime where gravitational collapse is classically expected to occur. In particular whether and how unitarity is preserved is still an unanswered question. We shall therefore make a less detailed account of those aspects of the problem.

In the last few years, in the wake of the first direct observation of gravitational waves by ground-based interferometers [25,26], the same semiclassical approximation turned out to be a useful tool for studying collisions of very heavy objects, provided they remain at sufficiently large distances with respect to their size. Typical candidates are, of course, astrophysical black holes which, besides being extremely compact, are characterized by the same quantum numbers as those of an elementary particle, i.e. mass and spin (plus possibly some conserved charges) as long as the process is only sensitive to the geometry outside their horizon.

Since semiclassical black holes, and a fortiori astrophysical ones, are much heavier than M_P , one does not have to consider, for them, a highly relativistic regime in order to justify the approximation (1.1) and, in particular, the eikonal approach. And for current physical applications it is often rather the opposite, i.e. a nonrelativistic approximation is viable. For black holes such as those featuring in the events observed by LIGO/Virgo [25], with masses of about $m_i \sim 30M_{\odot}$ (with $M_{\odot} \simeq 2 \times 10^{30}$ kg the solar mass), the ratio between the Schwarzschild radii $\frac{2Gm}{c^2}$ and the reduced Compton wavelength $\frac{h}{m_c}$ obeys

$$\frac{2Gm^2}{\hbar c} \simeq 10^{79} \gg 1\,,\tag{1.2}$$

Therefore, for such systems, the classical limit is obviously an excellent approximation. What is less obvious is the use of a point-particle approximation. This is expected to be good for black hole collisions at distances much larger than their Schwarzschild radii while for neutron stars the detailed interior of the star matters and has to be incorporated through some non-minimal coupling to gravity of the effective point particle.

It thus comes as no surprise that the traditional techniques for studying the collision and merger of compact astrophysical objects have been based on classical GR. Among the most important ones are those using numerical relativity (as pioneered in [27] and reviewed e.g. in [28]), the self-force approach [29], the Post-Newtonian expansion (see e.g. the nice review in [30]), and the Effective-One-Body (EOB) framework [31,32], although the latter does make use of quantum mechanical concepts for the determination of effective potential's parameters.

More recently, however, the idea that scattering amplitudes—the bread and butter of quantum field theory calculations—could be recycled for use in the physics of compact astrophysical systems has made its way in the scientific community [33–42]. Furthermore, as stressed by Damour in [35], even the ultra-relativistic high-energy regime we have already mentioned could provide very valuable information on the classical GR problem. As is also immediately clear from (1.2), in an amplitude-based approach, one is not dealing here with straight perturbation theory, which can be thought as an expansion for "small-G". One should therefore identify the appropriate way of resumming certain infinite sets of diagrams. The essential, simplifying feature, which is shared by WKB and eikonal methods, is that in the semiclassical limit the amplitude is controlled by a large phase, typically representing a large, classical action in units of \hbar , in which the coupling constant sits in the exponent. To obtain this form, it is crucial that particular contributions of entire classes of loop diagrams "exponentiate", i.e. appear to the appropriate power and with the right combinatoric factor to all loop orders. Sometimes this can be explicitly checked, at least at the level under scrutiny, sometimes it can be justified by other methods (like in the world-line approaches), and sometimes will just be assumed. The validity of the eikonal approximation thus needs to be checked on a case-by-case basis, but, as we shall discuss, it has been successfully applied to semiclassical scattering for both massive and massless scattering [15,43–47], in various spacetime dimensions [48,49]. Amplitude techniques, including also the eikonal, have been applied the case in which the colliding objects can be subject to tidal deformations [50–57] or carry spin [58–84]. Tidal effects [85–95] may in particular provide clues on the equation of states of neutron stars [96], on the nature of black holes [97] and on possible exotic astrophysical objects [98–100].

A very important feature of the semiclassical/eikonal approximation is thus that it allows one to take easily the classical limit itself, usually through a saddle-point approximation. On the one hand agreement with classical expectations provides a check of the quantum result. On the other hand, and perhaps more interestingly, the eikonal approximation offers a new tool for computing classical observables. The perturbative nature of such calculations is recovered by looking for an expansion of the exponent, after the resummation. In this way, the eikonal exponentiation provides a simple way

_

to recover the classical limit by matching to the Post-Minkowskian (PM) regime, in which the colliding objects remain sufficiently far apart and interact weakly, or, equivalently, undergo sufficiently small deflections,

$$\frac{G\sqrt{s}}{b} \ll 1. \tag{1.3}$$

For reference, to obtain a rough estimate of this parameter, we may consider again the situation of a merger event in the early inspiral phase, as measurable by LIGO/Virgo's detectors, where the typical relative separation r_0 between the two objects is such that

$$\frac{Gm_i}{r_0} \simeq \frac{1}{6} \,. \tag{1.4}$$

This small but non-negligible ratio motivates us to investigate higher orders in the PM expansion, which translate in higher-order approximations of the eikonal phase [47,48,101–112]. In this way, the eikonal exponentiation provides an alternative approach for computing classical gravitational observables from scattering amplitudes, complementing various types of EFT setups [37,49,113–115], the KMOC framework [116–122] as well as PM worldline EFT methods [55,56,123–140] (see [141] for a comparison between these two approaches).

As already mentioned the gravitational eikonal approximation can be justified in a large variety of situations and this review will try to cover as many of them as possible indicating, in each case, both the achievements and the challenges lying ahead. One such challenge is represented by inelastic processes, such as gravitational radiation or internal excitations (as it happens in string theory), in which eikonal phase becomes an operator [110,112,119,142-148]. We organized the material according to the order in *G* at which the eikonal phase (or operator) is computed. We believe that by starting from tree-level and then working our way up to two loops, we could organize the material in a pedagogical style and make the presentation as accessible as possible. At each loop level we will consider the case of pure Einstein gravity and of some supersymmetric extensions of it. At tree level and one loop, we will discuss the case of string theory as well. In the same spirit, we also describe first the two-to-two amplitude and only then turn to radiative processes and to radiation reaction.

A very interesting development of the eikonal methods that we shall not cover in this report concerns their application to scattering processes taking place in asymptotically AdS spacetimes, and their connection to holographic CFTs. This analysis has been initiated in [149–153] focusing on the high-energy scattering of light states as was done in the eighties in flat space. Through the AdS/CFT duality, the eikonal approach highlighted the existence of bounds on particular couplings in holographic CFTs, see for instance [154–157]. The eikonal regime has been also used in the analysis of the CFT correlators with heavy operators [158–163] and applied to the study of tidal excitations in AdS [164], and of higher point correlators [165]. Another recent development which we will not discuss is the use of the eikonal approach in the context of the so-called celestial CFT [166].

With this general plan in mind, let us discuss how the report is organized. In the remainder of the present section, we shall spell out our kinematics conventions, review the definition of partial waves and discuss qualitative features of different regimes of gravitational scattering. In Section 2, we provide a self-contained, elementary treatment of the leading eikonal resummation for a gravitational $2 \rightarrow 2$ process involving massless scalar objects. We show how to calculate the leading eikonal phase $2\delta_0$, prove that it exponentiates and how it directly gives the deflection angle. This very simple example is mainly meant to whet the reader's appetite and lends itself to two quite different derivations that lead to the same result. One is based on the amplitude for a single graviton exchange and the other one is based on solving the geodesic equation in the Aichelburg–Sexl metric. We also illustrate how partial-wave unitarity, which is violated at finite loop order, is recovered via the resummation.

In Section 3, we calculate the leading-order eikonal phase $2\delta_0$ in several different field theories, ranging from minimally coupled massive scalars to graviton scattering off a massive scalar, dilaton gravity, $\mathcal{N} = 8$ supergravity, higherderivative corrections of GR and scattering of spinning objects. We then turn to tree-level string amplitudes, for which we mainly focus on string-brane scattering (the analog of a probe-limit calculation) and also introduce the eikonal operator, which accounts for transitions between different excited modes of the string. Section 4 is instead devoted to one-loop calculations, which allow us on the one hand to check the first constraint arising from the exponentiation of the tree-level result and on the other hand to calculate the first sub-leading correction to the eikonal phase $2\delta_1$. Like for the previous one, we discuss both fields and strings. In Section 5, we encounter for the first time $2 \rightarrow 3$ amplitudes, which enter the discussion of the unitarity cuts of the $2 \rightarrow 2$ amplitudes. We review in detail this point, focusing on the interplay between momentum-space convolutions and their impact-parameter Fourier transforms. This also serves as an occasion to anticipate the calculation of the imaginary part of the two-loop results, which is instead presented in Section 6 for $\mathcal{N} = 8$ supergravity and for GR. There, we also comment more in detail on the definition of the impact parameter and on the connection between eikonal phase, radial action and phase shifts.

The remainder of the report is devoted to the inclusion of radiation in the final state. This is done first following soft theorems as in Section 7, restricting one's attention to low-frequency radiation, and then, to leading order in the PM expansion, but capturing the full spectrum, by exponentiating the tree-level $2 \rightarrow 3$ amplitude as in Section 8. We summarize our conclusions and prospects for the future in Section 9.

Several appendices are included in order to make the material sufficiently self-contained. Appendix A collects conventions about Feynman rules and useful results concerning Fourier transforms from momentum to impact-parameter

space. In Appendix B we present the calculation of the deflection angle in the probe limit for several theories. A brief Appendix C collects useful identities involving the completely antisymmetric tensor. Appendix D contains material that serves as background for the string-theory content of Sections 3, 4. In Appendix E we summarize a list of kinematic relations that apply to the $2 \rightarrow 3$ process involving graviton emissions and include expressions for the waveforms that complement those presented in Section 8. Finally, in Appendix F we illustrate how the on-shell metric fluctuation is linked to the asymptotic waveform.

1.2. Kinematics and conventions

Before proceeding further, let us spell out here for later convenience the main conventions that will be employed in the rest of the review.

The basic object of study of this work is the collision of two energetic objects with masses m_1 and m_2 . These define the initial states of a scattering process. For convenience, all external momentum vectors will be regarded as outgoing, so that $-p_1$ and $-p_2$ represent the physical momenta for such incoming particles. We work with the mostly-plus signature for the metric,

$$\eta_{\mu\nu} = \text{diag}(-1, 1, \dots, 1), \qquad \mu, \nu = 0, 1, \dots, D-1,$$
(1.5)

so that

$$p_1^2 = -m_1^2, \qquad p_2^2 = -m_2^2.$$
 (1.6)

When the two incoming states have nonzero masses, we define their velocities according to²

$$p_1^{\mu} = -m_1 v_1^{\mu}, \qquad p_2^{\mu} = -m_2 v_2^{\mu}, \qquad v_1^2 = -1, \qquad v_2^2 = -1.$$
 (1.7)

Then their relative speed is sized by the invariant

$$\sigma = -\frac{p_1 \cdot p_2}{m_1 m_2} = -v_1 \cdot v_2 = \frac{1}{\sqrt{1 - v^2}},$$
(1.8)

where v is the velocity of either particle as seen from the rest frame of the other one. Of course, $\sigma \rightarrow 1^+$ in the near-static limit, while $\sigma \rightarrow \infty$ in the ultra-relativistic limit. It is also convenient to introduce the "dual" velocities

$$\check{v}_{1}^{\mu} = \frac{\sigma v_{2}^{\mu} - v_{1}^{\mu}}{\sigma^{2} - 1}, \qquad \check{v}_{2}^{\mu} = \frac{\sigma v_{1}^{\mu} - v_{2}^{\mu}}{\sigma^{2} - 1}, \qquad v_{1} \cdot \check{v}_{1} = v_{2} \cdot \check{v}_{2} = -1, \qquad v_{1} \cdot \check{v}_{2} = v_{2} \cdot \check{v}_{1} = 0.$$

$$(1.9)$$

In particular, these vectors allow one to conveniently decompose any given vector ξ^{μ} in terms of its longitudinal and transverse components according to

$$\xi^{\mu} = \xi_{\parallel 1} \check{v}_{1}^{\mu} + \xi_{\parallel 2} \check{v}_{2}^{\mu} + \xi_{\perp}^{\mu} , \qquad (1.10)$$

with

$$\xi_{\parallel 1} = -\xi \cdot v_1, \qquad \xi_{\parallel 2} = -\xi \cdot v_2, \qquad \xi_{\perp} \cdot v_1 = 0, \qquad \xi_{\perp} \cdot v_2 = 0.$$
(1.11)

In a center-of-mass frame, the initial momenta take the form

$$-p_1 = (E_1, \vec{p}), \quad -p_2 = (E_2, -\vec{p}), \tag{1.12}$$

where \vec{p} is a (D - 1)-dimensional spatial vector. Let us collect here a few useful relations that link the relativistic factor σ and the masses $m_{1,2}$ to the total energy $E = E_1 + E_2$, the single-particle energies $E_{1,2}$ and the absolute value $p = |\vec{p}|$ of the spatial momentum in such a frame,

$$Ep = m_1 m_2 \sqrt{\sigma^2 - 1},$$
 (1.13)

$$E = E_1 + E_2 = \sqrt{m_1^2 + 2m_1m_2\sigma + m_2^2}, \qquad (1.14)$$

$$E_1 = \frac{m_1}{E} (m_1 + \sigma m_2), \tag{1.15}$$

$$E_2 = \frac{m_2}{E} (m_2 + \sigma m_1). \tag{1.16}$$

We define the scattering amplitude for the process $\alpha \rightarrow \beta$ in the standard way by decomposing the *S* operator according to S = 1 + iT and by letting

$$\langle \beta | T | \alpha \rangle = (2\pi)^D \delta^{(D)}(P_\alpha + P_\beta) \mathcal{A}_{\alpha \to \beta} , \qquad (1.17)$$

² While $p_{1,2}^{\mu}$ are past-directed to comply with the all-outgoing convention for external momenta, $v_{1,2}^{\mu}$ are future-directed, hence the minus signs in (1.7) and also in (1.28) below.

where the momentum conserving delta function takes into account that all momenta are formally outgoing. We normalize single-particle momentum eigenstates with mass *m* according to the Lorentz-invariant convention

$$2\pi\theta(p^{0})\delta(p^{2}+m^{2})\langle p|-p'\rangle = (2\pi)^{D}\delta^{(D)}(p+p'), \qquad (1.18)$$

which is equivalent to the more common normalization³

$$\langle p| - p' \rangle = 2\sqrt{|\vec{p}|^2 + m^2 (2\pi)^{D-1} \delta^{(D-1)}(\vec{p} + \vec{p}')}.$$
(1.19)

A case that we will often consider in the following is that of an *elastic* $2 \rightarrow 2$ scattering depicted in Fig. 1. We shall denote the corresponding amplitude simply by A and label the incoming particles by 1, 2 and the outgoing ones by 3, 4, in such a way that

$$p_1^2 = p_4^2 = -m_1^2, \qquad p_2^2 = p_3^2 = -m_2^2.$$
 (1.20)

As already mentioned, all external momenta are regarded as outgoing so that they satisfy the conservation condition

$$p_1 + p_2 + p_3 + p_4 = 0 \tag{1.21}$$

and thus define the standard Mandelstam variables via

$$s = -(p_1 + p_2)^2, \quad t = -(p_1 + p_4)^2, \quad u = -(p_1 + p_3)^2.$$
 (1.22)

As usual, *s* is linked to the total energy in the center-of-mass frame *E* by $E = \sqrt{s}$. We also define the momentum transfer,

$$q = p_1 + p_4 = -p_2 - p_3, (1.23)$$

which is related to the Mandelstam invariant t by $t = -q^2$. Of course, u can be written in terms of s and t using momentum conservation and the mass-shell conditions,

$$s + t + u = 2(m_1^2 + m_2^2).$$
(1.24)

More explicitly, following the notation in (1.17),

$$\langle p_4, p_3 | T | - p_2, -p_1 \rangle = (2\pi)^D \delta^{(D)}(p_1 + p_2 + p_3 + p_4) \mathcal{A}(s, t).$$
 (1.25)

Let us note, for later convenience, that one can also factor the overall momentum-conserving delta function by recasting the S-matrix element as follows,

$$2\pi\theta(p_4^0)\delta(p_4^2+m_1^2)2\pi\theta(p_3^0)\delta(p_3^2+m_2^2)\langle p_4, p_3|S|-p_2, -p_1\rangle = (2\pi)^D\delta^{(D)}(p_1+p_2+p_3+p_4)S$$
(1.26)

where, using $\delta^{(D)}(p_1 + p_4)\delta^{(D)}(p_2 + p_3) = \delta^{(D)}(p_1 + p_2 + p_3 + p_4)\delta^{(D)}(q)$ for the disconnected piece, with *q* as in (1.23),⁴

$$S = S(p_1, p_2; q) = (2\pi)^D \delta^{(D)}(q) + 2\pi \delta(2p_1 \cdot q - q^2) 2\pi \delta(2p_2 \cdot q + q^2) i\mathcal{A}(s, -q^2).$$
(1.27)

It can be convenient to introduce "average" momenta \bar{p}_i^{μ} and velocities u_i^{μ} according to [167]

$$-p_{1}^{\mu} = \bar{p}_{1}^{\mu} - \frac{1}{2}q^{\mu}, \qquad p_{4}^{\mu} = \bar{p}_{1}^{\mu} + \frac{1}{2}q^{\mu}, -p_{2}^{\mu} = \bar{p}_{2}^{\mu} + \frac{1}{2}q^{\mu}, \qquad p_{3}^{\mu} = \bar{p}_{2}^{\mu} - \frac{1}{2}q^{\mu},$$
(1.28)

and, for i = 1, 2,

$$\bar{p}_i^{\mu} = \bar{m}_i u_i^{\mu}, \qquad u_i^2 = -1.$$
(1.29)

In this way the mass-shell conditions turn into the following relations: by $p_1^2 - p_4^2 = 0$ and $p_2^2 - p_3^2 = 0$,

$$\bar{p}_i \cdot q = 0, \qquad u_i \cdot q = 0, \tag{1.30}$$

for i = 1, 2, while by $p_1^2 + p_4^2 = -2m_1^2$ and $p_2^2 + p_3^2 = -2m_2^2$,

$$\bar{m}_1^2 = m_1^2 + \frac{1}{4}q^2, \qquad \bar{m}_2^2 = m_2^2 + \frac{1}{4}q^2.$$
(1.31)

The Fourier transform of the S-matrix element S defined in (1.27) is then

$$\int \frac{d^{D}q}{(2\pi)^{D}} \mathcal{S} e^{ib \cdot q} = 1 + i \int \frac{d^{D}q}{(2\pi)^{D}} e^{ib \cdot q} 2\pi \delta(2\bar{p}_{1} \cdot q) 2\pi \delta(2\bar{p}_{2} \cdot q) \mathcal{A}(s, -q^{2}) = 1 + i \operatorname{FT}[\mathcal{A}](s, b).$$
(1.32)

³ One can multiply both sides of (1.18) by $2\pi\theta(-p'^0)\delta(p'^2+m^2)$ to make it manifestly symmetric.

⁴ For simplicity, we leave the θ functions implicit since they are irrelevant for sufficiently small q, given the fact that $p_4^0 = -p_1^0 + q^0$ and $-p_1^0$ is positive.



Fig. 1. To the left, a diagrammatic picture of the elastic $2 \rightarrow 2$ amplitude A. To the right, a cartoon of two-body scattering in the center-of-mass frame.

Going to a center-of-mass frame, one has

$$-p_1 = (E_1, \vec{p}), \quad -p_2 = (E_2, -\vec{p}), \quad p_3 = (E_2, -\vec{p}'), \quad p_4 = (E_1, \vec{p}'), \quad (1.33)$$

where now \vec{p} and \vec{p}' are (D-1)-dimensional space vectors with $|\vec{p}| = |\vec{p}'| \equiv p$. In this frame, E_1 , E_2 , p obey (1.13), (1.14), (1.15), (1.16), and

$$q^{\mu} = (0, \vec{q}) = (0, \vec{p}' - \vec{p}), \qquad q^2 = |\vec{q}|^2 = 2p^2(1 - \hat{p} \cdot \hat{p}'), \tag{1.34}$$

where $\hat{p} = \vec{p}/p$ and similarly for \hat{p}' . In fact, these properties correspond to further factorized forms of the *S*-matrix element (1.27),

$$S = 2\pi \,\delta(q^0) \left[(2\pi)^{D-1} \delta^{(D-1)}(\vec{q}\,) + i2\pi \,\delta(p - |\vec{p} + \vec{q}\,|) \frac{\mathcal{A}(s, -|\vec{q}\,|^2)}{4Ep} \right]$$
(1.35)

$$= 2\pi \,\delta(q^0)(2\pi)^{D-1} \frac{\delta(p-p')}{p^{D-2}} \left[\delta^{(D-2)}(\hat{p}, \hat{p}') + i \left(\frac{p}{2\pi}\right)^{D-2} \frac{\mathcal{A}(s, -2p^2(1-\hat{p}\cdot\hat{p}'))}{4Ep} \right]$$
(1.36)

where in the second line $\vec{p}' = \vec{p} + \vec{q}$ and $\delta^{(D-2)}(\hat{p}, \hat{p}')$ is the invariant delta function on the (D-2)-sphere (for instance $\delta^{(2)}(\hat{p}, \hat{p}') = \delta(\theta - \theta')\delta(\phi - \phi')/\sin\theta$ in D = 4 with standard spherical coordinates, which obviously implies $\int_0^{2\pi} d\phi \int_0^{\pi} d\theta \sin \theta \, \delta^{(2)}(\hat{p}, \hat{p}') = 1$).

Aligning the "average momenta" so that $\bar{p}_1^{\mu} = (\bar{E}_1, 0, \dots, 0, \bar{p})$ and $\bar{p}_2^{\mu} = (\bar{E}_2, 0, \dots, 0, -\bar{p})$, one obtains

$$-p_{1} = \left(\bar{E}_{1}, -\frac{1}{2}\mathbf{q}, \bar{p}\right), \qquad p_{4} = \left(\bar{E}_{1}, \frac{1}{2}\mathbf{q}, \bar{p}\right), -p_{2} = \left(\bar{E}_{2}, \frac{1}{2}\mathbf{q}, -\bar{p}\right), \qquad p_{3} = \left(\bar{E}_{2}, -\frac{1}{2}\mathbf{q}, -\bar{p}\right),$$
(1.37)

and in this way

$$q = (0, \mathbf{q}, 0),$$
 (1.38)

while the energies $\bar{E}_{1,2}$ can be obtained from the mass-shell conditions:

$$\bar{E}_{1,2}^2 = \bar{p}^2 + \frac{q^2}{4} + m_{1,2}^2 .$$
(1.39)

As is clear from (1.37), this parametrization corresponds to the so-called Breit (or brick-wall) frame, where each particle bounces back in the transverse directions and moves unperturbed along the longitudinal directions.

Another amplitude that we will employ in the following, especially when dealing with emitted radiation, is the $2 \rightarrow 3$ amplitude $\mathcal{A}^{(5)}$ describing the collision of two energetic particles with incoming momenta $-p_1$, $-p_2$ and outgoing momenta k_1 , k_2 with the additional emission of a single graviton with momentum k, as depicted in Fig. 2 so that

$$p_1^2 = p_4^2 = -m_1^2, \qquad p_2^2 = p_3^2 = -m_2^2, \qquad k^2 = 0$$
 (1.40)

and

$$p_1 + p_2 + k_1 + k_2 + k = 0. (1.41)$$

In this case, we find it convenient to define two "momentum transfers"

$$q_1 = p_1 + k_1, \qquad q_2 = p_2 + k_2, \qquad q_1 + q_2 + k = 0.$$
 (1.42)

Employing the velocities (1.7), the amplitude will depend on the invariants

$$\sigma = -v_1 \cdot v_2 \ge 1, \qquad \omega_1 = -v_1 \cdot k \ge 0, \qquad \omega_2 = -v_2 \cdot k \ge 0$$
(1.43)



Fig. 2. To the left, a diagrammatic picture of the inelastic 2 \rightarrow 3 amplitude $A^{(5)}$. To the right, a cartoon of two-body scattering with emission of gravitational radiation.

where ω_j is the graviton's frequency as seen in the rest frame of particle *j* (for *j* = 1, 2). Using (1.10), we can decompose k^{μ} as

$$k^{\mu} = \omega_1 \check{v}_1^{\mu} + \omega_2 \check{v}_2^{\mu} + k_{\perp}^{\mu} \,, \tag{1.44}$$

and $k^2 = 0$ implies

$$\frac{\mathcal{P}}{\sigma^2 - 1} = k_{\perp}^2 \ge 0, \qquad \mathcal{P} = -\omega_1^2 + 2\omega_1\omega_2\sigma - \omega_2^2.$$
(1.45)

As we shall discuss, the classical limit is obtained by considering q_1 , q_2 , k simultaneously small compared with the incoming particles' momenta. Therefore the exact mass-shell conditions for the massive states

$$2p_1 \cdot q_1 = q_1^2, \qquad 2p_2 \cdot q_2 = q_2^2, \tag{1.46}$$

read to leading order⁵

$$p_1 \cdot q_1 \approx 0, \qquad p_2 \cdot q_2 \approx 0.$$
 (1.47)

In this way, we find

$$q_1^{\mu} \approx -\omega_2 \check{v}_2^{\mu} + q_{1\perp}^{\mu}, \qquad q_2^{\mu} \approx -\omega_1 \check{v}_1^{\mu} + q_{2\perp}^{\mu}, \tag{1.48}$$

as follows from the last two equations in (1.9), and

$$q_1^2 \approx \frac{\omega_2^2}{\sigma^2 - 1} + q_{1\perp}^2, \qquad q_2^2 \approx \frac{\omega_1^2}{\sigma^2 - 1} + q_{2\perp}^2.$$
 (1.49)

This follows immediately from $\check{v}_1^2 = \check{v}_2^2 = \frac{1}{\sigma^2 - 1}$ and from the last equation in (1.11) for $\xi = q_i$. In a $2 \rightarrow 2 + N$ process with emission of N gravitons, the conservation condition would read

$$p_1 + p_2 + k_1 + k_2 + P = 0, (1.50)$$

with P the sum of all graviton momenta. Squaring this relation one finds

$$\sigma' \equiv -\frac{k_1 \cdot k_2}{m_1 m_2} = \sigma - \frac{2P \cdot (p_1 + p_2) + P^2}{2m_1 m_2}, \qquad (1.51)$$

where $P \cdot (p_1 + p_2)/\sqrt{s} = E_{rad}$ is the energy lost to graviton emissions in the (incoming) center-of-mass frame, while $-P^2 > 0$ represents the gravitons' invariant mass squared. Clearly, by energy conservation $E_{rad} \le \sqrt{s}$, and thus

$$P \cdot (p_1 + p_2) + P^2 = 2\sqrt{s} E_{\text{rad}} - (E_{\text{rad}})^2 + \vec{P}_{\text{rad}}^2 \ge (E_{\text{rad}})^2 + \vec{P}_{\text{rad}}^2 \ge 0.$$
(1.52)

Therefore (1.51) shows that the massive particles' relative velocity always decreases as a result of the emissions, $\sigma' \leq \sigma$. We shall return to this point in Section 8 and in the outlook.

1.3. Partial waves, unitarity, phase shifts and the eikonal phase

For later convenience, let us recall how the $2 \rightarrow 2$ scattering amplitude can be decomposed into partial waves by expressing it in terms of angular momentum eigenstates, and how unitarity holds for each partial wave. For simplicity we will discuss the procedure in D = 4 although generalization to arbitrary D is straightforward (see for instance [7,168]). We shall also focus on scalar massive particles, see e.g. [169,170] for generalizations to the spinning case.

⁵ We could introduce "average" momenta that are exactly orthogonal to q_1 , q_2 respectively, analogous to $\bar{p}_{1,2}^{\mu}$ introduced in (1.29), but we will not need the relations (1.47) beyond leading order.

Let us start from the S-matrix element obtained in (1.36) by factoring out the motion of the center of mass, which we may regard as the definition of the "reduced" S and T operators in this frame,

$$\langle \hat{p}'|\mathcal{S}|\hat{p}\rangle = \delta^{(2)}(\hat{p},\hat{p}') + i\langle \hat{p}'|T|\hat{p}\rangle, \qquad \langle \hat{p}'|T|\hat{p}\rangle = \left(\frac{p}{2\pi}\right)^2 \frac{\mathcal{A}(s,-q^2)}{4Ep},$$
(1.53)

where

$$\langle \hat{p}' | \hat{p} \rangle = \delta^{(2)}(\hat{p}, \hat{p}'), \qquad q = 2p \sin \frac{\theta}{2}, \qquad \hat{p} \cdot \hat{p}' \equiv \cos \theta.$$
 (1.54)

It is convenient to go to a basis with well-defined properties under rotations by means of the spherical harmonics $Y_{im}(\hat{p})$,

$$|j,m\rangle = \int d\Omega(\hat{p}) Y_{jm}(\hat{p})|\hat{p}\rangle, \qquad \langle j',m'|j,m\rangle = \delta_{jj'}\delta_{mm'}.$$
(1.55)

For concreteness, aligning \hat{p}' along the *z* axis, the three components of \hat{p} can be expressed in terms of θ and ϕ as $\hat{p} = (\cos \phi \sin \theta, \sin \phi \sin \theta, \cos \theta)$ and $d\Omega(\hat{p}) = \sin \theta \, d\theta \, d\phi$ with $0 \le \phi < 2\pi$ and $0 \le \theta \le \pi$. The states $|j, m\rangle$ in (1.55) are eigenstates of the total angular momentum J^2 and of its component along a given axis, say J_z ,

$$J^{2}|j,m\rangle = \hbar^{2}j(j+1)|j,m\rangle, \qquad J_{z}|j,m\rangle = \hbar m|j,m\rangle.$$
(1.56)

Since we are dealing with collisions of scalar particles, the invariance of S and T under rotations implies via the Wigner–Eckart theorem that their matrix elements take the following diagonal form,

$$\langle j', m' | \mathcal{S} | j, m \rangle = s_j(s) \delta_{jj'} \delta_{mm'}, \qquad \langle j', m' | T | j, m \rangle = 2f_j(s) \delta_{jj'} \delta_{mm'}, \qquad (1.57)$$

where $f_i(s)$ are suitable, *m*-independent functions, which we may identify as partial waves, and of course

$$s_j(s) = 1 + 2if_j(s)$$
. (1.58)

It is also common to express $f_i(s)$ in terms of the so-called phase shifts $\delta_i(s)$, according to

$$1 + 2if_j(s) = e^{2i\delta_j(s)}, \qquad f_j(s) = \frac{e^{2i\delta_j(s)} - 1}{2i} = e^{i\delta_j(s)} \sin \delta_j(s).$$
(1.59)

Therefore, using (1.55) and the definition (1.57) of $f_i(s)$ we obtain

$$\langle \hat{p}' | T | \hat{p} \rangle = \sum_{j=0}^{\infty} \frac{2j+1}{2\pi} P_j(\hat{p} \cdot \hat{p}') f_j(s), \qquad (1.60)$$

after recalling the addition theorem that links the spherical harmonics to the Legendre polynomials P_j , (see for instance DLMF (14.30.09))

$$\sum_{m=-j}^{j} Y_{jm}^{*}(\hat{p}) Y_{jm}(\hat{p}') = \frac{2j+1}{4\pi} P_{j}(\hat{p} \cdot \hat{p}').$$
(1.61)

Comparing with (1.53), we see that the original amplitude can be decomposed as follows,

$$\mathcal{A}(s, -q^2) = \frac{8\pi E}{p} \sum_{j=0}^{\infty} (2j+1) f_j(s) P_j(\cos\theta), \qquad (1.62)$$

$$f_{j}(s) = \frac{p}{16\pi E} \int_{-1}^{+1} d(\cos\theta) P_{j}(\cos\theta) \mathcal{A}(s, -q^{2}) \Big|_{q=2p\sin\frac{\theta}{2}},$$
(1.63)

where the latter relation can be obtained from the former using the orthogonality properties of the Legendre polynomials,

$$\int_{-1}^{+1} dx P_j(x) P_k(x) = \frac{2}{2j+1} \delta_{jk}, \qquad P_j(1) = 1.$$
(1.64)

The virtue of the partial waves $f_j(s)$ is that the full non-linear unitarity condition $S^{\dagger}S = 1$, which takes the following form for the *T*-matrix,

$$-i(T - T^{\dagger}) = T^{\dagger}T, \qquad (1.65)$$

becomes diagonal in *j* and its elastic contribution is particularly simple. Indeed, by taking the expectation value of (1.65) on $\langle j, m | \cdots | j, m \rangle$ and inserting a complete set of states on the right-hand side, by (1.57) one finds that the sum over intermediate two-body states simplifies to a single term, and the result is

$$\operatorname{Im} f_j(s,j) = |f_j(s)|^2 + \operatorname{inelastic}, \qquad (1.66)$$

where each (*j*-dependent) inelastic contribution is non negative. This implies the bound

_ _ _ _ . . .

$$Im f_{i}(s,j) \ge |f_{i}(s)|^{2}$$
(1.67)

or the (weaker) inequality

$$|f_i(s)| \le 1$$
. (1.68)

Indeed, (1.66) can be obtained starting directly from $S^{\dagger}S = 1$ and recalling the definition of the phase shifts (1.59),

$$|s_i(s)|^2 + \text{inelastic} = 1, \qquad e^{-2 \operatorname{Im} 2\delta_j(s)} + \text{inelastic} = 1$$
(1.69)

and thus one obtains the following bounds equivalent to (1.67),

$$|s_j(s)| \le 1$$
, $\ln 2\delta_j(s) \ge 0$ (1.70)

with the equality signs holding true for perfectly elastic scattering. Note that, unlike the Froissart bound, which depends on the absence of massless particles, the bound on partial waves should also hold in the presence of long range forces, in particular for gravity.

The eikonal phase, which constitutes the focus of this report, is closely related to the phase shifts discussed above. The main difference is that the eikonal is introduced by performing the Fourier transform of the S-matrix elements rather than their partial wave decomposition

$$\tilde{\mathcal{S}}(s,b) = 1 + i\tilde{\mathcal{A}}(s,b) = 1 + \int \frac{d^{D-2}q}{(2\pi)^{D-2}} e^{ib \cdot q} \frac{i\mathcal{A}(s,-q^2)}{4Ep} , \qquad (1.71)$$

where we are assuming the kinematics of the Breit frame (1.37). The technical advantage is that one does not have to deal with Legendre (or Gegenbauer for D > 4) polynomials, while the drawback is that the exact diagonal form in (1.57) is lost as we shall see in Eq. (5.26) below. However in the classical limit there is an approximate diagonalization and so, in analogy with (1.59), it is convenient to introduce the eikonal phase $2\delta(s, b)$

$$\tilde{\mathcal{S}}(s,b) = 1 + i\tilde{\mathcal{A}}(s,b) = (1+\cdots)e^{2i\delta(s,b)},$$
(1.72)

where the eikonal 2δ scales as $1/\hbar$ while the dots in the prefactor stand for quantum corrections. The impact parameter *b*, unlike *j*, is a continuous variable directly related to a classical quantity. The precise relation between $2\delta_j(s)$ and $2\delta(s, b)$ will be discussed in Section 2.2 and more generally in Section 6.1, where we will also elaborate on the relation between (1.71) and the covariant definition of the Fourier Transform introduced in (A.31).

1.4. Regimes of $2 \rightarrow 2$ scattering

As intuitively clear, the classical impulse Q, i.e. the total momentum transfer during a classical $2 \rightarrow 2$ collision, emerges from the quantum description via the exchange of a very large number of gravitons, each carrying a momentum of order q. The resummation mechanism described in the next sections will make this intuitive picture precise, effectively capturing an infinite number of graviton exchanges and thus predicting the classical limit of the amplitude from its conventional expansion in perturbation theory. Expressing the amplitude as a function of the collision's impact parameter, this resummation leads to the exponential of the original perturbative amplitude. Of course, due to the multiple interactions, the transferred momentum Q exchanged in the classical process is much larger than the perturbative momentum transfer q,

$$Q \gg q \,. \tag{1.73}$$

In particular, the ratio Q/q goes to infinity in the classical limit, i.e. in the formal $\hbar \rightarrow 0$ limit.

As far as the classical process is concerned, we are particularly interested in the regime where Q is small in comparison with the center-of-mass energy

$$s \gg Q^2 \,, \tag{1.74}$$

while the masses are kept arbitrary so as to explore various limits. A first regime is when the total center of mass energy is of the order of the masses

s,
$$m_1^2$$
, $m_2^2 \gg Q^2$ black-hole scattering, (1.75)

which we dub as "black hole scattering". Indeed, this can be seen as a relativistic, open-trajectory analogue of a black-hole binary system tracing out a non-relativistic, closed trajectory motion during the inspiral phase. From the conceptual point of view, it is interesting to take the extreme case where the kinetic energy is much larger than the (large) rest mass energy of the external states

$$s \gg m_i^2 \gg Q^2$$
 high-energy scattering. (1.76)

This ultrarelativistic limit is in principle different from the case where the masses are set to zero and one describes the scattering of two shock waves

$$s \gg Q^2$$
, $m_i^2 = 0$ shock-wave scattering, (1.77)

as will become clear especially in Section 7.3. It will be interesting to compare the two cases above. In particular we will see that for some quantities, such as the deflection angle up to 3PM, the transition from $m_i^2 \gg Q^2$ to $m_i^2 \rightarrow 0$ is smooth. while others quantities have rather different behaviors in these two regimes. Finally another interesting case is that of an elastic process where the mass of one state is much larger than the mass and kinetic energy of the other one

$$s, m_1^2 \gg s - m_1^2 - m_2^2, m_2^2 \gg Q^2$$
 probe limit. (1.78)

In this limit the light external state can be seen as a probe particle traveling in the gravitational background produced by the heavy object. Thus, in this case, the results can be checked by studying the geodetic motion of a test mass in a fixed classical metric.

2. Leading exponentiation: an appetizer

In this section we describe, in the technically simpler massless case, how to extract the leading eikonal phase $e^{2i\delta_0}$ both from perturbative quantum field theory, following [36,101,102,171], and from purely classical considerations, following [6,172]. We also show how $2\delta_0$ is directly linked to the deflection angle for the classical trajectory via a stationary-phase approximation, and illustrate some related properties of the amplitude that become manifest in its partial-wave decomposition.

2.1. Minimally coupled massless scalars: from amplitudes to geometry

In order to illustrate the leading-order exponentiation in the simplest possible gravitational setup, let us start from the case of two massless scalar fields minimally coupled to gravity,

$$S = \int \frac{R}{2\kappa_D^2} \sqrt{-g} \, d^D x - \frac{1}{2} \sum_{a=1,2} \int \partial_\mu \phi_a \, g^{\mu\nu} \, \partial_\nu \phi_a \, \sqrt{-g} \, d^D x \,, \tag{2.1}$$

with $\kappa_D = \sqrt{8\pi G_D}$. Considering fluctuations about a flat Minkowski background $g_{\mu\nu} = \eta_{\mu\nu} + 2\kappa_D h_{\mu\nu}$ and adopting De Donder gauge leads to the Feynman rules described in Appendix A.2. In particular, the propagators for the graviton and for the scalar with "flavor" index a = 1, 2 are given by (G and G_a are the same function, but it is somewhat helpful to keep track of their subscript)⁶

$$G_{\mu\nu,\rho\sigma}(k) = P_{\mu\nu,\rho\sigma}G(k), \qquad G(k) = \frac{-i}{k^2 - i0}, \qquad G_a(k_a) = \frac{-i}{k_a^2 - i0},$$
(2.2)

where

$$P_{\mu\nu,\rho\sigma} = \frac{1}{2} \left(\eta_{\mu\rho} \eta_{\nu\sigma} + \eta_{\mu\sigma} \eta_{\nu\rho} - \frac{2}{D-2} \eta_{\mu\nu} \eta_{\rho\sigma} \right).$$
(2.3)

Moreover, the leading scalar-graviton-scalar vertex reads

$$\tau_a^{\mu\nu}(p,p') = -i\kappa_D \Big[p^\mu p'^\nu + p^\nu p'^\mu - \eta^{\mu\nu}(pp') \Big],$$
(2.4)

where the scalar lines are regarded as both outgoing. One may consider replacing $P_{\mu\nu,\rho\sigma}$ with the projector over physical graviton states (see (5.45) below), but this replacement is immaterial for our present purposes because $\tau_a^{\mu\nu}$ is transverse, i.e. $\tau_a^{\mu\nu}p_{\mu} = 0 = \tau_a^{\mu\nu}p'_{\mu}$ (and thus also $\tau_a^{\mu\nu}k_{\mu} = 0$ with $k^{\mu} = -p^{\mu} - p'^{\mu}$) when p^{μ} and p'^{μ} are on-shell. Let $\mathcal{A}(s, t)$ be the amplitude for the elastic scattering of 1 and 2. We shall consider its behavior in the Regge limit,

$$s = -(p_1 + p_2)^2 \gg -t = -q^2$$
. (2.5)

In this regime, the incoming particles are highly energetic and barely graze off each other, so that p_1 , p_2 are formally very large compared to q, i.e. the deflection is very small. Moreover, we shall focus on the contribution to the amplitude at L loops, $A_{l}(s, t)$, that arises from the exchange of n = L + 1 virtual gravitons between the two energetic lines as depicted in Fig. 3. As we shall discuss shortly, these "ladder" diagrams indeed provide the dominant contribution to the L-loop amplitude in the Regge limit. We further denote by $\ell_1, \ell_2, \ldots, \ell_n$ the momenta of such exchanges, flowing from 2 to 1 (see Fig. 3). Clearly this contribution takes the form

$$i\mathcal{A}_{n-1}(s,t) = \int \left[\prod_{j=1}^{n} G(\ell_j) \frac{d^D \ell_j}{(2\pi)^D}\right] (2\pi)^D \delta^{(D)} (q-\ell) J^{(n)},$$
(2.6)

⁶ More precisely, since the propagator is diagonal in "flavor" space, $G_{ab}(k) = G_a(k) \delta_{ab}$.



Fig. 3. An example of (crossed) ladder topology. The thick line at the top (blue) refers to particle 1 and the thick line at the bottom (green) refers to particle 2. The exchanged graviton lines are attached and labeled by ℓ_1, \ldots, ℓ_7 in a random way. They flow from bottom to top, as indicated for the ℓ_4 line.

where $\ell = \sum_{j} \ell_{j}$ and $J^{(n)}$ is given by a sum over all ladder topologies of suitable products of massless scalar propagators together with the appropriate contractions between the projectors $P_{\mu\nu,\rho\sigma}$ and the vertices $\tau_{a}^{\mu\nu}$. Before spelling out $J^{(n)}$ explicitly, let us examine one such contraction and retain only the leading order in the Regge or near-forward limit, $\ell_{i} \sim q \ll \sqrt{s}$. We find

$$\tau_1^{\mu\nu}(p_1 - \ell_i, -p_1 + \ell_j) \simeq 2i\kappa_D p_1^{\mu} p_1^{\nu} \tag{2.7}$$

and therefore

$$\tau_1^{\mu\nu}(p_1 - \ell_i, -p_1 + \ell_j)P_{\mu\nu,\rho\sigma}\tau_2^{\rho\sigma}(p_2 + \ell_k, -p_2 - \ell_l) \simeq -\kappa_D^2 s^2.$$
(2.8)

For instance, in the case of a single exchange (tree level),

$$\mathcal{A}_0(s,t) = \boxed{\qquad} \simeq \frac{\kappa_D^2 s^2}{q^2} \tag{2.9}$$

where the subscript 0 stands for tree level. Consequently, to leading order, all dependence on vertices and index contractions factorizes according to

$$J^{(n)} \simeq \left(-\kappa_D^2 s^2\right)^n I^{(n)},$$
(2.10)

and we can rewrite (2.6) as

$$i\mathcal{A}_{n-1}(s,t) \simeq \left(-\kappa_D^2 s^2\right)^n \int \left[\prod_{j=1}^n G(\ell_j) \frac{d^D \ell_j}{(2\pi)^D}\right] (2\pi)^D \delta^{(D)} (q-\ell) \ I^{(n)}$$
(2.11)

with $I^{(n)}$ a purely scalar object given by a sum over all ladder topologies of suitable products of G_1 and G_2 propagators, as we now describe.

Summing all possible ways of attaching the exchanged lines and averaging over their labeling, we can write $I^{(1)} = 1$ and

$$I^{(n)} = \frac{1}{n!} \sum_{\sigma \in S_n} G_1(p_1 - \ell_{\sigma_1}) \cdots G_1(p_1 - \ell_{\sigma_1} - \dots - \ell_{\sigma_{n-1}}) \\ \times \sum_{\sigma' \in S_n} G_2(p_2 + \ell_{\sigma'_1}) \cdots G_2(p_2 + \ell_{\sigma'_1} + \dots + \ell_{\sigma'_{n-1}})$$
(2.12)

for $n \ge 2$, where S_n is the permutation group for n objects and $\frac{1}{n!}$ compensates for multiple counting. Note that, thanks to the explicit averages in Eq. (2.12), we solved for $\ell_{\sigma_n} = q - \ell_{\sigma_1} - \cdots - \ell_{\sigma_{n-1}}$ (and similarly for $\ell_{\sigma'_n}$) without spoiling the permutation symmetry of $I^{(n)}$. Two-particle and three-particle exchanges are illustrated in Fig. 4. Denoting by a subscript \perp the components orthogonal to p_1 and p_2 , to leading order in the Regge limit (2.5) we can decompose the *D*-momentum conserving delta function as follows (the factor of 2s arises from the Jacobian of the appropriate change of variables, see Eq. (A.42))

$$\delta^{(D)}(q-\ell) \simeq 2s \,\,\delta(2p_1\ell)\,\delta(2p_2\ell)\,\delta^{(D-2)}(q_\perp - \ell_\perp)\,. \tag{2.13}$$

Similarly, to leading order, we can make the approximation

$$G_1(p_1 - \ell_k) = \frac{-i}{-2p_1\ell_k + \ell_k^2 - i0} \simeq \frac{-i}{-2p_1\ell_k - i0},$$
(2.14)



Fig. 4. Ladder diagrams with two and three exchanged massless lines. The labels ℓ_1 , ℓ_2 (and ℓ_3) should be assigned in all possible ways.

and similarly for $G_2(p_2 + \ell_k)$. In this way, we obtain

$$I^{(n)} \simeq \frac{(-i)^{2(n-1)}}{n!} \sum_{\sigma \in S_n} \frac{1}{-2p_1 \ell_{\sigma_1} - i0} \cdots \frac{1}{-2p_1 (\ell_{\sigma_1} + \dots + \ell_{\sigma_{n-1}}) - i0} \times \sum_{\sigma' \in S_n} \frac{1}{2p_2 \ell_{\sigma'_1} - i0} \cdots \frac{1}{2p_2 (\ell_{\sigma'_1} + \dots + \ell_{\sigma'_{n-1}}) - i0}.$$
(2.15)

Letting

$$f(a_1,\ldots,a_n) \equiv \sum_{\sigma \in \mathcal{S}_n} \frac{1}{a_{\sigma_1}} \cdots \frac{1}{a_{\sigma_1} + \cdots + a_{\sigma_{n-1}}},$$
(2.16)

we can then use the delta function cast in the form (2.13) and the identity (see Appendix A.1 and Eq. (A.14) in particular)

$$\delta(\omega_1 + \dots + \omega_n) f(\omega_1 - i0, \dots, \omega_n - i0) = (2i\pi)^{n-1} \delta(\omega_1) \cdots \delta(\omega_n)$$
(2.17)

to conclude that

$$i \frac{\mathcal{A}_{n-1}(s,t)}{2s} \simeq \frac{(-\kappa_D^2 s^2)^n}{n!} \int \left[\prod_{j=1}^n G(\ell_j) \delta(2p_1 \ell_j) \delta(2p_2 \ell_j) \frac{d^D \ell_j}{(2\pi)^{D-2}} \right]$$

$$\times (2\pi)^{D-2} \delta^{(D-2)} \left(q_\perp - \ell_\perp \right).$$
(2.18)

This leads naturally to define the following Fourier transform with respect to the (D - 2)-dimensional transverse momentum⁷ q_{\perp} ,

$$\tilde{\mathcal{A}}(s,b) = \int e^{ibq} 2\pi \,\delta(2p_1q) 2\pi \,\delta(2p_2q) \,\mathcal{A}(s,-q^2) \frac{d^D q}{(2\pi)^D} = \int e^{ibq_\perp} \frac{\mathcal{A}(s,-q_\perp^2)}{2s} \,\frac{d^{D-2} q_\perp}{(2\pi)^{D-2}} \,, \tag{2.19}$$

which yields

$$i\tilde{\mathcal{A}}_{n-1}(s,b) \simeq \frac{(-\kappa_D^2 s^2)^n}{n!} \left[\int G(\ell) e^{ib\ell} 2\pi \,\delta(2p_1\ell) 2\pi \,\delta(2p_2\ell) \,\frac{d^D \ell}{(2\pi)^D} \right]^n.$$
(2.20)

The Fourier transform to the impact parameter b simply had the effect of diagonalizing the convolution between singleparticle exchanges occurring in momentum space. The near-forward regime (2.5) translates in impact-parameter space into the large-distance limit

$$\frac{\hbar}{b} \ll \sqrt{s} \,, \tag{2.21}$$

where \hbar has been reinstated for clarity. In conclusion, to leading order in this limit, adding up all contributions coming from *n*-graviton exchanges leads to

$$1 + i \sum_{n=1}^{\infty} \tilde{\mathcal{A}}_{n-1}(s, b) \simeq e^{2i\delta_0} , \qquad (2.22)$$

⁷ See Appendix A.3, in particular Eq. (A.44), for more details concerning these Fourier transforms, including the relation between the two expressions appearing in (2.19).

P. Di Vecchia, C. Heissenberg, R. Russo et al.

with

$$2i\delta_0(s,b) = i\tilde{\mathcal{A}}_0(s,b) = \frac{i\kappa_D^2 s}{2\hbar} \int \frac{e^{ib\ell_\perp}}{\ell_\perp^2} \frac{d^{D-2}\ell_\perp}{(2\pi)^{D-2}} = \frac{iG_D s}{\hbar} \frac{\Gamma\left(\frac{D-4}{2}\right)}{(\pi b^2)^{\frac{D-4}{2}}}.$$
(2.23)

At this point, let us go back and explain what selects the *n*-graviton exchanges among all possible contributions to the elastic amplitude to all loop orders. To clarify this point, notice that when an extra graviton is attached to the energetic scalars the corresponding diagram is enhanced by a factor of $\kappa_D \sqrt{s}$, since the 3-point vertex itself brings a factor of *s*, see (2.7) and (2.8), and the extra energetic propagator (2.14) scales as $\frac{1}{\sqrt{s}}$. If a graviton is added to the diagram in any other way, one obtains a contribution at the same order in the gravitational coupling, but without extra factors of \sqrt{s} . Notice that this is true also if higher-point vertices are used as for instance a contact interaction involving two gravitons and two scalars: since we are working with a two-derivative action (A.15), this vertex can scale at most as $\kappa_D^2 s$, while two 3-point vertices connected by a propagator (2.14) would yield $(\kappa_D s)^2 \frac{1}{\sqrt{s}}$. By dimensional analysis every time we lose a factor of \sqrt{s} , there must be an extra factor of $q \sim \frac{h}{b}$ proving that non-ladder diagrams are subleading in the limit (2.21). In summary the classical, or *eikonal*, limit requires $G_D s b^{4-D} \gg h$ so the full amplitude oscillates infinitely rapidly as

In summary the classical, or *eikonal*, limit requires $G_D s b^{4-D} \gg \hbar$ so the full amplitude oscillates infinitely rapidly as intuitively should be the case. By further taking the near-forward $q \ll \sqrt{s}$ or large-distance $b\sqrt{s} \gg \hbar$ limit, we can make the problem tractable, so the regime of interest for our analysis is specified by the following hierarchy of length scales

$$\frac{\hbar}{\sqrt{s}} \ll G_D \sqrt{s} \, b^{4-D} \ll b \,, \tag{2.24}$$

where $\frac{\hbar}{\sqrt{s}}$ plays the role of a quantum wavelength and $G_D\sqrt{s}b^{4-D}$ is the effective size of the curvature that characterizes the colliding objects. In the leading-order large *b* approximation the full classical amplitude can be approximated by (2.22) and (2.23) which as we saw follows from the resummation of ladder diagrams. The physically relevant D = 4 of (2.24) is thus somewhat special, because the eikonal phase develops a *b*-independent infrared (IR) divergence: the combination $G_D\sqrt{s}$ coincides with (one half) the Schwarzschild radius of a black-hole of mass \sqrt{s} and, from the finite part of the eikonal, one can read the effective size $G\sqrt{s} \log(b\mu)$ to be used in D = 4 version of (2.24). It is natural to identify the IR cutoff μ with the inverse of the distance *r* between the binary and a far-away observer.

In the regime (2.24), it is indeed natural to take the impact parameter *b* and the energy \sqrt{s} (or a length scale *R*, defined in analogy with the Schwarzschild radius by $R^{D-3} = G_D \sqrt{s}$) as the classical quantities characterizing the collision. In terms of these variables the leading eikonal is given by:

$$2i\delta_0 = \frac{iG_{DS}}{\hbar} \frac{\Gamma\left(\frac{D-4}{2}\right)}{(\pi b^2)^{\frac{D-4}{2}}} = i\frac{b\sqrt{s}}{\hbar} \frac{G_D\sqrt{s}}{b^{D-3}} \frac{\Gamma\left(\frac{D-4}{2}\right)}{\pi^{\frac{D-4}{2}}} .$$
(2.25)

To reiterate, in the classical limit we must require that *R* is much bigger than the Compton wavelength of a massless particle $(R \gg \frac{\hbar}{\sqrt{s}})$. On the other hand, in order to apply perturbation theory, we need that the interaction be weak and this corresponds to large values of *b* ($b \gg R$). In conclusion, Eq. (2.25) is valid for large values of *b* and this is obtained by considering the amplitude in the Regge limit ($|t| \ll s$). The factor of $1/\hbar$ in (2.25) signals that this quantity should indeed appear in an exponential $e^{2i\delta_0}$ so it can describe the value of the classical action, while all non-exponentiated terms are of order \hbar^n with n > 0 and describe quantum corrections.

Let us now show how we can calculate from this semiclassical action two important classical observables: the impulse Q and the deflection angle Θ depicted in Fig. 5. In the leading approximation, the exponentiation takes the form (2.22), or briefly

$$\tilde{\mathcal{S}}(s,b) = 1 + i\tilde{\mathcal{A}}(s,b) \simeq e^{2i\delta_0(s,b)}.$$
(2.26)

After exponentiation in impact-parameter space, we can Fourier transform back to momentum space⁸

$$S(s, -Q^2) = 2s \int d^{D-2}b \, e^{-\frac{i}{\hbar}bQ + 2i\delta_0(s,b)} \,. \tag{2.27}$$

Here *Q* describes the full exchanged momentum, or impulse, in the classical process, obtained after the eikonal resummation. Therefore it is important to distinguish it from *q*, the momentum exchanged in the perturbative amplitude in (2.9) via a single graviton. In the classical limit we can approximate (2.27) in terms of the stationary-phase contribution getting, at the saddle point,

$$Q^{\mu} = \hbar \frac{\partial 2\delta_0}{\partial b_{\mu}} = -2G_D s \frac{\Gamma(\frac{D}{2} - 1)}{\pi^{\frac{D}{2} - 2} b^{D-3}} \frac{b^{\mu}}{b}, \qquad (2.28)$$

where *b* denotes the magnitude of b^{μ} . Note that, unlike $2\delta_0$, the resulting leading-order impulse (2.28) is not singular as $D \rightarrow 4$ thanks to the action of the derivative that eliminates the *b*-independent Coulombic pole at D = 4. In terms of the

⁸ $S(s, -Q^2)$ should be distinguished from $S(p_1, p_2; q)$ introduced in (1.27), which also includes the mass-shell delta functions.



Fig. 5. Classical $2 \rightarrow 2$ scattering in the center-of-mass frame (1.33). The difference between *b* and *b_J* is $O(G^2)$ and thus negligible in this section.

quantities introduced after Eq. (2.24), this becomes

$$\frac{Q}{\sqrt{s}} = \frac{2\Gamma(\frac{D}{2}-1)}{\pi^{\frac{D}{2}-2}} \frac{G_D \sqrt{s}}{b^{D-3}} \ll 1 , \qquad (2.29)$$

where the last inequality holds in the limit of large distances where the impact parameter is much larger than *R*. On the other hand, from the Fourier transform (2.19), we have the usual relation $b \sim \hbar/q$ between the impact parameter and the momentum of a *single* graviton exchanged. Then the ratio between the classical impulse (2.28) and this single-graviton momentum transfer can be written as

$$\frac{Q}{q} \simeq \frac{2G_D s}{\hbar} \frac{\Gamma(\frac{D}{2}-1)}{\pi^{\frac{D}{2}-2} b^{D-4}} \simeq \frac{\Theta b}{\hbar} \gg 1$$
(2.30)

and can be interpreted as the typical number of exchanged gravitons in the classical process. This makes it manifest that in the classical limit the number of gravitons exchanged is very large, as signaled by the factor of \hbar^{-1} , while classical observables, such as the impulse (2.29) or the deflection angle (see (2.32) below), remain small in the regime under consideration.

We can now use (2.28) and (2.29) to compute the classical deflection angle Θ defined as the angle between the space-like components of the momenta $-p_1$ and p_4 in the center-of-mass frame (1.33), as shown in Fig. 5. In this frame $Q^2 = (p_1 + p_4)^2 = 2p^2(1 - \cos \Theta)$ where $\hat{p} \cdot \hat{p}' = \cos \Theta$, and thus

$$Q = 2p\sin\frac{\Theta}{2} \quad \Rightarrow \quad \sin\frac{\Theta}{2} = \frac{1}{2p} \left(-\hbar \frac{\partial 2\delta_0}{\partial b} \right), \tag{2.31}$$

where the second expression is obtained by using (2.28). The minus sign comes from the fact that $|\partial_b 2\delta_0| = -\partial_b 2\delta_0$. Incidentally, this shows that the gravitational force is attractive because in this way the deflection of either particle points towards the other one (see Fig. 5). Since, in the massless case, $p = \frac{\sqrt{s}}{2}$, and the condition (2.29), which characterizes the eikonal regime, ensures that Q/p is also small, we can expand $\sin \frac{\Theta}{2} \simeq \frac{\Theta}{2}$, obtaining the leading-order deflection angle

$$\Theta \simeq \frac{4\Gamma(\frac{D}{2}-1)}{\pi^{\frac{D}{2}-2}} \frac{G_D \sqrt{s}}{b^{D-3}} \ll 1.$$
(2.32)

Similarly, one can go to the time domain by taking a Fourier transform over $E = \sqrt{s}$. Again a stationary phase approximation yields a relation between the eikonal and the Shapiro time delay ΔT which measures how much objects are slowed down by the gravitational force [173–175]

$$\Delta T = \hbar \, \frac{\partial 2\delta_0}{\partial E} \,, \tag{2.33}$$

as seen from the center-of-mass frame. Contrary to what happens for the deflection angle (2.28), in this case the Coulomb divergence in δ_0 (2.23) does not cancel, so a better IR-safe observable is for instance the variation of ΔT when *b* changes.

As we shall further discuss in Section 2.3, both deflection angle and time delay were found to agree with classical GR expectations in the approximation in which each particle produces a non-trivial curved metric that affects the geodetic motion of the other particle [6,176]. In conclusion, we have seen how the classical deflection and the time delay, which we may attribute to an effective metric, emerges from standard (quantum) scattering amplitude calculations

⁹ The massive case will be discussed in Section 3.1.1 below.

on flat background spacetime. Notably, this avoids possible challenges, particularly in string theory, with quantization in curved space-time. See however [150–153] for discussions of the eikonal exponentiation in AdS spacetime, as well as [121,122,177] for recent discussions in curved spacetimes.

In the ensuing sections we will see that the eikonal exponentiation generalizes well beyond the simple case sketched here. While the basic idea is the same, it becomes more intricate when extended to the R/b corrections (subleading eikonal) and to theories beyond GR including more fields or higher derivative couplings or to string theory. This mechanism is expected to hold in all consistent gravitational theories, as we will discuss in some detail in this review. However, first, we will comment on the connection between eikonal phase and partial waves, as well as on the interpretation of the eikonal phase as a classical action for shockwave scattering.

2.2. Eikonal phase and phase shift

In this subsection we discuss the relation between the partial-wave decomposition introduced in Section 1.3 and the eikonal phase. As we shall see, this will highlight how the eikonal exponentiation is also instrumental in solving an apparent tension between the explicit result for the tree-level amplitude (2.23) and the unitarity bound (1.70).

The partial wave expansion of the resummed amplitude (2.27) is most easily performed for D = 4, but in this case the leading eikonal phase (2.25) is divergent. On the other hand, after exponentiation, this divergence is an overall *b*-independent phase which can be ignored (this is the usual Coulomb divergence due to the long-range nature of the tree-level potential in three space dimensions). The finite *b*-dependent contribution is

$$2\delta_0(s,b) \to -\frac{Gs}{\hbar} \log(\mu b)^2 \,, \tag{2.34}$$

with μ an inverse length scale introduced for dimensional reasons, so that the integral in (2.27) can be performed exactly by using the general Fourier transform (A.49),

$$\mathcal{S}(s, -Q^2) \simeq i \frac{8\pi G s^2}{Q^2} \left(\frac{4\mu^2 \hbar^2}{Q^2}\right)^{-i\alpha_G} \frac{\Gamma(1 - i\alpha_G)}{\Gamma(1 + i\alpha_G)}.$$
(2.35)

Here

$$\alpha_G \equiv \frac{Gs}{\hbar} \tag{2.36}$$

is dimensionless and provides an analog of the fine structure constant of QED. We see that the expression (2.35) for the resummed amplitude is equal to the one for single-graviton exchanges (2.9) with the substitution $q \rightarrow Q$ times two extra phase factors (the ratio of the two Γ -functions is also a pure phase).

In order to apply the partial wave projection (1.63), we note that in the massless case $2p = \sqrt{s}$ and recall the Rodrigues formula,

$$P_j(x) = \frac{1}{2^j j!} \frac{d^j}{dx^j} (x^2 - 1)^j.$$
(2.37)

We then need to compute the following integral,

$$\frac{(-1)^{j}}{2^{j}j!} \int_{-1}^{+1} dx \frac{d^{j}}{\partial x^{j}} (1-x^{2})^{j} (1-x)^{-1+i\alpha_{G}}.$$
(2.38)

Integrating by parts, one can easily bring all the derivatives with respect to *x* on the term with α_G , then the integral can be performed by changing variable from *x* to $z = \frac{1+x}{2}$ and recognizing that one obtains a quantity proportional to the Euler Beta-function. By including all the factors of the amplitude (2.35) one obtains¹⁰

$$s_j(s) = e^{2i\delta_j(s)} = \left(\frac{s}{4\mu^2\hbar^2}\right)^{i\alpha_G} \frac{\Gamma(1+j-i\alpha_G)}{\Gamma(1+j+i\alpha_G)}.$$
(2.39)

We now consider the classical limit,

$$j \sim \alpha_G \gg 1$$
, $J = \hbar j$, (2.40)

where the quantum number j takes large values, in order to lead to a sizeable classical angular momenta J. Taking the limit (2.40) in (2.39), and using the Stirling approximation, we find

$$2\delta_j = -\left[\alpha_G \log(j^2 + \alpha_G^2) + 2j \arctan\frac{\alpha_G}{j}\right],\tag{2.41}$$

 $^{^{10}}$ At fixed *j* that ratio has singularities along the imaginary *s*-axis, known as 't Hooft's poles. However, their presence depends on the large-Q behavior of the resummed scattering amplitude which is beyond control of the leading eikonal approximation [16].

where we used $\arctan \frac{\alpha_G}{i} = \frac{1}{2i} \log \frac{j + i\alpha_G}{i - i\alpha_C}$. Moreover, to leading order in the PM limit $j \gg \alpha_G \gg 1$, this simplifies to

$$2\delta_j(s) \simeq -\frac{Gs}{\hbar} \log(j^2), \qquad (2.42)$$

and thus, comparing with (2.34),

$$2\delta_0(s,b) \simeq 2\delta_j(s), \qquad \hbar j = J \simeq pb, \qquad (2.43)$$

up to subleading, *b*-independent corrections. At higher PM orders, as we shall see, it is instead important to maintain the distinction between $2\delta_i(s)$ and $2\delta(s, b)$. For this reason, we introduce a special symbol for $2\delta_i(s)$ in the classical limit,

$$2\delta_i(s) = \chi(s, J), \tag{2.44}$$

which is a function of the continuous variable $J = \hbar j$. We shall come back to this point in Section 6.1.

This identification between *b* and *j*-expansions to leading order, which is also natural by geometric considerations (see Fig. 5), seems to raise a problem with the partial wave unitarity bound (1.70). Indeed, focusing on the tree-level amplitude, $|\tilde{A}_0(s, b)| \simeq |f_j(s)| \gg 1$ in the classical, high-energy regime, which seems to contradict (1.70). However the resummed amplitude makes perfect sense since the large quantity $\tilde{A}_0(s, b) = 2\delta_0(s, b) \simeq 2\delta_j(s)$ appears only as a phase $e^{2i\delta_0}$. Thus, S(s, b) and $s_i(s)$ simply saturate the bound (1.70), by neglecting inelastic contributions,

$$|s_j(s)|^2 = 1, (2.45)$$

as one can also see directly from (2.39). When taking inelastic processes into account, as we shall see in the following, 2δ develops a *positive* imaginary part, so that

$$|s_i(s)|^2 = e^{-2\ln 2\delta_j(s)} < 1$$
(2.46)

and the bound is still respected, but no longer saturated. In this way, the eikonal exponentiation is instrumental in resolving the apparent tension between the high-energy behavior of perturbative amplitudes in gravity and the bounds coming from the unitarity constraint. In fact, this was one of the main historical motivation for its study.

Let us now go back to the connection between partial waves and deflection angle. Starting from (1.62), we see that the Q-space representation of the amplitude is given by a sum of the type

$$S(s, Q) \propto \sum_{j} P_{j}(\cos \Theta) e^{2i\delta_{j}}, \qquad Q = 2p \sin \frac{\Theta}{2}.$$
 (2.47)

Using the large-*j* limit of $P_i(\cos \Theta)$, (see for instance DLMF (14.15.11) with $\mu = 0$ and $\nu = j$ and (6.4))

$$P_{j}(\cos\Theta) \underset{j \to \infty}{\sim} \sqrt{\frac{2\Theta}{\pi \sin\Theta}} \cos\left(\left(j + \frac{1}{2}\right)\Theta - \frac{\pi}{4}\right),$$
(2.48)

the classical deflection angle can then be obtained by applying the saddle point condition to sum in (2.47), obtaining¹¹

$$\Theta = -\frac{\partial 2\delta_j(s)}{\partial j} = -\hbar \frac{\partial \chi(s, J)}{\partial J}, \qquad (2.49)$$

where we used the definition of $\chi(s, J)$ (2.44) in the last step. Naturally, substituting the approximate expression (2.42) for $j \gg \alpha_G \gg 1$ we simply recover the result of (2.32) for the leading-order deflection angle in D = 4,

$$\Theta \simeq \frac{2Gs}{J} \simeq \frac{4G\sqrt{s}}{b}, \qquad pb \simeq J = \hbar j.$$
(2.50)

Of course, in a physical quantity as the deflection angle, the dependence on μ drops out.

However, we can obtain a relation which formally does not rely on the PM expansion by substituting (2.39) for $j \sim \alpha_G \gg 1$, according to which

$$\Theta = i\left(\psi(j+1-i\alpha_G) - \psi(j+1+i\alpha_G)\right) \tag{2.51}$$

where ψ is the logarithmic derivative of the Γ -function. Using the following equation for ψ :

$$\psi(1+z) = -\gamma + \sum_{n=1}^{\infty} \left(\frac{1}{n} - \frac{1}{n+z} \right)$$
(2.52)

we can write Θ as follows:

$$\Theta = \sum_{m=j+1}^{\infty} \frac{2\alpha_G}{m^2 + \alpha_G^2} \sim 2 \int_0^{\frac{\alpha_G}{j}} \frac{dx}{1 + x^2} = 2 \arctan \frac{\alpha_G}{j}$$
(2.53)

¹¹ Here we use that, out of the two saddle-points $\Theta = \pm \partial_j(2\delta_j)$, since $\Theta \in [0, \pi]$, only $\Theta = -\partial_j(2\delta_j)$ can be realized in theories like gravity where the interaction is attractive, $-\partial_i(2\delta_j) > 0$.

where in the middle step we have approximated the sum with an integral by using $x = \alpha_G/m$ for large values of *j*. Equivalently, we could have substituted in (2.51) directly the expression (2.41), obtaining the same conclusion. Then we get

$$\tan\frac{\Theta}{2} = \frac{\alpha_G}{j}.$$
(2.54)

Taking into account that, in the classical limit, $j\hbar = pb_J = J$ we can compare this equation with (2.31). They have precisely the same form in D = 4 except for the substitution of sin with tan and of b with b_J . Taking into account the difference between b and b_J in Fig. 5 they are in agreement up to order G^2 . Let us also mention that, as discussed in Appendix B.3, the probe-limit scattering of massive objects in $\mathcal{N} = 8$ obeys a relation formally very close to (2.54).

2.3. Minimally coupled massless scalars: from geometry to amplitudes

Interestingly enough, the result (2.25) was also obtained, and at exactly the same time, by 't Hooft [6] from a calculation that follows exactly the opposite logic with respect to the one of the two previous subsections. The starting point of 't Hooft's approach is the *classical* shock-wave metric produced by a massless particle of given energy $E^{(1)}$ moving, say, in the positive-*z* direction at $u \equiv t - z = 0$. That metric was obtained long ago by Aichelburg and Sexl (AS) in the classic paper [178]. It can be easily generalized to arbitrary *D* (see e.g. [176]) and, in suitable coordinates, it reads

$$ds^{2} = -du \, dv + f(x_{\perp})\delta(u)du^{2} + \sum_{i=1}^{D-2} dx^{i} \, dx^{i} \,, \qquad v \equiv t+z \,,$$
(2.55)

where $x_{\perp} = (x^1, \dots, x^{D-2})$ and, in order to satisfy Einstein's equations with a source,

$$\Gamma_{\mu\nu} = E^{(1)}\delta(u)\delta^{(D-2)}(x^{i})\delta^{u}_{\mu}\delta^{u}_{\nu},$$
(2.56)

 $f(x_{\perp})$ should satisfy the equation

$$\Box_{\perp} f(x_{\perp}) = -16\pi G_D E^{(1)} \delta^{(D-2)}(x_{\perp}), \qquad \Box_{\perp} \equiv \sum_{i=1}^{D-2} \partial_i^2.$$
(2.57)

Hence, f is proportional to the well-known Green function in D-2 dimensions. More specifically, one finds¹²

$$f(\mathbf{x}_{\perp}) = -16\pi G_D E^{(1)} \frac{r^{4-D}}{(4-D)\Omega_{D-2}}, \qquad r^2 \equiv \sum_{i=1}^{D-2} (x^i)^2, \qquad \Omega_{D-2} = \frac{2\pi^{\frac{D-2}{2}}}{\Gamma(\frac{D-2}{2})}$$
(2.58)

(note that, in any D, f has the correct dimensions of a length).

Consider now the null geodesics describing the worldline of massless (test) particles moving in the opposite direction, initially at v = 0 and at transverse coordinates $x^i = b^i$. They describe straight lines at t < 0, but suffer both a *v*-delay and a deflection for $t \ge 0$.

These geodesics were considered in [6,179] for the AS metric and generalized in [176] for arbitrary *D* as well as for extended sources. It turns out that, in these convenient albeit rather singular coordinates, the *v*-coordinate of the test particle suddenly changes from zero to f(b). This *v*-delay is instantaneous and simply given by (using $r \rightarrow b$):

$$v \sim \theta(u)f(b) \Rightarrow \Delta v = f(b) = 4G_D E^{(1)} b^{4-D} \frac{\Gamma(\frac{D-4}{2})}{\pi^{\frac{D-4}{2}}}$$
(2.59)

where $\theta(u)$ is the Heaviside step function. We refer to [176,180] for more details about the solution of the geodesic equation.

't Hooft next considered a Lorentz frame in which particle 1 is very energetic and produces the shock wave while particle 2 is very soft and behaves as a test particle suffering the *v*-shift (2.59). Quantum mechanically, particle 2 has a wave function $\Psi^{(2)} \sim \exp(i\frac{E^{(2)}v}{\hbar})$ for t < 0 but, as it goes through t = 0, its phase picks up a shift given by

$$\frac{E^{(2)}\Delta v}{\hbar} = \frac{E^{(2)}f(b)}{\hbar} = \frac{G_D s}{\hbar b^{D-4}} \frac{\Gamma(\frac{D-4}{2})}{\pi^{\frac{D-4}{2}}}$$
(2.60)

where we have used that $4E^{(1)}E^{(2)} = s$. The above equation is precisely the phase shift in (2.25).

As pointed out in [172] this raised a doubt: working in the boosted reference frame mentioned above, the shock wave produced by the soft particle (2) is extremely weak, but it cannot be neglected because, when it is used to calculate its contribution to the phase of the wave function of the energetic particle (1), it is multiplied by a large factor $E^{(1)}$ producing

¹² In [176] the AS metric was (almost trivially) extended to a "beam" of null energy which is still concentrated at u = 0 but has an arbitrary transverse profile. Such effective shock waves naturally appear when the source is extended, like in the case of strings. See Section 3.2.6.

in the end exactly the same contribution (2.60). So it seems that the total classical phase shift is twice the eikonal result (2.25). This puzzle was solved in [172] by realizing that, in this geometric description, the classical action receives an extra contribution from the boundary term needed to define properly the Einstein–Hilbert variational principle. In the case at hand, this extra contribution cancels precisely half of the result obtained by simply summing the contributions of the two particles, thus restoring the agreement with the amplitude-based result.

As we will see in Appendix B, the same idea can be applied, even more straightforwardly, to the scattering of massive particles at the leading order in the probe limit (1.78), where particle 1 can be described by a stationary geometry and the classical action is derived from the geodesic followed by the particle 2. A similar approach also works, with the appropriate modifications, in the case of string–string collisions as discussed in detail in Section 3.2.6.

As discussed in Section 4.1.5, the probe limit actually determines not only the 1PM, i.e. $\mathcal{O}(G)$ eikonal phase, but also the 2PM one, corresponding to a one-loop calculation on the amplitude side. Instead, new classical data that is not fixed by the probe limit appears at 3PM, two-loop order, both in the massless and in the massive setups. Let us finally remark that this elegant method based on approximating the problem as the one of a particle scattering in the metric produced by the other particle represents the starting point of the so-called self-force expansion [29,181–183].

2.4. Collapse criteria in GR

Following up on the discussion of the previous subsection we would like to consider now the opposite situation in which not only the probe approximation breaks down, but even a perturbative expansion in $\frac{R}{b}$ is bound to fail (diverge?). This is when classical GR arguments lead to the conclusion that a finite fraction of the initial energy should collapse to form a black hole.

Of course, the quantitative study of such a regime is out of reach both by analytic and by numerical relativity methods. Amusingly, however, some non trivial results can be obtained in the strictly massless case as first pointed out by Penrose [184] for the axisymmetric situation (b = 0). The basic simplifying feature of massless collisions is that, by causality, nothing happens before the two front waves (at u = 0 and v = 0) meet at t = 0. In other words, a simple linear superposition of two AS shock waves of the form (2.55) exactly satisfies Einstein's equations for t < 0.

The idea of Penrose, later extended to $b \neq 0$ by Eardley and Giddings [185], is to check whether or not a marginally trapped closed surface (MTCS)¹³ can be constructed using the exact, known geometry at $t \rightarrow 0^-$. If the answer were positive, then we would conclude from general theorems that such a surface evolves later into the horizon of a black hole whose mass is bounded from below by the area of the initial surface. Unfortunately, the opposite statement cannot be claimed to be true: even if one can prove that at t = 0 there is no MTCS, one cannot exclude that one such surface is formed at some later time t > 0. A fortiori, only a lower limit on the mass of the future black hole (and therefore an upper limit on the total energy reaching future null infinity) can be given.

In his work [184], Penrose found that, at b = 0, a MTCS can be easily constructed and, from the value of its area, he concluded that at least a fraction $\frac{1}{\sqrt{2}}$ of the total incoming energy goes into forming a black hole. Eardley and Giddings, instead, managed to construct a MTCS (in D = 4) for $b < \gamma_{EG}G\sqrt{s}$ with $\gamma_{EG} \simeq 1.61$, meaning that the critical ratio for collapse $(\frac{G\sqrt{s}}{b})_{cr} \leq 1/\gamma_{EG} \simeq 0.62$. An interesting generalization [186] (particularly in view of a collision of extended objects like strings) is the one in

An interesting generalization [186] (particularly in view of a collision of extended objects like strings) is the one in which one considers the collision on two null beams of massless particles (moving again in opposite directions along the z-axis). As physically expected, one finds that the transverse size of the beam "adds up" to the impact parameter (defined as the transverse distance of the two beam's center of mass). One starts from a simple generalization of (2.57):

$$\Box_{\perp} f^{(i)}(x_{\perp}) = -16\pi G \rho^{(i)}(x_{\perp}), \qquad i = 1, 2,$$
(2.61)

where $\rho^{(i)}(x_{\perp})$ is the energy density per unit transverse area of each beam. Since the solution (2.58) of (2.57) is nothing but the Green function of the problem, the general solution of (2.61) is also known.

At this point one can check whether a MTCS is formed already at t = 0. Different physical situations were discussed in [186]. Let us just mention two relatively simple cases:

• Axisymmetric collision on two non-homogeneous axisymmetric beams. This generalizes Penrose's construction to extended objects. In this case $\rho^{(i)}(x_{\perp})$ depends just on the radial distance r from the symmetry axis. Consider the energy in each beam below some given r:

$$E^{(i)}(r) = \Omega_{D-2} \int_0^r \rho^{(i)}(r') r'^{D-3} dr' \; ; \; \Omega_d = \frac{2\pi^{d/2}}{\Gamma(d/2)} \, . \tag{2.62}$$

One finds that a MTCS can be constructed whenever one can find an r_c such that:

$$\Omega_{D-2}r_c^{D-3} \le 8\pi G_D \sqrt{E^{(1)}(r_c)E^{(2)}(r_c)}$$
(2.63)

where r_c plays the role of an effective impact parameter. In the special case of two identical homogeneous disks of radius L (2.63) simply gives $\Omega_{D-2}L^{D-3} \leq 4\pi G_D\sqrt{s}$ ($L \leq 2G\sqrt{s}$ in D = 4).

¹³ A (marginally) closed trapped surface is, loosely speaking, a closed surface whose future directed light cone is (marginally) inside the surface itself. More precisely, it may be defined as a (D-2)-dimensional closed surface whose outer normals have zero "convergence" (or "expansion").

• Collision of two homogeneous beams at $b \neq 0$. Considering for simplicity the case of D = 4 one finds that a MTCS can be constructed at t = 0 provided:

$$G\sqrt{s} > 2L \text{ and } b < G\sqrt{s} - 2L. \tag{2.64}$$

Comparing this to the point particle criterion of Ref. [185], we see that the role of *b* is now played by $b_{\text{eff.}} = b + 2L$ suggesting that no black hole is formed if the total energy is such that $G\sqrt{s} < 2L$, i.e. that the associated Schwarzschild radius is smaller than the sizes of the two beams.

Many other interesting issues belong to this problematic. One is the nature of the "phase transition" between the collapse and "dispersive" regime, see e.g. [187]. Is it first order, i.e. with the mass of the black hole having a lower bound as soon as $\frac{R}{b} > (\frac{R}{b})_{cr}$, or second order, in which case we may talk about the critical exponent α appearing in $m_{BH} \sim (b_c - b)^{\alpha}$ for $b \rightarrow b_c^-$? This question is related to one raised recently by Don Page [188] of whether at $b = b_c$ the fraction of initial energy being radiated in a massive point-particle collision approaches 1 in the limit of infinite Lorentz boost. These topics being somewhat outside the scope of this report, we refer the interested reader to the literature.

3. Leading eikonal for generic gravitational 2-body scattering

In the introductory Section 2 we discussed how, in the massless case, the leading eikonal is obtained from the Fourier transform to impact parameter space of the tree diagram with one graviton exchange in the Regge limit where $s \gg |t|$. In this section we want to extend this result to the massive case in several gravitational theories. We start from Einstein's GR coupled to massive matter fields, then consider extended theories of gravity with extra massless fields (such as the dilaton) or with higher derivative couplings and finally discuss the case of string theory. We focus on the tree-level approximation and on the limit where the momentum exchanged is much smaller than both the center of mass energy and the masses, since these contributions determine the leading eikonal. For minimally coupled massive scalars in GR the exponentiation of the leading eikonal exponentiates for all gravitational theories considered in this section and then discuss how to generalize the analysis of Section 2.2 and extract the classical physics observables of interest. We will postpone to the next sections the study of the one and two loop amplitudes relevant to eikonal exponentiation: they will serve as a check of the assumption taken here and provide new information about the subleading terms in the eikonal expansion.

3.1. Field theory at tree-level

3.1.1. Minimally coupled massive scalars

We consider here pure *D*-dimensional GR minimally coupled to two scalar fields with masses m_1 and m_2 . This is a simple generalization of the equal-mass setup analyzed in [101]. The action reads

$$S = \int d^{D}x \sqrt{|g|} \left\{ \frac{R}{2\kappa_{D}^{2}} - \frac{1}{2} \sum_{i=1}^{2} \left[\partial_{\mu}\phi_{i}\partial_{\nu}\phi_{i}g^{\mu\nu} + m_{i}^{2}\phi_{i}^{2} \right] \right\}.$$
(3.1)

As discussed in the introduction, the eikonal exponentiation also lends itself to describe the classical interactions between two Schwarzschild black holes at large distances, so we take the parameters m_1 and m_2 to be classically large, e.g. of the order of ten solar masses. Because of the no-hair theorem, it is natural to use Eq. (3.1) as a starting point, since the mass is the only feature that is relevant classically. In order to describe other compact objects, such as neutron stars, one needs to add non-minimal couplings parametrizing tidal deformations or other classical quantities besides the mass, for instance to describe spin.

To leading order in the conventional Born approximation, the interaction between the scalars ϕ_1 and ϕ_2 is captured by the diagram with a single graviton exchanged between the two massive lines. By using the Feynman propagator (A.23) and the vertex (A.28), which generalizes the one in (2.4),

$$\tau_a^{\mu\nu}(p,p') = -i\kappa_D \left[p^{\mu} p'^{\nu} + p^{\nu} p'^{\mu} - \eta^{\mu\nu} (p \cdot p' - m_a^2) \right],$$
(3.2)

it is straightforward to calculate the contribution of this diagram, which in D dimension reads

$$\mathcal{A}_{0}(s,t) = \underline{\qquad} = \frac{2\kappa_{D}^{2}}{-t} \left[\frac{1}{2} (s - m_{1}^{2} - m_{2}^{2})^{2} - \frac{2m_{1}^{2}m_{2}^{2}}{D - 2} + \frac{t}{2} \left(s - m_{1}^{2} - m_{2}^{2} \right) \right].$$
(3.3)

We then focus on the non-analytic terms in the $t = -(p_1 + p_4)^2 \sim 0$ limit which, as in the massless case, will capture the long range interaction in impact parameter space. In this approximation the amplitude (3.3) is dominated by the pole located at $t = -q^2 = 0$,

$$\mathcal{A}_0(s, -q^2) = 2\kappa_D^2 \frac{\gamma(s, m_i)}{q^2} + \mathcal{O}((q^2)^0) , \qquad (3.4)$$

with

$$\gamma(s, m_i) \equiv \frac{1}{2}(s - m_1^2 - m_2^2)^2 - \frac{2m_1^2m_2^2}{D - 2} = 2m_1^2m_2^2\left(\sigma^2 - \frac{1}{D - 2}\right),\tag{3.5}$$

where in the final step we give the result in terms of the variable σ defined in (1.8). The leading term in (3.4) can be derived by replacing (3.2) with the effective vertex

$$\tau_{\rm eff}^{\mu\nu}(p,p') = 2i\kappa_D \bar{p}^\mu \bar{p}^\nu, \qquad p^\mu = -\bar{p}^\mu + \frac{1}{2}q^\mu, \qquad p'^\mu = \bar{p}^\mu + \frac{1}{2}q^\mu, \tag{3.6}$$

where one introduces an "average momentum" \bar{p}^{μ} as in (1.28), so that $\bar{p} \cdot q = 0$, and (3.6) differs from (3.2) by terms suppressed in the small q limit (thanks to (1.28) and $(p \cdot p' - m_a^2) = \frac{q^2}{2}$). The result for the massless amplitude is recovered by considering the ultrarelativistic limit,

$$\sigma \to \infty, \qquad m_1 m_2 \sigma \sim \frac{s}{2},$$
(3.7)

and it is easy to check that, in this limit, (3.4) reproduces (2.9).

Actually the Born approximation is not justified in the setup under analysis and, in the massive case, even the low velocity limit is outside the tree-level regime of validity. This can be seen by adapting the derivation in Section 2.1 to this new kinematic setup. As before, we take the Fourier transform of (3.4) by keeping the external states on-shell to rewrite the result in impact parameter space: as shown in Appendix A.3, we can decompose q in the component q_{\perp} , which is perpendicular to the $p_{1,2}$ and the longitudinal components and the latter can be neglected since they are proportional to q^2 , see (A.37), and kill the pole in (3.4) yielding to irrelevant analytic terms. Thus we effectively deal with the (D-2)-dimensional Fourier transform appearing in the leading term of (A.40)

$$\tilde{\mathcal{A}}_{0}(s,b) = \int \frac{d^{D-2}q_{\perp}}{(2\pi)^{D-2}} \frac{\mathcal{A}_{0}(s,-q_{\perp}^{2})}{4Ep} e^{ibq_{\perp}} , \qquad (3.8)$$

where *E* and *p* are respectively the total energy and the absolute value of the spatial momentum in the center-of-mass frame (see Eq. (1.33)). Because of the scaling (1.2), \tilde{A}_0 is large even at low velocities. Then, as discussed in detail in Section 4, the 1-loop correction, arising from diagrams where two gravitons are exchanged between the scalar particles, includes a term involving the convolution of Eq. (3.3) with itself. This yields in impact parameter space a contribution proportional to \tilde{A}_0^2 which signals a clear breakdown of standard perturbation theory since \tilde{A}_0 is large. However we expect that the eikonal exponentiation resums this class of contributions as in (2.22), so we can identify the leading tree-level contribution (3.4) in impact parameter space with the 1PM eikonal.

$$\tilde{\mathcal{A}}_0 = 2\delta_0 . \tag{3.9}$$

Then by using Eq. (A.45) to calculate (3.8), we find

$$2\delta_0 = \frac{2m_1m_2G_D\left(\sigma^2 - \frac{1}{D-2}\right)\Gamma\left(\frac{D-4}{2}\right)}{\sqrt{\sigma^2 - 1}(\pi b^2)^{\frac{D-4}{2}}} .$$
(3.10)

In the ultrarelativistic limit (3.7), Eq. (3.10) reduces to (2.23) and we recover the massless case once again. Notice that in this limit Eq. (2.25) is the leading eikonal for any gravitational field theory in the two derivative approximation. This universality is a consequence of the Regge limit thanks to the following two properties. First, in the high energy regime amplitudes are dominated by the exchange of the states with the highest spin: indicating with *j* the spin of such particles, their leading contribution to A_0 scales as s^j . Then, when we are interested in long range effects, we can focus just on exchanges of massless particles. In gravitational theories the highest spin massless particle is the graviton which gives rise to a universal¹⁴ contribution to A_0 scaling as s^2 . Thus, in the ultrarelativistic regime, the presence of other lower spin fields is irrelevant for the derivation of the leading eikonal.

Going back to general kinematics, we can proceed as in the massless case and use Eq. (2.31) to compute the contribution of the leading eikonal to the deflection angle

$$2p\sin\frac{\Theta_0}{2} = \frac{4G_D m_1 m_2 \left(\sigma^2 - \frac{1}{D-2}\right) \Gamma\left(\frac{D-2}{2}\right)}{\sqrt{\sigma^2 - 1\pi^{\frac{D-4}{2}} b^{D-3}}} .$$
(3.11)

As discussed in detail in the later sections, the subleading eikonal terms will provide further contributions to the physical deflection angle Θ and so, in a PM approach, this result can be trusted up to $\mathcal{O}(G_D)$

$$\Theta = \frac{4G_D E\left(\sigma^2 - \frac{1}{D-2}\right) \Gamma\left(\frac{D-2}{2}\right)}{(\sigma^2 - 1)\pi^{\frac{D-4}{2}} b^{D-3}} + \mathcal{O}(G_D^2), \qquad (3.12)$$

¹⁴ As we will see in Section 3.1.5 the universality of the graviton coupling can be spoiled in presence of higher derivative corrections.

where we expanded the sin function to leading order and rewrote the momentum *p* in terms of the center-of-mass energy *E* thanks to (1.13). However, also the expression in (3.11) is useful as it contains all the non-linear contribution to the angle arising from the leading eikonal. For instance it allows to recover the full leading answer in the standard D = 4 Post-Newtonian (PN) limit $\frac{1}{l_{PN}} \sim v_{\infty} \ll 1$ where¹⁵

$$\sigma = \sqrt{1 + v_{\infty}^2}, \qquad \frac{1}{j_{\rm PN}} = \frac{Gm_1m_2}{J}.$$
(3.13)

One can easily recast the angle (3.11) in terms of the angular momentum by using the relation¹⁶ $b_J = b \cos(\Theta/2) = b$, which is obvious from Fig. 5, where by definition the angular momentum equals $J = pb_J$. Then we have

$$\tan\frac{\Theta_0}{2} = \frac{Gm_1m_2\left(2\sigma^2 - 1\right)}{J\sqrt{\sigma^2 - 1}}, \qquad \frac{\Theta_{0\rm PN}}{2} = \arctan\left(\frac{1}{j_{\rm PN}\,v_\infty}\right). \tag{3.14}$$

where in the equation on the right we took the leading PN term finding the classic Newtonian result, see [189] and references therein. In this regime, even if the scattering angle is finite, the momentum transfer in (3.11) is small because $p \rightarrow 0$.

Instead, in the leading PM approximation, we can use $b_l = b + O(G_D^2)$ obtaining

$$\Theta = \frac{4G_D E\left(\sigma^2 - \frac{1}{D-2}\right) \Gamma\left(\frac{D-2}{2}\right)}{(\sigma^2 - 1)\pi^{\frac{D-4}{2}}} \frac{p^{D-3}}{J^{D-3}} + \mathcal{O}(G_D^2) \,.$$
(3.15)

Let us also collect here for later convenience the D = 4 expressions for the 1PM impulse and the deflection angle,

$$Q = \frac{4Gm_1m_2\left(\sigma^2 - \frac{1}{2}\right)}{b\sqrt{\sigma^2 - 1}} + \mathcal{O}(G^2), \qquad \Theta = \frac{4GE\left(\sigma^2 - \frac{1}{2}\right)}{b(\sigma^2 - 1)} + \mathcal{O}(G^2).$$
(3.16)

Another interesting limit is to take one mass much larger than the energy of the other, for instance $m_2 \gg E_1$. In this regime, we can see particle 1 as a probe propagating in the gravitational background of the heavy particle 2. So it should be possible to obtain (3.15) by a classical calculation studying the geodetic motion in the appropriate curved geometry. This probe limit provides a straightforward but important check on the diagrammatic calculations, in particular when testing the subleading terms in the eikonal expansion. In this limit the total energy coincides with the rest mass of the heavy particle (M_h) while σ and p can be written in terms of the energy and the mass of the light probe particle

$$m_2 \equiv M_h , \qquad \sigma \to \frac{E_1}{m_1} , \qquad E \to \sqrt{M_h^2 + 2M_h E_1} , \qquad p \to \sqrt{E_1^2 - m_1^2} .$$
 (3.17)

By using (3.17) in (3.4) we have

$$\mathcal{A}_0 \sim 4\kappa_D^2 M_h^2 \frac{(D-2)E_1^2 - m_1^2}{(D-2)(-t)} .$$
(3.18)

In this limit the factor of 4*Ep* in (3.8) reduces to $4M_h\sqrt{E_1^2 - m_1^2}$, so in impact parameter space we have

$$2\delta_0 = \tilde{\mathcal{A}}_0(s,b) = \frac{2G_D M_h}{(D-2)} \frac{(D-2)E_1^2 - m_1^2}{\sqrt{E_1^2 - m_1^2}} \frac{\Gamma\left(\frac{D-4}{2}\right)}{(\pi b^2)^{\frac{D-4}{2}}} \,. \tag{3.19}$$

Then, by using again (2.31), we can derive the contribution from δ_0 to the deflection angle

$$\Theta = \frac{\sqrt{\pi}\Gamma\left(\frac{D-2}{2}\right)}{2\Gamma\left(\frac{D-1}{2}\right)} \frac{(D-2)E_1^2 - m_1^2}{E_1^2 - m_1^2} \left(\frac{R}{b}\right)^{D-3} + \mathcal{O}(G_D^2) , \qquad (3.20)$$

where we used (B.2) to introduce the *D*-dimensional Schwarzschild radius. It is then easy to check that, at linear order in *G*, Eq. (3.20) agrees with the result of the classical geodesic calculation (B.14) (since at leading order $J = pb_J \simeq pb = \sqrt{E_1^2 - m_1^2} b$).

3.1.2. Graviton scattering off a massive scalar

Another instructive tree-level result describes the amplitude between two scalar and two gravitons. A direct derivation from (3.1) requires to evaluate four Feynman diagrams and involves the cumbersome three-graviton vertex. It is possible

¹⁵ We follow the conventions of [189].

 $^{^{16}}$ In Section 6.1 we will present a derivation of this relation based on the standard partial wave decomposition as done in Section 2.2 for the leading eikonal.

to obtain a more compact expression by taking the field theory limit of a string amplitude [48]

$$i\mathcal{A}^{\alpha\beta;\rho\sigma} = -i\frac{2\kappa_D^2(k_4q_2)(k_1q_2)}{(q_2q_3)} \left[\frac{k_4^{\rho}k_1^{\alpha}}{k_4q_2} + \frac{k_4^{\alpha}k_1^{\rho}}{k_1q_2} + \eta^{\rho\alpha} \right] \left[\frac{k_4^{\sigma}k_1^{\beta}}{k_4q_2} + \frac{k_4^{\beta}k_1^{\sigma}}{k_1q_2} + \eta^{\sigma\beta} \right],$$
(3.21)

where k_i label the momenta of the massive scalars and q_i those of the gravitons, see Fig. 6. The factorized form of the result is a by-product of using the KLT approach [190] at the string theory level. If we focus on the D = 4 case, then it is possible to rewrite the result in terms of the compact spinor helicity formalism. As usual one can express the massless momenta in terms of a pair of commuting SU(2) spinors¹⁷

$$p_{\alpha\dot{\alpha}} = p_{\mu}\sigma^{\mu}_{\alpha\dot{\alpha}} = \lambda_{\alpha}\lambda_{\dot{\alpha}} , \qquad (3.22)$$

where $\tilde{\lambda}$ is the complex conjugate of λ if one sticks to the standard Lorentzian signature. The two physical helicities of the gravitons can be written in terms of

$$\varepsilon_{\alpha\dot{\alpha}}^{+} = \sqrt{2} \, \frac{\mu_{\alpha}\bar{\lambda}_{\dot{\alpha}}}{\langle\mu\lambda\rangle} \qquad \varepsilon_{\alpha\dot{\alpha}}^{-} = \sqrt{2} \, \frac{\lambda_{\alpha}\tilde{\mu}_{\dot{\alpha}}}{[\tilde{\lambda}\tilde{\mu}]} \tag{3.23}$$

by using $\epsilon^{++} = (\epsilon^+)^2$ and $\epsilon^{--} = (\epsilon^-)^2$, where we defined

$$\begin{aligned} \langle \lambda \mu \rangle &= \lambda^{\alpha} \mu_{\alpha} = -\lambda_{\beta} \mu^{\beta} = -\langle \mu \lambda \rangle , \\ [\lambda \mu] &= \lambda_{\dot{\alpha}} \mu^{\dot{\alpha}} = -\mu^{\dot{\beta}} \lambda_{\dot{\beta}} = -[\mu \lambda] . \end{aligned}$$
 (3.24)

In (3.23) μ and $\tilde{\mu}$ are arbitrary spinors and the freedom in their choice reflects just the possibility of performing gauge transformation. Then one can contract the free indices in (3.21) with two graviton polarizations

$$i\mathcal{A}_{0} = -i\frac{2\kappa_{D}^{2}(k_{4}q_{2})(k_{1}q_{2})}{(q_{2}q_{3})} \left[\varepsilon_{2}\varepsilon_{3} + \frac{(\varepsilon_{3}k_{1})(\varepsilon_{2}k_{4})}{q_{2}k_{4}} + \frac{(\varepsilon_{2}k_{1})(\varepsilon_{3}k_{4})}{q_{2}k_{1}} \right] \\ \times \left[\tilde{\varepsilon}_{2}\tilde{\varepsilon}_{3} + \frac{(\tilde{\varepsilon}_{3}k_{1})(\tilde{\varepsilon}_{2}k_{4})}{q_{2}k_{4}} + \frac{(\tilde{\varepsilon}_{2}k_{1})(\tilde{\varepsilon}_{3}k_{4})}{q_{2}k_{1}} \right],$$
(3.25)

where the polarizations have been factorized according to $\epsilon_{\mu\nu} = \varepsilon_{\mu}\tilde{\varepsilon}_{\nu}$ and we rewrote the prefactor in (3.21) in terms of (3.5). In D = 4, we can use (3.23) and we have $\varepsilon^{\pm} = \tilde{\varepsilon}^{\pm}$. One can choose explicit expressions for the auxiliary spinors μ , $\tilde{\mu}$ to simplify the various scalar product: in particular it is convenient to take $\mu_2 = \lambda_3$, $\mu_3 = \lambda_2$ and similarly $\tilde{\mu}_2 = \tilde{\lambda}_3$, $\tilde{\mu}_3 = \tilde{\lambda}_2$, where $q_{i\alpha\dot{\alpha}} = \lambda_{i\alpha}\tilde{\lambda}_{i\dot{\alpha}}$. By using¹⁸

$$s_{ij} = -2z_i z_j = \langle ij \rangle [ji], \qquad (3.26)$$

for null vectors z_i , z_j , and introducing the notation

$$\langle i w j \rangle = \lambda_i^{\alpha} w_{\mu} \sigma_{\alpha \dot{\alpha}}^{\mu} \tilde{\lambda}_j^{\dot{\alpha}} , \qquad (3.27)$$

for any (not necessarily null) four-vector w^{μ} , we obtain the following results

$$\varepsilon_{2}^{+}k_{1} = -\varepsilon_{2}^{+}k_{4} = \frac{1}{\sqrt{2}} \frac{\langle 3k_{4}2 \rangle}{\langle 32 \rangle}, \quad -\varepsilon_{3}^{+}k_{1} = \varepsilon_{3}^{+}k_{4} = \frac{1}{\sqrt{2}} \frac{\langle 2k_{4}3 \rangle}{\langle 32 \rangle},$$

$$\varepsilon_{2}^{-}k_{1} = -\varepsilon_{2}^{-}k_{4} = -\frac{1}{\sqrt{2}} \frac{\langle 2k_{4}3 \rangle}{\langle 32 \rangle}, \quad \varepsilon_{3}^{-}k_{1} = -\varepsilon_{3}^{-}k_{4} = \frac{1}{\sqrt{2}} \frac{\langle 3k_{4}2 \rangle}{\langle 32 \rangle},$$

$$\varepsilon_{1}^{\pm}q_{j} = 0, \quad \varepsilon_{2}^{\pm}\varepsilon_{3}^{\pm} = \frac{[23]}{\langle 23 \rangle}, \quad \varepsilon_{2}^{\pm}\varepsilon_{3}^{-} = 0, \quad \langle 3k_{4}2 \rangle \langle 2k_{4}3 \rangle = -(2k_{2}k_{3})k_{4}^{2} + (2k_{2}k_{4})(2k_{3}k_{4}).$$
(3.28)

Then we can rewrite the tree-level amplitude separating the result for the various helicities of the gravitons we find
19

$$i\mathcal{A}^{++;++} = i\kappa_D^2 \frac{[23]^2}{(23)^2} \frac{m_1^4 t}{(s - m_1^2)(u - m_1^2)} , \qquad (3.29)$$

$$i\mathcal{A}^{++;--} = i\kappa_D^2 \frac{(3|k_4|2]^4}{t(s-m_1^2)(u-m_1^2)},$$
(3.30)

in agreement with Eq. (2.19) and (2.20) of [191]. Notice that the helicity violating amplitude $A^{++;++}$ does not have a $\frac{1}{t}$ pole since the ratio $\frac{[23]}{(23)}$ is finite as $t \to 0$. We are interested in terms dominated by the *t*-pole, which are automatically

¹⁷ As usual $\sigma^{\mu}_{\alpha\dot{\alpha}} = (1, \sigma^{i})_{\alpha\dot{\alpha}}$ and $\bar{\sigma}^{\mu\dot{\alpha}\alpha} = (1, -\sigma^{i})^{\dot{\alpha}\alpha}$ where σ^{i} are the Pauli matrices. The *SU*(2) indices are raised and lower as follows: $\lambda_{\alpha} = \epsilon_{\alpha\beta}\lambda^{\beta}$, $\lambda^{\alpha}_{\alpha} = \epsilon^{\alpha\beta}\lambda_{\beta}$ and similarly $\tilde{\lambda}^{\dot{\alpha}} = \epsilon^{\dot{\alpha}\dot{\beta}}\tilde{\lambda}_{\dot{\beta}}$, $\tilde{\lambda}_{\dot{\alpha}} = \epsilon_{\dot{\alpha}\dot{\beta}}\tilde{\lambda}^{\dot{\beta}}$ with $\epsilon^{12} = -\epsilon_{12} = 1$.

¹⁸ We have some unconventional signs since we are using the mostly plus metric and so $Tr(\sigma_{\mu}\bar{\sigma}_{\nu}) = -2\eta_{\mu\nu}$.

¹⁹ We take $s = -(k_1 + q_2)^2$, $t = -(k_1 + k_4)^2$, $u = -(k_1 + q_3)^2$.



Fig. 6. The kinematics of a two scalar, two graviton amplitude.

leading in the limit (1.74) and so only the helicity preserving structure $A^{++;--}$ survives. We can extract the pole by taking the approximation $(3|k_4|2]^4 \simeq (3|k_4|3]^4 = (s - m_1^2)^4$, since $q_2 \simeq -q_3$ in the limit of small momentum transfer. Thus the leading contribution to the amplitude with two massive scalars and two gravitons is again captured by (3.4) with of course $m_2 = 0$

$$i\mathcal{A}_0^{++;--} \simeq \frac{2i\kappa_D^2}{-t}\gamma(s,m_1,m_2=0).$$
 (3.31)

Moving then to the impact parameter space, we obtain the following eikonal phase describing the scattering of a graviton off a massive scalar

$$2\delta_0 = \frac{G_D \gamma(s, m_1, m_2 = 0)}{\sqrt{\sigma^2 - 1}} \frac{\Gamma\left(\frac{D-4}{2}\right)}{(\pi b^2)^{\frac{D-4}{2}}}.$$
(3.32)

3.1.3. The effect of the dilaton on the eikonal scattering

In the two previous subsections we considered theories in which the only massless particle is the graviton. In this subsection and in the next one, we study how the tree-level result (3.4) changes in theories with a richer spectrum of massless states. We start by considering the case where the massless spectrum, besides the graviton, includes also a massless scalar focusing in particular on the case of the string theory dilaton. In a QFT setup the dilaton is automatically included when considering a gravitational theory which is obtained via the double copy approach [192]. This technique was used in [123,193–200] to derive classical quantities in various setups. Even if we will not make extensive use of this approach, we sketch briefly the idea in an explicit example, since this has been a popular approach to derive (super)gravity amplitudes with the presence of extra massless fields like the dilaton.

One starts from a *gauge* theory with no dynamical gravity - in our example a minimally coupled massive scalar field in the adjoint representation of the gauge group. We will not need to specify the gauge group as only general properties, such as the Jacobi identities, are used. The three-point amplitude involving two scalars and a gluon is given by:

$$i\mathcal{A}_{0\,\mu}^{(3)} = igf^{a_1a_3b}(k_1 - k_3)_{\mu} = 2igf^{a_1a_3b}(k_1 - \frac{q}{2})_{\mu} , \qquad (3.33)$$

where k_1 , k_3 are the momenta of the scalars and q is the one of the gluon. We can use this ingredient to compute the four-point scalar amplitude with a gluon exchange

$$i\mathcal{A}_{0} = ig^{2} \left[f^{a_{1}a_{3}b} f^{a_{4}a_{2}b} \frac{(k_{1} - k_{3})(k_{4} - k_{2})}{(k_{1} + k_{3})^{2}} + f^{a_{1}a_{2}b} f^{a_{3}a_{4}b} \frac{(k_{1} - k_{2})(k_{3} - k_{4})}{(k_{1} + k_{2})^{2}} \right] + f^{a_{1}a_{4}b} f^{a_{2}a_{3}b} \frac{(k_{1} - k_{4})(k_{2} - k_{3})}{(k_{1} + k_{4})^{2}} \right] = ig^{2} \left[f^{a_{1}a_{3}b} f^{a_{4}a_{2}b} \frac{s - u}{(-t)} + f^{a_{1}a_{2}b} f^{a_{3}a_{4}b} \frac{u - t}{(-s)} + f^{a_{1}a_{4}b} f^{a_{2}a_{3}b} \frac{t - s}{(-u)} \right].$$
(3.34)

The basic idea of the double copy is that there exists a color-kinematics duality based on the observation that, in each term of the expression above, the color factors and the momentum dependent numerators satisfy appropriate Jacobi identities.²⁰ Indeed, the color factors obey the standard Jacobi identity:

$$f^{a_1a_3b}f^{a_4a_2b} + f^{a_1a_2b}f^{a_3a_4b} + f^{a_1a_4b}f^{a_2a_3b} = 0$$
(3.35)

and corresponding kinematic numerators

$$\begin{aligned}
f^{a_1a_3b}f^{a_4a_2b} &\longleftrightarrow \quad (k_1 - k_3)(k_4 - k_2) = s - u \\
f^{a_1a_2b}f^{a_3a_4b} &\longleftrightarrow \quad (k_1 - k_2)(k_3 - k_4) = u - t \\
f^{a_1a_4b}f^{a_2a_3b} &\longleftrightarrow \quad (k_1 - k_4)(k_2 - k_3) = t - s
\end{aligned}$$
(3.36)

 $^{^{20}}$ For general amplitudes it is not straightforward to write the result that makes this feature manifest.

satisfy a "kinematic" Jacobi identity, i.e. their sum vanishes. By following the double copy approach, one starts from (3.34) and obtains a gravitational amplitude by substituting the color factors with the corresponding momentum-dependent factors (3.36). This yields

$$i\mathcal{A}_0 = i\left(\frac{\kappa_D}{2}\right)^2 \left[\frac{(u-s)^2}{(-t)} + \frac{(u-t)^2}{(-s)} + \frac{(s-t)^2}{(-u)}\right],$$
(3.37)

where we mapped $g \rightarrow \kappa_D/2$. This result is appropriate to describe the interaction among identical external states and, for the case of two different scalars considered in (3.3), we can focus just on the first term in (3.37). By using $u = -s - t + 2m_1^2 + 2m_2^2$, it is easy to see that even the 1/t part does not match the pure gravity result. The reason for this is that the gravitational theory obtained by this construction contains extra massless fields, in addition to the graviton²¹: a massless antisymmetric tensor and, crucially for us, a dilaton whose coupling to scalars is proportional to square of the mass.

In summary, the long-range interaction between two massive scalars in a theory containing both gravitons and dilatons takes the same form as (3.4), but with a different kinematic factor

$$\gamma^{(\text{dil})}(s, m_i) \equiv \frac{1}{2}(s - m_1^2 - m_2^2)^2 .$$
 (3.38)

Thus the tree amplitude to be used to construct the leading eikonal is

$$\mathcal{A}_0(s, -q^2) = \frac{32\pi m_1^2 m_2^2 G_D \sigma^2}{q^2} + \mathcal{O}((q^2)^0)$$
(3.39)

instead of the one in (3.4). Then, by following the usual steps, we obtain the leading eikonal

$$2\delta_0 = \frac{2m_1m_2G_D\sigma^2\Gamma\left(\frac{D-4}{2}\right)}{\sqrt{\sigma^2 - 1}(\pi b^2)^{\frac{D-4}{2}}}.$$
(3.40)

It is easy to isolate the contribution due to the dilaton exchanges and

$$\mathcal{A}_0(s, -q^2) = \frac{32\pi m_1^2 m_2^2 G_D}{(D-2)q^2}$$
(3.41)

from which we can also read off the dilaton-scalar-scalar coupling,

$$\mathcal{A}_0^{(3)} = -\frac{2\kappa_D m_{1,2}^2}{\sqrt{D-2}}\,.\tag{3.42}$$

We can check the interpretation above simply by adding to the initial Lagrangian (3.1) a dilaton φ

$$S = \int d^{D}x \sqrt{|G|} \left\{ \frac{R}{2\kappa_{D}^{2}} - \frac{1}{2} \left[\partial_{\mu}\varphi \partial_{\nu}\varphi G^{\mu\nu} + \sum_{i=1}^{2} \left(\partial_{\mu}\phi_{i}\partial_{\nu}\phi_{i}G^{\mu\nu} + m_{i}^{2}e^{\frac{2\kappa_{D}\varphi}{\sqrt{D-2}}}\phi_{i}^{2} \right) \right] \right\}.$$
(3.43)

Then it is straightforward that the exchange diagram involving the dilaton yields an extra contribution canceling the final term in (3.5) and thus reproducing (3.38).

3.1.4. Maximally supersymmetric gravity

We can further enrich the theory including extra fields. An interesting case is that of supergravities whose perturbative expansion has been studied in great details at high loop order [201–205]. Focusing on the maximal supersymmetric case, the four point amplitude among massless states is determined by a single scalar function even in a string theory setup [206]. In the field theory limit we have

$$A_0 = 8\pi G \frac{\mathcal{K}}{stu} , \qquad (3.44)$$

where the kinematic prefactor \mathcal{K} can be written as a pair of 4-index tensors K appearing also in the corresponding string amplitude, see for instance Eq. (7.4.57) of [206] and Section 7.4.2 of the same reference for the explicit expressions of K (see also [109]). It is convenient to organize the graviton multiplet in terms of on-shell realization of supersymmetry, see [207] for a realization relevant to D = 4 and [208] for higher dimensional cases. The basic idea is that \mathcal{K} encodes a super-momentum conserving delta function $\delta^{(16)}(Q)$ involving the sum of the supercharges of the external states.

In D = 4, all $2^8 = 256$ states of the $\mathcal{N} = 8$ supermultiplet can be packaged into the on-shell superfield

$$\Phi(\eta) = h^{++} + \eta^{A}\psi_{A} + \dots + \eta^{A}\eta^{B}\eta^{C}\eta^{D}\varphi_{ABCD} + \dots + \eta^{1}\eta^{2}\eta^{3}\eta^{4}\eta^{5}\eta^{6}\eta^{7}\eta^{8}h^{--}, \qquad (3.45)$$

²¹ Notice that the double copy amplitude contains also a contribution linear in t which is absent in (3.3). This signals that the action contains an extra contact term interaction between the four scalars.

where η^A denotes auxiliary Grassmann variables that saturate the *R*-symmetry indices *A*, *B*, *C*, ... (taking values from 1 to 8) of the various fields. For instance, $h^{\pm\pm}$ denote the positive/negative-helicity gravitons h^{++} , while φ_{ABCD} collect the 70 scalars present in the theory. In this formalism, one can write a 2 \rightarrow 2 "super-amplitude", which concisely encodes all 2 \rightarrow 2 scattering amplitudes among any four given states of the supermultiplet, and takes the following very compact form [167,209]

$$A_0 = \frac{\kappa^2}{stu} \frac{[34]^4}{(12)^4} \,\delta^{(16)}(Q)\,, \tag{3.46}$$

where the "super-momentum" conserving delta function is given by

$$\delta^{(16)}(Q) = \frac{1}{2^4} \prod_{A=1}^8 \sum_{i,j=1}^4 \langle ij \rangle \, \eta_i^A \eta_j^A \,. \tag{3.47}$$

The super-amplitude (3.46) should be thought of as a function of four copies of the on-shell superfield (3.45), $\Phi(\eta_j)$, with i = 1, 2, 3, 4 labeling the states as in Fig. 7, and of course

$$s = -(p_1 + p_2)^2$$
, $t = -(p_1 + p_4)^2$, $u = -(p_1 + p_3)^2$. (3.48)

We consider three possible situations. The first one is given by the choice

$$\phi_1 = \phi_2 = \varphi_{1234} \equiv \tau , \qquad \phi_3 = \phi_4 = \varphi_{5678} \equiv \bar{\tau} . \tag{3.49}$$

The idea is to focus on a complex scalar τ (corresponding for instance to the axio-dilaton of type IIB via dimensional reduction) and its complex conjugate. In order to extract the contribution due to the corresponding states from the super-amplitude (3.46), one should take the appropriate derivatives with respect to the auxiliary Grassmann variables as dictated by the way their fields feature in the superfield (3.45). For our case, since

$$\tau = \partial \Phi(\eta) \Big|_{\eta=0}, \qquad \bar{\tau} = \bar{\partial} \Phi(\eta) \Big|_{\eta=0}$$
(3.50)

with

$$\partial \equiv \prod_{A=1}^{4} \frac{\partial}{\partial \eta^{A}}, \qquad \bar{\partial} \equiv \prod_{A=5}^{8} \frac{\partial}{\partial \eta^{A}}, \qquad (3.51)$$

we need to calculate

The differential operators, which could be also interpreted as Grassmann integrals, only act on $\delta^{(16)}(Q)$ and it is easy to see that the result is simply

$$\partial_1 \partial_2 \bar{\partial}_3 \bar{\partial}_4 \delta^{(16)}(\mathbf{Q})\Big|_{n=0} = \langle 12 \rangle^4 \langle 34 \rangle^4 \,. \tag{3.53}$$

In this way, we obtain

$$\mathcal{A}_0^{\tau\tau \to \bar{\tau}\bar{\tau}} = \frac{\kappa^2 s^4}{stu} \,, \tag{3.54}$$

which is of course symmetric in the exchange $1 \leftrightarrow 2$. Similarly, for the choice

$$\phi_1 = \phi_4 = \varphi_{1234} \equiv \tau , \qquad \phi_2 = \phi_3 = \varphi_{5678} \equiv \bar{\tau} , \tag{3.55}$$

we have

$$\mathcal{A}_0^{\tau\bar{\tau}\to\tau\bar{\tau}} = \frac{\kappa^2 t^4}{stu},\tag{3.56}$$

and for the choice

$$\phi_1 = \phi_3 = \varphi_{1234} \equiv \tau \,, \qquad \phi_2 = \phi_4 = \varphi_{5678} \equiv \bar{\tau} \,, \tag{3.57}$$

we have

$$\mathcal{A}_0^{\tau\bar{\tau}\to\bar{\tau}\tau} = \frac{\kappa^2 u^4}{stu} \,. \tag{3.58}$$

Eqs. (3.56), (3.58) can be deduced directly from (3.54) by crossing symmetry in the *t* and *u*-channel respectively, although it is instructive to derive them from the super-amplitude (3.46) by applying the appropriate differential operator,

$$\mathcal{A}_{0}^{\tau\bar{\tau}\to\tau\bar{\tau}} = \partial_{1}\bar{\partial}_{2}\bar{\partial}_{3}\partial_{4}A_{0}\Big|_{\eta=0}, \qquad \mathcal{A}_{0}^{\tau\bar{\tau}\to\bar{\tau}\tau} = \partial_{1}\bar{\partial}_{2}\partial_{3}\bar{\partial}_{4}A_{0}\Big|_{\eta=0}.$$
(3.59)



Fig. 7. Scattering of four states in $\mathcal{N} = 8$ supergravity.

Yet another choice is to consider linear combinations of the above states, for instance an axion and a dilaton, as both in- and out-states,

$$\phi_1 = \phi_4 = \frac{\tau + \bar{\tau}}{\sqrt{2}}, \qquad \phi_2 = \phi_3 = \frac{\tau - \bar{\tau}}{i\sqrt{2}}.$$
(3.60)

The corresponding amplitude is given by

$$\mathcal{A}_{0} = -\frac{1}{4} (\partial_{1} + \bar{\partial}_{1})(\partial_{2} - \bar{\partial}_{2})(\partial_{3} - \bar{\partial}_{3})(\partial_{4} + \bar{\partial}_{4})A_{0}\Big|_{\eta=0}$$
(3.61)

or, expanding the derivatives and recognizing the amplitudes calculated above,

$$\mathcal{A}_{0} = \frac{1}{2} \left(\mathcal{A}_{0}^{\tau\tau \to \tau\bar{\tau}} - \mathcal{A}_{0}^{\tau\bar{\tau} \to \tau\bar{\tau}} + \mathcal{A}_{0}^{\tau\bar{\tau} \to \bar{\tau}\tau} \right)$$
(3.62)

and finally

$$\mathcal{A}_0 = \kappa^2 \frac{s^4 - t^4 + u^4}{2stu} \,. \tag{3.63}$$

In this way, by construction, the result is $s \leftrightarrow u$ symmetric. This will be useful in Section 6 since the $s \leftrightarrow u$ symmetry simplifies the analysis of the analytic properties of this amplitudes at higher loops. At tree level, of course (3.63) coincides with (3.54) up to irrelevant analytic terms as $t \rightarrow 0$ that correspond to short-range contributions in *b*-space.

One can introduce masses for the scalars by means of a Kaluza–Klein compactification. To this end, we can re-interpret (3.63) as (say) 10-dimensional, and perform a toroidal compactification of 6 spatial dimensions, letting

$$p_1^M = (p_1^\mu, 0, \dots, 0, 0, m_1), \qquad p_4^M = (p_4^\mu, 0, \dots, 0, 0, -m_1), p_2^M = (p_2^\mu, 0, \dots, 0, m_2, 0), \qquad p_3^M = (p_3^\mu, 0, \dots, 0, -m_2, 0),$$
(3.64)

in such a way that

$$0 = p_1^M p_{1M} = p_1^\mu p_{1\mu} + m_1^2, \qquad 0 = p_2^M p_{2M} = p_2^\mu p_{2\mu} + m_2^2$$
(3.65)

and similarly for p_3 , p_4 . Likewise, from the 4D perspective,

$$-2p_1 \cdot p_2 = s - m_1^2 - m_2^2, \qquad -(p_1 + p_4)^2 = t, \qquad -2p_1 \cdot p_3 = u - m_1^2 - m_2^2, \qquad (3.66)$$

and therefore

$$\mathcal{A}_{0} = \kappa^{2} \frac{(s - m_{1}^{2} - m_{2}^{2})^{4} - t^{4} + (u - m_{1}^{2} - m_{2}^{2})^{4}}{2(s - m_{1}^{2} - m_{2}^{2})t(u - m_{1}^{2} - m_{2}^{2})} \,.$$
(3.67)

Calculating the leading eikonal by going to impact-parameter space is now straightforward, and neglecting contributions that lack the 1/t pole, one finds again the result

$$2\delta_0 = \frac{2m_1m_2G_D\sigma^2\Gamma\left(\frac{D-4}{2}\right)}{\sqrt{\sigma^2 - 1}(\pi b^2)^{\frac{D-4}{2}}},$$
(3.68)

which we had obtained in (3.40) when discussing the $\mathcal{N} = 0$ theory emerging from the double copy. Therefore, as far as the long-range effects captured by the leading eikonal exponentiation are concerned, switching on supergravity is equivalent to taking into account the additional dilaton exchanges between the two massive particles.²²

²² In the approach above, we assumed that the compactification scale is much smaller than the classical length scales relevant to the binary, such as the impact parameter and the effective size *GE* of the colliding objects. Thus, long-range effects are captured just by the exchange of states with zero Kaluza–Klein numbers yielding for instance Eq. (3.68). See [210] for a treatment in which the dynamics of the Kaluza–Klein modes is included in an eikonal context.

P. Di Vecchia, C. Heissenberg, R. Russo et al.

Following the familiar steps, we can take $-p\frac{\partial}{\partial b}$ of (3.68) to obtain the leading deflection angle

$$\Theta = \frac{4GE(\pi b^2)^{\epsilon} \sigma^2 \Gamma(1-\epsilon)}{(\sigma^2 - 1)b} + \mathcal{O}(G^2) \xrightarrow[D \to 4]{} \frac{4GE\sigma^2}{(\sigma^2 - 1)b} + \mathcal{O}(G^2).$$
(3.69)

Let us conclude by looking again at the probe limit $m_1 \ll m_2$, in which $E \simeq m_1 = M$ and $\sigma \simeq E_p/m_p$, so that from (3.69) we find

$$\Theta = \frac{4GM(\pi b^2)^{\epsilon} \sigma^2 \Gamma(1-\epsilon)}{(\sigma^2 - 1)b} + \mathcal{O}(G^2) \xrightarrow{D \to 4} \frac{4GM\sigma^2}{(\sigma^2 - 1)b} + \mathcal{O}(G^2)$$
(3.70)

in agreement with (B.41) in D = 4 to leading order in *G*. In fact, show that the 1PM term determines completely the probe limit of the maximally supersymmetric case in four dimension [211], and the result can be obtained by proceeding as we did in Section 2.2. Starting from the phase shift determined by (3.68),

$$2\delta_j = \chi(J) = -\frac{2m_1 m_2 G \sigma^2}{\sqrt{\sigma^2 - 1}} \log J, \qquad (3.71)$$

one obtains an equation analogous to (2.54),

$$\tan\frac{\Theta}{2} = \frac{2m_1m_2G\sigma^2}{\sqrt{\sigma^2 - 1}}\frac{1}{J} \simeq \frac{2MG\sigma^2}{b(\sigma^2 - 1)},$$
(3.72)

where we have used that J = pb and, in the probe limit, $p \simeq m_p \sqrt{\sigma^2 - 1}$, thus recovering (B.41).

3.1.5. Effective field theories beyond GR

All cases discussed so far focus on gravitational theories whose action contains just two-derivative terms. At large distance this is certainly a reliable approximation, but it is interesting to see how higher derivative corrections can modify high energy scattering. By following [154] we start discussing the case where the standard gravitational action is modified by quadratic and cubic terms schematically as follows

$$S_{\rm gr} = \frac{1}{2\kappa_D^2} \int d^D x \sqrt{|G|} \left[R + l_2^2 R^2 + l_4^4 R^3 \right] , \qquad (3.73)$$

where l_i are the length scales determining when the new higher derivative terms start becoming relevant, which we take to be parametrically larger than ℓ_P . From an effective field theory point of view, it may appear strange to assume that the l_i are decoupled from ℓ_P , but this is quite natural in setups where the modification in (3.73) arise from a *classical* microscopic dynamics, as in string theory, or from integrating out non-gravitational degrees of freedom as in [169,175]. In explicit string constructions the scales l_i can be written in terms of the string length ℓ_s , but in this section we will work within an effective field theory approach and keep them arbitrary just assuming $l_i \gg \ell_P$. It is interesting to focus on the corrections in (3.73) because they are the only ones that can modify the on-shell 3-graviton amplitude. Strictly speaking this amplitude vanishes because the on-shell conditions force the momenta of the external states to be collinear. However one can either complexify the momenta or work in a signature with two times to define a non-trivial on-shell 3-graviton amplitude. So from (3.73) we have

$$A_3 = A_3^{(0)} + A_3^{(2)} + A_3^{(4)} , (3.74)$$

where $A_3^{(0)}$ is the contribution from the Einstein–Hilbert part of the action one and the other two terms are higher derivative corrections with two and four additional derivatives respectively. The Einstein–Hilbert contribution reads

$$iA_{3}^{(0)} = -2i\kappa_{D}\left[(\varepsilon_{1}\varepsilon_{2})(\varepsilon_{3}q_{1}) + \text{cycl.}\right]\left[(\tilde{\varepsilon}_{1}\tilde{\varepsilon}_{2})(\tilde{\varepsilon}_{3}q_{1}) + \text{cycl.}\right],$$
(3.75)

where $\varepsilon_{i\mu}\tilde{\varepsilon}_{i\nu} = h_{i\mu\nu}$ is a symmetric tensor representing the polarization of one of the external gravitons and in each parenthesis one needs to sum over the cyclical permutations. The correction proportional to l_2^2 reads

$$iA_3^{(2)} = 2i\kappa_D l_2^2 \left\{ [(\varepsilon_1 \varepsilon_2)(\varepsilon_3 q_1) + \text{cycl.}][(\tilde{\varepsilon}_1 q_2)(\tilde{\varepsilon}_2 q_3)(\tilde{\varepsilon}_3 q_1)] + \varepsilon \leftrightarrow \tilde{\varepsilon} \right\},$$
(3.76)

while the one proportional to l_{4}^{4} is

$$iA_3^{(4)} = 2i\kappa_D l_4^4 \left[(\varepsilon_1 q_2)(\varepsilon_2 q_3)(\varepsilon_3 q_1) \right] \left[(\tilde{\varepsilon}_1 q_2)(\tilde{\varepsilon}_2 q_3)(\tilde{\varepsilon}_3 q_1) \right] .$$
(3.77)

Since these on-shell couplings are written in the factorized form following from the KLT or the double copy relations, one can immediately see that they are obtained by taking the products of the space–time part of the standard 3-gluon coupling and its higher derivative modification related to a $Tr(F^3)$ in the Lagrangian. In the supersymmetric theories the modification in (3.77) is not allowed and in the maximally supersymmetric case also (3.76) is set to zero and so, in this case, the 3-point vertex cannot be deformed. In the case of pure gravity in D = 4, the modification (3.76) does not contribute to the eikonal since it can be derived by the Gauss–Bonnet term which is a total derivative in four dimension.

In the following we will keep *D* generic and we will not decouple the dilaton and the antisymmetric tensor arising from the double copy construction.

We can now discuss how the higher derivative corrections modify the tree-level amplitude (3.21) describing scattering of a graviton off a massive scalar. We focus in particular on the high-energy limit and look for terms that scale at least as the factor of $\gamma(s, m_1, m_2 = 0)$ present in the Einstein–Hilbert case (3.31). In order to do this we can simply combine, by using a de Donder propagator (A.23), an on-shell 3-graviton vertex obtained from (3.74) and the vertex with the external scalars (A.25). As a warming up exercise one can derive (3.31) from (3.75): by combining the scalar vertex with the de Donder propagator, one obtains

$$A_{\mu\nu}^{(1)} = -\frac{\kappa_D}{q^2} \left(k_{1\mu} k_{4\nu} + k_{1\nu} k_{4\mu} - \eta_{\mu\nu} \frac{2m_1^2}{D-2} \right) , \qquad (3.78)$$

which can be saturated with the variation of (3.75) with respect to $\epsilon_{1\mu\nu}$ (identifying that graviton as the particle exchanged in the *t*-channel). Only the term proportional to $q_2\epsilon_1q_2$ can produce a leading contribution in the energy since, when saturated with (3.78), it yields the scalar product $2(k_1q_2)(k_4q_2) \simeq -\gamma(s, m_1^2, m_2 = 0)$ where in the last step we neglected a term proportional to *t*. This reproduces exactly the result (3.31). One can follow the same approach by using (3.76) which provides the corrections proportional to l_2^2 to the scattering discussed above. Again in this effective vertex there is a term proportional to $q_2\epsilon_1q_2$ which for the same reasons above provides a contribution that scales with the energy exactly in the same way as the term coming from the standard Einstein–Hilbert 3-graviton vertex. The main qualitative difference that we find in this case is that the polarization of the external graviton appear dotted with the exchange momentum q

$$i\mathcal{A}_0 \sim \frac{2i\kappa_D^2}{-t}\gamma(s, m_1, m_2 = 0)(2l_2^2)[q\epsilon_1\epsilon_2 q].$$
 (3.79)

By following similar steps, one can obtain the correction related to l_4

$$i\mathcal{A}_0 \sim -\frac{2i\kappa_D^2}{-t}\gamma(s, m_1, m_2 = 0)(2l_4^4)\left[(q\epsilon_1 q)(q\epsilon_2 q)\right].$$
(3.80)

Seen as contributions to a tree-level amplitude the results in (3.79) and (3.80) do not have anything unusual, but since they behave as (3.31) in the limit (1.74) we expect them to exponentiate and contribute to leading *classical* eikonal. As before we rewrite the result in impact parameter space:

$$2\delta_{0} = \frac{G_{D}\gamma(s, m_{1}, m_{2} = 0)}{\sqrt{\sigma^{2} - 1}} \frac{\Gamma\left(\frac{D-4}{2}\right)}{(\pi b^{2})^{\frac{D-4}{2}}} \left[\epsilon_{1ij}\epsilon_{2ij} + (D-4)\frac{2l_{2}^{2}}{b^{2}}\epsilon_{1ij}\Pi_{jh}\epsilon_{2hi} - (D-2)(D-4)\frac{l_{4}^{4}}{b^{4}}\epsilon_{1ih}\epsilon_{2jk}\Pi_{ijhk} \right],$$
(3.81)

where we labeled the physical polarizations with the indices i, j, \ldots of the transverse space and

$$\Pi_{ij} = \delta_{ij} - (D-2)\frac{b_i b_j}{b^2} ,$$

$$\Pi_{ijhk} = \delta_{hk}\delta_{ij} + \delta_{hj}\delta_{ik} + \delta_{jk}\delta_{ih} - \frac{D}{b^2} \left(b_h b_k \delta_{ij} + b_h b_j \delta_{ik} + b_i b_k \delta_{ih} + b_i b_k \delta_{jh} + b_i b_j \delta_{hk} - (D+2)\frac{b_i b_j b_h b_k}{b^2} \right) .$$

$$(3.82)$$

Of course the first term in (3.81) is just the Einstein–Hilbert contribution of Eq. (3.32) which is identical for all the polarizations of the graviton. Notice, on the contrary, that the corrections related to l_2 and l_4 are not universal and depend on whether the polarizations involved have a trivial projection (or not) along the direction *b* of the impact parameter. The main novelty of the eikonal (3.81) with respect to the cases discussed so far is that it acts non-trivially in the space of the polarizations of the incident massless particles. Thus, instead of being a phase it becomes an operator mixing different helicities of the graviton, the dilaton and the antisymmetric tensor. As discussed in Section 4.2.4 this is the origin of causality violating contributions in the eikonal scattering for these modified theories when the impact parameter becomes of the order of $l_{2,4}$ [154].

3.1.6. Including classical spin

In this subsection we generalize the calculation of the leading eikonal phase to the case in which the massive particles also have a "classical spin", i.e. an intrinsic angular momentum as opposed to the one associated to their orbital motion. At the level of solutions of Einstein's equations, introducing spin corresponds to going from the Schwarzschild solution, characterized by its mass, to the Kerr solution describing a black hole with both a mass *m* and spin J_a , provided the inequality $J_a \leq Gm^2$ is satisfied (with the equality sign corresponding to the "extremal" case). Both the Schwarzschild and Kerr solutions involve the full non-linear structure of GR.

One can, however, construct a linearized version of the Kerr black hole by keeping in the GR Lagrangian only the kinetic term of the gravitational field and a term that describes its interaction with the energy–momentum tensor of the spinning matter. Then, from it, one can extract the three-point amplitude involving two massive particles with spin and a graviton [59,60,63] (the round parenthesis indicates symmetrization without additional factors, $A^{(\alpha}B^{\beta)} = A^{\alpha}B^{\beta} + A^{\beta}B^{\alpha}$):

$$\tau^{\mu\nu}(p,p',k;a) = \kappa_D \bar{p}^{(\mu} \exp(i\epsilon^{\nu)}{}_{\rho\alpha\beta} a^{\alpha} k^{\beta}) \bar{p}^{\rho} , \qquad (3.83)$$

where all three momenta entering the vertex are regarded as outgoing, and can take complex values in order to obey both momentum conservation and the mass shell relations,

$$p^{\mu} + p'^{\mu} + k^{\mu} = 0$$
, $\frac{1}{2} \left(p'^{\mu} - p^{\mu} \right) = \bar{p}^{\mu}$, $p^2 = -m^2 = p'^2$, $k^2 = 0 = \bar{p} \cdot k$. (3.84)

The effective vertex (3.83) involves the spin vector a^{μ} of the massive object. This is related to the spin tensor $S^{\mu\nu}$ through the following relations,²³

$$a^{\rho} = -\frac{1}{2m^2} \epsilon^{\rho\alpha\beta\gamma} \bar{p}_{\alpha} S_{\beta\gamma} , \qquad S^{\mu\nu} = \epsilon^{\mu\nu\rho\sigma} \bar{p}_{\rho} a_{\sigma}$$
(3.85)

and satisfies in particular

$$a \cdot \bar{p} = 0, \qquad J_a = ma, \tag{3.86}$$

where J_a is the magnitude of the spin angular momentum, as can be seen by going to the rest frame. Of course, in the $a_i \rightarrow 0$ limit, Eq. (3.83) reduces to the non-spinning effective vertex (3.6).

Using the above properties as well as the identities involving $\epsilon_{\mu\nu\rho\sigma}$ collected for convenience in Appendix C, one can show that

$$\epsilon^{\nu}{}_{\mu\alpha\beta}a^{\alpha}k^{\beta}\epsilon^{\mu}{}_{\rho\gamma\delta}a^{\gamma}k^{\delta}\ \bar{p}^{\rho} = -(a\cdot k)^{2}\delta^{\nu}{}_{\rho}\ \bar{p}^{\rho}.$$
(3.87)

As a result, the matrix exponential in (3.83) can be rewritten in the more explicit form

$$e^{i\epsilon^{\nu}{}_{\rho\alpha\beta}a^{\alpha}k^{\beta}}\bar{p}^{\rho} = \left[\cosh(a\cdot k)\delta^{\nu}{}_{\rho} + i\frac{\sinh(a\cdot k)}{a\cdot k}\epsilon^{\nu}{}_{\rho\alpha\beta}a^{\alpha}k^{\beta}\right]\bar{p}^{\rho}, \qquad (3.88)$$

and the vertex reads

$$\tau^{\mu\nu}(p,p',k;a) = i\kappa \left[\cosh(a \cdot k) 2\bar{p}^{\mu} \bar{p}^{\nu} + i \frac{\sinh(a \cdot k)}{a \cdot k} \left(\bar{p}^{\mu} \epsilon^{\nu}{}_{\rho\alpha\beta} a^{\alpha} k^{\beta} \bar{p}^{\rho} + \bar{p}^{\nu} \epsilon^{\mu}{}_{\rho\alpha\beta} a^{\alpha} k^{\beta} \bar{p}^{\rho} \right) \right].$$
(3.89)

We note that the trace of this vertex only comes from the cosh part,

$$\eta_{\mu\nu}\tau^{\mu\nu}(p,p',k;a) = 2i\kappa\bar{p}^2\cosh(a\cdot k).$$
(3.90)

We can now sew together two copies of the on-shell vertex (3.89), describing each the motion of a distinct particle labeled by 1 and 2, with the de Donder propagator (A.23), obtaining the following $2 \rightarrow 2$ amplitude for the scattering of two massive spinning objects,

$$i\mathcal{A}_0 = \tau^{\mu\nu}(p_1, p_4, -q; a_1)G_{\mu\nu,\rho\sigma}(q)\tau^{\rho\sigma}(p_2, p_3, q; a_2).$$
(3.91)

Here of course q^2 is not zero, unlike for k^{μ} in (3.83). Therefore, the expression (3.91) is only accurate up to contact terms, i.e. contributions whose residue at $q^2 \rightarrow 0$ vanishes, which only contribute to short-range effects. Vice-versa, the relations

$$\bar{p}_{1,2} \cdot q = 0, \quad \bar{p}_1 \cdot a_1 = 0, \quad \bar{p}_2 \cdot a_2 = 0$$
(3.92)

still hold and we may now consider real kinematics. Writing explicitly (3.91) we get

$$\mathcal{A}_{0} = \frac{2\kappa^{2}}{q^{2}} \left[\cosh(a_{1}q)\bar{p}_{1}^{\mu}\bar{p}_{1}^{\nu} - \frac{i}{2} (\epsilon^{\mu}{}_{\gamma\alpha\beta}\bar{p}_{1}^{\nu} + \epsilon^{\nu}{}_{\gamma\alpha\beta}\bar{p}_{1}^{\mu})\bar{p}_{1}^{\gamma}a_{1}^{\alpha}q^{\beta}\frac{\sinh(a_{1}q)}{qa_{1}} \right]$$

$$\times (2\eta_{\mu\rho}\eta_{\nu\sigma} - \eta_{\mu\nu}\eta_{\rho\sigma}) \left[\cosh(a_{2}q)\bar{p}_{2}^{\rho}\bar{p}_{2}^{\sigma} + \frac{i}{2} (\epsilon^{\rho}{}_{\lambda\delta\eta}\bar{p}_{2}^{\sigma} + \epsilon^{\sigma}{}_{\lambda\delta\eta}\bar{p}_{2}^{\rho})\bar{p}_{\lambda}^{\lambda}a_{2}^{\delta}q^{\eta}\frac{\sinh(a_{2}q)}{qa_{2}} \right].$$

$$(3.93)$$

The term with $cosh(qa_1)cosh(qa_2)$ can be easily computed and one gets

$$\frac{2\kappa^2}{q^2}m_1^2m_2^2(2\sigma^2 - 1)\cosh(qa_1)\cosh(qa_2), \qquad (3.94)$$

²³ To leading order in k^{μ} , we may disregard the difference between \bar{p}^{μ} , p'^{μ} and $-p'^{\mu}$.

For the computation of the term with $\sinh(qa_1)\sinh(qa_2)$ we can take advantage of the fact that only the first term of the de Donder propagator contributes (the second term vanishes according to the observation before (3.90)) and one gets:

$$\frac{2\kappa^{2}}{q^{2}}\frac{\sinh(a_{1}q)}{qa_{1}}\frac{\sinh(a_{2}q)}{qa_{2}}$$

$$\times\left(-(\epsilon_{\gamma\sigma\alpha\beta}\bar{p}_{1}^{\gamma}\bar{p}_{2}^{\sigma}a_{1}^{\alpha}q^{\beta})(\epsilon_{\rho\lambda\delta\eta}\bar{p}_{1}^{\rho}\bar{p}_{2}^{\lambda}a_{2}^{\delta}q^{\eta})+(p_{1}p_{2})\epsilon_{\gamma\alpha\beta}^{\rho}\bar{p}_{1}^{\gamma}a_{1}^{\alpha}q^{\beta}\epsilon_{\rho\lambda\delta\eta}\bar{p}_{2}^{\lambda}a_{2}^{\delta}q^{\eta}\right)$$
(3.95)

The second term in the big round parenthesis can be computed using the relation

$$\epsilon^{\rho}{}_{\gamma\alpha\beta}\bar{p}_{1}^{\gamma}a_{1}^{\alpha}q^{\beta}\epsilon_{\rho\lambda\delta\eta}\bar{p}_{2}^{\lambda}a_{2}^{\delta}q^{\eta} = (\bar{p}_{1}\cdot\bar{p}_{2})(a_{1}\cdot q)(a_{2}\cdot q) + (a_{1}\cdot\bar{p}_{2}a_{2}\cdot\bar{p}_{1}-\bar{p}_{1}\cdot\bar{p}_{2}a_{1}\cdot a_{2})q^{2}$$
(3.96)

that can be obtained from the formulas in Appendix C, while for the first term we need the following relation

$$-(\epsilon_{\mu\nu\alpha\beta}\bar{p}_{1}^{\mu}\bar{p}_{2}^{\nu}a_{1}^{\alpha}q^{\beta})(\epsilon_{\mu\nu\alpha\beta}\bar{p}_{1}^{\mu}\bar{p}_{2}^{\nu}a_{2}^{\alpha}q^{\beta}) = \left[(\bar{p}_{1}\cdot\bar{p}_{2})^{2} - \bar{p}_{1}^{2}\bar{p}_{2}^{2}\right]\left(a_{1}\cdot qa_{2}\cdot q - a_{1}\cdot a_{2}q^{2}\right) + \bar{p}_{1}\cdot\bar{p}_{2}a_{1}\cdot\bar{p}_{2}a_{2}\cdot\bar{p}_{1}q^{2}$$
(3.97)

Actually any term proportional to q^2 in (3.96) and (3.97) can be neglected because it will give an analytic piece. This means that the term with $\sinh(qa_1)\sinh(qa_2)$ is equal to

$$\frac{2\kappa_D^2}{q^2}m_1^2m_2^2(2\sigma^2 - 1)\sinh(qa_1)\sinh(qa_2).$$
(3.98)

Finally, the two mixed terms can be computed similarly, and one gets the following final result

$$\mathcal{A}_{0} = \frac{2\kappa^{2}m_{1}^{2}m_{2}^{2}\sigma}{q^{2}} \left[\sigma(1+v^{2})\cosh(a \cdot q) - 2i\epsilon_{\mu\nu\alpha\beta}v_{1}^{\mu}v_{2}^{\nu} \left(a_{2}^{\alpha}q^{\beta}\frac{\sinh(a_{2} \cdot q)}{a_{2} \cdot q} \cosh(a_{1} \cdot q) + a_{1}^{\alpha}q^{\beta}\frac{\sinh(a_{1} \cdot q)}{a_{1} \cdot q} \cosh(a_{2} \cdot q) \right) \right],$$

$$(3.99)$$

where the term in the first line comes from the sum of the terms in (3.94) and (3.98) and the identity $\sigma^2(1+v^2) = 2\sigma^2 - 1$. In the previous equation we used

$$\bar{p}_1^{\mu} \simeq m_1 v_1^{\mu}, \qquad \bar{p}_2^{\mu} \simeq m_2 v_2^{\mu}, \qquad \sigma = -v_1 \cdot v_2$$
(3.100)

and introduced

- 2

$$a^{\mu} = a_1^{\mu} + a_2^{\mu}, \qquad \sigma = \frac{1}{\sqrt{1 - v^2}}.$$
 (3.101)

In the center of mass,

$$\bar{p}_1^{\mu} \simeq (E_1, \vec{p}), \qquad \bar{p}_2^{\mu} \simeq (E_2, -\vec{p}), \qquad q^{\mu} = (0, \vec{q}),$$
(3.102)

letting

$$= (0, 0, p), \qquad \vec{q} = (\mathbf{q}, 0) = (q^{x}, q^{y}, 0), \qquad (3.103)$$

we have

 \vec{p}

$$\epsilon_{\mu\nu\alpha\beta}\bar{p}_1^{\mu}\bar{p}_2^{\nu}a_1^{\alpha}q^{\beta} = -m_1m_2\sigma v(\hat{p}\times\vec{a}_1)\cdot\vec{q}\,,\tag{3.104}$$

so that (3.99) can be rewritten as follows

$$\mathcal{A}_{0} = \frac{2\kappa^{2}m_{1}^{2}m_{2}^{2}\sigma^{2}}{q^{2}} \left[(1+v^{2})\cosh(a \cdot q) + 2iv \left((\hat{p} \times \vec{a}_{2}) \cdot \vec{q} \frac{\sinh(a_{2} \cdot q)}{a_{2} \cdot q} \cosh(a_{1} \cdot q) + (\hat{p} \times \vec{a}_{1}) \cdot \vec{q} \frac{\sinh(a_{1} \cdot q)}{a_{1} \cdot q} \cosh(a_{2} \cdot q) \right) \right].$$
(3.105)

Comparing (3.97) and (3.104), we see that

$$(a_1 \cdot q)^2 = a_1^2 q^2 - ((\hat{p} \times \vec{a}_1) \cdot \vec{q})^2.$$
(3.106)

We note that taking a square root of (3.106) and setting $q^2 = 0$, one can effectively replace

$$a_1 \cdot q \to \pm i(\hat{p} \times \vec{a}_1) \cdot \vec{q}$$
 (3.107)

in (3.99) at the price of short-range corrections (similarly for a_2). One can check this in the $m_2 = 0$ case [75] by writing $q^2 = (p_2 + p_3)^2$ in terms of the spinor formalism as in (3.26). Then one can extract the leading term for small q by sending either (23) or [23] to zero. This choice is related to the potential sign ambiguity introduced by the square root (3.107),

see [75] for a discussion of this point. However this sign is immaterial in (3.105) and the terms in the second line of that equation recombine via $\sinh \alpha \cosh \beta + \sinh \beta \cosh \alpha = \sinh(\alpha + \beta)$, leading to

$$\mathcal{A}_{0} = \frac{2\kappa^{2}m_{1}^{2}m_{2}^{2}\sigma^{2}}{q^{2}} \Big[(1+v^{2})\cosh(i(\hat{p}\times\vec{a})\cdot\vec{q}) + 2v\sinh(i(\hat{p}\times\vec{a})\cdot\vec{q}) \Big],$$
(3.108)

which can be also recast in the compact form

$$\mathcal{A}_{0} = \frac{\kappa^{2} m_{1}^{2} m_{2}^{2} \sigma^{2}}{q^{2}} \sum_{\eta = \pm 1} (1 + \eta v)^{2} e^{i \eta \vec{c} \cdot \vec{q}}, \qquad \vec{c} = \hat{p} \times \vec{a}.$$
(3.109)

Starting from (3.109), and going to impact-parameter space in the usual way,

$$2\delta_0 = \tilde{\mathcal{A}}_0 = \frac{\kappa^2 m_1 m_2 \sigma}{4\nu} \sum_{\eta=\pm 1} (1+\eta\nu)^2 \int \frac{d^{2-2\epsilon} q}{(2\pi)^{2-2\epsilon}} \frac{e^{i(\tilde{b}+\eta\tilde{c}\,)\cdot\tilde{q}}}{q^2} \,, \tag{3.110}$$

and, using (A.45), we obtain the leading eikonal phase including classical spin,

$$2\delta_0 = \frac{\kappa^2 m_1 m_2 \sigma}{4\nu} \frac{1}{4\pi^{1-\epsilon}} \sum_{\eta=\pm 1} (1+\eta \nu)^2 \frac{\Gamma(-\epsilon)}{\left(\left|\vec{b}+\eta \vec{c}\right|^2\right)^{-\epsilon}}.$$
(3.111)

We remark that \vec{c} is orthogonal to \vec{p} by its definition (3.109). The impulse is therefore given by

$$-\vec{Q} = -\frac{\partial 2\delta_0}{\partial \vec{b}} = \frac{\kappa^2 m_1 m_2 \sigma}{2v} \frac{1}{4\pi} \sum_{\eta=\pm 1} (1+\eta v)^2 \frac{\vec{b}+\eta \vec{c}}{\left|\vec{b}+\eta \vec{c}\right|^2}.$$
(3.112)

We see that the entire spin dependence is encoded in the shift $\vec{b} \rightarrow \vec{b} \pm \vec{c}$, which is reminiscent of the Newman–Janis shift [212], relating Kerr to Schwarzschild black holes.

These results are valid for generic spin orientations (see Fig. 8). Let us now consider the case in which both spins are parallel to the angular momentum in the center of mass frame as in Fig. 8(a). That is, we align the impact parameter by letting

$$b^{\mu} = (0, -b, 0, 0), \tag{3.113}$$

so that $\vec{L} = \vec{b} \times \vec{p} = (0, pb, 0)$, then

$$\vec{a}_{1,2} = (0, a_{1,2}, 0), \qquad a_{1,2} > 0, \qquad \vec{c} = (-a, 0, 0),$$
(3.114)

as summarized in Fig. 8(a). In this case, (3.111) reduces to

$$2\delta_0 = \frac{\kappa^2 m_1 m_2 \sigma}{2\nu} \left[-\frac{1+\nu^2}{4\pi^{1-\epsilon}} \left(\frac{1}{\epsilon} + \log(b^2 - a^2) + \gamma \right) - \frac{2\nu}{4\pi} \log \frac{b+a}{b-a} \right] + \mathcal{O}(\epsilon), \tag{3.115}$$

and the impulse is given by

$$-Q^{\mu} = Q \frac{b^{\mu}}{b}, \qquad Q = \frac{\kappa^2 m_1 m_2 \sigma}{v} \frac{1}{4\pi b} \frac{(1+v^2) - \frac{2va}{b}}{1 - \frac{a^2}{b^2}}, \qquad (3.116)$$

which agrees with Eq. (82) of Ref. [59]. The previous result corresponds to the case where the two spins are parallel to the orbital angular momentum (see (3.114) and the equation for \vec{L} after (3.113)). If we had instead taken them to be anti-parallel, we would have obtained a plus sign in the last term in the numerator of (3.116). This implies that the sign in the last term in the numerator of (3.116) is such that for parallel (antiparallel) orbital angular momentum and spin the deflection angle decreases (increases) in agreement with expectations [213].

We remark that, in view of the bound on the spin, $a \le Gm$, and of the PM limit $Gm \ll b$, the parameter a/b is always very small. For this reason, the final results (3.112), (3.116) should be thought of only as a convenient way to package their power series around a/b = 0.

3.2. String theory

Because of the presence of a massless spin-two state in its spectrum, string theory naturally contains gravity and, at sufficiently low energies (large distances), it reduces to an effective gravitational field theory. It is therefore natural to study the gravitational 2-body scattering in a string theory framework.²⁴ The focus of this section is to derive the leading eikonal in a truly stringy regime, i.e. when an effective field theory description is not reliable and the results obtained in

 $^{^{24}}$ Although the large distance physics discussed in this report appears to be insensitive to the ultraviolet completion of the theory, it is also desirable to work in a framework in which such a completion is explicitly implemented.



Fig. 8. Scattering of spinning objects in the center-of-mass frame.

the previous subsections cannot be used. Particular emphasis is given to the new phenomena arising in a string context which are related to the existence of Regge trajectories and to the extended nature of the fundamental objects. This analysis was first performed in [8,10] focusing on the string theory version of the high-energy massless scattering²⁵ discussed in Section 2. Here we start from the slightly simpler fixed-target setup where a light string scatters off a stack of D*p*-branes [22]. Then we will summarize the results obtained in the string–string collision case in Section 3.2.6.

Conceptually, we follow exactly the same approach discussed in the field theory context: we use scattering amplitudes to derive the dynamical information relevant for the leading eikonal and then, as a second step, show how to interpret the result as the motion of a probe in a curved background (in the spirit of Section 2.3). In the main text we focus mostly on type II superstring theories, which are free of tachyonic instabilities, and we refer to the original references for the derivations of several string results that are the starting point of our analysis. In Appendix D we provide some of these derivations in the context of bosonic string theory where it is possible to focus on the key features by using a simpler formalism. It turns out that the tachyonic instabilities do not play an important role in the definition of the string eikonal, since as we shall see they do not contribute to long-range effects, which instead arise from the expansion of the leading Regge trajectory around t = 0. Thus bosonic string theory provides the perfect arena both for developing the technical analysis and for discussing some interesting features of non-supersymmetric setups (see for instance Section 3.2.5). In order to make the comparison between supersymmetric and bosonic string theories easier we indicate the number of spacetime dimensions with d: of course we have d = 10 for type II theories or d = 26 for the bosonic theory, but several results at high energies take the same functional form if the spacetime dimension is formally indicated with d (while D will denote the number of noncompact dimensions).

3.2.1. String-brane scattering at tree level

We consider the scattering of a massless closed string off a stack of *N* coincident D*p*-branes in type II theories where some space-like directions can be compactified on circles. While we will stick to this simple setup, it should be possible to extend this analysis to more complicated configurations with orbifold compactifications and more general boundary conditions [214]. The leading contribution is captured by a worldsheet with the topology of the disk and two punctures in its interior (see Figs. 9 and 10). The boundary of the disk lies on one of the D*p*-branes and the punctures describe the ingoing/outgoing closed string that, for sake of simplicity, we take to be a massless NS–NS state. We focus on the case $p \le 6$ so that the gravitational backreaction of the D*p*-branes decays far away (see Section 4.2.3 for a discussion of this viewpoint). The leading tree-level contribution to this $1 \rightarrow 1$ scattering takes the usual form of a ratio of Γ -functions [215–217]

$$\mathcal{A}_{0}(p_{i},\epsilon_{i}) = -\frac{\kappa_{d}NT_{p}}{2}\mathcal{K}(p_{i},\epsilon_{i})\frac{\Gamma\left(-\alpha'E_{s}^{2}\right)\Gamma\left(-\frac{\alpha'}{4}t\right)}{\Gamma\left(1-\alpha'E_{s}^{2}-\frac{\alpha'}{4}t\right)} \equiv \frac{\mathcal{K}(p_{i},\epsilon_{i})}{(\alpha'E_{s}^{2})^{2}}\mathcal{A}_{0} , \qquad (3.117)$$

where our string conventions on the gravitational coupling κ_d and the D-brane tensions T_p are summarized in Appendix D.1.3. The closed string momentum is not conserved in the transverse directions $\mu = p + 1, \ldots, d - 1$ and $p_1 + p_2 = q$ is the momentum transferred to the D*p*-branes with $t = -q^2$. Momentum is conserved along the D*p*-branes, in the directions $\mu = 0, 1, \ldots, p$, so we introduce $E_s^2 = -(p_i)_{\parallel}^2$ and we will be interested in the high-energy eikonal regime corresponding to the limit $E_s^2 \gg |t|$. The result (3.117) is closely related to the tree-level amplitude for four open strings, as is the case for general mixed open/closed string amplitudes, see [218] for a detailed discussion at tree-level. Then the kinematic factor κ is the same one discussed for supergravity in Section 3.1.4 and the closed string polarizations $(\epsilon_i)_{\mu\nu}$ should be thought as pairs of open string polarizations $(\varepsilon_i)_{\mu}$, as done for scalar-graviton scattering in Section 3.1.2.

 $^{^{25}}$ As we will see, this high-energy limit is dominated by the leading Regge trajectory of the graviton, while subleading Regge trajectories, such as the one of the dilaton, become irrelevant.



Fig. 9. Scattering of a massless string off a Dp-brane. The blue circles represent punctures associated to the incoming and outgoing closed string and the green disk rests on the Dp-brane. The thin red line represents a heavy open string produced in the *s*-channel (Fig. 10(b) below), while the thick black line represents a closed string exchanged in the *t*-channel (Fig. 10(a) below).



Fig. 10. Two degeneration limits of Fig. 9. The blue dots represent punctures associated to the incoming and outgoing string states, while the boundary of the green disk lies on the Dp-branes. The picture on the left depicts the corner of the worldsheet moduli space relevant in the large distance regime where the closed strings in gray are those of the leading Regge trajectory, see Section 3.2.3. The picture on the right depicts the region capturing the closed/open string transition, see Section 3.2.2.

two Γ -functions in the numerator of (3.117) encode different sets of poles: the Gamma that depends on t contains the poles that lie on the closed string Regge trajectory $\alpha_{closed}(t) = 2 + \frac{\alpha' t}{2}$, while the other contain the *s*-channel poles on the open string Regge trajectory $\alpha_{open}(E_s^2) = 1 + \alpha' E_s^2$. We prefer to indicate the total energy in the center of mass for the string brane scattering as E_s rather than \sqrt{s} as perhaps would be more natural by using the relations with the 4-point open string amplitude mentioned above. The reason is that one can see this process as the probe limit of a $2 \rightarrow 2$ scattering where the D-brane stack is the heavy state and we reserve *s* for the total center-of-mass energy of this scattering, including the rest energy of the D-branes.

At leading order in the limit $E_s^2 \gg |t|$ we have

$$\mathcal{K}(p_1, \epsilon_1; p_2, \epsilon_2) = (\alpha' E_s^2)^2 \epsilon_1_{\mu\nu} \epsilon_2^{\mu\nu} + \cdots,$$
(3.118)

where $\epsilon_{1,2\mu\nu}$ are the polarizations of bosons in the NS–NS, and then, by using

$$\frac{\Gamma(a+b)}{\Gamma(a+c)} \sim a^{b-c}, \quad \text{as} \quad a \gg b, c , \qquad (3.119)$$

we get

$$\mathcal{A}_0 \simeq \frac{\kappa_d N T_p}{2} \Gamma\left(-\frac{\alpha'}{4}t\right) e^{-i\pi \frac{\alpha' t}{4}} (\alpha' E_s^2)^{1+\frac{\alpha' t}{4}} . \tag{3.120}$$

The overall normalization is directly related to the classical length scale R_p of the geometry describing the gravitational backreaction of the Dp-branes, see Eq. (D.7). Notice that we wrote explicitly the phase factor of $(-\alpha' E_s^2)^{\frac{\alpha' t}{4}}$ arising from the Stirling approximation (3.119): a simple way to make the appropriate choice of branch cut in the factor of $e^{-i\pi \frac{\alpha' t}{4}}$ is to make sure that the imaginary part of the amplitude is positive (recalling that t < 0). As usual, this follows from the relation between this imaginary part and the production rate of new degrees of freedom, in this case open strings attached to the Dp-branes. We will discuss this interpretation of the phase in Section 3.2.2 showing that there is a quantitative

link between the imaginary part of (3.120) and the amplitudes describing the transition from the external closed string and a particular class of open string states.

When the momentum transferred is small in comparison to the string scale $\alpha'|t| \ll 1$, the string amplitude is dominated by the exchange of massless states and, since we already took the high energy limit, we are just isolating the graviton propagating between the Dp-branes and the external closed strings

$$\mathcal{A}_0 \simeq \frac{4\kappa_d N T_p}{2} \left(-\frac{E_s^2}{t} \right) \,. \tag{3.121}$$

It is possible to make contact with the probe limit discussed at the end of Section 3.1.1 by identifying the Dp-branes with the heavy particle. We can wrap the extended space-like direction of the Dp-branes on a p-dimensional torus of volume V_c and then they would describe a target of mass M_h in the noncompact D = (d - p) dimensions (see also the comments around (D.4))

$$M_h = N\tau_p V_c = \frac{NT_p V_c}{\kappa_d}, \qquad \frac{1}{\kappa_D^2} = \frac{V_c}{\kappa_d^2}.$$
 (3.122)

The string calculation already includes the factor $1/(2M_h)$ which is part of the Fourier transform (3.8) used when both particles in the scattering are dynamical, as in the probe limit we have $4Ep \rightarrow (2M_h)(2E_s)$, where we used (3.17) with $E_1 \rightarrow E_s$ and $m_1 \rightarrow 0$ since we are considering a massless probe. So we have to compare (3.121) with (3.18) divided by $2M_h$ and for m = 0. By using (3.122), we find full agreement. This approximation is reliable when the impact parameter b of the process is much larger that the string length $\ell_s = \sqrt{\alpha'\hbar}$, as it follows from the usual relation $|q| \sim \hbar/b$ connecting the impact parameter b and the momentum $|q| \sim \sqrt{-t}$ of a single graviton exchanged between the string probe and the D*p*-branes.

For general values of $\alpha'|t|$ there are two novelties of the string result with respect to the field theory one: the first is the appearance of a phase and the second is the fact that the simple *t*-dependence related to the field theory propagator for the graviton is substituted by a Γ -function. Both are string effects: the first related to the existence of an open string sector supported by the D*p*-branes and the second related to the existence of higher-spin massive closed string states. Of course these features are a consequence of the extended nature of the elementary objects used in this case and appear generically in all string theories.²⁶ Then it is natural to expect that the string eikonal is qualitatively different from the field theory one when the impact parameter is of the order of the string scale, however it turns out that there are deviations at distances that are *parametrically* larger. In order to provide a first example of such phenomenon let us introduce

$$2\delta_0 = \tilde{\mathcal{A}}_0(E_s, b) = \int \frac{d^{d-p-2}q}{(2\pi)^{d-p-2}} \frac{\mathcal{A}_0(E_s, q^2)}{2E_s} e^{ibq} , \qquad (3.123)$$

where, as mentioned above, we need to include just the factor of $1/(2E_s)$ because the factor of $(2M_s)$ is already included in the amplitude. Following the treatment in [10], we make the massless pole of the string amplitude explicit

$$\mathcal{A}_0 = \frac{\kappa_d N T_p}{2} \frac{4E_s^2}{-t} \Gamma\left(1 - \frac{\alpha' t}{4}\right) e^{\frac{t}{4}Y},\tag{3.124}$$

where we introduced

$$Y = l_s^2(E_s) - i\pi \alpha', \quad \text{with} \quad l_s(E_s) = \sqrt{\alpha' \ln(\alpha' E_s^2)}.$$
(3.125)

The appearance of the effective string length $l_s(E_s)$, with the extra factor of $\ln(\sqrt{\alpha'}E_s)$, will play an important role and signals that elementary energetic strings are larger than expected from naive dimensional analysis. When (3.124) is used in (3.123), we can rewrite the dependence on q of all factors that are analytic as $q \rightarrow 0$ in terms of derivatives with respect to the impact parameter. For instance we have

$$t = -q^2 = \frac{\partial^2}{\partial b^i \partial b^i} \equiv \nabla^2 \quad . \tag{3.126}$$

Then we find the following integral representation of the leading string eikonal

$$2\delta_{0} = \Gamma\left(1 + \frac{\alpha'}{4}\nabla^{2}\right) \int \frac{d^{d-p-2}q}{(2\pi)^{d-p-2}} \frac{\kappa_{d}NT_{p}E_{s}}{q^{2}} \exp\left(-\frac{Y}{4}q^{2} + iqb\right)$$

$$= \kappa_{d}NT_{p}E_{s} \Gamma\left(1 + \frac{\alpha'}{4}\nabla^{2}\right) \int_{0}^{\infty} dT \int \frac{d^{d-p-2}q}{(2\pi)^{d-p-2}} e^{-q^{2}\left(T + \frac{Y}{4}\right) + iqb}$$

$$= \kappa_{d}NT_{p}E_{s} \Gamma\left(1 + \frac{\alpha'}{4}\nabla^{2}\right) \left(\frac{b^{2}}{4}\right)^{-\frac{d-p-4}{2}} \int_{0}^{\frac{b^{2}}{Y}} \frac{d\hat{T}}{(4\pi)^{\frac{d-p-2}{2}}} \hat{T}^{\frac{d-6-p}{2}} e^{-\hat{T}} , \qquad (3.127)$$

 $^{^{26}}$ Details change depending on the particular setups: for instance in the case of the string graviton-graviton scattering of Section 3.2.6 the degrees of freedom responsible for the imaginary part of the tree-level amplitude are closed strings in the *s*-channel.
where in the last step we introduced $\hat{T} = b^2/[4(T + \frac{Y}{4})]$. For real values of *Y*, the integral yields the incomplete Γ -function $\gamma(z, a)$ with

$$\gamma(z,a) = \int_0^a dt \, t^{z-1} e^{-t} = \sum_{k=0}^\infty \frac{a^{z+k} e^{-a}}{z(z+1)\dots(z+k)} \,, \tag{3.128}$$

so by analytical continuation we can write

$$2\delta_0 = \frac{\kappa_d N T_p E_s}{4\pi} \Gamma\left(1 + \frac{\alpha' \nabla^2}{4}\right) \left[(\pi b^2)^{-\frac{d-4-p}{2}} \gamma\left(\frac{d-4-p}{2}, \frac{b^2}{Y}\right) \right],$$
(3.129)

also when Y is given by (3.125). As usual, we expect that this result exponentiates after including the leading contributions of worldsheet diagrams with h + 1 boundaries on the Dp-branes. This is based on the same counting used in the QFT case: by using (D.7) we see that the first factor in (3.129) is proportional to the classical scale R_p^{d-p-3} , so by dimensional analysis δ_0 scales as $(R_p/b)^{d-p-3}E_sb/\hbar$ and is a large quantity in the classical limit. In Section 4.2 we will provide an explicit check of this exponentiation at the string level, focusing just on a particular regime since an exhaustive analysis is not yet available in the literature. Here we assume the eikonal exponentiation and discuss the behavior of (3.129) at large and small distances with respect to the effective string scale introduced in (3.125). In the large distance regime $b \gg l_s(E_s)$ we can use

$$\gamma\left(z,\frac{b^2}{Y}\right) \sim \Gamma(z) - e^{-\frac{b^2}{Y}} \left(\frac{b^2}{Y}\right)^{z-1} . \tag{3.130}$$

To leading order, the above expression does not depend on the impact parameter and yields a real contribution to δ_0 in Eq. (3.129). In the large distance limit we can neglect the overall Γ -function in (3.129) as it yields corrections suppressed by α'/b^2 ,

$$2\delta_{0} \sim \frac{\sqrt{\pi}E_{s}}{2} \frac{\Gamma\left(\frac{d-p-4}{2}\right)}{\Gamma\left(\frac{d-p-3}{2}\right)} \frac{R_{p}^{d-p-3}}{b^{d-p-4}} + \frac{i\pi E_{s}}{2\Gamma\left(\frac{d-p-3}{2}\right)} \sqrt{\frac{\pi\alpha'}{\ln(\alpha'E_{s}^{2})}} \left(\frac{R_{p}}{l_{s}(E_{s})}\right)^{d-p-3} e^{-\frac{b^{2}}{l_{s}^{2}(E_{s})}},$$
(3.131)

where we wrote the leading real and imaginary terms and used (D.7). This is obtained by expanding the γ -function around the first term of the following equation

$$\frac{b^2}{Y} \simeq \frac{b^2}{l_s^2(E_s)} + i \frac{\pi}{\ln(\alpha' E_s^2)} \frac{b^2}{l_s^2(E_s)} , \qquad (3.132)$$

since the second one is suppressed by $1/\ln(\alpha' E_s^2)$ at high energy. Then the expansion of the γ -function is obtained from the integral expression in (3.128)

$$\gamma\left(\frac{d-4-p}{2},\frac{b^2}{Y}\right) \simeq \gamma\left(\frac{d-4-p}{2},\frac{b^2}{l_s^2(E_s)}\right) + i\pi \frac{e^{-\frac{b^2}{l_s^2(E_s)}}\left(\frac{b}{l_s(E_s)}\right)^{d-4-p}}{\ln(\alpha' E_s^2)} .$$
(3.133)

The real part of (3.131) can also be written as $2G_D M_h E_s \Gamma \left(\frac{D-4}{2}\right) (\sqrt{\pi}b)^{-(D-4)}$ with the identifications $\kappa_D^2 = 8\pi G_D$, D = d-p and (3.122). This reproduces the QFT result in Eq. (3.19) for $m_1 = 0$, as expected since in this regime the field theory setup of Section 3.1.1 is a good effective description. The exponentially suppressed contributions are relevant for the imaginary part and can be derived directly from the amplitudes (3.120) as done in (3.146). Taking $-\frac{1}{E_s} \frac{\partial}{\partial b}$ of the real part of (3.130), one finds the deflection angle (B.28) obtained in Appendix B.2 by solving the geodesic equation.

In the opposite regime, where the impact parameter is small with respect to the effective string length $b < l_s(E_s)$, we can keep just the first term in the expansion (3.128) obtaining

$$2\delta_0 \sim \frac{\kappa_d N T_p E_s}{4\pi} \frac{(\pi Y)^{-\frac{d-4-p}{2}}}{\frac{d-4-p}{2}} \left(1 + O\left(\frac{b^2}{Y}, \frac{1}{\ln(\alpha' E_s^2)}\right) \right) .$$
(3.134)

It is interesting to consider the high-energy stringy regime where $b^2 \ll |Y|$, which is possible because of the $\ln(\alpha' E_s^2)$ enhancement in (3.125). Then, the Γ -function in (3.134) can be approximated to 1 (since the shift proportional to $\alpha' \nabla^2$ yields corrections suppressed by α'/Y) and the leading contribution to $2\delta_0$ takes a simple *b*-independent form. Notice, however, that this result is not real: by using (3.125) and expanding the factor *Y* in the regime $\sqrt{\alpha'} \sim b < l_s(E_s)$, one can see that the leading imaginary contribution takes the same form as in (3.131) without the exponential factor which is negligible in this regime

$$\operatorname{Im}(2\delta_0) \simeq \frac{\pi}{2\Gamma\left(\frac{d-p-3}{2}\right)} \left(\frac{R_p}{l_s(E_s)}\right)^{d-p-3} \frac{\sqrt{\pi\alpha'}E_s}{\sqrt{\ln(\alpha' E_s^2)}} .$$
(3.135)

Because of the presence of this imaginary part the elastic partial wave unitarity discussed in Section 1.3 is violated. This is not surprising since, when the probe gets close to the Dp-branes, it can excite the open strings degrees of freedom, i.e.

the closed string can "touch" the target and make a transition to an open string whose endpoints are anchored to one of the D*p*-branes. The less obvious aspect of this phenomenon is that it starts when $b \sim l_s(E_s)$ rather than simply when the impact parameter is of the order of the string length. In order to restore unitarity, one would need to include explicitly the open string degrees of freedom and promote $2\delta_0$ to an operator that can describe a closed/open string transition as discussed in Section 3.2.2.

3.2.2. The closed/open string transition

Let us discuss in some more detail the imaginary part of (3.120)

$$\operatorname{Im}(\mathcal{A}_0) \simeq \frac{\kappa_d N T_p}{2} (\alpha' E_p^2)^{1 + \frac{\alpha' t}{4}} \sin\left(-\pi \frac{\alpha' t}{4}\right) \Gamma\left(-\frac{\alpha' t}{4}\right) = \pi \frac{\kappa_d N T_p}{2} \frac{(\alpha' E_p^2)^{1 + \frac{\alpha' t}{4}}}{\Gamma\left(1 + \frac{\alpha' t}{4}\right)},$$
(3.136)

alt

where in the final step we used

$$\Gamma(z)\Gamma(1-z) = \frac{\pi}{\sin(\pi z)} .$$
(3.137)

A similar formula holds also for the imaginary part of the bosonic string amplitude (D.33)

$$\operatorname{Im}(\mathcal{A}_0) \simeq \pi \frac{\kappa_d N T_p}{2} \frac{(\alpha' E_p^2)^{1+\frac{\alpha' t}{4}}}{\Gamma\left(\xi + \frac{\alpha' t}{4}\right)}, \qquad (3.138)$$

with $\xi = 2$, instead of $\xi = 1$ as in (3.136).

This imaginary part is clearly a string effect since it vanishes in the $\alpha' \rightarrow 0$ limit and we will now show that it is due to the propagation of new degrees of freedom in the *s*-channel which in our setup are open strings representing excitations of the target *Dp*-branes. Let us illustrate this by starting from the full amplitude (3.117) (or (D.26) in the bosonic case). For simplicity in the superstring case (3.117) we choose the external states to be a Kalb–Ramond field with polarizations along the spacelike directions of the D-brane worldvolume: then the factor $\mathcal{K}(p_i, \epsilon_i)$ takes the form given in the square parenthesis of (3.140b) [219]. However, as expected, only the leading term in E_s will be relevant for the final result, so one can also focus directly in \mathcal{A}_0 in (3.117). By using the relation

$$\frac{\Gamma(x)}{\Gamma(x+a)} = \sum_{n=0}^{\infty} \frac{(-1)^n}{n!\Gamma(a-n)} \frac{1}{x+n} , \qquad (3.139)$$

it is possible to rewrite the string amplitudes mentioned above as a series of the s-channel poles

$$\mathcal{A}_{0}^{T} = \frac{\kappa_{d} N T_{p}}{2} \sum_{n=0}^{\infty} \frac{1}{n!} \frac{\Gamma\left(2 + \frac{\alpha' t}{4} + n\right)}{\Gamma\left(2 + \frac{\alpha' t}{4}\right)} \frac{1}{-\alpha' E_{s}^{2} + n - 1} , \qquad (3.140a)$$

$$\mathcal{A}_0^B = \frac{\kappa_d N T_p}{2} \left[(\alpha' E_s^2)^2 + (\alpha' E_s^2) \left(\frac{\alpha' t}{4} \right) \right] \sum_{n=0}^{\infty} \frac{1}{n!} \frac{\Gamma\left(\frac{\alpha' t}{4} + n \right)}{\Gamma\left(1 + \frac{\alpha' t}{4} \right)} \frac{1}{-\alpha' E_s^2 + n} , \qquad (3.140b)$$

where we used again (3.137). The amplitude in the first line refers to bosonic string theory where the probe particle is a tachyon, while the second line refers to the superstring case with the Kalb–Ramond state mentioned above. The square parenthesis in superstring case follows from the prefactor in (3.117) as one can see with a calculation similar to the one discussed for supergravity in Section 3.1.4. The poles in $\alpha' E_s^2$ correspond to open strings of mass $M_n^2 = n/\alpha'$ (or $M_n^2 = (n-1)/\alpha'$ in the bosonic case) propagating on the disk and the residue is a polynomial in *t* of degree *n* which is the maximum spin for the states of mass M_n . While the result in (3.140) seems to be real, there is actually an imaginary part which is hidden in the *i* ϵ prescription which is not written explicitly and should be reinstated. This can be done as follows

$$\frac{1}{-E_s^2 + \frac{n - (\xi - 1)}{\alpha'}} \to \frac{1}{-E_s^2 + M_n^2 - i\epsilon} = \Pr\left[\frac{1}{-E_s^2 + M_n^2}\right] + i\pi\,\delta(-E_s^2 + M_n^2)\,,\tag{3.141}$$

where the first term in the last step indicates the Cauchy Principal Value and the second one is the imaginary contribution due to $i\epsilon$ Feynman's prescription. Notice that this fixes the sign of the imaginary part of (3.141) providing a first principle justification for the prescription discussed after (3.120). Then we can write the imaginary part of the amplitude in a unified way for the bosonic and the superstring case

$$2 \operatorname{Im}(\mathcal{A}_0) \simeq \pi \kappa_d N T_p \sum_{n=0}^{\infty} \frac{1}{\Gamma(n+\xi-1)} \frac{\Gamma\left(\xi + \frac{\alpha't}{4} + n\right)}{\Gamma\left(\xi + \frac{\alpha't}{4}\right)} \delta(-\alpha' E_s^2 + n - (\xi-1)) , \qquad (3.142)$$

where we used the delta function to simplify the square parenthesis in the second equation of (3.140b). Because of perturbative unitarity, each term in the sum represents the modulus square of a 2-point amplitude describing a closed

string transforming into a linear combination of open strings at level *n*. Since we are interested in the limit $\alpha' E_s^2 \gg 1$, we can approximate the sum as an integral over the continuous variable $x = \frac{n}{\alpha' F^2}$

$$\sum_{n=0}^{\infty} \delta(-\alpha' E_s^2 + n - (\xi - 1)) \to \int_0^\infty dx \, \delta(-1 + x)$$
(3.143)

and rewrite the ratio of the *n*-dependent Γ -functions by using the limit (3.119) obtaining

$$\operatorname{Im}(\mathcal{A}_0) \simeq \pi \frac{\kappa_d N T_p}{2} \frac{\left(\alpha' E_s^2\right)^{1 + \frac{\alpha' t}{4}}}{\Gamma\left(\xi + \frac{\alpha' t}{4}\right)} , \qquad (3.144)$$

in agreement with (3.138).

Exactly as in the field theory case, one can extract the long-range description of this string scattering amplitude by taking its Fourier transform to impact parameter space (3.123). Starting from the imaginary part, we can extract the leading large distance behavior by approximating $\Gamma\left(\xi + \frac{a't}{4}\right) \simeq 1$ and then, since $t = -q^2$, the Fourier transform transform translates into a Gaussian integral

$$\operatorname{Im}(2\delta_0) = \int \frac{d^{d-2-p}q}{(2\pi)^{d-2-p}} \frac{\operatorname{Im} \mathcal{A}_0(E_s, q^2)}{2E_s} e^{ibq} \simeq \pi \frac{\kappa_d N T_p}{4} \frac{\alpha' E_s e^{-\frac{b^2}{\alpha' \ln(\alpha' E_s^2)}}}{(\pi \alpha' \ln(\alpha' E_s^2))^{\frac{d-2-p}{2}}} .$$
(3.145)

Then the standard eikonal $e^{2i\delta_0}$ is not a phase as it includes a real factor $e^{-\ln 2\delta_0}$ with

. . .

$$\operatorname{Im}(2\delta_0) \simeq \frac{\pi}{2\Gamma\left(\frac{d-p-3}{2}\right)} \frac{\sqrt{\pi\alpha' E_s}}{\sqrt{\ln(\alpha' E_s^2)}} \left(\frac{R_p}{l_s(E_s)}\right)^{d-3-p} e^{-\frac{b^2}{l_s'(E_s)}} , \qquad (3.146)$$

where we used (D.7). This result is consistent with the one obtained in (3.135) from (3.134).

It is interesting to notice that the ξ dependence drops out in (3.146), so the stringy imaginary part of the tree-level eikonal takes the same form both in the superstring and in the bosonic case. Clearly supersymmetry does not play a crucial role in the closed/open string transition at high energies. Thus for the rest of this subsection we can focus on the slightly easier setup of bosonic string theory without losing any physically interesting qualitative feature. Our aim is to provide a characterization of the open strings produced when the probe is captured by the *Dp*-branes. Since the incident closed string probe has zero space-like momentum along the *Dp*-brane worldvolume, the open strings responsible for the imaginary part (3.146) will be "static" and have just a non-zero energy component $\alpha' E_s^2 = n_0 - 1$, where n_0 indicates the level of the open string produced. So one should be able to retrieve (3.146) from the "square" of the operator \hat{V} describing the tree level transition between a generic closed string V_c and a generic open string V_0 .

To be more specific, by following [220], we introduce the open-closed string vertex \hat{V} that, when saturated with a particular open string state $\langle V_o |$ on the left and a particular closed string state $|V_c V_c\rangle$ on the right, provides their coupling

$$\langle V_0 | \hat{V} | (V_c \bar{V}_c)(p) \rangle = \sqrt{\frac{\kappa_d N T_p}{2\alpha'}} \langle V_c(p) \bar{V}_c(p) V_o \rangle, \qquad (3.147)$$

where *p* is the momentum of the closed state which is described as a ket-vector $|V_c \bar{V}_c\rangle$ on the left hand side to stress that it contains a holomorphic and an antiholomorphic part. In formulae we expect

$$\operatorname{Im}(\mathcal{A}_0) = \pi \, \alpha' \langle (V_c V_c)(p_2) | V^{\dagger} \, V | (V_c V_c)(p_1) \rangle \,. \tag{3.148}$$

The explicit form of \hat{V} was first discussed in [221,222] (in the case of all Neumann boundary conditions) and can also be derived from the overlap of generic Del Giudice, Di Vecchia, Fubini (DDF) states [223] by following [224]. This is the same formalism that we will use in Section 3.2.4 to describe the *closed* string transitions that are relevant for the tidal excitations. We refer to the literature for the derivation (see [220] and references therein) and here it suffices to say that the operator describing the closed/open transition takes the form $\hat{V} \sim \exp[\hat{a}_k N_{kl} \hat{a}_l]$, where the operators \hat{a} indicate the light-cone creation (for the open string) or annihilation (for the closed string) oscillators of level k, l and N_{kl} are explicit coefficients known as Neumann coefficients. The index for the level includes also zero where it represents the energy/momentum of the states. The key idea is to start from the exact expression of the Neumann coefficients and take the Regge limit to obtain simplified expressions in the gauge where the space-like part of the light-cone vector is along the direction of momentum of the closed string (this choice will also be used later in the closed string context, see (3.161)). For instance we can write the transition between an incident closed string tachyon with vanishing transverse momentum and a generic open string state as follows

$$\hat{V}_{p_{\perp}=0}|0\rangle = \sqrt{\frac{\kappa_d N T_p}{2\alpha'}} \,\delta\left(\alpha' E_s^2 + 1 - \sum \hat{a}_{-n}^i \hat{a}_n^i\right) \,e^{\frac{1}{2}\sum_{k,l} N_{kl}^{33} \hat{a}_{-k}^i \hat{a}_{-l}^i}|0\rangle \,, \tag{3.149}$$

where here the \hat{a} 's represent the open string oscillators satisfying $[\hat{a}_{-n}^i, \hat{a}_k^j] = n\delta_{k+n,0}\delta^{ij}$. The delta function enforces the energy conservation and, as mentioned, means that the level of the open string produced is large $n_o \sim \alpha' E_s^2$. In the high energy regime we have $n_o \gg 1$ and, as shown in [220], the Neumann coefficient

$$N_{kl}^{33} = -\frac{\alpha_2}{\alpha_1(k+l)} \binom{-k\frac{\alpha_1}{\alpha_3}}{k} \binom{-l\frac{\alpha_1}{\alpha_3}}{l}, \qquad (3.150)$$

where α_1 (α_2) are the dimensionless left (right) moving light-cone momenta for the incident closed string, while α_3 is the momentum of the open string produced. It is convenient to align the light-cone along the direction of motion of the closed string which we take to be a tachyon. Then we have

$$\begin{aligned} \alpha_{1} &= \sqrt{\alpha'} \left(E_{s} + p \right) = \sqrt{n_{o} - 1} + \sqrt{n_{o} + 3} \sim 2\sqrt{n_{o}} \left(1 + \frac{1}{2n_{o}} \right), \\ \alpha_{2} &= \sqrt{\alpha'} \left(E_{s} - p \right) = \sqrt{n_{o} - 1} - \sqrt{n_{o} + 3} \sim -\frac{2}{\sqrt{n_{o}}} \left(1 - \frac{1}{2n_{o}} \right), \\ \alpha_{3} &= -2\sqrt{\alpha'} E_{s} = -2\sqrt{n_{o} - 1} \sim -2\sqrt{n_{o}} \left(1 - \frac{1}{2n_{o}} \right), \end{aligned}$$
(3.151)

where we used $\alpha' E_s^2 = n_o - 1$ and $\alpha' p^2 = n_o + 3$ for the tachyon and took the large energy limit. Then one can check that the Neumann coefficients (3.150) are suppressed as $1/n_o$ in this limit if k, l are kept fixed. Instead if both k, l scale with n_o the Neumann coefficients stay finite

$$N_{kl}^{33} \sim n^{x+y-2} \frac{x^{x} y^{y}}{\Gamma(1+x)\Gamma(1+y)} \frac{1}{x+y} , \quad k = n_{0}x , \quad l = n_{0}y .$$
(3.152)

This means that we can describe the final open string state in terms of few oscillators in the light-cone gauge (3.151). It is difficult to prove analytically (3.148) even in the case q = 0, but, in this case, it is interesting to notice that it is easy to obtain a convincing numerical check in the light-cone gauge mentioned above: one can expand the exponential in (3.149) and focus on the contribution that is linear in N^{33} for each vertex. Then in (3.148) we obtain a contribution with two sums over the open string levels which at high energy can be approximated with two integrals. By recalling that in the bosonic case the indices *i*, *j* run from 1 to 24, we obtain the following numerical estimate for (3.148)

$$12\pi\alpha'\frac{\kappa_d NT_p}{2\alpha'}n_o \int_0^1 dx \int_0^1 dy \frac{x^{2x+1}y^{2y+1}\delta(x+y-1)}{\Gamma^2(1+x)\Gamma^2(1+y)} \sim 0.929 \frac{\kappa_d NT_p}{2}\alpha' E_s^2 , \qquad (3.153)$$

which is already very close to the full result (3.144). By going to next order and including in (3.149) the quadratic terms in the Neumann coefficients coming from the expansion of the exponential, one can check numerically that this new contribution is subleading and changes the numerical coefficient (3.153) from 0.929 to 0.998 [220]. It is clear that in this gauge the expansion is converging very quickly so one can describe the open string produced in the transition at t = 0as a linear combination of states with a pair of \hat{a}_k^i oscillators plus small corrections with the insertion of few other pairs (with the levels summing up to n_o globally).

We refer again to [220] for the generalization of (3.149) to the case of a non-vanishing transferred momentum q. By indicating the resulting closed/open vertex as $\hat{V}_{q_{\perp}}$, one can introduce its impact parameter version via the usual Fourier transform obtaining

$$\hat{V}_{b}|0\rangle = \int \frac{d^{d-p-2}q_{\perp}}{(2\pi)^{d-p-2}} e^{ib\vec{q}_{\perp}} \hat{V}_{q_{\perp}}|0\rangle , \qquad (3.154)$$

where $\hat{V}_{q_{\perp}}$ is similar to (3.149), but contains also linear terms in the exponential with the open string oscillators [220]. In the high energy limit, the final result takes the form of a squeezed coherent state

$$\hat{V}_{b}|0\rangle = \frac{1}{\left(\pi \,\alpha' \log \alpha' E_{s}^{2}\right)^{\frac{d-p-2}{2}}} \hat{V} \, e^{-\frac{1}{\alpha' \log \alpha' E_{s}^{2}} \left(b^{i} + i\sqrt{\frac{\alpha'}{2}} \sum_{k} \frac{1}{k} \hat{a}_{-k}^{i}\right)^{2}} |0\rangle , \qquad (3.155)$$

with the expected exponential factor $e^{-\frac{b^2}{l_s^2(E_s)}}$ [220] which explicitly shows that the closed/open string transition is suppressed unless the impact parameter is smaller than the effective string length $l_s(E_s)$. It would of course be interesting to generalize this analysis to the case where two open strings are produced by looking at the cuts of the annulus contribution to the closed string scattering. In this case there is a richer phase space for the final states and we expect that the total energy is shared preferably equally among the open strings. However, as far as we know, this analysis has not been carried out explicitly even in the bosonic case.

3.2.3. The Reggeon vertex formalism

As usual in tree-level string theories, Eq. (3.117) displays an infinite set of equally spaced poles. As discussed in the previous sections, in the high energy regime $E_s \gg |t|$ the poles in the *s*-channel merge to form a cut, while the poles in

t-channel appear explicitly in the result (3.120) and in the eikonal (3.134). In this regime the dominant states exchanged between the *Dp*-branes and the scattered string have maximal spin at each mass level, i.e. they should lie along the leading Regge trajectory of the closed string spectrum. In Appendix D.2.1 we show this quite explicitly by following the approach of [225,226] and even if our summary there focuses on the bosonic theory, the generalization to the superstring case is straightforward. Here we follow the approach taken in [149] that was applied to the string-brane scattering in [227,228].

The basic idea is that, at high energy, the dominant region in the integral over the worldsheet moduli (see for instance (D.28)) corresponds to diagrams where two closed string vertex operators are very close: the size of the relevant region scales as $(\alpha' E_s^2)^{-1}$, as pointed out also in the discussion after (D.31) for the bosonic string. Thus instead of calculating the full string amplitude and then take the high energy limit, it is possible to first take a generalized OPE to define a new ingredient, the "Reggeon vertex", that then can be attached to the D*p*-branes (or another Reggeon vertex in the case of the string–string scattering) to obtain directly the high energy limit of the amplitude. In this discussion we can keep the vertex operators v_i general as the only feature we need is the exponential factor $e^{ip_{1,2}X}$ that is always present for states with nonzero momentum. A disk amplitude with two closed strings takes a form similar to (D.28)

$$\mathcal{A}_{0} = C_{S_{2}} \frac{\alpha' \kappa_{d}}{8\pi} N \int \frac{d^{2} z_{1} d^{2} z_{2}}{dV_{SL(2,R)}} \langle 0 | \mathcal{V}_{1}^{(-1)}(z_{1}, \bar{z}_{1}) \mathcal{V}_{2}^{(0)}(z_{2}, \bar{z}_{2}) | B \rangle , \qquad (3.156)$$

but with generic vertex operators²⁷ \mathcal{V}_i . In (3.156), C_{S_2} is fixed as in (D.13) and the vertex operators have a normalization factor $\frac{k_d}{2\pi}$ as in (D.27). By introducing $z = \frac{z_1+z_2}{2}$ and $w = z_1 - z_2$ we can write the contribution from the exponential to the OPE as follows

$$\mathcal{V}_{1}^{(-1)}(z_{1},\bar{z}_{1})\mathcal{V}_{2}^{(0)}(z_{2},\bar{z}_{2}) \sim |w|^{\alpha' p_{1} p_{2}} e^{iqX(z,\bar{z}) + i\frac{p_{1} - p_{2}}{2}(w\partial_{z}X + \bar{w}\partial_{\bar{z}}X)} \frac{\mathcal{O}(z,z)}{|w|^{2n_{1}}|w|^{2n_{2}}} + \cdots,$$
(3.157)

where $\mathcal{O}(z, \bar{z})$ comes from the contribution of the non-exponential part of the vertices \mathcal{V}_i and $\alpha' p_i^2 = -4n_i$. We are focusing only on the most singular term as $w \to 0$ as the subleading contributions are suppressed at high energies. As mentioned, the motivation for taking the OPE approximation above is that the high energy result is dominated by the region of $|w|^2 \leq 1/(\alpha' E_s^2) \ll 1$. However this means that we need to treat exactly the terms where w is enhanced by factor of E_s , as in the combination proportional to $(p_1 - p_2)$ in the exponential of (3.157). By using the on-shell conditions we can rewrite the first contribution in this exponential as $\alpha' p_1 p_2 = -\frac{\alpha' t}{2} + 2n_1 + 2n_2$. In order to obtain a non-trivial result in the high energy limit, the OPE between the polynomial part of the vertices should compensate the factor $|w|^{2n_1+2n_2}$ coming from contraction of the exponential part leaving a pole and this is why we are focusing on the leading term in (3.157). One can check that such contribution can exist by considering states in the leading Regge trajectory $\mathcal{V}_j \sim \mathcal{P}_j^{[M_i]} \bar{\mathcal{P}}_j^{[N_i]} e^{ip_j X}$ that contain a polynomial of degree n_1 in the picture -1 (or n_2+1 in the picture 0) in ∂X and similarly for the antiholomorphic sector

$$\mathcal{P}_{1}^{\{M_{i}\}} = \psi^{(M_{0}} \sqrt{\frac{2}{\alpha'}} i \partial X^{M_{1}} \dots \sqrt{\frac{2}{\alpha'}} i \partial X^{M_{n_{1}}} e^{-\varphi} ,$$

$$\mathcal{P}_{2}^{\{M_{i}\}} = \left(\sqrt{\frac{2}{\alpha'}} i \partial X^{(M_{0}} \sqrt{\frac{2}{\alpha'}} i \partial X^{M_{1}} + (p_{2}\psi) \psi^{(M_{0}} i \partial X^{M_{1}} - n_{2}\psi^{(M_{0}} \partial \psi^{M_{1}} \right)$$

$$\sqrt{\frac{2}{\alpha'}} i \partial X^{M_{1}} \dots \sqrt{\frac{2}{\alpha'}} i \partial X^{M_{n_{2}}} .$$
(3.158)

The leading contribution is when all factors of ∂X and $\overline{\partial} X$ in both vertices are contracted either among themselves or with the exponential part of the other vertex, yielding a factor of $|w|^{-2n_1-2n_2}$, and finally there is an extra factor of $|w|^{-2}$ from the contraction of the terms with $p_2\psi$ (and $p_2\overline{\psi}$) in the second term of $\mathcal{P}_2^{\{M_i\}}$ (and $\overline{\mathcal{P}}_2^{\{M_i\}}$). When the leading term in the sum (3.157) is trivial, then the transition between the states described by \mathcal{V}_1 and \mathcal{V}_2 is suppressed at high energies. Thus we can isolate the dependence on w obtaining

$$\int \frac{d^2 z_1 d^2 z_2}{dV_{SL(2,R)}} \mathcal{V}_1(z_1, \bar{z}_1) \mathcal{V}_2(z_2, \bar{z}_2) \sim \mathcal{O}(z, \bar{z}) e^{iqX} \int \frac{d^2 w}{2\pi} |w|^{\frac{-\alpha' t}{2} - 2} e^{i\frac{p_1 - p_2}{2} (w \partial_z X + \bar{w} \partial_{\bar{z}} X)},$$
(3.159)

where we can set z = 1 and the factor of 2π in the measure comes from the residual conformal invariance. We can then perform the integral over w and \bar{w} by using²⁸

$$\int d^2 u \left(|u|^2\right)^{-A - \frac{\alpha' t}{4}} e^{B(u + \bar{u})} = 2\pi \frac{\Gamma\left(1 - A - \frac{\alpha' t}{4}\right)}{\Gamma\left(A + \frac{\alpha' t}{4}\right)} (-B^2)^{A - 1 + \frac{\alpha' t}{4}} , \qquad (3.160)$$

which can be checked by introducing a Schwinger parameter τ to rewrite the first factor as an exponential and then by carrying out the Gaussian integration and the integration over τ .

²⁷ Focusing on NS–NS states we need to take one vertex, say v_1 in the superghost picture (-1, -1) and the other in the picture (0, 0).

²⁸ The factor 2 in front comes from the normalization of d^2w discussed after (D.14).

In order to extract the energy dependence it is convenient to introduce a set of two light-cone vectors e^{\pm} such that

$$(e^{-})^{\mu} = \frac{1}{\sqrt{2}} \lim_{E_{s} \to \infty} \frac{p_{2}^{\mu}}{E_{s}} = -\frac{1}{\sqrt{2}} \lim_{E_{s} \to \infty} \frac{p_{1}^{\mu}}{E_{s}} , \qquad (e^{+}e^{-}) = 1 .$$
(3.161)

Then in the case at hand, O contains just the superghost contributions and a factor of $(p_2\psi)(p_2\bar{\psi})$ from the vertex in the zero picture, and we can use (3.160) to capture the high energy behavior

$$\int \frac{d^{2}z_{1}d^{2}z_{2}}{dV_{SL(2,R)}} \mathcal{V}_{1}^{(-1)}(z_{1},\bar{z}_{1})\mathcal{V}_{2}^{(0)}(z_{2},\bar{z}_{2}) \simeq e^{-i\pi\frac{\alpha't}{4}} (\alpha' E_{s}^{2})^{1+\frac{\alpha't}{4}} \frac{\Gamma\left(-\frac{\alpha't}{4}\right)}{\Gamma\left(1+\frac{\alpha't}{4}\right)} \mathcal{O}' e^{iqX}$$

$$\psi^{+}e^{-\varphi}\bar{\psi}^{+}e^{-\bar{\varphi}} \left(i\sqrt{\frac{2}{\alpha'}}\partial X^{+}\right)^{\frac{\alpha't}{4}} \left(i\sqrt{\frac{2}{\alpha'}}\bar{\partial}X^{+}\right)^{\frac{\alpha't}{4}} ,$$
(3.162)

where \mathcal{O}' is the result of the contractions between the string coordinates in \mathcal{V}_1 and \mathcal{V}_2 as discussed after (3.158). When inserted in (3.156), the operatorial part of (3.162) is trivial as the holomorphic and antihomorphic parts are contracted among them, while \mathcal{O}' is a *c*-number that by construction follows from the correlator $\langle \mathcal{V}_1^{(-1)} e^{iqX} \mathcal{V}_2^{(0)} \rangle$ Thus we can factorize the amplitude (3.156)

$$\mathcal{A}_0 \simeq \langle \mathcal{V}_1^{(-1)} \mathcal{V}_2^{(0)} \mathcal{V}_R^{(-1)} \rangle \Pi_R \langle \mathcal{V}_R^{(-1)} | B \rangle , \qquad (3.163)$$

where we introduced a Reggeon vertex $v_R^{(-1)}$ describing the collective contributions of the states exchanged in the closed-string channel, the Reggeon propagator Π_R and the couplings to the Dp-branes described by the boundary state $|B\rangle$ [229,230]. The Reggeon vertex is basically given by (3.162)

$$\mathcal{V}_{R}^{(-1)} = \kappa_{d} \left[\psi^{+} \mathrm{e}^{-\varphi} \left(\sqrt{\frac{2}{\alpha'}} i \partial X^{+} \right)^{\frac{\alpha' t}{4}} \right] \left[\bar{\psi}^{+} \mathrm{e}^{-\bar{\varphi}} \left(\sqrt{\frac{2}{\alpha'}} i \bar{\partial} X^{+} \right)^{\frac{\alpha' t}{4}} \right] e^{iqX} .$$
(3.164)

The factors of $\sqrt{\alpha'}E_s$ in (3.162) are automatically produced when inserting the Reggeon vertex in the 3-point correlator in (3.163) by performing the contraction between ∂X^+ and the exponential factors $e^{ip_{1,2}X}$, see (D.38). The split between the propagator and the boundary state contribution is defined as follows for later convenience, see the discussion after (3.178),

$$\Pi_R = \frac{1}{2\pi} e^{-i\pi\frac{\alpha' t}{4}} \frac{\Gamma\left(-\frac{\alpha' t}{4}\right)}{\Gamma\left(1+\frac{\alpha' t}{4}\right)} , \quad \langle \mathcal{V}_R^{(-1)} | B \rangle = 2\pi \frac{NT_p}{2} \Gamma\left(1+\frac{\alpha' t}{4}\right) . \tag{3.165}$$

We can use the OPE with the supercurrent to obtain the Reggeon vertex in the (0, 0) picture from the expression in the (-1, -1) picture given in (3.164) obtaining

$$\mathcal{V}_{R}^{(0)} = \kappa_{d} \left[-\frac{2}{\alpha'} \partial X^{+} \partial X^{+} - iq\psi\psi^{+} \partial X^{+} - \frac{\alpha't}{4}\psi^{+} \partial\psi^{+} \right] \left(\sqrt{\frac{2}{\alpha'}} i\partial X^{+} \right)^{\frac{\alpha't}{4} - 1}$$

$$\left[-\frac{2}{\alpha'} \bar{\partial} X^{+} \bar{\partial} X^{+} - iq\psi\bar{\psi}^{+} \bar{\partial} X^{+} - \frac{\alpha't}{4}\bar{\psi}^{+} \bar{\partial}\bar{\psi}^{+} \right] \left(\sqrt{\frac{2}{\alpha'}} i\bar{\partial} X^{+} \right)^{\frac{\alpha't}{4} - 1} e^{iqX} .$$

$$(3.166)$$

In conclusion we have constructed a Reggeon vertex operator v_R both in the picture 0 and in the picture -1 that, when inserted in (3.163), gives the correct high energy behavior of the amplitude for any choice of the two vertex operators v_1 and v_2 .

3.2.4. The string eikonal operator: the closed string sector

The main advantage of the approach discussed in the previous section is that the external states in the string-brane scattering are arbitrary. So it is possible to go beyond the scattering of massless states discussed in Section 3.2.1 and consider inelastic transitions of course focusing always on the leading contribution at high energy that has the same scaling as the elastic amplitudes (3.120). As we will see, such transitions are unavoidable in string theory and promote the eikonal phase discussed in Section 3.2.1 to an eikonal operator that acts on the Hilbert space of the closed string excitations [8,10]. This is an effect of the tidal forces acting on the scattered string that get enhanced at high energies [231]. As we will see explicitly in Section 4.2.2, in the case at hand these tidal forces are related to the gravitational field produced by the Dp-branes.

There are several ways to derive the eikonal operator related to tidal excitations. In the original approach [8,10], one obtains the result by factorizing elastic loop amplitudes, where arbitrary tidally excited states appear in the intermediate

channel. Here we follow the opposite approach and study directly inelastic tree-level amplitudes. Various techniques can be used to connect the eikonal operator and the string amplitudes [227,228,232]: here we will use the results obtained in the previous section and use the Reggeon vertex as the object that encodes all the contributions relevant at high energies. In practice we will derive an explicit expression for the correlator $\langle V_2 V_R V_1 \rangle$ appearing in (3.163) (even though it will be simpler to use the (0, 0) picture version in (3.166) so that both external states $V_{1,2}$ can be written in the (-1, -1) picture). Alternative approaches include the direct evaluation of the inelastic correlators by using vertex operators or the use of the Green–Schwarz 3-string vertex [233,234], as discussed in detail in [232].

The basic idea is to use the DDF formalism [223] (see also [235–237] for an explicit discussion of the NS sector of superstring theory) to deal with the vertex operators $v_{1,2}$. For more recent work on the DDF states see [238–242]. So we briefly review the key ideas of such construction. The starting point is an auxiliary tachyonic state with momentum p_T

$$|p_T; 0\rangle, \qquad \frac{\alpha'}{2}p_T^2 = 1.$$
 (3.167)

Then we need to introduce a null vector k whose scalar product with p_T is one, and (d-2) space-like vectors ϵ_j that are perpendicular to k. In summary we have

$$\frac{\alpha}{2}p_T k = 1, \qquad \epsilon_j k = 0, \qquad \epsilon_i \epsilon_j = \delta_{ij} . \tag{3.168}$$

Focusing on the NS sector for simplicity, the physical states are constructed by acting on the ground state (3.167) with the following DDF oscillators

$$A_{-n,j} = -i \oint_{0} dw \ (\epsilon_{j})_{\mu} \left(\sqrt{\frac{2}{\alpha'}} \partial X^{\mu} + in(k\psi)\psi^{\mu} \right) e^{-inkX_{L}(w)} , \qquad (3.169)$$
$$B_{-r,j} = i \oint_{0} dw \ (\epsilon_{j})_{\mu} \left(\sqrt{\frac{2}{\alpha'}} \partial X^{\mu} (k\psi) - \psi^{\mu} (k\partial X) + \frac{1}{2} \psi^{\mu} (k\psi) \frac{(k\partial \psi)}{(k\partial X)} \right) \frac{e^{-irkX_{L}(w)}}{(ik\partial X)^{\frac{1}{2}}} ,$$

where n(r) is a positive integer (half-integer). Of course a similar definition holds for the anti-holomorphic part with the exchange $X_L \to X_R$, $\psi \to \bar{\psi}$, see Appendix D.1.2 for our string conventions. As usual, it is necessary to impose the GSO projection: in order to describe the matter part of the states in the (-1, -1) picture, we have to select only the states containing an odd number of $B_{-r,j}$, so the first non trivial physical state is obtained by applying the operator $B_{-\frac{1}{2},j}$.

We are now ready to write explicitly the Reggeon vertex (3.166) in the DDF basis for the incoming and outgoing states

$$\langle \mathcal{V}_2^{(-1)} \mathcal{V}_R^{(0)} \mathcal{V}_1^{(-1)} \rangle$$
, (3.170)

where $|\mathcal{V}_i^{(-1)}\rangle$ are the operators corresponding to the DDF states introduced above. The general overlap between three DDF states was discussed in [224] in the bosonic case and in [237] for the NS string and evaluating the contour integrals in the definition of the DDF oscillators (3.169) yields the Neumann coefficients defining the generic couplings among three string states.²⁹ Evaluating (3.170) turns out to be much simpler and the key point is that we can choose *k* to simplify the high-energy limit of (3.170), which can be done by choosing *k* to be along e^+ introduced in (3.161). Then things become particularly simple in the fermionic sector: the Reggeon vertex does not contain any ψ^- (or $\bar{\psi}^-$) insertions, so no contractions are possible for the terms proportional to $k\psi$ (or $k\bar{\psi}$) which then can be set to zero. Then for our propose we can approximate the DDF oscillators as follows

$$A_{-n,j} \rightarrow -i \oint_{z} dw \ (\epsilon_{j})_{\mu} \sqrt{\frac{2}{\alpha'}} \partial X^{\mu} e^{-inkX_{L}(w)} ,$$

$$B_{-r,j} \rightarrow -i \oint_{z} dw \ (\epsilon_{j})_{\mu} \psi^{\mu} (k \partial X)^{\frac{1}{2}} e^{-irkX_{L}(w)}$$
(3.171)

where the contour integrals are around the tachyon exponential factor $e^{ip_T X(z)}$. We can also approximate the Reggeon vertex in (3.166) keeping only:

$$\mathcal{V}_{R}^{(0)} \simeq \kappa_{d} \left(\sqrt{\frac{2}{\alpha'}} i \partial X^{+} \right)^{\frac{\alpha' t}{4} + 1} \left(\sqrt{\frac{2}{\alpha'}} i \bar{\partial} X^{+} \right)^{\frac{\alpha' t}{4} + 1} \mathrm{e}^{i q X} \,. \tag{3.172}$$

The Reggeon vertex does not depend on the transverse fermionic coordinates $\epsilon_j \psi$, so the oscillators *B* and \overline{B} must pair up between the incoming and the outgoing state and the contour integrals in their definition reduces to the ones appearing in the 2-point function. In summary the Reggeon vertex in the fermionic sector acts simply as the identity operator. For the bosonic transverse oscillators (3.171) there are two options: they can be paired between the two external states as in the

²⁹ The same result is obtained in the light-cone approach, see [243] for the bosonic case and [233,234] for the superstring.

fermionic case, which of course requires that they appear in identical pairs, or they can be contracted with the exponential factor of the Reggeon vertex (3.172). The latter option is the technical origin for the inelastic transitions we are interested in. Each one of such contractions yields a factor of $\pm \sqrt{\frac{\alpha'}{2}} \epsilon_j q$ times an integral over the insertion w which turns out to be equal to one. The same result holds also for the anti-holomorphic modes. Finally the level matching condition implies that the difference of the total holomorphic and anti-holomorphic mode number vanishes for each external state. So we can equate this difference for the first and the second state and then take away the contribution of all modes that are paired since they contribute equally to both sides. Thus we see that the bosonic modes that are contracted with the exponential factor in the Reggeon vertex satisfy the following constraint

$$\sum n_1 - \sum \bar{n}_1 = \sum n_2 - \sum \bar{n}_2 . \tag{3.173}$$

We can then summarize the action of the Reggeon vertex in (3.170) as an exponential constructed with the DDF bosonic oscillators and the transferred momentum q

$$\mathcal{V}_{R}^{(0)} \to \kappa_{d} \left(\alpha' E_{s}^{2} \right)^{1 + \frac{\alpha' t}{4}} \int_{0}^{2\pi} \frac{d\sigma}{2\pi} : e^{iq\hat{X}} : , \qquad (3.174a)$$

$$\hat{X}^{j} = i \sqrt{\frac{\alpha'}{2}} \sum_{n \neq 0} \left(\frac{A_{n,j}}{n} e^{in\sigma} + \frac{\bar{A}_{n,j}}{n} e^{-in\sigma} \right), \tag{3.174b}$$

where the integral over σ enforces the constraint (3.173) and the energy dependent factors follows from the contraction of the fields ∂X^+ in (3.172) with the exponential part of the external states. The exponential is normal ordered so the positive modes are contracted with the incoming state $|V_1\rangle$ and the negative modes with the outgoing one $\langle V_2|$.

An advantage of the approach based on the Reggeon vertex discussed here is that it is fully covariant and so it can be used to provide a full characterization (in terms of the little group SO(d - 1)) of the (massive) excited states produced by tidal excitations. It is then possible to carry out explicit checks between the results obtained by using (3.174) and those obtained by the direct evaluation of the corresponding covariant amplitudes, see [232] for a detailed discussion of the transition between the ground state to the first and the second massive level. All results are consistent with the key properties of (3.174): the excitations added or damped by the tidal forces involve only the bosonic oscillators and are always in the spatial direction perpendicular to the (fast) motion of the scattering string

We can use (3.174) to write the high-energy result for the string-brane scattering as an operator instead of a matrix element between two specified states as in (3.163)

$$\hat{\mathcal{A}}_{0} \simeq \frac{NT_{p}\kappa_{d}}{2} e^{-i\frac{\alpha't}{4}} \Gamma\left(-\frac{\alpha't}{4}\right) (\alpha' E_{s}^{2})^{1+\frac{\alpha't}{4}} \int_{0}^{2\pi} \frac{d\sigma}{2\pi} : e^{iq\hat{X}} :, \qquad (3.175)$$

where we used (3.165). Notice that the non-operatorial overall factor matches that of Eq. (3.120). As usual we can derive the eikonal in the impact parameter space by using (3.123). As we discussed in Section 3.2.1, when the impact parameter becomes of the order of $l_s(E) = \sqrt{\alpha' \ln(\alpha' E_s^2)}$, the dynamics is dominated by the open strings attached to the Dp-branes and the picture of a closed string to closed string scattering is not reliable. So, let us focus on the case $b \gg l_s(E_s)$ where $\alpha' t$ is very small and the elastic factor in (3.175) reduces to the field theory result (3.121). Then it is straightforward to perform formally the Fourier transform (3.123) at the operatorial level, since the last factor in (3.175) provides just a shift $b \rightarrow b + \hat{X}$ in the result

$$2\hat{\delta}_{0} = \int \frac{d^{d-p-2}q}{(2\pi)^{d-p-2}} \frac{\hat{\mathcal{A}}_{0}(E_{s}, q^{2})}{2E_{s}} e^{ibq}$$

$$\simeq \int \frac{d^{d-p-2}q}{(2\pi)^{d-p-2}} NT_{p}\kappa_{d}E_{s} \int_{0}^{2\pi} \frac{d\sigma}{2\pi} \frac{e^{iq(b+\hat{X})}}{q^{2}} = \int_{0}^{2\pi} \frac{d\sigma}{2\pi} : 2\delta_{0}(b+\hat{X}):,$$
(3.176)

where we followed the steps in (3.127) for $b^2 \gg l_s^2(E_s)$ and wrote explicitly the normal ordering prescription. Thus, at the level of the leading eikonal, the generalization to the full string level takes the appealing form of an average of the effective QFT result, but with the appropriate "local" impact parameter for each point along the string: the part of the closed string that are closer to the Dp-branes feel a stronger gravitational force than those that are further apart. As we will see in Sections 4.2.2 and 4.2.3 this is at the basis of the tidal effects that are enhanced in the high energy limit.

3.2.5. Graviton scattering in bosonic string theory

In this subsection we show that, in sharp contrast with what happens in field theory discussed in Section 3.1.5, the terms appearing in the graviton scattering in the bosonic string that come from a quadratic (Riemann)² and a cubic term (Riemann)³ do not contribute to leading order at small impact parameter. Therefore in the bosonic string there is no risk of negative Shapiro time delay. The disk amplitude with two closed string tachyons is discussed in Appendix D.2. In the Regge limit it reduces to Eq. (D.33) from which one can derive the eikonal (D.35) for the tachyon scattering off a stack of

N coincident Dp-branes. In the same regime the disk amplitude of a massless state is given instead by:

$$\mathcal{A}_{0}^{\mathcal{GG}} \sim \frac{\kappa_{d} T_{p} N}{2} \mathrm{e}^{-i\pi \frac{\alpha' t}{4}} (\alpha' E_{s}^{2})^{1+\frac{\alpha' t}{4}} \Gamma \left(-1-\frac{\alpha' t}{4}\right)$$

$$\times \left((\epsilon_{1}\epsilon_{2})-\frac{\alpha'}{2}(\epsilon_{1}q)(\epsilon_{2}q)\right) \left((\bar{\epsilon}_{1}\bar{\epsilon}_{2})-\frac{\alpha'}{2}(\bar{\epsilon}_{1}q)(\bar{\epsilon}_{2}q)\right) , \qquad (3.177)$$

where, as usual, we split the closed string polarization into its holomorphic and anti-holomorphic part $\mathcal{G}_{\mu\nu} = \epsilon_{\mu}\bar{\epsilon}_{\nu}$. Notice that the quantity in the first line in (3.177) is just the elastic scattering of a tachyon on the Dp-branes (D.33). This result can be compared with that obtained for the supersymmetric case in Section 3.2.1, which we report below by combining (3.118) and (3.120)

$$\mathcal{A}_0 \sim (\epsilon_1 \epsilon_2) (\bar{\epsilon}_1 \bar{\epsilon}_2) \frac{\kappa_d T_p N}{2} \Gamma \left(-\frac{\alpha' t}{4} \right) \mathrm{e}^{-i\pi \frac{\alpha' t}{4}} (\alpha' E_s^2)^{1 + \frac{\alpha' t}{4}} . \tag{3.178}$$

A first qualitative difference between (3.177) and (3.178) is that the leading Regge trajectory in the latter includes the tachyonic (ground) state of the bosonic theory. The second difference is that even in the high energy Regge limit the bosonic amplitude has a non-trivial dependence on the polarization tensors, see the second line in (3.177). As emphasized in [154], this is a direct consequence of the modification of the three-graviton vertex in the bosonic theory which yields a quadratic (Riemann)² and a cubic term (Riemann)³ in the effective action, while in the maximally supersymmetric case these corrections are forbidden by supersymmetry. Because of this, the Lorentz structure in (3.177) is the same as the one appearing in Section 3.1.5 except that in the tree-level graviton-brane scattering there is a single parameter ($\sqrt{\alpha'}$) in the three-point vertex, while in the effective QFT description (3.74) there are in general two independent parameters (l_2 in (3.76) and l_4 in (3.77)).

We can follow the same approach discussed in Section 3.2.1 for the superstring and Appendix D.2 for the bosonic case and derive the eikonal for the graviton scattering: it is sufficient to rewrite the momentum transfer q appearing in the second line of (3.177) in terms of a derivative with respect to the impact parameter and use the result (D.35) for the first line of (3.177). Thus we obtain

$$2\delta_{0} = \left((\epsilon_{1}\epsilon_{2}) + \frac{\alpha'}{2}(\epsilon_{1}\partial_{b})(\epsilon_{2}\partial_{b}) \right) \left((\bar{\epsilon}_{1}\bar{\epsilon}_{2}) + \frac{\alpha'}{2}(\bar{\epsilon}_{1}\partial_{b})(\bar{\epsilon}_{2}\partial_{b}) \right) \\ \times \frac{\kappa_{d}NT_{p}E_{s}}{4\pi} \frac{\Gamma\left(1 + \frac{\alpha'\nabla_{b}^{2}}{4}\right)}{1 - \frac{\alpha'\nabla_{b}^{2}}{4}} \left[(\pi b^{2})^{-\frac{d-4-p}{2}} \gamma\left(\frac{d-4-p}{2}, \frac{b^{2}}{Y}\right) \right].$$

$$(3.179)$$

For the current analysis we ignore the absorptive effects, related to the imaginary part of *Y*, and concentrate our attention on the leading real part by replacing *Y* with $l_s^2(E_s)$, see Eq (3.125). By focusing on the regime $\sqrt{\alpha'} \ll b \ll l_s(E_s)$ we can use (3.128), keeping only the term with k = 0, and approximate the square parenthesis in (3.179) as follows

$$\left[(\pi b^2)^{-\frac{d-4-p}{2}} \gamma \left(\frac{d-4-p}{2}, \frac{b^2}{Y} \right) \right] = \frac{(\pi l_s^2(E_s))^{-\frac{d-4-p}{2}}}{\frac{d-4-p}{2}} + \mathcal{O}\left(\frac{b^2}{l_s^2(E_s)} \right) .$$
(3.180)

This means that the differential operators in the Γ -functions and in the polarization dependent prefactor act on a function of $\frac{b^2}{l_s^2(E_s)}$ that starts with a constant, thus they do not contribute to leading order at small impact parameter. Notice that also the tachyonic pole of the bosonic string becomes harmless.³⁰

The pattern discussed above is in sharp contrast with the QFT case discussed in Section 3.1.5, since in that case the eikonal (3.81) for the scattering of massless states contained terms that grow in the regime where the impact parameter is smaller than the length scales weighting the higher derivative corrections $b \ll l_{2,4}$ (as we mentioned ℓ_s in the string analysis plays the role of both l_2 and l_4 of the effective description). As we will see in Section 4.2.4 this plays a crucial difference in the behavior of the deflection angle and the Shapiro time delay obtained in two cases.

3.2.6. String-string scattering at tree level

Although for pedagogical reasons we have started our discussion of gravitational scattering in string theory from the case of string-brane collisions, historically the first case considered was the one of (massless) string-string collisions at transplanckian energy in (Type II) critical (i.e. d = 10) superstring theory (see Fig. 11). Such a theory is ghost and tachyon free and looks like a fully consistent quantum theory of gravity at least in perturbation theory. In that same approximation it differs from GR, even at large distance, because of the presence of other gravitationally coupled massless modes, besides the graviton. However, by going to high-energy, graviton exchange (or better the Regge trajectory on which it lies) dominates because of its higher spin and therefore by studying string-string collisions at transplanckian energy one can draw interesting lessons for generic theories of quantum gravity involving extended objects. For the original motivations for studying such gedanken experiments we refer to the introductory Section 1.1.

 $^{^{30}}$ This is essentially due to the fact that tachyon exchange is suppressed by two powers of the energy with respect to graviton exchange and therefore it is negligible in the high-energy limit.



Fig. 11. Scattering of two massless closed strings represented by blue and green circles. The thick black line corresponds to closed strings exchanged in the *t*-channel. The thin red line to very massive closed strings in the *s*-channel.

The original idea, in the late eighties, was to consider the collision of massless strings (e.g. gravitons or dilatons) and to study the process in a parameter space containing three relevant length scales: the fundamental length ℓ_s of string theory, see Appendix D, the impact parameter b of the process, and the characteristic scale of the geometry associated with the total center-of-mass energy $R \sim (GE)^{\frac{1}{D-3}}$. Note that this latter scale depends on Newton's constant G, which, in string theory can be traded for the string coupling g_s . By considering the regime of weakly-coupled string theory, $g_s \ll 1$, one can arguably dispose of a fourth length scale, the Planck length ℓ_P , since it will be much smaller than ℓ_s , see again Appendix D. Since the physics of the process can only depend on dimensionless ratios, the final parameter space is effectively two-dimensional.³¹ The other free parameter is the number D of non-compact spatial dimensions, having assumed the remaining (d - D) to be very small and static.

The parameter space is naturally divided in three regions, each one characterized by which length scale dominates over the other two (see Fig. 12). In region I, one has $b \gg l_s(s)$, *R* corresponding to a weak-gravity, point-like regime. In region II, the string scale $l_s(s)$ dominates over *b* and *R*. We may call it the string-gravity regime because it is here that string-size effects are strongly enhanced and deviations from GR are most evident. Finally, when $R > \ell_s$, *b* we enter in the strong-gravity regime where, classical gravitational collapse is expected to occur. A sketch of a possible critical line separating the collapse from the "dispersion" regime is shown in Fig. 12. It uses the physical expectation – following from the collapse criteria discussed in Section 2.4 – according to which the size of the colliding objects plays the role of an additional contribution to the effective marameter. This is why the critical line is expected to bend downwards and hit the real axis when $R \sim \ell_s$ (or $\sqrt{s} \sim g_s^{-2}M_s$). Obviously, if one were able to solve completely the problem in the collapse region by constructing a unitary *S*-matrix, one would solve the (in)famous information paradox raised by Hawking in the seventies [19].

We should finally mention that dividing the parameter space in just three regions is a gross approximation. There are interesting sub-regions, as we will discuss in Section 4.2.5.

This being said, let us start by considering, as the simplest example, the tree-level four-dilaton amplitude. It is fully symmetric in all three Mandelstam variables and given by

$$\mathcal{A}_{0}(s,t) = 8\pi G \left(\frac{tu}{s} + \frac{su}{t} + \frac{st}{u}\right) \frac{\Gamma(1 - \frac{\alpha's}{4})\Gamma(1 - \frac{\alpha'u}{4})\Gamma(1 - \frac{\alpha't}{4})}{\Gamma(1 + \frac{\alpha's}{4})\Gamma(1 + \frac{\alpha't}{4})\Gamma(1 + \frac{\alpha't}{4})}$$
(3.181)

with s + t + u = 0. In the Regge limit, $s \to \infty$ at fixed *t*, using (3.119) on the four Γ -functions that depend on *s*, it becomes:

$$\mathcal{A}_0(s,t) \sim \frac{32\pi G}{\alpha'} \frac{\Gamma(-\frac{\alpha'}{4}t)}{\Gamma(1+\frac{\alpha'}{4}t)} \left(\frac{\alpha'}{4}s\right)^{2+\frac{\alpha'}{2}t} e^{-i\pi\frac{\alpha'}{4}t}.$$
(3.182)

It is possible to obtain directly the high-energy formula above by using the approach of Section 3.2.3 where the amplitude is constructed by gluing together two 3-point correlators involving a Reggeon vertex

/ . . .

$$C_{S_2}\langle \mathcal{V}_1 \mathcal{V}_4 \mathcal{V}_R \rangle \left(\frac{\alpha'}{8\pi}\Pi_R\right) C_{S_2}\langle \mathcal{V}_R \mathcal{V}_2 \mathcal{V}_3 \rangle = \frac{4}{\alpha'} \frac{\Gamma\left(-\frac{\alpha't}{4}\right)}{\Gamma\left(1+\frac{\alpha't}{4}\right)} e^{-i\pi\frac{\alpha't}{4}} \left[\kappa_d \left(\alpha' E_s^2\right)^{1+\frac{\alpha't}{4}}\right]^2.$$
(3.183)

 $^{^{31}}$ One can also neglect the mass of the colliding strings, first by taking the initial states to be massless, and then by using the dynamical fact that only the excitation of relatively light massive strings is induced by tidal forces (see Section 4.2.2).



Fig. 12. A broad-brush phase diagram from collapse criteria, where the effective string scale $l_s(s)$ appearing on the vertical axis is defined in (3.187). We also show, qualitatively, the impact parameters b_t and b_r below which string-size (tidal) and radiative corrections, respectively, start to be relevant in the weak gravity regime. We also illustrate in the two panels how the relative importance of the two kinds of corrections strongly depends on *D*. See Section 4.2.5 for further details.

The square parenthesis follows from the two correlators involving the Reggeon vertex, where each field along the plus light-cone direction takes the leading value $\sqrt{\alpha'}E_s$, where E_s is the energy of each initial state so $s = (2E_s)^2$ (recalling that we are working in the center-of-mass frame). The overall prefactor combines the various normalizations and Reggeon propagator (3.165) and, by using (D.13) and (D.3), one can check that (3.183) agrees with (3.182). In the field theory limit this result reduces to

$$\mathcal{A}_0(s,t) = 8\pi G \frac{s^2}{(-t)} \,. \tag{3.184}$$

In the same limit, the leading eikonal phase is given by

$$2\delta_0(s,b) = \int \frac{d^{D-2}q}{(2\pi)^{D-2}} e^{iqb} \frac{\mathcal{A}_0(s,t)}{2s} = \pi G \frac{s\Gamma(\frac{D-4}{2})}{\pi^{\frac{D-2}{2}}b^{D-4}}.$$
(3.185)

This can be compared with (2.23) showing that, indeed, GR and string theory share the same ultra-relativistic large impact parameter limit.

However, even at arbitrarily high energy, this agreement fails to persist at somewhat lower values of *b*. Naively, one would expect this to happen when *b* becomes $O(\ell_s)$, possibly modulo a logarithmic enhancement as in (3.125). This turns out not to be the case. As we will discuss in Section 4.2.5, at a b_t parametrically larger than ℓ_s (see Eq. (4.120)), the phenomenon of tidal excitation kicks in³² as a result of the gravi-Reggeon exchange already discussed in Sections 3.2.3, 3.2.4 in the case of string-brane collisions, with the only difference that now the Reggeon vertex affects both strings.

One can follow the same steps discussed in (3.127) and rewrite (3.182) in impact parameter space as follows

$$2\delta_0(s,b) = Gs \frac{\Gamma(1 - \frac{\alpha'}{4}\nabla^2)}{\Gamma(1 + \frac{\alpha'}{4}\nabla^2)} \left[(b^2 \pi)^{-\frac{D-4}{2}} \gamma \left(\frac{D-4}{2}; \frac{b^2}{Y_c} \right) \right],$$
(3.186)

where

$$Y_c = l_s^2(s) - i\pi \alpha'$$
, $l_s^2(s) = 2\alpha' \log \frac{\alpha' s}{4}$. (3.187)

³² This phenomenon was first discussed in [10] where it was called diffractive excitation in analogy with a well-known phenomenon in hadronic physics. The true physical interpretation in terms of tidal forces was first given by Giddings [244].

For $b \gg l_s(s)$ we can ignore the ratio of two Γ -functions and use (3.130) obtaining

$$2\delta_0(s,b) \sim \frac{Gs \, \Gamma(\frac{D-4}{2})}{(\sqrt{\pi}b)^{D-4}} - Gs \, (\pi b^2)^{-\frac{D-4}{2}} e^{-\frac{b^2}{Y_c}} \left(\frac{b^2}{Y_c}\right)^{\frac{D-6}{2}} + \cdots$$
(3.188)

From (3.188) we have

$$\operatorname{Re} 2\delta_0(s,b) \sim \frac{Gs \,\Gamma(\frac{D-4}{2})}{(\sqrt{\pi}b)^{D-4}} - Gs \,(\pi b^2)^{-\frac{D-4}{2}} e^{-\frac{b^2}{l_s^2(s)}} \left(\frac{b^2}{l_s^2(s)}\right)^{\frac{D-6}{2}}$$
(3.189)

for the real part and

$$\operatorname{Im} 2\delta_0(s,b) \sim \frac{\pi \alpha'}{l_s^2(s)} Gs(\pi l_s^2(s))^{-\frac{D-4}{2}} e^{-\frac{b^2}{l_s^2(s)}} \left(1 - \frac{D-6}{2} \frac{l_s^2(s)}{b^2}\right)$$
(3.190)

for the imaginary part. As in the string-brane case, the exponentially suppressed term provides the leading imaginary part.

The real part of (3.186) has an amusing shock-wave interpretation in terms of the generalized Aichelburg–Sexl metrics discussed in Section 2.4. Neglecting again the correction coming from the ratio of the two Gamma-functions, it can be shown [245] to correspond to a shock-wave metric where the function $f(x_{\perp})$ is obtained from Eq. (2.57) by substituting, as in Eq. (2.61), $E^{(1)}$ times the transverse δ -function with a Gaussian profile:

$$\rho(\mathbf{x}_{\perp}) = E^{(1)} e^{-\frac{\mathbf{x}_{\perp}^{*}}{l_{s}^{2}(s)}} (\sqrt{\pi} \, l_{s}(s))^{2-D} \,, \tag{3.191}$$

i.e. with an approximately uniform-density beam of size $\sim l_s(s)$ that reduces to the delta function of (2.57) for $l_s(s) \rightarrow 0$. In other words, while the delta-function profile provides only the first term of (3.188) as shown in (2.60), the Gaussian profile provides instead the entire real part of (3.186). The proof of the above statement is straightforward. Following 't Hooft's derivation of the leading eikonal phase sketched in Section 2.3, we know that what determines the phase is, up to an energy factor, the time delay Δv suffered by one particle as it moves in the shock-wave produced by the other particle. According to Eq. (2.59), this time delay is determined by the quantity $f(x_{\perp})$ appearing in the (generalized) AS metric (2.55). Therefore, we can either extract $f(x_{\perp})$ directly from (3.186) and determine the transverse profile of the beam by using (2.57) with the profile (3.191) or, given the profile (3.191), by using (2.57) to fix first f(b) and then, from it, the real part of (3.186).

Without repeating here the discussion given above in the case of string-brane collisions we now simply give the analog of (3.176) for string-string scattering:

$$\hat{\delta}_0 = \frac{1}{4\pi^2} \int_0^{2\pi} \int_0^{2\pi} d\sigma_1 d\sigma_2 : \delta_0(E, b + \hat{X}_1(\sigma_1) - \hat{X}_2(\sigma_2)) : , \qquad (3.192)$$

where $\hat{X}_{1,2}$ are Hermitian (and commuting) closed-string position operators for each incoming string, and δ_0 is the leading eikonal (3.185).

For $b \gg l_s(s)$, $\delta_0(E, b)$ is real and thus $\hat{\delta}_0$ is Hermitian up to exponentially small corrections. The physical interpretation of (3.192) is that the graviton is exchanged between one point on one string and one point on the other string. The (transverse) coordinates of each string are however operators and (3.192) picks up an expectation value of the transverse distance between those two points. The physical consequences of the shift in *b*, tidal excitations due to the extended nature of quantum strings, will be discussed in Section 4.2.2. Here we just mention that, unlike the singular point-like case, the string-corrected eikonal has a finite $b \rightarrow 0$ limit and a smooth expansion in *b* around it. This is the string-string counterpart to what we have already discussed in the string-brane collision, see (3.134). One finds [10]:

$$\operatorname{Re} 2\delta_0 \sim \frac{2Gs}{\hbar} \frac{1}{\left(\pi l_s^2(s)\right)^{\frac{D-4}{2}}} \left(\frac{1}{D-4} - \frac{b^2}{(D-2)l_s^2(s)} + \cdots \right).$$
(3.193)

Another, even more distinctive feature of string theory as opposed to traditional QFT, is that, already at tree level, graviton exchange does not correspond to a real scattering amplitude. This can be seen immediately in (3.182) through its last factor, typical of Regge-pole behavior. This imaginary part comes from the presence of *s*-channel poles in the tree-level Shapiro–Virasoro amplitude corresponding to the heavy (i.e. of mass \sqrt{s}) closed-string intermediate states produced in that channel. It is the very essence of Dolen–Horn–Schmid duality [246,247], as it is incorporated in string theory, that this imaginary part matches, on average, the imaginary part corresponding to the Regge-pole behavior given in (3.182).

Note, however, that the imaginary part of the amplitude, unlike the real part, is order α' and lacks the graviton pole at t = 0. This lack of a singularity corresponds, in *b*-space to an exponential cut-off at large impact parameter. More quantitatively one finds [10]:

$$\operatorname{Im} 2\delta_0(s,b) \sim Gs\pi^2 \alpha' (\pi l_s^2(s))^{-\frac{D-2}{2}} e^{-\frac{b^2}{l_s^2(s)}}.$$
(3.194)

This result fully justifies having neglected Im $2\delta_0$ in the previous, weak-gravity (large *b*) regime. At higher string-loop level we expect multiple string formation in the *s*-channel. In Section 4.2.5 we will discuss how one can have a qualitative idea of the dominant process as a function of the total initial energy up to the already mentioned expected threshold of gravitational collapse, $\sqrt{s} \sim M_s g_s^{-2}$.

From (3.189) and (3.193) we can compute the deflection angle for small and large impact parameter. We get

$$\Theta = \frac{4G\sqrt{s}}{b^{D-3}} \frac{\Gamma(\frac{D-2}{2})}{\pi^{\frac{D-4}{2}}} \to \frac{4G\sqrt{s}}{b}, \qquad b \gg l_s(s),$$
(3.195a)

$$\Theta = \frac{8G\sqrt{s}b}{(D-2)\pi^{\frac{D-4}{2}}l_s^{D-2}(s)} \to \frac{4G\sqrt{s}b}{l_s^2(s)}, \qquad b \ll l_s(s),$$
(3.195b)

where the arrow indicates the limit $D \rightarrow 4$. The first, for large impact parameter, coincides with the one of a massless point-particle moving in the Aichelburg–Sexl metric (see (2.50) that can be derived from (2.60)). The second instead is intrinsically stringy because of the presence of α' . Since the deflection angle increases with *b* for small *b* and decreases with *b* for large *b*, it is to be expected that it will have a maximum at some intermediate *b*. We will discuss this and what happens in the intermediate region in Section 4.2.5.

3.2.7. Brane-brane scattering at leading order

It is possible, and of course interesting, to study the scattering between non-perturbative objects in string theory. When dealing with D-branes, one can use an exact CFT description in terms of the open strings attached to them as we have already done in the string-brane analysis. The calculation by Polchinski [248] for two static parallel D-branes was immediately generalized to the case of moving D-branes [249], see [250] for a derivation in the closed string channel by using the boundary state formalism briefly summarized in Appendix D.1.4. The case of D0-brane scattering is of particular interest because in this setup one can see the emergence of the 11-dimensional Planck length ℓ_{11} which is the fundamental scale of M-theory [251]. Here we will not provide a detailed analysis of the eikonal scattering for the brane-brane case, but will just highlight some similarities and differences with the more detailed analysis of the previous sections.

Technically the leading order eikonal for the D-brane scattering is captured by a world-sheet whose topology is an annulus. The boundary conditions are given by reflection matrices that generalize the one discussed in (D.23), as they now have to encode the information about the velocities of the D-branes. At large distances, this diagram is best described as a long thin cylinder and is dominated by the exchange of massless closed strings. Focusing on the case of the scattering between two D0-brane bound states, the fields mediating such interaction are the dilaton, the Ramond–Ramond vector and of course the graviton. Since type IIA string theory can be viewed as M-theory compactified on a circle and D0-branes are just 11-dimensional supergravity fields with non trivial Kaluza–Klein numbers, it is not surprising that the scattering between D0-branes falls in the setup described in Section 3.1.4. A difference with respect to the configuration summarized in (3.64) (beside the fact that the starting point is now a 11-dimensional theory) is that the KK-modes lie in the same direction. As discussed in [167,211] this amounts to generalizing the factor of σ^2 in the numerator of (3.68) to ($\sigma - \cos \phi$)² with $\phi = 0$ for KK-modes in parallel directions (rather $\phi = \pi/2$ which is appropriate for the choice made in (3.64)). Thus for D0-brane scattering we have the following large distance eikonal phase

$$2\delta_0 = \frac{2m_1m_2G_d(\sigma-1)^2\Gamma\left(\frac{d-4}{2}\right)}{\sqrt{\sigma^2 - 1}(\pi b^2)^{\frac{d-4}{2}}}.$$
(3.196)

In this case we need to set d = 10 for type IIA theory and the mass of each D0-brane bound state is simply $m_i = n_i \tau_0$, where n_i is the number of D0-branes constituents (identified with the KK-mode in the M-theory picture) and τ_0 is the mass of a single D0-brane see (D.5). In the discussion below we focus for simplicity on the case $n_i \sim O(1)$.

By lowering the impact parameter, we expect to see the inelastic channels typical of string theory. For instance at high energy ($s \gg m_i^2$), there should be a scale b_t below which the internal degrees of freedom of each D0-brane bound state are excited and the elastic amplitudes is suppressed. This is the analogue of what was discussed in Section 3.2.4 for the case of elementary strings, but now the excitations are open strings starting and ending on the same D0-brane bound state. From the argument given in Eq. (4.122) we expect that, in the case of D-brane scattering, the scale for these tidal excitations is $b_t^8 \sim g_s G_{10}\alpha's$. The extra factor of the string coupling g_s in comparison to the string case is due to the tension of the D-branes which is larger by $1/g_s$ with respect to that of fundamental strings (equivalently the production of a D-brane excitation requires a closed-open string vertex and so it is suppressed by g_s). It would be interesting to check this with an explicit calculation and provide a precise description for these inelastic transitions in terms of an eikonal operator. To our knowledge, this has not been done yet in the literature.

Instead, the string gravity regime depicted in Fig. 12 was already discussed in the original paper [249] by looking at the imaginary part of the annulus partition function in the full string result. In this case the degrees of freedom responsible for this suppression of the elastic D-brane scattering are the open strings stretched between the two colliding D-brane bound states. Naively one would expect that this channel opens when the impact parameter is of the order of the string scale, but exactly as it happens in the string-brane (Section 3.2.1) and the string–string (Section 3.2.6) cases, there is an enhancement factor scaling logarithmically with the center of mass energy and so again the relevant scale is $l_s(s)$ defined in (3.187). Thus in the ultrarelativistic regime string effects dominate short-distance brane–brane scattering.

Finally, at low velocities the interaction between D0-branes is weaker as already suggested by (3.196) which vanishes as $(\sigma - 1)^2 \sim p_{\infty}^4$. In this regime the scattering between two D0-branes remains essentially elastic, although quantum mechanical, up to 11-dimensional Planck length [251]. It remains to be seen whether the process becomes (semi)classical down to $b = \ell_{11}$ if one considers the collision of two bound states with $n_i \gg 1$ D0-branes.

4. Exponentiation and the subleading 2-body eikonal

In the previous two sections we have extracted the leading eikonal δ_0 in various gravitational theories from the treelevel amplitude. Its exponentiation, as we have seen, is equivalent to the statement that, in the (n-1)-loop amplitude in impact-parameter space \tilde{A}_{n-1} , the leading classical term equals $-\frac{i}{n!}(2i\delta_0)^n$. From the leading asymptotics of the resummed amplitude $1+i\tilde{A} \simeq e^{2i\delta_0}$, one can then compute classical observables, like the deflection angle and the Shapiro time delay, for collisions with large impact parameter, $b \gg R$ with R the typical size of the colliding objects.

However, the result obtained in this way is only accurate to leading order in the small parameter $\frac{R}{b}$. How can we compute the corrections to this leading behavior? Since this regime is the one characterized by weak gravitational interactions between the colliding objects, an equivalent question is: How do we retrieve the classical PM expansion of the deflection angle? The answer is that we have to calculate higher-order corrections in the eikonal phase $\delta = \delta_0 + \delta_1 + \cdots$ and, as we shall see, this can be done by looking at subleading terms in the classical expansion of loop diagrams. In this section, we perform this analysis for the one-loop amplitude \tilde{A}_1 .

Indeed, by the exponentiation of $2i\delta_0$, the leading term in the classical limit of $i\tilde{\mathcal{A}}_1$ must be given by $\frac{1}{2!}(2i\delta_0)^2$ and is thus proportional to $\frac{1}{\hbar^2}$, but the one-loop amplitude contains also an additional sub-leading term that is again proportional to $\frac{1}{\hbar}$ as it happened for the tree diagram itself. The former term is sometimes referred to as "super-classical" or as an "iteration", because it is the most singular in the classical limit and it does not provide any new information compared to the tree-level amplitude. The latter identifies instead a new classical term and one can extract from it the sub-leading eikonal $2i\delta_1$. It is natural to conjecture that also this sub-leading eikonal $2i\delta_1$ exponentiates and this can be argued on general grounds [36,102,252], although we will not discuss this proof in detail here. As we will see, its contribution is indeed subdominant for large impact parameter with respect to the leading one, i.e. it is suppressed by one more power of $\frac{R}{h}$.

Of course, the one-loop amplitude also contains quantum terms that scale like \hbar^n with $n \ge 0$. We will see that, although these extra terms do not contribute to the eikonal phase at one loop, they play an important role in the extraction of the sub-sub-leading parts of the eikonal at two loops.

More concretely, since at each order of the perturbative expansion we find additional classical terms that ought to exponentiate contributing to higher sub-leading parts of the eikonal and quantum terms that do not need to exponentiate,³³ we are then led to the conclusion that, in the classical limit, the full amplitude in impact parameter space is encoded in the following expression:

$$1 + i\tilde{\mathcal{A}}(s,b) = [1 + 2i\Delta(s,b)]e^{2i\delta}$$

$$\tag{4.1}$$

where, schematically, in D = 4

$$2\delta = 2\delta_0 + 2\delta_1 + 2\delta_2 + \dots = \frac{GE^2}{\hbar} \left(\log b + \frac{GE}{b} + \left(\frac{GE}{b}\right)^2 + \dots \right)$$
(4.2)

is the classical eikonal and

$$2\Delta = 2\Delta_1 + 2\Delta_2 + \dots = \left(\frac{GE}{b}\right)^2 \left[1 + \frac{\hbar}{Eb} + \dots\right] + \left(\frac{GE}{b}\right)^3 \left[1 + \frac{\hbar}{Eb} + \dots\right]$$
(4.3)

is the quantum remainder. In the bottom-up approach we are going to adopt, we shall first calculate the perturbative amplitude order by order in the loop expansion $A = A_0 + A_1 + A_2 + \cdots$, concentrating on the limit of small momentum transfer, take the Fourier transform from momentum space to impact-parameter space, and then determine δ_n and Δ_n by solving

$$i\tilde{\mathcal{A}}_0 = 2i\delta_0\,,\tag{4.4}$$

$$i\tilde{\mathcal{A}}_1 = \frac{1}{2!} (2i\delta_0)^2 + 2i\delta_1 + 2i\Delta_1 \,, \tag{4.5}$$

$$i\tilde{\mathcal{A}}_2 = \frac{1}{3!} (2i\delta_0)^3 + 2i\delta_0 2i\delta_1 + [2i\delta_2 + 2i\delta_0 2i\Delta_1] + 2i\Delta_2.$$
(4.6)

³³ One could exponentiate also the quantum part $1 + 2i\Delta = e^{2i\delta^{quant}}$ which amounts to a redefinition of the coefficients of the expansion (4.3). We prefer not to do so in order to define an object, 2δ , which directly yields the classical observables such as the deflection angle.

These relations are predicted by the formal re-expansion of (4.1) for small *G* and define $2\delta_n$, $2\Delta_n$ at an operative level, while also dictating the structure of super-classical terms, which only arise, at each level, from iterations of lower-order terms. In particular, the $\mathcal{O}(b^{-1})$ terms of $\tilde{\mathcal{A}}_1$ determine $2\delta_1$ and its $\mathcal{O}(b^{-2})$ terms determine $2\Delta_1$. Moreover, while $2\Delta_1$ is irrelevant for determining $2\delta_1$ at one-loop level, it is needed at two-loop level, together with $2\delta_0$, in order to solve (4.6) for the unknown $2\delta_2$, once the $\mathcal{O}(b^{-2})$ terms of $\tilde{\mathcal{A}}_2$ are computed.

Let us also mention that, while the leading eikonal is always real, sub-leading eikonals have, in general, also an imaginary part that is connected to the existence of inelastic channels. In this case, the classical deflection angle can be computed from the real part of the sub-leading eikonals, since the real part is the one contributing to classical phase oscillations, while the imaginary part only gives rise to an overall exponential suppression. This issue is going to be particularly acute in the two-loop eikonal that we will discuss in the next section, where the imaginary part actually contains long-range infrared divergences associated to soft graviton emissions.

In this section we restrict ourselves to the one-loop level, however, and analyze in detail the classical limit of the one-loop scattering amplitude $i\tilde{A}_1$ in massive $\mathcal{N} = 8$ supergravity and in GR. This gives us the opportunity to check that the leading super-classical term is indeed obtained from the quadratic term of the expansion of the leading eikonal $e^{2i\delta_0}$, i.e. that $i\tilde{A}_1 \sim \frac{1}{2!}(2i\delta_0)^2$ as in (4.5). We will then extract, from the next to the leading term, the sub-leading eikonal $2\delta_1$ and also, from the next to the next to the leading term, the quantum $2\Delta_1$ that is important in reproducing the two-loop amplitude according to Eq. (4.1).

4.1. 1-loop (2PM) in QFT

This section is divided in two sub-sections. In the first one we study the case of massive $\mathcal{N} = 8$ supergravity and in the second one the case of GR. In the massive case under consideration it is convenient to use the relative Lorentz factor σ , already introduced in (1.8), and also to define a related variable z which has the advantage of rationalizing some square roots, via

$$\sigma = -\frac{p_1 p_2}{m_1 m_2} = \frac{s - m_1^2 - m_2^2}{2m_1 m_2} = \frac{1}{2} \left(z + \frac{1}{z} \right), \qquad z = \sigma - \sqrt{\sigma^2 - 1},$$
(4.7)

where *s* is the Mandelstam variable. In particular $1 \le \sigma < \infty$ and $0 < z \le 1$, with $\sigma = 1$ or z = 1 corresponding to the case of two particles mutually at rest, and $\sigma \to \infty$ or $z \to 0^+$ to the case of an ultrarelativistic collision.

4.1.1. Massive $\mathcal{N} = 8$ supergravity

Let us start from the maximally supersymmetric case, introducing the masses via Kaluza–Klein compactification as discussed in Section 3.1.4. For the *s*-*u* symmetric collision discussed there, corresponding to the elastic scattering of an axion and a dilaton, the one-loop amplitude in $\mathcal{N} = 8$ supergravity with massive external states is given by [167,211]

$$\mathcal{A}_{1} = (8\pi G)^{2} \frac{c(\epsilon)}{2} \left[\left(s - m_{1}^{2} - m_{2}^{2} \right)^{4} + \left(u - m_{1}^{2} - m_{2}^{2} \right)^{4} - t^{4} \right] \left(I_{\mathrm{II}} + I_{\overline{\mathrm{II}}} \right)$$
(4.8)

where

$$c(\epsilon) = \frac{e^{-\gamma_E \epsilon}}{(4\pi)^{2-\epsilon}}, \qquad (4.9)$$

while I_{II} stands for the box integral,

$$I_{\rm II} = \int_{\ell} \frac{1}{(-2p_1\ell + \ell^2 - i0)(2p_2\ell + \ell^2 - i0)(\ell^2 - i0)(\ell - q)^2 - i0)},\tag{4.10}$$

with the shorthand notation

$$\int_{\ell} = e^{\gamma_E \epsilon} \int \frac{d^{4-2\epsilon}\ell}{i\pi^{2-\epsilon}} \,, \tag{4.11}$$

and I_{II} stands for the crossed box integral, obtained from I_{II} replacing p_1 with p_4 . A schematic representation of these topologies is given in Fig. 13. Let us also recall that $q = p_1 + p_4$.

We then face the task of evaluating the integrals I_{II} , $I_{\overline{II}}$. A key simplification in this respect is that we do not need their exact expression, but, since the typical perturbative momentum exchange is $q \sim \frac{\hbar}{b}$ and we focus on the regime of large impact parameters, we can restrict our attention to the limit

$$s, m_1^2, m_2^2 \gg |t| = q^2 \tag{4.12}$$

characterizing the classical PM regime. More precisely, since we are interested in the long-range terms in *b*-space after Fourier transform, we can concentrate on non-analytic terms in the small- q^2 expansion. A systematic strategy that is well suited to this type of calculations is the method of regions [253,254]. In this approach, the guiding principle is that the



Fig. 13. The box and crossed-box topologies.

asymptotic expansion of the desired integrals as $q \rightarrow 0$ is captured by Taylor-expanding the *integrand* with respect to all possible scalings of the loop momentum ℓ that give rise to non-scaleless integrals in dimensional regularization. In our case, the nontrivial scaling choices are $\ell \sim \mathcal{O}(q^0)$, which defines the hard region, and $\ell \sim \mathcal{O}(q)$, which defines to the soft region. The hard region is effectively a power series in q^2 and thus only gives rise to strictly localized contributions in *b*-space. Conversely, the soft region is the one responsible for the non-analytic terms we are after, so we can safely discard the former and concentrate on the latter.

Although we refrain here from exhibiting the full soft-region calculation of I_{II} , which was performed in detail for instance in [48,49,167], let us show how the leading soft term in $I_{II} + I_{\overline{II}}$ can be retrieved, following [109]. Starting from (4.10), the leading soft term in I_{II} reads

$$I_{\rm II} \simeq \int_{\ell} \frac{1}{(-2p_1\ell - i0)(2p_2\ell - i0)(\ell^2 - i0)((\ell - q)^2 - i0)},$$
(4.13)

where we used the scaling condition $\ell \sim q$, while p_1 , p_2 are q-independent to leading order. Sending $\ell \rightarrow q - \ell$, recalling $q = p_1 + p_4 = -p_2 - p_3$ and using $2p_1q = q^2$, $2p_2q = -q^2$, Eq. (4.13) can be rewritten to leading order in the form

$$I_{\rm II} \simeq \frac{1}{2} \int_{\ell} \frac{1}{(\ell^2 - i0)((\ell - q)^2 - i0)} \times \left[\frac{1}{(-2p_1\ell - i0)(2p_2\ell - i0)} + \frac{1}{(2p_1\ell - i0)(-2p_2\ell - i0)} \right].$$
(4.14)

Adding the crossing symmetry partner $I_{\overline{II}}$ and applying very similar manipulations, one then finds

$$I_{\rm II} + I_{\overline{\rm II}} \simeq \frac{1}{2} \int_{\ell} \frac{1}{(\ell^2 - i0)((\ell - q)^2 - i0)} \times \left[\frac{1}{2p_1\ell - i0} + \frac{1}{-2p_1\ell - i0} \right] \left[\frac{1}{2p_2\ell - i0} + \frac{1}{-2p_2\ell - i0} \right].$$
(4.15)

This localizes the integration on a (D - 2)-dimensional subspace

$$I_{\rm II} + I_{\overline{\rm II}} \simeq -2\pi^2 \int_{\ell} \frac{\delta(2p_1\ell)\delta(2p_2\ell)}{(\ell^2 - i0)((\ell - q)^2 - i0)} \,. \tag{4.16}$$

Choosing $p_1 = (-E_1, 0, ..., 0, -p)$, $p_2 = (-E_2, 0, ..., 0, p)$, $q = (0, q_{\perp}, 0)$ one can use the delta functions to perform the integrals with respect to ℓ^0 and ℓ_{\parallel} in $\ell = (\ell^0, \ell_{\perp}, \ell_{\parallel})$, obtaining

$$I_{\mathrm{II}} + I_{\overline{\mathrm{II}}} \simeq e^{\gamma_E \epsilon} \, \frac{i\pi^{\epsilon}}{2pE} \int \, \frac{d^{2-2\epsilon} \ell_{\perp}}{\ell_{\perp}^2 (\ell_{\perp} - q_{\perp})^2} \,, \tag{4.17}$$

with $E = E_1 + E_2$. The last integral is straightforward to perform using Schwinger or Feynman parameters, and one gets

$$I_{\rm II} + I_{\overline{\rm II}} \simeq e^{\gamma_E \epsilon} \, \frac{i\pi}{2pE} \, \frac{\Gamma(1+\epsilon)}{(q^2)^{1+\epsilon}} \frac{\Gamma(-\epsilon)^2}{\Gamma(-2\epsilon)} \,. \tag{4.18}$$

When substituted into (4.8), recalling that $pE = m_1 m_2 \sqrt{\sigma^2 - 1}$ as in (1.13), we thus obtain the leading contribution

$$\frac{i\mathcal{A}_1(s,q^2)}{4pE} \simeq \frac{(8\pi G)^2}{(4\pi)^2} \left(\frac{4\pi}{q^2}\right)^{\epsilon} \frac{(-2\pi m_1^2 m_2^2)}{q^2} \frac{\sigma^4}{\sigma^2 - 1} \frac{\Gamma(1+\epsilon)\Gamma^2(-\epsilon)}{\Gamma(-2\epsilon)} \,. \tag{4.19}$$

Sub-leading and sub-sub-leading terms in q are obtained by retaining higher-order contributions in the soft-region expansion. This can be done via direct integration in the soft region as exemplified here [48,49,108,109] or, more systematically, using Integration-By-Parts identities, reduction to master integrals and soft differential equations adapted to the soft region [47,167].

P. Di Vecchia, C. Heissenberg, R. Russo et al.

The resulting expression, complete to sub-sub-leading order, reads

$$\frac{iA_{1}(s,q^{2})}{4pE} = 4G^{2} \left(\frac{4\pi}{q^{2}}\right)^{\epsilon} \left\{ \frac{-2\pi m_{1}^{2}m_{2}^{2}}{q^{2}} \frac{\sigma^{4}}{\sigma^{2} - 1} \frac{\Gamma(1+\epsilon)\Gamma^{2}(-\epsilon)}{\Gamma(-2\epsilon)} + \frac{2i\sqrt{\pi}m_{1}m_{2}(m_{1}+m_{2})}{\sqrt{q^{2}}} \frac{\sigma^{4}}{(\sigma^{2}-1)^{\frac{3}{2}}} \frac{\Gamma(\epsilon+\frac{1}{2})\Gamma^{2}(\frac{1}{2}-\epsilon)}{\Gamma(-2\epsilon)} - \frac{i\sigma^{3}}{(\sigma^{2}-1)^{2}} \frac{\Gamma^{2}(-\epsilon)\Gamma(1+\epsilon)}{\Gamma(-2\epsilon)} + \left[m_{1}m_{2} \left[(1+2\epsilon) \left(\sigma^{2}\log z + \sigma\sqrt{\sigma^{2}-1} \right) + 2i\pi(\sigma^{2}-1) \right] + \frac{i\pi\epsilon}{2}s\sigma \right] \right\}.$$
(4.20)

Let us now introduce a notation to distinguish leading, sub-leading and sub-sub-leading terms in the q-expansion of $\mathcal{A}_1(s, q^2)$ according to

$$\mathcal{A}_1 = \mathcal{A}_1^{[2]} + \mathcal{A}_1^{[1]} + \mathcal{A}_1^{[0]} + \cdots,$$
(4.21)

in such a way that $\mathcal{A}_1^{[k]} \sim \mathcal{O}(q^{-k-2\epsilon})$. The $\mathcal{A}_1^{[2]}$ term, corresponding to the first line of Eq. (4.20), is the super-classical term. Its Fourier transform in impact parameter space (see Eq. (A.45)) indeed reproduces the quadratic iteration term of the expansion of the leading-order eikonal,

$$i\tilde{\mathcal{A}}_{1}^{[2]}(s,b) = \frac{1}{2} \left(\frac{2im_{1}m_{2}G(\pi b^{2})^{\epsilon}\sigma^{2}\Gamma(-\epsilon)}{\sqrt{\sigma^{2}-1}} \right)^{2} = \frac{1}{2}(2i\delta_{0})^{2} , \qquad (4.22)$$

in agreement with the leading-order term in (4.5). The Fourier transform of $\mathcal{A}_1^{[1]}$, which appears in the second line of Eq. (4.20) and is sub-leading for small q, gives instead the sub-leading eikonal according to

$$2i\delta_1 = i\tilde{\mathcal{A}}_1^{[1]} \tag{4.23}$$

so that³⁴

$$2\delta_1 = \frac{4(\pi b^2)^{2\epsilon} G^2 m_1 m_2(m_1 + m_2)}{\sqrt{\pi b^2}} \frac{\sigma^4}{(\sigma^2 - 1)^{\frac{3}{2}}} \frac{\Gamma(\frac{1}{2} - 2\epsilon)\Gamma^2(\frac{1}{2} - \epsilon)}{\Gamma(-2\epsilon)} \,.$$
(4.24)

Note that this next-to-leading eikonal, arising from box integrals, represents the first correction to (3.68) and is proportional to the sum of the masses. Taking the by now familiar derivative with respect to b, this translates into the following expression for the deflection angle.

$$\Theta = \frac{4GE(\pi b^2)^{\epsilon} \sigma^2 \Gamma(1-\epsilon)}{(\sigma^2 - 1)b} + \frac{4G^2 E(\pi b^2)^{2\epsilon} \sigma^4(m_1 + m_2)}{(\sigma^2 - 1)^2 \sqrt{\pi} b^2} \frac{\Gamma(\frac{1}{2} - 2\epsilon)\Gamma^2(\frac{1}{2} - \epsilon)}{\Gamma(-2\epsilon)} + \mathcal{O}(G^3).$$
(4.25)

Consistently with the above considerations, the $\mathcal{O}(G^2)$ correction vanishes identically for the collision of two (massless) shockwaves. Moreover, even for massive objects, it happens to be zero in D = 4 [211], while it does provide a nontrivial $\mathcal{O}(\frac{R^2}{b^2})$ correction to the deflection angle in higher-dimensional spacetimes. Finally, going to impact parameter with sub-sub-leading term $\mathcal{A}_1^{[0]}$ one gets the first contribution to the quantum

remainder,

$$2i\Delta_1 = i\tilde{\mathcal{A}}_1^{[0]}, \qquad (4.26)$$

finding a real part given by

$$\operatorname{Re} 2\Delta_{1} = \frac{8G^{2}m_{1}m_{2}(\pi b^{2})^{2\epsilon}}{\pi b^{2}} \frac{\sigma^{4}\left(\sigma \log z + \sqrt{\sigma^{2} - 1}\right)}{(\sigma^{2} - 1)^{2}} (1 + 2\epsilon)\Gamma^{2}(1 - \epsilon), \qquad (4.27)$$

and an imaginary part given by

$$\operatorname{Im} 2\Delta_{1} = \frac{8G^{2}\sigma^{3}(\pi b^{2})^{2\epsilon}\Gamma^{2}(1-\epsilon)}{b^{2}(\sigma^{2}-1)^{2}} \left[\frac{\epsilon}{2}s\sigma + 2m_{1}m_{2}(\sigma^{2}-1)\right].$$
(4.28)

In this section we focused on the elastic amplitude (4.8). Looking more in general at the 2 \rightarrow 2 scattering of massless states in $\mathcal{N} = 8$ supergravity, we remark that there can be *inelastic* contributions at the level of $\mathcal{A}_1^{[1]}$. However this does not imply that the 2PM eikonal becomes an operator, because there is an $\mathcal{O}(G)$ contribution to Δ that encodes the transition between different massless states (so this Δ_0 is nontrivial, in this more general setup, and becomes an operator). By using the natural generalization of (4.1), one can verify that the amplitudes in the massless sector are consistent with the 2PM eikonal (4.24), see [43] for an explicit check of this point in the probe limit where massless states scatter off a stack of Dp-branes.

³⁴ Let us point out a typo in the expression for $2\delta_1$ in Ref. [47], which should be multiplied by 1/2.

P. Di Vecchia, C. Heissenberg, R. Russo et al.

4.1.2. Real-analytic, crossing-symmetric reformulation

In the previous subsection we have collected the contribution of the different one-loop diagrams noticing that they combine into much simpler expressions than those of individual diagrams. They are in the form of an expansion in powers of q^2 i.e. of the quantum level at which they contribute. Furthermore, as computer outputs, they simply collect independently real and imaginary terms and express the result in terms of a choice of the two independent Mandelstam variables, $q^2 = -t$ and s (or of quantities like σ and z, themselves functions of s and the masses).

On the other hand, we expect the full amplitude to satisfy two exact properties:

• Real analyticity i.e. $\mathcal{A}(s^*, q^2) = (\mathcal{A}(s, q^2))^*$;

• Crossing symmetry i.e.
$$A(s, q^2) = A(u \equiv -s + q^2 + 2m_1^2 + 2m_2^2, q^2)$$
.

Those two properties, which should hold order by order in q^2 , are not at all apparent in the formulae of the previous subsection, but they must be hidden somewhere, of course up to the order in q^2 at which we stop our expansion. Here we will explicitly show how to rewrite the tree and one-loop amplitudes, given in the previous section in a real-analytic and crossing-symmetric form. Besides its usefulness for the interpretation of the result this also serves as a rather stringent test of the calculations themselves. A similar analysis will be carried out at the two-loop level in Section 6 where analyticity and crossing will turn out to be very useful for actually fixing the entire two-loop amplitude from its imaginary part. Let us mention in passing that analyticity and crossing also played a key role in recent developments [255,256].

To this purpose it is useful to introduce variables which are related to those introduced earlier in Section 1.2 by $s \leftrightarrow u$ exchange:

$$\begin{split} \bar{\sigma} &= \frac{u - m_1^2 - m_2^2}{2m_1 m_2} = -\left(\sigma - \frac{q^2}{2m_1 m_2}\right), \\ \bar{z} &= \bar{\sigma} - \sqrt{\bar{\sigma}^2 - 1} = -\frac{1}{z} \left(1 - \frac{q^2/(2m_1 m_2)}{\sqrt{\sigma^2 - 1}}\right) + O(q^4), \\ 2\bar{\sigma} &= \left(\bar{z} + \frac{1}{\bar{z}}\right), \qquad 2\sqrt{\bar{\sigma}^2 - 1} = \left(\frac{1}{\bar{z}} - \bar{z}\right). \end{split}$$
(4.29)

We also have to bear in mind that the factor $\frac{1}{4pE}$ appearing in some formulae cannot be ignored when discussing analyticity and crossing symmetry of the amplitude.

The tree-level amplitude (3.67) is already manifestly crossing symmetric and real analytic,

$$\mathcal{A}_0 = -\frac{\pi G}{q^2} \frac{16m_1^4 m_2^4 (\sigma^4 + \bar{\sigma}^4) - q^4}{m_1^2 m_2^2 \sigma \bar{\sigma}} \,. \tag{4.30}$$

Up to terms that lack the pole at $q^2 = 0$, we may equivalently write

$$\mathcal{A}_0(s,q^2) = \frac{16\pi G m_1^2 m_2^2}{q^2} \left(\sigma^2 + \bar{\sigma}^2\right). \tag{4.31}$$

At one-loop, Eq. (4.20) suggests trying the following analytic, crossing-symmetric ansatz for the super classical term:

$$\mathcal{A}_{1}^{\text{scl.}}(s,q^{2}) = \frac{16G^{2}m_{1}^{3}m_{2}^{3}}{q^{2}} \left(\frac{4\pi}{q^{2}}\right)^{\epsilon} \frac{\Gamma^{2}(-\epsilon)\Gamma(1+\epsilon)}{\Gamma(-2\epsilon)} \left[(\sigma^{4} + \bar{\sigma}^{4}) \left(\frac{\log(-z)}{\sqrt{\sigma^{2} - 1}} + \frac{\log(-\bar{z})}{\sqrt{\bar{\sigma}^{2} - 1}}\right) \right].$$
(4.32)

It is easy to check that, when expanded in q^2 using (4.29) and the branch choice $\log(-z) = \log z + i\pi$, (4.32) reproduces both the super-classical (first line in (4.20)) and the $O(\epsilon^0 m_1 m_2)$ terms in the square bracket of (4.20).

The remaining terms are as follows:

- A classical contribution proportional to $(q^2)^{-1/2}$ (second line in (4.20)). It is real and can be trivially symmetrized in s u since the error in doing so is of order $(q^2)^{1/2}$;
- A quantum contribution of $O(\epsilon)$ that can be written in the crossing-symmetric form:

$$\begin{aligned} \mathcal{A}_{1}^{qu.}(s,q^{2}) &= 16G^{2}m_{1}^{2}m_{2}^{2}\left(\frac{4\pi}{q^{2}}\right)^{\epsilon} \frac{\Gamma^{2}(-\epsilon)\Gamma(1+\epsilon)}{\Gamma(-2\epsilon)} \hat{\mathcal{A}}_{1}^{qu.}(s,q^{2}), \\ \hat{\mathcal{A}}_{1}^{qu.}(s,q^{2}) &= -\epsilon \left[\left(\frac{\sigma^{5}\log(-z)}{(\sigma^{2}-1)^{3/2}} + \frac{\bar{\sigma}^{5}\log(-\bar{z})}{(\bar{\sigma}^{2}-1)^{3/2}}\right) + \left(\frac{\sigma^{4}}{\sigma^{2}-1} + \frac{\bar{\sigma}^{4}}{\bar{\sigma}^{2}-1}\right) \right] \\ &-\epsilon \frac{(m_{1}^{2}+m_{2}^{2})}{2m_{1}m_{2}} \left(\frac{\sigma^{4}\log(-z)}{(\sigma^{2}-1)^{3/2}} + \frac{\bar{\sigma}^{4}\log(-\bar{z})}{(\bar{\sigma}^{2}-1)^{3/2}} \right). \end{aligned}$$
(4.33)

Note that in (4.34) we can again neglect the difference between $\bar{\sigma}$ and $-\sigma$ as well as the difference between \bar{z} and -1/z. As a result, the last line of (4.34) gives a purely imaginary contribution in the physical region.

In the above equations crossing symmetry is manifest. Concerning real analyticity, it can be checked by using, in particular, the analytic properties of the combination $\log(-z)(\sigma^2 - 1)^{-1/2}$. We will come back to the checks of real analyticity after we include the two-loop results in Section 6.



Fig. 14. The two-graviton cut needed to obtain the one-loop integrand for the amplitude in the classical limit. Each blob represents a tree-level amplitude with two scalars and two gravitons.

4.1.3. General relativity

Let us now turn to the collision of two massive scalars minimally coupled to GR. The logic we will apply is the same as for the N = 8 example discussed above. That is, we want to calculate the one-loop amplitude for the collision of two massive scalars minimally coupled to gravity, focusing on the non-analytic terms in the limit of small momentum transfer q^2 .

Of course, in the absence of supersymmetry, the amplitude integrand is not as simple as (4.8), and deriving it from Feynman diagram techniques would pose a nontrivial challenge. Following [48,108], a powerful integrand construction technique that we can apply to overcome this problem is the method of generalized unitarity. The one loop integrand we are after is a rational function, and its residues at the poles corresponding to the on-shell limits for certain internal lines are nothing but (sums of products of) tree-level amplitudes obtained by "cutting" such lines. A further simplification arises from the observation that, in order to capture all non-analytic terms associated to the long-range eikonal dynamics, we do not need to determine the full integrand, but rather only the contributions coming from the two-graviton cut depicted in Fig. 14. Indeed, we can neglect any topology involving contact interactions between the two massive lines, which would correspond to strictly localized effects.

We shall disregard topologies associated to vertex corrections and self-energy diagrams, which give rise to integrals that always vanish in the soft region. This may seem in principle not justified, because, although arising from the hard region only, these dressings of the one-graviton exchange naively appear in the amplitude with a kinematic prefactor $1/q^2$ (non-analytic). The key observation is that they also give rise to divergent terms that ought to be treated by appropriate coupling and wave-function renormalization. However one can check that the IR-divergences due to the two-graviton cut in Fig. 14 in fact exhaust the full IR-divergence of the one-loop amplitude predicted by the general exponential pattern [1,107]. Therefore, all infrared divergences arising from vertex corrections and self-energy diagrams must cancel against one another, and we need not worry about them. The cancellation of the corresponding finite terms then ought to follow from the Ward identity linking the renormalization constants for charge and wavefunction, as discussed in Ref. [257] for the case of electrodynamics, although we will not analyze it in detail here.

Let us then turn to the evaluation of the two-graviton cut. The amplitude for each blob in Fig. 14 is the one already discussed in Section 3.1.2,

$$\mathcal{A}_{\rho\sigma,\alpha\beta}(k_1,k_2,q_1,q_2) = \underset{k_1}{\overset{\rho\sigma}{\longleftarrow}} \overset{q_1}{\underset{\leftarrow}{\longleftarrow}} \overset{q_2}{\underset{\leftarrow}{\longleftarrow}} \overset{\alpha\beta}{\underset{\leftarrow}{\longleftarrow}} k_2$$
(4.35)

can be taken as follows, using momentum conservation to eliminate k_2 , [48]

$$\mathcal{A}_{\rho\sigma,\alpha\beta} = 2\kappa^{2} \frac{q_{1} \cdot (k_{1} + q_{2})q_{1} \cdot k_{1}}{q_{1} \cdot q_{2}} \times \left[\frac{(k_{1} + q_{2})^{\rho}k_{1}^{\alpha}}{q_{1} \cdot (k_{1} + q_{2})} - \frac{(k_{1} + q_{1})^{\alpha}k_{1}^{\rho}}{k_{1} \cdot q_{1}} + \eta^{\rho\alpha} \right] \left[\frac{(k_{1} + q_{2})^{\sigma}k_{1}^{\beta}}{q_{1} \cdot (k_{1} + q_{2})} - \frac{(k_{1} + q_{1})^{\beta}k_{1}^{\sigma}}{k_{1} \cdot q_{1}} + \eta^{\sigma\beta} \right].$$

$$(4.36)$$

This expression is of course highly non-unique, because one can always perform gauge transformations, i.e. shifts proportional to $q_1^{\rho}\xi^{\sigma} + q_1^{\sigma}\xi^{\rho}$ or to $q_2^{\alpha}\zeta^{\beta} + q_2^{\beta}\zeta^{\alpha}$ obtaining an equivalent amplitude. The advantage of the specific form



Fig. 15. Triangle and bubble topologies.

(4.36) is that it is exactly transverse with respect to each graviton momentum [48,258],

$$q_1^{\rho}\mathcal{A}_{\rho\sigma,\alpha\beta} = q_1^{\sigma}\mathcal{A}_{\rho\sigma,\alpha\beta} = 0, \qquad q_2^{\alpha}\mathcal{A}_{\rho\sigma,\alpha\beta} = q_2^{\beta}\mathcal{A}_{\rho\sigma,\alpha\beta} = 0,$$
(4.37)

as can be checked using the mass-shell conditions. This simplifies the sum over intermediate graviton polarizations involved in the cut in Fig. 14, and one can cast the amplitude of interest in the form

$$i\mathcal{A}_{1}(s,q) = \frac{1}{2} \int \frac{d^{D}q_{1}}{(2\pi)^{D}} \frac{d^{D}q_{2}}{(2\pi)^{D}} (2\pi)^{D} \delta^{(D)}(q-q_{1}-q_{2}) \times \mathcal{A}_{\rho\sigma,\alpha\beta}(p_{1},p_{4},-q_{1},-q_{2}) G^{\rho\sigma,\rho'\sigma'}(q_{1}) G^{\alpha\beta,\alpha'\beta'}(q_{2}) \mathcal{A}_{\rho'\sigma',\alpha'\beta'}(p_{2},p_{3},q_{1},q_{2}),$$
(4.38)

where $G^{\mu\nu,\rho\sigma}$ denotes the De Donder propagator (A.23),

$$G^{\mu\nu,\rho\sigma}(\ell) = \frac{-i}{2(\ell^2 - i0)} \left(\eta^{\mu\rho} \eta^{\nu\sigma} + \eta^{\mu\sigma} \eta^{\nu\rho} - \frac{2}{D-2} \eta^{\mu\nu} \eta^{\rho\sigma} \right).$$
(4.39)

The overall factor of $\frac{1}{2}$ in (4.38) is due to the reflection symmetry of Fig. 14 about the vertical axis, or equivalently to the two possible choices of labeling for the loop momentum, either $q_1 = \ell$, $q_2 = \ell - q$ or $q_2 = \ell$, $q_1 = q - \ell$.

The resulting integrand involves not only box and crossed box topologies (Fig. 13), as its $\mathcal{N} = 8$ counterpart, but also triangle and bubble topologies (Fig. 15). Moreover, while the numerators of box and crossed box are constant in the loop momentum, the remaining topologies involve nontrivial numerators. As we already mentioned above, the task of performing such integrals is considerably simplified by employing integration-by-parts identities after expansion in the soft region. Using these techniques, one can reduce the calculation to a simple set of three master integrals [167], the leading-order box and triangle, plus a trivial scalar bubble integral.

For illustrative purposes, let us consider a scalar triangle integral (let us recall the shorthand notation \int_{ℓ} introduced in (4.11))

$$I_{\Delta} = \int_{\ell} \frac{1}{(-2p_1\ell + \ell^2 - i0)(\ell^2 - i0)((\ell - q)^2 - i0)} \,. \tag{4.40}$$

Again focusing on the soft region, where $\ell \sim O(q)$, and on the leading-order contributions, we can massage this integral as we did for the leading box, obtaining

$$I_{\Delta} \simeq \frac{1}{2} \int_{\ell} \left[\frac{1}{2p_1\ell - i0} + \frac{1}{-2p_1\ell - i0} \right] \frac{1}{(\ell^2 - i0)((\ell - q)^2 - i0)}$$
(4.41)

so that

$$I_{\Delta} \simeq i\pi \int_{\ell} \frac{\delta(2p_1\ell)}{(\ell^2 - i0)((\ell - q)^2 - i0)} \,. \tag{4.42}$$

To evaluate this integral it is convenient to pick a reference frame where

$$p_1 = \left(-\sqrt{m_1^2 + \frac{1}{4}|\vec{q}|^2}, \frac{1}{2}\vec{q}\right), \qquad p_4 = \left(\sqrt{m_1^2 + \frac{1}{4}|\vec{q}|^2}, \frac{1}{2}\vec{q}\right)$$
(4.43)

so that $q = (0, \vec{q})$ and, to leading order for small q,

$$I_{\Delta} \simeq e^{\gamma_{E}\epsilon} \frac{\pi^{-1+\epsilon}}{2m_{1}} \int \frac{d^{3-2\epsilon}\vec{\ell}}{|\vec{\ell}|^{2}|\vec{\ell}-\vec{q}|^{2}} = e^{\gamma_{E}\epsilon} \frac{\sqrt{\pi}}{2m_{1}} \frac{\Gamma\left(\frac{1}{2}+\epsilon\right)}{(q^{2})^{\frac{1}{2}+\epsilon}} \frac{\Gamma\left(\frac{1}{2}-\epsilon\right)^{2}}{\Gamma\left(1-2\epsilon\right)}.$$

$$(4.44)$$

Combining this result with the box and bubble integrals, one can obtain an expression for the amplitude complete up to sub-sub-leading order in the small-*q* expansion, which we now illustrate. For simplicity, let us again break down the result as follows

$$\mathcal{A}_1 = \mathcal{A}_1^{[2]} + \mathcal{A}_1^{[1]} + \mathcal{A}_1^{[0]}, \qquad (4.45)$$

where each term scales according to $A_1^{(n)} \sim q^{-n-2\epsilon}$. The leading term, which is entirely determined by the box and crossed-box contributions, and is given by

$$\frac{i\mathcal{A}_{1}^{[2]}}{4pE} = -\frac{(8\pi G)^{2}}{(4\pi)^{2}} \left(\frac{4\pi}{q^{2}}\right)^{\epsilon} \frac{2\pi m_{1}^{2}m_{2}^{2}(\sigma^{2} - \frac{1}{D-2})^{2}}{q^{2}(\sigma^{2} - 1)} \frac{\Gamma(1+\epsilon)\Gamma^{2}(-\epsilon)}{\Gamma(-2\epsilon)}.$$
(4.46)

The subleading one reads instead

$$\frac{i\mathcal{A}_{1}^{[1]}}{4pE} = \frac{i(8\pi G)^{2}}{(4\pi)^{2}} \left(\frac{4\pi}{q^{2}}\right)^{\epsilon} \frac{\Gamma(\epsilon + \frac{1}{2})\Gamma(\frac{1}{2} - \epsilon)^{2}}{\Gamma(1 - 2\epsilon)} \left[-\frac{4\epsilon\sqrt{\pi}m_{1}m_{2}(m_{1} + m_{2})(\sigma^{2} - \frac{1}{D-2})^{2}}{q(\sigma^{2} - 1)^{\frac{3}{2}}} + \frac{2\sqrt{\pi}m_{1}m_{2}(m_{1} + m_{2})}{q(\sigma^{2} - 1)^{\frac{1}{2}}} \left(\sigma^{2} - \frac{4 + (1 - 2\epsilon)(\sigma^{2} - 1)}{16(1 - \epsilon)^{2}}\right) \right],$$
(4.47)

where the first line of this expression comes from the subleading expansion of box and crossed box, while the second line comes from the leading-order triangle contributions. Note that the first line vanishes in D = 4, as $\epsilon \rightarrow 0$. The subsubleading term instead combines all three types of topologies, and we find it convenient separate box/crossed box, triangle and bubble contributions according to

$$\mathcal{A}_{1}^{[0]} = \mathcal{A}_{1\Box}^{[0]} + \mathcal{A}_{1\Delta}^{[0]} + \mathcal{A}_{1\Box}^{[0]} \,. \tag{4.48}$$

We find, from the box and crossed box topologies [43],

$$\frac{i\mathcal{A}_{1\Box}^{[0]}}{4pE} = \frac{(8\pi G)^2}{(4\pi)^2} \left(\frac{4\pi}{q^2}\right)^{\epsilon} \left(\sigma^2 - \frac{1}{D-2}\right) \frac{\Gamma(1-\epsilon)^2 \Gamma(1+\epsilon)}{\Gamma(1-2\epsilon)} \left[\frac{-\pi s \left(\sigma^2 - \frac{1}{D-2}\right)}{\left(\sigma^2 - 1\right)^2} + \frac{2im_1 m_2}{\epsilon \sqrt{\sigma^2 - 1}} \left(\frac{4\sigma \operatorname{arccosh} \sigma}{\sqrt{\sigma^2 - 1}} + \frac{(1+2\epsilon) \left(\sigma^2 - \frac{1}{D-2}\right)}{\sigma^2 - 1} \left(1 - \frac{\sigma \operatorname{arccosh} \sigma}{\sqrt{\sigma^2 - 1}}\right)\right)\right],$$

$$(4.49)$$

from the triangle topologies,

$$\frac{i\mathcal{A}_{1\Delta}^{[0]}}{4pE} = \frac{(8\pi G)^2}{(4\pi)^2} \left(\frac{4\pi}{q^2}\right)^{\epsilon} \frac{im_1m_2}{2\sqrt{\sigma^2 - 1}} \frac{\Gamma(1 - \epsilon)^2 \Gamma(2 + \epsilon)}{\epsilon(1 - \epsilon)^2 \Gamma(1 - 2\epsilon)} \times \left[1 - 2\sigma^2(1 - \epsilon)\frac{11 - 18\epsilon + 8\epsilon^2}{(1 + \epsilon)(1 - 2\epsilon)}\right],$$
(4.50)

and, from the bubble,

[0]

$$\frac{i\mathcal{A}_{1\bigcirc}^{[0]}}{4pE} = \frac{(8\pi G)^2}{(4\pi)^2} \left(\frac{4\pi}{q^2}\right)^{\epsilon} \frac{im_1m_2}{\sqrt{\sigma^2 - 1}} \frac{\Gamma(1 - \epsilon)^2 \Gamma(1 + \epsilon)}{4\epsilon(5 - 2\epsilon)(3 - 2\epsilon)\Gamma(1 - 2\epsilon)} \times \left[\frac{\sigma^2 \left(16\epsilon^3 - 210\epsilon^2 + 633\epsilon - 522\right)}{2\epsilon - 1} - \frac{17\epsilon^3 - 68\epsilon^2 + 65\epsilon - 2}{2(\epsilon - 1)^2}\right].$$
(4.51)

We can now go to impact parameter space using Eq. (A.45). Let us start from the superclassical term $\mathcal{A}_{12}^{[2]}$ in (4.46), which gives

$$i\tilde{\mathcal{A}}_{1}^{[2]}(s,b) = \frac{1}{2} \left(\frac{2im_{1}m_{2}G(\pi b^{2})^{\epsilon}(\sigma^{2} - \frac{1}{D-2})\Gamma(-\epsilon)}{\sqrt{\sigma^{2} - 1}} \right)^{2} = \frac{1}{2}(2i\delta_{0})^{2} .$$
(4.52)

Thus, as in the previous case, we again obtain a cross check of the exponentiation of δ_0 , according to which the leading superclassical term in the one-loop amplitude must be given by $\frac{1}{2!}(2i\delta_0)^2$ as dictated by (4.5). The Fourier transform of the subleading, classical term $\mathcal{A}_1^{[1]}$, appearing in (4.47), gives the next to the leading classical

eikonal

$$2i\delta_1 = i\tilde{\mathcal{A}}_1^{[1]}(s,b)\,. \tag{4.53}$$

so that

$$2i\delta_1 = \frac{iG^2m_1m_2(m_1+m_2)}{(\pi b^2)^{\frac{1}{2}-2\epsilon}} \left[\frac{\Gamma^2(\frac{1}{2}-\epsilon)\Gamma(\frac{1}{2}-2\epsilon)\left(\sigma^2-\frac{1}{D-2}\right)^2}{\Gamma(-2\epsilon)(\sigma^2-1)^{\frac{3}{2}}} \right]$$
(4.54)

$$+ \frac{4\Gamma(\frac{1}{2} - \epsilon)^{2}\Gamma(\frac{1}{2} - 2\epsilon)}{\Gamma(1 - 2\epsilon)\sqrt{\sigma^{2} - 1}} \left(\sigma^{2} - \frac{4 + (1 - 2\epsilon)(\sigma^{2} - 1)}{16(1 - \epsilon)^{2}}\right) \right].$$
(4.55)

Like the leading eikonal, $2\delta_1$ is also a purely real quantity. Taking a derivative with respect to b according to

$$2p\sin\frac{\Theta}{2} = -\frac{\partial 2\delta}{\partial b} \implies \Theta^{2PM} = -\frac{1}{p}\frac{\partial 2\delta_1}{\partial b}, \qquad (4.56)$$

we can then obtain the 2PM correction to the deflection angle, in generic spacetime dimensions. Combining with the 1PM result (3.12), we thus obtain the following expression for the deflection angle,

$$\Theta = \frac{4GE\left(\sigma^{2} - \frac{1}{D-2}\right)\Gamma(1-\epsilon)}{(\sigma^{2} - 1)\pi^{-\epsilon}b^{1-2\epsilon}} + \frac{(1-2\epsilon)G^{2}E(m_{1}+m_{2})}{\pi^{\frac{1}{2}-2\epsilon}b^{2-2\epsilon}} \left[\frac{\Gamma^{2}(\frac{1}{2}-\epsilon)\Gamma(\frac{1}{2}-2\epsilon)\left(\sigma^{2} - \frac{1}{D-2}\right)^{2}}{\Gamma(-2\epsilon)(\sigma^{2}-1)^{2}} + \frac{4\Gamma(\frac{1}{2}-\epsilon)^{2}\Gamma(\frac{1}{2}-2\epsilon)}{\Gamma(1-2\epsilon)(\sigma^{2}-1)}\left(\sigma^{2} - \frac{4+(1-2\epsilon)(\sigma^{2}-1)}{16(1-\epsilon)^{2}}\right)\right] + \mathcal{O}(G^{3}).$$
(4.57)

Note that the 2PM correction is present only if both masses are non vanishing. In contrast with the N = 8 case, although the box and cross-box contributions (4.54) do not contribute for D = 4, the triangle contribution (4.55) survives, and we get:

$$2\delta_1 = \frac{3\pi G^2 m_1 m_2 (m_1 + m_2)(5\sigma^2 - 1)}{4b\sqrt{\sigma^2 - 1}}$$
(4.58)

and for the deflection angle [259]

$$\Theta = \frac{4GE(\sigma^2 - \frac{1}{2})}{b(\sigma^2 - 1)} + \frac{3\pi G^2 E(m_1 + m_2)(5\sigma^2 - 1)}{4(\sigma^2 - 1)b^2} + \mathcal{O}(G^3).$$
(4.59)

Finally the Fourier transform of the sub-sub-leading term $\mathcal{A}_{1}^{[0]}$ given by (4.48) identifies the leading quantum remainder

$$2i\Delta_1 = i\tilde{\mathcal{A}}_1^{[0]}. \tag{4.60}$$

This is a quantum term and indeed will not contribute to the classical eikonal. However, it will be needed at two-loop level in order to solve (4.6) for the unknown $2\delta_2$. Actually, since we will be able to obtain the two-loop amplitude only to the first few orders in the ϵ expansion around $\epsilon = 0$, i.e. D = 4, let us evaluate this quantum remainder for small ϵ , although performing the Fourier transform of the full expression (4.48) is straightforward using (A.45). We thus obtain

$$2i\Delta_{1} = -\epsilon \frac{G^{2}s(2\sigma^{2}-1)^{2}}{b^{2}(\sigma^{2}-1)^{2}} + \frac{iG^{2}m_{1}m_{2}(\pi b^{2}e^{\gamma_{E}})^{2\epsilon}}{b^{2}\pi(\sigma^{2}-1)^{3/2}} \left(\frac{2\sigma(2\sigma^{2}-1)(6\sigma^{2}-7)\operatorname{arccosh}\sigma}{\sqrt{\sigma^{2}-1}} - \frac{1-49\sigma^{2}+18\sigma^{4}}{15}\right) + i\epsilon \frac{G^{2}m_{1}m_{2}}{\pi b^{2}\sqrt{\sigma^{2}-1}} \left(-\frac{8\sigma(2\sigma^{2}+1)\operatorname{arccosh}\sigma}{(\sigma^{2}-1)^{3/2}} + \frac{9234\sigma^{2}-1783}{450}\right) + \mathcal{O}(\epsilon^{2}).$$

$$(4.61)$$

This expression for $2i\Delta_1$ is accurate through $\mathcal{O}(\epsilon)$ for both real and imaginary part. In particular its $\mathcal{O}(1)$ real part vanishes, while its imaginary part is nontrivial to $\mathcal{O}(1)$ and $\mathcal{O}(\epsilon)$. The terms involving $\operatorname{arccosh} \sigma$ of course only come from the box/crossed box contributions (4.49) already obtained in [48]. Moreover, the $\mathcal{O}(1)$ imaginary part and the $\mathcal{O}(\epsilon)$ real part agree with [109]. As we will discuss in the next chapter, this quantum piece will play a role in our discussion of the two-loop eikonal. On the one hand, it is needed to perform the subtractions dictated by (4.6). Moreover, it will be instrumental in determining the amplitude itself from its analyticity and crossing-symmetry properties.

4.1.4. Real-analytic, crossing-symmetric reformulation

Following the procedure we used for the N = 8 case in Section 4.1.2, let us recast the 3PM GR result in an explicitly real-analytic and crossing-symmetric form. Using the same notations as in (4.29) the tree level amplitude (3.4) can be written, up to higher orders in q^2 , as

$$\mathcal{A}_{0}(s,q^{2}) = \frac{16\pi G}{q^{2}} m_{1}^{2} m_{2}^{2} \Big[\left(\sigma^{2} - \frac{1}{D-2} \right) + \left(\bar{\sigma}^{2} - \frac{1}{D-2} \right) \Big].$$
(4.62)

At one-loop, in analogy with Eq. (4.20), we can deal separately and rather trivially with the purely real classical terms (4.47) behaving like $\frac{1}{\sqrt{q^2}}$. For the remaining terms, inspired again by the $\mathcal{N} = 8$ case of Section 4.1.2, we try the

ansatz:

$$\mathcal{A}_{1}^{\text{scl.}}(s,q^{2}) = \frac{(8\pi G)^{2}}{(4\pi)^{2}} \left(\frac{4\pi}{q^{2}}\right)^{\epsilon} \frac{\Gamma^{2}(-\epsilon)\Gamma(1+\epsilon)}{\Gamma(-2\epsilon)} \frac{4m_{1}^{2}m_{2}^{2}}{q^{2}}$$

$$\left[\left(\sigma^{2} - \frac{1}{2(1-\epsilon)}\right)^{2} + \left(\bar{\sigma}^{2} - \frac{1}{2(1-\epsilon)}\right)^{2} \right] \left[\frac{\log(-z)}{\sqrt{\sigma^{2}-1}} + \frac{\log(-\bar{z})}{\sqrt{\bar{\sigma}^{2}-1}} \right].$$
(4.63)

This reproduces, in *b*-space, the iteration of the tree contribution (4.46) and, furthermore, takes care of some quantum terms. Up to purely real pieces, the full structure of (4.48) is reproduced by the following additional real-analytic crossing-symmetric terms:

$$\begin{aligned} \mathcal{A}_{1}^{\text{add.}}(s,q^{2}) &= \frac{(8\pi G)^{2}}{(4\pi)^{2}} \left(\frac{4\pi}{q^{2}}\right)^{\epsilon} \frac{\Gamma^{2}(-\epsilon)\Gamma(1+\epsilon)}{\Gamma(-2\epsilon)} 4m_{1}^{2}m_{2}^{2}\hat{\mathcal{A}}_{1}^{\text{add.}}, \end{aligned} \tag{4.64} \\ \hat{\mathcal{A}}_{1}^{\text{add.}} &= \left[\left(\sigma^{2} - \frac{1}{2(1-\epsilon)}\right) + \left(\bar{\sigma}^{2} - \frac{1}{2(1-\epsilon)}\right) \right] \left[\frac{\sigma \log(-z)}{\sqrt{\sigma^{2}-1}} + \frac{\bar{\sigma}\log(-\bar{z})}{\sqrt{\bar{\sigma}^{2}-1}} \right] \\ &- \frac{\epsilon}{2} \left[\left(\sigma^{2} - \frac{1}{2(1-\epsilon)}\right)^{2} + \left(\bar{\sigma}^{2} - \frac{1}{2(1-\epsilon)}\right)^{2} \right] \left[\frac{\sigma \log(-z)}{(\sigma^{2}-1)^{3/2}} + \frac{\bar{\sigma}\log(-\bar{z})}{(\bar{\sigma}^{2}-1)^{3/2}} \right] \\ &- \epsilon \frac{m_{1}^{2} + m_{2}^{2}}{4m_{1}m_{2}} \left[\left(\sigma^{2} - \frac{1}{2(1-\epsilon)}\right)^{2} + \left(\bar{\sigma}^{2} - \frac{1}{2(1-\epsilon)}\right)^{2} \right] \left[\frac{\log(-z)}{(\sigma^{2}-1)^{3/2}} + \frac{\log(-\bar{z})}{(\bar{\sigma}^{2}-1)^{3/2}} \right]. \end{aligned}$$

We have checked that the sum of (4.63) and (4.64) reproduces the superclassical term (4.46), as well as the box and crossed box contributions (4.49), modulo a single real term given by:

$$\hat{\mathcal{A}}_{1}^{\text{res.}} = -2\epsilon \frac{\left(\sigma^{2} - \frac{1}{D-2}\right)^{2}}{\sigma^{2} - 1} \,. \tag{4.65}$$

We also note that the triangle (4.50) and bubble (4.51) contributions are real (indeed they do not exhibit any on shell intermediate states). Together with (4.65) they can be made crossing symmetric trivially. This exhausts the analytic study of the one-loop terms in GR.

4.1.5. The probe limit

It is instructive to study the regime where one mass is much larger than all other energy scales as was done at leading PM order at the end of Section 3.1.1 for GR and of Section 3.1.4 for $\mathcal{N} = 8$ supergravity. Of course in this limit one should reproduce the results obtained from the classical motion of a probe in a fixed background describing the heavy object, see Appendix B.1. This is easily checked by taking the limit $m_2 \gg E_1$ summarized in (3.17). Focusing first on the case of GR, it is sufficient to take (4.57) in the probe limit, $m_1 = m_p \ll M = m_2$, to obtain

$$\Theta = \frac{\sqrt{\pi} \Gamma\left(\frac{D-2}{2}\right)}{2\Gamma\left(\frac{D-1}{2}\right)} \frac{(D-2)E^2 - m_p^2}{E^2 - m_p^2} \left(\frac{R_s}{b}\right)^{D-3} + \frac{\sqrt{\pi} \Gamma\left(D - \frac{5}{2}\right)}{8\Gamma(D-2)} \left(\frac{R_s}{b}\right)^{2(D-3)} \frac{(2D-5)(2D-3)E_p^4 - 6(2D-5)E_p^2m_p^2 + 3m_p^4}{(E_p^2 - m_p^2)^2} + \mathcal{O}(G^3)$$

$$(4.66)$$

where we used (B.2) for R_s . The leading term matches (B.14) while the subleading correction matches (B.15). It is straightforward to repeat the check in the case of $\mathcal{N} = 8$ supergravity by starting from (4.25), for which we get

$$\Theta = \frac{\sqrt{\pi} \Gamma\left(\frac{D-2}{2}\right)}{\Gamma\left(\frac{D-2}{2}\right)} \frac{E^2}{E^2 - m_p^2} \left(\frac{R_s}{b}\right)^{D-3} + \frac{\sqrt{\pi} \Gamma\left(D - \frac{5}{2}\right)}{\Gamma\left(D - 4\right)} \left(\frac{E^2}{E^2 - m_p^2}\right)^2 \left(\frac{R_s}{b}\right)^{2(D-3)} + \mathcal{O}(G^3).$$
(4.67)

As already noticed, for D = 4 the 2PM correction in (4.67) vanishes and the 1PM result agrees with the probe limit (B.41).

At 2PM order, it is actually possible to make the link in the opposite direction and reconstruct the deflection angle for generic masses from the result in the probe limit. The idea is to consider the following ansatz for the classical impulse *Q* in an elastic scattering in the PM expansion [260],

$$Q = \frac{2Gm_1m_2}{J/p} \sum_{k_1,k_2=0}^{\infty} Q_{k_1k_2}(\sigma) \left(\frac{2Gm_1}{b_J}\right)^{k_1} \left(\frac{2Gm_2}{b_J}\right)^{k_2}$$
(4.68)

where we restrict to D = 4. The key assumption here, sometimes referred to as "good mass polynomiality", is that the dependence on the masses m_i always enters via the combination $2Gm_i/J$ and it is thus tied to the PM expansion itself.

Each coefficient $Q_{k_1k_2}(\sigma)$ is only a function of σ , which contributes to the result at PM order $k_1 + k_2 + 1$. Moreover, since the impulse Q must be symmetric under particle-interchange symmetry, in the elastic case, we must have

$$Q_{k_1k_2}(\sigma) = Q_{k_2k_1}(\sigma).$$
(4.69)

In the probe limit $m_1 \ll m_2$, in Eq. (4.68) only the terms with $k_1 = 0$ survive to leading order, and thus one can deduce the functions $Q_{0k}(\sigma) = Q_{k0}(\sigma)$ from the deflection angle calculated in Appendix B.1 by studying the motion of particle 1, with mass $m_1 = m_p$, in the background sourced by particle 2, with mass $m_2 = M$. In the case of GR, by (B.17), (B.18) we have at the first two orders

$$Q_{00}(\sigma) = \frac{2\sigma^2 - 1}{\sqrt{\sigma^2 - 1}}, \qquad Q_{01}(\sigma) = \frac{3\pi}{16} \frac{5\sigma^2 - 1}{\sqrt{\sigma^2 - 1}}, \tag{4.70}$$

where we used $E/m_p \simeq \sigma$ in the probe limit. Of course, substituting back Q_{00} and Q_{01} into (4.68) one recovers precisely the 2PM accurate deflection angle (4.59). On the contrary, at 3PM a new function $Q_{11}(\sigma)$ appears that cannot be fixed by looking at the probe limit, and needs to be determined by studying the problem where both particles are fully dynamical. This will be the subject of Section 6 for both $\mathcal{N} = 8$ supergravity and GR. Moreover, as we shall see in Section 8, in general the impulse Q_1 of particle 1 and that Q_2 of particle 2 can differ in the presence of radiative effects, thus introducing additional structures. For recent developments concerning the assumption of good mass polynomiality to $\mathcal{O}(G^4)$ see [138,140,261].

4.1.6. Tidal effects in field theory

Following Refs. [50,51,54,262], one can conveniently include finite-size effects associated to tidal deformations in the scattering amplitude approach by introducing higher derivative operators that are quadratic in the scalar field and involve powers of the Weyl tensor. Focusing for simplicity on the leading, quadratic order, these can be decomposed into the so-called *E* ("electric") or mass-type and *B* ("magnetic") or current-type tidal operators. The resulting "tidal vertex" involves two massive lines and two graviton lines, and thus produces corrections to the sewing procedure of Fig. 14 (while it does not affect the single-graviton, tree-level exchange). Without going into details, let us quote here for completeness the expression for the resulting one-loop correction to the impulse, which are of course equivalent to the tidal modifications of $2\delta_1$ up to a derivative with respect to *b*. For the case in which particle 1 is subject to tidal deformations (the analogous case in which object 2 can be deformed is obtained trivially by interchanging particle labels),

$$Q_{E_1^2} = \frac{R_f b}{G} \frac{3c_{E_1^2}}{m_1^2} \frac{35\sigma^4 - 30\sigma^2 + 11}{\sqrt{\sigma^2 - 1}}, \qquad Q_{B_1^2} = \frac{R_f b}{G} \frac{15c_{B_1^2}}{m_1^2} \sqrt{\sigma^2 - 1} \left(7\sigma^2 + 1\right), \tag{4.71}$$

with (i = 1, 2)

$$R_f = 15\pi G^3 m_1^2 m_2^2 / (64b^7) , \qquad c_{E_i^2} = \frac{1}{6} k_i^{(2)} R_i^5 / G , \qquad c_{B_i^2} = \frac{1}{32} j_i^{(2)} R_i^5 / G$$
(4.72)

and k_i , j_i are the colliding objects' Love numbers, while R_i is the radius of object *i*. For typical compact objects, $R_i = Gm_i/K_i$ where K_i is the body's "compactness" of order 0.1, 0.2 for neutron stars [30]. Thus, we see that the leading tidal corrections (4.71) are weighted by the dimensionless power-counting parameter $(Gm/b)^5/K^5$ (for $m \simeq m_{1,2}$ and $K \simeq K_{1,2}$) relative to the leading-order impulse (3.16). This shows how PM effects measured by Gm/b can compete with finite-size effects measured by the compactness parameter K.

4.2. Eikonal exponentiation in string theory at one loop

The first goal of this section is to analyze an explicit example of a one-loop amplitude in string theory. We will see that also in this case the leading term in the classical limit does not contain new information as it provides just a contribution towards the exponentiation of the tree-level result. The main difference with the QFT cases analyzed so far is that already the leading eikonal $\hat{\delta}_0$ (3.176) is an operator: as discussed in Section 3.2.4, this operator acts on the space of the possible string states. The non-trivial check presented here is that the leading term of the one-loop result is the square of $\hat{\delta}_0$, extending Eq. (4.6) to the more general setup where the colliding objects have a non-trivial structure. After gaining confidence that the full eikonal operator exponentiates, we revisit the physics related to tidal excitation. We show that, at the leading order, this classical effect is captured by a probe analysis focusing on the string dynamics in a non-trivial background. This generalizes to the string case the idea that the leading deflection angle for a point-like particle can be derived by solving the geodesic equation, see (3.20) and (B.14). We conclude this part on string theory by discussing more in detail how the exponentiated form of the leading eikonal captures the tidal effects on the string probe and how this can exponentially suppress the elastic scattering even at very large distances if the probe energy is large enough.

4.2.1. 1-loop in string theory

We work in the setup discussed in Section 3.2.1 and consider the first correction to the $1 \rightarrow 1$ scattering of a NS–NS massless state in presence of a stack of *N* coincident D*p*-branes in type II theories. From the world-sheet point of view,



Fig. 16. Scattering of a massless string off a Dp-brane at one loop.



Fig. 17. Alternative representation of the one-loop scattering of a closed string off a Dp-brane. The green circles are the boundaries of the worldsheet and rest on the Dp-brane, while the blue dots represent punctures associated to the asymptotic closed-string states.

this amplitude is captured by a diagram with the topology of an annulus with two boundaries (which are supported by the *Dp*-branes) and two punctures in the interior of the annulus representing the external closed string states, as depicted in Fig. 16. We will parameterize this surface as done in [22]: the "thickness" of the annulus is related to $e^{-\pi\lambda}$, with $0 \le \lambda < \infty$, while for the location of the punctures we use

$$z_{i} = e^{2\pi(-\lambda\rho_{i} + i\omega_{i})}, \qquad 0 < \rho_{i} < \frac{1}{2}, \qquad 0 \le \omega_{i} < 1.$$
(4.73)

(see Fig. 17). In type II theories, the behavior of this amplitude is constrained by supersymmetry and the kinematic dependence is the same as in the tree-level case [206], so we can extract a scalar function A_1

$$\mathcal{A}_1(p_i,\epsilon_i) = \mathcal{K}(p_i,\epsilon_i) \mathcal{A}_1 \tag{4.74}$$

that captures all the dynamical information. The derivation of A_1 is conceptually similar to the one summarized in Appendix D.3 for bosonic string theory, but technically requires to deal with some of the complications of superstring theory such as the sum over the spin structures. Here we start from the final result below and refer to the literature [263, 264] for a derivation and further references. In the superstring case we have

$$\mathcal{A}_{1} = \left(\frac{\kappa_{d}NT_{p}}{2}\right)^{2} \frac{\alpha'}{16\pi} (2\pi^{2}\alpha')^{-\frac{d-p-1}{2}} \int_{0}^{\infty} d\lambda \int d^{2}z_{1} d^{2}z_{2} \lambda^{-\frac{d-p-1}{2}} e^{-\alpha' E_{s}^{2} V_{s} - \frac{\alpha' t}{4} V_{t}} , \qquad (4.75)$$

with

$$V_{s} = -2\pi\lambda\rho^{2} + \log\left[\frac{\theta_{1}(i\lambda(\zeta+\rho)|i\lambda)\theta_{1}(i\lambda(\zeta-\rho)|i\lambda)}{\theta_{1}(i\lambda\zeta+\omega|i\lambda)\theta_{1}(i\lambda\zeta-\omega|i\lambda)}\right],$$

$$V_{t} = 8\pi\lambda\rho_{1}\rho_{2} + \log\left[\frac{\theta_{1}(i\lambda\rho+\omega|i\lambda)\theta_{1}(i\lambda\rho-\omega|i\lambda)}{\theta_{1}(i\lambda\zeta+\omega|i\lambda)\theta_{1}(i\lambda\zeta-\omega|i\lambda)}\right],$$
(4.76b)

where for the Jacobi theta-functions θ_i we follow the conventions of [265] where

$$\theta_1(\nu|\tau) = -2e^{\frac{\pi i\tau}{4}}\sin(\pi\nu)\prod_{n=1}^{\infty} \left[(1 - e^{2\pi in\tau})(1 - e^{2\pi i\nu}e^{2\pi in\tau})(1 - e^{-2\pi i\nu}e^{2\pi in\tau}) \right]$$
(4.77)

and we used the change of variables

$$\omega = \omega_1 - \omega_2 , \quad \rho = \rho_1 - \rho_2 , \quad \zeta = \rho_1 + \rho_2 . \tag{4.78}$$

Since we are interested in the long-range interaction between the stack of Dp-branes and the external probe, we wrote the amplitude in the so-called closed string channel where the large- λ limit is simple and describes a superposition of closed string states exchanged between the Dp-branes and the scattered closed string. In this limit the world-sheet has the shape of a sphere connected to the Dp-branes by two thin tubes and with two punctures representing the external states. The factor of $\lambda - \frac{d-p-1}{2}$ in (4.75) follows from the Gaussian integration of the center-of-mass momentum of the virtual closed string propagating between the two boundaries, which can be non-trivial only along the d-p-1 directions where the Dp-branes impose Dirichlet boundary conditions. After implementing the change of variables in (4.73) and (4.78), Eq. (4.75) reads

$$\mathcal{A}_{1} = \left(\frac{\kappa_{d}NT_{p}}{2}\right)^{2} \frac{\alpha'}{16\pi} (2\pi^{2}\alpha')^{-\frac{d-p-1}{2}} 2(2\pi)^{4} \int_{0}^{\infty} d\lambda \int_{0}^{1} d\zeta \int_{\mathcal{R}(\zeta)} d\rho \int_{0}^{1} d\omega \,\lambda^{-\frac{d-p-5}{2}} e^{-\alpha' E_{s}^{2} V_{s} - \frac{\alpha' t}{4} V_{t}} , \qquad (4.79)$$

where the region of integration for ρ depends on ζ : max{ $-\zeta, -(1-\zeta)$ } < ρ < min{ $\zeta, 1-\zeta$ }. The precise definition of $\mathcal{R}(\zeta)$ will not play any role in our analysis, since we focus on the leading contribution dominated by the $\rho \sim 0$ region, see however [228] for a more general setup where this point becomes relevant. As expected the integrand in (4.75) does not depend on $\omega_1 + \omega_2$ and so, in (4.79), we have trivially performed the integration over this variable. Our goal is to extract the leading contribution in the classical limit of the integral (4.75) and show that it provides the first term needed to exponentiate the 1PM stringy eikonal operator (3.176). In this context the classical parameters are R_p in (D.6), describing the gravitational backreaction of the D*p*-branes, and E_s , describing the probe state. So, as in the tree-level analysis, we are interested in the large E_s limit which from (4.79) is suppressed unless one takes also ρ small. Thus, for our purposes, we can use in (4.79) the approximate expressions

$$V_s \simeq -2\pi\lambda\rho^2 - 4\sin^2\pi\omega \left(e^{-2\pi\lambda\zeta} + e^{-2\pi\lambda(1-\zeta)}\right) , \qquad (4.80)$$

$$V_t \simeq -2\pi\lambda\zeta(1-\zeta) + \ln\left(4\sin^2\pi\omega\right) .$$

In the expression for V_t we kept only the terms that do not have any exponential factor involving λ , while for V_s we kept also the first exponential terms that become relevant in the region $\lambda \sim \ln(\alpha' E_s^2)$ and considered only the ρ -dependent terms that are enhanced by a factor of λ . In order to extract the classical terms contributing to the 2PM eikonal operator $\hat{\delta}_1$, one would need a better approximation for V_s and V_t as partially discussed in [22]. The complete expression for $\hat{\delta}_1$ at string level is not known, so we will not pursue further this analysis here.

By following closely the derivation discussed in Appendix D.3 for the bosonic string, we can rewrite (4.79) in terms of $d_{\perp} = (d - p - 2)$ integral involving the full tree-level amplitude (3.120) and a kernel written in the square bracket below

$$\mathcal{A}_{1} \simeq \frac{i}{4E_{s}} \int \frac{d^{d_{\perp}}\mathbf{k}}{(2\pi)^{d_{\perp}}} \mathcal{A}_{0}(E_{s},\mathbf{k}) \mathcal{A}_{0}(E_{s},\mathbf{q}-\mathbf{k}) \left[\frac{2^{\alpha'\mathbf{k}(\mathbf{q}-\mathbf{k})}}{\pi} B\left(\frac{1}{2} + \frac{\alpha'}{2}\mathbf{k}(\mathbf{q}-\mathbf{k}),\frac{1}{2}\right) \right] .$$
(4.81)

The structure is familiar from the QFT case: the convolution over **k** becomes just a standard product in impact parameter space, the factor of $2E_s$ is the one needed to pass from the amplitude to the eikonal (see Eq. (3.123)) and the extra factor of $\frac{i}{2}$ comes from the expansion at the second order of the exponential with the leading eikonal (see the first term on the r.h.s. of (4.5)). The kernel in the square parenthesis is a string effect as one can see by checking that in the $\alpha' \rightarrow 0$ limit it becomes the identity plus subleading corrections

$$\left[\frac{2^{\alpha'\mathbf{k}(\mathbf{q}-\mathbf{k})}}{\pi}B\left(\frac{1}{2}+\frac{\alpha'}{2}\mathbf{k}(\mathbf{q}-\mathbf{k}),\frac{1}{2}\right)\right]\simeq 1+\frac{\pi^2}{6}\left(\frac{\alpha'}{2}\mathbf{k}\cdot(\mathbf{q}-\mathbf{k})\right)^2+\cdots$$
(4.82)

In order to completely factorize the result in (4.81), it is sufficient to rewrite the kernel as an expectation value of operators written in terms of the string coordinates \hat{X} introduced in Eq. (3.174)

$$\frac{1}{\pi}B\left(\frac{1}{2}+\frac{\alpha'}{2}\mathbf{k}_{1}\mathbf{k}_{2},\frac{1}{2}\right)=2^{-\alpha'\mathbf{k}_{1}\mathbf{k}_{2}}\langle0|\prod_{i=1}^{2}\int_{0}^{2\pi}\frac{d\sigma_{i}}{2\pi}:e^{i\mathbf{k}_{1}\hat{X}(\sigma_{1})}::e^{i\mathbf{k}_{2}\hat{X}(\sigma_{2})}:|0\rangle,$$
(4.83)

where the fields \hat{X} were introduced in (3.174) and contain the physical excitations of the string involving the DDF oscillators, see Section 3.2.4 for a general discussion. Then the interpretation of the kernel appearing in this string theory case is clear: it is related to the possibility of having excited string states in the intermediate steps. Even when focusing

on a particular elastic amplitude, where the initial and the final states are dilatons, the gravitational field of the D*p*brane can stretch the string probe and excite it to a different state while it propagates. As already discussed, this is an entirely classical effect capturing the tidal forces on an extended probe [244]. As a further check of this interpretation, this phenomenon can be quantitatively described by quantizing the string worldsheet action in the background produced by the D*p*-branes, as we will discuss in Section 4.2.3. However, as usual with the eikonal exponentiation, classical effects provide constraints on loop amplitudes and in particular in this case the leading contribution (as $\alpha' E_s^2$ becomes large) of the *h*-loop diagram (i.e. a worldsheet with h + 1 boundaries and no handles³⁵) should match the following convolution involving the tree-level result and the kernel discussed above

$$\frac{\mathcal{A}_{h}^{L}}{2E_{s}} = \frac{i^{h}}{(h+1)!} \langle 0| \prod_{i=1}^{h+1} \int \frac{d^{d_{\perp}} \mathbf{k}_{i}}{(2\pi)^{d_{\perp}}} \int_{0}^{2\pi} \frac{d\sigma_{i}}{2\pi} \frac{\mathcal{A}_{0}(E_{s}, \mathbf{k}_{i})}{2E_{s}} : e^{i\mathbf{k}_{i}\hat{X}(\sigma_{i})} : |0\rangle \,\delta^{d_{\perp}} \left(\sum_{i=1}^{h+1} \mathbf{k}_{i} - \mathbf{q}\right) \,. \tag{4.84}$$

This indeed reduces to (4.81) for h = 1 (one loop) thanks to (4.83). Thanks to this factorized form of the leading contributions, it is straightforward to resum them in terms of the leading eikonal operator that was derived by studying the inelastic transition at tree level in Section 3.2.4. As usual, after going to impact parameter space by taking the Fourier transform (3.123) the convolution becomes a product and we obtain

$$\sum_{h=0}^{\infty} \frac{i\tilde{\mathcal{A}}_{h}^{L}}{2E_{s}} = \langle 0| \left[e^{2i\hat{\delta}_{0}} - 1 \right] |0\rangle \quad \Rightarrow \quad S = e^{2i\hat{\delta}_{0}} , \qquad (4.85)$$

where the operator $\hat{\delta}_0$ was introduced in Eq. (3.176) where the impact parameter *b* is shifted by the corresponding string coordinate \hat{X} . Then, at leading PM order, the evolution of an initial (light) string probe interacting with the D*p*-branes is captured by the S-matrix *S* above. Notice that while $\hat{\delta}_0$ is normal ordered, this is not the case for the evolution operator *S*, and this plays an important role in the analysis of tidal forces as discussed in the next section.

4.2.2. Tidal forces in string theory

In this subsection we mainly discuss the suppression of the elastic channel as a consequence of the tidal excitations of the colliding string. This is a direct consequence of the eikonal operator derived in the previous section: as we will see, when this operator is used to estimate an elastic amplitude a dumping factor arises from the normal ordering of the oscillators. A brief summary on the nature of the inelastic channels is postponed to Section 4.2.5 where the same problem is considered for the case of string–string collisions. Thus, while leading eikonal for a point-like scalar object is a pure phase and describes an elastic scattering, in the string case a new dynamical scale b_t appears: when the impact parameter is below this scale the elastic scattering is exponentially suppressed as the internal degrees of freedom of the string are excited during the scattering. In formulae we have

$$|\langle 0|S|0\rangle| \sim \exp\left[-\left(\frac{b_t}{b}\right)^{d-p-2}\right], \qquad b_t^{d-p-2} = \frac{\sqrt{\pi}E_s}{4T}(d-p-3)\frac{\Gamma\left(\frac{d-p-2}{2}\right)}{\Gamma\left(\frac{d-p-3}{2}\right)}R_p^{d-p-3},$$
(4.86)

where we approximated *S* by the leading eikonal operator as in (4.85) and wrote b_t in terms of the string tension $T = \frac{1}{2\pi\alpha'}$ to stress that it is a classical quantity. See also the discussion around Eq. (4.120) below.

Notice that at high energies b_t is parametrically larger than the string scale so tidal forces can become relevant also at large distances. In this regime we can expand the leading eikonal operator (3.176) by taking $|\hat{X}| \ll b$: at the zeroth order, one of course recovers the point-like eikonal (3.185), the linear term is absent since the integral over σ of \hat{X} in (3.174) vanishes, and the leading effects are encoded by the quadratic term. We have

$$\frac{1}{2}\frac{\partial^2 \delta_0(b)}{\partial b_i \partial b_j} = \frac{1}{2} \left\{ \mu_{\hat{y}}^2 \left[\delta_{ij} - \frac{b_i b_j}{b^2} \right] + \mu_0^2 \frac{b_i b_j}{b^2} \right\},\tag{4.87}$$

where

$$\mu_{\hat{y}}^{2} = -\frac{\sqrt{\pi}E_{s}}{2} \frac{\Gamma\left(\frac{d-p-2}{2}\right)}{\Gamma\left(\frac{d-p-3}{2}\right)} \frac{R_{p}^{d-p-3}}{b^{d-p-2}},$$

$$\mu_{0}^{2} = \frac{\sqrt{\pi}E_{s}}{2} (d-p-3) \frac{\Gamma\left(\frac{d-p-2}{2}\right)}{\Gamma\left(\frac{d-p-3}{2}\right)} \frac{R_{p}^{d-p-3}}{b^{d-p-2}} = -(d-p-3)\mu_{\hat{y}}^{2}.$$
(4.88)

Note that $\mu_{\hat{y}}^2 < 0$ while $\mu_0^2 > 0$, but this will not play an important role in the following. Then at the second order in the $|\hat{X}| \ll b$ expansion we have

$$e^{2i\hat{\delta}_{0}} \simeq e^{2i\delta_{0}} \exp\left\{i\int_{0}^{2\pi} \frac{d\sigma}{2\pi} \hat{X}^{i} \hat{X}^{j} \left[\mu_{\hat{y}}^{2} \left(\delta_{ij} - \frac{b_{i}b_{j}}{b^{2}}\right) + \mu_{0}^{2} \frac{b_{i}b_{j}}{b^{2}}\right]\right\}.$$
(4.89)

 $^{^{35}}$ As in Section 3.2, in the string-brane case we are taking the probe approximation where the string coupling is small, but with $g_s N$ fixed, so we ignore the contributions of worldsheets with handles.

By using the mode expansion (3.174) we have

$$\int_{0}^{2\pi} \frac{d\sigma}{2\pi} \hat{X}^{i} \hat{X}^{j} = \alpha' \sum_{n=1}^{\infty} \frac{1}{n} \left[2T_{0,n}^{ij} - T_{+,n}^{ij} - T_{-,n}^{ij} - \delta^{ij} \right] , \qquad (4.90)$$

where, by following [10], we introduced the combinations

$$T_{\pm,n}^{ij} = \frac{1}{2n} A_{\mp n}^{(i} \bar{A}_{\mp n}^{j)} , \quad 2T_{0,n}^{ij} = \delta^{ij} + \frac{1}{2n} \left(A_{-n}^{(i} A_{n}^{j)} + \bar{A}_{-n}^{(i} \bar{A}_{n}^{j)} \right) .$$
(4.91)

In order to evaluate the first string corrections to the leading eikonal, it is convenient to choose a basis parametrized by \hat{y}_i , where i = 1, ..., (d - p - 3) indicate the coordinates orthogonal to the scattering plane and i = 0 indicates the direction along the impact parameter. This diagonalizes the structure in (4.89) and we can consider (d - p - 2) decoupled families of generators $T_n^{\hat{y}_i}$ that satisfy the commutation relations of the *sl*(2, *R*) algebra

$$[T_{-,n}^{\hat{y}_i}, T_{+,m}^{\hat{y}_j}] = \delta^{ij} \,\delta_{nm} \,2T_{0,n}^{\hat{y}_i} \,, \quad [T_{0,n}^{\hat{y}_i}, T_{\pm,m}^{\hat{y}_j}] = \pm \delta^{ij} \,\delta_{nm} \,T_{\pm,n}^{\hat{y}_i} \,. \tag{4.92}$$

It is now possible to use the properties of the sl(2, R) algebra to derive the exponential suppression of the elastic scattering of an unexcited string mentioned in (4.86). Starting from

$$e^{2i\hat{\delta}_{0}} \simeq e^{2i\delta_{0}} \exp\left\{\sum_{n=1}^{\infty} \frac{i\alpha'}{n} \left[\mu_{0}^{2} \left(2T_{0,n}^{\hat{y}_{0}} - T_{+,n}^{\hat{y}_{0}} - T_{-,n}^{\hat{y}_{0}} - 1 \right) + \mu_{\hat{y}}^{2} \sum_{i=1}^{d-p-3} \left(2T_{0,n}^{\hat{y}_{i}} - T_{+,n}^{\hat{y}_{i}} - T_{-,n}^{\hat{y}_{i}} - 1 \right) \right] \right\}$$
(4.93)

we can use the identity

$$e^{x\left(2T_{0,n}^{\hat{y}_{i}}-T_{+,n}^{\hat{y}_{i}}-T_{-,n}^{\hat{y}_{i}}\right)} = e^{-\frac{x}{1-x}T_{+,n}^{\hat{y}_{i}}} e^{-2\ln(1-x)T_{0,n}^{\hat{y}_{i}}} e^{-\frac{x}{1-x}T_{-,n}^{\hat{y}_{i}}}$$
(4.94)

to calculate the elastic amplitude $\langle 0|S|0\rangle$. The terms proportional to $T_{\pm,n}^{\hat{y}_i}$ vanish when acting on the unexcited initial/final state and the same happens for the oscillator contribution to $T_{0,n}^{\hat{y}_i}$, see (4.91). Then (4.93) reads

$$\langle 0|e^{2i\hat{\delta}_{0}}|0\rangle \simeq e^{2i\delta_{0}} \prod_{n=1}^{\infty} \left\{ \frac{e^{-\frac{i\alpha'\mu_{0}^{2}}{n}}}{1-\frac{i\alpha'\mu_{0}^{2}}{n}} \left(\frac{e^{-\frac{i\alpha'\mu_{0}^{2}}{n}}}{1-\frac{i\alpha'\mu_{\hat{y}}^{2}}{n}} \right)^{d-p-3} \right\}$$
(4.95)

By using Weierstrass's formula for the Γ -function and then taking the absolute value square which is directly related to the probability of the transition, we obtain

$$\left| \langle 0 | e^{2i\delta_0} | 0 \rangle \right|^2 \simeq \left| e^{2i\delta_0} \Gamma \left(1 - i\alpha' \mu_0^2 \right) \Gamma^{d-p-3} \left(1 - i\alpha' \mu_{\hat{y}}^2 \right) \right|^2$$

$$\simeq e^{-2\mathrm{Im}2\delta_0} \left(\frac{\pi \alpha' \mu_0^2}{\sinh\left(\pi \alpha' \mu_0^2\right)} \right) \left(\frac{\pi \alpha' \mu_{\hat{y}}^2}{\sinh\left(\pi \alpha' \mu_{\hat{y}}^2\right)} \right)^{d-p-3}$$

$$(4.96)$$

where we used $\Gamma(1 + ix)\Gamma(1 - ix) = \frac{\pi x}{\sinh \pi x}$. As discussed in Section 3.2.1, the imaginary part of the leading eikonal is negligible when $b \gg l_s(E_s)$, so in this regime we can focus on the last two factors. In the leading eikonal approximation $b \gg R_p$, we have

$$|\langle 0|e^{2i\delta_0}|0\rangle|^2 \simeq (d-p-3)(2\pi\alpha'|\mu_{\hat{y}}^2|)^{d-p-2}e^{-2\pi\alpha'|\mu_0^2|}$$
(4.97)

where we used the last identity in the second equation (4.88). Then, by using (4.88), we obtain (4.86) in the regime where μ_0^2 (and $|\mu_{\hat{v}}^2|$) is large, i.e. $b_t \gg b$.

In summary, when the impact parameter is below the scale (4.86), internal bosonic excitations of the string along the directions \hat{y} and \hat{b} are excited according to parameters summarized in (4.88), while no excitations are created at this order along the remaining direction. As we will see in the next section, the same result can be recovered by studying the propagation of a string in a curved background.

4.2.3. A geometric description of the eikonal scattering

In the QFT setup the probe limit of the eikonal has a simple geometric interpretation as it is entirely captured by the classical motion in the background sourced by the heavy object, see Section 4.1.5 and Appendix B. It is natural to expect that a similar pattern holds also for the string eikonal operator and the aim of this section is to show that this is indeed the case. Again we focus on the case of the string-brane scattering that has been the main example analyzed, but the same idea applies more generally (see Section 4.2.5 below). It is natural to see stack of Dp-branes as the heavy object

and the scattering string state as the probe. Thus the first ingredient we need is the supergravity solution describing the gravitational backreaction induced by the Dp-branes. In the string frame it reads [266]

1

$$ds^{2} = [H(r)]^{-\frac{1}{2}} \eta_{\alpha\beta} dx^{\alpha} dx^{\beta} + [H(r)]^{\frac{1}{2}} \delta_{ij} dx^{i} dx^{j} ,$$

$$e^{2\phi(x)} = g_{s}^{2} [H(r)]^{\frac{3-p}{2}} , \quad C_{01\cdots p}(x) = \frac{1}{H(r)} - 1 ,$$
(4.98)

where the indices α , $\beta = 0, \dots, p$ are along the D*p*-brane worldvolume while the indices *i*, *j* = *p* + 1, ..., *d* - 1 span the transverse directions. The fields in the second line are the dilaton (ϕ) and the Ramond–Ramond (RR) p + 1-form potential that couples minimally to the D*p*-branes and the harmonic function H(r) is

$$H(r) = 1 + \left(\frac{R_p}{r}\right)^{d-p-3}, \qquad r^2 = x_i^2,$$
(4.99)

where the scale R_n is given by (D.6) in terms of the string coupling g_s and the number of Dp-branes N. The classical motion of a point-like object in the geometry above is discussed in Appendix B.2, but we now need to generalize this analysis to the full string case starting from a string action that includes the couplings to the non-trivial background fields. The presence of RR fields makes the exact description more complicated as one would need to resort to the Green-Schwarz [206,233,234,267] or the pure-spinor (see e.g. [268,269] and references therein) formalisms. However the analysis of the Reggeon vertex in Section 3.2.3 suggests that the fermionic degrees of freedom on the worldsheet do not play any role at the level of the leading eikonal as the string excitations induced by the tidal forces are fully captured by the bosonic field X^{μ} . Thus we can focus just on the universal part of the (gauge fixed) string action

$$S = -\frac{1}{4\pi\alpha'} \int d\tau d\sigma \left[\partial_{\tau} X^{\mu} \partial_{\tau} X^{\nu} - \partial_{\sigma} X^{\mu} \partial_{\sigma} X^{\nu}\right] g_{\mu\nu} , \qquad (4.100)$$

where the metric is the one given in the first line of (4.98). Even this simplified starting point contains much more information than needed for our purposes since we are interested in the classical dynamics of an energetic string probe. In this case we can follow the approach of [270] and approximate the background metric by taking just its behavior around the null geodesic describing at leading order the trajectory of the string center of mass. We sketch the main steps of this approach below.

First it is convenient to choose coordinates adapted to the motion of the center of mass which takes place in a plane of the transverse space. In polar coordinates (B.25) this plane is parametrized by r and ϕ and it is convenient to choose

$$du = \frac{\sqrt{H}}{F}dr, \qquad dv = -dt + b_J d\phi + Fdr, \qquad dz = d\phi + \frac{d\bar{\phi}}{du} du, \qquad (4.101)$$

where we introduced the shorthand notation $F = \sqrt{H(r) - \frac{b_I}{r^2}}$ and $\bar{\phi} \equiv \phi(u)$ is the value of the angle along the geodesic parametrized by *u*. So, from (4.101) and (B.27) we have

$$\frac{d\bar{\phi}(u)}{du} = \frac{d\bar{\phi}}{d\bar{r}}\frac{d\bar{r}}{du} = -\frac{b_J}{\bar{r}^2\sqrt{H(\bar{r})}},$$
(4.102)

where, following the convention above, \bar{r} is the value of r along the geodesic describing the motion of the string center of mass. This is at constant values of v, z and the (d - p - 3) coordinates orthogonal to the scattering plane that we indicate with y^i . In these coordinates the metric in (4.98) takes the following form

$$ds^{2} = 2du \, dv - \frac{dv^{2}}{\sqrt{H}} + \frac{2b \, dv \, dz}{\sqrt{H}} + \frac{(rF)^{2}}{\sqrt{H}} dz^{2} + \frac{dx_{a}^{2}}{\sqrt{H}} + r^{2}\sqrt{H} \sin^{2}(\phi - \bar{\phi}) d\Omega_{d-p-3}^{2} .$$
(4.103)

A similar change of coordinates can be implemented on the RR gauge field and then it is possible to take the Penrose limit on the full solution by introducing a small parameter ϵ and rescaling the coordinates $(x^a, y^i, z) \rightarrow \epsilon(x^a, y^i, z)$ together with $v \to \epsilon^2 v$. It was shown in [270] that in the limit $\epsilon \to 0$, the leading term is $\mathcal{O}(\epsilon^2)$ and yields a solution of the same supergravity field equations relevant for the original configuration (4.98). In this limit the geometry (4.103) reduces to

$$ds^{2} \simeq 2dudv + \frac{(\bar{r}F(\bar{r}))^{2}}{\sqrt{H(\bar{r})}}dz^{2} + \frac{dx_{a}^{2}}{\sqrt{H(\bar{r})}} + \bar{r}^{2}\sqrt{H(\bar{r})}\sin^{2}(\bar{\phi})dy^{i2}.$$
(4.104)

We can take this result in the canonical form by introducing the coordinates \hat{y}_0 , \hat{y}_i and \hat{x}_a

$$z = \frac{H^{\frac{1}{4}}}{\bar{r}F} \hat{y}_{0} , \quad y^{i} = \frac{\hat{y}_{i}}{H^{\frac{1}{4}}\bar{r}\sin\bar{\phi}} , \quad x^{a} = H^{\frac{1}{4}} \hat{x}_{a}$$

$$= \hat{v} + \frac{1}{2} \left[\hat{y}_{0}^{2} \partial_{u} \ln\left(\frac{\bar{r}F}{H^{\frac{1}{4}}}\right) + \hat{y}_{i}^{2} \partial_{u} \ln\left(\bar{r}\sin\bar{\phi}H^{\frac{1}{4}}\right) + \hat{x}_{a}^{2} \partial_{u} \ln\left(H^{-\frac{1}{4}}\right) \right]$$
(4.105)

obtaining

v

$$ds^{2} \simeq 2dud\hat{v} + d\hat{y}_{0}^{2} + d\hat{x}_{a}^{2} + d\hat{y}_{i}^{2} + \left(\mathcal{G}_{\hat{x}}\hat{x}_{a}^{2} + \mathcal{G}_{\hat{y}}\hat{y}_{i}^{2} + \mathcal{G}_{0}\hat{y}_{0}^{2}\right)du^{2}, \qquad (4.106)$$

P. Di Vecchia, C. Heissenberg, R. Russo et al.

with

$$\mathcal{G}_{\hat{x}} = \partial_{u}^{2} \ln H^{-\frac{1}{4}} + \left(\partial_{u} \ln H^{\frac{1}{4}}\right)^{2} , \quad \mathcal{G}_{0} = \partial_{u}^{2} \ln \left(\frac{\bar{r}F}{H^{\frac{1}{4}}}\right) + \left(\partial_{u} \ln \frac{\bar{r}F}{H^{\frac{1}{4}}}\right)^{2} , \quad (4.107)$$

$$\mathcal{G}_{\hat{y}} = \partial_u^2 \ln\left(\bar{r}\sin\bar{\phi}H^{\frac{1}{4}}\right) + \left(\partial_u \ln\bar{r}\sin\bar{\phi}H^{\frac{1}{4}}\right)^2 \,. \tag{4.108}$$

In order to capture the dynamics of energetic semiclassical string probes it is then sufficient to use the action

$$S = S_{\eta} - \frac{1}{4\pi\alpha'} \int d\tau \int_{0}^{2\pi} d\sigma \ \eta^{\alpha\beta} \partial_{\alpha} U \partial_{\beta} U \left(\mathcal{G}_{\hat{x}} \hat{X}_{a}^{2} + \mathcal{G}_{\hat{y}} \hat{Y}_{i}^{2} + \mathcal{G}_{0} \hat{Y}_{0}^{2} \right) , \qquad (4.109)$$

where S_{η} is the string action in flat space (i.e Eq. (4.100) with $g_{\mu\nu} = \eta_{\mu\nu}$). As standard, we are using uppercase symbols to indicate the string embedding fields corresponding to the various coordinates. The string analysis is greatly simplified by choosing a light cone gauge where the string oscillations along $U(\sigma, \tau)$ are gauged away so that we have

$$U(\sigma,\tau) = \alpha' E_s \tau , \qquad (4.110)$$

so we can keep using a lowercase symbol to indicate this coordinate. Then, at high energies, the interaction between the string probe and the background becomes localized at $\tau \sim 0$ since $\tau = u/(\alpha' E_s)$, so by changing variables from τ to u, we can write

$$S \simeq S_{\eta} - \frac{E_{s}}{2} \int_{-\infty}^{\infty} du \int_{0}^{2\pi} d\sigma \, \left(\mathcal{G}_{\hat{x}} \hat{X}_{a}^{2}(\sigma, 0) + \mathcal{G}_{\hat{y}} \hat{Y}_{i}^{2}(\sigma, 0) + \mathcal{G}_{0} \hat{Y}_{0}^{2}(\sigma, 0) \right).$$
(4.111)

As usual for semiclassical strings captured by a Penrose limit, the worldsheet action is described by free massive fields where the effective mass square parameters are

$$\frac{E_s}{2} \int_{-\infty}^{\infty} du \,\mathcal{G}_{\hat{x}} = E_s \int_{0}^{\infty} du \left[\partial_u^2 \ln H^{-\frac{1}{4}} + \left(\partial_u \ln H^{\frac{1}{4}} \right)^2 \right] \simeq 0 \,, \tag{4.112a}$$

$$\frac{E_s}{2} \int_{-\infty}^{\infty} du \ \mathcal{G}_{\hat{y}} = E_s \int_0^{\infty} du \left[\partial_u^2 \ln\left(\bar{r}\sin\bar{\phi}H^{\frac{1}{4}}\right) + \left(\partial_u\ln\left(\bar{r}\sin\bar{\phi}H^{\frac{1}{4}}\right)\right)^2 \right] \simeq \mu_{\hat{y}}^2 , \tag{4.112b}$$

$$\frac{E_s}{2} \int_{-\infty}^{\infty} du \ \mathcal{G}_0 = E_s \int_0^{\infty} du \left[\partial_u^2 \ln \left(\frac{\bar{r}F}{H^{\frac{1}{4}}} \right) + \left(\partial_u \ln \frac{\bar{r}F}{H^{\frac{1}{4}}} \right)^2 \right] \simeq \mu_0^2 , \qquad (4.112c)$$

where the final results are valid at 1PM (i.e. at order $(R_p/b)^{d-p-3}$) and the parameters μ_0^2 , $\mu_{\hat{y}}^2$ have been introduced in (4.88). In the equations above, we restricted the integration over *u* from infinity to the inversion point r_* (which corresponds to u = 0) and included the overall factor of two to account for the contribution of the second part of the trajectory from r_* back to infinity. The two integrals in (4.112a) vanish at 1PM order. The second one can be neglected since it scales as $(R_p/b)^{2(d-p-3)}$, while the first one can be written as a total derivative of a function that vanishes at the extremes, as done below

$$\int_{0}^{\infty} du \,\partial_{u}^{2} \ln H^{-\frac{1}{4}} = \int_{r_{*}}^{\infty} dr \,\partial_{r} \left(\frac{F}{\sqrt{H}} \partial_{r} \ln H^{-\frac{1}{4}} \right) = 0.$$
(4.113)

Here we used (4.101) to rewrite ∂_u as $(\partial_u r)\partial_r$ and the special values $F(r_*) = 0$, while $H(r_*)$ is finite, $F(\infty) = 1$ and $\partial_r H \to 0$ as $r \to \infty$. Evaluating the remaining two integrals requires a slightly more detailed discussion which is summarized at the end of Appendix B.2 and the result is summarized in (4.112b) and (4.112c). Thus, as promised, we rederived from a geometric analysis the parameters (4.88) determining the strength of the tidal excitations. It would be interesting to extend the comparison between the eikonal operator and geometric description of this section at subleading PM order to see whether the geometric interpretation of the string excitations holds beyond the leading eikonal.

4.2.4. String eikonal and classical causality

In the previous sections we discussed how the leading string eikonal exponentiates as an operator and what novelties this implies with respect to the field theory case. This exponentiation holds in the case of bosonic string theory, see Appendix D for more details, which provides an interesting setup to study the higher derivative corrections to the on-shell 3-graviton vertex discussed in 3.1.5 from an EFT point of view. The main point we wish to convey is the following: the string-brane scattering in bosonic string theory provides *classical* observables that are affected by α' -corrections that take the form of (3.76) and (3.77) with $\ell_2 \sim \ell_4 \sim \ell_s$. A naive field-theory limit yields causality violations, in the form of negative time delays, when the impact parameter is of the order of $\ell_{2,4,s}$ [154]. As discussed below, in the eikonal context, this can be seen by looking at the energy derivative of the leading eikonal. Equivalently, one can look at deflection angle and notice that the higher derivative corrections seem to make gravity repulsive when $b \sim \ell_{2,4,s}$. We will show how this pathological behavior is avoided in string theory. A first observation is that, in bosonic string theory, the 3-graviton amplitude takes the form of (3.74) with $\ell_4^2 = -2\ell_2^2 = \frac{\alpha'}{2}$. Then we saw that, at the level of the leading eikonal, string effects are captured by the shift $b \rightarrow b + \hat{X}$ (3.176). By applying this recipe to the long-range eikonal (3.131)

$$2\delta_0 \sim \frac{\sqrt{\pi}E_s}{2} \frac{\Gamma\left(\frac{d-p-4}{2}\right)}{\Gamma\left(\frac{d-p-3}{2}\right)} \frac{R_p^{d-p-3}}{b^{d-p-4}}$$
(4.114)

one obtains an eikonal operator that is directly related to the EFT eikonal (3.81): the latter can be obtained from the former simply by taking the expectation value in the massless sector

$$\langle \varepsilon_1, \tilde{\varepsilon}_1 | 2 \tilde{\delta}_0(b+X) | \varepsilon_2, \tilde{\varepsilon}_2 \rangle$$
, (4.115)

where $|\varepsilon_a \tilde{\varepsilon}_a \rangle = \varepsilon_{ai} \tilde{\varepsilon}_{aj} A^i_{-1} \bar{A}^j_{-1} |0\rangle$.

However, this result does not describe the elastic amplitude in the classical regime since one needs instead to take the expectation value of the exponentiated eikonal $e^{2i\hat{\delta}_0}$ as discussed in Section 4.2.2. The novelty here is that $e^{2i\hat{\delta}_0}$ is not a diagonal phase even in the massless sector. Thus in order to calculate the deflection angle of a massless state, one should first diagonalize $\hat{\delta}_0$ in the massless sector and then apply the usual relation (2.31) for the deflection angle or (2.33) for the time delay.

Let us then start by analyzing the EFT eikonal (3.81). The key observation of [154] is that regardless of the signs of the higher derivative corrections (ℓ_2^2 , $\ell_4^4 \ge 0$) there always exists an eigenstate of the EFT eikonal with a *negative* eigenvalue. In order to show this let us focus on the eigenvectors that lie in the space of the two states with polarizations

$$\varepsilon_i \tilde{\varepsilon}_j \to \epsilon^a_{ij} = \frac{1}{\sqrt{D-3}} \left(\delta_{ij} - \hat{b}_i \hat{b}_j \right) , \quad \varepsilon_i \tilde{\varepsilon}_j \to \epsilon^b_{ij} = \hat{b}_i \hat{b}_j$$

$$(4.116)$$

(let us recall that $\hat{b}_i = b_i/b$). This space is orthogonal to the other physical states and so it can be studied separately. From (3.81), one obtains

$$\begin{pmatrix} 2\hat{\delta}_{0}^{aa} & 2\hat{\delta}_{0}^{ab} \\ 2\hat{\delta}_{0}^{ba} & 2\hat{\delta}_{0}^{bb} \end{pmatrix} = 2\delta_{0} \begin{pmatrix} 1+2c_{2}+(D-1)c_{4} & -(D-1)\sqrt{D-3}c_{4} \\ -(D-1)\sqrt{D-3}c_{4} & 1-2(D-3)c_{2}+(D-1)(D-3)c_{4} \end{pmatrix},$$

$$c_{2} = (D-2)\frac{2\ell_{2}^{2}}{b^{2}}, \quad c_{4} = D(D-2)\frac{\ell_{4}^{4}}{b^{4}}.$$

$$(4.117)$$

Let us also emphasize that the matrix appearing in (4.117) is symmetric and real. One can check that the determinant of (4.117) always becomes negative when $|c_2|$, $|c_4| \gtrsim 1$, which means that the two eigenvalues

$$1 - (D-4)c_2 + \frac{1}{2}(D-2)(D-1)c_4 \pm \frac{D-2}{2}\sqrt{4c_2^2 - 4\frac{(D-4)(D-1)}{D-2}c_2c_4 + (D-1)^2c_4^2}$$
(4.118)

have generically opposite sign and one of them must be negative. States corresponding to eigenvectors with a negative eigenvalue have a non-standard behavior: both the deflection angle and the Shapiro time delay due to the gravitational scattering would be negative as they are related to derivatives of the eikonal. This would happen when $b \gtrsim \ell_2$, ℓ_4 which may seem to be within the range of validity of an EFT approach if $\ell_{2,4} \gg \ell_P$. However, the string theory analysis shows that, even when the higher derivative corrections in (3.73) have a classical origin, this pathological behavior is avoided.

In order to see this, we must recall that the string eikonal (D.35) has a more complicated structure than the EFT eikonal used for the argument above. In particular when $b < l_s(E_s) \equiv \sqrt{\alpha' \ln(\alpha' E_s^2)}$ the string eikonal becomes a constant plus corrections proportional to b^2 , see for instance (3.134). Then the corresponding eikonal operator, which is as before obtained with the shift $b \rightarrow b + \hat{X}$, differs from EFT eikonal at scales much larger than $b \sim \ell_s$: instead of growing at small distances, it becomes almost constant. The key ingredient for this is the Regge behavior which modifies the functional form in the impact parameter as showed in (3.127). It is important that the effective string scale $l_s(E_s)$ (3.125), where the softer behavior kicks in, is enhanced by a factor of $\ln(\alpha' E_s^2)$ with respect to the scale of the higher derivative terms in the effective action. This is the effect of the exchange of the whole leading Regge trajectory and the same mechanism would not work in a theory with a finite number of extra higher spin states beyond the graviton [154,271].

4.2.5. String-string scattering beyond tree level

In Section 3.2.6 we have discussed the main features of high-energy string–string scattering at tree level and anticipated some results that follow from the exponentiation in *b*-space of the tree-level result discussed in this section. Such an exponentiation indeed goes through in the case of string–string collisions as well: the only difference with respect to the case of string-brane collisions discussed in detail in Sections 4.2.1 and 4.2.2 is that now both incoming strings get excited through the exchange of the (gravi)reggeon: this is a consequence of factorization of Regge-pole residues, related, in turn, to *t*-channel unitarity. Without repeating here the discussion about tidal excitations given above in the case of string-brane collisions we simply give its analog for string–string scattering:

$$S = \exp(2i\delta) \tag{4.119}$$

with $\hat{\delta}$ now given in Eqs. (3.192), (3.185) (instead of (3.176)). For $b \gg l_s(s)$, $\hat{\delta}$ is Hermitian and thus S is unitary. This result was first obtained in [10] by proving the exponentiation of the tree-level string scattering amplitude. An elegant rederivation [272-275] follows from considering the quantization of one string in the Aichelburg-Sexl metric (see Section 2.3) produced by the other string. The corresponding classical action takes the form of (4.100) with $g_{\mu\nu}$ now given by (3.192). It is greatly simplified by choosing the gauge $u = \sqrt{\alpha' \tau}$ which allows, thanks to the $\delta(u)$ in (2.55), to carry out the integration over τ . This leaves just the σ integral of the profile function $f(\mathbf{x}_{\perp}(\sigma))$ of the transverse string coordinates as in (3.192). Actually, the explicit form of the eikonal operator, including α' corrections, allows one to reconstruct [273] the generalization of the AS metric produced by the profile (3.191).

At a qualitative level we can estimate the importance of the string corrections by noting that the normal ordering in $\hat{\delta}$ produces, at the level of *S*, corrections to the eikonal phase of order $\frac{\partial^2 2\delta_0}{(\partial b)^2} \ell_s^2$ which are, naively, of relative order $\frac{\ell_s^2}{b^2}$ with respect to $2\delta_0$ itself. The crucial point, however, is that some of these corrections are imaginary and thus cannot be neglected as soon as $\frac{\partial^2 2\delta_0}{(\partial b)^2} \ell_s^2$ becomes $\mathcal{O}(1)$. A simple calculation, using $2\delta_0$ in (2.25), shows that this happens at a critical value of *b* –that we denote by b_t – given by:

$$b_t^{D-2} \sim \frac{Gs}{\hbar} \ell_s^2 \gg \ell_s^{D-2} \quad \text{for} \quad \frac{Gs}{\hbar} \gg \ell_s^{D-4}$$

$$(4.120)$$

with $R^{D-3} \sim G\sqrt{s}$. The above result can also be guessed from the two following qualitative physical arguments. One of them goes as follows. The Newton potential acting on two points of a string at a transverse distance $2\Delta b$ from each other is given by

$$U(b \pm \Delta b) = \left(\frac{R}{b \pm \Delta b}\right)^{D-3} \Longrightarrow U(b + \Delta b) - U(b - \Delta b) = 2\frac{R^{D-3}}{b^{D-2}}\Delta b.$$
(4.121)

Taking the distance $2\Delta b$ of the two points on the string to be of order ℓ_s , and assuming that the two points move at the speed of light, we can compute the difference between the two forces acting on them. The tidal forces start to be relevant when such a difference is equal to the string tension times their distance ℓ_s :

$$2\frac{R^{D-3}}{b_t^{D-2}}\sqrt{s}\ell_s \sim T\ell_s \Longrightarrow b_t^{D-2} \sim 2\frac{\sqrt{s}R^{D-3}}{T} \sim 4\pi\alpha' Gs \sim \frac{4\pi Gs}{\hbar}\ell_s^2$$
(4.122)

where we have used that $R^{D-3} \sim G\sqrt{s}$ and $T = \frac{1}{2\pi\alpha'}$. Alternatively, (4.120) can be obtained by considering how the different bits of one string (say the one of energy E_1) suffer, as a result of the metric produced by the other string (of energy E_2), slightly different deflection angles. The spread $\Delta \Theta_1$ of these deflection angles is roughly:

$$\Delta\Theta_1 \sim \frac{GE_2}{b^{D-3}} \frac{\ell_s}{b} = GE_2 \ell_s b^{2-D} \,. \tag{4.123}$$

Such a spread corresponds to an invariant excitation mass $\Delta M_1 \sim E_1 \Delta \Theta_1 \sim GE_1 E_2 \ell_s b^{2-D}$. Requiring $\Delta M_1 \sim M_s$ (the mass of the first excited level) leads to the same estimate (4.120) for b_t after using $s \sim E_1 E_2$ as in (2.60).

Quantitatively, the phenomenon is described by the same gravi-reggeon vertex operator defined in (3.166), acting now on both external strings. After exponentiation of the tree-level amplitude in *b*-space, the resulting *S*-matrix is explicitly unitary within a Hilbert space containing just two-(massive or massless) strings. We just stress here that, although single gravi-reggeon exchange can only excite a small number of string oscillators, the full S-matrix (4.119) will populate a large number of excited states. Some details of this S-matrix have already been discussed in Section 4.2.2. We just mention that the elastic amplitude gets suppressed as a result of the opening of the inelastic channels. One finds:

$$|A_{\rm el}| \sim e^{-\frac{G_s}{\hbar} \frac{l_s^2}{b^{D-2}}} \sim e^{-(\frac{b_t}{b})^{D-2}} \ll 1 , \ (b \ll b_t).$$
(4.124)

The exponential suppression of the elastic amplitude for $b < b_t$ implies, by unitarity, that inelastic final states, consisting of two tidally excited strings, are copiously produced. The differential cross section $d\sigma/(dM_1 dM_2)$ for producing two massive strings of mass M₁ and M₂ (summed over the degeneracy of each mass level) was studied in [10]. Interestingly enough, the cross section starts growing like the (exponential) degeneracy of each mass level as if all states of a given mass were democratically produced. However, such a growth stops when $M_i \sim M_s (b_t/b)^{D-2}$. The distribution has a maximum around that value of $M_{1,2}$ and then falls off as a Gaussian.

A much more detailed analysis of the tidal excitation spectrum was made in [232] for the case of string-brane scattering. This shows how the apparently incoherent sum of different contributions actually comes from a unitary evolution of the system. Indeed, each gravi-Reggeon exchange induces elementary transitions obeying very simple selection rules while the final spectrum takes its apparently statistical form as a consequence of the eikonal exponentiation.

In the above discussion we have not considered graviton (and other massless particle) emission, a topic discussed in the following sections in the point particle limit. As one lowers b, radiation phenomena become important and things get increasingly complicated. The parameter controlling the importance of radiative corrections, relative to the leading order deflection, is formally $(R^2/b^2)^{\hat{D}-3}$ but, once again, they become relevant as soon as their contribution to the imaginary part of the eikonal phase becomes O(1). A naive estimate of the critical impact parameter b_r at which this happens is given by

$$\frac{Gs}{\hbar}b_r^{4-D}\left(\frac{R}{b_r}\right)^{2(D-3)} \sim 1, \quad \text{i.e.} \quad b_r \gg R \text{ for } \frac{Gs}{\hbar}b_r^{4-D} \gg 1.$$
(4.125)

This is a fair estimate of b_r for quantities that do not suffer from infrared divergences which is generically the case for D > 4. For D = 4, instead, many quantities are infrared sensitive: for instance, the elastic amplitude goes to zero together with the IR regulator. The estimate (4.125), however, should still apply to infrared safe quantities, such as the total radiated energy. Comparing now b_t of (4.120) with b_r of (4.125) we see that which kind of corrections comes first as one lowers b, depends on D and on the ratio R/ℓ_s : as shown in Fig. 12, radiative corrections take the upper (lower) hand at small (large) D. In any case the above discussion would imply that, in going towards $b \sim R$, we cannot keep just the string-size corrections and neglect the classical radiative corrections. Unfortunately, to this date, no calculation taking both corrections simultaneously into account is available.

Fortunately, a big simplification occurs if we lower *b* as much as we want while keeping $R < \ell_s$ (thus entering region II in Fig. 12). In this case, string-size effects should replace the expansion parameter R^2/b^2 by $R^2/(b^2 + \ell_s^2) < R^2/\ell_s^2 < 1$ justifying a perturbative (yet stringy) approach.³⁶ This amounts to saying that string-size corrections shield the bad short-distance regime of local quantum gravity.

Let us list now the most significative results in this string-gravity regime:

• The classical deflection angle reaches a maximal value Θ_M given by

$$\Theta_M \sim (R/l_s(s))^{D-3} \sim \frac{G\sqrt{s}}{l_s^{D-3}(s)},$$
(4.126)

which is reached for $b \sim l_s(s)$, i.e. when the two strings graze each other. This result simply comes from the smooth behavior of the stringy eikonal at small *b*, Eq. (3.193). Indeed, taking the derivative of Re 2 δ with respect to *b*, one obtains (see (3.195)):

$$\Theta = \frac{8G\sqrt{s}b}{(D-2)\pi^{\frac{D-4}{2}}l_s^{D-2}(s)} + \cdots, \qquad b \ll l_s(s),$$
(4.127)

which grows monotonically with *b*. It reaches the above-mentioned maximal value at $b \sim l_s(s)$ before starting to decrease according to the second equation in (3.195).

• For $\Theta > \Theta_M$ there is no real saddle point in *b* and, as a result, the elastic amplitude is exponentially suppressed. In order to find the actual damping of the elastic amplitude one has to find the dominant complex saddle point. This was done in Ref. [13] of which we give below a more streamlined version. Let us rewrite (3.186) in the form:

$$2\delta_0(s,b) = \frac{Gs(b^2)^{\frac{4-D}{2}}}{\pi^{\frac{D-4}{2}}} \left[\Gamma\left(\frac{D-4}{2}\right) - \int_{\frac{b^2}{Y_c}}^{\infty} dt \, e^{-t} t^{\frac{D-4}{2}-1} \right] \quad ; \quad Y_c = l_s^2(s) - i\pi\alpha' \,. \tag{4.128}$$

The first term is the field theory leading eikonal for the scattering of massless particles. It is easy to show that, after adding to it the tidal corrections, such term cannot provide a relevant complex saddle point. This can come, instead, from the second term, denoted by $2\delta_0^{(s)}$, which contains string-size (i.e. finite-beam, see (3.191)) effects. Indeed, at large complex values of *b* (i.e. at $|b|^2 > l_s^2(s)$), $2\delta_0^{(s)}$ can be exponentially enhanced. Its large |b| behavior is easily obtained by integration by parts and reads

$$2\delta_0^{(s)}(s,b) = -\frac{Gs(b^2)^{\frac{4-D}{2}}}{\pi^{\frac{D-4}{2}}}e^{-\frac{b^2}{Y_c}} \left(\frac{b^2}{Y_c}\right)^{-3+\frac{D}{2}} = -\frac{Gs}{\pi^{\frac{D-4}{2}}b^2Y_c^{\frac{D}{2}-3}}e^{-\frac{b^2}{Y_c}}$$
(4.129)

We can now go back to the amplitude in momentum/angle space by a standard Fourier transform. Keeping $\Theta \ll 1$ and fixed we get:

$$A(s,\Theta) = \int d^{D-2}b \ e^{-ib\frac{\sqrt{s\Theta}}{2}} e^{-iGs\pi^{-\frac{D-4}{2}}b^{-2}Y_c^{3-\frac{D}{2}}e^{-\frac{b^2}{Y_c}}}.$$
(4.130)

Looking for a (complex) saddle point gives the equation:

$$\frac{\sqrt{s}}{2}\Theta = \frac{2Gs}{\pi^{\frac{D-4}{2}}}\frac{Y_c^{\frac{1}{2}}e^{-\frac{b\zeta}{Y_c}}}{Y_c^{\frac{D-3}{2}}b_s} = \frac{\sqrt{s}}{2}\Theta_M e^{-\frac{b_s^2}{Y_c}}\left(\frac{4Y_c^{\frac{1}{2}}}{b_s\pi^{\frac{D-4}{2}}}\right), \qquad \Theta_M \equiv \frac{G\sqrt{s}}{l_s^{D-3}(s)}.$$
(4.131)

 $^{^{36}}$ This is still a physically motivated guess leading, as we shall see, to very sensible consequences. It would be interesting to have further explicit checks of its validity.

It is now clear that for $\Theta \gg \Theta_M$ the saddle point has to be predominantly imaginary and with a negative imaginary part in order to produce damping. Writing $b_s = -i\beta_s$ the previous equation becomes:

$$-i\frac{\Theta}{\Theta_{M}} = \frac{4}{\pi^{\frac{D-4}{2}}} \frac{Y_{c}^{\frac{1}{2}}}{\beta_{s}} e^{\frac{\beta_{s}^{2}}{Y_{c}}} \Longrightarrow \log\left(-i\frac{\Theta}{\Theta_{M}}\right) = \frac{\beta_{s}^{2}}{Y_{c}} + \log\left(\frac{4Y_{c}^{\frac{1}{2}}}{\beta_{s}\pi^{\frac{D-4}{2}}}\right).$$
(4.132)

The last term is subleading for large values of $\frac{\beta_s}{\sqrt{V_e}}$ and thus we get:

$$\beta_s = Y_c^{\frac{1}{2}} \sqrt{\log \frac{\Theta}{\Theta_M} - \frac{i\pi}{2}} \Longrightarrow b_s = -iY_c^{\frac{1}{2}} \sqrt{\log \frac{\Theta}{\Theta_M} - \frac{i\pi}{2}}.$$
(4.133)

Inserting this saddle point in (4.130) we find that the first exponential dominates over the second one by a $\log \frac{\Theta}{\Theta_M} \gg 1$ and arrive at the final expression [13]:

$$|\mathcal{A}(s,\Theta)| \equiv |\mathcal{A}_{ACV}| \sim e^{-\frac{\sqrt{s}}{2}\Theta \gamma_c^{\frac{1}{2}}\sqrt{\log\frac{\Theta}{\Theta_M} - \frac{i\pi}{2}}} \sim e^{-\frac{\sqrt{s}}{2\hbar}\Theta l_s(s)\sqrt{\log\frac{\Theta}{\Theta_M}}}.$$
(4.134)

Taking into account that $\frac{\Theta_M}{\Theta} \sim g_s^2 \ll 1$, we can also approximate (4.134) by

$$|\mathcal{A}_{ACV}| \sim e^{-\frac{\sqrt{5}}{2\hbar}\Theta\sqrt{2\alpha'}\sqrt{\log(\frac{\alpha's}{4})\log g_s^{-2}}}.$$
(4.135)

The parametric form of the suppression is consistent with the behavior found, using a completely different method, by Gross–Mende–Ooguri (GMO) [9,11,14] in the overlapping kinematical regime in which Θ is small and kept fixed at small g_s . We recall that the Gross–Mende approach [9,11] is based on the behavior at high-energy, fixed-angle, of the elastic (closed) string–string scattering amplitude at each loop level, generalizing the original tree-level (open string) result in [276]. This regime exhibits an exponential damping of the amplitude:

$$\mathcal{A}(s,t)_{\text{tree}} \sim g_s^2 e^{-\frac{\alpha'}{2}sf(\Theta)}, \qquad f(\Theta) = -\sin^2\frac{\Theta}{2}\log\sin^2\frac{\Theta}{2} - \cos^2\frac{\Theta}{2}\log\cos^2\frac{\Theta}{2} \ge 0.$$
(4.136)

This tree-level behavior is in tension, at least at small Θ , with expectations based on classical gravitational deflection which would suggest an unsuppressed amplitude at high energy. Thus, at fixed angle, one is actually in the opposite situation with respect to the fixed-*t* large-*s* behavior where, as already mentioned, the tree-level result is too large to be compatible with unitarity bounds. We have seen how the eikonal resummation of loops in that kinematical regime is capable of restoring unitarity bounds. Similarly, loops also come to the rescue at fixed angle. As shown in [9,11], loop amplitudes, in spite of being of higher order in the string coupling, are enhanced, relative to the tree:

$$\mathcal{A}(s,t)_{h-\text{loops}} \sim g_s^{2(1+h)} e^{-\frac{\alpha' s f(\Theta)}{2(1+h)}},$$
(4.137)

where the milder exponential overcompensates the extra powers of g_s at sufficiently large *s*. That means, of course, that the perturbative series diverges at high energy and that a non perturbative resummation is needed. In [14] it was shown that, in a finite high energy window, such a divergent series can be Borel resummed. It is

precisely in such an energy window that we can compare the result of [14] to the one based on the leading eikonal in the stringy regime at small deflection angle. One finds quite an amazing agreement between the two results:

$$|\mathcal{A}_{ACV}| \sim e^{-\sqrt{\frac{\alpha' s}{2}}\Theta\sqrt{\log(\frac{\alpha' s}{4})\log(g_s^{-2})}}, \qquad |\mathcal{A}_{GMO}| \sim e^{-\sqrt{\frac{\alpha' s}{2}}\Theta\sqrt{\log(\frac{4}{\Theta^2})\log(g_s^{-2})}}, \tag{4.138}$$

where the second equation follows from Eq. (3.31) of [14] (which uses units in which $\alpha' = \frac{1}{2}$) for small Θ :

$$|\mathcal{A}_{GMO}| \sim e^{-\sqrt{2\alpha' sf(\Theta) \log(g_s^{-2})}}.$$
(4.139)

The difference (a $\log(\frac{\alpha's}{4})$ replacing a $\log(\frac{4}{\Theta^2})$) is already apparent in the transition between the fixed- $t((\frac{\alpha's}{4})^{\frac{\alpha't}{2}})$ with $t = -s \sin^2 \frac{\theta}{2}$ and small fixed- Θ regime $(e^{-\frac{\alpha's}{8}\Theta^2 \log \frac{4}{\Theta^2}})$ at tree level. Let us stress that the above damping in $|\mathcal{A}_{ACV}|$

is only reliable in the stringy regime and has to be modified at higher energies when gravitational radiation further damps the elastic channel. Such effects are also (presumably) neglected in the GMO approach. Furthermore, the Borel resummation leading to $|A_{GMO}|$ could only be justified in a finite energy range. In Eq. (3.31) of Ref. [14] is claimed to be valid in the window $\log g_s^{-2} < s/M_s^2 < g_s^{-\frac{4}{3}}$. The impact parameters that can be actually probed at high energy at fixed Θ never go below the string-length scale.

• The impact parameters that can be actually probed at high energy at fixed Θ never go below the string-length scale. Indeed, for $\Theta < \Theta_M$ one finds two real saddle points in *b*, one above and one below ℓ_s , but the former is found to be the dominant one [13]. On the other hand, as we have just seen, for $\Theta > \Theta_M$ the imaginary saddle point (4.133) too has an absolute value larger than ℓ_s . All this suggests an effective generalized uncertainty principle (GUP) holding true in string theory and reading [13,277–279]:

$$\Delta X \ge \frac{\hbar}{\Delta P} + \alpha' \Delta P \ge 2\sqrt{\alpha' \hbar} = 2\ell_s \,. \tag{4.140}$$

• Let us now turn to the consequences of exponentiating the full tree-level string–string scattering amplitude including its imaginary part given in (3.194) at $b \le l_s(s)$.³⁷ This problem was already addressed in [10], was further analyzed in [280] (see also [281]), and was also used more recently in [282].

The basic point is how to interpret the imaginary part of the exponentiated scattering amplitude at a fixed loop order in terms of its *s*-channel cuts through the exchanged Reggeized gravitons. Fortunately, old work by Abramovsky, Gribov and Kancheli (AGK) [283] (see also [284]) provides very simple "cutting rules" for the relative weights when cutting *m* out of the total number *n* of exchanged gravitons at (n-1) loop order.³⁸ These rules (when directly applied in impact parameter space) tell us that the full imaginary part of the *n*-graviton exchange graph is the result of a contribution

$$\sigma_m^n = (-1)^{n-m} \frac{(4 \text{Im} \delta)^n}{m!(n-m)!}, \qquad n = 1, 2, \dots, \qquad m = 1, 2, \dots, n-1, n$$
(4.141)

due to cutting *m* out of *n* gravi-Reggeons, and a contribution

$$\sigma_0^n = (-1)^n \frac{(4 \text{Im}\delta)^n}{n!} - 2\text{ReS}^{(n)}, \qquad n = 1, 2, \dots,$$
(4.142)

when no gravi-Reggeon is cut, where $S^{(n)}$ is the full *n*-gravi-Reggeon exchange contribution to the S-matrix. As the symbol σ_m indicates, these are also to be interpreted as cross sections into inelastic channels with *m* cut gravi-Reggeons (i.e. *m* closed strings). It is trivial to check that, for any given *n*, the sum of all contributions from m = 0 to m = n gives back twice the full imaginary part of $T^{(n)} \equiv i(1 - S^{(n)})$, as it should.

In analogy with what was done for the description of tidal excitations, we shall again promote the eikonal phase to an eikonal operator [280] acting on both the tidally-excited pairs of strings and, at a more "coarse-grained" level, on a Hilbert space labeled by the number of cut gravi-Reggeons. A new unitary *S*-matrix reproducing the AGK cutting rules takes the form:

$$S = \exp(il) , \qquad (4.143)$$

where the Hermitian operator \hat{I} is given by:

$$\hat{I} = (\hat{\delta} + \hat{\delta}^{\dagger}) + \sqrt{-2i(\hat{\delta} - \hat{\delta}^{\dagger})(C + C^{\dagger})} = \hat{I}^{\dagger} , \qquad (4.144)$$

and the operators $\hat{\delta}$, $\hat{\delta}^{\dagger}$, *C* and *C*^{\dagger} satisfy the commutation relations

$$[C, C^{\dagger}] = 1, \ [\hat{\delta}, \hat{\delta}^{\dagger}] = [C, \hat{\delta}] = [C, \hat{\delta}^{\dagger}] = 0.$$
(4.145)

Using well-known harmonic-oscillator formulae, Eqs. (4.143), (4.144) lead to the more convenient form:

$$S = e^{2i\hat{\delta}} e^{i\sqrt{-2i(\hat{\delta}-\hat{\delta}^{\dagger})} C^{\dagger}} e^{i\sqrt{-2i(\hat{\delta}-\hat{\delta}^{\dagger})} C} .$$

$$(4.146)$$

Eqs. (4.144), (4.146) imply that, as long as δ is real (i.e. for $b \gg l_s(s)$), $\hat{\delta}$ is essentially Hermitian, the oscillators C, C^{\dagger} are turned off, and one recovers the unitary S-matrix (4.119) with $\hat{\delta}$ given by the usual recipe (3.192). On the other hand, as one goes to values of b of order $l_s(s)$ or lower, δ in (3.131) picks up an imaginary part and consequently (4.119) is no longer unitary, signaling the opening up of new channels. These correspond to unitarity cuts going through the gravi-reggeons themselves. In order to recover unitarity we should further extend the Hilbert space by including, in the final state, whatever goes on shell after cutting the gravi-reggeons. A detailed account of this "whatever" being unavailable, we limit ourselves to counting the number of such gravi-reggeons through the number-counting operator $N_C \equiv C^{\dagger}C$.

Indeed, from (4.146) one can calculate the average number of heavy strings produced. For Im 2δ of Eq. (3.194) of $\mathcal{O}(1)$ or smaller, i.e. for $\sqrt{s} < M_s g_s^{-1}$, a single closed string carrying the whole center of mass energy is formed. Instead, in the energy interval $M_s g_s^{-1} < \sqrt{s} < M_s g_s^{-2}$ the average number of produced strings grows like Im 2δ , i.e. like *s*. Consequently, by energy conservation, the average energy of each produced string decreases like $1/\sqrt{s}$. More precisely:

$$\langle E \rangle \sim \frac{M_s^2 g_s^{-2}}{\sqrt{s}} \,.$$
 (4.147)

As one approaches the supposed threshold of black hole production $s \sim M_s g_s^{-2}$ the most probable final state will consist of g_s^{-2} strings of energy $M_s = \frac{h}{\ell_s}$ which appears to be consistent with a smooth transition to a Hawking-like behavior in which the Hawking temperature matches the Hagedorn temperature of string theory and the string entropy matches the Bekenstein–Hawking entropy of a black hole of radius ℓ_s . Unfortunately, the approximations made cease to be valid when the energy going across each cut Reggeon goes below the string scale.

³⁷ Interestingly, as $b \rightarrow l_s(s)$, the imaginary part due to tidal excitations "saturates" and becomes of the same order as the one due to formation of *s*-channel resonances.

³⁸ Given that the topology of the diagram is that of a sphere with (n - 1)-handles, any value of $0 \le m \le n$ is allowed.

• Comparatively less work has been devoted to the study of heavy string scattering. An exception can be found in [285] where one considers the high-energy collision of a light/massless string on a heavy one (with the energy of the light string smaller than the mass of the heavy one). The idea is to take the heavy string mass to lie as close as possible to the so-called correspondence point, $M = M_s g_s^{-2}$, where fundamental strings and black holes share many properties, including their entropy $\sim g_s^{-2} \gg 1$ (see [286,287] and references therein).

One then checks whether the probe string can be sensitive to properties of the heavy one other than its mass or spin. One finds that, at least below the corresponding point (where the heavy string's Schwarzschild radius is smaller than ℓ_s), the light strings is also sensitive to the quadrupole moment of the heavy string, which therefore acts as some kind of "quantum hair". If that feature would persist all the way till the correspondence point, and possibly beyond, this would imply that some quantum hair can be detected for stringy black holes. Unfortunately, this conclusion cannot be reached on firm grounds since the approximations used become unjustified precisely as one approaches the correspondence point.

5. Unitarity and radiation-reaction

Radiation reaction effects are contributions are due to the effective force acting on the system as a result of the emission of gravitational waves (and of additional massless modes, in the supersymmetric case). Intuitively, as the two colliding objects deflect due to the mutual "potential" attraction, they emit energy and angular momentum due to Bremsstrahlung. This in turn further bends their trajectories, giving rise to dissipative terms in the eikonal. This reflects in additional contributions to the deflection angle, which appear to lowest order at two loops (3PM). This chapter and the next are devoted to the calculation of these interesting new effects. While in the next chapter we will derive them from a complete two-loop calculation of the elastic scattering amplitude, in this chapter we follow very closely the approach of Refs. [47,106] and obtain them by combining the contribution to the unitarity relation of the three-particle cut, involving two massive particles and a graviton, with the properties of real analyticity and crossing symmetry of the elastic scattering amplitude. We apply this method to both GR and massive $\mathcal{N} = 8$ supergravity. In particular, to obtain the radiation-reaction we do not need the complete three-particle cut, but only the part that is divergent as $\frac{1}{\epsilon}$ that is obtained from the five-point amplitude involving four massive particles and a graviton keeping only the leading term in the Weinberg limit of small graviton momenta $k^{\mu} \rightarrow 0$, which behaves as $\frac{1}{\omega}$ with $\omega = k^{0}$.

The chapter is organized as follows. In the first section, starting from the unitarity relation, we discuss the two-particle and three-particle cuts in momentum and impact-parameter space. In the second section we explicitly compute Im $2\delta_2$ for both GR and massive $\mathcal{N} = 8$ supergravity. Finally, in the third section, we discuss the constraints of real analyticity that relate this imaginary part to the radiation-reaction effects.

5.1. Unitarity in momentum space and in impact-parameter space

The unitarity of
$$S = 1 + iT$$
, i.e. $S^{\dagger}S = 1$ translates to

$$-iT + iT^{\dagger} = T^{\dagger}T, \qquad (5.1)$$

whose matrix elements between states $|a\rangle$, $|b\rangle$ yield

$$-i(2\pi)^{D}\delta(P_{a}-P_{b})\left(\mathcal{M}_{a\to b}-\overline{\mathcal{M}}_{b\to a}\right)=\langle b|T^{\dagger}T|a\rangle.$$
(5.2)

Inserting a complete set of free states on the right-hand side and restricting to $P_a = P_b$, then gives

$$-i\left(\mathcal{M}_{a\to b} - \overline{\mathcal{M}}_{b\to a}\right) = \sum_{I} (2\pi)^{D} \delta(P_{a} + P_{I}) \mathcal{M}_{a\to I} \overline{\mathcal{M}}_{b\to I}, \qquad (5.3)$$

where the sum includes Lorentz-invariant phase-space integrals over all the intermediate states $j \in I$ and, in general, sums over their spins/helicities. Since we will focus on the case in which $a \rightarrow b$ is an elastic $2 \rightarrow 2$ process for (real) massive scalars, for which the amplitude is a function A(s, t) of the two Mandelstam invariants (1.22), we have

$$2 \operatorname{Im} \mathcal{M}_{a \to b} = \sum_{I} (2\pi)^{D} \delta(P_{a} + P_{I}) \mathcal{M}_{a \to I} \overline{\mathcal{M}}_{b \to I} .$$
(5.4)

Let us write,

$$\operatorname{Im} \mathcal{A}(s,t) = \operatorname{Im}_{2pc} \mathcal{A}(s,t) + \operatorname{Im}_{3pc} \mathcal{A}(s,t) + \cdots$$
(5.5)

and discuss each cut separately (see below). Taking the Fourier transform of (5.5) according to (1.32), and noting that FT[Im A](s, b) = Im FT[A](s, b) because $A(s, -q^2)$ is a symmetric function of q, we get

$$Im \, FT[\mathcal{A}](s, b) = Im_{2pc} \, FT[\mathcal{A}](s, b) + Im_{3pc} \, FT[\mathcal{A}](s, b) + \cdots .$$
(5.6)
5.1.1. Elastic unitarity

The elastic contributions come from the two-particle cut

$$2 \operatorname{Im}_{2pc} \mathcal{A} = \int d(\operatorname{LIPS})_2 \mathcal{A}_{\mathrm{L}} \mathcal{A}_{\mathrm{R}}^*$$
(5.7)

(where L. and R. stand for "left" and "right") or in the pictorial form

$$2 \operatorname{Im}_{2pc} \begin{array}{c} p_{1} \xleftarrow{} \\ (q \uparrow) \\ p_{2} \xleftarrow{} \\ p_{3} \end{array} \begin{array}{c} p_{4} \\ = \int d(\operatorname{LIPS})_{2} \\ p_{2} \xleftarrow{} \\ p_{2} \xleftarrow{} \\ p_{2} \xleftarrow{} \\ p_{3} \end{array} \begin{array}{c} p_{1} \xleftarrow{} \\ (q \uparrow) \\ (q \uparrow) \\ p_{2} \xleftarrow{} \\ p_{2} \xleftarrow{} \\ p_{3} \end{array} \begin{array}{c} p_{4} \\ (q \uparrow) \\ (q \downarrow) \\ (q \uparrow) \\ (q \downarrow) \\ (q \uparrow) \\ (q \downarrow) \\$$

with $d(LIPS)_2$ the Lorentz-invariant phase space measure for the intermediate two-particle states. For simplicity, we do not consider inelastic 2 \rightarrow 2 processes for the time being.

Let us now check that the impact-parameter Fourier transform diagonalizes this two-particle phase space convolution. We can start from (5.8) and go to impact-parameter space via the Fourier transform for the $2 \rightarrow 2$ process (1.32). When applying this Fourier transform to both sides of (5.8), we obtain

$$2 \operatorname{Im}_{2pc} \operatorname{FT}[\mathcal{A}] = \int \frac{d^{D}q_{1}}{(2\pi)^{D}} 2\pi \delta(-2p_{1} \cdot q_{1} + q_{1}^{2}) 2\pi \delta(2p_{2} \cdot q_{1} + q_{1}^{2}) \xrightarrow{(q_{1}+q_{1}^{2})}{p_{2}} \xrightarrow{(q_{1}+q_{1}^{2})}{p_{3}}}$$

$$(5.9)$$

Then we change integration variables according to $q = q_1 + q_1'$ and use

$$\delta(-2p_1 \cdot q_1 + q_1^2)\delta(-2p_1 \cdot q + q^2) = \delta(-2p_1 \cdot q_1 + q_1^2)\delta(-2(p_1 - q_1) \cdot q_1' + q_1'^2)$$
(5.10)

and similarly

$$\delta(2p_2 \cdot q_1 + q_1^2)\delta(2p_2 \cdot q + q^2) = \delta(2p_2 \cdot q_1 + q_1^2)\delta(2(p_2 + q_1) \cdot q_1' + q_1'^2)$$
(5.11)

which hold thanks to the properties of the delta function. Then we get,

As shown in Appendix A.3, the Fourier transform FT[A] differs from the linearized one \tilde{A} employed in the definition of the eikonal exponentiation (4.1) by *G*-independent corrections suppressed by $1/b^2$ (more precisely, by powers of the

Physics Reports 1083 (2024) 1-169

dimensionless quantity $1/(pb)^2$), and thus we arrive at

$$2 \operatorname{Im}_{2pc} \tilde{\mathcal{A}} = \left| \tilde{\mathcal{A}}(s, b) \right|^2 \left(1 + \mathcal{O}\left(\frac{1}{b^2}\right) \right).$$
(5.13)

Using the eikonal exponentiation (4.1) on the right-hand side we find

$$2 \operatorname{Im}_{2pc} \tilde{\mathcal{A}} = \left| [1 + 2i\Delta(s, b)] e^{2i\delta(s, b)} - 1 \right|^2 \left(1 + \mathcal{O}\left(\frac{1}{b^2}\right) \right) \,. \tag{5.14}$$

We can expand this in *G* to obtain various constraints. To leading order in *G*, of course we get that the tree-level amplitude is real

$$2 \operatorname{Im}_{2pc} \tilde{\mathcal{A}}_0 = 0 \tag{5.15}$$

and hence the leading eikonal is real,

$$\operatorname{Im} 2\delta_0 = 0. \tag{5.16}$$

To first subleading order in *G*, i.e. $\mathcal{O}(G^2)$, we get

$$2 \operatorname{Im}_{2pc} \tilde{\mathcal{A}}_1 = (2\delta_0)^2 \left(1 + \mathcal{O}\left(\frac{1}{b^2}\right) \right) \,. \tag{5.17}$$

This is the first interesting constraint, since it reveals that the imaginary part of the one-loop amplitude up to $O(1/(b^2)^{1-2\epsilon})$ is exhausted by the eikonal exponentiation. In this way, this constraint ensures that $2\delta_1$ is real

$$\operatorname{Im} 2\delta_1 = 0, \qquad (5.18)$$

while in general $2\Delta_1$, being of the same order as the corrections in the right-hand side of (5.17) can itself develop an imaginary part. Additionally, Im $2\Delta_1$ receives contributions from inelastic $2 \rightarrow 2$ processes, e.g. from intermediate two-gravitino states in supergravity.

To $\mathcal{O}(G^3)$, we find instead

$$\operatorname{Im}_{2pc} \tilde{\mathcal{A}}_{2} = (2\delta_{0} \, 2\delta_{1} + 2\delta_{0} \operatorname{Re} 2\Delta_{1}) \left(1 + \mathcal{O}\left(\frac{1}{b^{2}}\right) \right) \,.$$
(5.19)

Therefore, the imaginary part of the two-loop amplitude due to the two-particle cut up to $O(1/(b^2)^{3/2-2\epsilon})$ is exhausted by the exponentiation. On the other hand, the imaginary part of $2\delta_2$ is not captured by the two-particle cut. By power counting in *G*, it must be due to the three-particle cut,

$$\operatorname{Im}_{2\delta_2} = \operatorname{Im}_{3nc} \tilde{\mathcal{A}}_2, \qquad (5.20)$$

which we shall illustrate in the next subsection.

Before proceeding, for illustrative purposes, let us explicitly calculate the *b*-space contribution of the two-particle cuts to order G^2 . We start from Eq. (5.12) and consider the situation in which the amplitudes appearing on its right-hand side are the tree-level ones, i.e. such that

$$\mathcal{A}_0(s, -q^2) = \frac{a_0}{q^2}, \qquad \tilde{\mathcal{A}}_0(s, b) = 2\delta_0(s, b) = \frac{1}{4Ep} \frac{a_0}{4\pi} \frac{\Gamma(-\epsilon)}{(\pi b^2)^{-\epsilon}}.$$
(5.21)

We retain here a generic q-independent prefactor a_0 , for the sake of generality; for instance

$$a_0 = 32\pi G m_1^2 m_2^2 \left(\sigma^2 - \frac{1}{2-2\epsilon}\right)$$
(5.22)

for collisions of minimally coupled massive scalars in GR. We start by evaluating the Fourier transform in the second line of (5.12). We can write rewrite it as follows

$$\int \frac{d^{D}q'_{1}}{(2\pi)^{D}} 2\pi \,\delta(2p'_{1} \cdot q'_{1} - q'^{2}_{1}) 2\pi \,\delta(2p'_{2} \cdot q'_{1} + q'^{2}_{1}) e^{ib \cdot q'_{1}} \mathcal{A}_{0}(s, -q'^{2}_{1}) = \int \frac{d^{D}q'_{1}}{(2\pi)^{D}} 2\pi \,\delta(2p'_{1} \cdot q'_{1}) 2\pi \,\delta(2p'_{2} \cdot q'_{1}) e^{ib \cdot q'_{1}} \frac{a_{0}}{q'^{2}_{1}}$$
(5.23)

where we have introduced the notation $p'_1 = p_1 - q_1$, $p'_2 = p_2 + q_1$ and we have used the fact that any correction involving positive integer powers of q'^2_1 , such as those due to the application of Eq. (A.40), can be neglected because they give rise to contact terms in *b*. At this point we may use (A.44) and (A.45) in the familiar way, up to an important point: the result only depends on the projection of b^{μ} orthogonal to p'_1 and p'_2 (and not to p_1 , p_2), so that

$$\int \frac{d^D q'_1}{(2\pi)^D} 2\pi \,\delta(2p'_1 \cdot q'_1) 2\pi \,\delta(2p'_2 \cdot q'_1) e^{ib \cdot q'_1} \frac{a_0}{q'_1^2} = \frac{1}{4Ep} \frac{a_0}{4\pi} \frac{\Gamma(-\epsilon)}{(\pi b'^2(q_1))^{-\epsilon}} \,, \tag{5.24}$$

where $b'(q_1)^2$ is the square of such projection,

$$b^{\prime 2}(q_1) = b^2 - \frac{1}{p^2} (b \cdot q_1)^2 \,. \tag{5.25}$$

At this point we can substitute back into (5.12), taking again into account that any positive integer power of q_1^2 can be safely dropped, so that

$$2 \operatorname{Im} \operatorname{FT}[\mathcal{A}_1] = \int \frac{d^D q_1}{(2\pi)^D} 2\pi \,\delta(2p_1 \cdot q_1) 2\pi \,\delta(2p_2 \cdot q_1) e^{ib \cdot q_1} \left[\frac{a_0}{q_1^2} \right] \left[\frac{1}{4Ep} \frac{a_0}{4\pi} \frac{\Gamma(-\epsilon)}{(\pi b'^2(q_1))^{-\epsilon}} \right].$$
(5.26)

In this step, we need to expand (5.25) appearing in the second factor to subleading order for large *b*. This term also depends on q_1 , which manifests the lack of complete factorization, if such subleading terms are taken into account. In the end, using (A.45) and its derivatives with respect to b^{μ} , one arrives at

$$2 \operatorname{Im} \operatorname{FT}[\mathcal{A}_1] = \left[2\delta_0(s, b)\right]^2 - \frac{2(1 - 2\epsilon)}{p^2 (b^2)^{1 - 2\epsilon}} \left[\frac{1}{4Ep} \frac{a_0}{4\pi} \frac{\Gamma(1 - \epsilon)}{\pi^{-\epsilon}}\right]^2 + \mathcal{O}(b^{-4 + 4\epsilon}).$$
(5.27)

To summarize, the FT of the two-body convolution of the tree-level amplitude is the product of the FT only to leading order in $1/b^2$, while this factorization receives corrections to first subleading order. On the one hand, here we calculated it using the properties of the Fourier transform under considerations to achieve the "almost factorized" form (5.26). On the other hand, in Eq. (6.111) below we provide an expression for the two-particle convolution accurate to leading and subleading order in q^2 . One can verify that, using (A.40), taking FT of (6.111) one indeed recovers (5.27).

5.1.2. Inelastic unitarity

The first inelastic contributions come from the three-particle cut of the 2 \rightarrow 2 amplitude A. This is obtained by "gluing together" two copies of a 2 \rightarrow 3 amplitude



where blue and green lines represent classical states with masses m_1 and m_2 , while massless lines are drawn in black. The drawing inside the dashed bubbles does not represent a specific topology, but just provides a visual help to recall the definition of the q_i variables, such that

$$q_1 + q_2 + k = 0. (5.29)$$

We will discuss momentarily on the precise form of the product of five point amplitudes appearing in this gluing, which of course involve index contractions with appropriate projector insertions. For the moment, let us write it schematically as

$$2 \operatorname{Im}_{3pc} \mathcal{A} = \int d(\operatorname{LIPS}) \mathcal{A}_{\mathrm{L}}^{(5)} \mathcal{A}_{\mathrm{R}}^{(5)*}$$
(5.30)

T.

(where L. and R. stand for "left" and "right") or in the pictorial form

$$2 \operatorname{Im}_{3pc} \bigvee_{p_{2}} p_{3} \xrightarrow{(\ q \uparrow) } p_{3} = \int d(\operatorname{LIPS}) \bigvee_{q_{2} \downarrow \xrightarrow{(\ q \uparrow) } k_{2}} p_{4} \xrightarrow{(\ q \uparrow) } p_{4} \xrightarrow{(\ q \downarrow) } p_{4} \xrightarrow$$

where

$$d(\text{LIPS}) = \frac{d^D k}{(2\pi)^D} 2\pi \theta(k^0) \delta(k^2) \frac{d^D k_1}{(2\pi)^D} 2\pi \theta(k_1^0) \delta(k_1^2 + m_1^2) 2\pi \theta(k_2^0) \delta(k_2^2 + m_2^2)$$
(5.32)

is the Lorentz-invariant phase-space measure for the intermediate states. Using momentum conservation we can retain only, say, k and k_1 as independent integration variables in the measure.

We enforce the classical limit as follows. As by now familiar, we assume that the momentum transfer $q = p_1 + p_4$ for the full $2 \rightarrow 2$ process be small compared to the momenta of particle 1 and 2, depicted by thick colored lines in (5.31), so that to leading order

$$p_1 \sim -m_1 u_1 + \mathcal{O}(q) \sim -k_1, \qquad p_2 \sim -m_2 u_2 + \mathcal{O}(q) \sim -k_2.$$
 (5.33)

By momentum conservation, this is tantamount to assuming the scaling

$$q_1 \sim q_2 \sim k \sim \mathcal{O}(q) \,. \tag{5.34}$$

We thus take the momenta $q_{1,2}$ and k to be simultaneously small, of the order of the elastic momentum transfer q. Equivalently, reinstating momentarily \hbar , this can be regarded as a formal $\hbar \rightarrow 0$ limit in which the wavenumbers $q_{1,2}/\hbar$ and k/\hbar are held fixed. Eq. (5.34) expresses the soft-region scaling that we discussed when evaluating loop integrals and is justified by the fact that it captures all non-analytic contributions in q^2 . It also allows us to simplify the integration in a similar way as before. Using $q_1 = p_1 + k_1$ and k as independent integration variables simplifies the LIPS measure as follows to leading order in the scaling (5.34),

$$d(\text{LIPS}) \simeq \frac{d^{D}k}{(2\pi)^{D}} 2\pi \theta(k^{0}) \delta(k^{2}) \frac{d^{D}q_{1}}{(2\pi)^{D}} 2\pi \delta(2p_{1} \cdot q_{1}) 2\pi \delta(2p_{2} \cdot (q_{1} + k)), \qquad (5.35)$$

Т

and leads us to adopt the following routing

$$2 \operatorname{Im}_{3pc} \begin{array}{c} p_{1} \xleftarrow{} \\ (q \uparrow) \\ p_{2} \xleftarrow{} \\ p_{3} \xleftarrow{} \\ p_{4} \xleftarrow{} \\ p_{5} \xleftarrow{} \\$$

for the integrated momenta. Since the massless momentum k will mostly remain as a spectator in the following manipulations let us also introduce a shorthand notation for its on-shell phase space integration,

$$\int_{k} \equiv \int \frac{d^{D}k}{(2\pi)^{D}} 2\pi \theta(k^{0})\delta(k^{2}).$$
(5.37)

Our basic ingredient (5.30) will be the 2 \rightarrow 3 tree-level amplitude in Eq. (3.1) of [104], which had been obtained from the low energy limit of a string amplitude in a toroidal compactification. In this way it naturally includes the contributions of the dilaton, of vectors and scalars arising in the Kaluza–Klein compactification, and of the graviton. To include all such contributions simultaneously, we find it convenient to formally promote all spacetime indices μ , ν , . . . to 10-dimensional ones M, N, . . . according to conventions specified below. The tree-level contribution to $\mathcal{A}^{(5)}$ in the classical limit reproduces the result of [123,127,288] and can be then written in the following convenient form,³⁹

$$\begin{aligned} \mathcal{A}_{0}^{(5)MN} &= (8\pi G)^{\frac{3}{2}} \left\{ \beta \left[-\frac{p_{1}^{M} p_{1}^{N} (k \cdot q_{1})}{(p_{1} \cdot k)^{2} q_{2}^{2}} - \frac{p_{2}^{M} p_{2}^{N} (k \cdot q_{2})}{(p_{2} \cdot k)^{2} q_{1}^{2}} \right. \\ &+ \frac{p_{1}^{M} (q_{1} - q_{2})^{N} + p_{1}^{N} (q_{1} - q_{2})^{M}}{2(p_{1} \cdot k) q_{2}^{2}} - \frac{p_{2}^{M} (q_{1} - q_{2})^{N} + p_{2}^{N} (q_{1} - q_{2})^{M}}{2(p_{2} \cdot k) q_{1}^{2}} \\ &+ \frac{(q_{1} - q_{2})^{M} (q_{1} - q_{2})^{N}}{2q_{1}^{2} q_{2}^{2}} \right] + 8 \frac{((p_{1} \cdot k) p_{2}^{M} - (p_{2} \cdot k) p_{1}^{M}) ((p_{1} \cdot k) p_{2}^{N} - (p_{2} \cdot k) p_{1}^{N})}{q_{1}^{2} q_{2}^{2}} \\ &+ (2p_{1} \cdot p_{2}) \left(\frac{4p_{1}^{M} p_{1}^{N} \frac{k \cdot p_{2}}{1 k \cdot p_{1}} - 2(p_{1}^{M} p_{2}^{N} + p_{1}^{N} p_{2}^{M})}{q_{2}^{2}} + \frac{4p_{2}^{M} p_{2}^{N} \frac{k \cdot p_{1}}{k \cdot p_{2}} - 2(p_{1}^{M} p_{2}^{N} + p_{1}^{N} p_{2}^{M})}{q_{1}^{2}} \\ &+ \frac{(q_{1} - q_{2})^{M} \left(-2(p_{1} \cdot k) p_{2}^{N} + 2(p_{2} \cdot k) p_{1}^{N} \right) + (q_{1} - q_{2})^{N} \left(-2(p_{1} \cdot k) p_{2}^{M} + 2(p_{2} \cdot k) p_{1}^{M} \right)}{q_{1}^{2} q_{2}^{2}} \right) \right\} \end{aligned}$$

the quantity β is defined in (5.41) below depending on the theory under consideration. The main feature of (5.38) is that it satisfies $k_M A_0^{(5)MN} = k_N A_0^{(5)MN} = 0$ for arbitrary values of the free index, which makes the calculations in general dimensions easier [48,258]. Notice that the terms proportional to β and the remaining terms are independently gauge invariant. It is also symmetric in its two spacetime indices, $A_0^{(5)MN} = A_0^{(5)MN}$.

³⁹ The full theory also allows for other $2 \rightarrow 3$ processes involving fermionic external states. However, these would yield contributions that are subleading in the limit of small momentum transfer.

When focusing on $\mathcal{N} = 8$ supergravity, we choose the following 10D kinematics $[106]^{40}$

$$p_1^M = (p_1^\mu; 0, \dots, 0, 0, m_1), \qquad p_2^M = (p_2^\mu; 0, \dots, 0, m_2, 0),$$
(5.39)

where $p_{1,2}^{\mu}$ are $(4-2\epsilon)$ -dimensional momenta and the dots stand for $2+2\epsilon$ zero entries. In contrast, in the case of GR,

$$p_1^M = (p_1^\mu; 0, \dots, 0, 0, 0), \qquad p_2^M = (p_2^\mu; 0, \dots, 0, 0, 0).$$
 (5.40)

The momenta k^M and $q_{1,2}^M$ are always non-trivial only in the $4 - 2\epsilon$ non-compact directions so that they effectively coincide with k^{μ} and $q_{1,2}^{\mu}$, e.g. $q_1^M = (q_1^{\mu}; 0, ..., 0, 0, 0)$. Another difference between $\mathcal{N} = 8$ and GR is related to the contribution of the dilaton as an internal state exchanged between the massive objects. This exchange can only occur in $\mathcal{N} = 8$ supergravity, while for GR the contribution of the dilaton has to be subtracted. This can be taken into account by specifying the parameter β as follows in the two theories,

$$\beta^{\mathcal{N}=8} = 4m_1^2 m_2^2 \sigma^2 , \qquad \beta^{\text{GR}} = 4m_1^2 m_2^2 \left(\sigma^2 - \frac{1}{D-2}\right).$$
(5.41)

Let us now describe more precisely how the N = 8 supergravity and GR amplitudes enter the integrand (5.30). In both cases, the kinematics is dictated by the momentum flow depicted in Eq. (5.31), so that

$$\mathcal{A}_{\rm L}^{(5)} = \mathcal{A}^{(5)}(p_1, p_2, k_1, k_2, k), \qquad \mathcal{A}_{\rm R}^{(5)*} = \mathcal{A}^{(5)*}(p_4, p_3, -k_1, -k_2, -k).$$
(5.42)

Index contractions are instead theory dependent. In the maximally supersymmetric case, we simply work in M, N, \ldots indices and saturate the Lorentz indices with the Minkowski metric, contracting the two amplitudes according to

$$\mathcal{A}_{L}^{(5)}\mathcal{A}_{R.}^{(5)*} = \mathcal{A}_{L}^{(5)MN}\eta_{MR}\eta_{NS}\mathcal{A}_{R.}^{(5)*RS}, \qquad (5.43)$$

because in this case the amplitude $\mathcal{A}^{(5)MN}$ is transverse and traceless. As already stressed, the advantage of this trick is that it effectively encompasses graviton, dilaton and Kaluza–Klein vectors/scalars at the same time. For GR we work with μ, ν, \ldots indices and on the top of this we need to subtract the contribution of the dilaton, and thus we employ the contraction

$$\mathcal{A}_{\rm L}^{(5)}\mathcal{A}_{\rm R}^{(5)*} = \mathcal{A}_{\rm L}^{(5)\mu\nu} \left[\eta_{\mu\rho} \eta_{\nu\sigma} - \frac{1}{D-2} \eta_{\mu\nu} \eta_{\rho\sigma} \right] \mathcal{A}_{\rm R}^{(5)*\rho\sigma} \,.$$
(5.44)

The structure within square brackets arises from the transverse-traceless projector over physical degrees of freedom $\Pi_{\mu\nu,\rho\sigma}(k)$. Explicitly, for any reference null vector r^{μ} , letting $\lambda^{\mu} = -r^{\mu}/(r \cdot k)$ and $\Pi^{\mu\nu} = \eta^{\mu\nu} + \lambda^{\mu}k^{\nu} + \lambda^{\nu}k^{\mu}$, one can construct this projector according to

$$\Pi^{\mu\nu,\rho\sigma}(k) = \frac{1}{2} \left(\Pi^{\mu\rho} \Pi^{\nu\sigma} + \Pi^{\mu\sigma} \Pi^{\nu\rho} - \frac{2}{D-2} \Pi^{\mu\nu} \Pi^{\rho\sigma} \right).$$
(5.45)

The transversality condition $k_{\mu}A^{(5)\mu\nu} = 0$ then turns (5.45) into the combination within square brackets in (5.44), which is the one appearing also in the de Donder propagator, simplified by using $A^{(5)\mu\nu} = A^{(5)\nu\mu}$. We will use (5.43) and (5.44) as a convenient shorthand notation to suppress explicit index contractions between $2 \rightarrow 3$ amplitude also in Section 8.

We can now start from (5.36) go to impact-parameter space via the usual Fourier transform for the $2 \rightarrow 2$ process (see Appendix A.3)

In the remainder of this section, we need not worry about the difference between FT[A] (A.31) and \tilde{A} (A.44), since we will be working to leading order in $q \sim 1/b$. When applying this Fourier transform to both sides of (5.36), we change integration variable by letting $q = q_1 + q_4$. After doing this, an important step is to use

$$\delta(p_1 \cdot q_1)\delta(2p_1 \cdot q) = \delta(p_1 \cdot q_1)\delta(2p_1 \cdot (q_1 + q_4)) = \delta(p_1 \cdot q_1)\delta(2p_1 \cdot q_4)$$
(5.47)

and similarly

$$\delta(p_2 \cdot (q_1 + k))\delta(2p_2 \cdot q) = \delta(p_2 \cdot (q_1 + k))\delta(2p_2 \cdot (q_4 - k)), \tag{5.48}$$

⁴⁰ For sake of simplicity here we focus directly on the case $\phi = \frac{\pi}{2}$ in the notations of that reference.

which hold thanks to the properties of the delta function. Then we get,

$$2 \operatorname{Im}_{3pc} \tilde{\mathcal{A}} = \int_{k} \int \frac{d^{D}q_{1}}{(2\pi)^{D}} 2\pi \,\delta(2p_{1} \cdot q_{1}) 2\pi \,\delta(2p_{2} \cdot (q_{1} + k)) e^{ib \cdot q_{1}} \xrightarrow{\left(\begin{array}{c} q_{1} \\ q_{1} \end{array}\right)} \xrightarrow{\left(\begin{array}{c} q_{1} \end{array}\right)} \xrightarrow{\left$$

We may reinterpret the 3 \rightarrow 2 amplitude in the second line as a 2 \rightarrow 3 one by flipping all external momenta and relabel $q_4 = -q'_1$ in the integration. Since moreover to leading order $p_4 \simeq -p_1$, $p_3 \simeq -p_2$ we can write

$$2 \operatorname{Im}_{3pc} \tilde{\mathcal{A}} = \int_{k} \left[\int \frac{d^{D}q_{1}}{(2\pi)^{D}} 2\pi \delta(2p_{1} \cdot q_{1}) 2\pi \delta(2p_{2} \cdot (q_{1} + k)) e^{ib \cdot q_{1}} \bigvee_{i=1}^{j} \frac{1}{(q_{1} \uparrow_{i})} \right] \\ \times \left[\int \frac{d^{D}q_{1}}{(2\pi)^{D}} 2\pi \delta(2p_{1} \cdot q_{1}') 2\pi \delta(2p_{2} \cdot (q_{1}' + k)) e^{ib \cdot q_{1}'} \bigvee_{i=1}^{j} \frac{1}{(q_{1} \uparrow_{i})} \right]^{*}$$
(5.50)

In this equation, the two quantities appearing within square brackets are equal to each other and can be identified as the Fourier transform of the five-point amplitude, $\tilde{\mathcal{A}}^{(5)}$. Of course, (5.50) only fixes the definition of $\tilde{\mathcal{A}}^{(5)}$ up to an overall phase, which would of course cancel in the product $\tilde{\mathcal{A}}^{(5)}\tilde{\mathcal{A}}^{(5)*}$. In order to treat the two momenta q_1 and q_2 on the same footing, it is natural to define

$$\tilde{\mathcal{A}}^{(5)} = \int \frac{d^{D}q_{1}}{(2\pi)^{D}} 2\pi \,\delta(2p_{1} \cdot q_{1}) 2\pi \,\delta(2p_{2} \cdot q_{2}) e^{ib_{1} \cdot q_{1} + ib_{2} \cdot q_{2}} \xrightarrow{(q_{1})} k_{1} \qquad (5.51)$$

with $b^{\alpha} = b_1^{\alpha} - b_2^{\alpha}$ and of course $q_1 + q_2 + k = 0$ as indicated in the figure. We shall come back to this Fourier transform in Section 8. As we shall see, this is indeed the appropriate Fourier transform to leading order in the PM expansion. We have thus related the imaginary part of the *b*-space $2 \rightarrow 2$ amplitude to the integral over the massless particle's phase space of the product of two *b*-space $2 \rightarrow 3$ amplitudes,

$$2 \operatorname{Im} 2\delta_2 = 2 \operatorname{Im}_{3pc} \tilde{\mathcal{A}} = \int_k \tilde{\mathcal{A}}^{(5)} \tilde{\mathcal{A}}^{(5)*} \,.$$
(5.52)

Let us recall that index contractions are left implicit and can be restored by using (5.43) for $\mathcal{N} = 8$ supergravity and (5.44) for GR.

To conclude, we have discussed how the imaginary part of the amplitude in *b*-space arising from the three-particle cut can be calculated in two different ways. The first is to calculate the three-particle cut (5.30) in momentum space and then take its Fourier transform at the very end. The second is to first take Fourier transforms of the five-point amplitudes via (5.51), contract them as in (5.43), (5.44) and integrate over the graviton phase space, according to (5.52). The former method is more powerful as far as loop integration techniques are concerned. The reason is that, in momentum space, one can interpret the phase-space delta functions as "cut propagators" and apply the full machinery of master integrals and differential equations for two-loop topologies. The latter makes the interpretation of this imaginary part more transparent,

5.2. Calculation of Im $2\delta_2$

By the relation (5.52), it is easy to see that the divergent part of

$$\operatorname{Im} 2\delta_2 = \operatorname{Im}_{3pc} \tilde{\mathcal{A}}_2 \tag{5.53}$$

can be obtained by only considering the leading term in the soft limit $k^{\mu} \rightarrow 0$ of the five-point amplitude in (5.38) that diverges as $\frac{1}{\omega}$, with $\omega = k^0$, and that arises from the first two lines of (5.38). Since in this limit $q_1 \simeq -q_2$, the Fourier transform (5.51) can be easily performed and one obtains the results summarized in Section 4 of [106]. In this way, one obtains

$$(\operatorname{Im}2\delta_{2})_{\mathcal{N}=8}(b) = -\frac{G^{3}(\beta^{\mathcal{N}=8})^{2}}{\pi b^{2} \epsilon (\sigma^{2}-1)^{2}} \left[\sigma^{2} + \frac{\sigma(\sigma^{2}-2)}{(\sigma^{2}-1)^{\frac{1}{2}}} \operatorname{arccosh} \sigma \right] + \mathcal{O}(\epsilon^{0})$$
(5.54)

for massive $\mathcal{N} = 8$ supergravity, where $\beta^{\mathcal{N}=8}$ is given in (5.41), and [289]

$$(\operatorname{Im} 2\delta_{2})_{\rm gr}(b) \simeq -\frac{1}{2\epsilon} \frac{G^{3}(\beta^{\rm GR})^{2}}{\pi b^{2}(\sigma^{2}-1)^{2}} \left[\frac{8-5\sigma^{2}}{3} - \frac{\sigma(3-2\sigma^{2})}{(\sigma^{2}-1)^{\frac{1}{2}}} \operatorname{arccosh} \sigma \right] + \mathcal{O}(\epsilon^{0})$$
(5.55)

for the graviton contribution in GR, where β^{GR} is given in (5.41). We refer to Section 5.1 of [106] for more details on these steps, while in this report we shall obtain Eqs. (5.55) and (5.54) as special cases of a more general setup from Eqs. (7.92) and (7.106) below.

The complete expression for $(\text{Im } 2\delta_2)$, including not only the divergent contribution but also the finite one, can be obtained from reverse unitarity. In GR one gets Eq. (6.28) of [47]

$$\begin{aligned} \operatorname{Im} 2\delta_{2} &= \frac{2m_{1}^{2}m_{2}^{2}(2\sigma^{2}-1)^{2}G^{3}}{\pi b^{2}(\sigma^{2}-1)^{2}} (\pi b^{2}e^{\gamma_{E}})^{3\epsilon} \\ &\times \left[\left(-\frac{1}{\epsilon} + \log(4(\sigma^{2}-1)) \right) \left(\frac{\sigma \left(2\sigma^{2}-3 \right) \operatorname{arccosh} \sigma}{\sqrt{\sigma^{2}-1}} + \frac{8-5\sigma^{2}}{3} \right) \right. \\ &- (\operatorname{arccosh} \sigma)^{2} \left(\frac{\sigma \left(2\sigma^{2}-3 \right)}{\sqrt{\sigma^{2}-1}} + \frac{2(\sigma^{2}-1)(4\sigma^{4}-12\sigma^{2}-3)}{(1-2\sigma^{2})^{2}} \right) \\ &+ (\operatorname{arccosh} \sigma) \frac{\sigma \left(88\sigma^{6}-240\sigma^{4}+240\sigma^{2}-97 \right)}{3\left(1-2\sigma^{2} \right)^{2}\sqrt{\sigma^{2}-1}} \\ &+ \operatorname{Li}_{2}(1-z^{2}) \frac{\sigma \left(3-2\sigma^{2} \right)}{\sqrt{\sigma^{2}-1}} + \frac{-140\sigma^{6}+220\sigma^{4}-127\sigma^{2}+56}{9\left(1-2\sigma^{2} \right)^{2}} \right]. \end{aligned}$$
(5.56)

while in massive $\mathcal{N} = 8$ supergravity one gets Eq. (4.21) of [47]

2 2 2

$$\operatorname{Im} 2\delta_{2}^{\mathcal{N}=8} = -\frac{16m_{1}^{2}m_{2}^{2}G^{3}}{\pi b^{2}} \frac{\sigma^{4}(\pi b^{2}e^{\gamma_{E}})^{3\epsilon}}{(\sigma^{2}-1)^{2}} \left\{ \frac{1}{\epsilon} \left(\sigma^{2} + \frac{\sigma(\sigma^{2}-2)}{(\sigma^{2}-1)^{\frac{1}{2}}} \operatorname{arccosh} \sigma \right) - \log(4(\sigma^{2}-1)) \left[\sigma^{2} + \frac{\sigma(\sigma^{2}-2)}{(\sigma^{2}-1)^{\frac{1}{2}}} \operatorname{arccosh} \sigma \right] + (\sigma^{2}-1) \left[1 + \frac{\sigma(\sigma^{2}-2)}{(\sigma^{2}-1)^{\frac{3}{2}}} \right] (\operatorname{arccosh} \sigma)^{2} + \frac{\sigma(\sigma^{2}-2)}{(\sigma^{2}-1)^{\frac{1}{2}}} \operatorname{Li}_{2}(1-z^{2}) + 2\sigma^{2} \right\},$$
(5.57)

Notice that the previous equations have both terms that, at high energy, behave as $(\log \sigma)^2$. However these terms cancel each others and one is left with an high energy behavior as $\log \sigma$ in agreement with the analysis of [104] based on analyticity and crossing symmetry.

5.3. Radiation reaction from Im $2\delta_2$ by real analyticity

In this section we follow the approach of Ref. [106] and using unitarity and real analyticity, we obtain a relation that connects two terms of Im $2\delta_2$ to the radiation reaction contribution, Re $2\delta_2^{(RR)}$, of the two-loop eikonal. Such a relation is

$$-\frac{i}{\pi\epsilon}\operatorname{Re} 2\delta_2^{\mathrm{RR}} \mapsto \left[1 + \frac{i}{\pi}\left(-\frac{1}{\epsilon} + \log(\sigma^2 - 1)\right)\right]\operatorname{Re} 2\delta_2^{\mathrm{RR}}.$$
(5.58)

The part in the round bracket comes from the integral over the frequency of the graviton given by⁴¹

$$(b^2)^{-1+3\epsilon} \int_0^{\omega b} \frac{d\omega}{\omega} (\omega b)^{-2\epsilon}$$
(5.59)

where the factor $(b^2)^{-1+3\epsilon}$ is precisely the one expected (also on dimensional ground) to appear in $2\delta_2$ and $\overline{\omega b}$ is an appropriate upper limit on the classical dimensionless quantity ωb . On the other hand, the integral over ω produces a $\frac{1}{\epsilon}$ divergence in the particular combination:

$$\int_{0}^{\omega b} \frac{d\omega}{\omega} (\omega b)^{-2\epsilon} = -\frac{1}{2\epsilon} (\overline{\omega b})^{-2\epsilon} = -\frac{1}{2\epsilon} + \log \overline{\omega b} + \mathcal{O}(\epsilon)$$
(5.60)

To obtain an estimate of the cutoff $\overline{\omega b}$, one can use (1.47) for single-graviton exchanges, according to which

$$|p_1^0 q_1^0| \approx |\vec{p}_1 \cdot \vec{q}_1| \le |\vec{p}_1| \, |\vec{q}_1| \,, \qquad |p_2^0 q_2^0| \approx |\vec{p}_2 \cdot \vec{q}_2| \le |\vec{p}_2| \, |\vec{q}_2| \tag{5.61}$$

and therefore

$$\omega = k^{0} = -q_{1}^{0} - q_{2}^{0} \le |q_{1}^{0}| + |q_{2}^{0}| \le \frac{|\vec{p}_{1}|}{|p_{1}^{0}|} |\vec{q}_{1}| + \frac{|\vec{p}_{2}|}{|p_{2}^{0}|} |\vec{q}_{2}|.$$
(5.62)

In the classical limit, $|\vec{q}_i| \sim 1/b$ and thus ωb is bounded by

$$\overline{\omega b} = \frac{|\vec{p}_1|}{|p_1^0|} + \frac{|\vec{p}_2|}{|p_2^0|} \,. \tag{5.63}$$

Using the explicit expressions (1.13), (1.15), (1.16) in the CM frame, one finds

$$\overline{\omega b} = \sqrt{\sigma^2 - 1} \left(1 + \mathcal{O}(\sigma - 1) \right) \tag{5.64}$$

In this way we have obtained the term in the round bracket in (5.58). In order to obtain the last term outside of the round bracket one must observe that, because of real analyticity, the amplitude between the two branch points $\sigma = \pm 1$ must be real and therefore in this region the term in the round bracket should be $\log(1 - \sigma^2)$. When we go from this region to the physical region $\sigma \ge 1$ then we get

$$\log(1 - \sigma^2 - i0) = \log(\sigma^2 - 1) - i\pi$$
(5.65)

obtaining Eq. (5.58) that allows us to determine the radiation reaction from the infrared divergent term of Im $2\delta_2$. For GR we obtain:

Re
$$2\delta^{\text{RR,GR}}(b) = \frac{8G^3m_1^2m_2^2(\sigma^2 - \frac{1}{2})^2}{\hbar b^2(\sigma^2 - 1)^2} \left[\frac{8 - 5\sigma^2}{3} - \frac{\sigma(3 - 2\sigma^2)}{(\sigma^2 - 1)^{\frac{1}{2}}}\operatorname{arccosh}\sigma\right],$$
 (5.66)

while in the case of massive N = 8 supergravity one gets:

$$\operatorname{Re} 2\delta_{2}^{\operatorname{RR},\mathcal{N}=8}(b) = \frac{16G^{3}m_{1}^{2}m_{2}^{2}\sigma^{4}}{\hbar b^{2}(\sigma^{2}-1)^{2}} \left[\sigma^{2} + \frac{\sigma(\sigma^{2}-2)}{(\sigma^{2}-1)^{\frac{1}{2}}}\operatorname{arccosh}\sigma\right].$$
(5.67)

In conclusion, we have derived the radiation reaction from Eq. (5.52) and from the properties of unitarity and real analyticity, without needing to construct the full two-loop amplitude. The arguments that brought us to the structure in the round bracket of (5.58) are confirmed by the first two terms of the explicit results for Im $2\delta_2$ given in (5.56) and (5.57).

⁴¹ We need to keep $D = 4 - 2\epsilon$ only for the integral over $|\vec{k}|$ while the integration over the angular variables can be done for $\epsilon = 0$, so that effectively $d^{D-1}\vec{k} = |\vec{k}|^{2-2\epsilon} d|\vec{k}| \sin \theta \, d\theta \, d\varphi$.

6. The 2-body eikonal at 3PM

The exponentiation pattern of Eq. (4.4), which we explicitly illustrated in Section 4 up sub-leading order, suffices to eliminate all super-classical redundancies arising from tree-level and one-loop order. Thus, it allows one to extract from the amplitude genuine classical information up to two loops, i.e. $\mathcal{O}(G^3)$ or 3PM order. The purpose of this chapter is to perform this step explicitly. We will start by sketching the calculation of the two-loop amplitude in the classical limit, to then check that super-classical terms cancel out consistently with Eq. (4.4) and finally obtain the 3PM correction to the eikonal, $2\delta_2$.

A novelty compared to lower loop-orders is the emergence of an imaginary part in the amplitude that is due, via the unitarity relation, to the presence of nontrivial three-particle cuts, as we discussed in Section 5. At one loop, the imaginary part of the amplitude is due to two-particle cuts only and is effectively subtracted in the exponentiation (4.4) at classical level, so that $2\delta_1$ is real and only the quantum remainder $2\Delta_1$ can develop an imaginary part. By contrast, at two loops, the three-particle inelastic channels leave behind a nonzero (in fact, infrared divergent!) imaginary part of $2\delta_2$, as in (5.20). In ensuing chapters we will discuss a more systematic way to deal with such additional channels, restoring the expected Hermiticity of the eikonal 2δ , which requires to promote it to an operator.

The focus of this chapter is instead on illustrating an interesting physical phenomenon that is captured by $2\delta_2$: radiation-reaction effects, which first occur at 3PM order and can be singled out by their time-reversal odd expansion in the small-velocity limit. On the other hand, such radiation-reaction contributions are crucial in order to ensure consistency, finiteness and universality of the results at high energies (or high velocities). In particular they are instrumental in ensuring agreement between the high-energy behavior of massive scattering and the universal result for the deflection angle of massless objects.

As for the one-loop level, we shall first focus on the technically simpler $\mathcal{N} = 8$ case, considering the *s*-*u* symmetric scattering of massive states obtained via Kaluza–Klein compactification, and then move to collisions of massive scalars minimally coupled to GR. We also provide more details on the relation between the eikonal phase, the phase shifts and the radial action, which as we shall see is also instrumental in performing the analytic continuation from the case of unbound orbits to bound systems.

6.1. Partial wave decomposition: from b to b_{I}

In Section 2.2 we discussed how classical observables emerge in the leading eikonal approximation for the massless scattering. Starting from the resummed amplitude in the impact parameter representation, in the classical regime we can perform the Fourier transform back to momentum space (2.27) by using a saddle-point approximation (2.28). The resummed amplitude can then be decomposed in partial waves, as in (1.63), and in the classical limit (2.40), as expected, the orbital angular momentum of the system becomes large $J/\hbar = j \gg 1$. The partial wave analysis of that section is limited to the case of the leading eikonal in D = 4, while the derivation based on the impact parameter representation seems more general and was already used in (4.59) to derive the 2PM deflection angle in GR. The purpose of this section is to show that the same results can be obtained also by performing the partial wave decomposition in the classical regime starting from the eikonal amplitude at an arbitrary PM order [290] (see also [169]).

We start by generalizing (2.27) to the massive scattering in D = 4 by first defining

$$\tilde{S}(s,b) = 1 + i\tilde{\mathcal{A}}(s,b) = e^{i\operatorname{Re}2\delta(s,b)}.$$
(6.1)

Here we remove the $1/\epsilon$ pole in $2\delta_0$, associated to the Coulombic falloff of the field, and introduce an appropriate cutoff scale in the leftover logarithm, as we did in (2.34), and we temporarily drop the imaginary part of $2\delta(s, b)$ that enters at 3PM, which will not enter the ensuing stationary-phase arguments. In Q-space, Eq. (6.1) translates to

$$S(s, Q) = 4Ep \int d^2b \, e^{-iQ \cdot b} e^{i\operatorname{Re} 2\delta(s,b)} = 8\pi Ep \int db \, b J_0\left(\frac{Qb}{\hbar}\right) e^{i\operatorname{Re} 2\delta(s,b)},\tag{6.2}$$

where we used the fact that 2δ depends only on b^2 and the identity

$$J_0(x) = \int_{-\pi}^{+\pi} \frac{d\phi}{2\pi} \, e^{ix\sin\phi} \,. \tag{6.3}$$

(together with the fact that the integrand is a periodic function of ϕ with period 2π). For large arguments $x = Qb/\hbar \gg 1$, one also has

$$J_0(x) \sim \frac{1}{\sqrt{2\pi}} \left(e^{i(x - \frac{\pi}{4})} + e^{-i(x - \frac{\pi}{4})} \right).$$
(6.4)

Using this relation and picking the first term, which corresponds to Q^{μ} being anti-aligned to b^{μ} , as in our conventions for gravitational scattering, we obtain

$$S(s,Q) \sim (\ldots) \int db \, b \, e^{\frac{i}{\hbar}Qb+i\operatorname{Re}2\delta(s,b)}.$$
(6.5)

Here and in the next few equations we need only worry about rapidly oscillating phase factors, and accordingly omit irrelevant slowly-varying prefactors. We then go to angular momentum-space by projecting on the *j*th Legendre polynomial as dictated by (1.63),

$$S_j(s) \sim (\cdots) \int db \, b \int_{-1}^{+1} d(\cos \Theta) P_j(\cos \Theta) e^{\frac{i}{\hbar} 2pb \sin \frac{\Theta}{2} + i \operatorname{Re} 2\delta(s,b)}, \qquad (6.6)$$

where we employed the relation between Q and Θ , which is again the same as the one encountered in (2.31),

$$Q = 2p\sin\frac{\Theta}{2}.$$
(6.7)

Using finally the large-*j* limit of the Legendre polynomial (2.48) we arrive at

$$S_{j}(s) \sim \int_{0}^{\pi} d\Theta \int db (\cdots) e^{-ij\Theta + \frac{i}{\hbar} 2pb \sin \frac{\Theta}{2} + i\operatorname{Re} 2\delta(s,b)}.$$
(6.8)

Collecting the terms appearing in the exponent, we recover the phase shift in the classical limit from (2.44) and

$$\chi(s,J) = \frac{1}{\hbar} \left(-J\Theta + 2pb \sin \frac{\Theta}{2} \right) + \operatorname{Re} 2\delta(s,b), \qquad (6.9)$$

which must be extremized by a double saddle point in Θ and b at fixed

$$b_J = \frac{J}{p} \,. \tag{6.10}$$

Taking derivatives of (6.9), the two saddle point conditions give

$$b_J = b \cos \frac{\Theta}{2}, \qquad 2p \sin \frac{\Theta}{2} = -\hbar \frac{\partial \operatorname{Re} 2\delta(s, b)}{\partial b}.$$
 (6.11)

Eqs. (6.7), (6.10) and (6.11) identify the sought after relations among the properties of the trajectory depicted in Fig. 18, linking them to the eikonal phase. In the classical limit, $S_j(s)$ is itself a rapidly oscillating phase factor, $S_j(s) \propto e^{2i\delta_j(s)}$ with $2\delta_j(s) = \chi(s, J)$ given by (6.9). Using the saddle point conditions, one can also recast it as follows in terms of *b* and b_J ,

$$\chi(s,J) = \frac{2J}{\hbar} \left(-\arccos \frac{b_J}{b} + \sqrt{\frac{b^2}{b_J^2} - 1} \right) + \operatorname{Re} 2\delta(s,b).$$
(6.12)

The derivative of $\chi(s, J)$ in (6.9) with respect to $j = J/\hbar$ then takes the following form

$$-\hbar \frac{\partial \chi(s,J)}{\partial J} = \Theta + p \left[b_J - b \cos \frac{\Theta}{2} \right] \frac{\partial \Theta}{\partial J} - \left[2p \sin \frac{\Theta}{2} + \hbar \frac{\partial \operatorname{Re} 2\delta(s,b)}{\partial b} \right] \frac{\partial b}{\partial J},$$
(6.13)

so that, owing to the saddle point conditions (6.11), it gives back the saddle-point value of Θ ,

$$\Theta = -\hbar \, \frac{\partial \chi(s, J)}{\partial J} \,. \tag{6.14}$$

For this reason $\chi(s, J)$ is closely connected to the radial action I(s, J), which obeys [291,292]

$$\Theta + \pi = -\frac{\partial I(s,J)}{\partial J} \tag{6.15}$$

so that

$$I(s, J) = -\pi J + \hbar \chi(s, J).$$
(6.16)

The expression (6.9) then provides the connection with the eikonal phase.

Eqs. (6.9) and (6.12) provide this connection only implicitly, since one ought to solve the saddle point conditions in order to express the right-hand side as a function *J*. In particular, in (6.12), *b* should be eliminated using the first condition (6.11), which in turn involves the deflection angle calculated using the second condition in (6.11). In order to illustrate how this works in practice let us consider the case of the perturbative PM expansion⁴²:

$$\operatorname{Re}(2\delta) = -d_0 \ln(b) + \sum_{n=1}^{\infty} \frac{d_n}{nb^n} , \qquad (6.17)$$

⁴² Although the relation (6.12) is non-perturbative in *G* it may actually break down at sufficiently small impact parameter if the relation between *b* and b_l cannot be straightforwardly inverted.



Fig. 18. Classical $2 \rightarrow 2$ scattering in the center-of-mass frame (1.33). \vec{Q} denotes the impulse and Θ the deflection angle. The impact parameter \vec{b}_J is perpendicular to the spatial momentum of either particle in the far past, and thus defines the angular momentum according to $J = p b_J$, with $p = |\vec{p}| = |\vec{p}'|$. Its magnitude differs by that of the eikonal impact parameter \vec{b} by terms of order Θ^2 as in (6.11).

where d_n captures the (n + 1)PM correction. It is straightforward to derive explicitly the first few terms of the classical deflection angle expressed as a function of J perturbatively to any given order. For instance up to order $O(G^5)$ we have

$$\chi(s,J) = -d_0 \log J + \frac{d_1 p}{J} + \frac{d_2 p^2 - \frac{d_0^2}{12}}{2J^2} + \frac{d_3 p^3 - \frac{3}{8} d_0^2 d_1 p}{3J^3} + \frac{d_4 p^4 - \frac{d_0 p^2}{2} \left(d_2 d_0 - d_1^2\right) + \frac{1}{80} d_0^5}{4J^4} + \mathcal{O}(G^6)$$
(6.18)

so that via (6.14),

$$\begin{split} \Theta &= \frac{d_0}{J} + \frac{d_1 p}{J^2} + \frac{d_2 p^2 - \frac{d_0^2}{12}}{J^3} \\ &+ \frac{d_3 p^3 - \frac{3}{8} d_0^2 d_1 p}{J^4} + \frac{d_4 p^4 - \frac{d_0 p^2}{2} \left(d_2 d_0 - d_1^2\right) + \frac{1}{80} d_0^5}{J^5} + \mathcal{O}(G^6). \end{split}$$
(6.19)

Likewise, via (6.16), Eq. (6.18) also gives the corresponding expansion of the radial action I(s, J).

The derivation presented above linking $\chi(s, J)$ to the eikonal phase via saddle point conditions generalizes straightforwardly to generic *D*, as one can check thanks to known properties of Gegenbauer polynomials: [DLMF §18.15(i) using (18.7.1)] and [DLMF §10.17(i)].

6.2. Massive $\mathcal{N} = 8$ supergravity

We go back to the elastic $2 \rightarrow 2$ scattering in $\mathcal{N} = 8$ supergravity where the external states are massive thanks to non-trivial Kaluza–Klein momenta in the compact directions [167,211]. We focus on the *s*-*u* symmetric process already defined in Section 3.1.4, where the states labeled by p_1 and p_4 are dilatons with KK momentum in one compact direction and those labeled by p_2 and p_3 are axions with KK momentum in another orthogonal compact direction [47,106,167]

As before, the starting ingredient for our analysis is the momentum-space amplitude in the small-*q* expansion, where, as already emphasized, we can restrict to the terms that are non-analytic in q^2 . We will also neglect terms of order $\mathcal{O}(q^2)^{\frac{1}{2}-2\epsilon}$ or smaller, which are irrelevant for determining $2\delta_2$ and would only enter the calculation of the quantum remainder $2\Delta_2$. In view of this, the relevant part of the *s* – *u* symmetric two-loop scattering amplitude [117] is given by the following sum of scalar Feynman integrals I_T where *T* indicates the integral's topology:

$$\mathcal{A}_{2}(s, q^{2}) = (8\pi G)^{3} \frac{1}{2(4\pi)^{4-2\epsilon}} \left[(s - m_{1}^{2} - m_{2}^{2})^{4} + (u - m_{1}^{2} - m_{2}^{2})^{4} \right]$$

$$\times \left[(s - m_{1}^{2} - m_{2}^{2})^{2} (I_{\mathrm{III}} + 2I_{\mathrm{IX}}) + (u - m_{1}^{2} - m_{2}^{2})^{2} (I_{\mathrm{III}} + 2I_{\mathrm{IX}}) + t^{2}(I_{\mathrm{H}} + I_{\mathrm{H}}) \right].$$
(6.20)

Let us write down the integrals explicitly and sketch their evaluation as $q \rightarrow 0$. We follow the kinematic conventions of Section 1.2, in particular the variables introduced in (1.40), (1.29). In terms of these variables, the planar double box

integral reads

$$I_{\rm HI} = \int \frac{d^{\rm D}\ell_1}{i\pi^{D/2}} \frac{d^{\rm D}\ell_2}{[2\bar{m}_1\ell_1 \cdot u_1 + (\ell_1^2 - \ell_1 \cdot q)] \left[-2\bar{m}_2\ell_1 \cdot u_2 + (\ell_1^2 - \ell_1 \cdot q)\right]}{1}$$

$$\frac{1}{\left[-2\bar{m}_1\ell_2 \cdot u_1 + (\ell_2^2 - \ell_2 \cdot q)\right] \left[2\bar{m}_2\ell_2 \cdot u_2 + (\ell_2^2 - \ell_2 \cdot q)\right] \ell_1^2\ell_2^2 \left(\ell_1 + \ell_2 - q\right)^2},$$
(6.21)

the non-planar double box integral reads

$$I_{\rm IX} = \int \frac{d^{\rm D}\ell_1}{i\pi^{D/2}} \frac{d^{\rm D}\ell_2}{i\pi^{D/2}} \frac{1}{\left[-2\bar{m}_1\ell_1 \cdot u_1 + (\ell_1^2 + \ell_1 \cdot q)\right] \left[2\bar{m}_2\ell_1 \cdot u_2 + (\ell_1^2 + \ell_1 \cdot q)\right]} \frac{1}{\left[2\bar{m}_1\ell_2 \cdot u_1 + (\ell_2^2 + \ell_2 \cdot q)\right] \left[2\bar{m}_2(\ell_1 + \ell_2) \cdot u_2 + ((\ell_1 + \ell_2)^2 + (\ell_1 + \ell_2) \cdot q)\right]}$$

$$(6.22)$$

$$\ell_1^2 \ell_2^2 \left(\ell_1 + \ell_2 + q \right)^2$$

and the H diagram reads

$$I_{\rm H} = \int \frac{d^{\rm D}\ell_1}{i\pi^{D/2}} \frac{d^{\rm D}\ell_2}{i\pi^{D/2}} \frac{1}{[-\bar{m}_2\ell_1 \cdot u_2 + (\ell_1^2 - \ell_1 \cdot q)][-2\bar{m}_1\ell_2 \cdot u_1 + (\ell_2^2 - \ell_2 \cdot q)]} \frac{1}{\ell_1^2\ell_2^2(\ell_1 + \ell_2 - q)^2(\ell_1 - q)^2(\ell_2 - q)^2} .$$
(6.23)

The Feynman -i0 prescription is left implicit throughout. The remaining "barred" topologies are obtained from the previous three by crossing symmetry, i.e. replacing u_1 by $-u_1$ (or equivalently u_2 by $-u_2$).

The expansion of these integrals for small q can be simplified by resorting to the method of regions. The relevant, non-analytic contributions emerge from the expansion in the soft region, where $\ell_1 \sim \ell_2 \sim \mathcal{O}(q)$, while $\bar{m}_1 u_1$, $\bar{m}_2 u_2$ are formally held fixed.⁴³ Conveniently, all dependence on the masses factorizes in this step: for instance, to leading order,

$$I_{\rm H} \simeq \frac{1}{\bar{m}_1 \bar{m}_2} \int \frac{d^D \ell_1}{i \pi^{D/2}} \frac{d^D \ell_2}{i \pi^{D/2}} \frac{1}{(-\ell_1 \cdot u_2)(-2\ell_2 \cdot u_1)\ell_1^2 \ell_2^2 (\ell_1 + \ell_2 - q)^2 (\ell_1 - q)^2 (\ell_2 - q)^2} \,. \tag{6.24}$$

The overall dependence on q^2 can be also easily factorized by rescaling the loop momenta. The resulting integrals are thus functions of the single invariant $y = -u_1 \cdot u_2$. Still, such integrals are not elementary for generic ϵ , in contrast with the situation at one loop. Convenient strategies to evaluate them are centered on the method of ordinary differential equations in canonical form in the variable $x = y - \sqrt{y^2 - 1}$. These equations afford simple iterative solutions order by order in ϵ in terms of polylogarithms in the variable x. The boundary conditions for these equations can be determined by performing yet another expansion by regions in the limit $y \rightarrow 1^+$, which luckily does yield elementary integrals for generic ϵ [47].

Performing the calculation according to this strategy, we get the following result:

$$\begin{aligned} \mathcal{A}_{2}(s,q^{2}) &= \frac{(8\pi G)^{3}}{(4\pi)^{4}} \left(\frac{4\pi e^{-\gamma E}}{q^{2}}\right)^{2\epsilon} \left\{ -\frac{2\pi^{2}m_{1}^{4}m_{2}^{4}}{\epsilon^{2}q^{2}} \left[\left(\frac{1+z^{2}}{1-z^{2}}\right)^{2} \left(\frac{1+z^{2}}{z}\right)^{4} \right] \right. \\ &+ \frac{4\pi m_{1}^{3}m_{2}^{3}}{\epsilon^{2}} \frac{(1+z^{2})^{5}}{(1-z^{2})^{4}z^{3}} \left[\pi (1-z^{2})^{2} + i\left(-1-4z^{2}\log z+z^{4}\right) \right] \\ &- \frac{i\pi m_{1}^{2}m_{2}^{2}}{3\epsilon} \left[\left(\frac{1+z^{2}}{1-z^{2}}\right)^{4} \frac{1}{z^{3}} \right] \left[6i\pi (m_{1}^{2}+m_{2}^{2})z(1+z^{2})^{2} \\ &+ 6m_{1}m_{2} \left[4(1+z^{2})^{2}(1-z^{2}) + 2(i\pi+2\log z)(1+z^{2})^{2} \\ &+ \left(\frac{\pi^{2}}{6} - \text{Li}_{2}(z^{2})\right)(z^{2}+1)(z^{4}-6z^{2}+1) - 2(1+z^{2})^{2}(1-z^{2})\log(1-z^{2}) \\ &+ 2(i\pi\log z+\log^{2} z)\left(z^{2}+4z^{4}-z^{6}\right) \left] \right] \right] \right\}, \end{aligned}$$

$$(6.25)$$

where as in (4.7)

$$z = \sigma - \sqrt{\sigma^2 - 1}, \qquad \sigma = \frac{1}{2} \left(z + \frac{1}{z} \right). \tag{6.26}$$

⁴³ The asymptotic expansion of the H topology also has non-analytic contributions that emerge from the mixed soft-hard region, where $\ell_1 \sim O(q)$, $\ell_2 \sim O(m)$. However, these only contribute to order $O(q^{3-2\epsilon})$ to the amplitude.

The above result can be organized in the following convenient form (we recall that $c(\epsilon)$ was defined in (4.9)):

$$\mathcal{A}_{2} = (8\pi G)^{3} \frac{c(\epsilon)^{2}}{(q^{2})^{2\epsilon}} \left[\frac{\mathcal{A}_{2}^{(2,2)}}{\epsilon^{2}q^{2}} + \frac{\mathcal{A}_{2}^{(0,1)}}{q} + \frac{\mathcal{A}_{2}^{(2,0)}}{\epsilon^{2}} + \frac{\mathcal{A}_{2}^{(1,0)}}{\epsilon} + \cdots \right]$$
(6.27)

up to terms that are further suppressed in q^2 , i.e. $\mathcal{O}((q^2)^{\frac{1}{2}-2\epsilon})$ or smaller, or that are subleading in ϵ . Note in particular that the coefficient of $((q^2)^{1+2\epsilon}\epsilon)^{-1}$ vanishes identically. A term of $\mathcal{O}(\epsilon^0(q^2)^{-1-2\epsilon})$ is also needed in order to fully reproduce the $\mathcal{O}(\delta_0^3)$ exponentiation and will not be discussed further. Similarly, $\mathcal{O}((q^2)^{-\frac{1}{2}-2\epsilon})$ terms start contributing to order $\mathcal{O}(\epsilon^0)$ via $\mathcal{A}_2^{(0,1)}$. Using the method described in [47], one can check that $\mathcal{A}_2^{(0,1)}$ is consistent with the iteration term $2\delta_0 2i\delta_1$, to leading order in ϵ . The fact that the $\mathcal{O}((q^2)^{-\frac{1}{2}-2\epsilon})$ contributions indeed reproduce such iteration term for generic ϵ has been checked in Ref. [108].

Explicitly, in terms of the variable z and σ , the above coefficients read

$$\mathcal{A}_{2}^{(2,2)} = -\frac{2\pi^{2}m_{1}^{4}m_{2}^{4}\left(z^{2}+1\right)^{6}}{z^{4}\left(z^{2}-1\right)^{2}} = -\frac{32\pi^{2}m_{1}^{4}m_{2}^{4}\sigma^{6}}{(\sigma^{2}-1)},$$

$$\mathcal{A}_{2}^{(2,0)} = \frac{4\pi^{2}m_{1}^{3}m_{2}^{3}\left(z^{2}+1\right)^{5}}{z^{3}\left(z^{2}-1\right)^{2}} + i\frac{4\pi m_{1}^{3}m_{2}^{3}\left(z^{2}+1\right)^{5}\left(z^{4}-4z^{2}\log\left(z\right)-1\right)}{z^{3}\left(z^{2}-1\right)^{4}}$$

$$(6.28)$$

$$\mathcal{A}_{2}^{(2,0)} = \frac{4\pi^{2}m_{1}^{3}m_{2}^{3}\left(z^{2}+1\right)^{5}}{z^{3}\left(z^{2}-1\right)^{2}} + i\frac{4\pi m_{1}^{3}m_{2}^{3}\left(z^{2}+1\right)^{5}\left(z^{4}-4z^{2}\log\left(z\right)-1\right)}{z^{3}\left(z^{2}-1\right)^{4}}$$

$$= \frac{32\pi^2 m_1^3 m_2^3 \sigma^5}{(\sigma^2 - 1)} - i \, 32\pi m_1^3 m_2^3 \left(\frac{\sigma^6}{(\sigma^2 - 1)^{3/2}} + \frac{\sigma^5}{(\sigma^2 - 1)^2} \log(z) \right), \tag{6.29}$$

and

$$\operatorname{Re} \mathcal{A}_{2}^{(1,0)} = \frac{2\pi^{2}m_{1}^{2}m_{2}^{2}\left(z^{2}+1\right)^{6}\left(\left(m_{1}^{2}+m_{2}^{2}\right)z+2m_{1}m_{2}\right)}{z^{3}\left(z^{2}-1\right)^{4}} \\ - \frac{4\pi^{2}m_{1}^{3}m_{2}^{3}\left(z^{2}+1\right)^{4}\left(z^{4}-4z^{2}-1\right)\log\left(z\right)}{z\left(z^{2}-1\right)^{4}}, \\ = 8\pi^{2}m_{1}^{2}m_{2}^{2}\left[\left(m_{1}^{2}+m_{2}^{2}\right)\frac{\sigma^{6}}{\left(\sigma^{2}-1\right)^{2}}+2m_{1}m_{2}\left(\frac{\sigma^{7}}{\left(\sigma^{2}-1\right)^{2}}+\frac{\sigma^{6}}{\left(\sigma^{2}-1\right)^{3/2}}\right)\right] \\ + 16\pi^{2}m_{1}^{3}m_{2}^{3}\sigma^{4}\left(\frac{1}{\left(\sigma^{2}-1\right)^{1/2}}-\frac{\sigma\left(\sigma^{2}-2\right)}{\left(\sigma^{2}-1\right)^{2}}\right)\log(z), \end{aligned}$$

$$(6.30)$$

together with

$$\begin{split} \operatorname{Im} \mathcal{A}_{2}^{(1,0)} &= \frac{2\pi m_{1}^{3} m_{2}^{3} \left(z^{4} - 6z^{2} + 1\right) \left(z^{2} + 1\right)^{5} \operatorname{Li}_{2} \left(z^{2}\right)}{z^{3} \left(z^{2} - 1\right)^{4}} \\ &- \frac{\pi m_{1}^{3} m_{2}^{3} \left(-24 \left(z^{4} - 1\right) + \pi^{2} \left(z^{4} - 6z^{2} + 1\right)\right) \left(z^{2} + 1\right)^{5}}{3z^{3} \left(z^{2} - 1\right)^{4}} \\ &+ \frac{4\pi m_{1}^{3} m_{2}^{3} \left(z^{4} - 4z^{2} - 1\right) \left(z^{2} + 1\right)^{4} \log^{2} \left(z\right)}{z \left(z^{2} - 1\right)^{4}} - \frac{4\pi m_{1}^{3} m_{2}^{3} \left(z^{2} + 1\right)^{6} \log \left(1 - z^{2}\right)}{z^{3} \left(z^{2} - 1\right)^{3}} \\ &- \frac{8\pi m_{1}^{3} m_{2}^{3} \left(z^{2} + 1\right)^{6} \log \left(z\right)}{z^{3} \left(z^{2} - 1\right)^{4}}, \end{split}$$
(6.31)
$$&= \frac{16\pi m_{1}^{3} m_{2}^{3} \sigma^{5} \left(\sigma^{2} - 2\right) \operatorname{Li}_{2} \left(z^{2}\right)}{\left(\sigma^{2} - 1\right)^{2}} - 64\pi m_{1}^{3} m_{2}^{3} \frac{\sigma^{6}}{\left(\sigma^{2} - 1\right)^{3/2}} - \frac{8}{3} \pi^{3} m_{1}^{3} m_{2}^{3} \frac{\sigma^{5} \left(\sigma^{2} - 2\right)}{\left(\sigma^{2} - 1\right)^{2}} \\ &- 16\pi m_{1}^{3} m_{2}^{3} \sigma^{4} \left(\frac{1}{\left(\sigma^{2} - 1\right)^{1/2}} - \frac{\sigma\left(\sigma^{2} - 2\right)}{\left(\sigma^{2} - 1\right)^{2}}\right) \log^{2} \left(z\right) + \frac{32\pi m_{1}^{3} m_{2}^{3} \sigma^{6} \log \left(1 - z^{2}\right)}{\left(\sigma^{2} - 1\right)^{3/2}} \\ &- 32\pi m_{1}^{3} m_{2}^{3} \sigma^{6} \left(\frac{\sigma}{\left(\sigma^{2} - 1\right)^{2}} + \frac{1}{\left(\sigma^{2} - 1\right)^{3/2}}\right) \log \left(z\right) . \end{split}$$

6.2.1. Eikonal and deflection angle to 3PM order

We have already computed $2\delta_0$ in Section 3.1.4 and $2\Delta_1$ in Section 4.1.1. This information can now be used to verify that the eikonal exponentiation takes place as expected up to two loops and to extract $2\delta_2$ from the previous amplitude through the following relations:

$$\operatorname{Re}(2\delta_2) = \operatorname{Re} \tilde{\mathcal{A}}_2 + \frac{1}{6}(2\delta_0)^3 + 2\delta_0 \operatorname{Im} 2\Delta_1,$$

(6.32)

 $\operatorname{Im}(2\delta_2) = \operatorname{Im} \tilde{\mathcal{A}}_2 - 2\delta_0 \operatorname{Re} 2\Delta_1.$

We obtain the following result, where we highlight in blue the terms associated to radiation-reaction,

$$\operatorname{Re}(2\delta_{2}) = \frac{4m_{1}^{2}m_{2}^{2}G^{3}(1+z^{2})^{4}}{b^{2}z^{2}(1-z^{2})^{5}} \left[(1+z^{2})^{2}(1-z^{2}) + 2z^{2}(1+4z^{2}-z^{4})\log z \right]$$

$$= \frac{16G^{3}m_{1}^{2}m_{2}^{2}}{b^{2}} \left\{ \frac{\sigma^{6}}{(\sigma^{2}-1)^{2}} - \operatorname{arccosh} \sigma \left[\frac{\sigma^{4}}{\sigma^{2}-1} - \frac{\sigma^{5}(\sigma^{2}-2)}{(\sigma^{2}-1)^{\frac{5}{2}}} \right] \right\}$$
(6.33)

and

$$\operatorname{Im} 2\delta_{2} = \frac{4m_{1}^{2}m_{2}^{2}G^{3}}{\pi b^{2}} \frac{(1+z^{2})^{4}(\pi b^{2}e^{\gamma E})^{3\epsilon}}{(1-z^{2})^{5}z^{2}} \left\{ \frac{1}{\epsilon}(1+z^{2})\left[\left(1-6z^{2}+z^{4}\right)\log z-1+z^{4} \right] - \left[2z^{2}\left(1+4z^{2}-z^{4}\right)\log^{2}z+(1+z^{2})^{2}(1-z^{2})\left[2-2\log(1-z^{2})+2\log z\right] + \left(\frac{\pi^{2}}{6}-\operatorname{Li}_{2}(z^{2})\right)(z^{2}+1)(z^{4}-6z^{2}+1) \right] \right\}$$
(6.34)

or equivalently

$$\operatorname{Im} 2\delta_{2} = -\frac{16m_{1}^{2}m_{2}^{2}G^{3}}{\pi b^{2}} \frac{\sigma^{4}(\pi b^{2}e^{\gamma_{E}})^{3\epsilon}}{(\sigma^{2}-1)^{2}} \left\{ \frac{1}{\epsilon} \left(\sigma^{2} + \frac{\sigma(\sigma^{2}-2)}{(\sigma^{2}-1)^{\frac{1}{2}}} \operatorname{arccosh} \sigma \right) - \log(4(\sigma^{2}-1)) \left[\sigma^{2} + \frac{\sigma(\sigma^{2}-2)}{(\sigma^{2}-1)^{\frac{1}{2}}} \operatorname{arccosh} \sigma \right] + (\sigma^{2}-1) \left[1 + \frac{\sigma(\sigma^{2}-2)}{(\sigma^{2}-1)^{\frac{3}{2}}} \right] (\operatorname{arccosh} \sigma)^{2} + \frac{\sigma(\sigma^{2}-2)}{(\sigma^{2}-1)^{\frac{1}{2}}} \operatorname{Li}_{2}(1-z^{2}) + 2\sigma^{2} \right\},$$
(6.35)

where we made the branch-cut singularity for z > 1 more explicit via

$$\operatorname{Li}_{2}(z^{2}) = -\operatorname{Li}_{2}(1-z^{2}) + \frac{\pi^{2}}{6} - \log(z^{2})\log(1-z^{2}).$$
(6.36)

and used

$$-\log z = \operatorname{arccosh} \sigma = \log\left(\sigma + \sqrt{\sigma^2 - 1}\right) = 2\operatorname{arcsinh} \sqrt{\frac{\sigma - 1}{2}}, \qquad (6.37)$$

$$\log(1 - z^2) = \frac{1}{2}\log(4(\sigma^2 - 1)) - \arccos \sigma .$$
(6.38)

An important observation is that, in the ultrarelativistic limit $\sigma \gg 1$ the terms proportional to $(\sigma \log \sigma)^2$ present in the second and third last lines of Eq. (6.35) cancel out against each other, yielding

$$\operatorname{Im} 2\delta_2 \sim \frac{\log(s)}{s} \frac{(8Gs)^3 \Gamma(1-\epsilon)^3}{32(\pi b^2)^{1-3\epsilon}} \left[-\frac{1}{4\epsilon} + \frac{1}{2} + \mathcal{O}(\epsilon) \right].$$
(6.39)

This is crucial in order to ensure agreement between the present calculation and the universal massless result [15] according to the general pattern discussed in [104].

For later convenience, let us also remark that the function defined by

$$f(\zeta) = \frac{\operatorname{arccosh} \zeta}{\sqrt{\zeta^2 - 1}} \tag{6.40}$$

for $\zeta > 1$ can be actually analytically continued to all complex ζ except for a branch cut at $\zeta < -1$. In particular, $f(\zeta)$ is analytic for $-1 < \zeta < 1$ where it takes the form

$$f(\zeta) = \frac{\arccos \zeta}{\sqrt{1 - \zeta^2}} \,. \tag{6.41}$$

More explicitly, the Taylor series for f(1 + z) around z = 0,

$$f(1+z) = \sum_{n=0}^{\infty} \frac{n! (-z)^n}{(2n+1)!!},$$
(6.42)

is absolutely convergent for |z| < 2, and convergent for z = 2, where it yields $f(3) = \frac{1}{\sqrt{2}} \log(1 + \sqrt{2})$, while it diverges for |z| > 2 and for z = -2. These properties lie at the heart of the analytic continuation from unbound to bound trajectories [291-293]. Indeed, $\zeta = \sigma > 1$ for scattering kinematics, such that the center-of-mass energy reads E = -2. $\sqrt{m_1^2 + 2m_1m_2\sigma + m_2^2} > m_1 + m_2$, while $-1 < \zeta < 1$ for bound states, such that $E = \sqrt{m_1^2 + 2m_1m_2\zeta + m_2^2} < m_1 + m_2$. The fact that $f(\zeta)$ has a branch point at $\zeta = -1$ will also provide useful information for the analysis of soft spectra carried out in Section 7.

Let us finally turn to the deflection angle Θ , which follows straightforwardly from 2 δ . Keeping in mind the link between b and b_l derived in Section 6.1, we obtain

$$\Theta = \Theta_{1PM} + \Theta_{2PM} + \Theta_{3PM} \tag{6.43}$$

with, to leading non-vanishing order in ϵ at each PM order,

$$\Theta_{1\text{PM}} = \frac{4Gm_1m_2\sigma^2}{J(\sigma^2 - 1)^{1/2}},$$
(6.44)

$$\Theta_{2PM} = -\frac{8\pi m_1^2 m_2^2 (m_1 + m_2) G^2 \epsilon \sigma^4}{J^2 \sqrt{m_1^2 + m_2^2 + 2m_1 m_2 \sigma} (\sigma^2 - 1)},$$
(6.45)

$$\Theta_{3PM} = -\frac{16m_1^3m_2^3\sigma^6G^3}{3J^3(\sigma^2 - 1)^{3/2}} + \frac{32m_1^4m_2^4G^3}{J^3(m_1^2 + m_2^2 + 2m_1m_2\sigma)} \left\{ \frac{\sigma^6}{\sigma^2 - 1} - \left[\sigma^4 - \frac{\sigma^5(\sigma^2 - 2)}{(\sigma^2 - 1)^{\frac{3}{2}}} \right] \operatorname{arccosh} \sigma \right\}.$$
(6.46)

The 1PM contribution corresponds to a tree diagram where both the graviton and the dilaton are exchanged, the 2PM contribution is absent for $\epsilon = 0$ in agreement with the results of Ref. [211] at one loop. The first term in Θ_{3PM} comes from the expansion of $\tan(\frac{\Theta}{2})$ at small Θ while the remaining terms are genuine new contributions from Re δ_2 . In the ultra-relativistic limit $\sigma \gg 1$ the leading term $\mathcal{O}(\sigma^4 \log \sigma)$ in the second line of (6.46) cancels and only the first

line survives reproducing the universal and finite ultra-relativistic result of [15],

$$\operatorname{Re} 2\delta_2 \sim \frac{4G^3 s^2}{b^2} \,. \tag{6.47}$$

Thanks to analyticity and crossing arguments, as discussed in [104], this cancellation can be seen as a consequence of the one occurring in the imaginary part as highlighted below Eq. (6.38).

It is also instructive to look at the opposite limit $\sigma \to 1$ which is relevant to the PN regime. In this respect, one should note that, as can be better appreciated looking at the second form of Eq. (6.33), the first and last term (highlighted in blue color) contain only even powers of $p \sim \sqrt{\sigma^2 - 1}$, while the second term has only odd powers. This means that the first and last term represent half-integer-PN corrections to the deflection angle, which are a consequence of dissipative processes, i.e. the signature of radiation reaction. Instead, the second term in (6.33) corresponds to the more conventional integer-PN expansion due to the conservative dynamics. More precisely, in the deflection angle (6.46), such dissipative effects appear starting at 1.5PN. This seems in tension with the 2.5PN scaling of radiation reaction effects that is expected from GR calculations. As we shall see, this apparent mismatch is due to the presence of additional massless states propagating in the supersymmetric theory, while the same calculation in GR will lead to 2.5PN radiation-reaction effects, as it should.

6.2.2. Real-analytic, crossing-symmetric reformulation at two loops

We will now extend to two loops the procedure followed in Section 4.1.2 to recast the explicit results of the previous subsection in real-analytic and crossing-symmetric form. We first combine the different contributions in the known form of Eq. (6.25), which suggests the ansatz:

$$\mathcal{A}_{2}(s, q^{2}) = \frac{(8\pi G)^{3}}{(4\pi)^{4}} \left(\frac{4\pi e^{-\gamma_{E}}}{q^{2}}\right)^{2\epsilon} (\sigma^{4} + \bar{\sigma}^{4}) \hat{A}_{2}(s, q^{2}),$$
$$\hat{A}_{2}(s, q^{2}) = \frac{\hat{A}_{2}^{[2]}(s, q^{2})}{\epsilon^{2}} + \frac{\hat{A}_{2}^{[1]}(s, q^{2})}{\epsilon}, \qquad (6.48)$$

where $\bar{\sigma}$ has been introduced in (4.29). In order to reproduce Eqs. (6.28) and (6.29) we find:

$$\hat{A}_{2}^{[2]}(s,q^{2}) = \left[-\frac{8\pi^{2}m_{1}^{4}m_{2}^{4}}{q^{2}} \left(\frac{\sigma^{2}}{\sigma^{2}-1} + \frac{\bar{\sigma}^{2}}{\bar{\sigma}^{2}-1} \right) - 8m_{1}^{3}m_{2}^{3} \left(\frac{\sigma}{(\sigma^{2}-1)^{2}}\log^{2}(-z) + \frac{\bar{\sigma}}{(\bar{\sigma}^{2}-1)^{2}}\log^{2}(-\bar{z}) + 2\left[\frac{\sigma^{2}}{(\sigma^{2}-1)^{3/2}}\log(-z) + \frac{\bar{\sigma}^{2}}{(\bar{\sigma}^{2}-1)^{3/2}}\log(-\bar{z}) \right] \right) \right]$$

$$(6.49)$$

and for Eqs. (6.30) and (6.31):

$$\frac{\hat{A}_{2}^{[1]}(s,q^{2})}{m_{1}^{3}m_{2}^{2}} = 4\pi^{2} \frac{(m_{1}^{2}+m_{2}^{2})}{m_{1}m_{2}} \frac{\sigma^{2}}{(\sigma^{2}-1)^{2}}$$
(6.50a)

$$-32\frac{\sigma^2}{(\sigma^2 - 1)^{3/2}} \left(\log(-z) - \log(z)\right)$$
(6.50b)

$$+4\frac{\sigma(\sigma^{2}-2)}{(\sigma^{2}-1)^{2}}(\log(-z) - \log(z))\left(\text{Li}_{2}(z^{2}) - \text{Li}_{2}\left(\frac{1}{z^{2}}\right)\right)$$
(6.50c)

$$-\frac{8}{3}\frac{1}{(\sigma^2-1)^{1/2}}\left[\left(\log^3(-z)-\log^3(z)\right)+\pi^2\left(\log(-z)-\log(z)\right)\right]$$
(6.50d)

$$-8\frac{\sigma^{3}}{(\sigma^{2}-1)^{2}}\left(\log(-z) - \log(z)\right)\left(\log(-z) + \log(z)\right)$$
(6.50e)

$$+8\frac{\sigma^2}{(\sigma^2-1)^{3/2}}\left(\log(-z) - \log(z)\right)\left[\log(1-z^2) + \log\left(1-\frac{1}{z^2}\right)\right]$$
(6.50f)

The two previous equations have been derived using $\log(-z) = i\pi + \log z$ and the following identities and branch choices:

$$-\operatorname{Li}_{2}\left(\frac{1}{z^{2}}\right) = \operatorname{Li}_{2}(z^{2}) + \frac{\pi^{2}}{6} + \frac{1}{2}\log^{2}(-z^{2}) = \operatorname{Li}_{2}(z^{2}) - \frac{\pi^{2}}{3} + 2\log^{2}z + 2i\pi\log z;$$

$$\log\left(1 - \frac{1}{z^{2}}\right) = \log(1 - z^{2}) - 2\log z - i\pi.$$
(6.51)

Combining these with (6.36) we can also write the combination appearing in (6.50c) in a form:

$$\operatorname{Li}_{2}(z^{2}) - \operatorname{Li}_{2}\left(\frac{1}{z^{2}}\right) = -2\left(\operatorname{Li}_{2}(1-z^{2}) + \log z^{2}\log(1-z^{2}) - \log^{2} z - i\pi \log z\right) , \qquad (6.52)$$

that is useful to discuss the non relativistic (σ , $z \rightarrow 1$) limit.

We can check that the expressions (6.49) and (6.50) satisfy crossing symmetry and real-analyticity. The first property follows simply from the fact that, at this order in q^2 , we can use the identification $\bar{z} = -\frac{1}{z}$.

Checking real analyticity is a bit more subtle. For (6.49) one needs to take into account the subleading (in q^2) term origination from the first line. This (purely real) term exactly cancels a similar term coming from the second line. Then one is left with a purely imaginary term from the second line:

$$-\frac{16m_1^3 m_2^3 \pi \sigma}{(\sigma^2 - 1)^2} i \log z , \qquad (6.53)$$

that can be written in real-analytic crossing-symmetric form as

$$-8m_1^3m_2^3\frac{\sigma}{(\sigma^2-1)^2}\frac{1}{2}(\log(-z)+\log(-\bar{z}))(\log(z^2)-\log(\bar{z}^2)).$$
(6.54)

At this point checking that the amplitude is real in the unphysical region $\sigma^2 < 1$ is straightforward once one realizes that, in that region, $|z|^2 = 1$, in other words *z* is a pure phase and thus $\log z$ is purely imaginary. This implies that the term in Eq. (6.53) is real and the same is true for the last term in (6.49) because of an extra factor *i* coming from $\sqrt{\sigma^2 - 1}$.

Coming now to (6.50), real analyticity is easily checked along the same lines for (6.50a), (6.50b), (6.50d) and (6.50e). Concerning instead (6.50c) and (6.50f) one has to remember that, in the unphysical region, $z^{-1} = \overline{z}$. Since both Li₂(z^2) and log($1 - z^2$) are real-analytic functions, the combinations involving them, appearing in (6.50c) and (6.50f) give a purely imaginary and purely real factor, respectively. Because of the different powers of ($\sigma^2 - 1$) appearing in the two contributions this is exactly as needed for real analyticity. In conclusion, we have shown that Eq. (6.49) and Eqs. (6.50) are real analytic functions. Note that this would not have been the case for the eikonal itself.

6.3. General relativity

In this section we illustrate the calculation of the 3PM eikonal from the $2 \rightarrow 2$ amplitude for the collision of two massive scalars in GR. This calculation was carried out for the first time in full detail in [109]. The underlying principles of this calculation are the same as those presented in the maximally supersymmetric case. One aims to reconstruct the amplitude A_2 to its first few orders in the small-*q* expansion, perform the Fourier transform to *b*-space, obtaining \tilde{A}_2 , and retrieve the classical information from Re $2\delta_2$ by matching with the exponentiation. More explicitly, armed with the *q*-expansion of A_2 or equivalently with the 1/*b*-expansion of \tilde{A}_2 and with the lower-loop order data $2\delta_0$, $2\delta_1$, $2\Delta_1$, one checks that in the equation

$$i\tilde{\mathcal{A}}_{2} = \frac{(2i\delta_{0})^{3}}{3!} + 2i\delta_{0}\,2i\delta_{1} + 2i\delta_{2} + i2\delta_{0}\,2i\Delta_{1}$$
(6.55)

the first two terms on the right hand side cancel out against analogous superclassical terms of the right-hand side, and obtains $2\delta_2$ by subtracting $i2\delta_0 2i\Delta_1$ from the rest. The main novelty is the fact that the gravity integrand takes a much more involved form compared to the neat expression (6.20), where only scalar integrals appeared. Still, one can reconstruct this integrand in a convenient way using generalized unitarity techniques [38,39,109,111,118] and then perform the integration using the same families of master integrals that were needed in the $\mathcal{N} = 8$ case [109].

Instead of following this route, here we shall adopt a different strategy, which combines unitarity, analyticity and crossing symmetry with the eikonal exponentiation itself. The idea is to start from (6.55), which we may rewrite as

$$i\tilde{\mathcal{A}}_{2} = \frac{(2i\delta_{0})^{3}}{3!} + 2i\delta_{0}\,2i\delta_{1} + i[\operatorname{Re} 2\delta_{2} + i\operatorname{Im} 2\delta_{2} + i2\delta_{0}\,2\Delta_{1}]$$
(6.56)

and regard \tilde{A}_2 and Re $2\delta_2$ as the unknown quantity. Indeed the tree-level and one-loop information determines $2\delta_0$, $2\delta_1$, $2\Delta_1$, and Im $2\delta_2$ can be obtained by performing a relatively simpler integral over three-particle phase space, as we have discussed in Section 5. This data determines all imaginary parts arising from the three-particle and two-particle cuts of the amplitude (plus a portion of the real part arising from $2\delta_0 \, \text{Im} \, 2\Delta_1$). We can then make a real-analytic and cross-symmetric ansatz for A_2 , up to the $(q^2)^{-2\epsilon}$ order relevant for the classical limit, complete of its real and imaginary parts. As we shall see, the available data will be enough to fix this ansatz up to a single unknown function $f_a(\sigma)$, which however appears with an extra power of s in front and is thus dominant in the limit of small mass ratio. Therefore, using the results of Appendix B.1, this last term can be fixed too by matching with the probe-limit calculation. This will lead us to the complete result of A_2 appropriate to then retrieve Re $2\delta_2$ and the classical deflection angle.

Let us start by adopting a notation for the small-q and small- ϵ expansion of the amplitude:

$$\mathcal{A}_{2} = \mathcal{A}_{2}^{[2]} + \mathcal{A}_{2}^{[1]} + \mathcal{A}_{2}^{[0]} + \mathcal{O}(q^{1-4\epsilon}) = \left(\frac{4\pi e^{-\gamma_{E}}}{q^{2}}\right)^{2\epsilon} \left[\frac{\mathcal{A}_{2}^{(2,2)}}{\epsilon^{2}q^{2}} + \frac{\mathcal{A}_{2}^{(1,1)}}{\epsilon q} + \frac{\mathcal{A}_{2}^{(2,0)}}{\epsilon^{2}} + \frac{\mathcal{A}_{2}^{(1,0)}}{\epsilon} + \mathcal{O}(q)\right],$$
(6.57)

where $\mathcal{A}_2^{[k]} \sim q^{-k-4\epsilon}$ and in the second line we introduced a notation similar to the one employed for the $\mathcal{N} = 8$ in (6.27) for convenience. As is by now familiar, $\mathcal{A}_2^{[2]}$ and $\mathcal{A}_2^{[1]}$ are simply determined by the iteration of classical terms, i.e. the first two terms on the right-hand side of (6.56). Therefore they must be given by the inverse Fourier transform of $-\frac{1}{3!}(2\delta_0)^3$ (see (3.10)).

$$A_{2}^{(2,2)} = -\frac{64\pi G^{3}m_{1}^{4}m_{2}^{4} \left(\sigma^{2} - \frac{1}{2(1-\epsilon)}\right)^{3} \Gamma(1-\epsilon)^{3} \Gamma(1+2\epsilon)}{\left(\sigma^{2} - 1\right) \Gamma(1-3\epsilon)} e^{2\epsilon\gamma_{E}},$$
(6.58)

and by the inverse Fourier transform of $2\delta_0 2i\delta_1$ (see (4.58)),

$$\mathcal{A}_{2}^{(1,1)} = \frac{6i\pi^{2}G^{3}\left(m_{1}+m_{2}\right)m_{1}^{3}m_{2}^{3}\left(2\sigma^{2}-1\right)\left(1-5\sigma^{2}\right)}{\sqrt{\sigma^{2}-1}} + \mathcal{O}\left(\epsilon\right),$$
(6.59)

where we expanded for simplicity to leading order in ϵ , although both $2\delta_0$ and $2\delta_1$ are of course known in any dimension. The new classical information is contained in $\mathcal{A}_2^{[0]}$, which, comparing with (6.56), must take the following form in b-space

$$\widetilde{\mathcal{A}}_{1}^{[0]} = \operatorname{Re} 2\delta_2 + 2\delta_0 \, 2i\Delta_1 + i \operatorname{Im} 2\delta_2 \,. \tag{6.60}$$

We may rewrite this in q-space as

$$\mathcal{A}_{2}^{[0]} = \mathcal{A}_{2}^{\text{new}} + \mathcal{A}_{2pc} + \mathcal{A}_{3pc} \,. \tag{6.61}$$

Our next task, following the above strategy, is to reconstruct the second and third term from two-particle and threeparticle cuts.

6.3.1. Two-particle and three-particle cuts

For the two-particle cut we can use

$$i\mathcal{A}_{2pc} = 4m_1 m_2 \sqrt{\sigma^2 - 1} \int d^{2-2\epsilon} b \, e^{-ib \cdot q} 2i\delta_0 \, 2i\Delta_1 \,.$$
(6.62)

The leading eikonal is (3.10)

$$2i\delta_0 = \frac{2im_1m_2G\left(\sigma^2 - \frac{1}{2-2\epsilon}\right)\Gamma(-\epsilon)}{\sqrt{\sigma^2 - 1}\left(\pi b^2\right)^{-\epsilon}}$$
(6.63)

and the quantum remainder at one-loop order, $2\Delta_1$, is given by (4.61). At this stage, we can use the inverse Fourier transform (A.49)

$$4m_1m_2\sqrt{\sigma^2 - 1} \int d^{2-2\epsilon} b \; \frac{e^{-ibq}}{(b^2)^{1-3\epsilon}} = \frac{4m_1m_2\sqrt{\sigma^2 - 1}}{(q^2)^{2\epsilon}} \frac{\pi^{1-\epsilon}\Gamma(2\epsilon)}{2^{-4\epsilon}\Gamma(1-3\epsilon)} \tag{6.64}$$

to obtain iA_{2pc} after multiplying $2i\delta_0$ and $2i\Delta_1$. Since $2\delta_0$ is $\mathcal{O}(\epsilon^{-1})$ and in addition (6.64) involves a singular factor $\Gamma(\epsilon)$, the first two nontrivial orders are captured by retaining up to $\mathcal{O}(\epsilon^{-1})$, for which we find

$$\begin{aligned} \mathcal{A}_{2pc} &= \frac{2G^{3}\pi sm_{1}^{2}m_{2}^{2} \left(2\sigma^{2}-1\right)^{3}}{\epsilon (q^{2})^{2\epsilon} \left(\sigma^{2}-1\right)^{2}} \\ &- \frac{i2G^{3}m_{1}^{3}m_{2}^{3} \left(2\sigma^{2}-1\right)}{\epsilon^{2} \left(\frac{q^{2}e^{\gamma_{E}}}{4\pi}\right)^{2\epsilon} \left(\sigma^{2}-1\right)^{\frac{3}{2}}} \left(\frac{2\sigma (2\sigma^{2}-1)(6\sigma^{2}-7) \operatorname{arccosh} \sigma}{\sqrt{\sigma^{2}-1}} - \frac{1-49\sigma^{2}+18\sigma^{4}}{15}\right) \\ &+ \frac{iG^{3}m_{1}^{3}m_{2}^{3}}{\epsilon (q^{2})^{2\epsilon} (\sigma^{2}-1)^{\frac{3}{2}}} \left(\frac{4\sigma \left(2\sigma^{2}-1\right) \left(8\sigma^{4}+2\sigma^{2}-11\right) \operatorname{arccosh} \sigma}{\sqrt{\sigma^{2}-1}} - \frac{18468\sigma^{6}-30728\sigma^{4}+13113\sigma^{2}-1753}{225}\right). \end{aligned}$$
(6.65)

Then, we need to add the imaginary part coming from the three-particle cut. As we know from Section 5, this is related to Im $2\delta_2$ by Fourier transform. In *b*-space, we have (5.56),

$$Im 2\delta_{2} = \frac{2m_{1}^{2}m_{2}^{2}(2\sigma^{2}-1)^{2}G^{3}}{\pi b^{2}(\sigma^{2}-1)^{2}} (\pi b^{2}e^{\gamma_{E}})^{3\epsilon} \\ \times \left[\left(-\frac{1}{\epsilon} + \log(4(\sigma^{2}-1)) \right) \left(\frac{\sigma \left(2\sigma^{2}-3 \right) \operatorname{arccosh} \sigma}{\sqrt{\sigma^{2}-1}} + \frac{8-5\sigma^{2}}{3} \right) \right. \\ \left. - \left(\operatorname{arccosh} \sigma \right)^{2} \left(\frac{\sigma \left(2\sigma^{2}-3 \right)}{\sqrt{\sigma^{2}-1}} + \frac{2(\sigma^{2}-1)(4\sigma^{4}-12\sigma^{2}-3)}{(1-2\sigma^{2})^{2}} \right) \right. \\ \left. + \left(\operatorname{arccosh} \sigma \right) \frac{\sigma \left(88\sigma^{6}-240\sigma^{4}+240\sigma^{2}-97 \right)}{3\left(1-2\sigma^{2} \right)^{2} \sqrt{\sigma^{2}-1}} \right. \\ \left. + \operatorname{Li}_{2}(1-z^{2}) \frac{\sigma \left(3-2\sigma^{2} \right)}{\sqrt{\sigma^{2}-1}} + \frac{-140\sigma^{6}+220\sigma^{4}-127\sigma^{2}+56}{9\left(1-2\sigma^{2} \right)^{2}} \right].$$

$$(6.66)$$

Since $\widetilde{A}_{3pc} = i \operatorname{Im} 2\delta_2$, we can again go to *q*-space using (6.64), finding

$$\begin{aligned} \mathcal{A}_{3pc} &= \frac{4im_1^3 m_2^3 (2\sigma^2 - 1)^2 G^3}{\epsilon (\sigma^2 - 1)^{\frac{3}{2}}} \left(\frac{4\pi e^{-\gamma_E}}{q^2}\right)^{2\epsilon} \\ &\times \left[\left(-\frac{1}{\epsilon} + \log(4(\sigma^2 - 1)) \right) \left(\frac{\sigma \left(2\sigma^2 - 3 \right) \operatorname{arccosh} \sigma}{\sqrt{\sigma^2 - 1}} + \frac{8 - 5\sigma^2}{3} \right) \right. \\ &- \left(\operatorname{arccosh} \sigma \right)^2 \left(\frac{\sigma \left(2\sigma^2 - 3 \right)}{\sqrt{\sigma^2 - 1}} + \frac{2(\sigma^2 - 1)(4\sigma^4 - 12\sigma^2 - 3)}{(1 - 2\sigma^2)^2} \right) \right. \\ &+ \left(\operatorname{arccosh} \sigma \right) \frac{\sigma \left(88\sigma^6 - 240\sigma^4 + 240\sigma^2 - 97 \right)}{3 \left(1 - 2\sigma^2 \right)^2 \sqrt{\sigma^2 - 1}} \\ &+ \operatorname{Li}_2(1 - z^2) \frac{\sigma \left(3 - 2\sigma^2 \right)}{\sqrt{\sigma^2 - 1}} + \frac{-140\sigma^6 + 220\sigma^4 - 127\sigma^2 + 56}{9 \left(1 - 2\sigma^2 \right)^2} \right]. \end{aligned}$$
(6.67)

Comparing the decomposition (6.57), (6.61) with the above explicit formulas for A_{2pc} and A_{3pc} , we discover that $A_2^{(2,0)}$ must be purely imaginary in order for A_2^{new} not to have $\mathcal{O}(\epsilon^{-2})$ real contributions that would result in a divergent eikonal *phase*, and is given by

$$i\mathcal{A}_{2}^{(2,0)} = \frac{2G^{3}m_{1}^{3}m_{2}^{3}\left(2\sigma^{2}-1\right)}{\left(\sigma^{2}-1\right)^{\frac{3}{2}}} \left(\frac{2\sigma(2\sigma^{2}-1)(6\sigma^{2}-7)\operatorname{arccosh}\sigma}{\sqrt{\sigma^{2}-1}} - \frac{1-49\sigma^{2}+18\sigma^{4}}{15}\right) + \frac{4G^{3}m_{1}^{3}m_{2}^{3}(2\sigma^{2}-1)^{2}}{\left(\sigma^{2}-1\right)^{\frac{3}{2}}} \left(\frac{\sigma\left(2\sigma^{2}-3\right)\operatorname{arccosh}\sigma}{\sqrt{\sigma^{2}-1}} + \frac{8-5\sigma^{2}}{3}\right).$$
(6.68)

Of course the imaginary part of $\mathcal{A}_2^{(1,0)}$ is also determined by the cuts, i.e. by the $\mathcal{O}(\epsilon^{-1}(q^2)^{-2\epsilon})$ terms of (6.65), (6.67),

$$\operatorname{Im} \mathcal{A}_{2}^{(1,0)} = \frac{G^{3}m_{1}^{3}m_{2}^{3}}{(\sigma^{2}-1)^{\frac{3}{2}}} \left(\frac{4\sigma \left(2\sigma^{2}-1\right) \left(8\sigma^{4}+2\sigma^{2}-11\right) \operatorname{arccosh} \sigma}{\sqrt{\sigma^{2}-1}} - \frac{18468\sigma^{6}-30728\sigma^{4}+13113\sigma^{2}-1753}{225} \right) + \frac{4m_{1}^{3}m_{2}^{3}(2\sigma^{2}-1)^{2}G^{3}}{(\sigma^{2}-1)^{\frac{3}{2}}} \times \left[\log(4(\sigma^{2}-1)) \left(\frac{\sigma \left(2\sigma^{2}-3\right) \operatorname{arccosh} \sigma}{\sqrt{\sigma^{2}-1}} + \frac{8-5\sigma^{2}}{3} \right) \right) - (\operatorname{arccosh} \sigma)^{2} \left(\frac{\sigma \left(2\sigma^{2}-3\right)}{\sqrt{\sigma^{2}-1}} + \frac{2(\sigma^{2}-1)(4\sigma^{4}-12\sigma^{2}-3)}{(1-2\sigma^{2})^{2}} \right) + (\operatorname{arccosh} \sigma) \frac{\sigma \left(88\sigma^{6}-240\sigma^{4}+240\sigma^{2}-97\right)}{3\left(1-2\sigma^{2}\right)^{2}\sqrt{\sigma^{2}-1}} + \frac{\operatorname{Li}_{2}(1-z^{2})\frac{\sigma \left(3-2\sigma^{2}\right)}{\sqrt{\sigma^{2}-1}} + \frac{-140\sigma^{6}+220\sigma^{4}-127\sigma^{2}+56}{9\left(1-2\sigma^{2}\right)^{2}} \right].$$
(6.69)

For convenience, we may trade the unknown function $\mathcal{A}_2^{\text{new}}$ in (6.61) for an equivalent one $\mathcal{A}_2^{(1,\text{new})}$, obtained by removing the contribution of the 2-particle cut (6.65) to the real part of $\mathcal{A}_2^{(1,0)}$:

$$\operatorname{Re} \mathcal{A}_{2}^{(1,0)} = \frac{2G^{3}\pi sm_{1}^{2}m_{2}^{2}\left(2\sigma^{2}-1\right)^{3}}{\left(\sigma^{2}-1\right)^{2}} + \mathcal{A}_{2}^{(1,\operatorname{new})}.$$
(6.70)

6.3.2. Real-analytic, crossing-symmetric ansatz in GR

In this subsection we will follow a shortcut, based on analyticity and crossing symmetry, in order to reconstruct the full two-loop scattering amplitude (at the relevant classical level) from its imaginary part, the idea being that the latter is easier to compute from lower-loop-order on shell amplitudes. As it turns out, the only missing information is contained in purely real terms that can be easily computed from the probe limit.

In order to pursue this program we start with an ansatz using the analytic structures we have already seen in the N = 8 case discussed in Section 6.2.2. Let us first notice that (6.59) can be written in the analytic crossing symmetric form:

$$\mathcal{A}_{2}^{(1,1)} = \frac{6\pi G^{3} (m_{1} + m_{2}) m_{1}^{3} m_{2}^{3} (2\sigma^{2} - 1) (1 - 5\sigma^{2})}{\sqrt{\sigma^{2} - 1}} \left(\log(-z) + \log(-\bar{z})\right) + \mathcal{O}(\epsilon) .$$
(6.71)

For the other terms we try, in analogy with (4.63), the following ansatz:

$$\mathcal{A}_{2}(s,q^{2}) = \frac{(8\pi G)^{3}}{(4\pi)^{4}} \left(\frac{4\pi e^{-\gamma_{E}}}{q^{2}}\right)^{2\epsilon} \hat{A}_{2}(s,q^{2}),$$

$$\hat{A}_{2}(s,q^{2}) = \frac{\hat{A}_{2}^{[2]}(s,q^{2},\epsilon)}{\epsilon^{2}} + \frac{\hat{A}_{2}^{[1]}(s,q^{2})}{\epsilon}.$$
(6.72)

In order to reproduce Eqs. (6.58) and (6.68) we try the same analytic structures appearing in (6.49). This uniquely determines:

$$\begin{aligned} \hat{A}_{2}^{[2]}(s,q^{2}) &= -\frac{8\pi^{2}m_{1}m_{2}}{q^{2}} \left(\frac{\sigma^{2} - \frac{1}{2(1-\epsilon)}}{\sigma^{2} - 1} + \frac{\bar{\sigma}^{2} - \frac{1}{2(1-\epsilon)}}{\bar{\sigma}^{2} - 1} \right) \left[\left(\sigma^{2} - \frac{1}{2-2\epsilon} \right)^{2} + \left(\bar{\sigma}^{2} - \frac{1}{2-2\epsilon} \right)^{2} \right] c_{\exp}(\epsilon) \\ &+ 4 \left[\left(\sigma^{2} - \frac{1}{2} \right)^{2} + \left(\bar{\sigma}^{2} - \frac{1}{2} \right)^{2} \right] \left(\left(\frac{\sigma(4\sigma^{2} - 5)}{(\sigma^{2} - 1)^{2}} \log^{2}(-z) + \frac{\bar{\sigma}(4\bar{\sigma}^{2} - 5)}{(\bar{\sigma}^{2} - 1)^{2}} \log^{2}(-\bar{z}) \right) \\ &+ \left[\frac{118\sigma^{4} - 259\sigma^{2} + 81}{60(\sigma^{2} - \frac{1}{2})(\sigma^{2} - 1)^{3/2}} \log(-z) + \frac{118\bar{\sigma}^{4} - 259\bar{\sigma}^{2} + 81}{60(\bar{\sigma}^{2} - \frac{1}{2})(\bar{\sigma}^{2} - 1)^{3/2}} \log(-\bar{z}) \right] \right), \end{aligned}$$
(6.73)

with

$$c_{\exp}(\epsilon) = \frac{\Gamma(1-\epsilon)^3 \Gamma(1+2\epsilon)}{\Gamma(1-3\epsilon)} e^{2\epsilon\gamma_E} .$$
(6.74)

Note that we have kept a more complete ϵ dependence in the first line of (6.73). In this way, when we expand that line to lowest order in q^2 (by setting $\bar{\sigma} = -\sigma$), it reproduces, at all ϵ , the superclassical (iteration) term $-\frac{i}{6}(2i\delta_0)^3$. At first order in $\bar{\sigma} + \sigma$ we get corrections which, at lowest order in ϵ , combine with the terms in the second and third line of (6.73) while at $\mathcal{O}(\epsilon)$ they contribute to $\hat{A}_2^{[1]}$.

Concentrating on the former, note that the last line of (6.73) gives a purely imaginary contribution (up to negligible corrections of higher order in q^2). The second line, instead, is proportional to $\log^2(-z) - \log^2(-\bar{z})$ and therefore carries both a real and an imaginary part. Amusingly, the former is exactly canceled by the subleading (and obviously real) contribution from the first line and the final result reproduces exactly the purely imaginary result (6.68).

Turning now to $\hat{A}_{2}^{[1]}$, we have to combine the above-mentioned leftover piece from the first line of (6.73) with Eqs. (6.30) and (6.31) with an appropriate ansatz for the real-analytic, crossing symmetric $\hat{A}_{2}^{[1]}$ itself. Based on the analogy with the $\mathcal{N} = 8$ case we try

$$\frac{\hat{A}_{2}^{[1]}(s,q^{2})}{m_{1}^{3}m_{2}^{3}} = \pi^{2} \frac{(m_{1}^{2} + m_{2}^{2})}{m_{1}m_{2}} \frac{\left[f_{a} + (2\sigma^{2} - 1)^{3}\right]}{(\sigma^{2} - 1)^{2}}$$
(6.75a)

$$+\frac{f_b}{(\sigma^2-1)^{3/2}} \left(\log(-z) - \log(z)\right)$$
(6.75b)

$$+\frac{f_c}{(\sigma^2-1)^2}\left(\log(-z) - \log(z)\right)\left(\operatorname{Li}_2(z^2) - \operatorname{Li}_2\left(\frac{1}{z^2}\right)\right)$$
(6.75c)

$$+\frac{f_d}{(\sigma^2-1)^{1/2}}\left[\left(\log^3(-z)-\log^3(z)\right)+\pi^2\left(\log(-z)-\log(z)\right)\right]$$
(6.75d)

$$+\frac{f_e}{(\sigma^2-1)^2} \left(\log^2(-z) - \log^2(z) \right)$$
(6.75e)

$$+\frac{f_f}{(\sigma^2-1)^{3/2}}\left(\log(-z) - \log(z)\right) \left[\log(1-z^2) + \log\left(1-\frac{1}{z^2}\right)\right]$$
(6.75f)

where the unknown functions f_a, \ldots, f_f are polynomials in σ . The first line of the ansatz (6.75a) is motivated by the analogue equation for the $\mathcal{N} = 8$ case, see (6.50a), but here we allow for an extra contribution, f_a , besides the one from the 2-particle cut in the first line in (6.65).

Matching onto the known terms in (6.69) fixes uniquely all the coefficients f_i except for f_a , which is purely real. We thus obtain the following result consisting of six distinct real-analytic crossing-symmetric structures paralleling exactly the ones we found in $\mathcal{N} = 8$:

$$\frac{\hat{A}_2^{[1]}(s,q^2)}{m_1^3 m_2^3} = \pi^2 \frac{(m_1^2 + m_2^2)}{m_1 m_2} \frac{\left[f_a + (2\sigma^2 - 1)^3\right]}{(\sigma^2 - 1)^2}$$
(6.76a)

$$-\frac{32468\sigma^{6} - 52728\sigma^{4} + 25813\sigma^{2} - 7353}{450(\sigma^{2} - 1)^{3/2}} \left(\log(-z) - \log(z)\right)$$
(6.76b)

$$+\frac{\sigma(2\sigma^2-1)^2(2\sigma^2-3)}{(\sigma^2-1)^2}\left(\log(-z)-\log(z)\right)\left(\text{Li}_2(z^2)-\text{Li}_2\left(\frac{1}{z^2}\right)\right)$$
(6.76c)

$$-\frac{4}{3}\frac{4\sigma^4 - 12\sigma^2 - 3}{(\sigma^2 - 1)^{1/2}} \left[\left(\log^3(-z) - \log^3(z) \right) + \pi^2 \left(\log(-z) - \log(z) \right) \right]$$
(6.76d)

$$-\frac{4}{3}\frac{\sigma(34\sigma^{6}-63\sigma^{4}+42\sigma^{2}-16)}{(\sigma^{2}-1)^{2}}\left(\log^{2}(-z)-\log^{2}(z)\right)$$
(6.76e)

$$+\frac{2}{3}\frac{(2\sigma^2-1)^2(8-5\sigma^2)}{(\sigma^2-1)^{3/2}}\left(\log(-z)-\log(z)\right)\left[\log(1-z^2)+\log\left(1-\frac{1}{z^2}\right)\right].$$
(6.76f)

6.3.3. The 3PM eikonal $2\delta_2$

Substituting (6.76), (6.73) into (6.72), and expanding for small q^2 as instructed by (6.57), we finally obtain

$$\operatorname{Re} \mathcal{A}_{2}^{(1,0)} = \frac{2G^{3}\pi sm_{1}^{2}m_{2}^{2} \left(2\sigma^{2}-1\right)^{3}}{\left(\sigma^{2}-1\right)^{2}} + 2\pi G^{3}m_{1}^{2}m_{2}^{2} \left[\frac{sf_{a}(\sigma)}{(\sigma^{2}-1)^{2}} - \frac{4}{3}m_{1}m_{2}\sigma \left(14\sigma^{2}+25\right)\right. \\ \left. + \frac{4m_{1}m_{2} \left(3+12\sigma^{2}-4\sigma^{4}\right)\operatorname{arccosh}(\sigma)}{\sqrt{\sigma^{2}-1}} + \frac{2m_{1}m_{2} \left(2\sigma^{2}-1\right)^{2}}{\sqrt{\sigma^{2}-1}} \left(\frac{8-5\sigma^{2}}{3 \left(\sigma^{2}-1\right)} + \frac{\sigma \left(2\sigma^{2}-3\right)\operatorname{arccosh}(\sigma)}{\left(\sigma^{2}-1\right)^{3/2}}\right)\right],$$

$$\left. \left. \left. \right. \right]$$

$$\left. \left. \left. \left. \right. \right]$$

$$\left. \left. \right. \right]$$

$$\left. \left. \left. \right. \right]$$

$$\left. \left. \right]$$

$$\left. \left. \left. \right] \right]$$

$$\left. \left. \right]$$

$$\left. \left. \left. \right] \right]$$

$$\left. \left. \right] \right]$$

$$\left. \left. \right] \left. \left. \right] \right] \left. \left. \right] \right] \right]$$

$$\left. \left. \right] \left. \left. \right] \left. \left. \right] \right] \left. \left. \right] \left. \left. \right] \right] \right] \left. \left. \right] \right] \left. \left. \right] \right] \left. \left. \right] \left. \left. \right] \right] \left. \left. \right] \left. \left. \right] \left. \left. \right] \right] \right] \right] \right] \left. \left. \right] \right] \right] \left. \left. \left. \right] \left. \left. \left. \right] \left. \left. \right] \left. \left. \right] \left. \left. \left. \right] \left. \left. \right] \left. \left. \left. \right] \left. \left. \left. \left. \right] \left. \left. \right] \left. \left. \left. \right] \left. \left. \right] \left. \left. \right. \right. \right. \right. \right. \right. \right. \right] \left. \left. \left. \left. \right] \left. \left. \left. \right. \right. \right. \right. \right] \left. \left. \left. \left. \left. \right. \right. \right. \right] \left. \left. \left. \left. \right. \right. \right. \right] \left. \left. \left. \left. \right. \right. \right. \right] \left. \left. \left. \left. \right. \right. \right. \right. \right] \left. \left. \left. \left. \right. \right. \right. \right. \right. \right. \right. \right] \left. \left. \left. \left. \left. \right. \right. \right. \right] \left. \left. \left. \left. \left. \right. \right. \right. \right. \right. \right] \left. \left. \left. \left. \left. \right. \right. \right. \right. \right] \left. \left. \left. \left. \left. \left. \left. \right. \right. \right. \right. \right. \right. \right. \right] \left. \left. \left. \left. \left. \right. \right. \right. \right. \right. \right. \right] \left. \left. \left. \left. \left. \right. \right. \right. \right. \right. \right] \left. \left. \left. \left. \left. \left. \right. \right. \right. \right. \right. \right. \right. \right. \right] \left. \left. \left. \left$$

where the last three lines provide the sought after contribution $\mathcal{A}_{2}^{(1,\text{new})}$ in (6.70).

Following the same steps as for $\mathcal{N} = 8$, we can start from the complete expression (6.57), go to *b*-space, subtract all iteration terms as dictated by (6.55), in particular those due to $2\delta_0 2\Delta_1$, and drop the imaginary part Im $2\delta_2$. However, since (6.77) is already the real part of the $\mathcal{O}(q^{-4\epsilon})$ term, this procedure is equivalent to starting from (6.77) and dropping the first term on the right-hand side. This term is just the real part of (6.65), i.e. the one arising from $2\delta_0 2\Delta_1$. The final step is to restore the appropriate overall factors appearing in (6.57) and go to *b*-space.

In other words Eq. (6.77), without the term in the first line, is the subtracted two-loop amplitude that, when translated in impact parameter space, directly gives the 3PM eikonal. In Section 6.3.4 we will instead show which subtractions are appropriate for obtaining the radial action and we will see that they are different from the ones for obtaining the eikonal (though of course closely related).

The result is the 3PM eikonal phase

. - - -

$$\operatorname{Re} 2\delta_{2}^{(gr)} = \frac{4G^{3}m_{1}^{2}m_{2}^{2}}{b^{2}} \left\{ \frac{\left(2\sigma^{2}-1\right)^{2}\left(8-5\sigma^{2}\right)}{6\left(\sigma^{2}-1\right)^{2}} - \frac{\sigma\left(14\sigma^{2}+25\right)}{3\sqrt{\sigma^{2}-1}} + \frac{sf_{a}(\sigma)}{4m_{1}m_{2}\left(\sigma^{2}-1\right)^{\frac{5}{2}}} + \operatorname{arccosh}\sigma \left[\frac{\sigma\left(2\sigma^{2}-1\right)^{2}\left(2\sigma^{2}-3\right)}{2\left(\sigma^{2}-1\right)^{\frac{5}{2}}} + \frac{-4\sigma^{4}+12\sigma^{2}+3}{\sigma^{2}-1} \right] \right\},$$

$$(6.78)$$

with $f_a(\sigma)$ as in (6.86) below. Eq. (6.78), together with the imaginary part (6.66), completes our discussion of the eikonal exponentiation up to 3PM. While in the approach presented here they all appear on the same footing, the various terms in (6.78) have different physical interpretations that we can now illustrate.

As already alluded to, the term in green in the second line will fixed by probe limit calculation momentarily. The terms in black contain instead genuine 3PM dynamical information, associated to the so-called "potential" interaction between the two objects. They are due to the fact that each body perceives gravitational attraction towards the other one, but this occurs in a fully relativistic manner as dictated by GR. Historically, they were the first 3PM effects to be calculated analytically [38,39], a result that was achieved by using amplitude techniques.

The terms highlighted in blue instead have a different meaning. They are due to the fact that a system of two objects undergoing nontrivial deflections is not a conservative one. The two objects can in general lose energy and angular momentum that can be stored in the gravitational field. The effect of this phenomenon is that the two bodies feel an additional, non-conservative force called radiation-reaction force [294,295]. To 3PM order, this reaction force is captured by the terms in blue and is actually due to a 2PM order loss of angular momentum that the two-body system transmits to the gravitational field [289,296].

The interplay between potential and radiation-reaction terms is crucial in order to ensure that the high-energy limit of (6.78) matches onto the corresponding result for the massless 3PM eikonal [15]. Taking $\sigma \rightarrow \infty$ with

$$s = E^2 = m_1^2 + 2m_1 m_2 \sigma + m_2^2 \sim 2m_1 m_2 \sigma , \qquad (6.79)$$

one finds again (6.47) in precise agreement with Ref. [15]. In particular, in this limit, a cancellation occurring in the square brackets of (6.78) between potential and radiation-reaction terms is crucial in order to ensure that terms of order $\frac{G^3s^2}{b^2} \log \frac{s}{m_1m_2}$ drop out. The universality of the result (6.47) for Re $2\delta_2$ (and of the associated log(*s*) in the Im $2\delta_2$ (6.39)) is due to the fact that, in the ultrarelativistic limit, graviton exchanges dominate the interactions because they couple with the highest power of the energy, both in $\mathcal{N} = 8$ and in GR.

We can now go from the eikonal to the deflection angle using the familiar saddle-point equation

$$2p\sin\frac{\partial}{2} = -\frac{\partial\operatorname{Re}2\delta}{\partial b}.$$
(6.80)

To this order, it is important to include in the left hand side the full deflection angle up to 3PM order, complete of its 1PM and 2PM terms, since the next-to-leading order term in the Taylor expansion gives rise to a contribution of order $(\Theta^{1\text{PM}})^3$. Let us collect here the result for the deflection angle complete up to 3PM:

. .

$$\Theta = \frac{4GE\left(\sigma^{2} - \frac{1}{2}\right)}{b\left(\sigma^{2} - 1\right)} + \frac{3\pi G^{2}E\left(m_{1} + m_{2}\right)\left(5\sigma^{2} - 1\right)}{4\left(\sigma^{2} - 1\right)b^{2}} + \frac{8G^{3}Em_{1}m_{2}}{b^{3}} \left\{ \frac{\left(2\sigma^{2} - 1\right)^{2}\left(8 - 5\sigma^{2}\right)}{6\left(\sigma^{2} - 1\right)^{\frac{5}{2}}} - \frac{\sigma\left(14\sigma^{2} + 25\right)}{3(\sigma^{2} - 1)} + \frac{sf_{a}(\sigma)}{4m_{1}m_{2}\left(\sigma^{2} - 1\right)^{3}} + \operatorname{arccosh}\sigma\left[\frac{\sigma\left(2\sigma^{2} - 1\right)^{2}\left(2\sigma^{2} - 3\right)}{2\left(\sigma^{2} - 1\right)^{3}} + \frac{-4\sigma^{4} + 12\sigma^{2} + 3}{\left(\sigma^{2} - 1\right)^{\frac{3}{2}}}\right] \right\} + \frac{G^{3}s^{\frac{3}{2}}\left(2\sigma^{2} - 1\right)^{3}}{3b^{3}\left(\sigma^{2} - 1\right)^{3}}, \tag{6.81}$$

with $f_a(\sigma)$ as in (6.86) below. To confirm the physical interpretation of the various contributions, it is instructive to take the small-velocity limit, i.e.

$$\sigma = \frac{1}{\sqrt{1 - v^2}}, \qquad v \to 0.$$
(6.82)

Then, the green term in (6.81), which is associated to the probe limit, scales like $G^3 v^{-4}$. In the PN counting where $G \sim v^2$, this corresponds to a 1PN effect. The potential interaction terms, appearing in black in the last two lines of (6.81), are proportional to $G^3 v^{-2}$, so they are 2PN effects. Instead, the radiation-reaction terms in blue scale as $G^3 v^{-1}$ indicating a 2.5PN effect. This half-odd PN order and the odd power of the velocity is indeed the hallmark of a dissipative effect. Comparing with the situation in $\mathcal{N} = 8$, we note that in that case radiation-reaction effects started showing up already to 1.5PN order. This is due to the presence of additional states in the spectrum of classical fields, in particular the ones associated to Kaluza-Klein vectors that couple to the dipole of the system, unlike the graviton, which couples to its quadrupole.

In order to express the final result in terms of the system's angular momentum, it is important to recall that b is not exactly orthogonal to the incoming particle velocities. It is instead related to the orthogonal impact parameter b_l such that $J = p b_l$ by the additional saddle-point equation $b_l = b \cos(\Theta/2)$ as in (6.11). This difference of order G^2 is important in the factor of 1/b appearing in 1PM term of (6.81), and leads to an additional contribution to the 3PM result when expressed in these variables. The deflection angle complete up to 3PM order can be then cast in the form

$$\Theta = \frac{4Gm_1m_2\left(\sigma^2 - \frac{1}{2}\right)}{J\sqrt{\sigma^2 - 1}} + \frac{3\pi G^2m_1^2m_2^2\left(m_1 + m_2\right)\left(5\sigma^2 - 1\right)}{4EJ^2} + \frac{8G^3m_1^4m_2^4}{sJ^3} \left\{ \frac{\left(2\sigma^2 - 1\right)^2\left(8 - 5\sigma^2\right)}{6\left(\sigma^2 - 1\right)} - \frac{\sigma\left(14\sigma^2 + 25\right)}{3(\sigma^2 - 1)^{-\frac{1}{2}}} + \frac{sf_a(\sigma)}{4m_1m_2(\sigma^2 - 1)^{3/2}} + \frac{\arccos n\sigma}{\sqrt{\sigma^2 - 1}} \left[\frac{\sigma\left(2\sigma^2 - 1\right)^2\left(2\sigma^2 - 3\right)}{2\left(\sigma^2 - 1\right)} - 4\sigma^4 + 12\sigma^2 + 3 \right] \right\} - \frac{2G^3m_1^3m_2^3\left(2\sigma^2 - 1\right)^3}{3J^3\left(\sigma^2 - 1\right)^{3/2}}.$$
(6.83)

Finally, in the probe limit $m_1 \ll m_2$, we find the following 3PM contribution⁴⁴

$$\Theta_{3PM,probe} = \frac{2G^3 m_1^3 m_2^3 f_a(\sigma)}{J^3(\sigma^2 - 1)^{3/2}} - \frac{2G^3 m_1^3 m_2^3 \left(2\sigma^2 - 1\right)^3}{3J^3 \left(\sigma^2 - 1\right)^{3/2}}$$
(6.84)

and we can compare this with (B.19) which requires (taking into account the prefactor in (B.13) with $R_s = 2Gm_2$ as in (B.2))

$$\Theta_{3PM,probe} = \left(\frac{2Gm_2\sqrt{E_p^2 - m_p^2}}{J}\right)^3 \frac{-120E_p^4m_p^2 + 60E_p^2m_p^4 + 64E_p^6 - 5m_p^6}{12\left(E_p^2 - m_p^2\right)^3}$$
(6.85)

with $E_p = m_1 \sigma$ and $m_p = m_1$. As promised, equating (6.84) with (6.85) fixes the unknown polynomial to be

$$f_a(\sigma) = 2(12\sigma^4 - 10\sigma^2 + 1)(\sigma^2 - 1).$$
(6.86)

6.3.4. Extracting the radial action from the amplitude

In this subsection we shall go back to the connection (6.9) between the eikonal phase, which is the main focus of the present report, and the phase shift, which is more directly related to the radial action as highlighted by (6.16). As

⁴⁴ The other terms in (6.83) are negligible because they are down by a factor $\frac{m_1m_2}{s} \sim \frac{m_1}{m_2}$.

.....

reviewed in Section 1.3, the main property of the states with well-defined angular momentum is that they diagonalize the 2 \rightarrow 2 elastic S-matrix, and the corresponding diagonal elements define the phase shift (see in particular (1.57) and (1.59)). Correspondingly, such a basis also diagonalizes the sum over intermediate two-particle states (1.66). On the other hand, the *b*-space Fourier transform at the basis of the eikonal formalism does not achieve this diagonalization exactly, as displayed in particular by Eq. (5.13), but only up to terms that are further suppressed by powers of $1/b^2$ or, equivalently, of q^2 .

In order to recover the phase shift from the amplitude one should subtract all super-classical terms by matching to (1.59) and not the *b*-space expression (4.1). Following the notation introduced in (2.44) for the phase shift in the classical limit, let us similarly define the symbol $\mathcal{F}(s, J)$

$$2f_j(s) = \mathcal{F}(s, J) \tag{6.87}$$

for the partial-wave amplitude in the same limit, when $J = \hbar j$ becomes classically sizeable. In this way, suppressing again \hbar from now on, (1.59) becomes

$$1 + i\mathcal{F}(s, J) = e^{i\chi(s,J)}(1 + i\rho(s,J)),$$
(6.88)

where $\rho(s, J)$ denotes the quantum remainder in the *J*-basis. The loop expansion for $\mathcal{F}(s, J)$,

$$\mathcal{F}(s,J) = \mathcal{F}_0(s,J) + \mathcal{F}_1(s,J) + \mathcal{F}_2(s,J) + \cdots, \qquad \mathcal{F}_L(s,J) \sim \mathcal{O}(G^{L+1}),$$
(6.89)

translates into a PM expansion for $\chi(s, J)$,

$$\chi(s,J) = \chi_0(s,J) + \chi_1(s,J) + \chi_2(s,J) + \cdots, \qquad \chi_L(s,J) \sim \mathcal{O}(G^{L+1})$$
(6.90)

and similarly for the quantum remainder $\rho(s, J)$. This in turn dictates the following relations,

$$1\mathcal{F}_0 = 1\chi_0$$
, (6.91)

$$i\mathcal{F}_{1} = \frac{(i\chi_{0})^{2}}{2!} + i\chi_{0} + i\rho_{1},$$

$$i\mathcal{F}_{2} = \frac{(i\chi_{0})^{3}}{3!} + (i\chi_{1})(i\chi_{0}) + [i\chi_{2} + (i\chi_{0})(i\rho_{1})].$$
(6.92)
(6.93)

Despite the resemblance with (4.4), (4.5), (4.6), at loop level these equations dictate slightly different subtractions, which do remove the superclassical terms, but in general can leave behind different classical corrections, as we now turn to illustrate.

As we checked in (2.43) in the massless setup, the leading-order expressions for the eikonal phase and the phase shift match,

$$\chi_0(s,J) = 2\delta_0(b)\Big|_{b \to b_1}.$$
(6.94)

This is not surprising because corrections in the relation (6.11) between *b* and $b_J = J/p$ are suppressed by powers of G^2 in the PM expansion. Similarly we can conclude that

$$\chi_1(s,J) = 2\delta_1(b)\big|_{b \to b_l}.$$
 (6.95)

We had already seen this in (6.18), whose first two terms on the right-hand side are equivalent to (6.94), (6.95), as follows from the definition (6.17) of d_1 , d_2 . In fact, (6.94) and (6.95) can be easily checked by noting that the characterizing property (6.14) is trivially equivalent to (6.11) to this order:

$$2\sin\frac{\Theta^{1\text{PM}} + \Theta^{2\text{PM}}}{2} = \Theta^{1\text{PM}} + \Theta^{2\text{PM}} + \mathcal{O}(G^3)$$
(6.96)

and correspondingly

$$-\frac{1}{p}\frac{\partial(2\delta_1(s,b)+2\delta(s,b))}{\partial b} = -\frac{\partial(2\delta_1(s,b_j)+2\delta(s,b_j))}{\partial J} + \mathcal{O}(G^3).$$
(6.97)

To subleading order in the amplitude, however, we encounter the first novelty. Indeed, repeating the steps in Section 5 that lead to (5.13), but exploiting the exact diagonalization granted by the *J*-projection, we obtain the following consequence of the unitarity relation:

$$2 \ln_{2pc} \mathcal{F}(s, J) = |\mathcal{F}(s, J)|^2 \tag{6.98}$$

without corrections (in contrast with (5.13)). In turn, this ensures that not only that

$$Im \chi_0(s, J) = Im \chi_1(s, J) = 0$$
(6.99)

(consistently with (6.94), (6.95) and with the fact that $2\delta_0$, $2\delta_1$ are real), but also that the quantum remainder is real to this order

$$Im \rho_1(s, J) = 0. (6.100)$$

This reflects at two loop level (6.93) into the following link:

Re
$$\chi_2 = \text{Re } \mathcal{F}_2 + \frac{1}{3!} (\chi_0)^3$$
, (6.101)

where, being purely real, ρ_1 has dropped out. In other words, there is no need to know the one-loop remainder in order to calculate Re χ_2 : one only needs to take into account the cubic contribution involving the leading-order phase shift, χ_0 , in the above equation.

Rather than working in *J*-space, it is more convenient to calculate the subtraction appearing in (6.101) in momentum space, where it translates to the (exact) triple convolution of the tree-level amplitude. Letting the subscript *M* stand for "momentum space", we have

$$(\chi_0)_M^3 = \int \int \bigcup_{p_2}^{p_1} \underbrace{\langle q_1 \rangle}_{q_1} d(\text{LIPS})_2 \underbrace{\langle q_2 \rangle}_{q_2} d(\text{LIPS})_2 \underbrace{\langle q_2 \rangle}_{q_1} d(\text{LIPS})_2 \underbrace{\langle q_1 \rangle}_{q_2} d(\text{LIPS})_2 \underbrace{\langle q_1 \rangle}_{q_2} p_3 \tag{6.102}$$

This expression is easy to calculate to any desired accuracy in q^2 , using the reverse unitarity strategy, since clearly it corresponds to a double cut of the double-box integral $I_{\rm III}$ appearing in the amplitude itself (6.20). Its leading-order contribution must cancel the super-classical terms in the real part of the partial wave, while its subleading contribution will contribute to classical order. More concretely, since all internal massive lines must be cut, one can obtain the result by looking at the *q*-expansion in Eq. (3.22) of [47] and the sought-for subleading contribution is given by (3.24 g) of that reference.

Let us consider, for the sake of generality, a tree-level amplitude of the form

$$\mathcal{A}_0(s, -q^2) = \frac{a_0}{q^2} \,, \tag{6.103}$$

so that

$$\tilde{\mathcal{A}}_0 = 2\delta_0 = \frac{1}{4Ep} \frac{a_0}{4\pi} \frac{\Gamma(-\epsilon)}{(\pi b^2)^{-\epsilon}}, \qquad \Theta^{1\text{PM}} = -\frac{1}{p} \frac{\partial 2\delta_0}{\partial b} = \frac{1}{4Ep^2} \frac{a_0}{2\pi^{1-\epsilon}} \frac{\Gamma(1-\epsilon)}{b^{1-2\epsilon}}.$$
(6.104)

In particular, for the case of minimally interacting massive scalars,

$$a_0 = 32\pi G m_1^2 m_2^2 \left(\sigma^2 - \frac{1}{2-2\epsilon}\right).$$
(6.105)

Then using reverse unitarity as explained above (see Eq. (3.11) of [110]), one obtains

$$(\chi_0)_M^3 = \left[1 - \frac{\epsilon q^2}{3p^2} + \mathcal{O}(q^4)\right] \frac{1}{16E^2 p^2} \frac{a_0^3 \Gamma(-\epsilon)^3}{(4\pi)^{2-2\epsilon} \Gamma(-3\epsilon)} \frac{\Gamma(1+2\epsilon)}{(q^2)^{1+2\epsilon}}.$$
(6.106)

Taking into account the small-*q* expansion of the two-loop amplitude

$$\mathcal{A}_2 = \mathcal{A}_2^{[2]} + \mathcal{A}_2^{[1]} + \mathcal{A}_2^{[0]} + \mathcal{O}(q^{1-4\epsilon})$$
(6.107)

where $\mathcal{A}_2^{[k]} \sim q^{-k-4\epsilon}$ as in (6.57), and going to *b*-space, we find

$$\operatorname{Re}\tilde{\mathcal{A}}_{2}(s,b) + \frac{1}{3!}\operatorname{FT}[(\chi_{0})_{M}^{3}] = \tilde{\mathcal{A}}_{2}^{[2]}(s,b) + \operatorname{Re}\tilde{\mathcal{A}}_{2}^{[0]}(s,b) + \frac{(2\delta_{0})^{3}}{3!} - \frac{(\Theta^{1\mathrm{PM}})^{3}}{3!} bp + \mathcal{O}(b^{-3-3\epsilon}).$$
(6.108)

In this way, the superclassical $\mathcal{O}(b^{3\epsilon})$ terms cancel out between the first term of each line. Note that the subleading corrections in the Fourier transform (A.40) do not play any role, precisely thanks to this cancellation.

The remaining classical terms that are left behind determine the 3PM phase shift according to

$$\operatorname{Re} \chi_{2}(s, J) = \left[\operatorname{Re} \tilde{\mathcal{A}}_{2}^{[0]}(s, b) - \frac{bp}{6} (\Theta^{1\mathrm{PM}})^{3}\right]_{b \to b_{J}}.$$
(6.109)

Here, in the last step, we have used the fact that the Fourier transform to *b*-space and the *J*-projection agree to leading order, and that *after the subtraction* the quantity we are after does not receive any subleading contribution (neither in 1/J nor in *G*).

This subtraction is not identical to the analogous one dictated by the eikonal exponentiation (4.5), (4.6). Indeed, as discussed in Section 5, provided that there are no available inelastic 2-particle channels, the imaginary part of $2\Delta_1$ is also

dictated by the tree-level amplitude (6.103) via

$$2 \operatorname{Im} \mathcal{A}_{1}(s, -q^{2}) = \int \bigvee_{p_{2}} \stackrel{()}{\longleftarrow} \stackrel{()}{\to} \stackrel{()$$

i.e using again reverse unitarity (see Eq. (A.19) of Ref. [110]),

$$2 \operatorname{Im} 2\mathcal{A}_1(s, -q^2) = \left[1 - \frac{\epsilon q^2}{4p^2} + \mathcal{O}(q^4)\right] \frac{1}{4Ep} \frac{a_0^2 \Gamma(-\epsilon)^2}{(4\pi)^{1-\epsilon} \Gamma(-2\epsilon)} \frac{\Gamma(1+\epsilon)}{(q^2)^{1+\epsilon}}.$$
(6.111)

As ensured by (5.13), going to *b*-space the leading term simply reproduces the contribution $(2\delta_0)^2$ dictated by the eikonal exponentiation, while the subleading term determines

$$2\operatorname{Im} 2\Delta_1(s,b) = \frac{2\epsilon}{(4Ep^2)^2} \frac{a_0^2 \Gamma(1-\epsilon)^2}{16\pi^{2-2\epsilon} (b^2)^{1-2\epsilon}}.$$
(6.112)

Therefore, using

$$\operatorname{Re} 2\delta_2(s, b) = \operatorname{Re} \tilde{\mathcal{A}}_2^{[0]}(s, b) + 2\delta_0(s, b) \operatorname{Im} 2\Delta_1(s, b), \qquad (6.113)$$

we obtain

$$\operatorname{Re} 2\delta_2(s,b) = \operatorname{Re} \tilde{\mathcal{A}}_2^{[0]}(s,b) - \frac{bp}{8} (\mathcal{O}^{1\mathrm{PM}})^3.$$
(6.114)

We are now in a position to make a precise comparison between the 3PM eikonal phase and phase shift. Subtracting (6.109) and (6.114),

$$\operatorname{Re} 2\delta_2(s, b)\Big|_{b \to J/p} - \operatorname{Re} \chi_2(s, J) = \frac{J}{24} (\Theta^{1\mathrm{PM}})^3.$$
(6.115)

We remark that Re $2\delta_2(s, b)$, which appears in (6.115), is the $\mathcal{O}(G^3)$ of Re $2\delta_2(s, b)$ regarded as a function of b. This should not be confused with the $\mathcal{O}(G^3)$ of Re $2\delta_2(s, b_J / \cos \frac{\Theta}{2})$, regarded as a function of b_J . As discussed for instance above (6.83), the substitution $b = b_J / \cos \frac{\Theta}{2}$ (6.11) would induce additional $\mathcal{O}(G^3)$ terms, hence the need to pay attention to this distinction. These terms, however, do not cancel out against the right-side of (6.115) so that the phase shift is not simply obtained from the eikonal phase by taking $b = b_J / \cos \frac{\Theta}{2}$ into account,

$$\operatorname{Re} 2\delta\left(s, b_{I}/\cos\frac{\Theta}{2}\right) \neq \operatorname{Re} \chi(s, J).$$
(6.116)

The difference (6.115) is precisely the one needed to reproduce the last term in the first line of (6.18), which we had derived formally from the saddle-point conditions. In turn, this is crucial to ensure that, while the eikonal phase is linked to the angle by (6.11), the phase shift obeys (6.14). Indeed, in the notations of (6.17), (6.18), (6.19),

$$\Theta^{1\text{PM}} = \frac{d_0}{J} p, \qquad \text{Re} \, 2\delta_2 = \frac{d_2 \, p^2}{2J^2}, \qquad \text{Re} \, \chi_2 = \frac{d_2 \, p^2 - \frac{d_0}{12}}{2J^2} \,. \tag{6.117}$$

Hence the difference between eikonal phase and phase shift exactly matches the one derived in (6.115).

To complete the spectrum of possible definitions for the subtractions, let us consider employing the Fourier transform with the complete mass-shell delta functions. We begin by recalling that the standard form of the exponentiation is dictated by (4.1), were the $\tilde{A}(s, b)$ is the Fourier transform with *linearized* delta functions as in (1.71) and (A.44). In contrast, suppose we were to define

$$1 + i \operatorname{FT}[\mathcal{A}](s, b_{l}) = [1 + 2i\check{\Delta}(s, b_{l})]e^{2i\delta(s, b_{l})},$$
(6.118)

with FT[A] as in (1.32) and (A.31). Then, while of course

$$2\check{\delta}_{0}(s, b_{J}) = 2\delta_{0}(s, b)\big|_{b \to b_{J}}, \qquad 2\check{\delta}_{1}(s, b_{J}) = 2\delta_{2}(s, b)\big|_{b \to b_{J}},$$
(6.119)

one finds a nontrivial difference for $2\check{\Delta}_1$ and $2\check{\delta}_2$ to the effect that

$$\operatorname{Re} 2\check{\delta}_{2}(s, b_{j}) = \left[\operatorname{Re} 2\delta_{2}(s, b) + \frac{bp}{8}(\Theta^{1\mathrm{PM}})^{3}\right]_{b \to b_{j}} = \left[\operatorname{Re} \tilde{\mathcal{A}}_{2}^{[0]}(s, b)\right]_{b \to b_{j}}$$
(6.120)

where we used (6.114) in the last step, and therefore

$$\operatorname{Re} 2\delta(s, b_{l}) = \operatorname{Re} 2\delta(s, b \cos \frac{\Theta}{2}) = \operatorname{Re} 2\delta(s, b)$$
(6.121)

up to 3PM order.

Let us conclude this section by presenting the radial action up to 3PM in GR for the scattering of minimally coupled massive scalars in terms of the coefficients defined in (6.17),

$$I(s,J) = -\pi J - d_0 \log J + \frac{d_1 p}{J} + \frac{d_2 p^2 - \frac{d_0^2}{12}}{2J^2},$$
(6.122)

where

$$d_0 = \frac{2Gm_1m_2(2\sigma^2 - 1)}{\sqrt{\sigma^2 - 1}}, \quad d_1 = \frac{3\pi G^2 m_1^2 m_2^2 (m_1 + m_2)(5\sigma^2 - 1)}{4Ep},$$
(6.123)

and

$$d_{2} = 8G^{3}m_{1}^{2}m_{2}^{2}\left\{\frac{(2\sigma^{2}-1)^{2}(8-5\sigma^{2})}{6(\sigma^{2}-1)^{2}} - \frac{\sigma(14\sigma^{2}+25)}{3\sqrt{\sigma^{2}-1}} + \frac{2s(12\sigma^{4}-10\sigma^{2}+1)}{m_{1}m_{2}(\sigma^{2}-1)^{\frac{3}{2}}} + \arccos \sigma \left[\frac{\sigma(2\sigma^{2}-1)^{2}(2\sigma^{2}-3)}{2(\sigma^{2}-1)^{\frac{5}{2}}} + \frac{-4\sigma^{4}+12\sigma^{2}+3}{\sigma^{2}-1}\right]\right\}.$$
(6.124)

6.3.5. Radial action, effective potential, PN limit and bound orbits

In this subsection, we determine the deflection angle and the radial action in terms of a potential present in the Hamiltonian describing the relative motion of two black holes. This serves as a tool to re-sum the PM contributions in a controlled way and access the PN limit that also applies to the case of bound orbits. We conclude by presenting the 2PN-accurate expressions for the periastron advance, revolution period for generic bound orbits and for the binding energy of circular orbits. The idea of matching to an effective potential has its roots in the EOB approach, and here we will follow the particularly simple incarnation given in [297].

We start from the Hamiltonian describing the relative motion in a plane of two black holes:

$$H = p_r^2 + \frac{J^2}{r^2} + V(r), \qquad V(r) = -\sum_{n=1}^{\infty} \frac{G^n}{r^n} f_n, \qquad (6.125)$$

where V(r) is the potential given as an expansion in the Newton constant. From the three Hamilton equations

$$\frac{\partial H}{\partial \theta} = -\dot{p}_{\theta} = 0, \qquad \dot{r} = \frac{\partial H}{\partial p_{r}} = 2p_{r}, \qquad \dot{\theta} = \frac{\partial H}{\partial p_{\theta}} = \frac{2p_{\theta}}{r^{2}}$$
(6.126)

we get that the angular momentum $p_{\theta} = J$ is a constant of motion and the relation

$$\frac{\partial \theta}{\partial r} = \frac{J}{r^2 p_r}, \quad \text{with} \quad p_r = \sqrt{p^2 - \frac{J^2}{r^2} - V(r)}. \tag{6.127}$$

Thus the deflection angle is

$$\Theta = -\pi + 2J \int_{r_*}^{\infty} \frac{dr}{r^2 \sqrt{p^2 - \frac{J^2}{r^2} - V(r)}} = -\pi - 2 \int_{r_*}^{\infty} dr \frac{\partial}{\partial J} \sqrt{p^2 - \frac{J^2}{r^2} - V(r)}, \qquad (6.128)$$

where p^2 is the constant value of the Hamiltonian and, in our case, p is equal to the momentum in the center of mass frame. The factor $-\pi$ is there to ensure that $\Theta = 0$ if the potential vanishes and the factor 2 takes care of the motion from infinity to the point of minimal distance r_* and from r_* back to infinity where r_* is the largest positive root of the condition of turning point, $p_r(r_*) = 0$. As shown in [298] one can avoid to determine it by computing the deflection angle by using instead the following equivalent expression (in Appendix B.1 we review a similar approach to calculate the deflection angle in the probe limit)

$$\Theta = \sum_{k=1}^{\infty} \Theta_k(b_J) \; ; \; \; \Theta_k(b_J) = \frac{2b}{k!} \int_0^\infty du \left(\frac{\partial}{\partial b_J^2}\right)^k \frac{\left(V(\sqrt{u^2 + b_J^2})\right)^k (u^2 + b_J^2)^{k-1}}{p^{2k}} \; , \tag{6.129}$$

- L

where as usual $J = pb_l$. From the deflection angle we can reconstruct the radial action I by using (6.15)

$$I = -\int (\Theta + \pi) \, dJ \tag{6.130}$$

and "the $+\pi$ " is so that $I \rightarrow -\pi J$ when $\Theta \rightarrow 0$; moreover the *J*-independent part of *I* is fixed so as to agree with the 1PM expression for $2\delta_0$ via (6.16) and (6.94). Then, the relation between the radial action and the potential is given by

$$I = -\pi J + 2 \int_{r_*}^{\infty} \sqrt{p^2 - \frac{J^2}{r^2} - V(r)}.$$
(6.131)

It can also be written in a more convenient way as

$$I = -\pi J - \sum_{k=1}^{\infty} \frac{1}{k!} \int_0^\infty du \left(\frac{\partial}{\partial b_j^2}\right)^{k-1} \frac{\left(V(\sqrt{u^2 + b_j^2})\right)^k (u^2 + b_j^2)^{k-1}}{p^{2k-1}}$$
(6.132)

The integral over *u* can be easily computed by changing variable to $u = b_l \sinh w$ and using

$$\int_0^\infty \frac{dw}{\cosh^{d-2} w} = \frac{\sqrt{\pi}\,\Gamma(\frac{d-2}{2})}{2\,\Gamma(\frac{d-2}{2})}\,.$$
(6.133)

In this way we get

$$I(J,\sigma) = -\pi J + \frac{f_1 G_D b_J^{2\epsilon} \Gamma(-\epsilon)}{2p} + \frac{f_2 G^2 \pi}{2J} + \frac{G^3}{J^2} \left(pf_3 + \frac{f_1 f_2}{2p} - \frac{f_1^3}{24p^3} \right) + \mathcal{O}(G^4) , \qquad (6.134)$$

where for the k = 1 contribution we reinstated the dimensional regularization parameter by using $-\frac{G_D f_1}{r^{D-3}}$ in place of $-\frac{G_1}{r}$ in the potential. Then, by using (6.15), we can derive the deflection angle (in D = 4)

$$\Theta = \frac{f_1 G}{pJ} + \frac{f_2 G^2 \pi}{2J^2} + \frac{2G^3}{J^3} \left(pf_3 + \frac{f_1 f_2}{2p} - \frac{f_1^3}{24p^3} \right) + \mathcal{O}(G^4) \,. \tag{6.135}$$

From the knowledge of the PM expanded deflection angle one can of course determine the coefficients f_n . Comparing the equation above with (6.83), we obtain up to 3PM

$$f_1 = \frac{2m_1^2 m_2^2 (2\sigma^2 - 1)}{E} , \quad f_2 = \frac{3m_1^2 m_2^2 (m_1 + m_2)(5\sigma^2 - 1)}{2E}$$
(6.136)

and

$$f_{3} = m_{1}^{2}m_{2}^{2} \left\{ \frac{2E(12\sigma^{4} - 10\sigma^{2} + 1)}{(\sigma^{2} - 1)} - \frac{3(2\sigma^{2} - 1)(5\sigma^{2} - 1)(m_{1} + m_{2})}{2(\sigma^{2} - 1)} + \frac{2m_{1}m_{2}}{E} \left[-\frac{2\sigma(14\sigma^{2} + 25)}{3} - 2\frac{4\sigma^{4} - 12\sigma^{2} - 3}{\sqrt{\sigma^{2} - 1}} \cosh^{-1}(\sigma) + \frac{(2\sigma^{2} - 1)^{2}}{\sqrt{\sigma^{2} - 1}} \left(\frac{8 - 5\sigma^{2}}{3(\sigma^{2} - 1)} + \frac{(2\sigma^{2} - 3)\sigma}{(\sigma^{2} - 1)^{\frac{3}{2}}} \cosh^{-1}(\sigma) \right) \right] \right\}.$$
(6.137)

It is possible also to derive the parameters f_n directly by matching the gravity scattering amplitudes against those of an effective theory encoding the potential (6.125) [37]. In this way one can derive at each PM order an object, $A_{nPM}(\sigma, q)$, that satisfies the following impetus formula [260,291,292,298]:

$$\frac{G^n}{r^n} f_n = \frac{1}{2E} \int \frac{d^3q}{(2\pi)^3} e^{iqr} A_{nPM}(\sigma, q) \,. \tag{6.138}$$

At the tree level A_{1PM} is the one in (3.4). At one loop A_{2PM} is given in (4.47) for $\epsilon = 0$. Finally at two loops,

$$\begin{aligned} A_{3PM}(\sigma, q^2) &= 4G^3 m_1^2 m_2^2 \pi \log q^2 E \\ \times \left[\frac{3(2\sigma^2 - 1)(5\sigma^2 - 1)(m_1 + m_2)}{2(\sigma^2 - 1)} - \frac{2E(12\sigma^4 - 10\sigma^2 + 1)}{(\sigma^2 - 1)} \right. \\ &+ \frac{2m_1 m_2}{E} \left(\frac{2(-3 - 12\sigma^2 + 4\sigma^4)\cosh^{-1}(\sigma)}{\sqrt{\sigma^2 - 1}} + \frac{2\sigma(14\sigma^2 + 25)}{3} \right. \\ &- \frac{(2\sigma^2 - 1)^2}{\sqrt{\sigma^2 - 1}} \left(\frac{8 - 5\sigma^2}{3(\sigma^2 - 1)} + \frac{(2\sigma^2 - 3)\sigma}{(\sigma^2 - 1)^{\frac{3}{2}}}\cosh^{-1}(\sigma) \right) \right) \right] \end{aligned}$$
(6.139)

which coincides with the part proportional to $\log q^2$ of Eq. (8) of [38], with the addition of the two terms coming from radiation reaction. Once the parameters f_n , and so the potential V(r) (6.125), are determined, one can use this information in (6.128) to find an "improved" deflection angle that resums a class of higher order corrections. A detailed analysis discussing also this approach is found in [297,299–302] showing excellent agreement with data from numerical relativity. The idea of resumming is at the basis of the EOB approach [31,32] that can be used to find accurate waveforms for bound systems by using an analytic approach.

A particularly simple case is the one in which the potential contains only f_1 and f_2 , which effectively reduces the calculations to the textbook Coulomb/Newton case. In this case the deflection angle can be computed exactly as a function of G, J and σ . To show this it is convenient to introduce the two dimensionless quantities:

$$\hat{f}_1 = \frac{f_1}{p^2 M}$$
; $\hat{f}_2 = \frac{f_2}{p^2 M^2}$; $M = m_1 + m_2$. (6.140)

Then Eq. (6.128) becomes

$$\Theta(J, E) + \pi = \frac{2J}{p} \int_{r_*}^{\infty} \frac{dr}{r^2 \sqrt{1 + \frac{\hat{f}_1 GM}{r} - \frac{\hat{f}^2}{p^2 r^2}}}$$
(6.141)

where

$$\hat{J}^2 = J^2 - \hat{f}_2 (GMp)^2 \tag{6.142}$$

and r_* is the positive root of the equation:

$$1 + \frac{\hat{f}_1 GM}{r} - \frac{\hat{J}^2}{p^2 r^2} = 0 \tag{6.143}$$

that has the two roots:

$$r_{\pm} = \frac{1}{2} \left(\pm \sqrt{(f_1 G M)^2 + \frac{4\hat{j}^2}{p^2}} - f_1 G M \right)$$
(6.144)

and $r_* = r_+$. Rewriting the quantity in the square root in (6.141) in terms of the two roots and changing the integration variable as

$$\frac{r_{+}}{r} = 1 - az^{2} \quad ; \quad a = \frac{r_{-} - r_{+}}{r_{-}} \tag{6.145}$$

we get

$$\frac{\Theta(J,E) + \pi}{2} = \frac{2J}{p\sqrt{r_+(-r_-)}} \int_0^{\frac{1}{\sqrt{a}}} \frac{dz}{\sqrt{1-z^2}} = \frac{2J}{p\sqrt{r_+(-r_-)}} \arcsin\left(\frac{1}{\sqrt{a}}\right).$$
(6.146)

By using (6.144) we get

$$2\sin^{2}\left[\frac{\Theta(J,E)+\pi}{4}\sqrt{1-\frac{\hat{f}_{2}(GMp)^{2}}{J^{2}}}\right] = 1+\frac{1}{\sqrt{1+\frac{4\hat{J}^{2}}{(p\hat{f}_{1}GM)^{2}}}}$$
(6.147)

which, by using trigonometric identities and recalling that $0 \le \Theta \le \pi$, can be shown to agree with Eq. (7.6) of [291] written below

$$\frac{\Theta(J,E) + \pi}{2} = \frac{1}{\sqrt{1 - \frac{\hat{f}_2 y^2}{\hat{f}_1^2}}} \left(\frac{\pi}{2} + \tan^{-1} \frac{\frac{y}{2}}{\sqrt{1 - \frac{\hat{f}_2 y^2}{\hat{f}_1^2}}} \right) , \tag{6.148}$$

where we introduced $y = \frac{GM\hat{f}_1p}{I}$.

An interesting feature of (6.148) is that its PN expansion captures the 1PN result to all order in the Newton constant *G*. We can see this by first using the results (6.136) for GR to rewrite (6.148) in terms of the parameters j_{PN} and v_{∞} introduced in (3.13) and then by taking the limit $\frac{1}{j_{\text{PN}}} \sim v_{\infty} \ll 1$ up to the first subleading order with $\alpha^{-1} = j_{\text{PN}}v_{\infty}$ fixed. Using, in this limit, the approximation $\frac{y}{2} \simeq \alpha(1 + \frac{2}{\alpha^2 j_{\text{PN}}^2})$ and $\frac{\hat{f}_2 y^2}{\hat{f}_1^2} \simeq \frac{6}{j_{\text{PN}}^2}$ we obtain

$$\frac{\Theta}{2} = \arctan \alpha + \frac{1}{j_{\text{PN}}^2} \left(3 \left(\arctan \alpha + \frac{\pi}{2} \right) + \frac{3\alpha^2 + 2}{\alpha(\alpha^2 + 1)} \right) + \mathcal{O} \left(j_{\text{PN}}^{-4} \right)$$
(6.149)

where we have used $\tan^{-1} \alpha (1+x) = \tan^{-1} \alpha + \frac{\alpha x}{1+\alpha^2} + \cdots$ with $x = \frac{3\alpha^2+2}{\alpha^2 j_{PN}^2}$. The first term above is the 0PN scattering angle obtained in (3.14), while the second term is the 1PN correction at all orders in the PM expansion, see for instance [189]. One can also obtain the result (6.149) from the probe limit calculation, as reviewed in Appendix B.1. Integrating (6.149)

as in (6.130), we have

$$I \simeq \frac{Gm_1m_2}{\alpha v_{\infty}} \left[2\alpha - 2 \arctan \alpha - \alpha \left(\log \left(\alpha^2 + 1 \right) + 2 \log \left(\frac{Gm_1m_2}{\alpha} \right) - 4 \log v_{\infty} \right) + \alpha v_{\infty}^2 \left(-2 \log \left(\alpha^2 + 1 \right) + 3\pi \alpha + 6\alpha \arctan \alpha - 4 \log \left(\frac{Gm_1m_2}{\alpha} \right) + 8 \log v_{\infty} \right) - \pi \right].$$
(6.150)

As discussed in [291–293], it is possible to extract information about bound binary systems starting from the resummed PN results for the scattering process. The basic idea is to analytically continue the unbound observables to the region where the angular momentum is kept fixed but the center-of-mass energy is below $m_1 + m_2$, due to the presence of a nontrivial binding energy. In practice, we need to extend the Lorentz factor from the region $\sigma > 1$ to $\tilde{\sigma} < 1$, which can be done in two ways because of the branch cut $p \sim \sqrt{\sigma^2 - 1}$. The recipe for defining the radial action \tilde{I} for the bound case is to take the linear combination of these two possibilities

$$\tilde{I} = I(\sqrt{\sigma^2 - 1} \to i\sqrt{1 - \tilde{\sigma}^2}, J) + I(\sqrt{\sigma^2 - 1} \to -i\sqrt{1 - \tilde{\sigma}^2}, J) .$$
(6.151)

In the PN limit this analytic continuation is equivalent to sending $v_{\infty} \rightarrow \pm i v_{\infty}$. Applying this recipe to (6.150), at 1PN order we find

$$\tilde{I} = -2\pi J + 2\pi G m_1 m_2 \frac{2\tilde{\sigma}^2 - 1}{\sqrt{1 - \tilde{\sigma}^2}} + 6\pi \frac{G^2 m_1^2 m_2^2}{J} + \mathcal{O}(2\text{PN}) .$$
(6.152)

This can be directly obtained from (6.149) by noting that all its terms are odd under $v_{\infty} \leftrightarrow -v_{\infty}$ (or equivalently $\alpha \rightarrow -\alpha$) except for $3\pi/(2j_{PN}^2)$ and therefore cancel out in (6.151). The *J*-independent term on the right hand side of (6.152), which we can take from the PM expansion (6.134), originates from the analytic continuation of $b_J^{2\epsilon}\Gamma(-\epsilon)$ in the second term of (6.134): since *J* is kept fixed in (6.151), we have to send $b_J \rightarrow \mp ib_J$ as $\sqrt{\sigma^2 - 1} \rightarrow \pm i\sqrt{1 - \tilde{\sigma}^2}$. The term $\mathcal{O}(G^2)$ in (6.152) follows from the only contribution in (6.149) that is even under $v_{\infty} \leftrightarrow -v_{\infty}$. We can rewrite (6.152) in the variables used by [30], introducing also the symmetric mass-ratio $\nu = m_1 m_2/(m_1 + m_2)^2$,

$$\tilde{\sigma} = 1 - \varepsilon_B \left(\frac{1}{2} - \frac{\nu}{8} \varepsilon_B \right), \qquad J = \frac{Gm_1m_2}{\sqrt{\varepsilon_B}} \sqrt{j_B}, \tag{6.153}$$

thus trading *J*, $\tilde{\sigma}$ for j_B , ε_B , and obtain, in the small ε_B limit (at fixed j_B),

$$\tilde{I} = -\pi G m_1 m_2 \left[2 \frac{\sqrt{j_B} - 1}{\sqrt{\varepsilon_B}} - \sqrt{\varepsilon_B} \left(6 \frac{1}{\sqrt{j_B}} + \frac{\nu - 15}{4} \right) \right] + \mathcal{O}(2\text{PN}) .$$
(6.154)

It was pointed out in [291] that the radial action for the bound case vanishes in the case of circular trajectories which is the most relevant one for describing the typical black hole binaries [30]. By using (6.154) we obtain the following relation

$$\tilde{I} = 0$$
, $\Rightarrow j_{B,\text{circ}} = 1 + \varepsilon_B \frac{\nu + 9}{4} + \mathcal{O}(2\text{PN})$, (6.155)

in agreement with [30] and with the closely related derivation given in [303].

Of course, it is also possible to introduce the bound radial action in the PM expansion. Substituting Eq. (6.134) into Eq. (6.151) we obtain at the 2PM approximation

$$\tilde{I} = -2\pi J + 2\pi G m_1 m_2 \frac{2\tilde{\sigma}^2 - 1}{\sqrt{1 - \tilde{\sigma}^2}} + \pi \frac{G^2 m_1^2 m_2^2}{J} (m_1 + m_2) \frac{3(5\tilde{\sigma}^2 - 1)}{2E} + \mathcal{O}(3\text{PM}) .$$
(6.156)

By using the result above in the boundary-to-bound dictionary [291,292], one can derive the periastron advance $K = \frac{\Delta \Theta}{2\pi}$ and the period T_b of the bound motion

$$\Delta \Theta = -\partial_J \tilde{I} , \qquad T_b = \partial_E \tilde{I} = \frac{\partial_{\tilde{\sigma}} \tilde{I}}{\partial_{\tilde{\sigma}} \sqrt{m_1^2 + 2m_1 m_2 \tilde{\sigma} + m_2^2}} . \tag{6.157}$$

It is straightforward to check that, using (6.156) in (6.157) and rewriting the result in terms of the variables (6.153), one obtains Eq. (347b) of [30] (at order $1/j_B$, but at all orders in ε_B) for the periastron advance *K* and Eq. (347a) of the same

reference (at order $1/\sqrt{j_B}$, but at all orders in ε_B) for the angular frequency $n = 2\pi/T_b^{45}$:

$$n = \frac{\varepsilon_B^{3/2} c^3}{G(m_1 + m_2)} \left\{ 1 + \frac{\varepsilon_B}{8} (-15 + \nu) + \frac{\varepsilon_B^2}{128} \left[555 + 30\nu + 11\nu^2 + \frac{192}{j_B^{1/2}} (-5 + 2\nu) \right] \right. \\ \left. + \frac{\varepsilon_B^3}{3072} \left[-29385 - 4995\nu - 315\nu^2 + 135\nu^3 + \frac{5760}{j_B^{1/2}} \left(17 - 9\nu + 2\nu^2 \right) \right. \\ \left. + \frac{16}{j_B^{3/2}} \left(-10080 + \left(13952 - 123\pi^2 \right) \nu - 1440\nu^2 \right) \right] + \mathcal{O} \left(\frac{1}{c^8} \right) \right\},$$

$$K = 1 + \frac{3\varepsilon_B}{j_B} + \frac{\varepsilon_B^2}{4} \left[\frac{3}{j_B} (-5 + 2\nu) + \frac{15}{j_B^2} (7 - 2\nu) \right] \\ \left. + \frac{\varepsilon_B^3}{128} \left[\frac{24}{j_B} \left(5 - 5\nu + 4\nu^2 \right) + \frac{1}{j_B^2} \left(-10080 + \left(13952 - 123\pi^2 \right) \nu - 1440\nu^2 \right) \right. \\ \left. + \frac{5}{j_B^3} \left(7392 + \left(-8000 + 123\pi^2 \right) \nu + 336\nu^2 \right) \right] + \mathcal{O} \left(\frac{1}{c^8} \right).$$

$$(6.158)$$

7. The eikonal operator in the soft limit

The 3PM eikonal for the elastic $2 \rightarrow 2$ scattering presents a few interrelated puzzling features. The most evident one is perhaps the appearance of an infrared-divergent (positive) imaginary part $e^{2i\delta} = e^{-\operatorname{Im} 2\delta}e^{i\operatorname{Re} 2\delta}$ indicating that the probability for this process to take place is in fact infinitely suppressed in the limit $D \rightarrow 4$. The appearance of an imaginary part also makes the eikonal manifestly non-unitary and this suggests a way out. It indicates that we neglected channels that are actually important also in the classical limit. Indeed, by considering a strictly $2 \rightarrow 2$ process, we neglected the fact that the two-body system emits radiation. The aim of this section is to take the first step in order to ameliorate this treatment by including the presence of soft radiation in the final state. To this end, we will promote the eikonal phase to a Hermitian operator [112] depending on the graviton creation/annihilation operators and on the Weinberg soft factor [1,304], taking inspiration from earlier approaches based on the Bloch–Nordsieck method and more recently on dressed states [143,305,306].

As we will see, this will highlight how soft gravitons are responsible for the infrared divergent imaginary part of 2δ . Moreover, another shortcoming of the elastic eikonal framework is that, by its very nature, it does not provide formulas to calculate observables associated to the gravitational field. Introducing soft radiation as well, via the soft eikonal operator, we will also gain access to the properties of gravitational waves at low frequencies: the Zero-Frequency Limit (ZFL) of the energy emission spectrum and the memory effect. In fact, thanks to the exact, nonperturbative nature of the soft theorem, the resulting formulas will even allow us to take a peek *beyond* the conventional PM approximation considered so far. As we shall discuss, this will be crucial in resolving yet another apparent puzzle concerning the high-energy limit. In the naïve large- σ limit the 3PM expression for the ZFL of the spectrum seems ill-behaved. However, this is not the signal of an actual pathological behavior, bur rather of the breakdown of the PM approximation. The exact expression derived thanks to the soft theorem instead provides the correct answer, even in the true ultrarelativistic limit, allowing us to make contact with the massless setup.

A subtle point is related to the effects of static gravitational fields, whose Fourier transform is localized exactly at zero frequency. Clearly these fields do not carry energy–momentum, so their inclusion is not relevant for evaluating the corresponding spectra. Instead, they can in general carry angular momentum and are therefore important for the angular momentum balance of the particles+field system. As shown in [289], and as we will obtain below, their contribution to the angular momentum of the full gravitational field is in fact the leading one in the PM expansion and starts at $\mathcal{O}(G^2)$. In order to accommodate for such effects, we will consider a different dressing of the elastic process, which essentially encodes static gravitational modes via the -i0 prescription.

Being based on the leading graviton theorem, the formulas presented in this section hold for generic deflections and are independent of the PM expansion. Moreover they apply not only to the two-body scattering of spinless, point-like objects, but also to collisions involving spinning or tidally-deformable bodies as well as to multi-body scatterings. This is due to the universality of the soft factor, whose expression only depends on the momenta of the hard particles. Of course, spin and tidal effects for instance do enter the explicit expression of the results, but only insofar as they influence the relation between final and initial momenta, i.e. the impulses.

In the present section, since we are restricting to leading order in the soft approximation, it is sufficient to work by using the momenta of energetic external particles in the elastic process as given, as in [1,304] and more recently in [143,307–310]. Taking into account radiation back-reaction on the trajectories of the massive objects will be instead crucial when considering the full gravitational-wave spectrum in the ensuing section.

⁴⁵ Let us recall that $\varepsilon_B \sim \frac{1}{c^2}$ and $j_B \sim c^0$ as in (344) of [30].

7.1. Soft eikonal operator without static modes

We can include soft radiation by following the method of Bloch–Nordsieck [311,312] and Weinberg [1,304] (see also [305,306,313]). Let us consider the S-matrix element $S_{s.r.,N}^{(M)}$ for the emission of N soft gravitons on top of a background hard process involving particles with momenta p_n , where n runs over the hard states. The subscript "s.r." emphasizes that we are restricting our attention to soft radiation. The total S-matrix element factorizes as the matrix element $S^{(M)}$ (the superscript (M) stands for "momentum space") for the hard process times N universal factors $f_i(k)$ expressed in terms of the polarization *i* of the graviton and of its momentum k, [1,304,311,312]

$$S_{s.r.,N}^{(M)} = \prod_{r=1}^{N} w_{j_r}(k_r) S^{(M)}, \quad w_j(k) = \varepsilon_j^{*\mu\nu}(k) w_{\mu\nu}(k), \quad w^{\mu\nu}(k) = \sum_n \frac{\kappa \, p_n^{\mu} p_n^{\nu}}{p_n \cdot k}, \tag{7.1}$$

where $\kappa = \sqrt{8\pi G}$. Of course an analogous formula holds for soft absorption processes, with $w_i(k)$ replaced by $-w_i^*(k)$. We keep graviton momenta always future-directed. For simplicity, we omit the p_n from the arguments of $w_i(k)$. As emphasized by Weinberg (see e.g. [314]), the formula (7.1) applies to the case in which the "bare" amplitude $S^{(M)}$ is connected, hence in our case to the *iT* part of S = 1 + iT. Extending (7.1) to the disconnected part of the S-matrix will be crucial for the inclusion of static effects and for the calculation of the angular momentum loss given in the next section.

We introduce creation/annihilation operators for the gravitons, obeying canonical commutation relations

$$2\pi\theta(k^{0})\delta(k^{2})[a_{i}(k),a_{j}^{\dagger}(k')] = (2\pi)^{D}\delta^{(D)}(k-k')\delta_{ij}, \qquad (7.2)$$

with *i*, *j* labeling physical polarizations, and we define

$$\int_{k}^{*} = \int \frac{d^{D}k}{(2\pi)^{D}} 2\pi \theta(k^{0})\delta(k^{2})\theta(\omega^{*} - k^{0}).$$
(7.3)

Following Weinberg [1], we have introduced a frequency scale ω^* below which the approximation (7.1) is valid, and in practice this can be taken $\omega^* \sim v/b$ for eikonal scattering. On the other hand we do not need an infrared frequency cutoff thanks to dimensional regularization. We can then capture the factorization (7.1) for soft emissions by defining an exponential operator depending on the oscillators, (the sum over repeated polarization indices is left implicit)

$$e^{2i\hat{\delta}_{\text{S.r.}}} = e^{\int_{k}^{*} \left[w_{j}\left(k\right) a_{j}^{\dagger}\left(k\right) - w_{j}^{*}\left(k\right) a_{j}\left(k\right) \right]},\tag{7.4}$$

and dressing the matrix element for the underlying process according to (here $|0\rangle$ denotes the oscillators' vacuum)

$$S_{s.r.}^{(M)} = e^{2i\hat{\delta}_{s.r.}} \frac{S^{(M)}}{\langle 0|e^{2i\hat{\delta}_{s.r.}}|0\rangle} \,.$$
(7.5)

Then, the matrix elements (7.1) are recovered using the commutation relations (7.2) in

$$S_{s.r.,N}^{(M)} = \langle 0|a_{j_1}(k_1)\cdots a_{j_N}(k_N)S_{s.r.}^{(M)}|0\rangle, \qquad (7.6)$$

and similarly, for absorption processes $\langle 0|S_{s.r.}^{(M)}a_{j_1}^{\dagger}\cdots a_{j_N}^{\dagger}|0\rangle$. Let us now focus on the case in which the background process is the elastic 2 \rightarrow 2 scattering which has been the main object of study in the previous sections of this Report. We consider the Fourier transform to b-space of the two factors in (7.5) separately. By construction the second factor, which describes the elastic process, gives the eikonal. However, thanks to the division by $\langle 0|e^{2i\delta_{s.r.}}|0\rangle$ in (7.5), one needs only the real part of 2δ , as the infrared divergent contribution to the imaginary part is automatically encoded in the new operator part, as we shall see momentarily.⁴⁶ The first factor in (7.5) is instead regular as $Q \to 0$, so we can write it as a differential operator acting on a delta-function $\delta^{D-2}(b)$ trading each Q with a derivative

$$Q^{\mu} \to -i\frac{\partial}{\partial b_{\mu}} \tag{7.7}$$

in the Fourier transform. Of course the product of these two factors in (7.5) becomes a convolution in b space. However, since one factor is just a delta function, the integral of the convolution can be performed straightforwardly, and one obtains

$$S_{s.r.} = e^{\int_{k}^{k} \left[w_{j}(k) a_{j}^{\dagger}(k) - w_{j}^{*}(k) a_{j}(k) \right]} \left[1 + 2i\Delta(b) \right] e^{i\operatorname{Re} 2\delta(b)},$$
(7.8)

where the external momenta p_n in the first line contain derivatives acting on the *b*-dependent functions in the second line.

 $^{^{46}}$ In this section, we focus on the infrared divergent contribution to this imaginary part. The finite contributions also involve non-soft modes and can be reproduced via a similar mechanism which will be discussed in Section 8.

Let us make a few general comments before using (7.8) in some concrete calculations. First, the classical *S*-matrix obtained by neglecting the quantum remainder Δ is explicitly unitary since only the real part of 2δ enters this equation and the inelastic prefactor is the exponential of an anti-Hermitian operator. Second, we obtain the dominant contribution in the classical limit when the derivatives hidden in the external momenta p_n due to (7.7) act on the rapidly oscillating eikonal phase, so that effectively

$$Q^{\mu} \to \frac{\partial \operatorname{Re} 2\delta(b)}{\partial b_{\mu}} \,. \tag{7.9}$$

Since the soft factor $w_j(k)$ becomes proportional to Q in the small-Q limit (see for instance Eq. (2.11) of [106]), we see explicitly how the disconnected term of the elastic S-matrix element drops out under the action of the derivative (7.9) in (7.8). Lastly, let us discuss how the factors involving graviton oscillators in (7.8) can be regarded as soft dressing of initial and final states. To this end, it is sufficient to define

$$w_j^{\text{out/in}}(k) = \varepsilon_{j\mu\nu}^*(k) \sum_{n \in \text{out/in}} \eta_n \frac{\kappa \, p_n^{\mu} p_n^{\nu}}{p_n \cdot k} \tag{7.10}$$

with $\eta_n = +1$ ($\eta_n = -1$) if *n* is a final (initial) state of the background process, and to introduce the dressed states

$$|\operatorname{out/in}\rangle = e^{\int_{k}^{*} \left(w_{j}^{\operatorname{out/in}(k)a_{j}^{\dagger}(k) - w_{j}^{\operatorname{out/in}*(k)a_{j}(k)} \right)} |\Psi_{\operatorname{out/in}}\rangle,$$
(7.11)

where $|\Psi_{out/in}\rangle$ only involve massive (hard) states. In this way, if $|\Psi_{out}\rangle = e^{i \operatorname{Re} 2\delta(b)} |\Psi_{in}\rangle$, then we can rewrite this relation in terms of dressed states as follows,

$$|\text{out}\rangle = e^{\int_{k}^{*} \left(w_{j}^{\text{out}(k)a_{j}^{\mathsf{T}}(k) - w_{j}^{\text{out}*}(k)a_{j}(k)} \right)} e^{\int_{k}^{*} \left(-w_{j}^{\text{in}(k)a_{j}^{\mathsf{T}}(k) + w_{j}^{\text{in}*}(k)a_{j}(k)} \right)} e^{i\operatorname{Re} 2\delta(b)} |\text{in}\rangle$$

$$= e^{\int_{k}^{*} \left((w_{j}^{\text{out}(k) - w_{j}^{\text{in}}(k))a_{j}^{\mathsf{T}}(k) - (w_{j}^{\text{out}(k) - w_{j}^{\text{in}}(k))^{*}a_{j}(k)} \right)} e^{i\operatorname{Re} 2\delta(b)} |\text{in}\rangle , \qquad (7.12)$$

since one can check that the two dressings for initial and final states commute as operators, owing to the reality of the combinations $w_j^{\text{out/in}}(k)$ themselves, and one obtains a total dressed state with $w_j(k) = w_j^{\text{out}}(k) - w_j^{\text{in}}(k)$. Therefore, $|\text{out}\rangle = S_{s.r.}|\text{in}\rangle$ with $S_{s.r.}$ precisely taking the overall dressing factor into account.

We can now apply the eikonal operator to discuss the contribution of low-energy gravitons to observables, including the waveforms, memory, and the particle-energy emission spectrum. The general procedure, given any quantum observable O, is to take its expectation value according to

$$\langle \mathcal{O} \rangle = \langle \Psi_{\rm in} | S_{\rm s,r.}^{\dagger} \mathcal{O} S_{\rm s,r.} | \Psi_{\rm in} \rangle \,. \tag{7.13}$$

Physically, this means to evaluate the mean value of O in the final state of the scattering event, obtained by the action of $S_{s.r.}$. Thus we follow the same strategy as in the KMOC approach [116], but here we take the classical limit as a first step approximating the full *S*-matrix with the eikonal (operator). Let us start from the classical field, which is obtained by inserting in the expectation value (7.13) the free gravitational field [119]

$$H_{\mu\nu}(\mathbf{x}) = \int_{k} \left[\varepsilon_{j\mu\nu}(k) a_{j}(k) e^{i\mathbf{k}\cdot\mathbf{x}} + \varepsilon_{j\mu\nu}^{*}(k) a_{j}^{\dagger}(k) e^{-i\mathbf{k}\cdot\mathbf{x}} \right].$$
(7.14)

This yields

$$h^{\mu\nu}(x) = \langle H^{\mu\nu}(x) \rangle = \int_{k}^{*} \left[w_{\text{TT}}^{\mu\nu}(k) e^{ikx} + w_{\text{TT}}^{*\mu\nu}(k) e^{-ikx} \right],$$
(7.15)

where

$$w_{\rm TT}^{\mu\nu}(k) = \Pi_{\rho\sigma}^{\mu\nu}(k) \, w^{\rho\sigma}(k) \,, \qquad w^{\mu\nu}(k) = \sum_{n} \frac{\kappa \, p_n^{\mu} p_n^{\nu}}{p_n \cdot k} \,, \tag{7.16}$$

and $\Pi_{\rho\sigma}^{\mu\nu}(k)$ is the standard transverse-traceless projector over physical degrees of freedom (5.45). Of course the prediction (7.15) is only accurate for a detector placed at a large distance *r* from the sources. Taking this limit (see Appendix F) at a fixed retarded time *u*, so that $r \gg |u|$, *b*, and moving along the null vector n^{μ} , which characterizes the angular direction, (7.15) yields⁴⁷

$$h^{\mu\nu}(x) \sim \frac{1}{4\pi r} \int_{-\infty}^{+\infty} \frac{d\omega}{2i\pi} w_{\rm TT}^{\mu\nu}(\omega n) e^{-i\omega u}, \qquad (7.17)$$

where we used that $w_{\mu\nu}(k) = -w^*_{\mu\nu}(-k)$. Adjusting the normalization by comparing

$$g_{\mu\nu} = \eta_{\mu\nu} + 2W_{\mu\nu} = \eta_{\mu\nu} + 2\kappa h_{\mu\nu} , \qquad (7.18)$$

⁴⁷ We can send $\omega^* \to \infty$ as long as we focus on the value of the resulting integral for large $|u| \gtrsim b$.

(7.19)

we define the waveform according to

$$W_{\mu\nu} = \kappa h_{\mu\nu}$$
.

Performing the Fourier transform in (7.17) requires in principle to specify how the $1/\omega$ singularity at $\omega = 0$ is circumvented (see (7.65) below) [144,307–310]. However, as stressed in [315,316], the key point is that the behavior of the waveform at large |u| is completely determined by this pole at $\omega = 0$, and possible ambiguities are in fact *u*-independent. Considering the invariant combination

$$\Delta W_{\mu\nu}(n) = W_{\mu\nu}(u > 0, n) - W_{\mu\nu}(u < 0, n), \qquad (7.20)$$

and recalling that $\Pi_{\rho\sigma}^{\mu\nu}(\omega n) = \Pi_{\rho\sigma}^{\mu\nu}(n)$, we thus obtain the memory effect [317]

$$\Delta W_{\mu\nu}(n) = \frac{2G}{r} \Pi^{\mu\nu}_{\rho\sigma}(n) \sum_{a} \frac{p_a^{\rho} p_a^{\sigma}}{(-p_a \cdot n)}, \qquad (7.21)$$

i.e. the leading result of [307–310] or the term indicated as $A_{\mu\nu}$ in [310]. However, this approach does not capture non-linear memory effects [289,318–320].

Let us instead consider the projection of $S_{s.r.}|\Psi_{in}\rangle$ back onto the initial state $|\Psi_{in}\rangle$ with no gravitons. Then one needs to normal order the inelastic exponential through the usual Baker–Campbell–Hausdorff formula $e^{A+B} = e^A e^B e^{-\frac{1}{2}[A,B]}$, so the amplitude for the elastic process is given by

$$\langle \Psi_{\rm in}|S_{s.r.}|\Psi_{\rm in}\rangle = \exp\left[-\frac{1}{2}\int_{k}^{*}w_{\mu\nu}^{*}(k)\Pi^{\mu\nu,\rho\sigma}(k)w_{\rho\sigma}(k)\right]e^{i\operatorname{Re}2\delta(b)},\tag{7.22}$$

where focused on the classical contribution (ignoring Δ). The transversality condition $k^{\mu}w_{\mu\nu} = 0$, which holds for gravity by momentum conservation, grants

$$w_{\mu\nu}^{*}(k)\Pi^{\mu\nu,\rho\sigma}(k)w_{\rho\sigma}(k) = w_{\mu\nu}^{*}(k)\left(\eta^{\mu\rho}\eta^{\nu\sigma} - \frac{1}{D-2}\eta^{\mu\nu}\eta^{\rho\sigma}\right)w_{\rho\sigma}(k) \equiv w^{*}(k)w(k),$$
(7.23)

where we introduced a useful condensed notation according to which explicit index contractions are suppressed, as in (5.44). We will use this repeatedly in the following. Recasting (7.22) as

$$\langle \Psi_{\rm in} | S_{s.r.} | \Psi_{\rm in} \rangle = e^{i \left[\operatorname{Re} 2\delta(b) + \frac{i}{2} \int_k^* w^*(k)w(k) \right]}$$
(7.24)

we see that the damping factor that emerged from the reordering of the exponential factors can be interpreted as an imaginary contribution to the classical eikonal: the infrared-divergent one as $\epsilon = (4 - D)/2 \rightarrow 0$,

$$\operatorname{Im} 2\delta(b) = \frac{1}{2} \int_{k}^{*} w^{*}(k)w(k) + \mathcal{O}(\epsilon^{0}) = \frac{1}{2} \int_{k} \theta(\omega^{*} - k^{0}) w^{*}(k)w(k) + \mathcal{O}(\epsilon^{0}).$$
(7.25)

Note that this integral is scale-invariant in the limit $\epsilon \to 0$, which means that its $1/\epsilon$ term is in fact independent of the cutoff ω^* . We thus see the origin of the infrared-divergent imaginary part we had already encountered in the previous section. It emerges because the elastic amplitude neglects the contributions of soft-graviton emissions [1]. The integral entering (7.25) can be evaluated as follows to leading order in ϵ , retaining also the logarithmic dependence on the cutoff. Introducing velocities analogous to (1.7) for each state according to

$$p_n^{\mu} = \eta_n m_n v_n^{\mu} \tag{7.26}$$

so that $v_n^2 = -1$ and v_n^{μ} is future-directed, we can use the following identity,

$$\int_{k}^{*} \frac{m_{n}m_{m}}{(p_{n}\cdot k)(p_{m}\cdot k)} = \int_{k}^{*} \frac{\eta_{n}\eta_{m}}{(v_{n}\cdot k)(v_{m}\cdot k)} = \left[\frac{(\omega^{*})^{-2\epsilon}}{-2\epsilon}\right] \frac{F_{nm}}{(2\pi)^{2}} + \mathcal{O}(\epsilon^{0})$$
(7.27)

where

$$F_{nm} = \frac{\eta_n \eta_m \operatorname{arccosh} \sigma_{nm}}{\sqrt{\sigma_{nm}^2 - 1}}, \qquad \sigma_{nm} = -v_n \cdot v_m = -\eta_n \eta_m \frac{p_n \cdot p_m}{m_n m_m}.$$
(7.28)

To show (7.27), it is convenient to perform a decomposition of the integrated momentum k^{μ} analogous to (1.10) for each pair *n*, *m*, letting

$$k^{\mu} = \omega_n \,\check{v}^{\mu}_n + \omega_m \,\check{v}^{\mu}_m + k^{\mu}_{\perp} \,, \qquad \check{v}^{\mu}_{n,m} = \frac{\sigma_{nm} v^{\mu}_{m,n} - v^{\mu}_{n,m}}{\sigma^2_{nm} - 1} \,. \tag{7.29}$$

Taking the Jacobian determinant $1/\sqrt{\sigma_{nm}^2 - 1}$ into account (see also Appendix A.3) and focusing on the rest frame of particle *n* where $k^0 = \omega_n$, we obtain,

$$\int_{k}^{*} \frac{1}{(v_{n} \cdot k)(v_{m} \cdot k)} = \frac{1}{\sqrt{\sigma_{nm}^{2} - 1}} \int_{0}^{\omega^{*}} \frac{d\omega_{n}}{\omega_{n}} \int \frac{d\omega_{m}}{\omega_{m}} \int \frac{d^{2-2\epsilon}k_{\perp}}{(2\pi)^{3-2\epsilon}} \delta\left(k_{\perp}^{2} - \frac{-\omega_{n}^{2} + 2\omega_{n}\omega_{m}\sigma_{nm} - \omega_{m}^{2}}{\sigma_{nm}^{2} - 1}\right),$$
(7.30)

where we have used $k \cdot v_{n,m} = -\omega_{n,m}$. Changing integration variables by letting $\omega_n = \omega$, $\omega_m = \omega x$ and $k_{\perp}^{\mu} = \omega x_{\perp}^{\mu}$, performing the integral over x_{\perp}^{μ} and focusing on the leading term as $\epsilon \to 0$, we find

$$\int_{k}^{*} \frac{1}{(v_{n} \cdot k)(v_{m} \cdot k)} = \frac{1}{(2\pi)^{2} \sqrt{\sigma_{nm}^{2} - 1}} \int_{0}^{\omega^{*}} \frac{d\omega}{\omega^{1 + 2\epsilon}} \int \frac{dx}{2x} \,\theta\left(-x^{2} + 2\sigma_{nm}x - 1\right) + \mathcal{O}(\epsilon^{0}) \,. \tag{7.31}$$

Noting that the Heaviside θ function restricts the integration over *x* to lie between the two (positive) roots $\sigma_{nm} \pm \sqrt{\sigma_{nm}^2 - 1}$ finally leads to

$$\int_{k}^{*} \frac{1}{(v_n \cdot k)(v_m \cdot k)} = \left[\frac{(\omega^*)^{-2\epsilon}}{-2\epsilon}\right] \frac{\operatorname{arccosh} \sigma_{nm}}{(2\pi)^2 \sqrt{\sigma_{nm}^2 - 1}} + \mathcal{O}(\epsilon^0)$$
(7.32)

where we used that $\arccos \sigma_{nm} = \pm \log(\sigma_{nm} \pm \sqrt{\sigma_{nm}^2 - 1})$. This shows (7.27), and using this equation in (7.25), one finds

$$\operatorname{Im} 2\delta(b) = \left[\frac{(\omega^*)^{-2\epsilon}}{-2\epsilon}\right] \frac{G}{\pi} \sum_{n,m} m_n m_m \left(\sigma_{nm}^2 - \frac{1}{2}\right) F_{nm} + \mathcal{O}(\epsilon^0).$$
(7.33)

A similar calculation concerns the insertion of the energy-momentum operator

$$P^{\alpha} = \int_{k}^{*} k^{\alpha} a_{j}^{\dagger}(k) a_{j}(k), \qquad \mathbf{P}^{\alpha} = \langle P^{\alpha} \rangle$$
(7.34)

as in Eq. (7.13), which leads to

$$\boldsymbol{P}^{\alpha} = \int_{k}^{*} k^{\alpha} w^{*}(k) w(k), \qquad (7.35)$$

thanks to (7.23). Since we employ the leading soft approximation, we can only resolve the ZFL of the energy emission spectrum,

$$\lim_{\omega \to 0} \frac{dE}{d\omega} \equiv \frac{\partial \mathbf{P}^0}{\partial \omega^*},\tag{7.36}$$

or, using (7.35),

$$\lim_{\omega \to 0} \frac{dE}{d\omega} = \int_{k} \delta(\omega^{*} - k^{0}) k^{0} w^{*}(k) w(k) = \omega^{*} \int_{k} \delta(\omega^{*} - k^{0}) w^{*}(k) w(k)$$
(7.37)

(with \int_k as in (5.37)). Comparing with (7.25), we then see that

$$\lim_{\omega \to 0} \frac{dE}{d\omega} = \lim_{\epsilon \to 0} 2\omega^* \frac{\partial}{\partial \omega^*} \operatorname{Im} 2\delta(b) + \mathcal{O}(\epsilon^0).$$
(7.38)

Using (7.38) and the explicit result (7.33), we see that this derivative cancels the divergence and extracts the coefficient of $\log \omega^*$ in the $\epsilon \to 0$ expansion, leading to

$$\lim_{\omega \to 0} \frac{dL}{d\omega} = \lim_{\epsilon \to 0} \left[-4\epsilon \ln 2\delta(b) \right]$$
(7.39)

or, more explicitly,

Jг

$$\lim_{\omega \to 0} \frac{dE}{d\omega} = \frac{2G}{\pi} \sum_{n,m} m_n m_m \left(\sigma_{nm}^2 - \frac{1}{2}\right) F_{nm} \,. \tag{7.40}$$

As already mentioned, the scope of validity of the formulas (7.33), (7.40) is rather wide, since they retain an exact dependence on the background hard kinematics, they generalize straightforwardly to scatterings involving an arbitrary number of initial and final states, and are also valid if the colliding objects have spin or an internal structure, e.g. if they are subject to tidal deformations. This highlights a general mechanism: the infrared divergences of the elastic amplitude determine the ZFL of the energy emission spectrum via massless quanta [1,106].

Focusing on the 2 \rightarrow 2 case, in order to obtain a more explicit formula it is sufficient to use $\sigma_{nn} = 1$ and $F_{nn} = 1$, while for $n \neq m$ we have $\sigma_{nm} = \sigma_{mn}$ and

$$\sigma_{12} = \sigma_{34} = \sigma , \quad \sigma_{13} = \sigma_{24} = \sigma_{Q} , \quad \sigma_{14} = 1 + \frac{Q^2}{2m_1^2} , \quad \sigma_{23} = 1 + \frac{Q^2}{2m_2^2} , \tag{7.41}$$

where we introduced the shorthand notation

$$\sigma_{\rm Q} = \sigma - \frac{Q^2}{2m_1m_2} = -\frac{u - m_1^2 - m_2^2}{2m_1m_2} \,. \tag{7.42}$$

th (7.25), we then see that

Then, Eq. (7.40) becomes

$$\lim_{\omega \to 0} \frac{dE}{d\omega} = \frac{4G}{\pi} \left\{ 2m_1 m_2 \left(\sigma^2 - \frac{1}{2}\right) \frac{\arccos \sigma}{\sqrt{\sigma^2 - 1}} - 2m_1 m_2 \left(\sigma_Q^2 - \frac{1}{2}\right) \frac{\operatorname{arccosh} \sigma_Q}{\sqrt{\sigma_Q^2 - 1}} \right. \\ \left. + \sum_{i=1,2} \left[\frac{m_i^2}{2} - m_j^2 \left(\left(1 + \frac{Q^2}{2m_i^2}\right)^2 - \frac{1}{2} \right) \frac{\operatorname{arccosh} \left(1 + \frac{Q^2}{2m_i^2}\right)}{\sqrt{\left(1 + \frac{Q^2}{2m_i^2}\right)^2 - 1}} \right] \right\},$$
(7.43)

where, as discussed, the transferred momentum Q should be interpreted by using (7.9), which corresponds to the substitution $Q^{\mu} \rightarrow -\hat{b}^{\mu} 2p \sin \frac{\Theta}{2}$. As already emphasized, while later on we will focus on certain interesting kinematic limits, the dependence of this formula on the dynamics of the background elastic process, and in particular on Q/m_i , is exact.

7.2. Eikonal operator including static modes

Let us now turn to the contribution of low-frequency gravitons to the waveform in position space, in particular its value at early retarded times, and to the angular momentum [289,321]. As we shall discuss, both quantities are sensitive to static-field effects, and therefore to how one approach the $\sim 1/\omega$ singularity at $\omega = 0$ in the radiation spectrum.

As we emphasized, the soft eikonal operator (7.8) is based on the standard Weinberg soft theorem, which includes soft gravitons with low but nonzero frequency. As such, it does not include effects that arise due to exactly static fields, whose Fourier transform is localized at zero frequency. To include them it is sufficient to replace the standard soft factor $w_i(k)$ in (7.8) by⁴⁸

$$f_{j}(k) = \varepsilon_{j\mu\nu}(k)^{*}F^{\mu\nu}(k), \qquad F^{\mu\nu}(k) = \sum_{n} \frac{\sqrt{8\pi G} p_{n}^{\mu} p_{n}^{\nu}}{p_{n} \cdot k - i0}.$$
(7.44)

and to consider the following operator (neglecting the quantum remainder for brevity)

$$S_{s.r.} = e^{\int_{k}^{*} \left[f_{j}(k) a_{j}^{\dagger}(k) - f_{j}^{*}(k) a_{j}(k) \right]} e^{2i\tilde{\delta}(b)},$$
(7.45)

where the definition of $2\tilde{\delta}$ will be specified momentarily. By including the -i0 prescription in (7.44) even for real emissions of gravitons, we are now dressing the full *S*-matrix, including the identity term, and thus include possible "emissions" localized at $\omega = 0$ from disconnected pieces of the hard matrix element. This construction is not a standard application of Weinberg's theorem, which only holds for *connected* amplitudes [314,322], and in this way it also captures static effects. In fact, the same static term in the asymptotic field also follows from the worldline approach in which one solves for the particle trajectories and the field using retarded propagators [129,134].

To see how this modification reflects the definition of dressed states, compared to the one discussed in the previous subsection, let us now consider

$$f_j^{\text{out/in}}(k) = \varepsilon_{j\mu\nu}^*(k) \sum_{n \in \text{out/in}} \eta_n \frac{\sqrt{8\pi G p_n^{\mu} p_n^{\nu}}}{p_n \cdot k - i0}$$
(7.46)

and

$$|\text{OUT/IN}\rangle = e^{\int_{k}^{*} \left(f_{j}^{\text{out/in}}(k) a_{j}^{\dagger}(k) - f_{j}^{\text{out/in}*}(k) a_{j}(k) \right)} |\Psi_{\text{out/in}}\rangle$$
(7.47)

(the notation $|OUT/IN\rangle$ is meant to distinguish these states from $|Out/in\rangle$ in (7.10)). In this way, if we start again from $|\Psi_{out}\rangle = e^{i \operatorname{Re} 2\delta(b)} |\Psi_{in}\rangle$ and we rewrite it in terms of dressed states, we get

$$|\text{OUT}\rangle = e^{\int_{k}^{*} \left(\int_{j}^{\text{out}(k)a_{j}^{\dagger}(k) - \int_{j}^{\text{out}*}(k)a_{j}(k) \right)} e^{-\int_{k}^{*} \left(\int_{j}^{\text{in}(k)a_{j}^{\dagger}(k) - \int_{j}^{\text{in}*}(k)a_{j}(k) \right)} e^{i\operatorname{Re} 2\delta(b)} |\text{IN}\rangle .$$
(7.48)

In this new setup, the two dressings for initial and final states no longer commute, and using the Baker–Campbell–Hausdorff formula $e^A e^B = e^{A+B} e^{+\frac{1}{2}[A,B]}$ one obtains

$$|\text{OUT}\rangle = e^{\int_{k}^{*} \left(f_{j}(k) a_{j}^{\dagger}(k) - f_{j}^{*}(k) a_{j}(k) \right)} e^{\frac{1}{2} \int_{k}^{*} \left(f_{j}^{\text{out}*}(k) f_{j}^{\text{in}}(k) - f_{j}^{\text{out}*}(k) f_{j}^{\text{in}*}(k) \right) + i \operatorname{Re} 2\delta(b)} |\text{IN}\rangle ,$$
(7.49)

where

$$f_j(k) = f_j^{\text{out}}(k) - f_j^{\text{in}}(k)$$
 (7.50)

⁴⁸ Let us note that the symmetric Lorentz tensor $F^{\mu\nu}$ in (7.44) should not be confused with the symmetric coefficients F_{mn} introduced in (7.28) earlier on.

and therefore, comparing with (7.45), we see that $|OUT\rangle = S_{s,r}|N\rangle$ provided the phase takes the value

$$2i\tilde{\delta}(b) = i\operatorname{Re} 2\delta(b) - 2i\delta^{\operatorname{dr.}}(b), \qquad 2i\delta^{\operatorname{dr.}}(b) = -\frac{1}{2}\int_{k}^{*} (f_{j}^{\operatorname{out}*}(k)f_{j}^{\operatorname{in}}(k) - f_{j}^{\operatorname{out}}(k)f_{j}^{\operatorname{in}*}(k)).$$
(7.51)

The last equation, whose right-hand side is manifestly imaginary, identifies a contribution to the phase due to the dressing (hence the superscript "dr."). The integral that we need to calculate in order to determine the resulting phase correction reads

$$2i\delta^{\mathrm{dr.}}(b) = \frac{1}{2} \sum_{\substack{n \in \mathrm{out}\\m \in \mathrm{in}}} \int_{k}^{*} \left[\frac{8\pi G m_{n}^{2} m_{m}^{2} (\sigma_{nm}^{2} - \frac{1}{2})}{(p_{n} \cdot k + i0)(p_{m} \cdot k - i0)} - \frac{8\pi G m_{n}^{2} m_{m}^{2} (\sigma_{nm}^{2} - \frac{1}{2})}{(p_{n} \cdot k - i0)(p_{m} \cdot k + i0)} \right].$$
(7.52)

Sending $k^{\mu} \rightarrow -k^{\mu}$ in the second term, we see that this expression recombines as follows,

$$2i\delta^{\text{dr.}}(b) = \frac{1}{2} \sum_{\substack{n \in \text{out} \\ m \in \text{in}}} 8\pi G m_n m_m (\sigma_{nm}^2 - \frac{1}{2}) I_{nm}$$
(7.53)

where, introducing the velocities v_n^{μ} according to (7.26)

$$I_{nm} = \int \frac{d^{4-2\epsilon}k}{(2\pi)^{4-2\epsilon}} \frac{2\pi \operatorname{sgn}(k^0)\delta(k^2)\theta(\omega^* - |k^0|)}{(\eta_n v_n \cdot k + i0)(\eta_m v_m \cdot k - i0)} \,.$$
(7.54)

Here $sgn(k^0) = \theta(k^0) - \theta(-k^0)$ takes the value +1 (resp. -1) if $k^0 > 0$ ($k^0 < 0$). The integrals I_{nm} can be evaluated in a manner similar to (7.27), as we turn to illustrate. Sending $\epsilon = \frac{4-D}{2} \rightarrow 0$, which as we shall see leads to a finite leading-order contribution, and focusing on the rest frame of particle *n* where $k^0 = \omega_n = -v_n \cdot k$, we can decompose the integrated momentum according to (7.29) to arrive at

$$I_{nm} = \frac{1}{\sqrt{\sigma_{nm}^2 - 1}} \int_{-\omega^*}^{+\omega^*} \frac{\operatorname{sgn}(\omega_n) \, d\omega_n}{(-\eta_n \omega_n + i0)} \int \frac{d\omega_m dk_\perp^2}{(-\eta_m \omega_m - i0)^2 (2\pi)^2} \delta\left(k_\perp^2 - \frac{-\omega_n^2 + 2\omega_n \omega_m \sigma_{nm} - \omega_m^2}{\sigma_{nm}^2 - 1}\right)$$
(7.55)

after taking the Jacobian determinant $1/\sqrt{\sigma_{nm}^2 - 1}$ into account. Changing integration variables by letting $\omega_n = \omega$, $\omega_m = \omega x$ and $k_{\perp}^2 = \omega^2 x_{\perp}^2$, so that

$$\operatorname{sgn}(\omega_n) \, d\omega_n \, d\omega_m \, dk_\perp^2 = \omega^3 \, d\omega \, dx \, dx_\perp^2 \,, \tag{7.56}$$

and performing the integral over x_{\perp}^2 by means of the delta function,

$$I_{nm} = \frac{1}{(2\pi)^2 \sqrt{\sigma_{nm}^2 - 1}} \int_{-\omega^*}^{+\omega^*} \frac{d\omega}{(-\eta_n \omega + i0)} \int \frac{\omega \, dx}{2(-\eta_m \omega x - i0)} \theta \left(-x^2 + 2\sigma_{nm} x - 1\right).$$
(7.57)

The Heaviside θ function restricts the integration over x to lie between the two positive roots $\sigma_{nm} \pm \sqrt{\sigma_{nm}^2 - 1}$, so that the integrals over ω and x factorize, leading to

$$I_{nm} = \frac{\operatorname{arccosh}\sigma_{nm}}{(2\pi)^2 \sqrt{\sigma_{nm}^2 - 1}} \int_{-\omega^*}^{\omega^*} \frac{\omega \, d\omega}{(-\eta_n \omega + i0)(-\eta_m \omega - i0)} \,.$$
(7.58)

The new ingredient that we need to evaluate is thus the integral over "positive and negative frequencies"

$$\int_{-\omega^*}^{+\omega^*} \frac{\omega \, d\omega}{(-\eta_n \omega + i0)(-\eta_m \omega - i0)} = -\frac{i\pi}{2} (\eta_n - \eta_m) \,. \tag{7.59}$$

To see this, let us make the *i*0 prescription more manifest by introducing a small $\lambda > 0$. Clearly the integral vanishes if $\eta_n = \eta_m$ because $\int_{-\omega^*}^{\omega^*} \omega d\omega / (\omega^2 + \lambda^2)$ is zero by parity. If $\eta_n = -\eta_m$, the integral reduces instead to

$$-\int_{-\omega^*}^{+\omega^*} \frac{\omega \, d\omega}{(\omega - i\lambda\eta_n)^2} = -\left[\log(\omega - i\lambda\eta_n) - \frac{i\lambda\eta_n}{\omega - i\lambda\eta_n}\right]_{-\omega^*}^{+\omega^*}$$

$$= -i\pi\eta_n + \frac{2i\eta_n\lambda\omega^*}{(\omega^*) + \lambda^2} \xrightarrow[\lambda \to 0^+]{} -i\pi\eta_n, \qquad (7.60)$$

which shows (7.59), and this leads to the final expression for I_{nm} ,

$$I_{nm} = -\frac{i\pi}{2}(\eta_n - \eta_m) \frac{\arccos h \sigma_{nm}}{(2\pi)^2 \sqrt{\sigma_{nm}^2 - 1}} \,.$$
(7.61)

Note that the result no longer depends on the cutoff ω^* , so we may now send it to zero. What we obtain is thus a contribution localized at the zero-frequency end of the spectrum, i.e. an effect intrinsically due to field configurations
that are static in time domain. Substituting (7.61) into (7.53), we obtain

$$2i\delta^{\text{dr.}}(b) = -iG \sum_{\substack{n \in \text{out} \\ m \in \text{in}}} m_n m_m (\sigma_{nm}^2 - \frac{1}{2}) \frac{\operatorname{arccosh} \sigma_{nm}}{\sqrt{\sigma_{nm}^2 - 1}}$$
(7.62)

so that, using (7.41) and expanding for small deflections $Q = Q_{1PM} + O(G^2)$,

$$2i\delta^{\rm dr.}(b) = \frac{iGQ_{\rm 1PM}^2}{2} \left[\frac{8-5\sigma^2}{3(\sigma^2-1)} + \frac{\sigma(2\sigma^2-3)\operatorname{arccosh}\sigma}{(\sigma^2-1)^{\frac{3}{2}}} \right] + \mathcal{O}(G^4) = i\operatorname{Re} 2\delta_2^{\rm RR}(b) + \mathcal{O}(G^4),$$
(7.63)

where we have recognized the 3PM radiation-reaction phase $\text{Re} 2\delta_2^{\text{RR}}(b)$. We also neglected a *b*-independent $\mathcal{O}(G)$ contribution, which would be tantamount to a finite shift of the (IR-divergent) Shapiro time delay. In this way, recalling that $2i\delta^{\text{dr.}}(b)$ defines the subtraction defining $2i\delta(b)$ in (7.51), we see that the inclusion of static modes in the dressing leads to an eikonal operator in which the overall phase $2i\delta(b)$ is the *conservative* eikonal phase up to 3PM, i.e. it does not include radiation-reaction effects. We shall see in later sections that the information about these effects is not lost, but is actually encoded in the "soft factor" $f_j(k)$ rather than in the explicit phase.

Let us now turn to the prediction of this eikonal operator for two observables: the memory waveform and the angular momentum of the field. Proceeding like for (7.17), but taking the expectation on the $S_{s,r,}|\Psi_{in}\rangle$ state including static effects, we find the following asymptotic limit for the gravitational field considering a large distance r at fixed retarded time u and angles given by n^{μ} (see also Appendix F)

$$h^{\mu\nu} \sim \frac{1}{4\pi r} \int_{-\infty}^{+\infty} \frac{d\omega}{2i\pi} f_{\rm TT}^{\mu\nu}(\omega n) e^{-i\omega u}, \qquad f_{\rm TT}^{\mu\nu}(k) = \Pi_{\rho\sigma}^{\mu\nu}(k) \sum_{n} \frac{\kappa \, p_n^{\rho} p_n^{\sigma}}{p_n \cdot k - i0}, \tag{7.64}$$

and performing the integral over ω by means of the Fourier representation of the Heaviside θ function,

$$\int_{-\infty}^{+\infty} \frac{d\omega}{i2\pi} \frac{e^{-i\omega u}}{-\eta_n \omega - i0} = \int_{-\infty}^{+\infty} \frac{d\omega}{i2\pi} \frac{e^{i\omega\eta_n u}}{\omega - i0} = \theta(\eta_n u),$$
(7.65)

this leads to

$$W^{\mu\nu} = \kappa h^{\mu\nu} \sim \frac{2G}{r} \Pi^{\mu\nu}_{\rho\sigma}(n) \sum_{a} \frac{p_{a}^{\rho} p_{a}^{\sigma} \,\theta(\eta_{a} u)}{(-\eta_{a} p_{a} \cdot n)} \,. \tag{7.66}$$

Note that for u > 0 (u < 0) only the out (in) contributions $\eta_n > 0$ ($\eta_n < 0$) survive, so that the difference $\Delta W_{\mu\nu}$ yields the memory effect (7.21). Expanding (7.66) to leading order in the PM regime, where the out states are equal to the in states up to $\mathcal{O}(G)$ deflections, one instead finds

$$W^{\mu\nu} \sim \frac{2G}{r} \Pi^{\mu\nu}_{\rho\sigma}(n) \sum_{a \in \mathbf{i} n} \frac{p_a^{\rho} p_a^{\sigma}}{p_a \cdot n} + \mathcal{O}(G^2), \qquad (7.67)$$

i.e. the waveform has a *u*-independent $\mathcal{O}(G)$ contribution. This is the leading-order static field, whose inclusion was possible precisely thanks to the -i0 prescription in (7.44). Before moving on to computing different contributions to the angular momentum of the process let us make a few remarks on the traditional GR perspective on this issue.

The issue of angular momentum and of its loss in GR is a subtle one and has a long history (see e.g. [323]). The approach based on the use of Bondi–Sachs coordinates [324,325], which is very convenient for describing the energy loss by the system via gravitational-wave emission, cannot be trivially extended to angular momentum. The problem is that the definition of Bondi's angular momentum, J_B , suffers from a gauge ambiguity related to the possibility of performing supertranslations, a subgroup of the BMS transformations [326,327] that preserves the Bondi–Sachs coordinate conditions.

As an example, the $\mathcal{O}(G)$ static term in (7.67) can always be removed completely by performing a BMS supertranslation,

$$2\kappa\delta_T h_{AB} = 2\kappa T(n)\,\partial_u h_{AB} - r(2D_A D_B - \gamma_{AB} D^C D_C) T(n)\,, \tag{7.68}$$

where *A*, *B*, *C*, . . . label angular directions and $\gamma_{AB} = (\partial_A n) \cdot (\partial_B n)$ is the metric on the sphere with covariant derivative *D*_A, by choosing [321]

$$T(n) = 2G \sum_{a \in \text{in}} (p_a \cdot n) \log(p_a \cdot n/m_a).$$
(7.69)

This raises the issue of whether such a static contribution has a physical meaning. In [321] it was argued that the true radiated angular momentum is the one computed by using a particular BMS gauge, called the canonical gauge in the GR literature, which amounts to setting the initial shear of the Bondi–Sachs metric to zero. In the post-Minkowskian expansion this contribution starts at $\mathcal{O}(G^3)$ and is insensitive to exactly zero frequency gravitons. It was also proposed that the additional static contribution to angular-momentum loss corresponds to choosing a different Bondi gauge, dubbed "intrinsic".

P. Di Vecchia, C. Heissenberg, R. Russo et al.

The issue has been further addressed in [328,329] and, more recently, in [330,331]. The outcome of these investigations is that both the radiative and the static contributions to the angular momentum loss by the two-body system can be defined in a BMS-invariant way. The former contribution is indeed the one computed in the canonical gauge, while the latter can also be given in gauge invariant form. In the approach of [321,329] it corresponds to a different gauge fixing of J_{B} , while in [328,330,331] the mechanical angular momentum loss is identified with the difference between the final and initial Bondi angular momentum computed in *two different* Bondi gauges. In formulae:

$$\Delta J_{\text{mech}} = J_B^{(+)}(u \to +\infty) - J_B^{(-)}(u \to -\infty), \qquad (7.70)$$

where the upper label (-) refer to the usual canonical gauge choice at $u \to -\infty$, while the label (+) is defined by setting the final shear to zero (because of the memory effect this actually means that at $u \to +\infty$ we are no longer in the canonical gauge). By adding and subtracting a term $J_B^{(-)}(u \to +\infty)$ one ends up with:

$$\Delta J_{\text{mech}} = \left(J_B^{(-)}(u \to +\infty) - J_B^{(-)}(u \to -\infty) \right) + \left(J_B^{(+)}(u \to +\infty) - J_B^{(-)}(u \to +\infty) \right)$$

$$\equiv \Delta J_{\text{mech}}^{\text{rad}} + \Delta J_{\text{mech}}^{\text{stat}} .$$
(7.71)

Eq. (7.71) makes it clear that the first term is insensitive to a *u*-independent (i.e. strictly zero-frequency) contribution to $J_B^{(-)}$ while the second term, being evaluated after radiation has stopped, is the difference of two constants and is proportional to the final shear once the initial shear is set to zero [331]. Such a difference is nothing but the well known memory and corresponds to the zero-frequency limit of the News tensor's Fourier transform. In [331] the two prescriptions have been shown to agree at $\mathcal{O}(G^2)$.

If the decomposition (7.71) can be confirmed at all orders in *G*, it would establish a clear and strict connection between the above BMS-based approach and the amplitude-based one discussed in this section where we distinguished strictly zero-frequency gravitons from those with any non-vanishing frequency. In [331] it has been explicitly checked that the definition (7.71) of the static contribution to the angular momentum loss is in full agreement with the one computed at $\mathcal{O}(G^3)$ in [144,332] and discussed hereafter. Perhaps another interesting remark before turning to the calculation is that the same technique to compute the static angular momentum loss applies also in the case of a massless scalar field (see (7.110) below), for which no BMS ambiguities arise.

To discuss the angular momentum of the gravitational field, we take the expectation in the $S_{s.r.}|\Psi_{in}\rangle$ state of the corresponding operator in De Donder gauge, which reads as follows

$$J_{\alpha\beta} = -i \int_{k} a^{\dagger}_{\mu\nu}(k) \Big(D^{\mu\nu,\rho\sigma} k_{[\alpha} \frac{\partial}{\partial k^{\beta}]} + 2\eta^{\mu\rho} \delta^{\nu}_{[\alpha} \delta^{\sigma}_{\beta]} \Big) a_{\rho\sigma}(k) , \qquad (7.72)$$

where

$$D^{\mu\nu,\rho\sigma} = \frac{1}{2} \left(\eta^{\mu\rho} \eta^{\nu\sigma} + \eta^{\mu\sigma} \eta^{\nu\rho} - \eta^{\mu\nu} \eta^{\rho\sigma} \right)$$
(7.73)

is the tensor structure appearing in the gauge-fixed De Donder action (see Appendix A.2). At this stage we face the problem of simplifying the intermediate state sum using the physical projector (5.45). In particular, we would like to see how the dependence on the reference vector λ^{α} appearing in that projector eventually cancels out. Relying on the transversality property $k_{\mu}F^{\mu\nu} = 0$, we obtain the following result, expressed as a sum of orbital and spin contributions,

$$\mathcal{J}_{\alpha\beta} = \mathcal{L}_{\alpha\beta} + \mathcal{S}_{\alpha\beta} \,. \tag{7.74}$$

Here the orbital angular momentum reads (with the soft $F^{\mu\nu}$ as in (7.44))

$$i\mathcal{L}_{\alpha\beta} = \int_{k}^{*} \left(F_{\mu\nu}^{*} k_{[\alpha} \frac{\overleftrightarrow{\partial} F^{\mu\nu}}{\partial k^{\beta]}} - \frac{F^{\prime*}}{D-2} k_{[\alpha} \frac{\overleftrightarrow{\partial} F^{\prime}}{\partial k^{\beta]}} + 2k_{[\alpha} F_{\beta]\mu}^{*} \lambda \cdot F^{\mu} - 2\lambda \cdot F_{\mu}^{*} k_{[\alpha} F_{\beta]}^{\mu} \right),$$
(7.75)

with $\lambda \cdot F^{\alpha} = \lambda_{\beta} F^{\alpha\beta}$, $F' = \eta_{\rho\sigma} F^{\rho\sigma}$ and

$$f\frac{\partial}{\partial k^{\beta}}g \equiv \frac{1}{2}\left(f\frac{\partial g}{\partial k^{\beta}} - \frac{\partial f}{\partial k^{\beta}}g\right),\tag{7.76}$$

while the spin angular momentum is given by

$$i\mathcal{S}_{\alpha\beta} = \int_{k}^{k} \left(2F_{\mu[\alpha}^{*}F_{\beta]}^{\mu} + 2k_{[\beta}F_{\alpha]\mu}^{*}\lambda \cdot F^{\mu} + 2\lambda \cdot F_{\mu}^{*}k_{[\alpha}F_{\beta]}^{\mu} \right).$$

$$(7.77)$$

Therefore, the dependence on the reference vector λ^{α} drops out in the sum and we obtain the simple expression [144,332]

$$\mathcal{J}_{\alpha\beta} = -i \int_{k}^{*} F_{\mu\nu}^{*} \left[\left(\eta^{\mu\rho} \eta^{\nu\sigma} - \frac{1}{D-2} \eta^{\mu\nu} \eta^{\rho\sigma} \right) k_{[\alpha} \frac{\overleftrightarrow{\partial}}{\partial k^{\beta]}} + 2 \eta^{\mu\rho} \delta^{\nu}_{[\alpha} \delta^{\sigma}_{\beta]} \right] F_{\rho\sigma} .$$
(7.78)

The two terms within square brackets are reminiscent of the orbital and spin part of the gravitational angular momentum, in particular the factor of 2 is associated to dealing with a spin-2 field. However, the two terms are not separately gauge invariant, as one can easily check, pointing to the fact that only their sum can be given a well defined physical meaning. Indeed, in order to obtain (7.78) where the intermediate transverse-traceless projections have dropped out, one needs to rely on nontrivial cancellations between the orbital and the spin term (we refer to [144] for further details on this derivation). The integrals appearing in (7.78) can be reduced to (7.54) by first combining pieces with "positive and negative frequencies" like we did for (7.51). In this way we can write (7.78)

$$\mathcal{J}^{\alpha\beta} = \frac{i\kappa^2}{2} \sum_{n,m} \left(\sigma_{nm}^2 - \frac{1}{2} \right) m_n^2 m_m^2 K^{[\alpha} p_m^{\beta]} - i\kappa^2 \sum_{n,m} (p_n p_m) p_n^{[\alpha} p_m^{\beta]} (p_m \cdot K)$$
(7.79)

in terms of the integral

$$K^{\mu} = \int \frac{d^4k}{(2\pi)^4} \frac{\operatorname{sgn}(k^0) 2\pi \,\delta(k^2) k^{\mu} \theta(\omega^* - |k^0|)}{(p_n k + i0)(p_m k - i0)^2} = A p_n^{\mu} + B p_m^{\mu} \,.$$
(7.80)

The integral $(p_m K)$ can be expressed in terms of I_{nm} in (7.61)

$$(p_m K) = \frac{I_{nm}}{m_n m_m} = \frac{i\pi}{2} \frac{\eta_n - \eta_m}{m_n m_m} \frac{\eta_n \eta_m \arccos \sigma_{nm}}{(2\pi)^2 \sqrt{\sigma_{nm}^2 - 1}},$$
(7.81)

while $(p_n K)$ can be easily computed

$$(p_n K) = \int \frac{d^4 k}{(2\pi)^4} \frac{\operatorname{sgn}(k^0) 2\pi \delta(k^2) \theta(\omega^* - |k^0|)}{(p_m k - i0)^2} = -\frac{i\pi \eta_m}{(2\pi)^2 m_m^2}$$
(7.82)

from the product of the two integrals

$$\int_{-\omega^*}^{\omega^*} \frac{\omega d\omega}{(-\eta_m \omega - i0)^2} = -i\pi \eta_m, \qquad \frac{1}{(2\pi)^2} \int_0^{\pi} \frac{\sin \theta d\theta}{(E_m - |\vec{p}_n| \cos \theta)^2} = \frac{1}{(2\pi)^2 m_m^2}.$$
(7.83)

As a result the coefficient A in (7.80) is given by

$$A = \frac{m_m^2(p_n K) + (p_n p_m) p_m K}{(p_n p_m)^2 - m_n^2 m_m^2} = -\frac{i\pi}{2} \frac{\eta_m + \eta_n + (\eta_n - \eta_m) (\sigma_{nm} \Delta_{nm} - 1)}{(2\pi)^2 m_n^2 m_m^2 (\sigma_{nm}^2 - 1)}$$
(7.84)

with Δ_{nm} as in (7.86) below, *B* is not needed because it will not contribute when inserted in the first term of (7.79), by antisymmetry. For the same reason, also the term with $\eta_n + \eta_m$ in (7.84) does not contribute, and we find the following expression for the angular momentum carried by the static gravitational field in D = 4, after sending $\omega^* \rightarrow 0$,

$$\mathcal{J}^{\alpha\beta} = -\sum_{n\in\text{in}}\sum_{m\in\text{out}}c_{nm}p_n^{[\alpha}p_m^{\beta]},\tag{7.85}$$

with

$$c_{nm} = 2G\left[\left(\sigma_{nm}^2 - \frac{1}{2}\right)\frac{\sigma_{nm}\Delta_{nm} - 1}{\sigma_{nm}^2 - 1} - 2\sigma_{nm}\Delta_{nm}\right], \qquad \Delta_{nm} = \frac{\operatorname{arccosh}\sigma_{nm}}{\sqrt{\sigma_{nm}^2 - 1}}.$$
(7.86)

The expression (7.85) is Lorentz covariant, translation invariant and valid for arbitrary kinematics p_n of the background hard process, i.e. it holds regardless whether or not the outgoing momenta can be regarded as small deflections of the incoming ones. In fact, just like Weinberg's theorem, the formula holds independently of the number and of the specific details of the hard particles taking part in the background hard process, which may also carry spin or be subject to tidal deformations: one need only assign their momenta. However, of course, it only captures the contribution to the angular momentum due to static/zero-frequency fields.

For a 2
$$\rightarrow$$
 2 process, $p_4^{\mu} = Q^{\mu} - p_1^{\mu}, p_3^{\mu} = -Q^{\mu} - p_2^{\mu}$ up to $\mathcal{O}(G^3)$ corrections, and thus

$$\mathcal{J}^{\alpha\beta} = p_1^{\mu} Q^{\beta} (c_{13} - c_{14}) - p_2^{\mu} Q^{\beta} (c_{24} - c_{23}) + \mathcal{O}(G^4).$$
(7.87)

where we have used that $\sigma_{13} = \sigma_{24}$. In the PM expansion, using $c_{14} - c_{13} = \frac{G}{2}\mathcal{I}(\sigma) + \mathcal{O}(G^3)$, $c_{23} - c_{24} = \frac{G}{2}\mathcal{I}(\sigma) + \mathcal{O}(G^3)$ with

$$\frac{1}{2}\mathcal{I}(\sigma) = \frac{8-5\sigma^2}{3(\sigma^2-1)} + \frac{\sigma(2\sigma^2-3)\operatorname{arccosh}\sigma}{(\sigma^2-1)^{3/2}},$$
(7.88)

we find

$$\mathcal{J}^{\alpha\beta} = -\frac{G}{2}(p_1 - p_2)^{[\alpha} Q^{\beta]} \mathcal{I}(\sigma) + \mathcal{O}(G^4).$$
(7.89)

In particular, substituting (3.16) for the 1PM impulse and using

$$Q^{\mu} = -\frac{b^{\mu}}{b}Q_{1\rm PM} + \mathcal{O}(G^3), \qquad (7.90)$$

we get

$$\mathcal{I}^{\alpha\beta} = \frac{G^2 m_1 m_2 (2\sigma^2 - 1)}{b^2 \sqrt{\sigma^2 - 1}^2} \mathcal{I}(\sigma) (p_1 - p_2)^{[\alpha} b^{\beta]} + \mathcal{O}(G^3),$$
(7.91)

which, in the center-of-mass frame where $-p_1 = (E_1, p)$ and $-p_2 = (E_2, -p)$, matches the $\mathcal{O}(G^2)$ results in Eq. (4.6) of [289] and in Eq. (12) of [332]. However, as already emphasized, thanks to the universality of the soft theorem (7.89) is also valid when the particles carry classical spin [81], provided one substitutes the appropriate 1PM deflection. Taking into account the 2PM correction to the deflection (4.58) on top of that, Eq. (7.89) also correctly accounts for the $\mathcal{O}(G^3)$ static angular momentum, as well tidal effects [57]. However, to that order, additional genuinely radiative terms also appear and we shall see in the next sections how they can be calculated in the eikonal framework.

7.3. The ultrarelativistic and massless limits

It is very instructive to inspect the behavior of the spectrum (7.43) in several physically relevant limits. The standard PM regime requires that $Q^2 \sim (p\Theta)^2 \ll 2m_i^2$. In this limit, one can extract the leading (3PM) contribution by Taylor-expanding the first line of (7.43) in Q^2 , while the second only give subleading contributions,

$$\lim_{\omega \to 0} \frac{dE}{d\omega} \simeq \frac{2G}{\pi} Q^2 \left[\frac{8 - 5\sigma^2}{3(\sigma^2 - 1)} + \frac{(2\sigma^2 - 3)\sigma \operatorname{arccosh} \sigma}{(\sigma^2 - 1)^{3/2}} \right].$$
(7.92)

When rewritten in terms of the divergent part of Im 2 δ , using the simple link (7.39), the result (7.92) then reproduces the divergent part of Im 2 δ_2 in (5.55) (and agrees with Eq. (5.14) of [106]), once we use $Q \simeq Q_{1PM}$ (see (3.16) for the explicit expression). Moreover, to the leading 3PM order, Eq. (7.92) has been explicitly shown to hold also if the colliding objects carry spin, for generic spin alignments [81]. If we then consider in the formal ultra-relativistic limit $\sigma \rightarrow \infty$, Eq. (7.43) gives:

$$\lim_{\omega \to 0} \frac{dE}{d\omega} \simeq \frac{4G}{\pi} Q^2 \left(\log \frac{s}{m_1 m_2} - \frac{5}{6} \right).$$
(7.93)

However, the exact expression (7.43) shows that the PM approximation breaks down, even when Θ is small, when one of the two particles becomes ultra-relativistic, $p \gg m_i$. Indeed, while in this regime the first line in (7.43) can always be expanded to first order in Q^2/s , the second one presents instead a branch point at $Q^2 = -4m_i^2$, corresponding to the *t*-channel thresholds (outside the physical region). This implies that the PM expansion starts to diverge whenever for at least one index *i*, (see Eq. (6.42) and comments below it)

$$\frac{Q}{2m_i} > 1. \tag{7.94}$$

Since $Q \simeq p \Theta$ which is of order $\sqrt{m_1 m_2 \sigma/2} \Theta$, we thus recover the bound first pointed out by D'Eath [333] for the validity of the PM approximation,

$$\max\left\{\sqrt{\frac{m_1}{m_2}}, \sqrt{\frac{m_2}{m_1}}\right\} \frac{\sqrt{\sigma}}{2\sqrt{2}} \lesssim 1.$$
(7.95)

In view of this bound one could say that the PM approximation holds for weakly gravitating systems with "generic" but not entirely arbitrary speeds.

For the ZFL of the emitted energy spectrum, however, the ultrarelativistic regime, or equivalently the massless limit, can be obtained from (7.43) by considering $2p \rightarrow \sqrt{s}$ and $m_1, m_2 \ll Q = \sqrt{s} \sin \frac{\Theta}{2}$. The mass singularities appearing separately in each line of the formula neatly cancel against each other and then (7.43) reduces to (see e.g. [143] where the result is extended to an arbitrary number of external massless legs)

$$\lim_{s \to 0} \frac{dE}{d\omega} \simeq \frac{4G}{\pi} \left[s \log \frac{s}{s - Q^2} + Q^2 \log \frac{s - Q^2}{Q^2} \right]$$
(7.96)

or in terms of the deflection angle

6

$$\lim_{\omega \to 0} \frac{dE}{d\omega} \simeq -\frac{4G}{\pi} s \left[\cos^2 \frac{\Theta}{2} \log \cos^2 \frac{\Theta}{2} + \sin^2 \frac{\Theta}{2} \log \sin^2 \frac{\Theta}{2} \right],$$
(7.97)

which agrees with the leading soft limit of Eq. (5.12) of [310]. If we then consider the small Θ limit of (7.97), after taking the ultrarelativistic limit, at leading order for $\Theta \ll 1$ we have

$$\lim_{\omega \to 0} \frac{dE}{d\omega} \simeq \frac{Gs\Theta^2}{\pi} \left[1 + \log \frac{4}{\Theta^2} \right],\tag{7.98}$$

which reproduces the result obtained in [334] (see also [333]) within a classical GR approach and in [142] from a scattering amplitudes perspective. As anticipated, the true ultrarelativistic behavior of the spectrum exhibits a non-analytic dependence on Θ and thus on the Newton constant *G*, compatibly with the breakdown of the PM approximation. These features are in fact shared by the leading-order angular momentum loss. For the component orthogonal to the scattering plane, one finds in the ultrarelativistic limit

$$\mathcal{J}^{xy} \sim 2Gs \, \sin \Theta \, \log \frac{\cos \frac{\Theta}{2}}{\sin \frac{\Theta}{2}} \,, \tag{7.99}$$

which when further expanded for small Θ yields

$$\mathcal{J}^{xy} \sim Gs \Theta \log \frac{4}{\Theta^2},$$
(7.100)

again exhibiting the characteristic non-analyticity in the Newton constant.

7.4. The $\mathcal{N} = 8$ eikonal operator in the soft limit

To include the presence of other massless fields (scalars and vectors), which are needed to discuss the case of N = 8 supergravity where the massive particles are described by KK modes, we can use the fact that *S*-matrix elements for soft emissions factorize in a way analogous to (7.1), with soft factors that instead of $w_i(k)$ are given by

$$w_{j}^{\text{vec}}(k) = \sum_{n} \eta_{n} e_{n} \frac{\varepsilon_{\mu,j}^{*}(k) p_{n}^{\mu}}{p_{n} \cdot k}, \qquad w_{j}^{\text{sc}}(k) = \sum_{n} \frac{g_{n}}{p_{n} \cdot k}$$
(7.101)

for vectors and scalars respectively. As above, η_n takes the value +1 for outgoing and -1 for incoming states, while e_n and g_n denote suitable couplings. These new soft particles are easily accommodated in the eikonal operator by including in the exponent of (7.8) the relevant operators a_d for the dilaton (with coupling $g_n = -\kappa m_n^2/\sqrt{D-2}$) and $a_{v_i,j}$ for two vectors ($e_n = \sqrt{2} \kappa m_i$) and a_{s_i} for two scalars ($g_n = \kappa m_i^2$). The corresponding "memory waveforms" are also obtained in a very similar way.

In analogy with (7.40), we find the following simple and general expressions for the energy emission spectra,

$$\lim_{\omega \to 0} \frac{dE^{\text{vec}}}{d\omega} = \frac{1}{4\pi^2} \sum_{n,m} e_n e_m (-\sigma_{nm}) F_{nm} , \qquad (7.102)$$

$$\lim_{\omega \to 0} \frac{dE^{\rm sc}}{d\omega} = \frac{1}{4\pi^2} \sum_{n,m} \frac{g_n g_m}{m_n m_m} F_{nm} \,. \tag{7.103}$$

Combining scalars, vectors and the graviton, we obtain a remarkably simple result for the $\mathcal{N} = 8$ spectrum,

$$\lim_{\omega \to 0} \frac{dE^{\mathcal{N}=8}}{d\omega} = \frac{2G}{\pi} \sum_{n,m} m_n m_m (\sigma'_{nm})^2 F_{nm} , \qquad (7.104)$$

where $\sigma'_{nm} = \sigma_{nm} - 1$ if *n* and *m* have momenta compactified along the same KK direction (so that $m_n = m_m$) and $\sigma'_{nm} = \sigma_{nm}$ otherwise. Specializing (7.104) to the 2 \rightarrow 2 kinematics as in (7.41), we have

$$\lim_{\omega \to 0} \frac{dE^{\mathcal{N}=8}}{d\omega} = \frac{4G}{\pi} \left[2m_1 m_2 \sigma^2 \frac{\arccos \sigma}{\sqrt{\sigma^2 - 1}} - 2m_1 m_2 \sigma_Q^2 \frac{\operatorname{arccosh} \sigma_Q}{\sqrt{\sigma_Q^2 - 1}} - \frac{(Q^2)^2}{4m_1^2} \frac{\operatorname{arccosh} \left(1 + \frac{Q^2}{2m_1^2}\right)}{\sqrt{\left(1 + \frac{Q^2}{2m_1^2}\right)^2 - 1}} - \frac{(Q^2)^2}{4m_2^2} \frac{\operatorname{arccosh} \left(1 + \frac{Q^2}{2m_2^2}\right)}{\sqrt{\left(1 + \frac{Q^2}{2m_2^2}\right)^2 - 1}} \right].$$
(7.105)

The standard relativistic PM regime, where $Q^2 \ll 2m_i^2$ in the equation above, has an analytic PM expansion whose leading term reads

$$\lim_{\omega \to 0} \frac{dE^{\mathcal{N}=8}}{d\omega} \simeq \frac{4GQ^2}{\pi} \left[\frac{\sigma^2}{\sigma^2 - 1} + \frac{(\sigma^2 - 2)\sigma\operatorname{arccosh}\sigma}{(\sigma^2 - 1)^{3/2}} \right].$$
(7.106)

Approximating $Q \simeq p\Theta$ with Θ the leading deflection angle given in (6.44), and exploiting the link (7.39), this reproduces the divergent part of Im $2\delta_2$ in (5.54).

On the contrary, in the regime (7.94), the small- Θ expansion is non-analytic and interestingly, in extreme ultrarelativistic kinematics where the masses can be neglected, one obtains again (7.96). Indeed, the contributions related to the dilaton, and the Kaluza–Klein scalars and vectors become negligible in this regime, as suggested by the fact that (7.102), (7.103) scale with lower powers of σ_{nm} compared to (7.40), and the graviton provides the dominant behavior. At ultra-high energies, in this way, only the contribution due to the emission of gravitons survives. This is a universal expression for two-derivatives theories in accordance with the expectation that gravity dominates the high-energy limit not just for the elastic scattering, as argued in [104], but also in the (soft) radiation sector.

The inclusion of static modes for vector and scalar fields proceeds along similar lines as for the graviton case discussed above by means of the -i0 prescription, i.e. by introducing the modified soft factors

$$f_{j}^{\text{vec}}(k) = \sum_{n} \eta_{n} e_{n} \frac{\varepsilon_{\mu,j}^{*}(k) p_{n}^{\mu}}{p_{n} \cdot k - i0}, \qquad f^{\text{sc}}(k) = \sum_{n} \frac{g_{n}}{p_{n} \cdot k - i0}$$
(7.107)

using which one then finds the following vector and scalar waveforms,

$$\langle A_{\mu}(x) \rangle \sim \frac{1}{4\pi r} \Pi^{\mu}_{\nu}(\hat{x}) \sum_{n} \frac{e_{n} p_{n}^{\nu} \theta(\eta_{n} u)}{E_{n} - \vec{k}_{n} \cdot \hat{x}}, \quad \langle \Phi(x) \rangle \sim \frac{1}{4\pi r} \sum_{n} \frac{g_{n} \theta(\eta_{n} u)}{E_{n} - \vec{k}_{n} \cdot \hat{x}}$$
(7.108)

where $\Pi^{\mu\nu}$ is the transverse projector. For the angular momenta one finds

$$\mathcal{J}_{\alpha\beta}^{\rm sc} = -\frac{i}{2} \int_{\vec{k}} \left(f^* k_{\alpha} \frac{\partial f}{\partial k^{\beta}} - k_{\alpha} \frac{\partial f^*}{\partial k^{\beta}} f \right)$$
(7.109)

and

(

$$\mathcal{J}^{\rm sc})^{\mu\nu} = \frac{1}{16\pi} \sum_{n,m} \frac{g_n g_m}{m_n^2 m_m^2} \frac{\sigma_{nm} \Delta_{nm} - 1}{\sigma_{nm}^2 - 1} (\eta_n - \eta_m) p_n^{[\mu} p_m^{\nu]}, \qquad (7.110)$$

for the scalar case, as well as

$$\mathcal{J}_{\alpha\beta}^{\text{vec}} = -i \int_{\vec{k}} F_{\mu}^{*} \left(\eta^{\mu\nu} k_{[\alpha} \frac{\partial}{\partial k^{\beta}]} + \delta^{\mu}_{[\alpha} \delta^{\nu}_{\beta]} \right) F_{\nu}$$
(7.111)

which evaluates to

$$\left(\mathcal{J}^{\text{vec}}\right)^{\alpha\beta} = \frac{1}{16\pi} \sum_{n,m} \frac{e_n e_m}{m_n m_m} \left[-\sigma_{nm} \frac{\sigma_{nm} \Delta_{nm} - 1}{\sigma_{nm}^2 - 1} + \Delta_{nm} \right] (\eta_n - \eta_m) p_n^{[\alpha} p_m^{\beta]}$$
(7.112)

for the vector. Combining this with the graviton result discussed above, one arrives at the following simple formula for the static angular momentum loss in N = 8,

$$\mathcal{J}_{\mathcal{N}=8}^{\alpha\beta} = \frac{G}{2} \sum_{n,m} \left[\sigma_{nm}^{\prime 2} \frac{\sigma_{nm} \Delta_{nm} - 1}{\sigma_{nm}^2 - 1} - 2\sigma_{nm}^{\prime} \Delta_{nm} \right] (\eta_n - \eta_m) p_n^{[\alpha} p_m^{\beta]} , \qquad (7.113)$$

where $\sigma'_{nm} = \sigma_{nm} - 1$ if *n* and *m* corresponds to states compactified along the same direction, so that $m_n = m_m$, while it equals $\sigma'_{nm} = \sigma_{nm}$ otherwise.

8. The eikonal operator beyond the soft limit

In this section, we discuss a framework that allows us to combine the dynamical information about graviton exchanges that is contained in the elastic eikonal, which, as we have seen, determines the deflection of the colliding black holes, with the information about graviton emissions that is contained in the inelastic amplitudes. Building on the idea of introducing creation/annihilation operators for gravitons already introduced in the previous section, we construct here an eikonal operator that dictates the final state of the collision out of two ingredients: the eikonal phase and coherent graviton emissions with generic, i.e. not necessarily soft, frequencies. As we shall see, this provides a comprehensive strategy to calculate observable quantities associated to the scattering up to $\mathcal{O}(G^3)$: the waveforms themselves, the emitted linear and angular momentum, and the changes in the linear and angular momentum of the colliding objects. In particular, this will allow us to explicitly check the corresponding balance laws.

8.1. The elastic eikonal revisited

So far the external states involved in the scattering have been described simply in terms of momentum eigenstates. Following [116], we now introduce wavepackets Φ_i to describe classical particles, so the initial state for the 2 \rightarrow 2 scattering reads

$$|\psi\rangle = \int_{-p_1} \int_{-p_2} \Phi(-p_1) \Phi(-p_2) e^{ip_1 b_1 + ip_2 b_2} |-p_1, -p_2\rangle .$$
(8.1)

The wavepackets are peaked around the classical value (that with an abuse of notation we will still indicate with $-p_i$) and we use the following notation for the on-shell integrals

$$\int_{p_i} = \int \frac{d^D p_i}{(2\pi)^D} 2\pi \theta(p_i^0) \delta(p_i^2 + m_i^2) .$$
(8.2)

By following [120], we can formally write the final state $S_c |\psi\rangle = (1 + iT) |\psi\rangle$ in the elastic case (the subscript *c* stands for "conservative") as

$$S_{c}|\psi\rangle = \int_{-p_{1}} \int_{-p_{2}} \int_{p_{3}} \int_{p_{4}} \Phi(-p_{1})\Phi(-p_{2})e^{ip_{1}b_{1}+ip_{2}b_{2}}|p_{3},p_{4}\rangle\langle p_{3},p_{4}|S|-p_{1},-p_{2}\rangle.$$
(8.3)

Of course we are interested only in the classical contribution to the S-matrix element which, as discussed in the previous sections, is more easily captured in an impact parameter representation, see (4.1). Thus we write the momentum-space S-matrix element

$$\langle p_3, p_4 | iT | -p_1, -p_2 \rangle = (2\pi)^D \int d^D Q \, \delta^{(D)}(p_1 + p_4 - Q) \delta^{(D)}(p_2 + p_3 + Q) \, i\mathcal{A}(-(p_1 + p_2)^2, Q^2)$$
(8.4)

as the inverse Fourier transform of the eikonal result

$$2\pi\,\delta(2\bar{p}_2Q)\,2\pi\,\delta(2\bar{p}_1Q)\,i\mathcal{A}(-(p_1+p_2)^2,\,Q^2) = \int d^Dx\,\left[(1+2i\Delta(b))\left(e^{2i\delta(b)}-1\right)\right]e^{-ixQ} \,. \tag{8.5}$$

In this subsection, for ease of notation, we simply drop the imaginary part of 2 δ . Let us discuss the quantities *b* and \bar{p}_i in this relation. The momenta are simply defined as in (1.28), with q^{μ} replaced by Q^{μ} ,

$$p_1 = -\bar{p}_1 + \frac{Q}{2}, \qquad p_2 = -\bar{p}_2 - \frac{Q}{2}, \qquad p_4 = \bar{p}_1 + \frac{Q}{2}, \qquad p_3 = \bar{p}_2 - \frac{Q}{2},$$
(8.6)

so the delta functions on the l.h.s. of (8.5) follow from the fact that the p_i are on-shell. Then in order to produce these delta functions from the r.h.s., the eikonal in the square parenthesis should not depend on the components of x along \bar{p}_i , thus we introduced b to indicate the components of x orthogonal to \bar{p}_i recovering the property $b\bar{p}_{1,2} = 0$ of the eikonal impact parameter. By using (8.4) and (8.5) in (8.3) we get

$$S_{c}|\psi\rangle = \int_{p_{3}} \int_{p_{4}} |p_{3}, p_{4}\rangle e^{-ib_{1}p_{4}} e^{-ib_{2}p_{3}} \int \frac{d^{D}Q}{(2\pi)^{D}} \int d^{D}x \\ \times e^{iQ(b_{1}-b_{2})} e^{2i\delta(b)} e^{-ixQ} (1+2i\Delta(b)) \Phi(p_{4}-Q) \Phi(Q+p_{3}) .$$
(8.7)

8.1.1. Saddle points

The key idea of the eikonal approach is to approximate the integrals with the stationary phase contribution and this can be done in (8.7) both for *x* and *Q* [120]. From the condition on *x* we obtain

$$Q_{\mu} = \frac{\partial 2\delta(b)}{\partial x^{\mu}} = \frac{\partial 2\delta(b)}{\partial b} \frac{b_{\mu}}{b}$$
(8.8)

where we used $\frac{\partial b}{\partial x^{\mu}} = \frac{b_{\mu}}{b}$. We thus recovered the second relation in Eq. (6.11). In this approach, the relation between *b* and the initial angular momentum follows from the other stationary phase condition

$$(b_1 - b_2)^{\mu} - x^{\mu} = -\frac{\partial 2\delta(b)}{\partial b} \frac{b_{\nu}}{b} \frac{\partial b^{\nu}}{\partial Q^{\mu}} , \qquad (8.9)$$

where the r.h.s. follows from the implicit dependence of *b* on *Q*. In order to make this explicit, we need to decompose *x* along *b* and the space spanned by \bar{p}_i

$$x^{\mu} = b^{\mu} + (\bar{p}_1 + \bar{p}_2)^{\mu} A_1 + (\bar{p}_1 - \bar{p}_2)^{\mu} A_2 = b^{\mu} + (p_4 + p_3)^{\mu} A_1 + (p_4 - p_3 - Q)^{\mu} A_2 , \qquad (8.10)$$

which implies

2

$$\frac{\partial b^{\nu}}{\partial Q^{\mu}} = -(\bar{p}_1 + \bar{p}_2)^{\nu} \frac{\partial A_1}{\partial Q^{\mu}} - (\bar{p}_1 - \bar{p}_2)^{\nu} \frac{\partial A_2}{\partial Q^{\mu}} + \delta^{\nu}_{\mu} A_2 , \qquad (8.11)$$

since the derivatives involved in the stationary point conditions are calculated by keeping x, $p_{3,4}$ fixed. By using (8.6), (8.10) and (8.11) in (8.9), and noting that only the last term in (8.11) gives a nonvanishing contribution to it, we get

$$b_{I}^{\mu} \equiv (b_{1} - b_{2})^{\mu} = b^{\mu} - (p_{1} + p_{2})^{\mu} A_{1} - (p_{1} - p_{2})^{\mu} A_{2}.$$
(8.12)

The classical values of $A_{1,2}$ can be determined by imposing that $b\bar{p}_{1,2} = 0$ and $b_J p_{1,2} = 0$ and by using (8.6) in (8.12). We obtain

$$A_1 = \frac{(m_1^2 - m_2^2)|Q|b}{4m_1^2 m_2^2 (\sigma^2 - 1)} , \quad A_2 = -\frac{s|Q|b}{4m_1^2 m_2^2 (\sigma^2 - 1)} .$$
(8.13)

By contracting (8.12) with b_j , we obtain $b_j^2 = b_j b$. Contracting it instead with b, and recalling that $Q \cdot b = -|Q|b$, since the force is attractive for gravitational theories, we get

$$b_J^2 = b^2 \left(1 - \frac{sQ^2}{4m_1^2 m_2^2 (\sigma^2 - 1)} \right) = b^2 \cos^2 \frac{\Theta}{2} , \qquad (8.14)$$

8.1.2. Impulse and angular momentum

We now use the elastic eikonal operator (8.7) to calculate the impulse by taking expectations of the momentum operator for particle 1,

$$P_1^{\alpha} = \int_p p^{\alpha} a_1^{\dagger}(p) a_1(p) \,. \tag{8.15}$$

Applying this operator to the initial state (8.1), we obtain

$$P_1^{\alpha}|\psi\rangle = \int_{-p_1} \int_{-p_2} \Phi_1(-p_1)\Phi_2(-p_2)|-p_1, -p_2\rangle \, e^{ib_1 \cdot p_1 + ib_2 \cdot p_2}(-p_1^{\mu}) \tag{8.16}$$

after using that $|-p_1\rangle = a_1^{\dagger}(-p_1)|0\rangle$ and therefore the expectation value,

$$\langle \psi | P_1^{\alpha} | \psi \rangle = \int_{-p_1} |\Phi_1(-p_1)|^2 (-p_1^{\mu}).$$
(8.17)

where we have used $\langle p|p'\rangle 2\pi\theta(p^0)\delta(p^2+m_1^2) = (2\pi)^D\delta^{(D)}(p-p')$ that follows from the canonical commutation relations. For the final state, using (8.7), we get

$$P_{1}^{\mu}S_{c}|\psi\rangle = \int_{p_{3}}\int_{p_{4}} e^{-ib_{1}\cdot p_{4}} \Phi_{1}(p_{4}-Q)e^{-ib_{2}\cdot p_{3}} \Phi_{2}(p_{3}+Q)p_{4}^{\mu}|p_{3},p_{4}\rangle$$

$$\times \int \frac{d^{D}Q}{(2\pi)^{D}} \int d^{D}x \, e^{i(b-x)\cdot Q+2i\delta(s,b)}$$
(8.18)

and for the expectation value calculated at the saddle point,

$$\langle \psi | S_c^{\dagger} P_1^{\mu} S_c | \psi \rangle = \int_{p_3} \int_{p_4} p_4^{\mu} | \Phi_1(p_4 - Q) |^2 | \Phi_2(p_3 + Q) |^2 .$$
(8.19)

Formally performing the shifts $p_4^\mu = Q^\mu - p_1^\mu$, $p_3^\mu = -Q^\mu - p_2^\mu$, we recover

$$\langle \psi | S_c^{\dagger} P_1^{\mu} S_c | \psi \rangle - \langle \psi | P_1^{\alpha} | \psi \rangle = \int_{-p_1} \int_{-p_2} Q^{\mu} | \Phi_1(-p_1) |^2 | \Phi_2(-p_2) |^2 \to Q^{\mu}.$$
(8.20)

Of course for particle 2 we find

$$\langle \psi | S_c^{\dagger} P_2^{\mu} S_c | \psi \rangle - \langle \psi | P_2^{\alpha} | \psi \rangle = -Q^{\mu}.$$
(8.21)

The Jacobian for the change of variables considered in the last step, as well as the integral over the fluctuations around the saddle points, would require further study. We shall return to this point in the outlook Section 9.

We will provide the analogous derivation for the angular momentum in the more complete eikonal framework discussed in the ensuing sections which will also include dissipative effects (see Eq. (8.152) for the complete final result). However, let us isolate here the conservative portion of the answer corresponding in particular to (8.156) for particle 1,

$$\Delta L_{(1c)\alpha\beta} = b_{1[\alpha} Q_{\beta]} + p_{4[\alpha} \frac{\partial 2\delta(b)}{\partial p_4^{\beta]}}, \qquad \Delta L_{(2c)\alpha\beta} = -b_{2[\alpha} Q_{\beta]} + p_{3[\alpha} \frac{\partial 2\delta(b)}{\partial p_3^{\beta]}}.$$
(8.22)

Focusing first on particle 1, we recall that 2δ depends on p_4^{μ} both via $\sigma = -p_3 \cdot p_4/(m_1m_2)$ and via the projection that relates b_1^{μ} to b^{μ} as in (8.10). As a result, we find

$$\Delta L_{(1c)\alpha\beta} = b_{1[\alpha}Q_{\beta]} + \left[p_{1[\alpha|}p_{2|\beta]} + (p_1 + p_2)_{[\alpha}Q_{\beta]}\right] \frac{\partial 2\delta(s,b)}{\partial p_1 \cdot p_2} + (A_1 + A_2)p_{1[\alpha}Q_{\beta]}$$
(8.23)

and similarly, for particle 2,

$$\Delta L_{(2c)\alpha\beta} = -b_{2[\alpha}Q_{\beta]} - \left[p_{1[\alpha|}p_{2|\beta]} + (p_1 + p_2)_{[\alpha}Q_{\beta]}\right] \frac{\partial 2\delta(s,b)}{\partial p_1 \cdot p_2} + (A_1 - A_2)p_{2[\alpha}Q_{\beta]}.$$
(8.24)

As a result

$$\Delta L_{(1c)\alpha\beta} + \Delta L_{(2c)\alpha\beta} = (b_J + A_1(p_1 + p_2) + A_2(p_1 - p_2))_{[\alpha} Q_{\beta]}.$$
(8.25)

Using (8.12), we see that the combination between round brackets is precisely b_{α} . Since $b^{[\alpha}Q^{\beta]} = 0$ by (8.8), we get

$$\Delta L_{(1c)\alpha\beta} + \Delta L_{(2c)\alpha\beta} = 0.$$
(8.26)

Therefore, as expected, no mechanical angular momentum is lost by the two-body system in the conservative approximation.

To study (8.23) on its own, let us choose a frame where the center-of-mass sits in the origin of the transverse plane

$$E_1 b_1^{\alpha} + E_2 b_2^{\alpha} = 0, \qquad (8.27)$$

i.e.

$$b_1^{\alpha} = \frac{E_2}{E} b_J^{\alpha} , \qquad b_2^{\alpha} = -\frac{E_1}{E} b_J^{\alpha} .$$
 (8.28)

Let us also decompose the transferred momentum using p_1^{μ} , p_2^{μ} and b_1^{μ} as basis vectors,

$$Q^{\mu} = \frac{2E_2}{E}\sin^2\frac{\Theta}{2}p_1^{\mu} - \frac{2E_1}{E}\sin^2\frac{\Theta}{2}p_2^{\mu} - \frac{p}{b}\sin\Theta b_J^{\mu}.$$
(8.29)

Using (8.29) we get

$$b_1^{[\alpha}Q^{\beta]} + (A_1 + A_2)p_1^{[\alpha}Q^{\beta]} = -2\sin^2\frac{\Theta}{2}\frac{E_1E_2}{E^2}b^{[\alpha}(p_1 + p_2)^{\beta]} + 2\sin^2\frac{\Theta}{2}\tan\frac{\Theta}{2}\frac{E_1E_2}{E^2}\frac{b}{p}p_1^{[\alpha}p_2^{\beta]}.$$
(8.30)

If we now go to the frame where the center-of-mass is also initially at rest,

$$-p_1^{\alpha} = (E_1, p^l), \qquad -p_2^{\alpha} = (E_2, -p^l), \tag{8.31}$$

we see that, if we restrict to spatial components, all terms in (8.23) vanish after using (8.30),

$$\Delta L^{g}_{(ic)} = 0. \tag{8.32}$$

This tells us that indeed angular momentum is conserved separately for each particle, in this special frame. The mixed components ΔL^{0l} do not vanish, in general. However, they are actually not well defined, since they depend on the Shapiro time delay $\partial 2\delta_0/\partial E \propto \frac{1}{4-D} + \log(b/b_0)$, which is infinite in D = 4 due to the long-range nature of the gravitational force. This requires a subtraction which leaves behind an arbitrary cutoff scale b_0 in the remaining logarithm.

8.2. Coherent state approximation beyond the soft limit

As discussed in the previous section, in order to include radiation in the eikonal framework we need to introduce creation/annihilation operators describing the physical gravitons (or in general massless particles) that can be produced/absorbed in the scattering process. So far we focused on the soft limit which in our context means that the typical energy of the radiation quanta is much smaller than that of the potential gravitons exchanged between the two energetic particles: $\omega \ll \hbar v/b$. In the previous section we used the approach by Bloch–Nordsieck/Weinberg [1,311,312] to describe soft radiation as an exponential dressing of the "hard" elastic scattering as done in several papers [15,142,143,307–310,335]. However, in the spirit of the eikonal exponentiation, we expect that classical radiation exponentiates at all frequencies at least in a PM expansion. The natural guess is that, to describe frequencies $\omega \gtrsim v/b$, the functions w_j in (7.8) should be derived from the classical limit of the whole 5-point function $\mathcal{A}^{(5)}$. We will use \mathcal{W}_j to indicate these more general functions appearing in the operatorial part of the eikonal beyond the soft approximation.

We start by taking the most straightforward generalization of the approach followed in the elastic case and introduce the impact parameter representation of $A^{(5)}$ by taking its Fourier transform

$$\tilde{\mathcal{A}}^{(5)\mu\nu}(x_1, x_2, k) = \int \frac{d^D q_1}{(2\pi)^{D-2}} \,\delta(2p_1 q_1 - q_1^2) \,\delta(2p_2 q_2 - q_2^2) e^{ix_1 \cdot q_1 + ix_2 \cdot q_2} \,\mathcal{A}^{(5)\mu\nu}(q_1, q_2, k) \,, \tag{8.33}$$

where $q_1 + q_2 + k = 0$ and the delta functions enforce the on-shell condition for the energetic particles. As usual, in the classical limit we need to expand the full amplitude for small values of q_i and k with respect to hard momenta p_i since the soft momenta are proportional to \hbar in the classical regime. We expect also in the inelastic case the same pattern discussed in Section 4 for the elastic scattering: the leading tree-level contribution as $\hbar \rightarrow 0$ should exponentiate and account for the leading term in the classical limit at each loop level. Of course perturbative amplitudes at a fixed order in *G* will also have terms that scale as the leading tree-level contribution: by comparing their form with the formal expansion in *G* of the eikonal operator $\hat{\delta}$ one should be able to fix in principle the form of $\hat{\delta}$ order by order in the PM approximation. As a consistency check, contributions of order $\mathcal{O}(G^n)$ that are more divergent than the classical terms should be completely determined by lower order data. In our analysis we will focus on the leading PM contribution to the operator part of the eikonal and discuss how the approaches of Sections 7.1 and 7.2 can be generalized beyond the soft approximation. In

both cases, we take the following generalization of the elastic eikonal (8.7)

$$S|\psi\rangle = \int_{p_3} \int_{p_4} e^{-ib_1 p_4} e^{-ib_2 p_3} \int \frac{d^D Q_1}{(2\pi)^D} \int \frac{d^D Q_2}{(2\pi)^D} \Phi(p_4 - Q_1) \Phi(p_3 - Q_2) \times \int d^D x_1 \int d^D x_2 \, e^{i(b_1 - x_1)Q_1 + i(b_2 - x_2)Q_2} e^{2i\hat{\delta}(x_1, x_2)} |p_3, p_4, 0\rangle , \qquad (8.34)$$

where the operator $e^{2i\hat{\delta}}$ is determined by the 4-point and 5-point function as sketched above.

Let us stress that the final state (8.34) contains a large number of gravitons as it has the same coherent state structure of the soft eikonal operator (7.8). While this may not be surprising, since we are describing the classical radiation produced during the scattering, it is not obvious that the operator $e^{2i\delta}$ can be determined just by the classical limit of the 4 and the 5-point amplitudes order by order in *G*. It is argued in [120] that this is a consistency requirement because it ensures that the variance in the distribution of the emitted gravitons becomes negligible in the classical limit. A first check is of course to verify that the tree-level 6-point function involving two gravitons in the final state does not yield classical contributions that should be included in the eikonal. This is discussed in [120] for the technically simpler case of QED and in [336] for gravity: in both cases the assumption above holds. This supports the intuitive picture that the soft quanta are emitted one at the time when the energetic particles bend in the scattering. However, as mentioned in the outlook, this point deserves further investigation studying also the inelastic amplitudes at loop level. We expect that the eikonal operator involves also corrections that are non-linear in the creation/annihilation operators for the soft quanta. For instance, terms of this type are probably needed to capture contributions related to the Compton-like 4-point amplitude with two gravitons and two energetic particles.

8.2.1. Eikonal operator without static modes

In order to define explicitly the eikonal operator (8.34), we then need to combine the classical information extracted from the 4 and the 5-point functions. A first approach is to define [120]

$$e^{2i\hat{\delta}(x_1,x_2)} = \int \frac{d^D Q}{(2\pi)^D} \int d^D x \, e^{-iQ(x-x_1+x_2)} e^{2i\delta_s(b)} e^{i\int_k \left[\mathcal{W}_j(x_1,x_2,k)a_j^{\dagger}(k) + \mathcal{W}_j^*(x_1,x_2,k)a_j(k)\right]},\tag{8.35}$$

where $\delta_s(b)$ is related to the elastic eikonal discussed in Section 8.1, while W_j is derived from the classical limit of the 5-point amplitude. Of course both objects will have a PM expansion and at 1PM and 2PM $\delta_s(b)$ is the same as the elastic eikonal discussed in Sections 3 and 4. The leading mixed term in this expansion, $2\delta_0W_{0,j}$ in the one-loop $2 \rightarrow 3$ amplitude has been recently checked to be consistent with the exponentiation (8.35) [146–148]. We know that at 3PM δ_2 develops an imaginary part, see Section 5, but here we do not need to include it as it will be automatically generated by the operator part we are adding: it is the same mechanism that in Section 7 yielded the divergent imaginary part of δ_2 from the normal ordering of the soft eikonal operator, see (7.25). Thus we identified $\delta_s(b)$ with the real part of the elastic eikonal and, following [145], we take its dependence on the external momenta as follows,

$$2\delta_s(b) = \frac{1}{2} \left[\text{Re} \, 2\delta(\sigma_{12}, b_e) + \text{Re} \, 2\delta(\sigma_{34}, b_e) \right], \tag{8.36}$$

where we symmetrize between incoming $\sigma_{12} = -\frac{p_1 \cdot p_2}{m_1 m_2}$ and outgoing $\sigma_{34} = -\frac{p_3 \cdot p_4}{m_1 m_2}$ momenta. This last step starts being relevant only at 4PM, so beyond the scope of this review, but we will briefly comment in Section 9 on why the approach taken in (8.36) is useful.

In the radiative sector we have a similar PM expansion where $W_j = W_{0j} + W_{1j} + \cdots$ starts at order $G^{3/2}$. Since we will work at the first PM order for the operator part, we focus on W_{0j} . This is obtained simply by using the classical tree-level 5-point function (5.38) in (8.33) and then contracting it with physical polarization tensors such as the $\varepsilon_{x,+}$ introduced in (8.84). For our analysis we need just the leading term and so we can use the linearized version of the delta functions, $\delta(2p_1q_1)\delta(2p_2q_2)$, since $2p_iq_i \gg q_i^2$

$$\mathcal{W}_{0}^{\mu\nu}(x_{1}, x_{2}, k) = \int \frac{d^{D}q_{1}}{(2\pi)^{D}} e^{ix_{1} \cdot q_{1} + x_{2} \cdot q_{2}} 2\pi \,\delta(2p_{1} \cdot q_{1}) 2\pi \,\delta(2p_{2} \cdot q_{2}) \,\mathcal{A}_{0}^{(5)\mu\nu}(q_{1}, q_{2}, k) \,, \tag{8.37}$$

where, as before, $q_1 + q_2 + k = 0$ and $\mathcal{A}_0^{(5)\mu\nu}(q_1, q_2, k)$ is given in (5.38). In the same spirit of Eq. (8.36), we take $\mathcal{W}^{\mu\nu}$ to depend on \tilde{p}_1 and \tilde{p}_2 ,

$$\tilde{p}_1 = \frac{1}{2}(p_4 - p_1) = p_4 - \frac{Q_1}{2}, \qquad \tilde{p}_2 = \frac{1}{2}(p_3 - p_2) = p_3 - \frac{Q_2}{2},$$
(8.38)

rather than on p_1 and p_2 . In order to clarify the meaning of the integrations in (8.34), it is convenient to change variables as follows $x_1 = x_+ + \frac{x_-}{2}$, $x_2 = x_+ - \frac{x_-}{2}$, $Q_1 = Q_e - \frac{p}{2}$ and $Q_2 = -Q_e - \frac{p}{2}$. By rewriting (8.37) in terms of x_{\pm} we see that W_0 depends on x_+ only through an overall factor of e^{-ix_+k} . When using this fact in (8.34), one can see that the integration over x_+ simply implies that P is equal to the sum of the momenta of the emitted gravitons, as it can be seen by expanding

the last exponential with the creation modes. Therefore, at a generic order $(a^{\dagger})^M$, we have $P = \sum_{m=1}^M k_m$. The factorized dependence on x_+ implies also the following transformation property under translations

$$x_{1,2}^{\mu} \to x_{1,2}^{\mu} + a^{\mu}, \qquad \mathcal{W}_{0}^{\mu\nu} \to e^{-ia\cdot k} \mathcal{W}_{0}^{\mu\nu}.$$
 (8.39)

The same property will hold also at the subleading PM orders as long as we start from a definition with the functional form of (8.37).

It is instructive to use the stationary phase approach to calculate the norm of the final state $\langle \psi | S^{\dagger}S | \psi \rangle$. Classical unitarity implies that it should be equal to the norm of the initial state $\langle \psi | \psi \rangle$ and here we would like to see that at least all large phases cancel at the stationary point. By using (8.34) and (8.35) we have

$$\langle \psi | S^{\dagger} S | \psi \rangle = \int_{P_{3}, P_{4}} \int \frac{d^{D} Q_{i}'}{(2\pi)^{D}} d^{D} x_{i}' \int \frac{d^{D} Q}{(2\pi)^{D}} d^{D} x' \, e^{-i \sum_{r} (b_{r} - x_{r}') \cdot Q_{r}' + iQ'(x' - x_{1}' + x_{2}') - i2\delta_{S}(b')} \\ \times \int \frac{d^{D} Q_{i}}{(2\pi)^{D}} d^{D} x_{i} \int \frac{d^{D} Q}{(2\pi)^{D}} d^{D} x \, e^{i \sum_{r} (b_{r} - x_{r}) \cdot Q_{r} - iQ(x - x_{1} + x_{2}) + i2\delta_{S}(b)} \\ \times e^{-\frac{1}{2} \int_{k} \mathcal{W}_{j}^{*}(x_{1}, x_{2}, k) \mathcal{W}_{j}(x_{1}, x_{2}, k) - \frac{1}{2} \int_{k} \mathcal{W}_{j}^{*}(x_{1}', x_{2}', k) \mathcal{W}_{j}(x_{1}', x_{2}', k) \mathcal{W}_{j}(x_{1}', x_{2}', k) \mathcal{W}_{j}(x_{1}, x_{2}, k)} \\ \times \Phi_{1}^{*}(p_{4} - Q_{1}') \Phi_{2}^{*}(p_{3} - Q_{2}') \Phi_{1}(p_{4} - Q_{1}) \Phi_{2}(p_{3} - Q_{2}) \,.$$

The stationary conditions for Q_i and Q'_i yield

$$\begin{aligned} x_{i\mu} &= b_{i\mu} + \frac{\partial 2\delta_{s}(b)}{\partial Q_{i}^{\mu}} + \int_{k} \left[\frac{i}{2} \frac{\partial (\mathcal{W}_{j}^{*}(x_{1}, x_{2}, k)\mathcal{W}_{j}(x_{1}, x_{2}, k))}{\partial Q_{i}^{\mu}} - i \mathcal{W}_{j}^{*}(x_{1}', x_{2}', k) \frac{\partial \mathcal{W}_{j}(x_{1}, x_{2}, k)}{\partial Q_{i}^{\mu}} \right] \\ x_{i\mu}' &= b_{i\mu} + \frac{\partial 2\delta_{s}(b')}{\partial Q_{i}'^{\mu}} - \int_{k} \left[\frac{i}{2} \frac{\partial (\mathcal{W}_{j}^{*}(x_{1}', x_{2}', k)\mathcal{W}_{j}(x_{1}', x_{2}', k))}{\partial Q_{i}'^{\mu}} - i \frac{\partial \mathcal{W}_{j}^{*}(x_{1}', x_{2}', k)}{\partial Q_{i}'^{\mu}} \mathcal{W}_{j}(x_{1}, x_{2}, k) \right]. \end{aligned}$$
(8.41)

Similarly from the variation of x_i and x'_i , we have

$$Q_{i\mu} = (-1)^{i+1}Q_{\mu} + \int_{k} \left[\frac{i}{2} \frac{\partial (\mathcal{W}_{j}^{*}(x_{1}, x_{2}, k)\mathcal{W}_{j}(x_{1}, x_{2}, k))}{\partial x_{i}^{\mu}} - i \mathcal{W}_{j}^{*}(x_{1}', x_{2}', k) \frac{\partial \mathcal{W}_{j}(x_{1}, x_{2}, k)}{\partial x_{i}^{\mu}} \right]$$

$$Q_{i\mu}' = (-1)^{i+1}Q_{\mu}' - \int_{k} \left[\frac{i}{2} \frac{\partial (\mathcal{W}_{j}^{*}(x_{1}', x_{2}', k)\mathcal{W}_{j}(x_{1}', x_{2}', k))}{\partial x_{i}^{\mu}} - i \frac{\partial \mathcal{W}_{j}^{*}(x_{1}', x_{2}', k)}{\partial x_{i}^{\mu}} \mathcal{W}_{j}(x_{1}, x_{2}, k) \right]$$
(8.42)

and from the variations of x, x', Q and Q'

$$x_{\mu} = (x_1 - x_2)_{\mu} + \frac{\partial 2\delta_s(b)}{\partial Q^{\mu}}, \quad Q_{\mu} = \frac{\partial 2\delta_s(b)}{\partial x^{\mu}}, \quad (8.43)$$

$$x'_{\mu} = (x'_{1} - x'_{2})_{\mu} + \frac{\partial 2\delta_{s}(b')}{\partial Q'^{\mu}}, \quad Q'_{\mu} = \frac{\partial 2\delta_{s}(b')}{\partial x'^{\mu}}.$$
(8.44)

We can satisfy these equations by requiring

$$x'_{\mu} = x_{\mu} = (x_1 - x_2)_{\mu} + \frac{\partial 2\delta_s(b)}{\partial Q^{\mu}}, \quad Q'_{\mu} = Q_{\mu} = \frac{\partial 2\delta_s(b)}{\partial x^{\mu}},$$

$$Q'_{i\mu} = Q_{i\mu} = (-1)^{i+1}Q_{\mu} - i\int_k W_j^*(x_1, x_2, k) \frac{\partial}{\partial x_i^{\mu}} W_j(x_1, x_2, k),$$

(8.45)

$$(x'_i - b_i)_{\mu} = (x_i - b_i)_{\mu} = \frac{\partial 2\delta(b)}{\partial Q_i^{\mu}} - i \int_k \mathcal{W}_j^*(x_1, x_2, k) \frac{\overleftrightarrow{\partial}}{\partial Q_i^{\mu}} \mathcal{W}_j(x_1, x_2, k),$$

where $f \partial g = (f \partial g - g \partial f)/2$. When (8.45) are satisfied, all large phases cancel in the evaluation of $\langle \psi | S^{\dagger}S | \psi \rangle$, so they define the classical values of various quantities. For instance, we can use (8.45) and the relations

$$p_1 + p_4 = Q_1 , \quad p_2 + p_3 = Q_2 ,$$
 (8.46)

which follow from the presence of the wavepackets Φ_i , to express Q_i in terms of the initial data $b_{1,2}$ and $p_{1,2}$.

As mentioned after (8.38), some of the integrals could have been performed exactly, but we preferred to treat all the variables on the same footing. By summing the results for the Q_i 's and using the factorized dependence on x_+ of W, we recover

$$Q_1^{\mu} + Q_2^{\mu} = -\int_k \mathcal{W}_j^*(x_1, x_2, k) k^{\mu} \mathcal{W}_j(x_1, x_2, k) = -P^{\mu}$$
(8.47)

ensuring momentum conservation.

8.2.2. Eikonal operator including static modes

As we saw in Section 7, for some applications it is convenient to include in the eikonal operator also the static contribution arising from the zero-frequency modes and this can be easily done also by following the formalism discussed in the previous section. To this end we introduce an auxiliary frequency scale ω^* such that the full result is well approximated by the soft limit for frequencies below ω^* . Then we can write the eikonal operator simply by splitting the operator part in two terms, one for $\omega > \omega^*$, where it coincides with the formulation in (8.35), and the other for $\omega < \omega^*$ for which we follow the approach of Section 7. This splitting is useful only in the intermediate steps so to implement the dressing with the zero-frequency modes discussed in Section 7.2, but on the final results we will take the limit $\omega^* \rightarrow 0$. Thus the differences between this approach and that of the previous section are limited to the static contributions related to zero-frequency contributions and the -i0 prescription, see the comments after Eq. (7.45). In formulae we have

$$e^{2i\hat{\delta}(x_{1},x_{2})} = \int \frac{d^{D}Q}{(2\pi)^{D}} \int d^{D}x \, e^{-iQ(x-x_{1}+x_{2})} e^{i2\delta_{S}(b)}$$

$$\times e^{\int_{k} \theta(\omega^{*}-k^{0}) \left[\int_{j}^{\text{out}} a_{j}^{\dagger} - \int_{j}^{\text{out}} a_{j} \right]} e^{-\int_{k} \theta(\omega^{*}-k^{0}) \left[\int_{j}^{\text{in}} a_{j}^{\dagger} - \int_{j}^{\text{in}} a_{j} \right]}$$

$$\times e^{i\int_{k} \theta(k^{0}-\omega^{*}) \left[\mathcal{W}_{j}(x_{1},x_{2},k)a_{j}^{\dagger}(k) + \mathcal{W}_{j}^{*}(x_{1},x_{2},k)a_{j}(k) \right]}.$$
(8.48)

We can follow the same steps of Section 7.2 and combine the oscillators in the second line of (8.48) into a single exponential. This is equivalent to simply redefining the eikonal phase as in (7.51), that is

$$2i\delta(b) = 2i\delta_s(b) - 2i\delta^{\text{dr.}}(b) \tag{8.49}$$

with

$$2i\delta^{\text{dr.}}(b) = -\frac{1}{2} \int_{k}^{\omega} (f_{j}^{\text{out}*}(k)f_{j}^{\text{in}}(k) - f_{j}^{\text{in}*}(k)f_{j}^{\text{out}}(k)) = \frac{i}{4} GQ_{1PM}^{2} \mathcal{I}(\sigma), \qquad (8.50)$$

where $2\delta^{dr.}(b)$ was derived in Eqs. (7.51) and (7.63), and has been reported again here just for convenience after using (7.88). Thus we can rewrite (8.48) as

$$e^{2i\hat{\delta}(x_{1},x_{2})} = \int \frac{d^{D}Q}{(2\pi)^{D}} \int d^{D}x \, e^{-iQ(x-x_{1}+x_{2})} e^{i2\tilde{\delta}(b)} \times e^{\int_{k} \theta(\omega^{*}-k^{0}) \left[f_{j}a_{j}(k)^{\dagger} - f_{j}^{*}(k)a_{j}(k) \right]} e^{i\int_{k} \theta(k^{0} - \omega^{*}) \left[\mathcal{W}_{j}(x_{1},x_{2},k)a_{j}^{\dagger}(k) + \mathcal{W}_{j}^{*}(x_{1},x_{2},k)a_{j}(k) \right]},$$
(8.51)

where $f_j(k) = f_j^{\text{out}}(k) - f_j^{\text{in}}(k)$. The operator (8.51) has to be used in (8.34) to obtain the full eikonal operator. The discussion of classical unitarity for this eikonal operator follows the same steps discussed from Eq. (8.40). One can check that the various integrals are dominated, in the classical limit, by the following values

$$x_{\mu} = (x_1 - x_2)_{\mu} + \frac{\partial 2\tilde{\delta}(b)}{\partial Q^{\mu}}, \qquad Q_{\mu} = \frac{\partial 2\tilde{\delta}(b)}{\partial x^{\mu}}, \qquad (8.52a)$$

$$Q_{i\mu} = (-1)^{i+1} Q_{\mu} - i \int_{k} \mathcal{W}_{j}^{*}(x_{1}, x_{2}, k) \frac{\stackrel{\leftrightarrow}{\partial}}{\partial x_{i}^{\mu}} \mathcal{W}_{j}(x_{1}, x_{2}, k) , \qquad (8.52b)$$

$$(\mathbf{x}_{i} - b_{i})_{\mu} = \frac{\partial 2\tilde{\delta}(b)}{\partial Q_{i}} - i \int_{k} \theta(\omega^{*} - k^{0}) f_{j}^{*}(k) \frac{\overleftrightarrow{\partial}}{\partial Q_{i}^{\mu}} f_{j}(k) - i \int_{k} \theta(k^{0} - \omega^{*}) \mathcal{W}_{j}^{*}(\mathbf{x}_{1}, \mathbf{x}_{2}, k) \frac{\overleftrightarrow{\partial}}{\partial Q_{i}^{\mu}} \mathcal{W}_{j}(\mathbf{x}_{1}, \mathbf{x}_{2}, k),$$

$$(8.52c)$$

where one should use (8.36) and (8.38).

8.2.3. The $\mathcal{N} = 8$ eikonal operator

It is straightforward to construct the eikonal operator (8.35) or (8.51) relevant to the classical limit of $\mathcal{N} = 8$ amplitudes for $2 \rightarrow 2$ processes involving massive scalars plus additional emissions of gravitons, dilatons, and KK scalars and vectors. For the elastic eikonal phase, one should use the $\mathcal{N} = 8$ result which we derived in the previous sections up to $\mathcal{O}(G^3)$. For the inelastic portion of the operator, instead, one can introduce new ladder operators associated to the new emitted massless states, as discussed in Section 7.4, paired with appropriate $2 \rightarrow 3$ amplitudes for the emission of such states. However, it is more convenient to adopt the approach of Section 5 and simply promote the appropriate Lorentz indices μ, ν, \ldots to 10-dimensional ones M, N, \ldots , employing of course the $\mathcal{N} = 8$ coupling in Eq. (E.22). This effectively captures all possible types of emissions and exchanges relevant to the classical limit.

8.3. Computing observables from the eikonal operator

In this section we illustrate how the eikonal operator allows us to calculate in a systematic way observable quantities that characterize gravitational collisions. The general strategy, as already discussed in the soft limit based on Eq. (7.13), is to take expectation values of the appropriate operators in the final state obtained by acting with the eikonal operator on the initial state. Since in this way the final state now includes radiation, this will allow us to obtain not only the deflection angle, but also the leading-order (L.O.) gravitational waveform as well as the 3PM linear and angular momenta of the gravitational field generated by the collision. Moreover, since the operator incorporates both field and the massive degrees of freedom in a dynamical fashion, it will also provide expressions for the corresponding changes of mechanical linear and angular momenta of each particle taking part in the collision that explicitly obey the corresponding balance laws.

8.3.1. Waveforms

As we discussed in order to obtain (7.17) in the zero-frequency limit, starting from the canonically normalized field operator $H_{\mu\nu}(x)$ and taking its expectation value in the final state dictated by the eikonal operator, we obtain the metric fluctuation $W_{\mu\nu}(x)$,

$$g_{\mu\nu}(x) = \eta_{\mu\nu} + 2W_{\mu\nu}(x). \tag{8.53}$$

Proceeding in the same way, but using this time the eikonal operator (8.35), we find the following waveform in D = 4 that goes beyond the soft approximation

$$W^{\mu\nu}(x) = \int_{-\infty}^{+\infty} \frac{d\omega}{2\pi} e^{-i\omega u} \widehat{W}^{\mu\nu}\left(\omega(1,\hat{x})\right), \qquad \widehat{W}^{\mu\nu}(k) = \frac{2G}{r} \frac{\mathcal{W}^{\mu\nu}}{\sqrt{8\pi G}},$$
(8.54)

where ω is the frequency measured by an asymptotic detector, u is the corresponding retarded time and $(1, \hat{x})^{\mu}$ a null vector characterizing its direction with respect to the source. To obtain the L.O. waveform, we may employ $W_0^{\mu\nu}$ given in (8.37), which is the impact-parameter version of the classical $2 \rightarrow 3$ amplitude, with the further approximation $x_{1,2}^{\alpha} \simeq b_{1,2}^{\alpha}$ of the saddle-point conditions (8.45). Using the kinematics conventions given in Section 1.2 (see in particular Eq. (1.10), and Eqs. (1.43) and following), we may cast it in the form

$$\mathcal{W}_{0}^{\mu\nu}(k) = \frac{1}{4m_{1}m_{2}} \int \frac{d^{4}q_{1}}{(2\pi)^{4}} e^{ib_{1}\cdot q_{1} + ib_{2}\cdot q_{2}} 2\pi \,\delta(v_{1}\cdot q_{1}) 2\pi \,\delta(v_{2}\cdot q_{2}) \mathcal{A}_{0}^{(5)\mu\nu}(q_{1}, q_{2}, k) \tag{8.55}$$

where $q_1 + q_2 + k = 0$. In this way we will recover the frequency-domain expression $\widehat{W}^{\mu\nu}(k)$ for the waveform [129]. Performing the frequency Fourier transform (8.54) leads to the expression $W^{\mu\nu}$ for the metric fluctuation in (retarded) time domain [128,337,338], which however we will not discuss explicitly here.

The starting point for computing the LO. waveform is thus the five-point amplitude $\mathcal{A}_0^{(5)\mu\nu}$ in the classical limit given in Eq. (5.38). We find it convenient to rewrite the theory-dependent coefficient β by factoring out the mass dependence according to

$$\beta = 2m_1^2 m_2^2 c_0 \,, \tag{8.56}$$

so that, in D = 4,

$$\beta_{GR} = 4m_1^2 m_2^2 \left(\sigma^2 - \frac{1}{2}\right), \qquad \beta_{\mathcal{N}=8} = 4m_1^2 m_2^2 \sigma^2.$$
(8.57)

and

$$c_0^{\text{GR}} = 2\sigma^2 - 1, \qquad c_0^{\mathcal{N}=8} = 2\sigma^2.$$
 (8.58)

We also recall that Eq. (5.38) is symmetric $A_0^{(5)\mu\nu} = A_0^{(5)\nu\mu}$ and encodes gauge invariance via the important property

$$k_{\mu}\mathcal{A}_{0}^{(5)\mu\nu} = 0.$$
(8.59)

In order to perform the Fourier transform, it can be useful to isolate the terms in $A_0^{(5)\mu\nu}$ that have either $1/q_1^2$ or $1/q_2^2$ from those that have both factors, writing

$$\mathcal{A}_{0}^{(5)\mu\nu} = \frac{1}{q_{1}^{2}} \mathcal{A}_{0,1}^{(5)\mu\nu} + \frac{1}{q_{2}^{2}} \mathcal{A}_{0,2}^{(5)\mu\nu} + \frac{1}{q_{1}^{2} q_{2}^{2}} \mathcal{A}_{0,\mathrm{irr}}^{(5)\mu\nu}$$
(8.60)

where $\mathcal{A}_{0,1}^{(5)\mu\nu}$, $\mathcal{A}_{0,2}^{(5)\mu\nu}$, $\mathcal{A}_{0,irr}^{(5)\mu\nu}$ only contain q_1 and q_2 in the numerators. For terms of the first kind, we may solve $q_2 = -q_1 - k$ in the integrand of (8.55), obtaining

$$\mathcal{W}_{0,1}^{\mu\nu}(k) = \frac{e^{-ib_2 \cdot k}}{4m_1m_2} \int \frac{d^4q_1}{(2\pi)^4} e^{ib \cdot q_1} 2\pi \,\delta(v_1 \cdot q_1) 2\pi \,\delta(v_2 \cdot q_1 - \omega_2) \frac{1}{q_1^2} \mathcal{A}_{0,1}^{(5)\mu\nu} \Big|_{q_2 = -q_1 - k} \tag{8.61}$$

with $b^{\alpha} = b_1^{\alpha} - b_2^{\alpha}$. Performing the decomposition (1.10) of the integrated momentum q_1^{μ} and taking the Jacobian factor $1/\sqrt{\sigma^2 - 1}$ into account (see also Appendix A.3), we see that the delta functions set $q_{1\parallel 1} = 0$ and $q_{1\parallel 2} = -\omega_2$ (as already noted in (1.48)) so that

$$\mathcal{W}_{0,1}^{\mu\nu}(k) = \frac{e^{-ib_2 \cdot k}}{4m_1 m_2 \sqrt{\sigma^2 - 1}} \int \frac{d^2 q_{1\perp}}{(2\pi)^2} e^{ib \cdot q_{1\perp}} \frac{1}{q_{1\perp}^2 + \frac{\omega_2^2}{\sigma^2 - 1}} \mathcal{A}_{0,1}^{(5)\mu\nu} \Big|_{q_1}^{q_2} = -q_1 - k \tag{8.62}$$

and this integral can be evaluated in terms of a modified Bessel function of the second kind $K_0(z)$,

$$\int \frac{d^2 q_{1\perp}}{(2\pi)^2} \frac{e^{ib \cdot q_{1\perp}}}{q_{1\perp}^2 + \frac{\omega_2^2}{\sigma^2 - 1}} = \frac{1}{2\pi} K_0(\Omega_2),$$
(8.63)

where we introduced the dimensionless combinations

$$\Omega_1 = \frac{\omega_1 b}{\sqrt{\sigma^2 - 1}}, \qquad \Omega_2 = \frac{\omega_2 b}{\sqrt{\sigma^2 - 1}}, \tag{8.64}$$

so that

$$\mathcal{W}_{0,1}^{\mu\nu}(k) = \frac{e^{-ib_2 \cdot k}}{8\pi m_1 m_2 \sqrt{\sigma^2 - 1}} \mathcal{A}_{0,1}^{(5)\mu\nu} \Big|_{\substack{q_2 = -q_1 - k \\ q_1 = -\omega_2 \check{\nu}_2 + q_{1\perp}}} K_0\left(\Omega_2\right).$$

$$(8.65)$$

In the last step we have used the identity $q_{1\perp}^{\mu}e^{ib\cdot q_{1\perp}} = -i\partial_b^{\mu}e^{ib\cdot q_{1\perp}}$ to rewrite all instances of $q_{1\perp}^{\mu}$ appearing in $\mathcal{A}_{0,1}^{(5)\mu\nu}$ as derivatives with respect to the impact parameter b^{μ} . The discussion of the second kind of terms in (8.60) is entirely analogous, with the only difference that for those it is more convenient to solve the momentum conservation condition as $q_1^{\mu} = -q_2^{\mu} - k^{\mu}$, so that

$$\mathcal{W}_{0,2}^{\mu\nu}(k) = \frac{e^{-ib_1 \cdot k}}{8\pi m_1 m_2 \sqrt{\sigma^2 - 1}} \mathcal{A}_{0,2}^{(5)\mu\nu} \Big|_{\substack{q_1 = -q_2 - k \\ q_2 = -\omega_1 \check{\nu}_1 + q_{2\perp}}} K_0(\Omega_1) \,. \tag{8.66}$$

Finally, we turn to the calculation of the third kind of terms in (8.60), which, proceeding as above, can be cast in the form

$$\mathcal{W}_{0,\mathrm{irr}}^{\mu\nu}(k) = \frac{e^{-ib_2 \cdot k}}{4m_1 m_2 \sqrt{\sigma^2 - 1}} \int \frac{d^2 q_{1\perp}}{(2\pi)^2} e^{ib \cdot q_{1\perp}} \frac{\mathcal{A}_{0,\mathrm{irr}}^{(5)\mu\nu} \Big|_{q_2} = -q_1 - k}{\left((q_{1\perp} + k_{\perp})^2 + \frac{\omega_1^2}{\sigma^2 - 1}\right) \left(q_{1\perp}^2 + \frac{\omega_2^2}{\sigma^2 - 1}\right)}$$
(8.67)

where $\mathcal{P} = -\omega_1^2 + 2\omega_1\omega_2\sigma - \omega_2^2$ as in (1.45). The basic integral to be performed in this case is

$$\int \frac{d^2 q_{1\perp}}{(2\pi)^2} \frac{e^{ib \cdot q_{1\perp}}}{\left((q_{1\perp} + k_{\perp})^2 + \frac{\omega_1^2}{\sigma^2 - 1}\right) \left(q_{1\perp}^2 + \frac{\omega_2^2}{\sigma^2 - 1}\right)} \equiv \frac{1}{2\pi} H \,. \tag{8.68}$$

Introducing Schwinger parameters, one finds

$$H = \int_{\mathbb{R}^2_+} \frac{dx_1 dx_2}{2(x_1 + x_2)} e^{-\frac{1}{x_1 + x_2} \left(\frac{b^2}{4} + ib \cdot kx_1 + \frac{x_1^2 \omega_1^2 + 2\sigma x_1 x_2 \omega_1 \omega_2 + x_2^2 \omega_2^2}{\sigma^2 - 1}\right)}.$$
(8.69)

Letting $x_1 = b^2 x/(4\lambda)$, $x_2 = b^2 y/(4\lambda)$ with Feynman parameters x, y obeying x + y = 1, and using the integral representation

$$\frac{K_1(z)}{z} = \int_0^\infty e^{-\lambda - \frac{z^2}{4\lambda}} \frac{d\lambda}{4\lambda^2},$$
(8.70)

for the modified Bessel function of the second kind $K_1(z)$, one can express H as the following parametric integral,

$$H = \frac{b^2}{2} \int_0^1 e^{-ixb \cdot k} \frac{K_1(\Omega(x))}{\Omega(x)} dx$$
(8.71)

where, again letting x + y = 1,

$$\Omega(x) = \sqrt{\Omega_1^2 x^2 + 2\Omega_1 \Omega_2 \sigma x y + \Omega_2^2 y^2}.$$
(8.72)

Therefore,

$$\mathcal{W}_{0,\mathrm{irr}}^{\mu\nu}(k) = \frac{e^{-ib_{2}\cdot k}}{16\pi m_{1}m_{2}\sqrt{\sigma^{2}-1}} \mathcal{A}_{0,\mathrm{irr}}^{(5)\mu\nu} \Big|_{\substack{q_{2} = -q_{1}-k\\q_{1} = -\omega_{2}\check{v}_{2}+q_{1}\perp}} \left[b^{2} \int_{0}^{1} e^{-ixb\cdot k} \frac{K_{1}(\Omega(x))}{\Omega(x)} dx \right].$$

$$(8.73)$$

To perform the remaining *b*-derivatives in (8.65), (8.66) and (8.73), it can be useful to recall that

$$K'_0(z) = -K_1(z), \qquad (zK_1(z))' = -zK_0(z).$$
(8.74)

Proceeding in this way, and collecting the three pieces according to (8.60), we obtain the following result for the covariant frequency-domain waveform (8.54), letting

$$\widehat{W}_{0}^{\mu\nu}(k) = \widehat{W}_{0,12}^{\mu\nu}(k) + \widehat{W}_{0,\text{irr}}^{\mu\nu}(k).$$
(8.75)

Then

$$\widehat{W}_{0,12}^{\mu\nu}(k) = \frac{2G^2 m_1 m_2}{br\omega_1 \omega_2(\sigma^2 - 1)} \sum_{j=1,2} \left[A_j^{\mu\nu} e^{-ib_j \cdot k} \frac{bK_0(\Omega_j)}{\sqrt{\sigma^2 - 1}} - 2ic_0 B_j^{\mu\nu} e^{-ib_j \cdot k} K_1(\Omega_j) \right]$$
(8.76)

with

$$A_{1}^{\mu\nu} = c_{0} \left(\sigma^{2} - 1\right) v_{1}^{(\mu} k^{\nu)} \omega_{2} + v_{1}^{\mu} v_{1}^{\nu} \omega_{2} \left(2\sigma \omega_{2} \left(c_{0} - 4(\sigma^{2} - 1)\right) + 2c_{0} \omega_{1}\right) + 2\sigma v_{1}^{(\mu} v_{2}^{\nu)} \omega_{1} \omega_{2} \left(2(\sigma^{2} - 1) - c_{0}\right) ,$$

$$B_{1}^{\mu\nu} = (b \cdot k) v_{1}^{\mu} v_{1}^{\nu} \omega_{2} + v_{1}^{(\mu} b^{\nu)} \omega_{1} \omega_{2}$$
(8.77)

and $A_2^{\mu\nu}$, $B_2^{\mu\nu}$ are obtained by interchanging particle labels $(v_1^{\mu} \leftrightarrow v_2^{\mu}, b_1^{\mu} \leftrightarrow b_2^{\mu})$ and hence $b^{\mu} \leftrightarrow -b^{\mu}$. Here and in the following $\xi^{(\mu}\xi'^{\mu')} = \xi^{\mu}\xi'^{\mu'} + \xi^{\mu'}\xi'^{\mu}$ without additional factors. Moreover,

$$\widehat{W}_{0,\text{irr}}^{\mu\nu}(k) = \frac{2G^2m_1m_2}{r(\sigma^2 - 1)^{3/2}} \int_0^1 e^{-ib(x)\cdot k} dx \Big[C^{\mu\nu} K_0(\Omega(x)) + D^{\mu\nu} \frac{K_1(\Omega(x))}{\Omega(x)} \Big]$$
(8.78)

where

$$b^{\mu}(x) = xb_1^{\mu} + yb_2^{\mu}$$
(8.79)

(let us recall that y = 1 - x) together with

$$C^{\mu\nu} = c_0(\sigma^2 - 1) \left[2\eta^{\mu\nu} - i(x - y)b^{(\mu}k^{\nu)} \right] - 2c_0 \left(v_1^{\mu}v_1^{\nu} - \sigma v_1^{(\mu}v_2^{\nu)} + v_2^{\mu}v_2^{\nu} \right) -ib^{(\mu}v_1^{\nu)} \left(2\sigma\omega_1 \left(2\left(\sigma^2 - 1\right) - c_0y \right) - 2c_0x\omega_2 \right) - iv_2^{(\mu}b^{\nu)} \left(2\sigma\omega_2 \left(2\left(\sigma^2 - 1\right) - c_0x \right) - 2c_0y\omega_1 \right),$$
(8.80)

and

$$D^{\mu\nu} = c_0(\sigma^2 - 1) \Big(\frac{b^2}{2} (x - y)^2 k^{\mu} k^{\nu} - \frac{2b^{\mu} b^{\nu}}{b^2} \Omega(x)^2 \Big)$$

$$+ b^2 v_1^{(\mu} k^{\nu)} (x - y) (c_0 x \omega_1 + \sigma \omega_2 (c_0 y - 2(\sigma^2 - 1)))$$

$$+ b^2 v_2^{(\mu} k^{\nu)} (y - x) (c_0 y \omega_2 + \sigma \omega_1 (c_0 x - 2(\sigma^2 - 1)))$$

$$+ \frac{2b^2 v_1^{\mu} v_1^{\nu}}{\sigma^2 - 1} (c_0 (x \omega_1 + \sigma y \omega_2)^2 + 2 (\sigma^2 - 1) \omega_2 (\omega_2 (\sigma^2 (x - y) - 1) - 2\sigma x \omega_1))$$

$$+ \frac{2b^2 v_2^{\mu} v_2^{\nu}}{\sigma^2 - 1} (c_0 (y \omega_2 + \sigma x \omega_1)^2 + 2 (\sigma^2 - 1) \omega_1 (\omega_1 (\sigma^2 (y - x) - 1) - 2\sigma y \omega_2))$$

$$+ \frac{2b^2 v_1^{(\mu} v_2^{\nu)}}{\sigma^2 - 1} (\sigma x \omega_1^2 (2 (\sigma^2 - 1) - c_0 x)$$

$$+ \omega_1 \omega_2 (2 (\sigma^2 - 1) - c_0 (\sigma^2 + 1) xy) + \sigma y \omega_2^2 (2 (\sigma^2 - 1) - c_0 y) \Big).$$
(8.81)

As a cross check, one can verify that $k_{\mu}W_{0}^{\mu\nu}(k) = 0$ by integrating by parts with respect to *x*, using $b \cdot ke^{ixb \cdot k} = -i\partial_{x}e^{ixb \cdot k}$. This causes the terms arising from integration by parts to cancel against the remaining integrated terms, while the boundary terms cancel against the non-integrated ones. The expressions we obtained in this way are valid for both GR and $\mathcal{N} = 8$ provided one takes into account the appropriate conventions discussed below Eq. (5.38).

In order to extract two gravitational physical degrees of freedom encoded in the waveform, we construct a pair of orthogonal polarization tensors as follows. We keep c_0 generic so as to capture both gravitational-wave emissions in GR and in $\mathcal{N} = 8$ supergravity. We first introduce reference vectors \tilde{e}^{μ}_{θ} and e^{μ}_{ϕ} such that (more explicit expressions for such reference vectors are given below in Eq. (8.91) and following)

$$\tilde{e}_{\theta} \cdot \tilde{e}_{\theta} = 1 = e_{\phi} \cdot e_{\phi} , \qquad \tilde{e}_{\theta} \cdot e_{\phi} = 0 , \qquad \tilde{e}_{\theta} \cdot k = e_{\phi} \cdot k = 0$$
(8.82)

and

$$\tilde{e}_{\theta} \cdot b = 0, \qquad e_{\phi} \cdot v_i = 0, \qquad -\frac{v_1 \cdot \tilde{e}_{\theta}}{\sigma \omega_1 - \omega_2} = \frac{v_2 \cdot \tilde{e}_{\theta}}{\sigma \omega_2 - \omega_1} = \frac{1}{\sqrt{\mathcal{P}}}.$$
(8.83)

Let us recall that \mathcal{P} had been introduced in (1.45) and is nonnegative. We then construct the following transverse-traceless polarization tensors

$$\varepsilon_{\times}^{\mu\nu} = \frac{1}{2} (\tilde{e}_{\theta}^{\mu} e_{\phi}^{\nu} + \tilde{e}_{\theta}^{\nu} e_{\phi}^{\mu}), \qquad \varepsilon_{+}^{\mu\nu} = \frac{1}{2} (\tilde{e}_{\theta}^{\mu} \tilde{e}_{\theta}^{\mu} - e_{\phi}^{\mu} e_{\phi}^{\nu})$$
(8.84)

and define

$$\widehat{W}_{\times}(k) = \varepsilon_{\times\mu\nu} \widehat{W}_{0}^{\mu\nu}(k), \qquad \widehat{W}_{+}(k) = \varepsilon_{+\mu\nu} \widehat{W}_{0}^{\mu\nu}(k).$$
(8.85)

Isolating for convenience two contributions for each polarization according to

$$\widehat{W}_{\times/+}(k) = \widehat{W}_{12,\times/+}(k) + \widehat{W}_{\text{irr},\times/+}(k), \qquad (8.86)$$

we find the following expressions. For the \times polarization,

$$\widehat{W}_{12,\times}(k) = -\frac{4iG^2m_1m_2c_0}{br(\sigma^2 - 1)} b \cdot e_{\phi} \left(e^{-ib_1 \cdot k} K_1\left(\Omega_1\right) v_1 \cdot \tilde{e}_{\theta} - e^{-ib_2 \cdot k} K_1\left(\Omega_2\right) v_2 \cdot \tilde{e}_{\theta} \right)$$

$$(8.87)$$

and

$$\widehat{W}_{\rm irr,\times}(k) = \frac{4iG^2m_1m_2}{r\sqrt{\sigma^2 - 1}} \left(\frac{c_0\omega_1\omega_2}{\sqrt{\mathcal{P}}} - 2\sigma\sqrt{\mathcal{P}}\right) b \cdot e_\phi \int_0^1 e^{-ib(x)\cdot k} K_0(\Omega(x)) dx$$
(8.88)

with $b^{\mu}(x)$, $\Omega(x)$ as in (8.72). For the + polarization, we find

$$W_{12,+}(k) = \frac{2G^2m_1m_2}{r\omega_1\omega_2(\sigma^2-1)} \Big[i\frac{b\cdot k}{b} c_0 \left(e^{-ib_2\cdot k} K_1(\Omega_2) \left(v_2 \cdot \tilde{e}_{\theta} \right)^2 \omega_1 - e^{-ib_1\cdot k} K_1(\Omega_1) \left(v_1 \cdot \tilde{e}_{\theta} \right)^2 \omega_2 \right)$$

$$+ \frac{e^{-ib_1\cdot k} K_0(\Omega_1) v_1 \cdot \tilde{e}_{\theta} \omega_2 - e^{-ib_2\cdot k} K_0(\Omega_2) v_2 \cdot \tilde{e}_{\theta} \omega_1}{\sqrt{\sigma^2 - 1}\sqrt{\mathcal{P}}} \left(\left(\sigma^2 - 1 \right) \left(4\mathcal{P}\sigma - c_0 \omega_1 \omega_2 \right) - c_0 \mathcal{P}\sigma \right) \Big]$$

$$(8.89)$$

and finally

$$\widehat{W}_{\text{irr,+}}(k) = \frac{2G^2m_1m_2}{r\sqrt{\sigma^2 - 1}} \int_0^1 dx \, e^{-ib(x) \cdot k} \left[\frac{(b \cdot e_{\phi})^2}{b^2} c_0 K_1(\Omega(x)) \Omega(x) - c_0 K_0(\Omega(x)) \right. \\ \left. + \frac{b^2 K_1(\Omega(x))}{\Omega(x)\mathcal{P}} \left(c_0 \omega_1^2 \omega_2^2 + 2\mathcal{P}^2 - 4\sigma \omega_1 \omega_2 \mathcal{P} \right) \right].$$

$$(8.90)$$

Introducing an additional vector \check{b}^{μ} satisfying $\check{b}^2 = b^2$ and $b \cdot \check{u}_{1,2} = b \cdot \check{b} = 0$, the explicit decomposition of the reference vectors \tilde{e}^{μ}_{θ} , e^{μ}_{ϕ} in terms of the basis vectors v^{μ}_1 , v^{μ}_2 , b^{μ} , \check{b}^{μ} reads

$$\tilde{e}^{\mu}_{\theta} = \frac{\omega_1 v_2^{\mu} - \omega_2 v_1^{\mu}}{\sqrt{\mathcal{P}}}, \qquad e^{\mu}_{\phi} = \frac{(\check{b} \cdot k) b^{\mu} - (b \cdot k) \check{b}^{\mu}}{b \sqrt{(b \cdot k)^2 + (\check{b} \cdot k)^2}}.$$
(8.91)

In the center-of-mass (CM) frame where

$$v_1^{\mu} = \frac{1}{m_1}(E_1, 0, 0, p), \quad v_2^{\mu} = \frac{1}{m_2}(E_2, 0, 0, -p), \quad b^{\mu} = (0, 0, b, 0), \quad \check{b}^{\mu} = (0, b, 0, 0), \quad (8.92)$$

we can introduce the explicit parametrization

$$k^{\mu} = \omega(1, \sin\theta \cos\phi, \sin\theta \sin\phi, \cos\theta)$$
(8.93)

and the polarization vectors

$$\tilde{e}^{\mu}_{\theta} = -\frac{1}{\sin\theta} (\cos\theta, 0, 0, 1), \qquad e^{\mu}_{\phi} = (0, -\sin\phi, \cos\phi, 0), \tag{8.94}$$

which satisfy (8.82), (8.83). In particular, $\mathcal{P} = \omega^2(\sigma^2 - 1)\sin^2\theta$. We provide a more complete list of kinematic relations in Appendix E, together with an expression for the × and + projections in terms of reference vectors e^{μ}_{θ} , e^{μ}_{ϕ} such that e^{μ}_{θ} is not necessarily orthogonal to b^{μ} .

Let us now explore the soft limit $k^{\mu} \to 0$ of the full waveform $\widehat{W}^{\mu\nu}(k)$ (8.75). To leading order in this limit, we find

$$\widehat{W}_{L}^{\mu\nu}(k) = \frac{4G^{2}im_{1}m_{2}c_{0}}{b^{2}r\sqrt{\sigma^{2}-1}\omega_{1}^{2}\omega_{2}^{2}} \times \left(v_{2}^{\mu}v_{2}^{\nu}\omega_{1}^{2}(b\cdot k) - (b\cdot k)v_{1}^{\mu}v_{1}^{\nu}\omega_{2}^{2} - v_{1}^{(\mu}b^{\nu)}\omega_{2}^{2}\omega_{1} + v_{2}^{(\mu}b^{\nu)}\omega_{2}\omega_{1}^{2}\right)$$

$$(8.95)$$

which is in agreement with the PM limit of Weinberg's soft theorem (translated to b-space), using the leading-order deflection $Q_{1PM} = 2Gm_1m_2c_0/(b\sqrt{\sigma^2-1})$. Indicating by $\log(\omega b)$ a generic log-enhanced dependence on the graviton's overall scale,⁴⁹ we find to subleading order in the soft limit

$$\widehat{W}_{S.L.}^{\mu\nu}(k) = \frac{4G^2 m_1 m_2 \left(4(\sigma^2 - 1) - c_0\right) \sigma \log(\omega b)}{r \left(\sigma^2 - 1\right)^{3/2} \omega_1 \omega_2} \left(v_1^{\mu} v_1^{\nu} \omega_2^2 - v_1^{(\mu} v_2^{\nu)} \omega_1 \omega_2 + \omega_1^2 v_2^{\mu} v_2^{\nu}\right) - \frac{2G^2 m_1 m_2 c_0 \log(\omega b)}{r \sqrt{\sigma^2 - 1} \omega_1 \omega_2} \left(k^{(\mu} \xi^{\nu)} - \eta^{\mu\nu} \xi \cdot k\right)$$

$$(8.96)$$

with $\xi^{\mu} = \omega_2 v_1^{\mu} + \omega_1 v_2^{\nu}$. Setting $c_0 = c_0^{\text{GR}}$ to discuss the case of particles interacting only gravitationally, we can compare with the subleading log-theorem [309,310], which predicts the form of the trace-reversed metric perturbation $e^{\mu\nu} = \widehat{W}^{\mu\nu} - \frac{1}{2}\eta^{\mu\nu}\eta_{\alpha\beta}\widehat{W}^{\alpha\beta}$ according to

$$e_{S.L.}^{\mu\nu}(k) = \frac{4G^2 m_1 m_2 \left(2\sigma^2 - 3\right) \sigma \log(\omega b)}{r \left(\sigma^2 - 1\right)^{3/2} \omega_1 \omega_2} \left(v_1^{\mu} v_1^{\nu} \omega_2^2 - v_1^{(\mu} v_2^{\nu)} \omega_1 \omega_2 + \omega_1^2 v_2^{\mu} v_2^{\nu}\right).$$
(8.97)

We see that the first line of (8.96) agrees with (8.97), while the second line of (8.96) vanishes when contracted with physical polarizations. Instead, we do not find agreement with the "sub-subleading" $\mathcal{O}(\omega \log \omega)$ soft contribution to waveforms emitted during collisions of massless objects [142], which lie above the bound (7.94). For instance, for the \times polarization, we find

$$\frac{W_{\times L.} + W_{\times,\mathcal{O}(\omega\log\omega)}}{\hat{W}_{\times L.}} \sim 1 + \frac{1}{4} (\omega b)^2 \log(\omega b) (\sin\theta)^2$$
(8.98)

in any center-of-mass frame, where $\vec{p}_1 + \vec{p}_2 = 0$ (the above relation is instead independent of the translation frame), as $\sigma \to \infty$, while the result of [142] gives rise to a similar relation but with the factor of $\frac{1}{4}$ replaced by $\frac{1}{2}$. Finally, let us calculate the dilatonic waveform. For this case, we set $c_0 = c_0^{\mathcal{N}=8}$ in order to include not only dilaton

emissions but also dilaton exchanges, and take the trace according to

$$\widehat{W} = \frac{1}{\sqrt{2}} \eta_{\mu\nu} \widehat{W}^{\mu\nu} \,. \tag{8.99}$$

Contracting indices generates some terms proportional to $b \cdot k$ from $\widehat{W}_{irr}^{\mu\nu}$ that we can integrate by parts to obtain a more compact expression. The result is $\widehat{W} = \widehat{W}_{12} + \widehat{W}_{irr}$ with

$$\begin{aligned} \widehat{W}_{12}(k) &= \frac{4\sqrt{2}G^2 m_1 m_2 \sigma^2}{r} \left\{ \frac{i}{b(\sigma^2 - 1)} \left[\frac{b \cdot k}{\omega_1} e^{-ib_1 \cdot k} K_1(\Omega_1) - \frac{b \cdot k}{\omega_2} e^{-ib_2 \cdot k} K_1(\Omega_2) \right] \right. \\ &+ \frac{(\sigma^2 - 2)\omega_2 + \sigma\omega_1}{\sigma\omega_1(\sigma^2 - 1)^{3/2}} e^{-ib_1 \cdot k} K_0(\Omega_1) + \frac{(\sigma^2 - 2)\omega_1 + \sigma\omega_2}{\sigma\omega_2(\sigma^2 - 1)^{3/2}} e^{-ib_2 \cdot k} K_0(\Omega_2) \right\} \end{aligned}$$
(8.100)

and

~

~

$$\widehat{W}_{irr}(k) = -\frac{4\sqrt{2}bG^2m_1m_2}{r}(\omega_1^2 + \omega_2^2)\int_0^1 e^{-ib(x)\cdot k}\frac{K_1(\Omega(x))}{\Omega(x)}.$$
(8.101)

8.3.2. Linear momentum

Once the waveforms are given one can easily write down the corresponding formulas for the spectrum of emitted energy and momentum for each polarization as a function of frequency or retarded time. In frequency domain, which is more relevant for interferometers, one has

$$d\mathbf{P}_{\times,+}^{\mu} = k^{\mu} \left| \omega \mathcal{W}_{\times,+} \right|^2 \frac{d\omega \, d\Omega(n)}{2\omega (2\pi)^3},\tag{8.102}$$

⁴⁹ For instance $\log(\Omega_1) = \log(\omega b) + \cdots$, $\log(\Omega_2) = \log(\omega b) + \cdots$ where the dots denote non-log-enhanced terms.

where, as discussed in Appendix F, the measured frequency and the detector's angular positions are characterized by $k^{\mu} = \omega n^{\mu}$, while $d\Omega(n)$ is the angular measure. In particular, using (8.93), the gravitational energy emitted in a given direction θ , ϕ takes the form (see e.g. [337,338])

$$d\mathbf{P}^{0}_{\times,+} = \omega \left| \omega \mathcal{W}_{\times,+} \right|^{2} \sin \theta \, \frac{d\omega \, d\theta \, d\phi}{2\omega (2\pi)^{3}} \,. \tag{8.103}$$

These differential spectra, which characterize the angular and frequency distribution of the energy carried by gravitational waves, have been studied, for instance, in [337,338] for massive objects at finite σ , and in [142,333] for the ultra-relativistic case. Leaving to future work a more detailed study of the spectrum in different kinematical regimes, let us mention a few of its broad features in the frequency domain after integration over the angular variables.

As we have already mentioned, in the limit $\omega \to 0$ the spectra tend to the well-known ZFL [339]. The characteristic frequency range for which a soft expansion in powers of ωb is valid is of order 1/b and in D = 4 contains logarithmically enhanced sub- and sub-sub-leading corrections. Such a mild ω dependence also holds above 1/b as long as ωb does not exceed $\sqrt{\sigma}$. Above this frequency the spectrum drops, first like a small power of ω , and then exponentially when $\omega b > \sigma^{\frac{3}{2}}$. This qualitative feature of the spectrum is responsible for the σ dependence of the total radiated energy to be discussed below. It would be interesting to compare this spectrum with the one of the ultra-relativistic limit i.e. above the bound (7.94), $\Theta \sqrt{\sigma} \gtrsim 1$, but, so far, this has only been done in the small- ω limit as already discussed in Section 7.

Here we shall focus on the calculation of the total emitted energy and momentum, which is obtained by integrating (8.102) over the graviton phase space, to leading order in the coupling,

$$\boldsymbol{P}^{\mu} = \int_{k} \tilde{\mathcal{A}}^{(5)} k^{\mu} \tilde{\mathcal{A}}^{(5)*} \,. \tag{8.104}$$

Following the steps discussed in Section 5, this can be conveniently rewritten in terms of the Fourier transform of a three-particle cut, according to

 $\mathbf{P}^{\mu} = \mathrm{FT} \int d(\mathrm{LIPS}) k^{\mu} \xrightarrow{\left(\begin{array}{c} & & \\ &$

where FT stands for the Fourier transform (5.46). The integral on the right-hand side can be calculated using the reverseunitarity method, i.e. by reinterpreting the delta functions involved in the LIPS as cut propagators. The result in GR is [47,117,118]

$$P_{\rm rad}^{\mu} = \frac{G^3 m_1^2 m_2^2}{b^3} \left(\check{v}_1^{\mu} + \check{v}_2^{\mu} \right) \mathcal{E}(\sigma) \,, \tag{8.106}$$

with

$$\frac{\mathcal{E}(\sigma)}{\pi} = f_1(\sigma) + f_2(\sigma) \log \frac{\sigma + 1}{2} + f_3(\sigma) \frac{\sigma \operatorname{arccosh} \sigma}{2\sqrt{\sigma^2 - 1}},
f_1(\sigma) = \frac{210\sigma^6 - 552\sigma^5 + 339\sigma^4 - 912\sigma^3 + 3148\sigma^2 - 3336\sigma + 1151}{48(\sigma^2 - 1)^{3/2}},
f_2(\sigma) = -\frac{35\sigma^4 + 60\sigma^3 - 150\sigma^2 + 76\sigma - 5}{8\sqrt{\sigma^2 - 1}},
f_3(\sigma) = \frac{\left(2\sigma^2 - 3\right)\left(35\sigma^4 - 30\sigma^2 + 11\right)}{8\left(\sigma^2 - 1\right)^{3/2}}.$$
(8.107)

Let us now check that this emission of energy and momentum is matched by corresponding radiative losses of energymomentum of the colliding objects, i.e. by the integrals in (8.52b), which we denote as follows to leading order in the PM expansion using the further saddle point conditions (8.52a),

$$\mathbf{Q}_{1\mu} = \frac{1}{2} \int_{k} \left[-i \frac{\partial \tilde{\mathcal{A}}^{(5)}}{\partial b_{1}^{\mu}} \tilde{\mathcal{A}}^{(5)*} + i \tilde{\mathcal{A}}^{(5)} \frac{\partial \tilde{\mathcal{A}}^{(5)*}}{\partial b_{1}^{\mu}} \right].$$
(8.108)

Proceeding in the by now familiar way (see Section 5), we can recast this as the following three-particle cut,

$$\mathbf{Q}_{1}^{\mu} = \frac{1}{2} \operatorname{FT} \int d(\operatorname{LIPS})(q_{1}^{\mu} - q_{4}^{\mu}) \xrightarrow{\left(\begin{array}{c} & & \\ &$$

We can then use the same techniques employed to calculate P_{rad}^{μ} , based on reverse-unitarity, obtaining the following result

$$\mathbf{Q}_{1}^{\mu} = -\frac{G^{3}m_{1}^{2}m_{2}^{2}}{b^{3}}\check{v}_{2}^{\mu}\,\mathcal{E}(\sigma)\,,\tag{8.110}$$

with $\mathcal{E}(\sigma)$ as in (8.107), and similarly for particle 2

$$\mathbf{Q}_{2}^{\mu} = -\frac{G^{3}m_{1}^{2}m_{2}^{2}}{b^{3}}\check{v}_{1}^{\mu}\,\mathcal{E}(\sigma).$$
(8.111)

We see that (8.106), (8.110), (8.111) obey the energy-momentum conservation condition

$$\mathbf{P}^{\mu} + \mathbf{Q}_{1}^{\mu} + \mathbf{Q}_{2}^{\mu} = 0.$$
(8.112)

This can be seen as a consequence at the level of integrals of the basic identity $q_1^{\mu} + q_2^{\mu} + k^{\mu} = 0$ among the integrated momenta. Taking into account that the additional Q_{μ} contribution enters the total impulse $Q_{i\mu}$ (8.52b) with opposite sign for the two particles, we thus recover the complete energy–momentum balance law,

$$P^{\mu} + Q_1^{\mu} + Q_2^{\mu} = 0.$$
(8.113)

Here we used that the only contribution to the energy–momentum of the gravitational field is the radiative one given in (8.104), $P^{\mu} = \mathbf{P}^{\mu}$.

Let us comment on the difference between (8.45) and (8.52) when it comes to the non-radiative part of the impulse. In the former case, Q^{μ} is determined by the derivative of the full eikonal phase Re 2 δ up to 3PM, which therefore includes both conservative and radiation-reaction effects, as we discussed in Section 6,

$$Q_1^{\mu} = -\frac{G}{2} Q_{1\text{PM}}^2 \mathcal{I}(\sigma) \frac{b^{\mu}}{b^2} = -Q_2^{\mu} \,. \tag{8.114}$$

In the latter case, Q^{μ} is determined by the derivative of the "reduced" phase 2δ , which does *not* include such effects, due to the subtraction of (8.50). However, this difference is accompanied by the appearance of a new static contribution in the relation (8.52c) linking x^{μ} to b^{μ} , which we denote by

$$\Delta x_{\mu} \equiv (x_1 - x_2 - b_J)_{\mu} = -i \int_{k} \theta(\omega^* - k^0) F^*(k) \Big(\frac{\overleftrightarrow}{\partial Q_1^{\mu}} - \frac{\overleftrightarrow}{\partial Q_2^{\mu}} \Big) F(k) + \mathcal{O}(G^3).$$
(8.115)

To the order under consideration, using the on-shell conditions and the same integrals involved in the evaluation of $\mathcal{J}^{\alpha\beta}$ in Section 7, one obtains

$$\Delta x_{\mu} \simeq Q_{\mu} \mathcal{G}, \qquad \mathcal{G} = \frac{1}{2} \left(c_{14} + c_{23} - c_{13} - c_{24} \right) \,. \tag{8.116}$$

where Q_{μ} in this case is the conservative part of the impulse.

This quantity contributes with an extra term to the relation between *b* and *b_j* in (8.12) and this implies that, in computing Q^{μ} in (8.52a), we can use the relation $b = x + \mathcal{GQ}(b = x)$ getting

$$Q^{\mu} = \frac{\partial 2\tilde{\delta}(x + \mathcal{G}Q(b = x))}{\partial x^{\mu}} = \frac{\partial 2\tilde{\delta}(x)}{\partial x^{\mu}} + \mathcal{G}Q\frac{\partial^2 2\tilde{\delta}(x)}{\partial x \partial x^{\mu}} + \mathcal{O}(G^4).$$
(8.117)

The second term gives the radiation reaction that can be written as

$$Q_1^{\mu} = \frac{G\mathcal{I}(\sigma)}{4b} b^{\mu} \frac{\partial Q^2}{\partial b} = \frac{G\mathcal{I}(\sigma)}{4} \frac{\partial Q^2}{\partial b_{\mu}} = -\frac{G}{2} Q_{1\text{PM}}^2 \mathcal{I}(\sigma) \frac{b^{\mu}}{b^2} + \mathcal{O}(G^4), \qquad (8.118)$$

where, at leading order, we have put x = b, and $\mathcal{G} = \frac{G}{2}\mathcal{I}(\sigma)$ by (8.114), as well as $Q = Q_{1\text{PM}}$. In this way, the predictions of the two formalisms for the full impulse agree.

8.3.3. Angular momentum

The angular momentum of the gravitational field sourced by the collision is given by the sum of two pieces,

$$J^{\alpha\beta} = \boldsymbol{J}^{\alpha\beta} + \mathcal{J}^{\alpha\beta}. \tag{8.119}$$

The second one, $\mathcal{J}^{\alpha\beta}$, is the contribution due to static modes, which starts at 2PM order and which we have already calculated in (7.85). The first one, $J^{\alpha\beta}$, is instead new. It starts at 3PM order and arises due to genuine gravitational wave emissions. Its expression is the same as (7.78), with the gravitational waveform $\tilde{\mathcal{A}}^{(5)\mu\nu}$ replacing the "soft factor" $F^{\mu\nu}$. For convenience, let us break it down as follows

$$\boldsymbol{J}_{\alpha\beta} = \boldsymbol{J}_{\alpha\beta}^{(0)} + \boldsymbol{J}_{\alpha\beta}^{(s)}, \qquad \boldsymbol{J}_{\alpha\beta}^{(0)} = -i \int_{k} k_{[\alpha} \frac{\partial \tilde{\mathcal{A}}^{(5)}}{\partial k^{\beta]}} \tilde{\mathcal{A}}^{(5)*}, \qquad \boldsymbol{J}_{\alpha\beta}^{(s)} = i \int_{k} 2 \tilde{\mathcal{A}}_{[\alpha}^{(5)\mu} \tilde{\mathcal{A}}_{\beta]\mu}^{(5)*}.$$
(8.120)

As already mentioned, the two terms in the previous equations are formally reminiscent of the orbital and spin terms of the gravitational angular momentum, but only their sum is physically meaningful and gauge invariant. One of the main advantages of the expression (8.120) for the angular momentum is that it is manifestly covariant under Lorentz transformation. To see how it transforms under translations,

$$b_{1,2}^{\mu} \mapsto b_{1,2}^{\mu} + a^{\mu} \,, \tag{8.121}$$

let us first remark that, under (8.121), $\tilde{A}^{(5)\mu\nu}(k)$ transforms according to

$$\tilde{\mathcal{A}}^{(5)\mu\nu}(k) \mapsto e^{-ia \cdot k} \tilde{\mathcal{A}}^{(5)\mu\nu}(k), \qquad (8.122)$$

as is clear from the explicit expression (5.51). Then, taking into account the differential operator in (8.120), we see that

$$\mathbf{J}^{\alpha\beta} \mapsto \mathbf{J}^{\alpha\beta} + a^{[\alpha} \mathbf{P}^{\beta]}, \tag{8.123}$$

after comparing with the radiated energy-momentum (8.104). In this way, we see that (8.120) is in fact Poincaré covariant.

Following the above strategy, we shall now recast these expression in a form which is amenable to the application of reverse-unitarity. Let us start from $J_{\alpha\beta}^{(s)}$, for which this manipulation is straightforward:

$$\mathbf{J}_{\alpha\beta}^{(s)} = 2i \operatorname{FT} \int d(\operatorname{LIPS}) \begin{bmatrix} p_1 & \overleftarrow{\qquad} & \overleftarrow{\qquad} \\ & & & & & \\ & & & & \\ & & & \\ & & & & & \\ & & & &$$

as follows by applying the same steps discussed in Section 5, carrying along the appropriate index contractions. Let us now turn to $J_{\alpha\beta}^{(o)}$, for which the presence of a derivative with respect to k^{μ} makes the connection between the *b*-space and the *q*-space representation less straightforward, unlike all quantities considered so far. We first rewrite the defining expression (8.120) in a frame where $b_2^{\alpha} = 0$ and thus $b^{\alpha} = b_1^{\alpha}$, which can always be attained by performing an appropriate translation according to Eqs. (8.121), (8.122), (8.123). In this way we find

$$i\mathbf{J}_{\alpha\beta}^{(0)} = \int_{k} k_{[\alpha} \frac{\partial}{\partial k^{\beta}]} \left[\int \frac{d^{D}q_{1}}{(2\pi)^{D}} 2\pi \,\delta(2p_{1} \cdot q_{1}) 2\pi \,\delta(2p_{2} \cdot (q_{1} + k)) e^{ib \cdot q_{1}} \bigvee_{q_{1}} \left(\begin{array}{c} q_{1} \\ q_{1}$$

where we have already appropriately "flipped" the 2 \rightarrow 3 amplitude in the second line by rewriting it as a 3 \rightarrow 2 one. Shifting $q_4 = q - q_1$, and using $\delta(2p_1 \cdot q_1)\delta(2p_1 \cdot q_4) = \delta(2p_1 \cdot q_1)\delta(2p_1 \cdot q)$, we have

$$\begin{split} \mathbf{j}_{\alpha\beta}^{(o)} &= \int \frac{d^{D}q}{(2\pi)^{D}} 2\pi \,\delta(2p_{1} \cdot q) e^{\mathbf{i} b \cdot q} \int_{k} \int \frac{d^{D}q_{1}}{(2\pi)^{D}} 2\pi \,\delta(2p_{1} \cdot q_{1}) \\ & \times k_{[\alpha} \frac{\partial}{\partial k^{\beta}]} \left[2\pi \,\delta(2p_{2} \cdot (q_{1} + k)) \stackrel{i}{\underset{p_{2}}{\overset{(q_{1} \uparrow \uparrow)}{\longleftrightarrow}}}_{p_{2} \overset{(q_{1} \uparrow \uparrow)}{\longleftrightarrow}} \right] 2\pi \,\delta(2p_{2} \cdot (q - q_{1} - k)) \stackrel{(q_{1} \uparrow \downarrow)}{\underset{p_{2}}{\overset{(q_{1} \uparrow \uparrow)}{\longleftrightarrow}}}_{q_{2} \overset{(q_{1} \uparrow \uparrow)}{\longleftrightarrow}} \right] 2\pi \,\delta(2p_{2} \cdot (q - q_{1} - k)) \stackrel{(q_{1} \downarrow)}{\underset{p_{2}}{\overset{(q_{1} \uparrow \downarrow)}{\longleftrightarrow}}}_{q_{1} \overset{(q_{1} \uparrow \downarrow)}{\longleftrightarrow}}_{q_{2} \overset{(q_{1} \uparrow \downarrow)}{\longleftrightarrow}} \right] 2\pi \,\delta(2p_{2} \cdot (q - q_{1} - k)) \stackrel{(q_{1} \downarrow)}{\underset{(q_{1} \downarrow \downarrow)}{\overset{(q_{1} \downarrow)}{\longleftrightarrow}}}_{q_{1} \overset{(q_{1} \uparrow \downarrow)}{\longleftrightarrow}}_{q_{2} \overset{(q_{2} \downarrow)}{\longleftrightarrow}}_{q_{2} \overset{(q_{1} \downarrow)}{\longleftrightarrow}}_{q_{2} \overset{(q_{2} \downarrow)}{\longleftrightarrow}}_{q_{2} \overset{(q_{1} \downarrow)}{\longleftrightarrow}}_{q_{1} \overset{(q_{1} \downarrow)}{\longleftrightarrow}}_{q_{2} \overset{(q_{2} \downarrow)}{\r}}_{q_{2} \overset{(q_{2} \downarrow)}{\longleftrightarrow}}_{q_{2} \overset{(q_{2$$

When we let the differential operator act on the square bracket, we need to distinguish two types of terms. For the terms where the derivative acts on the amplitude, we can use again

$$\delta(2p_2 \cdot (q_1 + k))\delta(2p_2 \cdot (q - q_1 - k)) = \delta(2p_2 \cdot (q_1 + k))\delta(2p_2 \cdot q).$$
(8.126)

Instead, for those where the derivative acts on the delta function we can use the following property,

$$\frac{\partial \delta(2p_2 \cdot (q_1 + k))}{\partial k^{\beta}} \,\delta(2p_2 \cdot (q - q_1 - k)) = \frac{\partial \delta(2p_2 \cdot (q_1 + k))}{\partial k^{\beta}} \,\delta(2p_2 \cdot q) \\ + \,\delta(2p_2 \cdot (q_1 + k)) \,\frac{\partial \delta(2p_2 \cdot q)}{\partial q^{\beta}} \,, \tag{8.127}$$

which follows from the distributional identity $\delta'(x)\delta(y - x) = \delta'(x)\delta(y) + \delta(x)\delta'(y)$. In total we get three terms: the first one involves the standard Fourier transform in the usual transverse plane $\delta(2p_1 \cdot q)\delta(2p_2 \cdot q)$ and the derivative acts on the amplitude, the second one involves the standard Fourier transform and the derivative acts on the phase-space delta function $\delta(2p_2 \cdot (q_1 + k))$, while in the last one the derivative acts on one of the delta functions appearing in the Fourier transform, $\delta(2p_2 \cdot q)$. Grouping together the first two terms, we thus find

$$i\mathbf{J}_{\alpha\beta}^{(0)} = \mathrm{FT} \int k_{\alpha} \frac{\partial}{\partial k^{\beta}} \begin{bmatrix} p_{1} \underbrace{\langle q_{1} \uparrow \rangle}_{(q_{1} \uparrow)} \\ d(\mathrm{LIPS}) \\ -q_{1} \\ k \\ p_{2} \underbrace{\langle q_{1} \uparrow \rangle}_{(q_{1} \uparrow)} \\ p_{2} \underbrace{\langle q_{1} \uparrow \rangle}_{(q_$$

with $FT_{\alpha}^{(2)}$ given by

$$\mathrm{FT}_{\alpha}^{(2)}\left[f_{\mu}(q)\right] \equiv \int \frac{d^{D}q}{(2\pi)^{D}} \, 2\pi\,\delta(2p_{1}\cdot q)\,2\pi\,\frac{\delta(2p_{2}\cdot q)}{\partial q^{\alpha}}\,e^{ib\cdot q}f_{\mu}(q)\,. \tag{8.129}$$

This modified Fourier transform acts on a quantity that looks like the *q*-space expression of \mathbf{P}^{α} . However, due to the derivative acting on $\delta(2p_2 \cdot q)$, we can no longer calculate this quantity by assuming $v_2 \cdot q = 0$. To evaluate (8.129), let us start by introducing velocities as in (1.7), $p_1^{\mu} = -m_1 v_1^{\mu}$ and $p_2^{\mu} = -m_2 v_2^{\mu}$ so that

$$\mathrm{FT}^{(2)\alpha}\left[f_{\mu}(q)\right] = \frac{u_{2}^{\alpha}}{4m_{1}m_{2}} \int \frac{d^{D}q}{(2\pi)^{D}} 2\pi\,\delta(v_{1}\cdot q)\,2\pi\,\delta'(v_{2}\cdot q)\,e^{ib\cdot q}\,f_{\mu}(q)\,. \tag{8.130}$$

Proceeding as in Appendix A.3, and in particular decomposing the integrated momentum q^{μ} as in (1.10), we find

$$\mathrm{FT}_{\alpha}^{(2)}\left[f_{\mu}(q)\right] = -v_{2\alpha} \int \frac{d^{D-2}q_{\perp}}{(2\pi)^{D-2}} \,\frac{\delta(q_{\parallel 1})dq_{\parallel 1} \, 2\pi \, \delta'(q_{\parallel 2})dq_{\parallel 2}}{4m_1 m_2 \sqrt{\sigma^2 - 1}} \, e^{ib \cdot q_{\perp}} f_{\mu}(q) \,. \tag{8.131}$$

Integrating by parts (i.e. using the definition of the δ' distribution) we can recast this as the ordinary Fourier transform of a derivative, obtaining

$$\operatorname{FT}_{\alpha}^{(2)}\left[f_{\mu}(q)\right] = v_{2\alpha}\operatorname{FT}\left[\frac{\partial}{\partial q_{\parallel 2}}f_{\mu}(q)\right].$$
(8.132)

The second line of (8.128) can be simplified using (8.132) so that

$$i\mathbf{y}_{\alpha\beta}^{(o)} = \operatorname{FT} \int_{k} k_{\alpha} \frac{\partial}{\partial k^{\beta}} \begin{bmatrix} p_{1} \underbrace{\langle q_{1} \uparrow \\ q_{1} \downarrow \\ q_{1} \downarrow \\ q_{1} \uparrow \\ q_{1} \uparrow \\ q_{1} \uparrow \\ q_{1} \downarrow \\ q$$

The dependence on $q_{\parallel 2}$ in the integrand appearing in the second line comes from $v_2 \cdot q$ and from the invariants that involve $q_{\parallel 2}$ and a loop momentum, or in q^2 itself: $I(v_2 \cdot q, q \cdot \ell_1, q \cdot \ell_2, q^2)$. Then, since

$$\begin{aligned} v_{2} \cdot q &= -q_{\parallel 2} , \\ \ell_{1} \cdot q &= \ell_{1} \cdot q_{\perp} + \ell_{1} \cdot \check{v}_{2} q_{\parallel 2} , \\ \ell_{2} \cdot q &= \ell_{2} \cdot q_{\perp} + \ell_{2} \cdot \check{v}_{2} q_{\parallel 2} , \\ q^{2} &= q_{\perp}^{2} + q_{\parallel 2}^{2} \end{aligned}$$

$$(8.134)$$

we find

$$\frac{\partial I}{\partial q_{\parallel 2}} = -\frac{\partial I}{\partial v_2 \cdot q} + \ell_1 \cdot \check{v}_2 \frac{\partial I}{\partial (q \cdot \ell_1)} + \ell_2 \cdot \check{v}_2 \frac{\partial I}{\partial (q \cdot \ell_2)}, \qquad (8.135)$$

because q^2 is quadratic in $q_{\parallel 2}$ so its derivative vanishes at $q_{\parallel 2} = 0$.

Let us comment as to why there are no ambiguities associated to derivatives of delta functions. As is clear from (5.51), one is allowed to choose different expressions for the five-point amplitude provided that they only differ by terms that vanish when

$$v_1 \cdot \ell_1 = 0 \quad \text{or} \quad v_2 \cdot \ell_1 + v_2 \cdot \ell_2 = 0,$$
 (8.136)

thanks to the delta functions. Different choices for these modifications will change each line in (8.133) separately, but will not change the total sum, provided of course that the *same* choice is made consistently in both lines. The easiest option is to just treat q_1 and k as completely independent integration variables, and let the derivative of the delta functions automatically take care of their interdependence on-shell due to the longitudinal components.

Although we discussed the above steps for $b_2^{\alpha} = 0$, we present the final results in a frame where $b_1^{\alpha} + b_2^{\alpha} = 0$, related to the previous one by a translation by $-b^{\alpha}/2$ (see Eqs. (8.121), (8.123)), where particle-interchange symmetry is manifest. Defining \mathcal{E}_{\pm} and \mathcal{F} in terms of the functions \mathcal{E} in (8.107) and \mathcal{C} given by

$$\begin{aligned} \frac{c}{\pi} &= g_1 + g_2 \log \frac{\sigma + 1}{2} + g_3 \frac{\sigma \arccos \sigma}{2\sqrt{\sigma^2 - 1}} ,\\ g_1 &= \frac{105\sigma^7 - 411\sigma^6 + 240\sigma^5 + 537\sigma^4 - 683\sigma^3 + 111\sigma^2 + 386\sigma - 237}{24(\sigma^2 - 1)^2} ,\\ g_2 &= \frac{35\sigma^5 - 90\sigma^4 - 70\sigma^3 + 16\sigma^2 + 155\sigma - 62}{4(\sigma^2 - 1)} ,\\ g_3 &= -\frac{(2\sigma^2 - 3)\left(35\sigma^5 - 60\sigma^4 - 70\sigma^3 + 72\sigma^2 + 19\sigma - 12\right)}{4\left(\sigma^2 - 1\right)^2} \end{aligned}$$
(8.137)

[117,118,332] by letting

$$\mathcal{C}\sqrt{\sigma^2 - 1} = -\mathcal{E}_+ + \sigma\mathcal{E}_-, \qquad \mathcal{F} = \mathcal{E}_+ - \frac{1}{2}\mathcal{E} = -\mathcal{E}_- + \frac{1}{2}\mathcal{E}, \qquad (8.138)$$

P. Di Vecchia, C. Heissenberg, R. Russo et al.

we find

$$\mathbf{J}^{\alpha\beta} \simeq \frac{G^3 m_1^2 m_2^2}{b^3} \mathcal{F}(\sigma) \left(b^{[\alpha} \check{v}_1^{\beta]} - b^{[\alpha} \check{v}_2^{\beta]} \right), \qquad \mathcal{F}(\sigma) = \frac{(\sigma-1)\mathcal{E} - 2\sqrt{\sigma^2 - 1}\mathcal{C}}{2(\sigma+1)}.$$
(8.139)

After a translation

$$a^{\mu} = \frac{E_2 - E_1}{2(E_1 + E_2)} b^{\mu} , \qquad (8.140)$$

which places the center of mass (or rather "center of energy") in the origin of the transverse plane, Eq. (8.139) becomes

$$J^{\alpha\beta} \simeq \frac{G^3 m_1 m_2}{b^3} \left[b^{[\alpha} (m_1 p_2 - m_2 p_1)^{\beta]} \frac{\mathcal{C}(\sigma)}{\sqrt{\sigma^2 - 1}} + \frac{m_1 m_2}{E^2} \mathcal{E}(\sigma) \frac{\sigma - 1}{\sigma^2 - 1} \left(b^{[\alpha} p_1^{\beta]} (m_1 + m_2 \sigma) - b^{[\alpha} p_2^{\beta]} (m_2 + m_1 \sigma) \right) \right]$$
(8.141)

that, in the center of mass system where $-p_1 = (E_1, p)$ and $-p_2 = (E_2, -p)$, reproduces Eq. (15) of Ref. [332] *except* for the static (zero-frequency) modes. Adding (8.141) with such contribution as given by (7.89) expanded up to $\mathcal{O}(G^3)$, one reproduces the full result of Ref. [332],

Let us now provide the analogous formulas expressing the angular momentum that is lost by each particle, which we derive in full detail. Acting with the angular momentum operator

$$L_{(1)\alpha\beta} = -i \int_{k_1} a_1^{\dagger}(k_1) k_{1[\alpha} \frac{\partial a_1(k_1)}{\partial k_1^{\beta]}}$$
(8.142)

on the initial state (8.1) and using the commutation relations, one finds

$$L_{(1)}^{\alpha\beta}|\psi\rangle = \int_{-p_1} \int_{-p_2} \left(-ip_1^{[\alpha}\partial_{p_1}^{\beta]} \Phi_1(-p_1) \right) \Phi_2(-p_2) e^{ib_1 \cdot p_1 + ib_2 \cdot p_2} |-p_1, -p_2\rangle + \int_{-p_1} \int_{-p_2} b_1^{[\alpha}(-p_1^{\beta]}) \Phi_1(-p_1) \Phi_2(-p_2) e^{ib_1 \cdot p_1 + ib_2 \cdot p_2} |-p_1, -p_2\rangle.$$
(8.143)

The corresponding expectation value is therefore

$$\langle \psi | L_{(1)}^{\alpha\beta} | \psi \rangle = \int_{-p_1} \Phi_1^*(-p_1) \left(-i p_1^{[\alpha} \partial_{p_1}^{\beta]} \Phi_1(-p_1) \right) + \int_{-p_1} b_1^{[\alpha}(-p_1^{\beta]}) |\Phi_1(-p_1)|^2 \,. \tag{8.144}$$

Let us now act on the final state dictated by the eikonal operator. For notational simplicity, we discuss the action of the eikonal operator (8.35) which does not include the static-mode contributions involving $F^{\mu\nu}$. They can be easily reinstated by replacing $W_j(x_1, x_2, k)$ with $\theta(k^0 - \omega^*)W(x_1, x_2, k) + \theta(\omega^* - k^0)f_j(x_1, x_2, k)$. Doing this and replacing Re 2δ with 2δ , one obtains the result for (8.51). For simplicity, we also suppress the additional integrals over x and Q, which do not play a role in the derivation. With this proviso, we start by calculating

$$\begin{split} L_{(1)\alpha\beta}S|\psi\rangle &= \int_{p_{3}}\int_{p_{4}} e^{-ib_{1}\cdot p_{4} - ib_{2}\cdot p_{3}} |p_{3}, p_{4}\rangle \\ &\times \int \frac{d^{D}Q_{1}}{(2\pi)^{D}} \int \frac{d^{D}Q_{2}}{(2\pi)^{D}} \int d^{D}x_{1} \int d^{D}x_{2} e^{i(b_{1}-x_{1})\cdot Q_{1}+i(b_{2}-x_{2})\cdot Q_{2}+2i\delta(b)} \\ &\times \left(b_{1[\alpha|}p_{4|\beta]} + p_{4[\alpha}\frac{\partial 2\delta(b)}{\partial p_{4}^{\beta]}} + \int_{k} p_{4[\alpha}\frac{\partial}{\partial p_{4}^{\beta]}}W(x_{1}, x_{2}, k) a^{\dagger}(k)\right) \\ &\times e^{i\int_{k}W(x_{1}, x_{2}, k)a^{\dagger}(k)} |0\rangle \, \Phi_{1}(p_{4}-Q_{1}) \, \Phi_{2}(p_{3}-Q_{2}) \\ &+ \int_{p_{3}}\int_{p_{4}} e^{-ib_{1}\cdot p_{4}-ib_{2}\cdot p_{3}} |p_{3}, p_{4}\rangle \\ &\times \int \frac{d^{D}Q_{1}}{(2\pi)^{D}} \int \frac{d^{D}Q_{2}}{(2\pi)^{D}} \int d^{D}x_{1} \int d^{D}x_{2} e^{i(b_{1}-x_{1})\cdot Q_{1}+i(b_{2}-x_{2})\cdot Q_{2}+2i\delta(b)} \\ &\times \left(-ip_{4[\alpha}\frac{\partial}{\partial p_{4}^{\beta]}} \Phi_{1}(p_{4}-Q_{1})\right) e^{i\int_{k}W(x_{1}, x_{2}, k)a^{\dagger}(k)} |0\rangle \, \Phi_{2}(p_{3}-Q_{2}) \end{split}$$

In order to make the expression slightly more compact, we used (5.52) in the exponents to reabsorb the imaginary part of 2δ arising from the reordering of the exponential factors. In the last line we can use

$$p_{4[\alpha} \frac{\partial}{\partial p_4^{\beta]}} \Phi_1(p_4 - Q_1) = (p_4 - Q_1)_{[\alpha} \frac{\partial}{\partial p_4^{\beta]}} \Phi_1(p_4 - Q_1) - Q_{1[\alpha} \frac{\partial}{\partial Q_1^{\beta]}} \Phi_1(p_4 - Q_1)$$
(8.146)

and integrate by parts the second term. Doing this and defining the operator

$$O_{(1)\alpha\beta} = p_{4[\alpha} \frac{\partial}{\partial p_4^{\beta]}} + Q_{1[\alpha} \frac{\partial}{\partial Q_1^{\beta]}}$$
(8.147)

we can write

$$\begin{split} L_{(1)\alpha\beta}S|\psi\rangle &= \int_{p_{3}}\int_{p_{4}}e^{-ib_{1}\cdot p_{4}-ib_{2}\cdot p_{3}}|p_{3},p_{4}\rangle \\ &\times \int \frac{d^{D}Q_{1}}{(2\pi)^{D}}\int \frac{d^{D}Q_{2}}{(2\pi)^{D}}\int d^{D}x_{1}\int d^{D}x_{2} e^{i(b_{1}-x_{1})\cdot Q_{1}+i(b_{2}-x_{2})\cdot Q_{2}+2i\delta(b)} \\ &\times \left(x_{1[\alpha|}Q_{1|\beta]}+O_{(1)\alpha\beta}2\delta(b)+\int_{k}O_{(1)\alpha\beta}\mathcal{W}(x_{1},x_{2},k)a^{\dagger}(k)\right) \\ &\times e^{i\int_{k}\mathcal{W}(x_{1},x_{2},k)a^{\dagger}(k)}|0\rangle \ \Phi_{1}(p_{4}-Q_{1}) \ \Phi_{2}(p_{3}-Q_{2}) \\ &+ \int_{p_{3}}\int_{p_{4}}e^{-ib_{1}\cdot p_{4}-ib_{2}\cdot p_{3}}|p_{3},p_{4}\rangle \\ &\times \int \frac{d^{D}Q_{1}}{(2\pi)^{D}}\int \frac{d^{D}Q_{2}}{(2\pi)^{D}}\int d^{D}x_{1}\int d^{D}x_{2} e^{i(b_{1}-x_{1})\cdot Q_{1}+i(b_{2}-x_{2})\cdot Q_{2}+2i\delta(b)} \\ &\times \left(b_{1[\alpha|}(p_{4}-Q_{1})|_{\beta]}-i(p_{4}-Q_{1})|_{\alpha}\frac{\partial}{\partial p_{4}^{\beta]}}\Phi_{1}(p_{4}-Q_{1})\right) e^{i\int_{k}\mathcal{W}(x_{1},x_{2},k)a^{\dagger}(k)}|0\rangle \ \Phi_{2}(p_{3}-Q_{2}) \,. \end{split}$$

For the expectation value, we find

$$\begin{split} \langle \psi | S^{\dagger} L_{(1)\alpha\beta} S | \psi \rangle \\ &= \int_{p_{3}, p_{4}} \int \frac{d^{D} Q_{1}'}{(2\pi)^{D}} \int \frac{d^{D} Q_{2}'}{(2\pi)^{D}} \int d^{D} x_{1}' \int d^{D} x_{2}' e^{-i(b_{1}-x_{1}') Q_{1}' - i(b_{2}-x_{2}') Q_{2}' - i\operatorname{Re} 2\delta(b')} \\ &\times \int \frac{d^{D} Q_{1}}{(2\pi)^{D}} \int \frac{d^{D} Q_{2}}{(2\pi)^{D}} \int d^{D} x_{1} \int d^{D} x_{2} e^{i(b_{1}-x_{1}) Q_{1} + i(b_{2}-x_{2}) Q_{2} + i\operatorname{Re} 2\delta(b)} \\ &\times \left(x_{1[\alpha|Q_{1}|\beta]} + O_{(1)\alpha\beta} 2\delta(b) - i \int_{k} W^{*}(x_{1}', x_{2}', k) O_{(1)\alpha\beta} W(x_{1}, x_{2}, k) \right) |0\rangle \\ &\times e^{-\operatorname{Im} 2\delta(b') - \operatorname{Im} 2\delta(b) + \int_{k} W^{*}(x_{1}', x_{2}', k) W(x_{1}, x_{2}, k)} \\ &\times \Phi_{1}^{*}(p_{4} - Q_{1}') \Phi_{2}^{*}(p_{3} - Q_{2}') \Phi_{1}(p_{4} - Q_{1}) \Phi_{2}(p_{3} - Q_{2}) \\ &+ \int_{p_{3}, p_{4}} \int \frac{d^{D} Q_{1}'}{(2\pi)^{D}} \int \frac{d^{D} Q_{2}'}{(2\pi)^{D}} \int d^{D} x_{1}' \int d^{D} x_{2}' e^{-i(b_{1}-x_{1}') Q_{1}' - i(b_{2}-x_{2}') Q_{2}' - i\operatorname{Re} 2\delta(b')} \\ &\times \int \frac{d^{D} Q_{1}}{(2\pi)^{D}} \int \frac{d^{D} Q_{2}}{(2\pi)^{D}} \int d^{D} x_{1} \int d^{D} x_{2} e^{i(b_{1}-x_{1}) Q_{1} + i(b_{2}-x_{2}) Q_{2} + i\operatorname{Re} 2\delta(b)} \\ &\times \left(b_{1[\beta|}(p_{4} - Q_{1})_{\alpha|} \Phi_{1}(p_{4} - Q_{1}) - i(p_{4} - Q_{1})_{[\alpha} \frac{\partial}{\partial p_{4}^{\beta]}} \Phi_{1}(p_{4} - Q_{1}) \right) \\ &\times e^{-\operatorname{Im} 2\delta(b') - \operatorname{Im} 2\delta(b) + \int_{k} W(x_{1}', x_{2}', k) W(x_{1}, x_{2}, k)} \\ &\times \Phi_{1}^{*}(p_{4} - Q_{1}') \Phi_{2}^{*}(p_{3} - Q_{2}') \Phi_{2}(p_{3} - Q_{2}) \,. \end{aligned}$$
saddle point

At the saddle point,

$$\begin{aligned} \langle \psi | S^{\dagger} L_{(1)\alpha\beta} S | \psi \rangle \\ &= \int_{p_{3}, p_{4}} | \Phi_{1}(p_{4} - Q_{1})|^{2} | \Phi_{2}(p_{3} - Q_{2})|^{2} \\ &\times \left(x_{1[\alpha|}Q_{1|\beta]} + O_{(1)\alpha\beta} \operatorname{Re} 2\delta(b) - i \int_{k} \mathcal{W}^{*}(x_{1}, x_{2}, k) \stackrel{\leftrightarrow}{O}_{(1)\alpha\beta} \mathcal{W}(x_{1}, x_{2}, k) \right) \\ &+ \int_{p_{3}, p_{4}} \Phi_{1}(p_{4} - Q_{1})^{*} | \Phi_{2}(p_{3} - Q_{2})|^{2} \\ &\times \left(b_{1[\beta|}(p_{4} - Q_{1})_{|\alpha|} \Phi_{1}(p_{4} - Q_{1}) - i(p_{4} - Q_{1})_{[\alpha} \frac{\partial}{\partial p_{4}^{\beta]}} \Phi_{1}(p_{4} - Q_{1}) \right) \end{aligned}$$
(8.150)

and shifting the integration like in Section 8.1.2, $p_4 = Q_1 - p_1$, $p_3 = Q_2 - p_2$ and recognizing the last line as the angular momentum of the initial state (8.144), we finally obtain

$$\langle \psi | S^{\mathsf{T}} L_{(1)\alpha\beta} S | \psi \rangle - \langle \psi | L_{(1)\alpha\beta} | \psi \rangle$$

$$= \int_{-p_1, -p_2} |\Phi_1(-p_1)|^2 |\Phi_2(-p_2)|^2$$

$$\times \left(x_{1[\alpha|} Q_{1|\beta]} + O_{(1)\alpha\beta} \operatorname{Re} 2\delta(b) - i \int_k \mathcal{W}^*(x_1, x_2, k) \overset{\leftrightarrow}{O}_{(1)\alpha\beta} \mathcal{W}(x_1, x_2, k) \right)$$
(8.151)

with $O_{(1)\alpha\beta}$ the operator (8.147).

Performing the appropriate replacements discussed below (8.144), if we are interested in the angular momentum up to G^3 including the static modes, we thus have

$$\Delta L_{(1)}^{\alpha\beta} = x_{1[\alpha|}Q_{1|\beta]} + p_{4[\alpha]}\frac{\partial 2\tilde{\delta}(b)}{\partial p_4^{\beta]}} - i\int_k \tilde{\mathcal{A}}^{(5)}p_{4[\alpha}\frac{\partial}{p_4^{\beta]}}\tilde{\mathcal{A}}^{(5)} - i\int_k F^*O_{(1)\alpha\beta}F + \mathcal{O}(G^4)\,, \tag{8.152}$$

where we have neglected the second term in the operator (8.147) in $O_{(1)\alpha\beta}2\tilde{\delta}$ and similarly in $\tilde{\mathcal{A}}^{(5)*}O_{(1)\alpha\beta}\tilde{\mathcal{A}}^{(5)}$, because it would give $\mathcal{O}(G^4)$ effects since $\tilde{\mathcal{A}}^{(5)*}\tilde{\mathcal{A}}^{(5)}$ is already $\mathcal{O}(G^3)$. Instead, as we shall see, it is important to keep it in $F^*O_{(1)\alpha\beta}F$ because it grants the identity

$$O_{(1)\alpha\beta}F^{\mu\nu} = p_{4[\alpha}\frac{\partial G^{\mu\nu}}{\partial p_4^{\beta]}} + p_{1[\alpha}\frac{\partial G^{\mu\nu}}{\partial p_1^{\beta]}},\tag{8.153}$$

where $G^{\mu\nu}(p_4, p_3, p_1, p_2)$ is the "soft factor" seen as a function of four independent hard momenta, i.e.

$$F^{\mu\nu}(p_4, p_3, Q_1, Q_2) = G^{\mu\nu}(p_4, p_3, Q_1 - p_4, Q_2 - p_3).$$
(8.154)

In the following, we will calculate (8.152), breaking it down into the following terms

$$\Delta L_1^{\alpha\beta} = \Delta L_{(1c)}^{\alpha\beta} + \Delta L_1^{\alpha\beta} + \Delta \mathcal{L}_1^{\alpha\beta}$$
(8.155)

and similarly for particle 2. The conservative term reads as follows, using that the difference between x_1^{μ} and b_1^{μ} is aligned with Q^{μ} as can be seen from the saddle point condition (8.52a) and from Eqs. (8.8), (8.9),

$$\Delta L_{(1c)}^{\alpha\beta} = x_{1[\alpha|}Q_{|\beta]} + p_{4[\alpha]}\frac{\partial \operatorname{Re} 2\delta(b)}{\partial p_4^{\beta]}} = b_{1[\alpha|}Q_{|\beta]} + p_{4[\alpha]}\frac{\partial \operatorname{Re} 2\delta(b)}{\partial p_4^{\beta]}}.$$
(8.156)

We already calculated (8.156) in Eq. (8.23) and following. The radiative terms is given by

$$\Delta \boldsymbol{L}_{i}^{\alpha\beta} = \mathrm{Im} \boldsymbol{J}_{i}^{\alpha\beta} + b_{i}^{[\alpha} \boldsymbol{Q}_{i}^{\beta]}$$
(8.157)

where \mathbf{Q}_{i}^{α} is the radiative contribution to the impulse as in (8.108) and we defined the shorthand

$$\boldsymbol{J}_{i\alpha\beta} = \int_{k} p_{i[\alpha} \frac{\partial \tilde{\mathcal{A}}^{(5)}}{\partial p_{i}^{\beta]}} \tilde{\mathcal{A}}^{(5)*} \,. \tag{8.158}$$

Focusing for definiteness on particle 2 in a frame where $b_2^{\alpha} = 0$, and proceeding in a manner entirely analogous to what we did for $J^{\alpha\beta}$, one arrives at the following expression in terms of three-particle cuts,

$$J_{2\alpha\beta} = \operatorname{FT} \int v_{2[\alpha} \frac{\partial}{\partial v_{2}^{\beta]}} \begin{bmatrix} p_{1} \underbrace{\langle q_{1} \uparrow \rangle}_{\langle q_{1} \uparrow \rangle} \\ d(\operatorname{LIPS}) \underbrace{\langle q_{1} \uparrow \rangle}_{p_{2}} \\ p_{2} \underbrace{\langle q_{1} \uparrow \rangle}_{\langle q_{1} \uparrow \rangle} \\ p_{2} \underbrace{\langle q_{1} \uparrow \rangle}_{\langle q_{1} \uparrow \rangle} \\ p_{2} \underbrace{\langle q_{1} \uparrow \rangle}_{\langle q_{1} \uparrow \rangle} \\ p_{2} \underbrace{\langle q_{1} \uparrow \rangle}_{\langle q_{1} \uparrow \rangle} \\ p_{2} \underbrace{\langle q_{1} \uparrow \rangle}_{\langle q_{1} \uparrow \rangle} \\ p_{2} \underbrace{\langle q_{1} \uparrow \rangle}_{\langle q_{1} \uparrow \rangle} \\ p_{2} \underbrace{\langle q_{1} \uparrow \rangle}_{\langle q_{1} \uparrow \rangle} \\ p_{2} \underbrace{\langle q_{1} \uparrow \rangle}_{\langle q_{1} \uparrow \rangle} \\ p_{2} \underbrace{\langle q_{1} \uparrow \rangle}_{\langle q_{1} \uparrow \rangle} \\ p_{2} \underbrace{\langle q_{1} \uparrow \rangle}_{\langle q_{1} \uparrow \rangle} \\ p_{2} \underbrace{\langle q_{1} \uparrow \rangle}_{\langle q_{1} \uparrow \rangle} \\ p_{2} \underbrace{\langle q_{1} \uparrow \rangle}_{\langle q_{1} \uparrow \rangle} \\ p_{2} \underbrace{\langle q_{1} \uparrow \rangle}_{\langle q_{1} \uparrow \rangle} \\ p_{2} \underbrace{\langle q_{1} \uparrow \rangle}_{\langle q_{1} \uparrow \rangle} \\ p_{2} \underbrace{\langle q_{1} \uparrow \rangle}_{\langle q_{1} \uparrow \rangle} \\ p_{2} \underbrace{\langle q_{1} \uparrow \rangle}_{\langle q_{1} \uparrow \rangle} \\ p_{2} \underbrace{\langle q_{1} \uparrow \rangle}_{\langle q_{1} \uparrow \rangle} \\ p_{3} \underbrace{\langle q_{1} \uparrow \rangle}_{\langle q_{1} \uparrow \rangle} \\ p_{4} \underbrace{\langle q_{1} \uparrow \rangle}_{\langle q_{1} \uparrow \rangle} \\ p_{4} \underbrace{\langle q_{1} \uparrow \rangle}_{\langle q_{1} \uparrow \rangle} \\ p_{4} \underbrace{\langle q_{1} \uparrow \rangle}_{\langle q_{1} \uparrow \rangle} \\ p_{4} \underbrace{\langle q_{1} \uparrow \rangle}_{\langle q_{1} \uparrow \rangle} \\ p_{4} \underbrace{\langle q_{1} \uparrow \rangle}_{\langle q_{1} \uparrow \rangle} \\ p_{4} \underbrace{\langle q_{1} \uparrow \rangle}_{\langle q_{1} \uparrow \rangle} \\ p_{4} \underbrace{\langle q_{1} \uparrow \rangle}_{\langle q_{1} \uparrow \rangle} \\ p_{4} \underbrace{\langle q_{1} \uparrow \rangle}_{\langle q_{1} \uparrow \rangle} \\ p_{4} \underbrace{\langle q_{1} \uparrow \rangle}_{\langle q_{1} \uparrow \rangle} \\ p_{4} \underbrace{\langle q_{1} \uparrow \rangle}_{\langle q_{1} \uparrow \rangle} \\ p_{4} \underbrace{\langle q_{1} \uparrow \rangle}_{\langle q_{1} \uparrow \rangle} \\ p_{4} \underbrace{\langle q_{1} \uparrow \rangle}_{\langle q_{1} \uparrow \rangle} \\ p_{4} \underbrace{\langle q_{1} \uparrow \rangle}_{\langle q_{1} \uparrow \rangle} \\ p_{4} \underbrace{\langle q_{1} \uparrow \rangle}_{\langle q_{1} \uparrow \rangle} \\ p_{4} \underbrace{\langle q_{1} \uparrow \rangle}_{\langle q_{1} \uparrow \rangle} \\ p_{4} \underbrace{\langle q_{1} \uparrow \rangle}_{\langle q_{1} \downarrow \rangle} \\ p_{4} \underbrace{\langle q_{1} \uparrow \rangle}_{\langle q_{1} \downarrow \rangle} \\ p_{4} \underbrace{\langle q_{1} \uparrow \rangle}_{\langle q_{1} \downarrow \rangle} \\ p_{4} \underbrace{\langle q_{1} \uparrow \rangle}_{\langle q_{1} \downarrow \rangle} \\ p_{4} \underbrace{\langle q_{1} \uparrow \rangle}_{\langle q_{1} \downarrow \rangle} \\ p_{4} \underbrace{\langle q_{1} \uparrow \rangle}_{\langle q_{1} \downarrow \rangle} \\ p_{4} \underbrace{\langle q_{1} \uparrow \rangle}_{\langle q_{1} \downarrow \rangle} \\ p_{4} \underbrace{\langle q_{1} \uparrow \rangle}_{\langle q_{1} \downarrow \rangle} \\ p_{4} \underbrace{\langle q_{1} \downarrow \rangle}_{\langle q_{1} \downarrow \rangle} \\ p_{4} \underbrace{\langle q_{1} \downarrow \rangle}_{\langle q_{1} \downarrow \rangle} \\ p_{4} \underbrace{\langle q_{1} \downarrow \rangle}_{\langle q_{1} \downarrow \rangle} \\ p_{4} \underbrace{\langle q_{1} \downarrow \rangle}_{\langle q_{1} \downarrow \rangle} \\ p_{4} \underbrace{\langle q_{1} \downarrow \rangle}_{\langle q_{1} \downarrow \rangle} \\ p_{4} \underbrace{\langle q_{1} \downarrow \rangle}_{\langle q_{1} \downarrow \rangle} \\ p_{4} \underbrace{\langle q_{1} \downarrow \rangle}_{\langle q_{1} \downarrow \rangle} \\ p_{4} \underbrace{\langle q_{1} \downarrow \rangle}_{\langle q_{1} \downarrow \rangle} \\ p_{4} \underbrace{\langle q_{1} \downarrow \rangle}_{\langle q_{1} \downarrow \rangle} \\ p_{4} \underbrace{\langle q_{1} \downarrow \rangle}_{\langle q_{1} \downarrow \rangle} \\ p_{4} \underbrace{\langle q_{1} \downarrow \rangle} \\ p_{4} \underbrace{\langle q_{1$$

and employing reverse-unitarity we obtain the result

 \leftrightarrow

$$\Delta \mathbf{L}_{1}^{\alpha\beta} \simeq \frac{G^{3}m_{1}^{2}m_{2}^{2}}{b^{3}} \left[+ \frac{\mathcal{E}_{+}b^{[\alpha}v_{1}^{\beta]}}{\sigma - 1} - \frac{1}{2}\mathcal{E}b^{[\alpha}\check{v}_{2}^{\beta]} \right],$$

$$\Delta \mathbf{L}_{2}^{\alpha\beta} \simeq \frac{G^{3}m_{1}^{2}m_{2}^{2}}{b^{3}} \left[-\frac{\mathcal{E}_{+}b^{[\alpha}v_{2}^{\beta]}}{\sigma - 1} + \frac{1}{2}\mathcal{E}b^{[\alpha}\check{v}_{1}^{\beta]} \right].$$
(8.160)

Of course the expression for particle 1 is obtained by interchanging particle labels everywhere. The radiative quantities (8.139) and (8.160) obey the balance law

$$\boldsymbol{J}^{\alpha\beta} + \Delta \boldsymbol{L}_{1}^{\alpha\beta} + \Delta \boldsymbol{L}_{2}^{\alpha\beta} = 0.$$
(8.161)

Finally, the static-mode contributions to the changes in the angular momenta are given by the following formula valid up to $\mathcal{O}(G^3)$,

$$\Delta \mathcal{L}_{1}^{\alpha\beta} = \int_{k} F^{*} \left(p_{4[\alpha} \frac{\overleftrightarrow{\partial}}{\partial p_{4}^{\beta]}} + Q_{1[\alpha} \frac{\overleftrightarrow{\partial}}{\partial Q_{1}^{\beta]}} \right) F + b_{1}^{[\alpha} \mathcal{Q}_{1}^{\beta]}, \qquad (8.162)$$

where \mathcal{Q}_1^{β} is the static contribution to the impulse, (8.114). Under a translation (8.121), $\Delta \mathcal{L}_i^{\alpha\beta} \rightarrow \Delta \mathcal{L}_i^{\alpha\beta} + a^{[\alpha} \mathcal{Q}_i^{\beta]}$. Making use of the identity (8.153), we see that

$$-i\int_{k}F^{*}\left(p_{4[\alpha}\frac{\overleftrightarrow{\partial}}{\partial p_{4}^{\beta]}}+Q_{1[\alpha}\frac{\partial}{\partial Q_{1}^{\beta]}}\right)F=-i\int_{k}G^{*}p_{1[\alpha}\frac{\overleftrightarrow{\partial}}{\partial p_{1}^{\beta]}}G-i\int_{k}G^{*}p_{4[\alpha}\frac{\overleftrightarrow{\partial}}{\partial p_{4}^{\beta]}}G=J_{(1)\alpha\beta}+J_{(4)\alpha\beta},$$
(8.163)

where we have defined

$$J_{(n)\alpha\beta} = -i \int_{k} G^* p_{n[\alpha} \frac{\partial}{\partial p_n^{\beta]}} G.$$
(8.164)

Of course, a similar manipulation works for particle 2. The integrals appearing in these combinations can be evaluated in the same way as those we discussed in the previous section and give the following result for m = 1, 2, 3, 4,

$$2\eta_m J_{(m)}^{\alpha\beta} = \sum_{\eta_n = -\eta_m} c_{nm} p_n^{[\alpha} p_m^{\beta]} - \sum_{\substack{\eta_n = \eta_m \\ n \neq m}} d_{nm} p_n^{[\alpha} p_m^{\beta]}, \qquad (8.165)$$

For instance,

$$2J_4^{\alpha\beta} = c_{14}p_1^{[\alpha}p_4^{\beta]} + c_{24}p_2^{[\alpha}p_4^{\beta]} - d_{12}p_3^{[\alpha}p_4^{\beta]}.$$
(8.166)

with c_{nm} as defined in (7.86) and

$$d_{nm} = 2G \frac{\sigma_{nm}^2 - \frac{1}{2}}{\sigma_{nm}^2 - 1} \,. \tag{8.167}$$

In conclusion, we find the following result for (8.162) and for the analogous expression for particle 2

$$\Delta \mathcal{L}_{1}^{\alpha\beta} = J_{1}^{\alpha\beta} + J_{4}^{\alpha\beta} + b_{1}^{[\alpha} \mathcal{Q}_{1}^{\beta]}, \qquad \Delta \mathcal{L}_{2}^{\alpha\beta} = J_{2}^{\alpha\beta} + J_{3}^{\alpha\beta} + b_{2}^{[\alpha} \mathcal{Q}_{2}^{\beta]}.$$
(8.168)

Here, $Q_1^{\alpha} = -Q_2^{\alpha}$ is the 3PM radiation-reaction contribution to the impulse,

$$Q_{1}^{\alpha} = -\frac{GQ_{1PM}^{2}b^{\alpha}}{2b^{2}}\mathcal{I}(\sigma), \qquad Q_{1PM} = \frac{2Gm_{1}m_{2}(2\sigma^{2}-1)}{b\sqrt{\sigma^{2}-1}}$$
(8.169)

with $\mathcal{I}(\sigma)$ as in (7.88). One can easily check that

$$J_1^{\alpha\beta} + J_2^{\alpha\beta} + J_3^{\alpha\beta} + J_4^{\alpha\beta} = -\mathcal{J}^{\alpha\beta} \,. \tag{8.170}$$

Moreover, $b_1^{[\alpha} Q_1^{\beta]} + b_2^{[\alpha} Q_2^{\beta]} = (b_1 - b_2)^{[\alpha} Q_1^{\beta]} = b_J^{[\alpha} Q_1^{\beta]} = 0$ (up to $\mathcal{O}(G^4)$ corrections as follows from (8.12)) which vanishes by antisymmetry. As a result, these static contributions obey the "separate" balance law

$$\mathcal{J}^{\alpha\beta} + \Delta \mathcal{L}_1^{\alpha\beta} + \Delta \mathcal{L}_2^{\alpha\beta} = 0 \tag{8.171}$$

so that, in conclusion,

$$J^{\alpha\beta} + \Delta L_1^{\alpha\beta} + \Delta L_2^{\alpha\beta} = 0.$$
(8.172)

9. Summary and outlook

Conventional perturbation theory provides, in the common practice, the main tool to approach scattering-amplitude calculations. It is based on the assumption that, in the regime of interest, the relevant coupling constant can be regarded as sufficiently small. A textbook example is Quantum Electrodynamics, in which elementary charges couple via the fine-structure constant, $e^2/\hbar \approx 1/137$. The situation is dramatically different when it comes to gravitational interactions of classically sizeable objects, such as black holes with masses (and, thus, energies) much larger than the Planck mass, whose effective coupling GE^2/\hbar is extremely large. This seemingly renders our most effective techniques to calculate scattering amplitudes useless to study the classical limit of gravity.

In this report, we illustrated how the eikonal exponentiation serves as a convenient, flexible and conceptually transparent tool to attack this problem in a variety of different theories and setups. This mechanism "builds" classical gravitational interactions by resumming the exchanges of many gravitons, in accordance with the intuition that their number, like any quantum number, should become large in the classical limit. Their contributions to the $2 \rightarrow 2$ amplitude A exponentiate, so that in the classical limit $1+i\tilde{A} \simeq e^{2i\delta_0}$ with $2\delta_0(s, b)$ the leading eikonal phase, which is simply dictated by the single-graviton exchange and indeed proportional to GE^2/\hbar . In this way, perturbation theory comes back into the game since, by matching to the eikonal exponentiation, the calculation of higher-loop amplitudes allows one to obtain successively more refined approximations for this exponent, $2\delta_0 + 2\delta_1 + \cdots$, weighted by the classical small PM parameter GE/b. In turn, the rapidly oscillating nature of the exponentiated amplitude fixes the values of the classical exchanged momentum, the impulse, via a saddle point condition.

After reviewing the combinatoric proof of the leading eikonal exponentiation, and illustrating its connection with the impulse and classical deflection angle, as well as the close relation with the phase shifts perhaps more familiar from elementary treatments of the angular momentum in quantum mechanics, we went on to discuss the single-exchange $\mathcal{O}(G)$ eikonal phase $2\delta_0$ and its manifold applications: minimally coupled massive particles in GR, both spinless and carrying a classical spin, massive particles in maximal supergravity, and string collisions. In many situations, proving the exponentiation to all orders is impractical, and one can resort to a more pragmatic approach of checking the constraints it imposes at each loop level. In this spirit, we presented the discussion of one-loop amplitudes in the classical limit both in order to retrieve the next-to-leading $\mathcal{O}(G^2)$ phase $2\delta_1$ and as a means of checking the $i(2\delta_0)^2/2!$ term dictated by the two-graviton exchanges. A similar pattern holds for high-energy string scattering, and in the string setup new interesting phenomena arise. In particular, already the leading string eikonal is promoted to an operator: for instance, excitations in the *s*-channel have to be included at short distances, while tidal deformations can become important also at large distance. We showed that the leading energy contribution of the one-loop elastic amplitude in string theory is consistent with the exponentiation of the leading eikonal operator, but currently a similar analysis for massive string states is lacking.

After a detour into the interconnections between unitarity and the *b*-space exponentiation, we reviewed the twoloop, $\mathcal{O}(G^3)$, calculation focusing on maximal supergravity and GR. The $\mathcal{O}(G^3)$ eikonal $2\delta_2$ is the lowest-order contribution to the classical exponent at which imprints of the dissipative nature of the scattering problem manifest themselves. It possesses both so-called radiation-reaction terms in its real part, which reflect into time-reversal-odd contributions to the deflection angle, and an infrared-divergent imaginary part. The latter is due to the fact that the $2 \rightarrow 2$ amplitude by its very nature neglects the emitted radiation, and, via unitarity, this corresponds to nontrivial three-particle cuts that the standard exponentiation does not account for. We therefore took a first step to amend this shortcoming by combining the eikonal exponentiation with the Weinberg exponentiation of soft quanta, which ought to be included in the description of any physical scattering event in order to obtain infrared-safe final quantities and to restore manifest unitarity. The general nature of soft theorems also allowed us to take a peek beyond the conventional PM regime, by relaxing the assumption that the impulse be parametrically small compared to the masses. The resulting exponential structure combining elastic collisions with inelastic emission processes takes the form of an operator, and in the last section we provided a self-contained discussion of this eikonal operator to $\mathcal{O}(G^3)$ beyond the soft approximation, where it neatly disentangles conservative and dissipative effects, and allows one to calculate a variety of different observables such as the emitted energy and angular momentum during the collision.

Together with the emergence of these new theoretical structures, which we illustrated in detail in this report, the classical limit also presents simplifying features, mainly when it comes to evaluating loop integrals. In this limit, as long as the colliding objects stay far apart, one can focus on those contributions to loop integrals that are non-analytic in *q*-space and therefore relevant to the long-range behavior in *b*-space [340,341]. A key simplification in this respect is due to the method of region [253,342], which permits to focus on such contributions directly, without the need to first evaluate the full integrals and then take their asymptotic expansions. We have briefly presented a few simple applications of this method, while dedicating more space to the physical lessons that can be extracted from the results of its application. Similarly, we have extensively applied, but only sketched, the powerful method of reverse-unitarity [117,118,343–346], which allows one to calculate phase-space integrals from discontinuities of more conventional loop integrals. In summary, we hope to have been able to convey the main physical ideas behind the eikonal exponentiation of gravity amplitude up to $O(G^3)$, and to have stimulated the reader's interest and curiosity towards this angle of approach to the problem of gravitational scattering, which not only led to the discovery of new patterns and to an improved theoretical understanding of gravity, but also to very concrete new predictions for gravitational observables.

At this stage, several open challenges lie ahead, both from a conceptual and from a technical standpoint. Remaining at "low" PM orders, all observables including the deflection angle and the emitted energy and angular momentum can be

determined up to 3PM from the knowledge of the 2 \rightarrow 2 amplitude up to two loops and of the tree-level 2 \rightarrow 3 amplitude in the classical limit [47,117,118,145,332]. However, as we have reviewed in the respective chapters, these observables behave very differently when one considers collisions with increasing center-of-mass energies. The deflection angle up to 3PM turns out to be perfectly smooth in the limit in which this energy *E* (equivalently, the Lorentz factor σ) is taken to be large by keeping the leading deflection angle *GE/b* small and fixed, i.e. in the ultrarelativistic limit. This is actually crucial in order to ensure agreement with an early calculation of this quantity for collisions of massless objects [15], and in order to clarify its universality. In this limit, the massless particle with the highest spin, the graviton, dominates, since it couples to the highest power of the energy, and the deflection angle becomes the same for any theory where this condition is satisfied. In contrast, the 3PM emitted energy and angular momentum only make sense below the bound given by (7.94), i.e. provided $\sqrt{\sigma}(GE/b)$ is at most of order one [333,337,338]. Trusting their 3PM expressions beyond this threshold leads to nonsensical conclusions, as the systems seems to be able to radiate much more energy and angular momentum than it possessed to begin with! A veritable "energy crisis".

Guided by the example of the Zero-Frequency Limit of the energy emission spectrum, which is governed by soft theorems and can thus be calculated independently of the PM approximation [112,144], it is highly likely that this singularity arises due to the fact that we are attempting to calculate the coefficient of a power series, the "expansion in powers of *G*" (or PM expansion), of a quantity which is actually not analytic in *G*. This issue emerged very clearly in the ZFL of the spectrum [112], where the full expression starts exhibiting a branch singularity precisely at the bound (7.94). In that case, the correct high-energy limit for the fraction of energy radiated at frequencies between 0 and 1/*b* does not behave, unphysically, like $(GE/b)^3 \log \sigma$, which is unbounded as $\sigma \to \infty$, but rather exhibits (bounded) non-analytic terms of the type $(GE/b)^3 \log(GE/b)$. It is tempting to speculate that a similar mechanism may apply to the full emitted energy and angular momentum as well, in order to produce a possibly non-analytic but still physically sensible answer for their ultrarelativistic limits. This task is made challenging by the fact that soft theorems no longer apply to such integrated quantities, but it is also of great interest since it would provide a connection with the results obtained for the scattering of massless objects in Refs. [142,143,335].

Scatterings of massless states have been the subject of intense studies up to two loops in string theory, already in the early days of the gravity eikonal [15] and, hopefully, the progress within a QFT approach discussed in this report will motivate new further studies in this string theory context. It would be interesting to derive the 2PM string eikonal operator $(2\hat{s}_1 \text{ in our notation})$ and use the new results on higher-loop string amplitudes obtained over the past 20 years (see for instance [347] and references therein) to investigate systematically the eikonal exponentiation in string theory beyond the one-loop level, which is what has been included in this report. Moreover, little is known as far as the exponentiation of string amplitudes involving massive states is concerned, see however [228]. Borrowing from the mileage and intuition gained in the field theory context, where the corresponding studies have reached three loops [40,41], in particular the focus on the non analytic terms that dominate the long range dynamics, will certainly provide a pivotal simplification, compared to the study of the full amplitude. Moreover, it could prove useful to introduce masses via Kaluza–Klein compactification rather than by considering excited string states.

Moving on to a broader perspective, outstanding efforts have been recently devoted to the endeavor of bringing amplitude-based techniques for gravitational-wave physics closer to timely phenomenological applications, along two main directions: analytically continuing the results obtained from scattering amplitudes to the bound-system kinematics, and including all relevant effects beyond the point-particle approximation. Moving forward, it will of course be interesting to undertake similar analysis by means of the eikonal exponentiation as well.

In the present report we have discussed a few basic steps towards making the information extracted from the amplitude more directly available for applications to bound systems, of which binary mergers represent the key relevant example for observational purposes, in the spirit of the so-called Boundary-to-Bound dictionary [291–293,348]. This connection is essentially based on an analytic continuation, one could say, from "positive" to "negative" energy (after subtracting the rest mass in the CM frame). In addition, since most binary systems revolve along on quasi-circular trajectories, the resulting information also needs to be matched from a regime of large eccentricities, more directly accessible from the PM expansion, to small or vanishing eccentricities. We have illustrated how this can be done in practice, starting from 1PM and 2PM data in order to retrieve all PN data up to 2PN, while we leave the investigation of how 3PM and 4PM eikonal data may eventually combine to yield the 3PN and 4PN information for future work.

String scattering represents a prototypical example where the colliding objects possess an internal structure, which manifests itself as a dependence of the eikonal operator on the string's excitation modes (see Section 3.2.4). For phenomenological applications, one can similarly introduce such deformations, encoding in particular tidal Love numbers and higher-multiple modes in the amplitude approach [50–54] in an EFT spirit [113] by means of suitable additional parameters. This is particularly relevant for neutron stars, whose tidal deformability properties are expected to provide insights into their Quantum Chromodynamical origin, internal structure and on the resulting equation of state. Including classical spin effects is also important and this poses a challenge as far as the amplitude approach is concerned. Indeed, while masses and tidal deformabilities can be simply encoded in a Lagrangian formulation via suitable continuous parameters, the intrinsically quantized nature of spin *s* in the quantum world poses an obstacle to taking the classical limit, in which $\hbar s$ should become classically sizeable and *s* attain very large values. In this report we have presented a self-contained account of a simple strategy to include classical spin effects in the leading eikonal, i.e. to first order in *GM/b* but formally to all orders in the spin parameter. At present, the inclusion of spin from one-loop order onward relies on a



Fig. 19. Dominant diagrams contributing to the semiclassical eikonal phase. The shaded blob stands for a generic connected tree-level topology.

strategy to break down the calculation into an expansion in spin multipoles. The missing ingredient to upgrade this to an all-order-in-spin calculation, the Compton-like amplitude with classical spin, remains partly elusive, despite encouraging recent developments [58–84,349–352] including possible ties with states built out of the string spectrum [353]. The only exception is the calculation of the radiation reaction at two-loop order [81] that, as in the spinless case discussed in [106], is only based on the soft limit of the five-point amplitude. The result obtained with this approach agrees in the small spin limit with the complete two-loop calculation for the spin one case fully described by an extended $\mathcal{N} = 2$ world-line supersymmetry [136].

Coming to the frontier in the *G*-expansion [38–41,47,104,118,131,135,138,289,332,354], a potentially loose end concerns the integrals over fluctuations around the saddle points characterizing the classical limit. Following the literature, we have presented the derivation of classical observables associated to the scattering by taking appropriate expectations of the relevant operators on the final state dictated by the action of the eikonal operator (minus their expectation on the initial state), and evaluated the integrals by means of the saddle point conditions. In principle, one should be able to check that the Gaussian integrals for the fluctuations about such classical values eventually produce suppressed contributions, either weighted by powers of \hbar or, perhaps more plausibly, of *GE/b*. In the latter case, it will be of course important to take them into account in the appropriate way when moving to higher orders in this parameter. Put another way, it should be possible to explicitly check the normalization condition for the final state, which, in a novel fashion, brings us back to the original issue of unitarity restoration in the classical limit.

In this connection let us mention, for completeness, a partially successful attempt [355] at constructing a semiclassical unitary S-matrix in the case of ultrarelativistic (massless) collisions. In Sections 3.2.6, 4.2.5 we have discussed what has been achieved in the two regimes labeled as I and II in Fig. 12. Unfortunately, the most interesting regime, the one leading classically to gravitational collapse, is also the most difficult one to analyze. So far, it has only been approached in the point-particle limit and in D = 4 (although going to D > 4 one would avoid having to deal with some infrared divergences). The problem at hand is that the PM expansion is an expansion in powers of R/b and, precisely because of gravitational collapse, it is expected to break down at some critical value of $\mathcal{O}(1)$ for that ratio (see Fig. 12). The only simplification that looks to be fully justified is that, order by order in the above expansion, the dominant diagrams contributing to the semiclassical eikonal phase consist of connected tree-diagrams involving the gravitons emitted by the two energetic particles as external lines (see Fig. 19). Adding disconnected diagrams with the same external legs should exponentiate the eikonal phase in a standard way (much like the two-particle reducible contributions of Ref. [111]). Summing tree-diagrams is supposed to be equivalent to solving a classical field theory. Using work by Lipatov [356] one can make a reasonable guess about the form of such field theory in the high-energy limit. After simplifying further Lipatov's action a two-dimensional field theory was proposed and studied in [355] with reasonable success in view of the rough approximations made. We refer to the literature concerning those developments and only mention here that critical values for *R/b* in good agreement with the classical expectations discussed in Section 2.4 were indeed found [357–359]. On the other hand, in spite of considerable effort [360,361], control of unitarity could not be achieved in any simple way above the critical value. At the same time, no strong claim of unitarity loss below b_c can be made in view of the many unjustified approximations made along the way.

Coming back to the PM expansion, there still are aspects of the full 4PM result [138] that need to be investigated including a more detailed comparison with the PN approaches [261,362]. It will certainly be interesting to continue the analysis of the classical dynamics at three loops beyond the case of scalar particles, along the lines of [354], and also to calculate the angular momenta at 4PM. Recently the conservative classical dynamics at four loops was studied for the relativistic scattering in electrodynamics [363], paving the way for the analysis of the GR case at 5PM. From the 4PM order, new exciting physical phenomena manifest themselves: tail and recoil effects. The tail effect is caused by an interaction between the two-body system and the gravitational energy that is first emitted and then reabsorbed by the system itself [30,364,365]. For this reason, it can be regarded as a "globally conservative" type of interaction which is however nonlocal in time, and thus very different from the typical "potential" interactions, whereby each objects feels the gravitational pull of the other one. Recoil is caused by the net emission of linear momentum and represents a novel dissipative effect that combines with the more familiar back-reaction of radiation emissions on the relative motion to that order. Notably, recoil implies that, even if the system is observed from the initial center-of-mass frame, the two bodies will in general experience *different* deflection angles compared to their original direction of motion. It will be important

to investigate how such phenomena fit together in an improved version of the eikonal operator, yet to be formulated, in order to clarify the classical limit of the amplitude(s) up to 4PM order. A first step in this direction has been taken in [145] where the linear reaction [296] at 4PM was derived from an expansion of the stationary phase conditions (8.52), using in particular the prescription (8.36), finding perfect agreement with the result of [332]. In addition the high-energy behavior of the 4PM deflection angle(s) currently available in the literature points to a breakdown of the smoothness property enjoyed by it up to 3PM. Whether or not this is to be regarded as a manifestation of the bound (7.94) certainly deserves further investigation.

This also relates to another open puzzle in massless $\mathcal{N} = 8$ supergravity amplitudes, where it was observed that the exponentiation of the elastic 2 \rightarrow 2 amplitude fails precisely starting to order $\mathcal{O}(G^4)$ due to a superclassical mismatch proportional to Im $2\delta_2$ [44,45]. Armed with the appropriate eikonal operator, whose phase is by construction manifestly real, it will hopefully be possible to move past this obstacle by appropriately including all relevant inelastic channels as well.

As we have discussed in the last two sections of this report, the first step in this direction, which up to 3PM completely clarifies the structure of the exponentiation and allows for a systematic calculation of the observables, is to combine eikonalized (virtual) graviton exchanges with coherent (real) graviton emissions, effectively producing an operator that links elastic $2 \rightarrow 2$ with inelastic $2 \rightarrow 3$ amplitude. One cannot help but wonder whether the ingredients entering the next order will eventually turn out to be essentially the same, or if instead new ones are to be included. For instance it is possible that new amplitudes will need to be exponentiated, in particular the Compton-like graviton scattering in Fig. 6, as suggested by the fact that they appear as possible sub-diagrams or cuts of higher loop $2 \rightarrow 3$ and $2 \rightarrow 2$ amplitudes [146–148,257] (while of course analogous $1 \rightarrow 3$ processes with two final gravitons in the final states are kinematically forbidden). A closely related issue concerns the possible emergence of correlations among graviton emissions, in particular due to the constraint of total energy conservation. Such effects, which ought to be encoded to leading order in the $2 \rightarrow 4$ amplitude with two graviton emissions, have been shown to be absent at tree level, and the picture of a coherent, uncorrelated emission has not received corrections so far [120,336]. Further research and careful scrutiny will hopefully help to clarify these points. Here we would like to mention some of the most recent developments which confirm the usefulness of the eikonal approach in the analysis of subleading PM corrections.

Extending the logic presented in this report, it is natural to generalize the eikonal operator in (8.35) or (8.48) including in W_j also the subleading PM contributions that can be extracted from the classical limit of the 2 \rightarrow 3 amplitude at 1-loop. It would then be possible to derive the NLO PM scattering waveform by following the same steps of Section 8.3.1. Starting from the result for the classical 1-loop 5-point amplitude [146–148,257], this problem is being under intense investigation. First, the relation between the classical amplitude and the gravitational waveform becomes more subtle at NLO. As discussed in [255], the gravitational waveform is one of the asymptotic observables requiring a non-standard timeordering from the S-matrix point of view, which, in the KMOC approach, is implemented by taking into account carefully some cut contributions. At NLO, these contributions implement a change of frame for the waveform [366], thus making the result very natural from the point of view of the eikonal exponentiation mentioned after (8.35) and explaining the need of such a rotation to match the PM waveform against known PN [367] or soft data [368]. Beside this, it was realized that the choice of BMS frame affects the NLO waveform also beyond its delta-function contribution at zero-frequency [366]. A detailed comparison between the NLO PM waveform and the PN results, including several non-linear effects, is discussed in [369,370] finding perfect agreement. It is certainly interesting to extend this comparison to higher order to see whether the intuition based on the eikonal exponentiation can again provide useful guidance.

Besides the applications mentioned above, the key idea underlying the gravitational eikonal (resummation of leading contributions leading to an exponentiation) is likely to provide new insights also in other contexts. For instance, the AdS/CFT duality is the perfect setup to study some of the fundamental questions that motivated the early analysis of the gravitational eikonal as mentioned in the introduction. Advances in the analytic calculation of holographic correlators through a bootstrap approach [371,372] can provide new tools to improve our understanding of the gravitational eikonal in this context and possibly use it to study the nature of black-hole microstates. The eikonal approach can be useful also to analyze the gravitational scattering in more general curved backgrounds, see for instance [373,374], which again can encode interesting information about the nature of black holes as for instance pointed out in [375].

CRediT authorship contribution statement

Paolo Di Vecchia: Conceptualization, Writing – original draft, Writing – review & editing. **Carlo Heissenberg:** Conceptualization, Writing – original draft, Writing – review & editing. **Rodolfo Russo:** Conceptualization, Writing – original draft, Writing – review & editing. **Conceptualization**, Writing – original draft, Writing – review & editing. **Conceptualization**, Writing – original draft, Writing – review & editing.

Declaration of competing interest

The authors declare that they have no known competing financial interests or personal relationships that could have appeared to influence the work reported in this paper.

Acknowledgments

We would like to thank Francesco Alessio, Fabian Bautista, Brando Bellazzini, Marcus Berg, Zvi Bern, Massimo Bianchi, Donato Bini, Andreas Brandhuber, Emil Bjerrum-Bohr, Costas Bachas, Poul Henrik Damgaard, Thibault Damour, Stefano De Angelis, Maurizio Firrotta, Stefano Foffa, Claudio Gambino, Alessandro Georgoudis, Riccardo Gonzo, Enrico Herrmann, Henrik Johansson, Gregor Kälin, David Kosower, Emil Martinec, Raffaele Marotta, Gustav Mogull, Alexander Ochirov, Donal O'Connell, Julio Parra-Martinez, Ludovic Planté, Massimo Porrati, Rafael Porto, Fabio Riccioni, Massimiliano Riva, Michael Ruf, Chia-Hsien Shen, Riccardo Sturani, Gabriele Travaglini, Pierre Vanhove, Filippo Vernizzi, Leong Khim Wong, Mao Zeng for several enlightening discussions.

RR and GV are grateful to IHES for hospitality during the part of this work. CH and RR are grateful to KITP, Santa Barbara, for hospitality during the workshop High-Precision Gravitational Waves.

This research was partially supported in part by the National Science Foundation, US under Grant No. NSF PHY-1748958. The research of RR is partially supported by the UK Science and Technology Facilities Council (STFC) Consolidated Grants ST/P000754/1 and ST/T000686/1. The research of CH (PDV) is fully (partially) supported by the Knut and Alice Wallenberg Foundation, Sweden under grant KAW 2018.0116. Nordita is partially supported by Nordforsk, Norway.

Appendix A. Field theory details and conventions

A.1. Permutation identities

Let us first show by induction that

$$F(a_1,\ldots,a_n) \equiv \sum_{\sigma \in S_n} \frac{1}{a_{\sigma_1}} \cdots \frac{1}{a_{\sigma_1} + \cdots + a_{\sigma_n}} = \frac{1}{a_1 \cdots a_n} \,. \tag{A.1}$$

For n = 2, we find

$$F(a_1, a_2) = \frac{1}{a_1(a_1 + a_2)} + \frac{1}{a_2(a_2 + a_1)} = \frac{1}{a_1 a_2}.$$
(A.2)

Moreover, denoting by $S_n(j)$ the permutations such that $\sigma_n = j$, we can write

$$F(a_1, \dots, a_n) = \frac{1}{a_1 + \dots + a_n} \sum_{j=1}^n \left[\sum_{\sigma \in S_n(j)} \frac{1}{a_{\sigma_1}} \cdots \frac{1}{a_{\sigma_1} + \dots + a_{\sigma_{n-1}}} \right].$$
 (A.3)

The quantity within brackets is equal to $F(a_1, \ldots, \widehat{a_j}, \ldots, a_n)$, where $\widehat{a_j}$ means that a_j is omitted. Using the induction hypothesis we then find

$$\sum_{j=1}^{n} F(a_1, \dots, \widehat{a_j}, \dots, a_n) = \sum_{j=1}^{n} \frac{1}{a_1 \cdots \widehat{a_j} \cdots a_n} = \frac{a_1 + \dots + a_n}{a_1 \cdots a_n},$$
(A.4)

so that (A.3) leads to (A.1). The quantity f in (2.16) is related to F by

$$f(a_1, ..., a_n) = (a_1 + \dots + a_n)F(a_1, \dots, a_n),$$
(A.5)

and therefore it is given by

$$f(a_1, \dots, a_n) = \sum_{j=1}^n \frac{1}{a_1 \cdots \widehat{a_j} \cdots a_n} \,. \tag{A.6}$$

We shall now use (A.6) together with

$$2\pi\delta(\omega) = \int e^{i\omega t} dt , \qquad \frac{1}{\omega_k - i0} = \int e^{-i(\omega_k - i0)t_k} \theta(t_k) i dt_k$$
(A.7)

to simplify

$$T(\omega_1,\ldots,\omega_n) \equiv 2\pi\,\delta(\omega_1+\cdots+\omega_n)f(\omega_1-i0,\ldots,\omega_n-i0),\,. \tag{A.8}$$

Let us again start from the n = 2 case, for simplicity. We have

$$T(\omega_1, \omega_2) = \int dt_1 dt_2 i\theta(t_1) e^{i\omega_1(t_2 - t_1)} e^{i\omega_2 t_2} + \int dt_1 dt_2 i\theta(t_2) e^{i\omega_2(t_1 - t_2)} e^{i\omega_1 t_1}$$
(A.9)

so that shifting $t_1 \rightarrow t_2 - t_1$ in the first term and $t_2 \rightarrow t_1 - t_2$ in the second leads to

$$T(\omega_1, \omega_2) = \int dt_1 dt_2 \, i \left[\theta(t_2 - t_1) + \theta(t_1 - t_2) \right] e^{i\omega_1 t_1} e^{i\omega_2 t_2} \tag{A.10}$$

and thus

$$\frac{1}{2\pi}T(\omega_1,\omega_2) = 2i\pi\,\,\delta(\omega_1)\delta(\omega_2)\,.\tag{A.11}$$

A similar manipulation goes through for generic *n*, where

$$T(\omega_1,\ldots,\omega_n) = \int dt_1 \cdots dt_n \sum_{j=1}^n e^{i\omega_j t_j} \prod_{k \neq j} i\theta(t_k) e^{i\omega_k(t_j - t_k)}.$$
(A.12)

In each term of the sum over *j*, we can send $t_k \rightarrow t_j - t_k$ for all $k \neq j$, to obtain

$$T(\omega_1, \dots, \omega_n) = i^{n-1} \int dt_1 \cdots dt_n \, e^{i(\omega_1 t_1 + \dots + \omega_n t_n)} \sum_{j=1}^n \prod_{k \neq j} \theta(t_j - t_k) \,. \tag{A.13}$$

Noting that the factor involving the theta functions is 1 for any ordering of t_1, \ldots, t_n finally yields

$$\frac{1}{2\pi}T(\omega_1,\ldots,\omega_n) = (2i\pi)^{n-1}\delta(\omega_1)\cdots\delta(\omega_n).$$
(A.14)

A.2. Feynman rules

In this appendix we collect useful Feynman rules. We start from the classical action describing a massless scalar ϕ minimally coupled to gravity,

$$S = \int \frac{R}{2\kappa^2} \sqrt{-g} \, d^D x - \frac{1}{2} \int \partial_\mu \phi \, g^{\mu\nu} \, \partial_\nu \phi \, \sqrt{-g} \, d^D x \,, \tag{A.15}$$

with $\kappa = \sqrt{8\pi G}$. Defining $g_{\mu\nu} = \eta_{\mu\nu} + 2\kappa h_{\mu\nu}$ and retaining quadratic terms only, one finds

$$S^{(2)} = \frac{1}{2} \int h^{\mu\nu} \left[\Box h_{\mu\nu} - \partial_{\mu} \partial^{\alpha} h_{\alpha\nu} - \partial_{\nu} \partial^{\alpha} h_{\alpha\mu} + \partial_{\mu} \partial_{\nu} h^{\alpha}_{\alpha} - \eta_{\mu\nu} \left(\Box h^{\alpha}_{\alpha} - \partial^{\alpha} \partial^{\beta} h_{\alpha\beta} \right) \right] d^{D}x + \frac{1}{2} \int \phi \Box \phi \, d^{D}x , \qquad (A.16)$$

where indices are raised and lowered using the Minkowski metric. To quantize the theory we go to De Donder gauge,

$$\partial^{\alpha}h_{\mu\alpha} = \frac{1}{2} \partial_{\mu}h^{\alpha}_{\alpha} \,. \tag{A.17}$$

This requires adding to the action a gauge-fixing term plus suitable ghost contributions, which however do not play any role in our classical analysis. The net effect is to replace $S^{(2)}$ with

$$S_{DD}^{(2)} = \frac{1}{2} \int h^{\mu\nu} \left(\Box h_{\mu\nu} - \frac{1}{2} \eta_{\mu\nu} \Box h_{\alpha}^{\alpha} \right) d^{D}x + \frac{1}{2} \int \phi \Box \phi \, d^{D}x \,.$$
(A.18)

Rewriting the graviton part in terms of

$$D_{\mu\nu,\rho\sigma} = \frac{1}{2} \left(\eta_{\mu\rho} \eta_{\nu\sigma} + \eta_{\mu\sigma} \eta_{\nu\rho} - \eta_{\mu\nu} \eta_{\rho\sigma} \right), \tag{A.19}$$

we find

$$S_{DD}^{(2)} = \frac{1}{2} \int h^{\mu\nu} D_{\mu\nu,\rho\sigma} \Box h^{\rho\sigma} d^D x + \frac{1}{2} \int \phi \Box \phi d^D x \,. \tag{A.20}$$

The inverse of $D_{\mu\nu,\rho\sigma}$ is

$$P_{\mu\nu,\rho\sigma} = \frac{1}{2} \left(\eta_{\mu\rho} \eta_{\nu\sigma} + \eta_{\mu\sigma} \eta_{\nu\rho} - \frac{2}{D-2} \eta_{\mu\nu} \eta_{\rho\sigma} \right), \tag{A.21}$$

which satisfies

$$D^{\mu\nu,\rho\sigma}P_{\rho\sigma,\alpha\beta} = \frac{1}{2} \left(\delta^{\mu}_{\alpha} \delta^{\nu}_{\beta} + \delta^{\nu}_{\alpha} \delta^{\mu}_{\beta} \right), \tag{A.22}$$

and this fixes the propagators to be

$$G_{\mu\nu,\rho\sigma}(k) = \frac{-iP_{\mu\nu,\rho\sigma}}{k^2 - i0}, \qquad G(k) = \frac{-i}{k^2 - i0}.$$
 (A.23)

The leading scalar-graviton interaction term

$$S_{I} = -\kappa \int h^{\mu\nu} \left[-\partial_{\mu}\phi \ \partial_{\nu}\phi + \frac{1}{2} \eta_{\mu\nu} (\partial\phi)^{2} \right] d^{D}x , \qquad (A.24)$$

which translates to the vertex

$$\tau^{\mu\nu}(p,p') = -i\kappa \Big[p^{\mu}p'^{\nu} + p^{\nu}p'^{\mu} - \eta^{\mu\nu}(p \cdot p') \Big], \tag{A.25}$$

where the scalar lines are regarded as both outgoing.

In case the scalar has mass m,

$$S = \int \frac{R}{2\kappa^2} \sqrt{-g} \, d^D x - \frac{1}{2} \int \left(\partial_\mu \phi \, g^{\mu\nu} \, \partial_\nu \phi + m^2 \phi^2 \right) \sqrt{-g} \, d^D x \,, \tag{A.26}$$

can be discussed along similar lines. The massive scalar propagator is given by

$$G_m(k) = \frac{-i}{k^2 + m^2}$$
(A.27)

while the leading scalar-graviton vertex in De Donder gauge reads

$$\tau^{\mu\nu}(p,p') = -i\kappa \left[p^{\mu}p'^{\nu} + p^{\nu}p'^{\mu} - \eta^{\mu\nu}(p \cdot p' - m^2) \right].$$
(A.28)

Note that this vertex is transverse with respect to the graviton momentum,

$$\tau^{\mu\nu}(p,p')(p_{\nu}+p'_{\nu}) = 0, \qquad (A.29)$$

whenever the scalar lines are on-shell $p^2 = p'^2 = -m^2$. Moreover, its trace reads

$$\eta_{\mu\nu}\tau^{\mu\nu}(p,p') = -i\kappa \left[\frac{2-D}{2}(p+p')^2 + 2m^2\right],\tag{A.30}$$

which is zero in the massless case, up to terms that vanish on the graviton's mass shell.

A.3. Useful Fourier transforms to impact parameter space

In this appendix we collect a few useful properties of the Fourier transform into impact parameter space. Taking the D-dimensional Fourier transform of the S-matrix element one obtains the invariant expression (1.32) and this leads us to consider Fourier transforms of the following type,

$$FT[f](b) = \int \frac{d^D q}{(2\pi)^D} 2\pi \,\delta(2p_1 \cdot q - q^2) 2\pi \,\delta(2p_2 \cdot q + q^2) \,e^{ib \cdot q} f(q^2) \,. \tag{A.31}$$

with $f(q^2)$ playing the role of the invariant amplitude $\mathcal{A}(s, -q^2)$. In order to recast this in a more explicit way, let us start from the case in which p_1 and p_2 are massive momenta, rewriting $p_1^{\mu} = -m_1 v_1^{\mu}$ and $p_2^{\mu} = -m_2 v_2^{\mu}$ as in (1.7) so that

$$FT[f](b) = \frac{1}{4m_1m_2} \int \frac{d^D q}{(2\pi)^D} 2\pi \delta\left(v_1 \cdot q + \frac{q^2}{2m_1}\right) 2\pi \delta\left(v_2 \cdot q - \frac{q^2}{2m_2}\right) e^{ib \cdot q} f(q^2).$$
(A.32)

It is convenient to change integration variables by decomposing the integrated momentum according to (1.10) so that

$$q^{\mu} = \check{v}_{1}^{\mu} q_{\parallel 1} + \check{v}_{2}^{\mu} q_{\parallel 2} + q_{\perp}^{\mu}$$
(A.33)

where by definition $q_{\perp} \cdot p_{1,2} \equiv 0$. Therefore

$$(dq)^{2} = \check{v}_{1}^{2} (dq_{\parallel 1})^{2} + 2\check{v}_{1} \cdot \check{v}_{2} dq_{\parallel 1} dq_{\parallel 2} + \check{v}_{2}^{2} (dq_{\parallel 2})^{2} + (dq_{\perp})^{2}$$
(A.34)

and the determinant of this metric is given by

$$-\det g = (\check{v}_1 \cdot \check{v}_2)^2 - \check{v}_1^2 \check{v}_1^2 = \frac{1}{\sigma^2 - 1}.$$
(A.35)

Therefore,

$$d^{D}q = \frac{dq_{\parallel 1} dq_{\parallel_{1}}}{\sqrt{\sigma^{2} - 1}} d^{D-2}q_{\perp}, \qquad (A.36)$$

and we find

$$FT[f](b) = \frac{1}{4m_1m_2} \int \frac{dq_{\parallel 1} dq_{\parallel 1}}{\sqrt{\sigma^2 - 1}} \frac{d^{D-2}q_{\perp}}{(2\pi)^{D-2}} \delta\left(q_{\parallel 1} - \frac{q^2}{2m_1}\right) \delta\left(q_{\parallel 2} + \frac{q^2}{2m_2}\right) e^{ib \cdot q} f(q^2).$$
(A.37)

We are only interested in evaluating this Fourier transform for very large *b*, orthogonal to p_1 and p_2 (up to $\mathcal{O}(G^2)$ corrections), and therefore we can expand the integrand for small $q \sim 1/b$. Using the Taylor expansion of the delta functions and noting that (cf. (1.45))

$$q^{2} = \frac{q_{\parallel 1}^{2} - 2\sigma q_{\parallel 1} q_{\parallel 2} + q_{\parallel 2}^{2}}{\sigma^{2} - 1} + q_{\perp}^{2}, \qquad (A.38)$$

P. Di Vecchia, C. Heissenberg, R. Russo et al.

we obtain

$$FT[f](b) = \frac{1}{4m_1m_2\sqrt{\sigma^2 - 1}} \int \frac{d^{D-2}q_{\perp}}{(2\pi)^{D-2}} e^{ib \cdot q_{\perp}} \left(f(q_{\perp}^2) + \frac{s}{4m_1^2m_2^2(\sigma^2 - 1)} \left[x^2 f(x) \right]_{x=q_{\perp}^2}' + \cdots \right)$$
(A.39)

(the superscript ' stands for a total derivative with respect to the argument x) or equivalently, using the identity (1.13),

$$FT[f](b) = \frac{1}{4Ep} \int \frac{d^{D-2}q_{\perp}}{(2\pi)^{D-2}} e^{ib \cdot q_{\perp}} \left(f(q_{\perp}^2) + \frac{1}{4p^2} \left[x^2 f(x) \right]_{x=q_{\perp}^2}' + \cdots \right).$$
(A.40)

In the massless case, $p_1^2 = p_2^2 = 0$, we can instead adopt the decomposition

$$q^{\mu} = \frac{1}{s} \left(p_1^{\mu} x_1 + p_2^{\mu} x_2 \right) + q_{\perp}^{\mu} \,, \tag{A.41}$$

where again $q_{\perp} \cdot p_{1,2} = 0$. Since $x_{1,2} = -2p_{2,1} \cdot q$, we then find

$$(dq)^2 = -\frac{dx_1 dx_2}{s} + (dq_\perp)^2, \qquad \sqrt{-\det g} = \frac{1}{2s}$$
 (A.42)

and in this way (A.31) evaluates to

$$FT[f](b) = \frac{1}{2s} \int \frac{d^{D-2}q_{\perp}}{(2\pi)^{D-2}} e^{ib \cdot q_{\perp}} \left(f(q_{\perp}^2) + \frac{1}{s} \left[x^2 f(x) \right]_{x=q_{\perp}^2}' + \cdots \right).$$
(A.43)

Recalling that in the massless case $E = 2p = \sqrt{s}$, we see that the expression (A.40) is thus valid for both massive and massless setups. It is easy to see that the first term on the right-hand side of (A.40) corresponds to simply dropping the q^2 in the arguments of the delta functions appearing in (A.31), for which we adopt the notation

$$\tilde{f}(b) = \int \frac{d^{D}q}{(2\pi)^{D}} 2\pi \,\delta(2p_{1} \cdot q) 2\pi \,\delta(2p_{2} \cdot q) \,e^{ib \cdot q} f(q^{2}) = \frac{1}{4Ep} \int \frac{d^{D-2}q_{\perp}}{(2\pi)^{D-2}} e^{ib \cdot q_{\perp}} f(q_{\perp}^{2}) \,. \tag{A.44}$$

Since in most applications $f(q^2)$ has a power-law dependence on q^2 , the basic type of integrals that we need to calculate is:

$$I_D(\nu) = \int \frac{d^{D-2}q}{(2\pi)^{D-2}} e^{ib \cdot q} \left(q^2\right)^{\nu} = \frac{2^{2\nu}}{\pi^{1-\epsilon}} \frac{\Gamma(1+\nu-\epsilon)}{\Gamma(-\nu)(b^2)^{\nu+1-\epsilon}}, \qquad D = 4 - 2\epsilon.$$
(A.45)

A quick way to see this is to introduce Schwinger parameters, so that

$$I_D(\nu) = \int_0^\infty dt \; \frac{t^{-1-\nu}}{\Gamma(-\nu)} \int \frac{d^{D-2}q}{(2\pi)^{D-2}} \, e^{ib \cdot q - tq^2} \tag{A.46}$$

and performing the Gaussian integral and letting t = 1/x leads to

$$I_D(\nu) = \frac{1}{(4\pi)^{\frac{D-2}{2}} \Gamma(-\nu)} \int_0^\infty dx \, x^{-1+\nu+\frac{D-2}{2}} e^{-\frac{b^2}{4}x} \,. \tag{A.47}$$

Recognizing the Γ -function in the last equation then yields Eq. (A.45).

Expanding the identity (A.45) around v = 0 on both sides, one can also deduce the Fourier transform of any power of the logarithm $\left[\log(q^2)\right]^n$. For instance expanding to linear order, we obtain

$$\int \frac{d^{D-2}q}{(2\pi)^{D-2}} e^{ib \cdot q} \log\left(q^2\right) = -\frac{\Gamma(1-\epsilon)}{\pi (b^2)^{1-\epsilon}}.$$
(A.48)

In the main text we also use the inverse Fourier transform (from *b* to *q*-space) which can be easily obtained from (A.45) by appropriately interchanging the roles of q and b,

$$\int d^{D-2}b \,\mathrm{e}^{-ib \cdot q} (b^2)^{-\nu} = \frac{\pi^{\frac{D-2}{2}}}{2^{2\nu+2-D}} \frac{\Gamma\left(\frac{D}{2}-1-\nu\right)}{\Gamma(\nu)} \left(q^2\right)^{1+\nu-\frac{D}{2}} \,. \tag{A.49}$$

Appendix B. The deflection angle in the probe limit

The deflection angle is a key classical observable in the $2 \rightarrow 2$ gravitational scattering which can be derived from scattering amplitudes thanks to the eikonal exponentiation. In the limit where the mass of one particle is much larger than any other energy scale in the problem, the result obtained from the eikonal approach should agree with a classical calculation describing the propagation of the other particle in the curved geometry produced by the heavy one. In order to carry out explicitly such a calculation, we need to know how the two particles couple to the massless fields in the theory under consideration. These couplings determine the classical solution for the massless fields produced by the heavy object, and this solution in turn determines the classical trajectory of the light probe in such background, *neglecting* its backreaction. In this appendix we will discuss some explicit examples of such probe-limit calculations for GR, brane scattering and $\mathcal{N} = 8$ supergravity.

B.1. Geodesic motion in Schwarzschild

The simplest case is of course that of a scalar particle with a large mass. While it is possible to describe its gravitational field by following a diagrammatic approach [376] (see also [48,377,378]), deriving the *D*-dimensional Schwarzschild metric order by order in the large distance expansions $R_s/r \ll 1$, here we will use directly the exact (Schwarzschild–Tangherlini) black-hole solution in *D* spacetime dimensions,

$$ds^{2} = -\left(1 - \left(\frac{R_{s}}{r}\right)^{D-3}\right)dt^{2} + \left(1 - \left(\frac{R_{s}}{r}\right)^{D-3}\right)^{-1}dr^{2} + r^{2}d\Omega_{D-2}^{2}.$$
(B.1)

The precise relation between the Schwarzschild radius R_s and the mass of the heavy scalar M is

$$R_{s}^{D-3} = \frac{16\pi GM}{(D-2)\Omega_{D-2}} = \frac{8\Gamma(\frac{D-1}{2})GM}{\pi^{\frac{D-3}{2}}(D-2)},$$
(B.2)

where in the last step we used the area of the *n*-dimensional sphere of unit radius $\Omega_n = \frac{2\pi^{\frac{n+1}{2}}}{\Gamma(\frac{n+1}{2})}$. One can write the action for a minimally coupled probe of mass m_p as

$$S = \frac{1}{2} \int d\tau \left(e(\tau)^{-1} \frac{dx^{\mu}}{d\tau} \frac{dx^{\nu}}{d\tau} g_{\mu\nu} - m_p^2 e(\tau) \right) , \qquad (B.3)$$

where τ parametrizes the trajectory and $e(\tau)$ is an auxiliary variable defining the world-line metric. The action is invariant under reparametrization $\tau \rightarrow \tau'(\tau)$ if x^{μ} is a scalar $(x'^{\mu}(\tau') = x^{\mu}(\tau))$ and $e'(\tau')d\tau' = e(\tau)d\tau$. Then we can choose a parametrization (a "gauge") in which $e(\tau)$ is constant and the equation of motions for $e(\tau)$ become a constraint. We shall also take into account that the motion takes place in the plane determined by the initial velocity and by the impact parameter, indicating with ϕ the angle in this plane. For later convenience, we write the constraint coming from the variation of $e(\tau)$ for a metric of the form (B.1) but with generic functions of the radial coordinate r: $ds^2 = g_{tt}dt^2 + g_{rr}dr^2 + r^2g_{\phi\phi}d\phi^2$,

$$|g_{tt}| \left(\frac{dt}{d\tau}\right)^2 - g_{rr} \left(\frac{dr}{d\tau}\right)^2 - r^2 g_{\phi\phi} \left(\frac{d\phi}{d\tau}\right)^2 = e^2 m_p^2 . \tag{B.4}$$

Since we work with metrics that do not depend explicitly on the time and are spherically symmetric, we obtain the following conservation laws for the energy E and the angular momentum J of the probe

$$eE_p = |g_{tt}| \frac{dt}{d\tau} , \qquad eJ = r^2 g_{\phi\phi} \frac{d\phi}{d\tau} .$$
(B.5)

By using these results in (B.4) we obtain

$$\frac{dr}{d\tau} = \pm \left[\frac{e^2 E_p^2}{|g_{tt}|g_{rr}} - \frac{e^2}{g_{rr}} \left(\frac{J^2}{r^2 g_{\phi\phi}} + m_p^2 \right) \right]^{\frac{1}{2}},\tag{B.6}$$

and

$$\frac{d\phi}{dr} = \pm \frac{1}{r^2} \left[\frac{g_{\phi\phi}^2}{|g_{tt}|g_{rr}} \frac{E_p^2}{J^2} - \frac{g_{\phi\phi}}{g_{rr}} \left(\frac{1}{r^2} + g_{\phi\phi} \frac{m_p^2}{J^2} \right) \right]^{-\frac{1}{2}}.$$
(B.7)

Here \pm refer to the incoming/outgoing portion of the trajectory, since a scattering process $r(\pm \tau) \rightarrow \infty$ and there is an inversion point r_* corresponding to the largest root of $\frac{dr}{d\tau}$

$$\left[\frac{E_p^2}{|g_{tt}|g_{rr}} - \frac{1}{g_{rr}}\left(\frac{J^2}{r^2 g_{\phi\phi}} + m_p^2\right)\right]_{r_*} = 0.$$
(B.8)

Thus the scattering angle reads

$$\Theta = 2 \int_{r_*}^{\infty} dr \left(\frac{d\phi}{d\tau}\right) \left(\frac{dr}{d\tau}\right)^{-1} - \pi$$

$$= 2J \int_{r_*}^{\infty} \frac{dr}{r^2} \left[\frac{g_{\phi\phi}^2 E_p^2}{|g_{tt}|g_{rr}} - \frac{g_{\phi\phi}}{g_{rr}} \left(\frac{J^2}{r^2} + g_{\phi\phi}m_p^2\right)\right]^{-\frac{1}{2}} - \pi$$
(B.9)

For the Schwarzschild metric we have $|g_{tt}| = g_{rr}^{-1} = 1 - \left(\frac{R_s}{r}\right)^{D-3}$ and $g_{\phi\phi} = 1$, so (B.9) reduces to an incomplete elliptic integral.

P. Di Vecchia, C. Heissenberg, R. Russo et al.

For our purposes it is useful to write an explicit perturbative solution in the PM expansion. We first start from (B.8) which for Schwarzschild's case reads

$$E_p^2 - \left(1 - \left(\frac{R_s}{r_*}\right)^{D-3}\right) \left(\frac{J^2}{r_*^2} + m_p^2\right) = 0.$$
(B.10)

. . . .

For general *D* we can solve this constraint perturbatively: the leading contribution is obtained by ignoring the term proportional to $(R_s/r_*)^{D-3}$ and then it is straightforward to find the corrections in an expansion for large *J*

$$r_{*} = \frac{J}{\sqrt{E_{p}^{2} - m_{p}^{2}}} \left[1 + \sum_{n=1}^{\infty} \frac{E_{p}^{2} c_{n}}{(E_{p}^{2} - m_{p}^{2})} \left(\sqrt{E_{p}^{2} - m_{p}^{2}} \frac{R_{s}}{J} \right)^{n(D-3)} \right],$$

$$c_{1} = -\frac{1}{2}, \quad c_{2} = \frac{(5 - 2D)E_{p}^{2} + 4m_{p}^{2}}{8(E_{p}^{2} - m_{p}^{2})},$$

$$c_{3} = -\frac{(D - 3)(3D - 8)E_{p}^{4} - 4(3D - 8)E_{p}^{2}m_{p}^{2} + 8m_{p}^{4}}{16(E_{p}^{2} - m_{p}^{2})^{2}}, \quad \dots$$
(B.11)

Then we can write (B.9) in the Schwarzschild case after introducing the variable $u = r_*/r$

$$\Theta = \frac{2J}{r_*} \int_0^1 du \left[E_p^2 - \left(1 - \left(\frac{R_s u}{r_*} \right)^{D-3} \right) \left(\frac{J^2 u^2}{r_*^2} + m_p^2 \right) \right]^{-\frac{1}{2}} - \pi .$$
(B.12)

We can then use (B.11) in the equation above to rewrite the integrand in term of *J* instead of r_* and then expand it for small values $\sqrt{E_p^2 - m_p^2} R_s/J$. The integral in *u* can be performed in terms of Euler's Beta and we obtain

$$\Theta = \sum_{n=1}^{\infty} \Theta_n \left(\frac{\sqrt{E_p^2 - m_p^2} R_s}{J} \right)^{n(D-3)}$$
(B.13)

whose explicit terms up to 3PM are

$$\Theta_1 = \frac{\sqrt{\pi} \Gamma\left(\frac{D}{2} - 1\right) \left[(D-2)E_p^2 - m_p^2 \right]}{2\Gamma\left(\frac{D-1}{2}\right) (E_p^2 - m_p^2)},\tag{B.14}$$

$$\Theta_2 = \frac{\sqrt{\pi}\Gamma\left(D - \frac{5}{2}\right)}{8\Gamma(D-2)} \frac{(2D-5)(2D-3)E_p^4 - 6(2D-5)E_p^2m_p^2 + 3m_p^4}{(E_p^2 - m_p^2)^2},$$
(B.15)

$$\Theta_{3} = \frac{\sqrt{\pi}\Gamma\left(\frac{3D}{2} - 4\right)}{48\Gamma\left(\frac{3D}{2} - \frac{7}{2}\right)} (E_{p}^{2} - m_{p}^{2})^{-3} \\
\times \left[(3D - 8)(3D - 6)(3D - 4)E_{p}^{6} - 15(3D - 8)(3D - 6)E_{p}^{4}m_{p}^{2} \\
+ 45(3D - 8)E_{p}^{2}m_{p}^{4} - 15m_{p}^{6} \right] .$$
(B.16)

For convenience, let us write out the D = 4 expressions as well:

$$\Theta_1 = \frac{2E_p^2 - m_p^2}{E_p^2 - m_p^2} \,, \tag{B.17}$$

$$\Theta_2 = \frac{3\pi \left(5E_p^2 - m_p^2\right)}{16 \left(E_p^2 - m_p^2\right)},\tag{B.18}$$

$$\Theta_3 = \frac{-120E_p^4 m_p^2 + 60E_p^2 m_p^4 + 64E_p^6 - 5m_p^6}{12\left(E_p^2 - m_p^2\right)^3} \,. \tag{B.19}$$

It is also instructive to study the result for the deflection angle in the PN (as opposed to PM) expansion, restricting for simplicity to D = 4. For this purpose, it is convenient to introduce the variables v_{∞} , j_{PN} (as in (3.13)) and α according to,

$$E_p = m_p \sqrt{1 + v_{\infty}^2}, \qquad v_{\infty} = \frac{1}{j_{\text{PN}}\alpha}, \qquad J = GMm_p j_{\text{PN}}.$$
 (B.20)

In these variables, Eq. (B.10) for the inversion point takes the following form

$$1 + \frac{1}{(j_{\rm PN}\alpha)^2} - \left(1 + \frac{G^2 M^2 j_{\rm PN}^2}{r_*^2}\right) \left(1 - \frac{2GM}{r_*}\right) = 0.$$
(B.21)
The PN limit can be then introduced by considering the scaling limit

$$j_{PN} \sim \mathcal{O}(c), \qquad G \sim \mathcal{O}(c^{-2}), \qquad \alpha \sim \mathcal{O}(1) \quad \text{as } c \to \infty,$$
(B.22)

which reflects the fact that the angular momentum becomes large and the velocity small, while keeping $\alpha = GMm_p/(Jv_{\infty})$ of order one. It is straightforward to solve (B.21) perturbatively in the limit (B.22), obtaining

$$r_* = GM\left[j_{\rm PN}^2\alpha(\sqrt{1+\alpha^2}-\alpha) - \left(1+\frac{\alpha}{\sqrt{1+\alpha^2}}\right)\right] + \mathcal{O}(c^{-4})$$
(B.23)

for the OPN and 1PN contributions. Substituting into (B.12), expanding in the same limit and performing the resulting elementary integrals, one then obtains the following PN-expanded deflection angle

$$\Theta = 2 \arctan \alpha + \frac{2}{j_{\text{PN}}^2} \left[3 \left(\arctan \alpha + \frac{\pi}{2} \right) + \frac{3\alpha^2 + 2}{\alpha(1 + \alpha^2)} \right] + \mathcal{O}(c^{-4}).$$
(B.24)

This calculation actually reproduces the deflection angle up to 1PN independently of the probe-limit approximation, i.e. the first two lines of Eq. (45) of [189].

B.2. Geodesic motion in D-brane metric

For the analysis of the string-brane scattering it is useful to discuss the geodetic motion in the gravitational backreaction of a stack of Dp-branes. It is straightforward to adapt the analysis of the previous section to the case where the metric is given by (4.98). By writing the transverse space in polar coordinate this metric takes the following form

$$ds^{2} = [H(r)]^{-\frac{1}{2}} \eta_{\alpha\beta} dx^{\alpha} dx^{\beta} + [H(r)]^{\frac{1}{2}} \left(dr^{2} + r^{2} d\phi^{2} d\Omega_{d-p-3}^{2} \right) , \qquad (B.25)$$

where H(r) is the harmonic function given in (4.99). We can then read the components of the metric involved in the geodesic equations (B.4) and (B.5)

$$|g_{tt}|^{-1} = g_{rr} = g_{\phi\phi} = \sqrt{H} , \qquad (B.26)$$

Since we are focusing on the motion of light string state with a large kinetic energy, we can neglect the terms involving the probe mass and from (B.7) we obtain

$$\frac{d\phi}{dr} = \frac{\pm b_J}{r^2 \sqrt{1 - \left(\frac{b_j}{r}\right)^2 + \left(\frac{R_p}{r}\right)^{d-p-3}}} = \pm \frac{b_J}{r^2 F} , \quad \Leftrightarrow \quad \frac{d\phi}{d\rho} = \frac{\pm \hat{b}_J}{\sqrt{1 - \hat{b}_J^2 \rho^2 + \rho^{d-p-3}}} , \tag{B.27}$$

where $F = \pm \sqrt{H(r) - \frac{b_J}{r^2}}$, $\rho = R_p/r$ and $\hat{b}_J = b_J/R_p$. One can follow the same steps as in the previous section to find the scattering angle by integrating (B.9), where the turning point r_* (or ρ_*) is defined as in (B.8). At the leading PM order we get

$$\Theta_1 = \sqrt{\pi} \, \frac{\Gamma\left(\frac{d-p-2}{2}\right)}{\Gamma\left(\frac{d-p-3}{2}\right)} \left(\frac{R_p}{r}\right)^{d-p-3} \,. \tag{B.28}$$

For completeness let us quote also some exact result when the integral in (B.9) can be expressed in terms of elementary functions

$$p = d - 4 \Rightarrow \Theta = 2 \arctan\left(\frac{1}{2\hat{b}_J}\right), \quad p = d - 5 \Rightarrow \Theta = \frac{\pi \hat{b}_J}{\sqrt{\hat{b}_J^2 - 1}} - \pi$$
 (B.29)

and when (B.9) reduces to a complete elliptic integral such as the case of a stack of D3-branes in type II theories

$$p = d - 7 \Rightarrow \Theta = 2\sqrt{1 + k_3^2}, K(k_3) - \pi, \quad k_3 = \frac{\dot{b}_J}{2} \left(\hat{b}_J - \sqrt{\hat{b}_J^2 - 4} \right) - 1.$$
 (B.30)

Here *K* is the complete elliptic integral of the first kind

$$K(k_3) = \int_0^1 \frac{dx}{\sqrt{(1-x^2)(1-k_3^2x^2)}} \,. \tag{B.31}$$

Notice that when we have 3 transverse directions (i.e. p = d - 4) the full deflection angle in the probe limit is determined by the leading eikonal as a consequence of supersymmetry as pointed out in [211].

We conclude this appendix by providing some details on the integrals in (4.112) relevant for the semiclassical dynamics of a string probe (rather than just a point-like object) in the D-brane metric. Starting from (4.112b), it is convenient to

proceed as done for (4.112a): we can neglect the second term in the square parenthesis since it scales as $(R_p/b)^{2(d-p-3)}$ and rewrite the first term as a total derivative with respect to the radial coordinate. Then we have

$$\frac{1}{2} \int_{-\infty}^{\infty} du \ \mathcal{G}_{\hat{y}} \simeq \int_{0}^{\infty} d\bar{r} \ \partial_{\bar{r}} \left[\frac{F}{\sqrt{H}} \partial_{\bar{r}} \ln\left(\bar{r}\sin\bar{\phi}H^{\frac{1}{4}}\right) \right] = - \left. \frac{b_J \cos\bar{\phi}}{\bar{r}^2 \sqrt{H}\sin\bar{\phi}} \right|_{\bar{r}=r_*} \simeq -\frac{\Theta_1}{b} \ . \tag{B.32}$$

The only non-trivial contribution arises when the derivative inside the square parenthesis acts on $\ln \sin \bar{\phi}$, since this produces a factor of $\frac{d\bar{\phi}}{dr}$ which, thanks to (B.27), cancel the overall factor of *F* and yields a finite contribution as $r \to r_*$. In the final step we used $b_J \simeq b$, $\bar{\phi}(r_*) = \frac{\Theta + \pi}{2}$ and (B.28). Then one can check that the result in (B.32) agrees with the one in (4.112b) with $\mu_{\hat{y}}^2$ defined in (4.88). When evaluating (4.112c), it is more convenient to evaluate explicitly the derivatives for both terms in the square parenthesis obtaining

$$\frac{1}{2} \int_{-\infty}^{\infty} du \ \mathcal{G}_0 = \int_0^{\infty} du \left[\frac{\partial_u^2 \left(\bar{r} F H^{-\frac{1}{4}} \right)}{\bar{r} F H^{-\frac{1}{4}}} \right] = \int_{r_*}^{\infty} \frac{d\bar{r}}{\bar{r} F H^{-\frac{1}{4}}} \ \partial_{\bar{r}} \left[\frac{F}{H^{\frac{1}{2}}} \partial_{\bar{r}} \left(\bar{r} F H^{-\frac{1}{4}} \right) \right] . \tag{B.33}$$

Then it is straightforward to evaluate the expression above to 1PM order obtaining

$$\frac{1}{2} \int_{-\infty}^{\infty} du \ \mathcal{G}_0 \simeq \int_b^{\infty} d\bar{r} \frac{d-p-3}{4\bar{r}\sqrt{\bar{r}^2-b^2}} \left[(d-p-4) - (d-p-1)\frac{b^2}{\bar{r}^2} \right] \left(\frac{R_p}{\bar{r}}\right)^{d-p-3},\tag{B.34}$$

where we used again $r_* \simeq b$ at 1PM. Thanks to the relation

$$\int_{b}^{\infty} dr \frac{r^{1-2x}}{r^{2}-b^{2}} = \frac{\sqrt{\pi}b^{1-2x}\Gamma\left(x-\frac{1}{2}\right)}{2\Gamma(x)},$$
(B.35)

one can check that, after performing the integral, (B.34) reproduces (4.112c) with μ_0^2 defined in (4.88).

B.3. Probe limit of the $\mathcal{N} = 8$ case

In order to perform the classical calculation for the massive N = 8 case it is easier to start from the setup discussed in Section 3.1.4 and dualize it to a frame where all the massive objects are D*p*-branes. This can be done by the following chain of dualities:

- lift the setup to M-theory and then go back to type IIA by compactifying along the 9th direction; in this way we the first particle is transformed into a bound state of *n*₁ D0-branes, while the second particle is unchanged;
- perform T-dualities along the directions 7, 8, 9 so that the external states become a bound state of n_1 D3-branes and a F1-string wrapped n_2 times along the direction 8;
- since we are now in type IIB string theory we can perform a S-duality and obtain bound states of n_1 D3-branes and n_2 D1-branes;
- for convenience we can T-dualize back to type IIA along the 8th direction to get n_1 D2-branes and n_2 D0-branes.

All objects in each step are point-like in the noncompact directions and the string frame metric produced by the bound state of D2-branes is

$$ds^{2} = \left(1 + \frac{4GM}{r}\right)^{-\frac{1}{2}} \left(-dt^{2} + dx_{7}^{2} + dx_{8}^{2}\right) + \left(1 + \frac{4GM}{r}\right)^{\frac{1}{2}} dx_{\perp}^{2} .$$
(B.36)

The classical solution includes also a non-trivial RR 2-form, which does not play any role in this problem, and a dilaton⁵⁰

$$e^{\phi} = \left(1 + \frac{4GM}{r}\right)^{\frac{1}{4}} , \qquad (B.38)$$

where *M* is the mass of the D2-brane bound state. The action for the D0-brane probes involves both the fields above and read

$$S = -m_{p=0} \int d\tau e^{-\phi} \sqrt{\left| g_{\mu\nu} \frac{dx^{\mu}}{d\tau} \frac{dx^{\nu}}{d\tau} \right|}, \qquad (B.39)$$

⁵⁰ If the number of noncompact direction is D we have

$$\frac{\Gamma\left(\frac{D-3}{2}\right)}{\pi^{\frac{D-1}{2}}}\frac{\kappa_D T_2 n_1}{2} = \frac{\Gamma\left(\frac{D-3}{2}\right)}{\pi^{\frac{D-3}{2}}} 4G_D M , \qquad (B.37)$$

and the harmonic function is $1 + \frac{\Gamma(\frac{D-3}{2})4G_DM}{\pi^{\frac{D-3}{2}}r^{D-3}}$.

where of course we restrict the motion in the noncompact space. This action is equivalent to (B.3) with an effective metric $g_{\mu\nu}^{\text{eff}} = e^{-2\phi}g_{\mu\nu}$. We can then calculate the deflection angle by using (B.9) with

$$g_{00}^{\text{eff}} = \left(1 + \frac{4GM}{r}\right)^{-1}, \quad g_{rr}^{\text{eff}} = g_{\phi\phi}^{\text{eff}} = 1.$$
 (B.40)

Then in this case the deflection angle is given by a circular integral

$$\Theta = 2J \int_{r_*}^{\infty} \frac{dr}{r^2} \left[\left(1 + \frac{4GM}{r} \right) E_p^2 - \left(\frac{J^2}{r^2} + m_p^2 \right) \right]^{-\frac{1}{2}} - \pi$$

= 2 arctan $\left(\frac{2GM}{b_J} \frac{\sigma^2}{\sigma^2 - 1} \right),$ (B.41)

where used $E_p = m_p \sigma$ and $J = m_p b \sqrt{\sigma^2 - 1}$.

Appendix C. Some relations satisfied by the Levi-Civita tensor in D = 4

In this Appendix we provide some relations satisfied by the product of two the four-dimensional Levi-Civita tensors. We list them here

$$\epsilon^{\mu\nu\rho\sigma}\epsilon_{\mu\nu\rho\sigma} = -4! \tag{C.1}$$

$$\epsilon^{\mu\nu\rho\sigma}\epsilon_{\mu\nu\rho\delta} = -6\delta^{\sigma}_{\delta} \tag{C.2}$$

$$\epsilon^{\mu\nu\rho\sigma}\epsilon_{\mu\nu\gamma\delta} = -2\delta^{\rho\sigma}_{\gamma\delta} \tag{C.3}$$

$$\epsilon^{\mu\nu\rho\sigma}\epsilon_{\mu\beta\gamma\delta} = -\delta^{\nu\rho\sigma}_{\beta\gamma\delta} \tag{C.4}$$

and

$$\epsilon^{\mu\nu\rho\sigma}\epsilon_{\alpha\beta\gamma\delta} = -\delta^{\mu}_{\alpha}\delta^{\nu\rho\sigma}_{\beta\gamma\delta} + \delta^{\mu}_{\beta}\delta^{\nu\rho\sigma}_{\alpha\gamma\delta} - \delta^{\mu}_{\gamma}\delta^{\nu\rho\sigma}_{\alpha\beta\delta} + \delta^{\mu}_{\delta}\delta^{\nu\rho\sigma}_{\alpha\beta\gamma}$$
(C.5)

where

$$\delta^{\rho\sigma}_{\gamma\delta} = \delta^{\rho}_{\gamma}\delta^{\sigma}_{\delta} - \delta^{\rho}_{\delta}\delta^{\sigma}_{\gamma} \tag{C.6}$$

and

$$\delta^{\nu\rho\sigma}_{\beta\gamma\delta} = \delta^{\nu}_{\beta} \left(\delta^{\rho}_{\gamma} \delta^{\sigma}_{\delta} - \delta^{\rho}_{\delta} \delta^{\sigma}_{\gamma} \right) - \delta^{\nu}_{\gamma} \left(\delta^{\rho}_{\beta} \delta^{\sigma}_{\delta} - \delta^{\rho}_{\delta} \delta^{\sigma}_{\beta} \right) + \delta^{\nu}_{\delta} \left(\delta^{\rho}_{\beta} \delta^{\sigma}_{\gamma} - \delta^{\rho}_{\gamma} \delta^{\sigma}_{\beta} \right) . \tag{C.7}$$

We use the convention where $\epsilon_{0123} = 1$.

Appendix D. String theory background

In this appendix we provide some details on how to derive the relevant amplitudes in the context of the bosonic theory which is technically simpler than the superstring case while capturing all the main features.

D.1. String theory conventions

Here we collect our string conventions and a short discussion about the boundary state that is needed to describe the stack of Dp-branes used in the main text.

D.1.1. Scales and coupling constants

Free string theory in flat spacetime is described by the Nambu-Goto action:

$$S_{\text{string}} = -T \int d\tau \int_0^{\pi} d\sigma \sqrt{(\dot{x} \cdot x')^2 - \dot{x}^2 x'^2},$$

$$\dot{x} \equiv \partial_{\tau} x(\sigma, \tau), \qquad x' \equiv \partial_{\sigma} x(\sigma, \tau), \qquad \dot{x} \cdot x' \equiv \eta_{\mu\nu} \dot{x}^{\mu} x'^{\nu},$$
(D.1)

where the double integral is nothing but the area swept by the string and *T* is the classical string tension with dimensions of an energy per unit length (in units in which c = 1). At the classical level (e.g. in the context of cosmic strings) *T* is the only free parameter and the classical equations of motion are obviously independent of it.⁵¹ The inverse of *T* has

⁵¹ When moving in a non trivial geometry the size of the string relative to the characteristic scale of the geometry does instead matter at the classical level.

the same dimensions as an angular momentum per squared mass (recall that we have set c = 1) and has been denoted traditionally by $2\pi \alpha'$ since it first made its appearance in hadronic physics as the slope of the linear Regge trajectories: $\alpha(t) = \alpha(0) + \alpha' t$.

In a quantum context what matters is the action in Planck units, a dimensionless quantity, and this necessarily introduces a fundamental area ℓ_s^2 as the natural replacement of Planck's constant in string theory [277]. In this review we adopt the following relation between these various quantities:

$$\frac{S_{\text{string}}}{\hbar} = \frac{T}{\hbar} (\text{Area Swept}) = \frac{\text{Area Swept}}{2\pi \ell_s^2}, \qquad \ell_s = \sqrt{\alpha' \hbar} = \sqrt{\frac{\hbar}{2\pi T}}, \qquad (D.2)$$

where we should note the analogy with the definition of the Planck length in four dimensions, $\ell_P = \sqrt{G\hbar}$, with *G* replacing the role of α' . The similarity goes even further: classical GR has no intrinsic length (or mass) scale while quantum gravity does. Physically, ℓ_s plays the role of a minimal length scale, a minimal size for fundamental quantum strings, and a shortdistance cutoff regularizing quantum corrections. Its inverse, $M_s \equiv \frac{\hbar}{\ell_s}$ is the energy/mass scale associated with string excitations and with the cutoff in momentum space. In string theory it is thus natural to take ℓ_s as the basic unit of length and to express any other quantity, according to its dimensionality, in terms of ℓ_s , c, \hbar , and of dimensionless numbers. The latter, when not directly fixed, are associated with the value of dimensionless scalar fields, called moduli.

In the rest of this Appendix, and elsewhere in this review, we will follow the common practice of using interchangeably α' and ℓ_s^2 (i.e. set $\hbar = 1$) except when the distinction is physically relevant. As an example, $\alpha(t)$ is dimensionless (being an exponent characterizing power-like Regge behavior) and therefore α' cannot be just the inverse of *T*. What appears in the above expression for $\alpha(t)$ is actually $\frac{\alpha' t}{\hbar} = \frac{t}{M_s^2}$, but we shall omit the \hbar^{-1} factor throughout.

In perturbative string theory the strength of the gravitational interaction depends on α' and on the moduli of the theory as follows:

$$2\kappa_d^2\hbar = 16\pi G_d\hbar = 16\pi \ell_d^{d-2} = 2^{-\frac{d-10}{2}} (2\pi)^{d-3} g_s^2 (\alpha'\hbar)^{\frac{d-2}{2}} = 2^{-\frac{d-10}{2}} (2\pi)^{d-3} g_s^2 \ell_s^{d-2},$$
(D.3)

where ℓ_d is the *d*-dimensional Planck length, g_s , the string coupling, is related to the expectation value of the dilaton and of course we have d = 10 for critical superstring theory and d = 26 for the bosonic theory. Note that at very weak string coupling $\ell_d/\ell_s \ll 1$. This physically means that string-size corrections intervene well before quantum gravity loops get out of control. That does not mean, however, that straight perturbation theory is always reliable at $g_s^2 \ll 1$: although there is a formal loop expansion in powers of g_s^2 , these can be enhanced by large kinematical factors such as powers of the energy or infrared singularities as discussed in the main body of this report.

When some of the directions are compactified on a manifold of volume V_c and only D directions are non-compact, it is useful to introduce the Newton constant $\kappa_D^2 = 8\pi G_D$ for the D-dimensional theory appropriate for describing physics at distances much larger than the size of the compact dimensions:

$$\kappa_D^2 = \frac{\kappa_d^2}{V_c} = 8\pi \,\ell_D^{D-2} \,. \tag{D.4}$$

It follows that the ratio ℓ_D/ℓ_s is further reduced if $V_c \gg \ell_s^{d-D}$, the case of "large extra dimensions". In this report only the case of string-size extra dimensions will be considered.

The spectrum of type II and bosonic string theories also contains Dp-branes, non-perturbative objects which can support the end-points of open strings [248]. In its simplest configuration a Dp-brane enforces Neumann boundary conditions on the string fields along p spatial directions and time, and Dirichlet boundary conditions along the remaining d - (p + 1) directions. In other words, the end-points of an open string move on a (p + 1)-dimensional Minkowskian submanifold. The tension (i.e. energy per unit p-dimensional surface) of these objects is

$$\tau_p = \frac{T_p}{\kappa_d}$$
, with $T_p = 2^{-\frac{d-10}{4}} \sqrt{\pi} (2\pi \sqrt{\alpha'})^{\frac{d-2p-4}{2}}$, (D.5)

which of course fixes its coupling to gravity. In the superstring case the Dp-branes are minimally coupled to the (p + 1) Ramond–Ramond potential with a RR charge density μ_p given by $\mu_p = \sqrt{2T_p}$ (i.e. charge per unit p-volume).

In Section 3.2, we use a stack of *N* D*p*-branes as a target in a (thought) scattering experiment with fundamental strings. It is then convenient to introduce the scale R_p of the geometry sourced by the D*p*-branes

$$R_p^{d-p-3} = \frac{\Gamma\left(\frac{d-p-3}{2}\right)}{\pi^{\frac{d-p-1}{2}}} \frac{g_s N}{4} \frac{(2\pi\sqrt{\alpha'})^{d-p-3}}{2^{\frac{d-10}{2}}} .$$
(D.6)

In terms of the tension T_p we have

$$\frac{\kappa_d T_p N}{2} = \frac{\pi^{\frac{d-p-1}{2}} R_p^{d-p-3}}{\Gamma\left(\frac{d-p-3}{2}\right)} \,. \tag{D.7}$$

D.1.2. String mode expansion

Our conventions on the string coordinates and their mode expansion in the closed string case are

$$X^{Im}(z, z) = X_{L}^{m}(z) + X_{R}^{m}(z) ,$$

$$X_{L}^{M}(z) = q_{L} - i\frac{\alpha'}{2}p_{L}^{M}\ln z + i\sqrt{\frac{\alpha'}{2}}\sum_{n\neq 0}\frac{\alpha_{n}^{M}}{n}z^{-n} ,$$

$$X_{R}^{M}(\bar{z}) = q_{R} - i\frac{\alpha'}{2}p_{R}^{M}\ln \bar{z} + i\sqrt{\frac{\alpha'}{2}}\sum_{n\neq 0}\frac{\bar{\alpha}_{n}^{M}}{n}\bar{z}^{-n} .$$
(D.8)

In the closed string sector the modes α_n and $\bar{\alpha}_n$ are independent and, after canonical quantization, satisfy the commutation relations

$$[\alpha_n^M, \alpha_m^N] = \eta^{MN} n \,\delta_{n+m} \,, \quad [\bar{\alpha}_n^M, \bar{\alpha}_m^N] = \eta^{MN} n \,\delta_{n+m} \,, \quad [\alpha_n^M, \bar{\alpha}_m^N] = 0 \,. \tag{D.9}$$

It is convenient to take also the left/right center of mass and momentum operators to be independent

$$[q_L^M, p_L^N] = \frac{i}{2} \eta^{MN}, \quad [q_R^M, p_R^N] = \frac{i}{2} \eta^{MN}; \quad [q_L^M, p_R^N] = [q_R^M, p_L^N] = 0$$
(D.10)

and impose, for the non-compact directions, an identification on their eigenvalues of the physical states $p_t^M = p_R^M = p^M$, where p^M is the total momentum. It is also convenient to introduce $\sqrt{\frac{\alpha'}{2}}p_L^M = \alpha_0^M$ and $\sqrt{\frac{\alpha'}{2}}p_R^M = \bar{\alpha}_0^M$, so we have

$$\partial X^{M} = -i\sqrt{\frac{\alpha'}{2}} \sum_{n} \alpha_{n}^{M} z^{-n-1} , \quad \bar{\partial} X^{M} = -i\sqrt{\frac{\alpha'}{2}} \sum_{n} \bar{\alpha}_{n}^{M} \bar{z}^{-n-1} .$$
 (D.11)

Then we have the following Operator Product Expansions (OPE)

$$\partial X^{M}(z_{1})\partial X^{N}(z_{2}) \sim -\frac{\alpha'}{2} \frac{\eta^{MN}}{(z_{1}-z_{2})^{2}} + \cdots$$

$$e^{ik_{1}X_{L}(z_{1})}e^{ik_{2}X_{L}(z_{2})} \sim (z_{1}-z_{2})^{\frac{\alpha'}{2}k_{1}k_{2}}e^{i(k_{1}+k_{2})X_{L}(z_{2})} + \cdots$$
(D.12)

and similarly for the anti-holomorphic part.

D.1.3. Normalizations for string amplitudes We associate a factor of $\frac{\kappa_d}{2\pi}$ to each string vertex operator and indicate with C_{S_2} the normalization of the closed string tree-level amplitudes where the worldsheet has the topology of a sphere. Imposing perturbative unitarity on the factorization of a 4-point function into two 3-point amplitudes one obtains

$$C_{S_2} \left(\frac{\kappa_d}{2\pi}\right)^2 \frac{\alpha'}{8\pi} = 1 . \tag{D.13}$$

The factor of $\alpha'/(8\pi)$ is related to the closed string propagator

$$P(a) = \frac{\alpha'}{2} (L_0 + \tilde{L}_0 - 2a)^{-1} = \frac{\alpha'}{8\pi} \int_{|z| \le 1} \frac{d^2 z}{|z|^2} z^{L_0 - a} \bar{z}^{\tilde{L}_0 - a} , \qquad (D.14)$$

where a = 1 in the bosonic theory and $a = \frac{1}{2}$ in the NS-NS sector of type II theories. The overall normalization of the left-hand side is chosen so to have residue one at various mass poles (for instance, close to the tachyon pole, behaves as $1/(p^2 + m_T^2)$, with $m_T^2 = -4/\alpha'$, while on the right-hand side we use the definition of the measure $d^2z = i dz \wedge d\overline{z} = 2 d\operatorname{Re}(z) d\operatorname{Im}(z)$ as in [265].

D.1.4. The boundary state

In this subsection we give a short introduction to the boundary state that describes the Dp-branes. For more details see for instance [229]. Dp-branes are extended p dimensional objects characterized by the fact that open strings have their endpoints attached to them. The open string with the endpoints at $\sigma = 0, \pi$ attached to two parallel Dp-branes satisfies the usual Neumann boundary conditions along the directions longitudinal to the world volume of the branes

$$\partial_{\sigma} X^{\alpha}|_{\sigma=0,\pi} = 0 \qquad \qquad \alpha = 0, 1, \dots, p \tag{D.15}$$

and Dirichlet boundary conditions along the directions transverse to the brane

$$X^{i}|_{\sigma=0} = y^{i} ; X^{i}|_{\sigma=\pi} = w^{i} \qquad \qquad i = p+1, \dots, d-1$$
(D.16)

where y^i and w_i are the coordinates of the two Dp-branes and we take σ and τ in the two intervals $0 \le \sigma \le \pi$ and $0 \leq \tau \leq T$.

The previous are the conditions in the so-called open string channel, but those characterizing the boundary state are instead those in the so-called closed string channel. This nomenclature follows from the fact that the annulus diagram can be constructed in two ways: as one-loop of open strings or as a tree diagram with a closed string propagators connecting two boundary states. These two descriptions are connected by a conformal transformation in terms of the variable $\zeta \equiv \sigma + i\tau$:

$$\zeta = \sigma + i\tau \to -i\zeta = \tau - i\sigma . \tag{D.17}$$

After the inversion $\sigma \to -\sigma$ the previous conformal transformation simply amounts to exchange σ with τ and viceversa

$$(\sigma, \tau) \to (\tau, \sigma) \,. \tag{D.18}$$

Finally in order to have the closed string variables σ and τ to vary in the intervals $\sigma \in [0, \pi]$ and $\tau \in [0, \hat{T}]$ corresponding to a closed string propagating between the two D branes one must perform the following conformal rescaling

$$\sigma \to \frac{\pi}{T}\sigma \qquad \tau \to \frac{\pi}{T}\tau$$
, (D.19)

with $\hat{T} = \pi^2 / T$.

The equations characterizing the boundary state are obtained by applying the conformal transformation previously constructed to the boundary conditions for the open string given in (D.15) and (D.16). At $\tau = 0$ we get the following conditions:

$$\partial_{\tau} X^{\alpha}|_{\tau=0}|B_X\rangle = 0 \qquad \alpha = 0, \dots, p , \qquad (D.20)$$

$$X^{i}|_{\tau=0}|B_{X}\rangle = y^{i}$$
 $i = p + 1, \dots, d - 1$. (D.21)

Analogous conditions can be obtained for the Dp-branes at $\tau = \hat{T}$.

The previous equations can be easily written in terms of the closed string oscillators by making use of the expansion in Eq. (D.8), obtaining

$$\begin{aligned} (\alpha_n^{\alpha} + \widetilde{\alpha}_{-n}^{\alpha})|B_X\rangle &= 0 , \quad (\alpha_n^i - \widetilde{\alpha}_{-n}^i)|B_X\rangle &= 0 \quad \forall n \neq 0 , \\ \hat{p}^{\alpha}|B_X\rangle &= 0 \quad (\hat{q}^i - y^i)|B_X\rangle &= 0 , \end{aligned}$$
(D.22)

where $\hat{q} = q_L + q_R$ and $\hat{p} = p_L = p_R$. Introducing the reflection matrix

$$\mathcal{R}_{p}^{\mu\nu} = (\eta^{\alpha\beta}, -\delta^{ij}), \qquad (D.23)$$

the equations for the non-zero modes can be rewritten as

$$(\alpha_n^{\mu} + \mathcal{R}_p^{\mu} \widetilde{\alpha}_{-n}^{\nu})|B_X\rangle = 0 \quad \forall \ n \neq 0 .$$
(D.24)

The state satisfying the previous equations can easily be determined to be

$$|B_X\rangle = \frac{T_p}{2}\delta^{d-p-1}(\hat{q}^i - y^i) \left(\prod_{n=1}^{\infty} e^{-\frac{1}{n}\alpha_{-n}\mathcal{R}_p\widetilde{\alpha}_{-n}}\right) |0\rangle_{\alpha}|0\rangle_{\widetilde{\alpha}}|p=0\rangle , \qquad (D.25)$$

where the normalization $\frac{T_p}{2}$ is fixed by imposing that the computation of the annulus diagram in the open and in the closed string channel gives the same result.

D.2. String-brane scattering: the bosonic theory at tree-level

The simplest possible setup where we can study the string theory leading eikonal is the scattering of two closed string tachyons of momenta p_1 and p_2 off a stack of *N* D*p*-branes in bosonic string theory. The tree-level string amplitude capturing the scattering process mentioned above is

$$\mathcal{A}_0^T = \frac{\kappa_d N T_p}{2} \frac{\Gamma(-\alpha' E_s^2 - 1) \Gamma\left(-\frac{\alpha'}{4}t - 1\right)}{\Gamma\left(-\alpha' E_s^2 - \frac{\alpha'}{4}t - 2\right)} , \qquad (D.26)$$

where the kinematics is identical to the one discussed in the superstring case after (3.117). It is useful to provide some detail on the derivation of Eq. (D.26) as this will help in clarifying the key physics novelty of the string eikonal with respect to the field theory setup.

As standard we start from the vertex operators describing the emission of the external states. For the closed string tachyon we have

$$\mathcal{V}_{T}(z_{i},\bar{z}_{i}) = \frac{\kappa_{d}}{2\pi} \hat{\mathcal{V}}_{T}(z_{i},\bar{z}_{i}) = \frac{\kappa_{d}}{2\pi} e^{ip_{i}X(z_{i},\bar{z}_{i})} , \qquad (D.27)$$

where $X^M(z_i, \bar{z}_i)$ are the string embedding coordinates given in (D.8), the momenta are on-shell $p_i^2 = \frac{4}{\alpha'}$ (see Appendix D.1 for our string theory conventions) and the exponentials are understood to be normal ordered. The tree-level amplitude corresponds to world-sheet with the topology of the disk with the two closed string insertions at two points (z_1 , z_2) in its interior

$$\mathcal{A}_{0}^{T} = C_{S_{2}} \frac{\alpha' \kappa_{d}}{8\pi} N \int \frac{d^{2} z_{1} d^{2} z_{2}}{d V_{SL(2,R)}} \langle 0 | \mathcal{V}_{T}(z_{1}, \bar{z}_{1}) \mathcal{V}_{T}(z_{2}, \bar{z}_{2}) | B \rangle .$$
(D.28)

Here $|B\rangle$ is the boundary state describing the stack of Dp-branes, see Eq. (D.25), $|0\rangle$ is the SL(2, C) invariant vacuum and our conventions on the string normalizations are summarized in Appendix D.1.3. The effect of $|B\rangle$ on the world-sheet fields is to identify the holomorphic and the anti-holomorphic parts through the reflection matrix \mathcal{R}_p which is the identity along the Dp-branes and minus the identity in the transverse directions. After the identification, the anti-holomorphic fields are placed at $1/\bar{z}_2$ and $1/\bar{z}_1$, while the holomorphic ones are inside the disk of unit radius at z_1 and z_2 and |z| = 1 represents the world-sheet boundary. Finally the measure can be easily obtained by fixing three punctures, say $1/\bar{z}_2$, $1/\bar{z}_1$ and z_2 and by integrating over the remaining one inserting the contribution of the *c*-ghost

$$\frac{d^{2}z_{1}d^{2}z_{2}}{dV_{SL(2,R)}} = dz_{1} \langle c\left(\frac{1}{\bar{z}_{2}}\right) c\left(\frac{1}{\bar{z}_{1}}\right) c(z_{2}) \rangle$$

$$= dz_{1}\left(\frac{1}{\bar{z}_{2}} - \frac{1}{\bar{z}_{1}}\right) \left(\frac{1}{\bar{z}_{1}} - z_{2}\right) \left(\frac{1}{\bar{z}_{2}} - z_{2}\right) = \left(\frac{1}{\bar{z}_{2}} - z_{1}\right)^{2} \left(\frac{1}{\bar{z}_{1}} - z_{2}\right)^{2} dx$$
(D.29)

where in the last step we introduced the SL(2, R) invariant cross-ratio

$$x = \frac{(z_1 - z_2)\left(\frac{1}{\bar{z}_2} - \frac{1}{\bar{z}_1}\right)}{\left(\frac{1}{\bar{z}_1} - z_2\right)\left(\frac{1}{\bar{z}_2} - z_1\right)}.$$
 (D.30)

The cross-ratio *x* is manifestly real and, since the z_i 's are inside the disk of unit radius, it lies between zero and one. In particular, the limit $x \to 0$ corresponds to $|z_1 - z_2| \to 0$ describing a world-sheet where the two external states interact first via a 3-point closed string vertex with one leg glued to the disk. Then by using the on-shell conditions, momentum conservation $(1 + \mathcal{R}_p)p_1 + (1 + \mathcal{R}_p)p_2 = 0$ and the ordering $\frac{1}{|z_2|} > \frac{1}{|z_1|} > |z_1| > |z_2|$ we have

$$\begin{aligned} \mathcal{A}_{0}^{T} &= \frac{\kappa_{d}T_{p}}{2}N\int_{0}^{1}dx\left(\frac{1}{\bar{z}_{1}}-z_{2}\right)^{2}\left(\frac{1}{\bar{z}_{2}}-z_{1}\right)^{2}\left[\left(\frac{1}{\bar{z}_{1}}-z_{1}\right)^{\frac{\alpha'}{2}p_{1}\mathcal{R}_{p}p_{1}}\left(\frac{1}{\bar{z}_{1}}-z_{2}\right)^{\frac{\alpha'}{2}p_{1}\mathcal{R}_{p}p_{2}}\right.\\ &\left.\left(\frac{1}{\bar{z}_{2}}-\frac{1}{\bar{z}_{1}}\right)^{\frac{\alpha'}{2}p_{1}p_{2}}\left(z_{1}-z_{2}\right)^{\frac{\alpha'}{2}p_{1}p_{2}}\left(\frac{1}{\bar{z}_{2}}-z_{1}\right)^{\frac{\alpha'}{2}p_{1}\mathcal{R}_{p}p_{2}}\left(\frac{1}{\bar{z}_{2}}-z_{2}\right)^{\frac{\alpha'}{2}p_{2}\mathcal{R}_{p}p_{2}}\right] \end{aligned} \tag{D.31} \\ &=\frac{\kappa_{d}T_{p}}{2}N\int_{0}^{1}dx\left(1-x\right)^{-\alpha'E_{s}^{2}-2}x^{-\frac{\alpha'}{4}t-2}, \end{aligned}$$

where we used (D.13) the normalization factor $\frac{T_p}{2}$ of the boundary state (see Eq. (D.25)) to simplify the overall normalization and

$$p_1 p_2 = \frac{-t + m_1^2 + m_2^2}{2}, \quad p_r \mathcal{R}_p p_r = -2E_s^2 + m_r^2, \quad p_1 \mathcal{R}_p = 2E_s^2 + \frac{t - m_1^2 - m_2^2}{2}, \quad (D.32)$$

where $m_i^2 = -\frac{4}{\alpha'}$. Thanks to the integral representation of the Euler Beta function one obtains (D.26).

Exactly as in the field theory setup the eikonal is captured by the limit $E_s^2 \gg |t|$ and by using (3.119)

$$\mathcal{A}_{0}^{T} \simeq \frac{\kappa_{d} N T_{p}}{2} (-\alpha' E_{s}^{2})^{1+\frac{\alpha' t}{4}} \Gamma\left(-1-\frac{\alpha' t}{4}\right) = \frac{\kappa_{d} N T_{p}}{2} (\alpha' E_{s}^{2})^{1+\frac{\alpha' t}{4}} \frac{e^{-i\pi \frac{\alpha' t}{4}}}{-\frac{\alpha' t}{4}} \frac{\Gamma\left(1-\frac{\alpha' t}{4}\right)}{1+\frac{\alpha' t}{4}} .$$
(D.33)

Notice that this result can be directly derived directly from the last line (D.31). When $\alpha' E_s^2 \gg 1$ then the *x*-integral can be performed by focusing on the region $x \to 0$ with $\alpha' E_s^2 x$ finite: by expanding the integrand in this regime and finally extending the region of integration to infinity, we have

$$\mathcal{A}_{0}^{T} \simeq \frac{\kappa_{d} T_{p}}{2} N \int_{0}^{\infty} dx \, e^{\alpha' E_{s}^{2} x} x^{-\frac{\alpha' t}{4} - 2} = \frac{\kappa_{d} T_{p}}{2} N(-\alpha' E_{s}^{2})^{1 + \frac{\alpha' t}{4}} \, \Gamma\left(-\frac{\alpha' t}{4} - 1\right) \,, \tag{D.34}$$

which agrees with (D.33). This shows explicitly that the result is entirely captured by the limit discussed before (D.31). We can now follow the same approach used in Section 3.2.1 for the superstring and perform the Fourier Transform (3.123)

of the bosonic result. Then we obtain

$$2\delta_{0} = \frac{\kappa_{d}NT_{p}E_{s}}{4\pi} \frac{\Gamma\left(1 + \frac{\alpha'\nabla_{b}^{2}}{4}\right)}{1 - \frac{\alpha'\nabla_{b}^{2}}{4}} \left[(\pi b^{2})^{-\frac{d-4-p}{2}} \gamma\left(\frac{d-4-p}{2}, \frac{b^{2}}{Y}\right) \right],$$
(D.35)

which is very similar to (3.129) except for the tachyonic pole and the fact that the critical dimension is d = 26. We recall that $\gamma(z, a)$ denotes the incomplete Γ -function.

D.2.1. The Reggeon vertex formalism for bosonic strings

1

In this section we follow [149,225,226] and show that the *t*-channel pole in (D.33) is related to a particular class of closed string states in the leading Regge trajectory. Actually it is not difficult to present the argument for the scattering of general closed string states instead of focusing on tachyons (D.27), so we consider directly the case where we have generic physical vertex operators \mathcal{V}_1 and \mathcal{V}_2 in (D.28) at the place of two \mathcal{V}_T . Notice that our discussion does not require the two vertex operators to be equal and this will be exploited later in Section 3.2.4.

As suggested by the discussion after Eq. (D.33), the eikonal result can be obtained by focusing on the degeneration channel where the world-sheet looks like a vertex involving the three closed string states with one of them being offshell and propagating until it interacts with the stack of Dp-branes. By inserting a complete set of states between the 3-point interaction and the boundary state we have

$$\mathcal{A}_{0}^{(12)} = C_{S_{2}} \frac{\alpha' \kappa_{d}}{8\pi} N \left[\sum_{\mathcal{Q}_{\ell}} \int \frac{d^{2}z}{2\pi} (z\bar{z})^{\ell - 2 - \frac{\alpha' t}{4}} \langle \mathcal{V}_{1} \mathcal{V}_{2} \mathcal{Q}_{\ell} \rangle \langle \mathcal{Q}_{\ell} | B \rangle \right] , \qquad (D.36)$$

where Q_{ℓ} is a generic state of momentum $q = (p_1 + p_2)$ at level ℓ and the sum, of course, covers all possible levels $\ell = 0, 1, \ldots$; the integral over the phase of z is trivial, as the level matching condition has already been implemented, and the factor involving $z\bar{z} \equiv x$ follows from the closed string propagator (D.14). We now need to see how the contributions of different states Q_{ℓ} in (D.36) scale with the energy of the incident string and, in the eikonal limit, we would like to focus on the leading terms. In order to do so it is convenient to decompose the polarizations of the states Q_{ℓ} by using the vectors (e^{\pm}) defined in (3.161) and the directions orthogonal to them. Factors of E_s can arise from the contractions between $\partial^r X^+ = e_M^+ \partial^r X^M$ in Q_ℓ and the universal exponential factors $e^{ip_{1,2}X}$ in $\mathcal{V}_{1,2}$. Thus, in the high-energy regime, the decomposition (D.36) is dominated by the states

$$Q_{\ell} = \frac{1}{\ell!} \left(i \sqrt{\frac{2}{\alpha'}} \partial X^+ \right)^{\ell} \left(i \sqrt{\frac{2}{\alpha'}} \bar{\partial} X^+ \right)^{\ell} e^{iqX} .$$
(D.37)

Below we summarize the contractions between the world-sheet coordinates in \mathcal{Q}_{ℓ} and $\mathcal{V}_{1,2}$ that are relevant for evaluating (D.36)

$$i\sqrt{\frac{2}{\alpha'}}\partial^{r}X^{+}(z)e^{ip_{1}X(w)} \sim (\sqrt{\alpha'}E_{s})\,\partial^{r-1}\left(\frac{1}{z-w}\right)e^{ip_{1}X(w)}$$

$$i\sqrt{\frac{2}{\alpha'}}\partial^{r}X^{+}(z)e^{ip_{2}X(w)} \sim -(\sqrt{\alpha'}E_{s})\,\partial^{r-1}\left(\frac{1}{z-w}\right)e^{ip_{2}X(w)}.$$
(D.38)

Another source of factors of E_s is the contraction of $\partial^r X^+(z)$ with the tensor part of the external vertex operator when this describe a massive state. In this case we have

$$i\sqrt{\frac{2}{\alpha'}}\partial^r X^+(z) \, i\partial^s X^M(w) \sim (\sqrt{\alpha'}E_s) \, \frac{v_i^M}{m} \partial_z^{r-1} \partial_w^{s-1} \frac{1}{(z-w)^2} \,, \tag{D.39}$$

where v_i^M indicates the longitudinal polarization for a massive state of spatial momentum \vec{p}_i . It is then clear that the leading contributions to (D.36) come from the states Q_{ℓ} that have the highest number of factors of X^+ , so, at level ℓ , it is convenient to have $(\partial X^+)^\ell$ rather then structures with higher derivatives such as $\partial^r X^+$. Thus in summary, in the high-energy regime, the relevant states exchanged between the incident string and the stack of Dp-branes are⁵²

$$Q_{\ell}^{R} = \frac{1}{\ell!} \left(i \sqrt{\frac{2}{\alpha'}} \partial X^{+} \right)^{\ell} \left(i \sqrt{\frac{2}{\alpha'}} \bar{\partial} X^{+} \right)^{\ell} e^{i q X(z, \bar{z})} .$$
(D.40)

Beside being off-shell these states do not satisfy exactly the BRST constraint also because $e^+q \sim 1/E$, but in the highenergy regime this violation becomes irrelevant so it does not affect our calculation. Notice that at level ℓ such state

⁵² The overall factor of $1/\ell!$ ensures that the two point function is normalized to one.

yields a contribution proportional to $(\sqrt{\alpha'}E_s)^{2\ell}$, so only the graviton ($\ell = 1$) yields a result compatible with a classical eikonal contribution, while all contributions of higher spin states ($\ell \ge 2$) grow too quickly with the energy. It is only after resumming the contributions of the whole tower of states in the leading Regge trajectory that one finds the eikonal result (D.34).

The basic idea of the Reggeon vertex approach is to perform the sum over the states (D.40) formally at the operator level. We first notice that the scalar product $\langle Q_{\ell}^{R}|B\rangle$ is independent of ℓ thanks to the normalization in (D.40) and one is left with only the normalization of the boundary state in (D.25). Thus, when focusing on the contributions of the states (D.40), Eq. (D.36) reads

$$\begin{aligned} \mathcal{A} &\simeq \frac{NT_p \kappa_d}{2} \langle \hat{\mathcal{V}}_1 \hat{\mathcal{V}}_2 \int_0^\infty dx \sum_{\ell=0}^\infty \frac{1}{\ell!} \left[x \left(i \sqrt{\frac{2}{\alpha'}} \partial X^+ i \sqrt{\frac{2}{\alpha'}} \bar{\partial} X^+ \right) \right]^\ell x^{-2 - \frac{\alpha' t}{4}} e^{iqX} \rangle \\ &= \frac{NT_p \kappa_d}{2} \langle \hat{\mathcal{V}}_1 \hat{\mathcal{V}}_2 \int_0^\infty dx e^{x \frac{2}{\alpha'} i \partial X^+ i \bar{\partial} X^+} x^{-2 - \frac{\alpha' t}{4}} e^{iqX} \rangle \end{aligned}$$

$$\begin{aligned} &= \frac{NT_p \kappa_d}{2} \langle \hat{\mathcal{V}}_1 \hat{\mathcal{V}}_2 \left[i \sqrt{\frac{2}{\alpha'}} \partial X^+ i \sqrt{\frac{2}{\alpha'}} \bar{\partial} X^+ \right]^{1 + \frac{\alpha' t}{4}} e^{iqX} \rangle \Gamma \left(-1 - \frac{\alpha' t}{4} \right) \\ &= \frac{NT_p \kappa_d}{2} e^{-i \frac{\alpha' t}{4}} \Gamma \left(-1 - \frac{\alpha' t}{4} \right) \langle \hat{\mathcal{V}}_1 \hat{\mathcal{V}}_2 \mathcal{V}_R \rangle = \Pi_R^{D_p} \langle \hat{\mathcal{V}}_1 \hat{\mathcal{V}}_2 \mathcal{V}_R \rangle . \end{aligned}$$

$$\end{aligned}$$

where the hat on the vertices means that they are stripped of their normalization as in (D.27). As already mentioned, $x = z\bar{z}$ and, at high energy, we can extend the integral over *x* from the interval (0, 1) to $(0, \infty)$. In the manipulations above we treated the combination appearing the square parenthesis as a positive quantity as its leading contribution when inserted in a correlator is αE_s^2 as follows from (D.38). Then the integral over *x* has to be evaluated via an analytic continuation on E_s^2 and this is the origin of the phase in the final step. The symbols $\Pi_R^{D_p}$ and \mathcal{V}_R indicate the Reggeon propagator glued to the boundary state and the Reggeon vertex

$$\Pi_{R}^{D_{p}} = \frac{NT_{p}}{2} e^{-i\frac{\alpha' t}{4}} \Gamma\left(-1 - \frac{\alpha' t}{4}\right) , \qquad (D.42)$$

$$\mathcal{V}_{R} = \kappa_{d} \left[\sqrt{\frac{2}{\alpha'}} i\partial X^{+} \sqrt{\frac{2}{\alpha'}} i\bar{\partial} X^{+}\right]^{1 + \frac{\alpha' t}{4}} e^{iqX} . \qquad (D.43)$$

It is convenient to separate a Reggeon propagator and its coupling to the Dp-branes $\Pi_R^{D_p} = \Pi_R \langle \mathcal{V}_R | B \rangle$ as follows

$$\Pi_{R} = \frac{1}{2\pi} e^{-i\frac{\alpha' t}{4}} \frac{\Gamma\left(-1 - \frac{\alpha' t}{4}\right)}{\Gamma\left(2 + \frac{\alpha' t}{4}\right)} , \qquad \langle \mathcal{V}_{R}|B \rangle = 2\pi \frac{NT_{p}}{2} \Gamma\left(2 + \frac{\alpha' t}{4}\right) . \tag{D.44}$$

The motivation for doing so is that the Reggeon vertex $\hat{\nu}_R$ and propagator Π_R can be used to derive the high-energy limit of pure closed string amplitudes, as done in Section 3.2.6. At this stage the split in (D.44) is somewhat arbitrary, but it can be justified by sketching how to adapt the steps in (D.41) to the tree-level amplitudes with four external closed strings. In this case, the integrand in (D.36) takes the schematic form $(z\bar{z})^{\ell-2-\frac{\alpha' t}{4}} \langle v_1 v_2 Q_\ell \rangle \langle Q_\ell v_3 v_4 \rangle$ and the holomorphic and the anti-holomorphic sectors are independent. It is then convenient to factorize the calculation by summing independently over the number of the holomorphic and the anti-holomorphic insertions since the level matching condition is imposed at the end by the integral over the phase of *z*. By focusing on the operator part, we need to consider the integral

$$\int d^{2}z \sum_{\ell,\bar{\ell}=0}^{\infty} \left[\frac{1}{\sqrt{\ell!}} \left(z \, i \sqrt{\frac{2}{\alpha'}} \partial X^{+} \right)^{\ell} \frac{1}{\sqrt{\bar{\ell}!}} \left(\bar{z} \, i \sqrt{\frac{2}{\alpha'}} \bar{\partial} X^{+} \right)^{\ell} \right] (z\bar{z})^{-2-\frac{\alpha' t}{4}} \\ \left[\frac{1}{\sqrt{\ell!}} \left(i \sqrt{\frac{2}{\alpha'}} \partial X^{+} \right)^{\ell} \frac{1}{\sqrt{\bar{\ell}!}} \left(i \sqrt{\frac{2}{\alpha'}} \bar{\partial} X^{+} \right)^{\bar{\ell}} \right].$$
(D.45)

A standard approach is to rewrite the factor of $(z\bar{z})^{-2-\frac{\alpha' t}{4}}$ as an exponential by introducing a Schwinger parameter

$$(z\bar{z})^{-2-\frac{\alpha' t}{4}} = \int_0^\infty d\tau \ e^{-|z|^2 \tau} \frac{\tau^{1+\frac{\alpha' t}{4}}}{\Gamma\left(2+\frac{\alpha' t}{4}\right)} \ . \tag{D.46}$$

Then the 2D integral over z in (D.45) is Gaussian and the final integral over $\tau' = 1/\tau$ yields a Γ -function. Then one can see that the tree-level amplitude with four closed strings is captured, at high energy, by the correlator $\langle \hat{\mathcal{V}}_1 \hat{\mathcal{V}}_2 \hat{\mathcal{V}}_R \rangle \Pi_R \langle \hat{\mathcal{V}}_R \hat{\mathcal{V}}_3 \hat{\mathcal{V}}_4 \rangle$ involving the same Reggeon vertex (D.43) as before and the propagator Π_R in (D.44).

It is straightforward to follow the steps discussed in Section 3.2.4 and obtain the bosonic eikonal operator from the Reggeon vertex (D.43). The final result has the same structure as in the superstring case (3.175), just with the overall factor which follows from (D.44)

$$\hat{\mathcal{A}}_{0} \simeq \frac{NT_{p}\kappa_{d}}{2}e^{-i\frac{\alpha' t}{4}} \Gamma\left(-1 - \frac{\alpha' t}{4}\right) (\alpha' E_{s}^{2})^{1 + \frac{\alpha' t}{4}} \int_{0}^{2\pi} \frac{d\sigma}{2\pi} : e^{iq\hat{X}} : .$$
(D.47)

D.3. String-brane scattering: the bosonic theory at one-loop

In the closed string channel the annulus amplitudes can be evaluated using two boundary states. For instance the one-loop correction to the amplitude in (D.28) is

$$\mathcal{A}_{1}^{T} = C_{S^{2}} \left(\frac{\alpha' \kappa_{d}}{8\pi} N\right)^{2} \left(\frac{\kappa_{d}}{2\pi}\right)^{2} \frac{1}{4\pi} \int \frac{d^{2}q}{|q|^{2}} d^{2}z_{1} d^{2}z_{2} \langle B | \hat{\mathcal{V}}_{T}(z_{1}) \hat{\mathcal{V}}_{T}(z_{2}) q^{L_{0}-1} \bar{q}^{\bar{L}_{0}-1} | B \rangle , \qquad (D.48)$$

where, as in (D.28), we inserted a factor of $\frac{\alpha' \kappa_d}{8\pi} N$ for each closed string propagator P (D.14), a factor of $\frac{\kappa_d}{2\pi}$ for each vertex and a symmetry factor $\frac{1}{4\pi}$ related to the residual symmetries of the annulus. The contribution of the zero modes q^{μ} and p^{μ} to (D.48) is

$$(2\pi)^{p+1}\delta^{(p+1)}(p_1+p_2)(2\pi^2\alpha'\lambda)^{-\frac{d-p-1}{2}}e^{\frac{\alpha'}{2\pi\lambda}\left[\left(E_s^2+\frac{4}{\alpha'}\right)\left(\log\frac{|z_1|}{|z_2|}\right)^2-t\log|z_1|\log|z_2|\right]},$$
(D.49)

where we introduced λ via $\log |q| = -\pi \lambda$ and used the on-shell conditions for the external tachyons $m_1^2 = m_2^2 = -4/\alpha'$. The contribution of the non-zero modes has two effects. First it yields the usual annulus measure

$$d\mu_1 = 2\pi d\lambda \frac{1}{|q|^2} \prod_{n=1}^{\infty} \frac{1}{\left(1 - |q|^{2n}\right)^{d-2}} .$$
(D.50)

Then it transforms the disk Green function $\log(z_i - z_i)$ into the annulus one, which can be expressed in terms of the prime form $\log E(z_i, z_i)$, where

$$E(z_i, z_j) = (z_i - z_j) \prod_{n=1}^{\infty} \frac{\left(1 - |q|^{2n} \frac{z_i}{z_j}\right) \left(1 - |q|^{2n} \frac{z_j}{z_i}\right)}{(1 - |q|^{2n})^2} ,$$
(D.51)

or in terms of the Jacobi θ -function (4.77)

$$E(z_i, z_j) = 2\pi i \ e^{i\pi(v_i + v_j)} \frac{\theta_1(v_i - v_j | i\lambda)}{\theta_1'(0|i\lambda)} \ .$$
(D.52)

Then we can generalize the disk integrand derived in Appendix D.2 to the annulus topology. As done in Section 4.2 we introduce the variables

$$z_i = e^{2\pi i v_i} \equiv e^{2\pi i (i\lambda\rho_1 - \omega_1)}, \quad z_j = e^{2\pi i v_j} \equiv e^{2\pi i (i\lambda\rho_2 - \omega_2)}$$
(D.53)

and obtain the following result for the one-loop amplitude

$$\begin{aligned} \mathcal{A}_{1}^{T} &= \left(\frac{\kappa_{d}T_{p}N}{2}\right)^{2} \frac{\alpha'}{16\pi} (2\pi^{2}\alpha')^{-\frac{d-p-1}{2}} (2\pi)^{4} \int_{0}^{\infty} \frac{d\lambda}{\lambda^{\frac{d-p-5}{2}}} \frac{1}{|q|^{2}} \prod_{n=1}^{\infty} \frac{1}{(1-|q|^{2n})^{d-2}} \\ &\times 4 \int_{0}^{\frac{1}{2}} d\rho_{1} \int_{0}^{\frac{1}{2}} d\rho_{2} \int_{0}^{1} d\omega_{1} \int_{0}^{1} d\omega_{2} \ e^{-(\alpha' E_{s}^{2}+2)V_{s} - \frac{\alpha' t}{4}V_{t}} \frac{(\theta_{1}'(0|i\lambda))^{4} \ e^{4\pi\lambda\rho^{2}}}{(2\pi)^{4}\theta_{1}^{2}(i\lambda\rho - \omega|i\lambda)\theta_{1}^{2}(-i\lambda\rho - \omega|i\lambda)} , \end{aligned}$$
(D.54)

where the factor 4 in the second line comes from the normalization of d^2z_1 and d^2z_2 discussed after (D.14), we switched to the variables introduced in (4.78) and the functions V_s , V_t are exactly those appearing in the superstring amplitude defined in (4.76). As in the superstring case, the kinematic configuration we are interested in (large E_s and small R_p/b) implies that the integral is dominated by the region of small ρ and large λ . In this limit we have

$$\mathcal{A}_{1}^{T} \sim \left(\frac{\kappa_{d}T_{p}N}{2}\right)^{2} \frac{\alpha'}{16\pi} (2\pi^{2}\alpha')^{-\frac{d-p-1}{2}} (2\pi)^{4} \int_{0}^{\infty} \frac{d\lambda}{\lambda^{\frac{d-p-5}{2}}} e^{2\pi\lambda} \int_{0}^{1} d\zeta \int_{0}^{1} d\omega$$

$$\times 2 \int_{\mathcal{R}(\zeta)} d\rho \ e^{2\pi\alpha' E_{s}^{2}\lambda\rho^{2}} e^{2\pi\lambda\zeta(1-\zeta)\frac{\alpha't}{4}} (4\sin^{2}\pi\omega)^{-\frac{\alpha't}{4}}$$

$$\times \exp\left[4\alpha' E_{s}^{2}\sin^{2}(\pi\omega) \left(e^{-2\pi\lambda\zeta} + e^{-2\pi\lambda(1-\zeta)}\right)\right] \left(4\sin^{2}(\pi\omega)\right)^{-2} ,$$
(D.55)

where the region of integration $\mathcal{R}(\zeta)$ is defined after (4.79). The final factor in the last line comes from the last fraction in Eq. (D.54) and is not present in the superstring case as one can see by comparing (D.54) and (4.79). As we will soon see,

P. Di Vecchia, C. Heissenberg, R. Russo et al.

this difference is related to the presence of a tachyonic state in the bosonic theory. The integral over $\rho \sim 0$ is Gaussian (after a Wick rotation $E_s \rightarrow iE_e$). By writing the exponential in the last line as a double series of terms proportional to $e^{-2\pi n\lambda\zeta}$ and $e^{-2\pi m\lambda(1-\zeta)}$ we obtain an expression very similar to the integrand I_1 in Appendix A of [22]. Then the integral over ω can also be performed and one obtains

$$\mathcal{A}_{1}^{T} \sim \left(\frac{\kappa_{d}T_{p}N}{2}\right)^{2} \frac{\alpha'}{8\pi} (2\pi^{2}\alpha')^{-\frac{d-p-1}{2}} (2\pi)^{4} \int_{0}^{1} d\zeta \frac{i}{\sqrt{2\alpha' E_{s}^{2}}} (\alpha' E_{s}^{2})^{2} 4^{-\frac{\alpha' t}{4}} \\ \times \sum_{n,m=0}^{\infty} \frac{1}{n!m!} \left(4\alpha' E_{s}^{2}\right)^{(n-1)+(m-1)} \frac{1}{\pi} B\left(\frac{1}{2}, n+m-2-\frac{\alpha' t}{4}+\frac{1}{2}\right) \\ \times \Gamma\left(-\frac{d-p-6}{2}\right) \pi^{\frac{d-p-6}{2}} \left[-2\zeta(1-\zeta)\frac{\alpha' t}{4}+2(n-1)\zeta+2(m-1)(1-\zeta)\right]^{\frac{d-p-6}{2}}.$$
(D.56)

where the last line comes from the integral over λ . This expression is very similar to the superstring case, except that n, m are shifted to n - 1, m - 1, due to the presence of the tachyon pole $(e^{2\pi\lambda})$ in the first line of (D.55) and of the last factor $(\sin^{-4}(\pi\omega))$ in the final line of the same equation. We can trade the integral over ζ for a momentum integral in D = d - p - 2 dimensional space by using the identity

$$\int_{0}^{1} d\zeta \ \Gamma\left(-\frac{d-p-6}{2}\right) \left[-2\zeta(1-\zeta)\frac{\alpha' t}{4} + 2(n-1)\zeta + 2(m-1)(1-\zeta)\right]^{\frac{d-p-6}{2}}$$
$$= (2\pi\alpha')^{\frac{d-p-2}{2}} \int \frac{d^{d-p-2}\mathbf{k}}{(2\pi)^{d-p-2}} \left[2(n-1) + \frac{\alpha'}{2}\mathbf{k}^{2}\right]^{-1} \left[2(m-1) + \frac{\alpha'}{2}(\mathbf{k}-\mathbf{q})^{2}\right]^{-1}, \tag{D.57}$$

where we used bold symbols to indicate the transverse (d-2) vectors following the convention introduced in (1.37). We can rewrite the sums as integrals by using

$$\sum_{m=0}^{\infty} \frac{1}{m!} \frac{f(m)s^m}{m+t} = -\int_{\mathcal{C}} \frac{dm}{2\pi i} e^{-i\pi m} \Gamma(-m) \frac{f(m)s^m}{m+t} , \qquad (D.58)$$

where the contour includes all the poles in the $\Gamma(-m)$ and not the other ones. We can then deform the contour and focus on the poles of the propagators in (D.57), which are the only ones that contribute to the leading term in the energy. We find

$$\begin{aligned} \mathcal{A}_{1}^{T} &\sim \left(\frac{\kappa_{d}T_{p}N}{2}\right)^{2} \frac{\alpha'}{8\pi} (2\pi^{2}\alpha')^{-\frac{d-p-1}{2}} (2\pi)^{4} \frac{i\pi^{\frac{d-p-6}{2}}}{\sqrt{2\alpha' E_{s}^{2}}} \frac{1}{4} (\alpha' E_{s}^{2})^{2} 4^{-\frac{\alpha'}{4}} \\ &\times (2\pi\alpha')^{\frac{d-p-2}{2}} \int \frac{d^{d-p-2}\mathbf{k}}{(2\pi)^{d-p-2}} \Gamma\left(-1+\frac{\alpha'}{4}\mathbf{k}^{2}\right) \Gamma\left(-1+\frac{\alpha'}{4}(\mathbf{q}-\mathbf{k})^{2}\right) e^{i\pi\frac{\alpha'}{4}\mathbf{k}^{2}+i\pi\frac{\alpha'}{4}(\mathbf{q}-\mathbf{k})^{2}} \\ &\times \left(4\alpha' E_{s}^{2}\right)^{-\frac{\alpha'}{4}\mathbf{k}^{2}-\frac{\alpha'}{4}(\mathbf{q}-\mathbf{k})^{2}} \frac{1}{\pi} B\left(\frac{1}{2}, -\frac{\alpha'}{4}\mathbf{k}^{2}-\frac{\alpha'}{4}(\mathbf{q}-\mathbf{k})^{2}-\frac{\alpha' t}{4}+\frac{1}{2}\right) + \cdots, \end{aligned}$$
(D.59)

where the dots stand for the contributions of the other poles that involve also the Euler Beta-function in the second line of (D.56). These contributions are needed to cancel spurious singularities in (D.59). For instance the last line has a pole when $\alpha'(\mathbf{k}^2 - \mathbf{kq}) \simeq 1$ which does not correspond to the propagation of a physical state. However, this region is suppressed by a factor of $(\alpha' E_s^2)^{-\frac{1}{2}}$ with respect to the $\alpha' \mathbf{k}^2$, $\alpha' \mathbf{q}^2 \ll 1$ since we have $(4\alpha' E_s^2)^{-\frac{\alpha'}{4}\mathbf{k}^2 - \frac{\alpha'}{4}(\mathbf{q} - \mathbf{k})^2} \sim (4\alpha' E_s^2)^{-\frac{\alpha'}{4}\mathbf{q}^2 - \frac{1}{2}}$, thus it is not reliably captured by (D.56). In particular at this subleading order one should add the contribution neglected in (D.56). Selecting a pole from the Euler Beta-function and the other from (D.57), one obtains a structure that exactly cancels the spurious pole in (D.56) mentioned above. We thus focus on the leading contribution written in (D.56) and recast the Euler Beta-function as the correlator introduced (4.83). By using Eq. (D.33) we can rewrite (D.59) in the following factorized form

$$\mathcal{A}_{1}^{T} \sim \left[\frac{\alpha'}{8\pi} (2\pi^{2}\alpha')^{-\frac{d-p-1}{2}} (2\pi)^{4} \frac{i\pi^{\frac{d-p-6}{2}}}{\sqrt{2\alpha' E_{s}^{2}}} \frac{1}{4} (2\pi\alpha')^{\frac{d-p-2}{2}} \right] \int \frac{d^{d-p-2}\mathbf{k}}{(2\pi)^{d-p-2}} \times \mathcal{A}_{0}^{T}(E_{s}, \mathbf{k}) \mathcal{A}_{0}^{T}(E_{s}, \mathbf{q} - \mathbf{k}) \langle 0| \prod_{i=1}^{2} \int_{0}^{2\pi} \frac{d\sigma_{i}}{2\pi} : e^{i\mathbf{k}\hat{X}(\sigma_{1})} :: e^{i(\mathbf{q} - \mathbf{k})\hat{X}(\sigma_{2})} : |0\rangle .$$
(D.60)

Since the square parenthesis on the first line is just $i/(4E_s)$, we indeed obtain (4.84) for h = 2 for the case of external tachyon states in bosonic string theory.

Appendix E. More on the $2 \rightarrow 3$ kinematics and on the x, + waveforms

In this appendix we provide a list of relations that apply to the 2 \rightarrow 3 kinematics discussed at the end of Section 1.2 and which can be useful, in particular, to manipulate the waveforms presented in Section 8.3.1. We also present alternative expressions for the ×, + projections of the *b*-space amplitude $\tilde{A}_0^{(5)\mu\nu}$, i.e. the leading-order gravitational waveform, that are equivalent to those presented in Section 8.3.1.

Let us begin by recalling that the incoming momentum vectors p_1 and p_2 define to the incoming particle's velocities v_1^{μ} and v_2^{μ} by (1.7), which satisfy $v_1^2 = -1$, $v_2^2 = -1$, and are related to the Lorenz factor σ by $\sigma = -v_1 \cdot v_2$ as in (1.8). The projector onto the plane spanned by v_1^{μ} and v_2^{μ} can be read off from (1.10),

$$P^{\mu}_{\nu} = -\check{v}^{\mu}_{1}v_{1\nu} - \check{v}^{\mu}_{2}v_{2\nu} = \frac{1}{\sigma^{2} - 1} \left[v^{\mu}_{1}v_{1\nu} - \sigma(v^{\mu}_{1}v_{2\nu} + v^{\mu}_{2}v_{1\nu}) + v^{\mu}_{2}v_{2\nu} \right],$$
(E.1)

where \check{v}_1^{μ} , \check{v}_2^{μ} are the dual velocities defined by (1.9). This projector of course satisfies

$$P^{\mu}_{\ \nu}v^{\nu}_{1} = v^{\mu}_{1}, \qquad P^{\mu}_{\ \nu}v^{\nu}_{2} = v^{\mu}_{2}, \qquad P^{\mu}_{\ \rho}P^{\rho}_{\ \nu} = P^{\mu}_{\ \nu}.$$
(E.2)

For any vector ξ^{μ} , we may then rewrite its decomposition (1.10) into longitudinal and transverse projections as follows,

$$\xi^{\mu} = \xi^{\mu}_{\parallel} + \xi^{\mu}_{\perp}, \qquad \xi^{\mu}_{\parallel} = P^{\mu}_{\ \nu} \xi^{\nu}, \qquad \xi_{\parallel} \cdot \xi_{\perp} = 0.$$
(E.3)

In terms of the invariant products $\sigma = -v_1 \cdot v_2 \ge 1$, $\omega_1 = -v_1 \cdot k \ge 0$ and $\omega_2 = -v_2 \cdot k$ introduced in (1.43), one can explicitly compute the longitudinal projections of various vectors. Since we always work in a regime in which the momentum transfers q_1 , q_2 are small, and thus (1.47) hold to leading order, we recover (1.48),

$$q_{1\parallel}^{\mu} \approx -\omega_2 \check{v}_2^{\mu} = \frac{\omega_2 (v_2^{\mu} - \sigma v_1^{\mu})}{\sigma^2 - 1}, \qquad q_{2\parallel}^{\mu} \approx -\omega_1 \check{v}_1^{\mu} = \frac{\omega_1 (v_1^{\mu} - \sigma v_2^{\mu})}{\sigma^2 - 1}$$
(E.4)

and define

$$\Delta_{\parallel}^{\mu} \equiv \frac{1}{2(\sigma^2 - 1)} \left[-v_1^{\mu}(\omega_1 + \sigma\omega_2) + v_2^{\mu}(\omega_2 + \sigma\omega_1) \right] \approx \frac{1}{2} (q_1 - q_2)_{\parallel}^{\mu}$$
(E.5)

for later convenience. From now on we shall not distinguish between \approx and = signs, for simplicity, since we always work to leading order in the approximation (1.47). Then from the following relations we find

$$b^2 q_{1\parallel}^2 = \frac{b^2 \omega_2^2}{\sigma^2 - 1} \equiv \Omega_2^2 , \qquad b^2 q_{2\parallel}^2 = \frac{b^2 \omega_1^2}{\sigma^2 - 1} \equiv \Omega_1^2$$
(E.6)

and

$$k \cdot q_{1\parallel} = \frac{\omega_2(-\omega_2 + \sigma\omega_1)}{\sigma^2 - 1}, \qquad k \cdot q_{2\parallel} = \frac{\omega_1(-\omega_1 + \sigma\omega_2)}{\sigma^2 - 1}.$$
(E.7)

When saturating (E.5) with the polarization vector \tilde{e}^{μ}_{θ} defined in (8.91), that is

$$\tilde{e}^{\mu}_{\theta} = \frac{\omega_1 v_2^{\mu} - \omega_2 v_1^{\mu}}{\sqrt{\mathcal{P}}} \,, \tag{E.8}$$

with $\mathcal{P} = -\omega_1^2 + 2\omega_1\omega_2\sigma - \omega_2^2$ as in (1.45), we find

$$\frac{1}{2}(q_1 - q_2)_{\parallel} \cdot \tilde{e}_{\theta} = \frac{\omega_1 \omega_2}{\sqrt{\mathcal{P}}} \,. \tag{E.9}$$

Let us now give the explicit expressions of the previously introduced quantities in terms of σ , in the center-of-mass frame, choosing the following explicit parametrization for the graviton momentum k^{μ} defined in (8.93),

$$k^{\mu} = \omega n^{\mu}, \qquad n^{\mu} = (1, \sin\theta\cos\phi, \sin\theta\sin\phi, \cos\theta). \tag{E.10}$$

We get

$$\omega_1 = \frac{\omega \left(m_1 + m_2 \sigma - \cos \theta m_2 \sqrt{\sigma^2 - 1}\right)}{\sqrt{s}}, \qquad \omega_2 = \frac{\omega \left(m_2 + m_1 \sigma + \cos \theta m_1 \sqrt{\sigma^2 - 1}\right)}{\sqrt{s}}$$
(E.11)

and⁵³

$$\sqrt{q_{1\parallel}^{2}} = \frac{\omega}{\sqrt{s}} \frac{m_{2} + m_{1}\sigma + m_{1}\sqrt{\sigma^{2} - 1}\cos\theta}{\sqrt{\sigma^{2} - 1}} = \frac{\omega_{2}}{\sqrt{\sigma^{2} - 1}},$$

$$\sqrt{q_{2\parallel}^{2}} = \frac{\omega}{\sqrt{s}} \frac{m_{1} + m_{2}\sigma - m_{2}\sqrt{\sigma^{2} - 1}\cos\theta}{\sqrt{\sigma^{2} - 1}} = \frac{\omega_{1}}{\sqrt{\sigma^{2} - 1}}$$
(E.12)

together with

$$k \cdot q_{1\parallel} = \omega^{2} \frac{\left(m_{2} + m_{1}\sigma + m_{1}\cos\theta\sqrt{\sigma^{2} - 1}\right)\left(m_{2}\sqrt{\sigma^{2} - 1} - (m_{1} + m_{2}\sigma)\cos\theta\right)}{s\sqrt{\sigma^{2} - 1}},$$

$$k \cdot q_{2\parallel} = \omega^{2} \frac{(m_{1} + m_{2}\sigma - m_{2}\cos\theta\sqrt{\sigma^{2} - 1})(m_{1}\sqrt{\sigma^{2} - 1} + (m_{2} + m_{1}\sigma)\cos\theta)}{s\sqrt{\sigma^{2} - 1}}.$$
(E.13)

Adopting the notation

$$\mathbf{k}^{\mu} = k^{\mu}_{\perp} \,, \tag{E.14}$$

we also find

$$\mathcal{P} = -\omega_1^2 + 2\omega_1\omega_2\sigma - \omega_2^2 = (\sigma^2 - 1)\mathbf{k}^2, \qquad \mathbf{k}^2 = \omega^2\sin^2\theta$$
(E.15)

In particular, the last two relations make it obvious that $P \ge 0$, consistently with (1.45).

We conclude this appendix by presenting expressions for the \times , + projections of the impact-parameter space 2 \rightarrow 3 amplitude in terms of polarization vectors e^{μ}_{θ} , e^{μ}_{ϕ} such that

$$e_{\theta} \cdot k = 0, \qquad e_{\phi} \cdot k = 0, \qquad e_{\phi} \cdot v_i = 0, \tag{E.16}$$

but such that e_{θ}^{μ} is not necessarily orthogonal to b^{μ} (in contrast with the vector \tilde{e}_{θ}^{μ} employed in Section 8.3.1, see Eq. (8.83)). We thus start from the five-point amplitude given in Eq. (5.38) (using $M = \mu$, $N = \nu$ and $\beta = \beta^{GR}$ as in (5.41) as appropriate for GR), perform the Fourier transform (8.55), thus obtaining $W_0^{\mu\nu} = \tilde{A}_0^{\mu\nu}$ by following the steps detailed in Section 8.3.1, and project it along the two polarizations according to

$$\tilde{\mathcal{A}}_{\times}^{(5)} = \tilde{\mathcal{A}}_{0}^{(5)\mu\nu} e_{\phi\mu} e_{\theta\nu} , \qquad \tilde{\mathcal{A}}_{+}^{(5)} = \tilde{\mathcal{A}}_{0}^{(5)\mu\nu} \frac{1}{2} \left(e_{\theta\mu} e_{\theta\nu} - e_{\phi\mu} e_{\phi\nu} \right) .$$
(E.17)

Explicitly, in the center-of-mass defined by (8.92) and (8.93), where the longitudinal directions are 0 and 3, the velocities have vanishing components along the 1, 2 axis, while

$$v_{1}^{0} = \frac{m_{1} + m_{2}\sigma}{\sqrt{s}}, \qquad v_{1}^{3} = \frac{m_{2}\sqrt{\sigma^{2} - 1}}{\sqrt{s}},$$

$$v_{2}^{0} = \frac{m_{2} + m_{1}\sigma}{\sqrt{s}}, \qquad v_{2}^{3} = -\frac{m_{1}\sqrt{\sigma^{2} - 1}}{\sqrt{s}}.$$
(E.18)

Then, we choose

 $e^{\mu}_{\theta} = (0, \cos\theta\cos\phi, \cos\theta\sin\phi, -\sin\theta), \qquad e^{\mu}_{\phi} = (0, -\sin\phi, \cos\phi, 0), \tag{E.19}$

so that (E.16) hold and

$$e_{\theta}^2 = 1, \qquad e_{\theta} \cdot e_{\phi} = 0, \qquad e_{\phi}^2 = 1.$$
 (E.20)

For the \times polarization, we get

$$\begin{split} \tilde{\mathcal{A}}_{\times}^{(5)} &= \frac{(8\pi G)^{\frac{3}{2}}}{4Ep(2\pi)} i(\hat{b}e_{\phi}) \left\{ \beta \left[\left(\frac{(p_{1}e_{\theta})\Omega_{1}}{(p_{1}k)b} e^{-ibk/2} K_{1}(\Omega_{1}) - \frac{(p_{2}e_{\theta})\Omega_{2}}{(p_{2}k)b} e^{ibk/2} K_{1}(\Omega_{2}) \right) \right. \\ &+ i \int_{0}^{1} dx \, e^{-i\frac{kb}{2}(2x-1)} \left((\hat{b}e_{\theta})\Omega(x) K_{1}(\Omega(x)) + i(x - \frac{1}{2}) (\mathbf{k}e_{\theta}) b K_{0}(\Omega(x)) \right) \\ &+ (e_{\theta}\Delta_{\parallel}) b \int_{0}^{1} dx \, e^{-i\frac{kb}{2}(2x-1)} K_{0}(\Omega(x)) \right] \\ &- 4p_{1}p_{2} ((p_{1}k)(p_{2}e_{\theta}) - (p_{2}k)(p_{1}e_{\theta})) \int_{0}^{1} dx \, e^{-i\frac{kb}{2}(2x-1)} b K_{0}(\Omega(x)) \right\}, \end{split}$$
(E.21)

 $[\]overline{53}$ In order to compare with the formalism used in Section 6 of Ref. [47] we need to use the following identities $c_{1,2}^2 \mathbf{k}^2 = q_{1,2\parallel}^2$, $d_{1,2} \mathbf{k}^2 = (kq_{1,2\parallel})$ and $b^2 \mathbf{k}^2 f(x) = \Omega^2 (1-x)$ where $\Omega(x)$ is given in (E.24).

$$\beta = \beta^{\rm GR} = 2m_1^2 m_2^2 (2\sigma^2 - 1)$$

while, for the polarization + we get

A

$$\begin{split} \tilde{q}_{\pm}^{(5)} &= \frac{(8\pi G)^{\frac{3}{2}}}{4kp(4\pi)} \left\{ \beta \left[\frac{(p_{1}e_{\theta})^{2}}{(p_{1}k)^{2}} e^{-ikb/2} \left((kq_{2\parallel})K_{0}(\Omega_{1}) - i(k\hat{b}) \frac{\Omega_{1}}{b}K_{1}(\Omega_{1}) \right) \right. \\ &+ \frac{(p_{2}e_{\theta})^{2}}{(p_{2}k)^{2}} e^{ikb/2} \left((kq_{1\parallel})K_{0}(\Omega_{2}) + i(k\hat{b}) \frac{\Omega_{2}}{b}K_{1}(\Omega_{2}) \right) \right. \\ &+ \frac{p_{1}e_{\theta}}{p_{1}k} e^{-ibk/2} \left((-(e_{\theta}\mathbf{k}) + 2(e_{\theta}\Delta_{\parallel}))K_{0}(\Omega_{1}) + 2i(e_{\theta}\hat{b}) \frac{\Omega_{1}}{b}K_{1}(\Omega_{1}) \right) \\ &- \frac{p_{2}e_{\theta}}{p_{2}k} e^{ibk/2} \left(((e_{\theta}\mathbf{k}) + 2(e_{\theta}\Delta_{\parallel}))K_{0}(\Omega_{2}) + 2i(e_{\theta}\hat{b}) \frac{\Omega_{2}}{b}K_{1}(\Omega_{2}) \right) \\ &+ 2(e_{\theta}\Delta_{\parallel})^{2} \int_{0}^{1} dx \, e^{-i\frac{kb}{2}(2x-1)} \left(\frac{b^{2}}{2} \frac{K_{1}(\Omega(x))}{\Omega(x)} \right) + 2(e_{\theta}\Delta_{\parallel}) \\ &\times \int_{0}^{1} dx e^{-i\frac{kb}{2}(2x-1)} \left((e_{\theta}\mathbf{k})(1 - 2x) \left(\frac{b^{2}}{2} \frac{K_{1}(\Omega(x))}{\Omega(x)} \right) + i(e_{\theta}\hat{b})bK_{0}(\Omega(x)) \right) \\ &+ \frac{1}{2} \int_{0}^{1} dx e^{-i\frac{kb}{2}(2x-1)} \left[2(e_{\theta\perp}e_{\theta\perp})K_{0}(\Omega(x)) - 2(e_{\theta}\hat{b})^{2}\Omega(x)K_{1}(\Omega(x)) \\ &+ (e_{\theta}\mathbf{k})^{2}(1 - 2x)^{2} \left(\frac{b^{2}}{2} \frac{K_{1}(\Omega(x))}{\Omega(x)} \right) + 2i(e_{\theta}\mathbf{k})(e_{\theta}\hat{b})(1 - 2x)bK_{0}(\Omega(x)) \right] \right] \\ &+ \int_{0}^{1} dx \, e^{-i\frac{kb}{2}(2x-1)} \left[-(e_{\phi}e_{\phi})K_{0}(\Omega(x)) + (e_{\phi}\hat{b})^{2}\Omega(x)K_{1}(\Omega(x)) \right] \\ &+ \left(8p_{1}p_{2} \right) \left[\left((p_{1}e_{\theta})^{2} \frac{kp_{2}}{p_{1}} - (p_{1}e_{\theta})(p_{2}e_{\theta}) \right) e^{-ibk/2}K_{0}(\Omega_{1}) \\ &+ \left((p_{2}e_{\theta})^{2} \frac{kp_{1}}{kp_{2}} - (p_{1}e_{\theta})(p_{2}e_{\theta}) \right) e^{-ibk/2}K_{0}(\Omega_{2}) \\ &+ ((p_{1}k)(p_{2}e_{\theta}) - (p_{2}k)(p_{1}e_{\theta})) \int_{0}^{1} dx \, e^{-i\frac{kb}{2}(2x-1)} \left(-i(e_{\theta}\hat{b})bK_{0}(\Omega(x)) \\ &- (e_{\theta}\mathbf{k})(\frac{1}{2} - x) \frac{b^{2}K_{1}(\Omega(x))}{\Omega(x)} - 2(e_{\theta}\Delta_{\parallel}) \left(\frac{b^{2}}{2} \frac{K_{1}(\Omega(x))}{\Omega(x)} \right) \right) \right] \right\}, \end{split}$$

where

$$\Omega(x) = \sqrt{\Omega_1^2 x^2 + \Omega_2^2 (1-x)^2 + 2\Omega_1 \Omega_2 \sigma x (1-x)}.$$
(E.24)

Note that the vector \tilde{e}^{μ}_{θ} employed in Section 8.3.1 and the vector e^{μ}_{θ} employed here differ by a vector proportional to $k^{\mu} = \omega n^{\mu}$

$$\tilde{e}^{\mu}_{\theta} = (-\cot\theta, 0, 0, -1/\sin\theta) = e^{\mu}_{\theta} - \cot\theta n^{\mu}.$$
(E.25)

Therefore, by gauge invariance, either choice must actually yield the same projected waveforms. We have verified explicitly, as a check of the correctness of our the expressions, that the sum of Eqs. (8.87), (8.88) for the × polarization agrees with Eq. (E.21) and the sum of Eqs. (8.89), (8.90) for the + polarization agrees with Eq. (E.23), as a consequence of $k^{\mu}\tilde{A}_{0}^{\mu\nu} = 0$ (taking into account the overall factor in (8.54)).

Appendix F. From the field to the asymptotic waveform

From the eikonal operator (8.35), or from its soft versions (7.8) or (7.45), we obtain expressions for the expectation of the canonically normalized field (7.14) in the final state of the form

$$h^{\mu\nu}(x) = \int_{k} \left[e^{ik \cdot x} i \tilde{\mathcal{T}}^{\mu\nu}(k) - e^{-ik \cdot x} i \tilde{\mathcal{T}}^{\mu\nu}(k)^{*} \right]$$
(F.1)

(E.23)

with $\tilde{\mathcal{T}}^{\mu\nu}(k) = \mathcal{W}_{TT}^{\mu\nu}(k)$ or $\tilde{\mathcal{T}}^{\mu\nu}(k) = -iw_{TT}^{\mu\nu}(k)$ or $\tilde{\mathcal{T}}^{\mu\nu}(k) = -iF_{TT}^{\mu\nu}(k)$ respectively in those three cases. An important property that can be seen to hold in all three applications considered in the text is

$$\tilde{\tau}^{\mu\nu}(-k) = \tilde{\tau}^{*\mu\nu}(k).$$
(F.2)

The prediction (F.1) in only reliable for the gravitational field sourced by the collision far away from the sources, and one thus faces the issue of taking the asymptotic limit of (F.1) in a null direction. This amounts to considering a detector placed, spatially, very far away for retarded times comparable to the one at which the scattering event takes place. To this end, we introduce the decomposition

$$x^{\mu} = u t^{\mu} + r n^{\mu} , \qquad (F.3)$$

where t^{μ} is the detector's four-velocity, $t^2 = -1$, and n^{μ} is a future-directed null vector, such $n^2 = 0$ and $-t \cdot n = 1$, characterizing its direction. In this way, *u* describes the retarded time of the asymptotic detector and r > 0 represents its distance from the source, so we want to take the limit

$$r \to \infty, \quad u, \ n^{\mu} \text{ fixed.}$$
 (F.4)

It is convenient to change integration variable in (F.1) in a similar way, letting $k^{\mu} = \rho t^{\mu} + \omega m^{\mu}$ with $\omega > 0$, and such that $m^2 = 0$ and $-t \cdot m = 1$. Taking into account that the corresponding metric reads

$$(dk)^{2} = -d\rho^{2} + 2d\rho \,d\omega + \omega^{2}(dm)^{2}, \tag{F.5}$$

we find that Eq. (F.1) takes the following form

$$h^{\mu\nu}(x) = i \int_0^\infty \frac{d\omega}{2\omega} \,\omega^{D-2} \int \frac{d^{D-2}m}{(2\pi)^{D-1}} \,\tilde{\mathcal{T}}^{\mu\nu}(\omega m) \, e^{-i\omega u + i\omega r \, n \cdot m} + (\text{c.c.})\,, \tag{F.6}$$

(where c.c. stands for "complex conjugate"). The fact that $r \to \infty$ in the last term in the exponent can be compensated by letting m = n + z, where the new integration variable z must obey $t \cdot z = 0$ and $z \cdot n = -z^2/2$, and focusing on the region $z \sim O(1/\sqrt{r})$. Then, to leading order we can approximate $m \sim n$ everywhere else, obtaining

$$h^{\mu\nu}(x) \sim i \int_{0}^{\infty} \frac{d\omega}{2\omega} \,\omega^{D-2} \tilde{\mathcal{T}}^{\mu\nu}(\omega n) \, e^{-i\omega u} \int \frac{d^{D-2}z}{(2\pi)^{D-1}} \, e^{-\frac{i}{2}\omega r z^{2}} + (\text{c.c.}) \,. \tag{F.7}$$

Performing the Gaussian integral over z, we thus obtain the desired asymptotic limit,

$$h^{\mu\nu}(x) \sim \frac{i}{2(2i\pi r)^{\frac{D-2}{2}}} \int_0^\infty \frac{d\omega}{2\pi} \,\omega^{\frac{D-4}{2}} \tilde{\mathcal{T}}^{\mu\nu}(\omega n) e^{-i\omega u} + (\text{c.c.})\,.$$
(F.8)

Although the previous expression holds in arbitrary dimensions, we can focus on D = 4,

$$h^{\mu\nu}(x) \sim \frac{1}{4\pi r} \int_0^\infty \frac{d\omega}{2\pi} \,\tilde{\tau}^{\mu\nu}(\omega \, n) e^{-i\omega u} + (\text{c.c.}) \tag{F.9}$$

and finally, using (F.2), we note that the two terms combine to yield a single integral over "positive and negative frequencies",

$$h^{\mu\nu}(x) \sim \frac{1}{4\pi r} \int_{-\infty}^{\infty} \frac{d\omega}{2\pi} \,\tilde{\tau}^{\mu\nu}(\omega n) e^{-iu\omega} \,. \tag{F.10}$$

References

- [1] S. Weinberg, Infrared photons and gravitons, Phys. Rev. 140 (1965) B516-B524, http://dx.doi.org/10.1103/PhysRev.140.B516.
- [2] J. Scherk, J.H. Schwarz, Dual models for nonhadrons, Nuclear Phys. B 81 (1974) 118-144, http://dx.doi.org/10.1016/0550-3213(74)90010-8.
- [3] J. Scherk, J.H. Schwarz, Dual models and the geometry of space-time, Phys. Lett. B 52 (1974) 347–350, http://dx.doi.org/10.1016/0370-2693(74)90059-8.
- [4] M.B. Green, J.H. Schwarz, Anomaly cancellation in supersymmetric D=10 gauge theory and superstring theory, Phys. Lett. B 149 (1984) 117–122, http://dx.doi.org/10.1016/0370-2693(84)91565-X.
- M. Soldate, Partial wave unitarity and closed string amplitudes, Phys. Lett. B186 (1987) 321-327, http://dx.doi.org/10.1016/0370-2693(87) 90302-9.
- [6] G. 't Hooft, Graviton dominance in ultrahigh-energy scattering, Phys. Lett. B198 (1987) 61-63, http://dx.doi.org/10.1016/0370-2693(87)90159-6.
- [7] I.J. Muzinich, M. Soldate, High-energy unitarity of gravitation and strings, Phys. Rev. D37 (1988) 359, http://dx.doi.org/10.1103/PhysRevD.37.359.

 [8] D. Amati, M. Ciafaloni, G. Veneziano, Superstring collisions at Planckian energies, Phys. Lett. B197 (1987) 81, http://dx.doi.org/10.1016/0370-2693(87)90346-7.

- [9] D.J. Gross, P.F. Mende, The high-energy behavior of string scattering amplitudes, Phys. Lett. B 197 (1987) 129–134, http://dx.doi.org/10.1016/ 0370-2693(87)90355-8.
- [10] D. Amati, M. Ciafaloni, G. Veneziano, Classical and quantum gravity effects from Planckian energy superstring collisions, Internat. J. Modern Phys. A3 (1988) 1615–1661, http://dx.doi.org/10.1142/S0217751X88000710.
- [11] D.J. Gross, P.F. Mende, String theory beyond the Planck scale, Nuclear Phys. B 303 (1988) 407-454, http://dx.doi.org/10.1016/0550-3213(88) 90390-2.

- [12] B. Sundborg, High-energy asymptotics: the one loop string amplitude and resummation, Nuclear Phys. B306 (1988) 545–566, http://dx.doi. org/10.1016/0550-3213(88)90014-4.
- [13] D. Amati, M. Ciafaloni, G. Veneziano, Can space-time be probed below the string size? Phys. Lett. B216 (1989) 41, http://dx.doi.org/10.1016/ 0370-2693(89)91366-X.
- [14] P.F. Mende, H. Ooguri, Borel summation of string theory for Planck scale scattering, Nuclear Phys. B339 (1990) 641–662, http://dx.doi.org/10. 1016/0550-3213(90)90202-0.
- [15] D. Amati, M. Ciafaloni, G. Veneziano, Higher order gravitational deflection and soft bremsstrahlung in Planckian energy superstring collisions, Nuclear Phys. B347 (1990) 550–580, http://dx.doi.org/10.1016/0550-3213(90)90375-N.
- [16] D. Amati, M. Ciafaloni, G. Veneziano, Planckian scattering beyond the semiclassical approximation, Phys. Lett. B289 (1992) 87–91, http: //dx.doi.org/10.1016/0370-2693(92)91366-H.
- [17] D. Amati, M. Ciafaloni, G. Veneziano, Effective action and all order gravitational eikonal at Planckian energies, Nuclear Phys. B403 (1993) 707-724, http://dx.doi.org/10.1016/0550-3213(93)90367-X.
- [18] S.W. Hawking, Particle creation by black holes, Comm. Math. Phys. 43 (1975) 199–220, http://dx.doi.org/10.1007/BF02345020; Commun. Math. Phys. 46 (1976) 206;
- [19] S.W. Hawking, Breakdown of predictability in gravitational collapse, Phys. Rev. D 14 (1976) 2460–2473, http://dx.doi.org/10.1103/PhysRevD. 14.2460.
- [20] J.J. Sakurai, J. Napolitano, Modern Quantum Mechanics, in: Quantum Physics, Quantum Information and Quantum Computation, Cambridge University Press, 2020, http://dx.doi.org/10.1017/9781108587280.
- [21] S. Weinberg, Lectures on Quantum Mechanics, second ed., Cambridge University Press, 2015.
- [22] G. D'Appollonio, P. Di Vecchia, R. Russo, G. Veneziano, High-energy string-brane scattering: Leading eikonal and beyond, J. High Energy Phys. 1011 (2010) 100, http://dx.doi.org/10.1007/JHEP11(2010)100, arXiv:1008.4773.
- [23] E.J. Martinec, N.P. Warner, The harder they fall, the bigger they become: Tidal trapping of strings by microstate geometries, J. High Energy Phys. 04 (2021) 259, http://dx.doi.org/10.1007/JHEP04(2021)259, arXiv:2009.07847.
- [24] N. Ceplak, S. Hampton, Y. Li, Toroidal tidal effects in microstate geometries, J. High Energy Phys. 03 (2022) 021, http://dx.doi.org/10.1007/ JHEP03(2022)021, arXiv:2106.03841.
- [25] B.P. Abbott, et al., Virgo, LIGO Scientific Collaboration, Observation of gravitational waves from a binary black hole merger, Phys. Rev. Lett. 116 (6) (2016) 061102, http://dx.doi.org/10.1103/PhysRevLett.116.061102, arXiv:1602.03837.
- [26] B.P. Abbott, et al., Virgo, LIGO Scientific Collaboration, GW151226: Observation of gravitational waves from a 22-solar-mass binary black hole coalescence, Phys. Rev. Lett. 116 (24) (2016) 241103, http://dx.doi.org/10.1103/PhysRevLett.116.241103, arXiv:1606.04855.
- [27] F. Pretorius, Evolution of binary black hole spacetimes, Phys. Rev. Lett. 95 (2005) 121101, http://dx.doi.org/10.1103/PhysRevLett.95.121101, arXiv:gr-qc/0507014.
- [28] E. Gourgoulhon, 3+1 formalism and bases of numerical relativity, 2007, arXiv:gr-qc/0703035.
- [29] L. Barack, A. Pound, Self-force and radiation reaction in general relativity, Rep. Progr. Phys. 82 (1) (2019) 016904, http://dx.doi.org/10.1088/ 1361-6633/aae552, arXiv:1805.10385.
- [30] L. Blanchet, Gravitational radiation from post-Newtonian sources and inspiralling compact binaries, Living Rev. Rel. 17 (2014) 2, http: //dx.doi.org/10.12942/lrr-2014-2, arXiv:1310.1528.
- [31] A. Buonanno, T. Damour, Effective one-body approach to general relativistic two-body dynamics, Phys. Rev. D59 (1999) 084006, http: //dx.doi.org/10.1103/PhysRevD.59.084006, arXiv:gr-qc/9811091.
- [32] A. Buonanno, T. Damour, Transition from inspiral to plunge in binary black hole coalescences, Phys. Rev. D62 (2000) 064015, http: //dx.doi.org/10.1103/PhysRevD.62.064015, arXiv:gr-qc/0001013.
- [33] D. Neill, I.Z. Rothstein, Classical space-times from the s matrix, Nuclear Phys. B877 (2013) 177–189, http://dx.doi.org/10.1016/j.nuclphysb.2013. 09.007, arXiv:1304.7263.
- [34] T. Damour, Gravitational scattering, post-Minkowskian approximation and effective one-body theory, Phys. Rev. D94 (10) (2016) 104015, http://dx.doi.org/10.1103/PhysRevD.94.104015, arXiv:1609.00354.
- [35] T. Damour, High-energy gravitational scattering and the general relativistic two-body problem, Phys. Rev. D97 (4) (2018) 044038, http: //dx.doi.org/10.1103/PhysRevD.97.044038, arXiv:1710.10599.
- [36] N.E.J. Bjerrum-Bohr, P.H. Damgaard, G. Festuccia, L. Planté, P. Vanhove, General relativity from scattering amplitudes, Phys. Rev. Lett. 121 (17) (2018) 171601, http://dx.doi.org/10.1103/PhysRevLett.121.171601, arXiv:1806.04920.
- [37] C. Cheung, I.Z. Rothstein, M.P. Solon, From scattering amplitudes to classical potentials in the post-Minkowskian expansion, Phys. Rev. Lett. 121 (25) (2018) 251101, http://dx.doi.org/10.1103/PhysRevLett.121.251101, arXiv:1808.02489.
- [38] Z. Bern, C. Cheung, R. Roiban, C.-H. Shen, M.P. Solon, M. Zeng, Scattering amplitudes and the conservative Hamiltonian for binary systems at third post-Minkowskian order, Phys. Rev. Lett. 122 (20) (2019) 201603, http://dx.doi.org/10.1103/PhysRevLett.122.201603, arXiv:1901.04424.
- [39] Z. Bern, C. Cheung, R. Roiban, C.-H. Shen, M.P. Solon, M. Zeng, Black hole binary dynamics from the double copy and effective theory, J. High Energy Phys. 10 (2019) 206, http://dx.doi.org/10.1007/JHEP10(2019)206, arXiv:1908.01493.
- [40] Z. Bern, J. Parra-Martinez, R. Roiban, M.S. Ruf, C.-H. Shen, M.P. Solon, M. Zeng, Scattering amplitudes and conservative binary dynamics at $\mathcal{O}(G^4)$, Phys. Rev. Lett. 126 (17) (2021) 171601, http://dx.doi.org/10.1103/PhysRevLett.126.171601, arXiv:2101.07254.
- [41] Z. Bern, J. Parra-Martinez, R. Roiban, M.S. Ruf, C.-H. Shen, M.P. Solon, M. Zeng, Scattering amplitudes, the tail effect, and conservative binary dynamics at o(G⁴), Phys. Rev. Lett. 128 (16) (2022) 161103, http://dx.doi.org/10.1103/PhysRevLett.128.161103, arXiv:2112.10750.
- [42] N.E.J. Bjerrum-Bohr, P.H. Damgaard, L. Plante, P. Vanhove, The SAGEX review on scattering amplitudes chapter 13: Post-Minkowskian expansion from scattering amplitudes, J. Phys. A 55 (44) (2022) 443014, http://dx.doi.org/10.1088/1751-8121/ac7a78, arXiv:2203.13024.
- [43] A.K. Collado, P. Di Vecchia, R. Russo, S. Thomas, The subleading eikonal in supergravity theories, J. High Energy Phys. 10 (2018) 038, http://dx.doi.org/10.1007/JHEP10(2018)038, arXiv:1807.04588.
- [44] P. Di Vecchia, A. Luna, S.G. Naculich, R. Russo, G. Veneziano, C.D. White, A tale of two exponentiations in $\mathcal{N} = 8$ supergravity, Phys. Lett. B798 (2019) 134927, http://dx.doi.org/10.1016/j.physletb.2019.134927, arXiv:1908.05603.
- [45] P. Di Vecchia, S.G. Naculich, R. Russo, G. Veneziano, C.D. White, A tale of two exponentiations in N=8 supergravity at subleading level, J. High Energy Phys. 03 (2020) 173, http://dx.doi.org/10.1007/JHEP03(2020)173, arXiv:1911.11716.
- [46] Z. Bern, H. Ita, J. Parra-Martinez, M.S. Ruf, Universality in the classical limit of massless gravitational scattering, Phys. Rev. Lett. 125 (3) (2020) 031601, http://dx.doi.org/10.1103/PhysRevLett.125.031601, arXiv:2002.02459.
- [47] P. Di Vecchia, C. Heissenberg, R. Russo, G. Veneziano, The eikonal approach to gravitational scattering and radiation at O(G³), J. High Energy Phys. 07 (2021) 169, http://dx.doi.org/10.1007/JHEP07(2021)169, arXiv:2104.03256.
- [48] A. Koemans Collado, P. Di Vecchia, R. Russo, Revisiting the second post-Minkowskian eikonal and the dynamics of binary black holes, Phys. Rev. D 100 (6) (2019) 066028, http://dx.doi.org/10.1103/PhysRevD.100.066028, arXiv:1904.02667.
- [49] A. Cristofoli, P.H. Damgaard, P. Di Vecchia, C. Heissenberg, Second-order post-Minkowskian scattering in arbitrary dimensions, J. High Energy Phys. 07 (2020) 122, http://dx.doi.org/10.1007/JHEP07(2020)122, arXiv:2003.10274.

- [50] C. Cheung, M.P. Solon, Tidal effects in the post-Minkowskian expansion, Phys. Rev. Lett. 125 (19) (2020) 191601, http://dx.doi.org/10.1103/ PhysRevLett.125.191601, arXiv:2006.06665.
- [51] Z. Bern, J. Parra-Martinez, R. Roiban, E. Sawyer, C.-H. Shen, Leading nonlinear tidal effects and scattering amplitudes, J. High Energy Phys. 05 (2021) 188, http://dx.doi.org/10.1007/JHEP05(2021)188, arXiv:2010.08559.
- [52] C. Cheung, N. Shah, M.P. Solon, Mining the geodesic equation for scattering data, Phys. Rev. D 103 (2) (2021) 024030, http://dx.doi.org/10. 1103/PhysRevD.103.024030, arXiv:2010.08568.
- [53] R. Aoude, K. Haddad, A. Helset, Tidal effects for spinning particles, J. High Energy Phys. 03 (2021) 097, http://dx.doi.org/10.1007/JHEP03(2021) 097, arXiv:2012.05256.
- [54] M. Accettulli Huber, A. Brandhuber, S. De Angelis, G. Travaglini, From amplitudes to gravitational radiation with cubic interactions and tidal effects, Phys. Rev. D 103 (4) (2021) 045015, http://dx.doi.org/10.1103/PhysRevD.103.045015, arXiv:2012.06548.
- [55] S. Mougiakakos, M.M. Riva, F. Vernizzi, Gravitational bremsstrahlung with tidal effects in the post-Minkowskian expansion, Phys. Rev. Lett. 129 (12) (2022) 121101, http://dx.doi.org/10.1103/PhysRevLett.129.121101, arXiv:2204.06556.
- [56] G.U. Jakobsen, G. Mogull, J. Plefka, B. Sauer, All things retarded: radiation-reaction in worldline quantum field theory, J. High Energy Phys. 10 (2022) 128, http://dx.doi.org/10.1007/JHEP10(2022)128, arXiv:2207.00569.

[57] C. Heissenberg, Angular momentum loss due to tidal effects in the post-Minkowskian expansion, 2022, arXiv:2210.15689.

- [58] N. Arkani-Hamed, T.-C. Huang, Y.-t. Huang, Scattering amplitudes for all masses and spins, J. High Energy Phys. 11 (2021) 070, http: //dx.doi.org/10.1007/JHEP11(2021)070, arXiv:1709.04891.
- [59] J. Vines, Scattering of two spinning black holes in post-Minkowskian gravity, to all orders in spin, and effective-one-body mappings, Classical Quantum Gravity 35 (8) (2018) 084002, http://dx.doi.org/10.1088/1361-6382/aaa3a8, arXiv:1709.06016.

[60] A. Guevara, A. Ochirov, J. Vines, Scattering of spinning black holes from exponentiated soft factors, 2018, arXiv:1812.06895.

- [61] M.-Z. Chung, Y.-T. Huang, J.-W. Kim, S. Lee, The simplest massive S-matrix: from minimal coupling to black holes, 2018, arXiv:1812.08752.
- [62] B. Maybee, D. O'Connell, J. Vines, Observables and amplitudes for spinning particles and black holes, J. High Energy Phys. 12 (2019) 156, http://dx.doi.org/10.1007/JHEP12(2019)156, arXiv:1906.09260.
- [63] A. Guevara, A. Ochirov, J. Vines, Black-hole scattering with general spin directions from minimal-coupling amplitudes, Phys. Rev. D 100 (10) (2019) 104024, http://dx.doi.org/10.1103/PhysRevD.100.104024, arXiv:1906.10071.
- [64] N. Arkani-Hamed, Y.-t. Huang, D. O'Connell, Kerr black holes as elementary particles, J. High Energy Phys. 01 (2020) 046, http://dx.doi.org/10. 1007/JHEP01(2020)046, arXiv:1906.10100.
- [65] H. Johansson, A. Ochirov, Double copy for massive quantum particles with spin, J. High Energy Phys. 09 (2019) 040, http://dx.doi.org/10.1007/ JHEP09(2019)040, arXiv:1906.12292.
- [66] M.-Z. Chung, Y.-T. Huang, J.-W. Kim, Classical potential for general spinning bodies, J. High Energy Phys. 09 (2020) 074, http://dx.doi.org/10. 1007/JHEP09(2020)074, arXiv:1908.08463.
- [67] P.H. Damgaard, K. Haddad, A. Helset, Heavy black hole effective theory, J. High Energy Phys. 11 (2019) 070, http://dx.doi.org/10.1007/ JHEP11(2019)070, arXiv:1908.10308.
- [68] Y.F. Bautista, A. Guevara, On the double copy for spinning matter, J. High Energy Phys. 11 (2021) 184, http://dx.doi.org/10.1007/JHEP11(2021) 184, arXiv:1908.11349.
- [69] R. Aoude, K. Haddad, A. Helset, On-shell heavy particle effective theories, J. High Energy Phys. 05 (2020) 051, http://dx.doi.org/10.1007/ JHEP05(2020)051, arXiv:2001.09164.
- [70] M.-Z. Chung, Y.-t. Huang, J.-W. Kim, S. Lee, Complete Hamiltonian for spinning binary systems at first post-Minkowskian order, J. High Energy Phys. 05 (2020) 105, http://dx.doi.org/10.1007/JHEP05(2020)105, arXiv:2003.06600.
- [71] Z. Bern, A. Luna, R. Roiban, C.-H. Shen, M. Zeng, Spinning black hole binary dynamics, scattering amplitudes, and effective field theory, Phys. Rev. D 104 (6) (2021) 065014, http://dx.doi.org/10.1103/PhysRevD.104.065014, arXiv:2005.03071.
- [72] A. Guevara, B. Maybee, A. Ochirov, D. O'Connell, J. Vines, A worldsheet for Kerr, J. High Energy Phys. 03 (2021) 201, http://dx.doi.org/10.1007/ JHEP03(2021)201, arXiv:2012.11570.
- [73] D. Kosmopoulos, A. Luna, Quadratic-in-spin Hamiltonian at O(G²) from scattering amplitudes, J. High Energy Phys. 07 (2021) 037, http: //dx.doi.org/10.1007/JHEP07(2021)037, arXiv:2102.10137.
- [74] R. Aoude, A. Ochirov, Classical observables from coherent-spin amplitudes, J. High Energy Phys. 10 (2021) 008, http://dx.doi.org/10.1007/ JHEP10(2021)008, arXiv:2108.01649.
- [75] Y.F. Bautista, A. Guevara, C. Kavanagh, J. Vines, Scattering in black hole backgrounds and higher-spin amplitudes: Part I, 2021, arXiv:2107.10179. [76] M. Chiodaroli, H. Johansson, P. Pichini, Compton black-hole scattering for s \leq 5/2, J. High Energy Phys. 02 (2022) 156, http://dx.doi.org/10. 1007/[HEP02(2022)156, arXiv:2107.14779.
- [77] K. Haddad, Exponentiation of the leading eikonal with spin, 2021, arXiv:2109.04427.
- [78] W.-M. Chen, M.-Z. Chung, Y.-t. Huang, J.-W. Kim, The 2PM Hamiltonian for binary Kerr to quartic in spin, J. High Energy Phys. 08 (2022) 148, http://dx.doi.org/10.1007/JHEP08(2022)148, arXiv:2111.13639.
- [79] R. Aoude, K. Haddad, A. Helset, Searching for Kerr in the 2PM amplitude, J. High Energy Phys. 07 (2022) 072, http://dx.doi.org/10.1007/ JHEP07(2022)072, arXiv:2203.06197.
- [80] Z. Bern, D. Kosmopoulos, A. Luna, R. Roiban, F. Teng, Binary dynamics through the fifth power of spin at $\mathcal{O}(G^2)$, 2022, arXiv:2203.06202.
- [81] F. Alessio, P. Di Vecchia, Radiation reaction for spinning black-hole scattering, 2022, arXiv:2203.13272.
- [82] F. Febres Cordero, M. Kraus, G. Lin, M.S. Ruf, M. Zeng, Conservative binary dynamics with a spinning black hole at O(G3) from scattering amplitudes, Phys. Rev. Lett. 130 (2) (2023) 021601, http://dx.doi.org/10.1103/PhysRevLett.130.021601, arXiv:2205.07357.
- [83] Y.F. Bautista, A. Guevara, C. Kavanagh, J. Vines, Scattering in black hole backgrounds and higher-spin amplitudes. Part II, J. High Energy Phys. 05 (2023) 211, http://dx.doi.org/10.1007/JHEP05(2023)211, arXiv:2212.07965.
- [84] N.E.J. Bjerrum-Bohr, G. Chen, M. Skowronek, Classical spin gravitational compton scattering, 2023, arXiv:2302.00498.
- [85] T. Damour, M. Soffel, C.-m. Xu, General relativistic celestial mechanics. 3. Rotational equations of motion, Phys. Rev. D 47 (1993) 3124–3135, http://dx.doi.org/10.1103/PhysRevD.47.3124.
- [86] T. Damour, M. Soffel, C.-m. Xu, General relativistic celestial mechanics. 4: Theory of satellite motion, Phys. Rev. D 49 (1994) 618–635, http://dx.doi.org/10.1103/PhysRevD.49.618.
- [87] W.D. Goldberger, I.Z. Rothstein, Dissipative effects in the worldline approach to black hole dynamics, Phys. Rev. D 73 (2006) 104030, http://dx.doi.org/10.1103/PhysRevD.73.104030, arXiv:hep-th/0511133.
- [88] T. Hinderer, Tidal Love numbers of neutron stars, Astrophys. J. 677 (2008) 1216-1220, http://dx.doi.org/10.1086/533487, arXiv:0711.2420.
- [89] E.E. Flanagan, T. Hinderer, Constraining neutron star tidal Love numbers with gravitational wave detectors, Phys. Rev. D 77 (2008) 021502, http://dx.doi.org/10.1103/PhysRevD.77.021502, arXiv:0709.1915.
- [90] T. Damour, A. Nagar, Relativistic tidal properties of neutron stars, Phys. Rev. D 80 (2009) 084035, http://dx.doi.org/10.1103/PhysRevD.80.084035, arXiv:0906.0096.
- [91] T. Binnington, E. Poisson, Relativistic theory of tidal Love numbers, Phys. Rev. D 80 (2009) 084018, http://dx.doi.org/10.1103/PhysRevD.80. 084018, arXiv:0906.1366.

- [92] T. Hinderer, B.D. Lackey, R.N. Lang, J.S. Read, Tidal deformability of neutron stars with realistic equations of state and their gravitational wave signatures in binary inspiral, Phys. Rev. D 81 (2010) 123016, http://dx.doi.org/10.1103/PhysRevD.81.123016, arXiv:0911.3535.
- [93] B. Kol, M. Smolkin, Black hole stereotyping: Induced gravito-static polarization, J. High Energy Phys. 02 (2012) 010, http://dx.doi.org/10.1007/ JHEP02(2012)010, arXiv:1110.3764.
- [94] T. Damour, A. Nagar, L. Villain, Measurability of the tidal polarizability of neutron stars in late-inspiral gravitational-wave signals, Phys. Rev. D 85 (2012) 123007, http://dx.doi.org/10.1103/PhysRevD.85.123007, arXiv:1203.4352.
- [95] M. Favata, Systematic parameter errors in inspiraling neutron star binaries, Phys. Rev. Lett. 112 (2014) 101101, http://dx.doi.org/10.1103/ PhysRevLett.112.101101, arXiv:1310.8288.
- [96] L. Baiotti, L. Rezzolla, Binary neutron star mergers: a review of Einstein's richest laboratory, Rep. Progr. Phys. 80 (9) (2017) 096901, http://dx.doi.org/10.1088/1361-6633/aa67bb, arXiv:1607.03540.
- [97] L. Barack, et al., Black holes, gravitational waves and fundamental physics: a roadmap, Classical Quantum Gravity 36 (14) (2019) 143001, http://dx.doi.org/10.1088/1361-6382/ab0587, arXiv:1806.05195.
- [98] A. Buonanno, B.S. Sathyaprakash, Sources of gravitational waves: Theory and observations, 2014, arXiv:1410.7832.
- [99] V. Cardoso, P. Pani, Testing the nature of dark compact objects: a status report, Living Rev. Rel. 22 (1) (2019) 4, http://dx.doi.org/10.1007/ s41114-019-0020-4, arXiv:1904.05363.
- [100] D. Baumann, H.S. Chia, R.A. Porto, J. Stout, Gravitational collider physics, Phys. Rev. D 101 (8) (2020) 083019, http://dx.doi.org/10.1103/ PhysRevD.101.083019, arXiv:1912.04932.
- [101] D.N. Kabat, M. Ortiz, Eikonal quantum gravity and Planckian scattering, Nuclear Phys. B388 (1992) 570-592, http://dx.doi.org/10.1016/0550-3213(92)90627-N, arXiv:hep-th/9203082.
- [102] R. Akhoury, R. Saotome, G. Sterman, High energy scattering in perturbative quantum gravity at next to leading power, 2013, arXiv:1308.5204.
- [103] C. Cheung, M.P. Solon, Classical gravitational scattering at $\mathcal{O}(G^3)$ from feynman diagrams, J. High Energy Phys. 06 (2020) 144, http: //dx.doi.org/10.1007/JHEP06(2020)144, arXiv:2003.08351.
- [104] P. Di Vecchia, C. Heissenberg, R. Russo, G. Veneziano, Universality of ultra-relativistic gravitational scattering, Phys. Lett. B 811 (2020) 135924, http://dx.doi.org/10.1016/j.physletb.2020.135924, arXiv:2008.12743.
- [105] M. Accettulli Huber, A. Brandhuber, S. De Angelis, G. Travaglini, Eikonal phase matrix, deflection angle and time delay in effective field theories of gravity, Phys. Rev. D 102 (4) (2020) 046014, http://dx.doi.org/10.1103/PhysRevD.102.046014, arXiv:2006.02375.
- [106] P. Di Vecchia, C. Heissenberg, R. Russo, G. Veneziano, Radiation reaction from soft theorems, Phys. Lett. B 818 (2021) 136379, http: //dx.doi.org/10.1016/j.physletb.2021.136379, arXiv:2101.05772.
- [107] C. Heissenberg, Infrared divergences and the eikonal exponentiation, Phys. Rev. D 104 (4) (2021) 046016, http://dx.doi.org/10.1103/PhysRevD. 104.046016, arXiv:2105.04594.
- [108] N.E.J. Bjerrum-Bohr, P.H. Damgaard, L. Planté, P. Vanhove, Classical gravity from loop amplitudes, Phys. Rev. D 104 (2) (2021) 026009, http://dx.doi.org/10.1103/PhysRevD.104.026009, arXiv:2104.04510.
- [109] N.E.J. Bjerrum-Bohr, P.H. Damgaard, L. Planté, P. Vanhove, The amplitude for classical gravitational scattering at third post-Minkowskian order, J. High Energy Phys. 08 (2021) 172, http://dx.doi.org/10.1007/JHEP08(2021)172, arXiv:2105.05218.
- [110] P.H. Damgaard, L. Plante, P. Vanhove, On an exponential representation of the gravitational S-matrix, J. High Energy Phys. 11 (2021) 213, http://dx.doi.org/10.1007/IHEP11(2021)213, arXiv:2107.12891.
- [111] A. Brandhuber, G. Chen, G. Travaglini, C. Wen, Classical gravitational scattering from a gauge-invariant double copy, J. High Energy Phys. 10 (2021) 118, http://dx.doi.org/10.1007/[HEP10(2021)118, arXiv:2108.04216.
- [112] P. Di Vecchia, C. Heissenberg, R. Russo, G. Veneziano, The eikonal operator at arbitrary velocities I: the soft-radiation limit, J. High Energy Phys. 07 (2022) 039, http://dx.doi.org/10.1007/JHEP07(2022)039, arXiv:2204.02378.
- [113] W.D. Goldberger, I.Z. Rothstein, An effective field theory of gravity for extended objects, Phys. Rev. D 73 (2006) 104029, http://dx.doi.org/10. 1103/PhysRevD.73.104029, arXiv:hep-th/0409156.
- [114] R.A. Porto, The effective field theorist's approach to gravitational dynamics, Phys. Rep. 633 (2016) 1–104, http://dx.doi.org/10.1016/j.physrep. 2016.04.003, arXiv:1601.04914.
- [115] M. Levi, Effective field theories of post-Newtonian gravity: A comprehensive review, Rep. Progr. Phys. 83 (7) (2020) 075901, http://dx.doi.org/ 10.1088/1361-6633/ab12bc, arXiv:1807.01699.
- [116] D.A. Kosower, B. Maybee, D. O'Connell, Amplitudes, observables, and classical scattering, J. High Energy Phys. 02 (2019) 137, http://dx.doi.org/ 10.1007/JHEP02(2019)137, arXiv:1811.10950.
- [117] E. Herrmann, J. Parra-Martinez, M.S. Ruf, M. Zeng, Gravitational bremsstrahlung from reverse unitarity, Phys. Rev. Lett. 126 (20) (2021) 201602, http://dx.doi.org/10.1103/PhysRevLett.126.201602, arXiv:2101.07255.
- [118] E. Herrmann, J. Parra-Martinez, M.S. Ruf, M. Zeng, Radiative classical gravitational observables at O(G³) from scattering amplitudes, J. High Energy Phys. 10 (2021) 148, http://dx.doi.org/10.1007/[HEP10(2021)148, arXiv:2104.03957.
- [119] A. Cristofoli, R. Gonzo, D.A. Kosower, D. O'Connell, Waveforms from amplitudes, Phys. Rev. D 106 (5) (2022) 056007, http://dx.doi.org/10. 1103/PhysRevD.106.056007, arXiv:2107.10193.
- [120] A. Cristofoli, R. Gonzo, N. Moynihan, D. O'Connell, A. Ross, M. Sergola, C.D. White, The uncertainty principle and classical amplitudes, 2021, arXiv:2112.07556.
- [121] T. Adamo, A. Cristofoli, A. Ilderton, Classical physics from amplitudes on curved backgrounds, J. High Energy Phys. 08 (2022) 281, http: //dx.doi.org/10.1007/JHEP08(2022)281, arXiv:2203.13785.
- [122] T. Adamo, A. Cristofoli, A. Ilderton, S. Klisch, All-order waveforms from amplitudes, 2022, arXiv:2210.04696.
- [123] W.D. Goldberger, A.K. Ridgway, Radiation and the classical double copy for color charges, Phys. Rev. D 95 (12) (2017) 125010, http: //dx.doi.org/10.1103/PhysRevD.95.125010, arXiv:1611.03493.
- [124] G. Kälin, R.A. Porto, Post-Minkowskian effective field theory for conservative binary dynamics, J. High Energy Phys. 11 (2020) 106, http: //dx.doi.org/10.1007/JHEP11(2020)106, arXiv:2006.01184.
- [125] G. Kälin, Z. Liu, R.A. Porto, Conservative dynamics of binary systems to third post-Minkowskian order from the effective field theory approach, 2020, arXiv:2007.04977.
- [126] G. Kälin, Z. Liu, R.A. Porto, Conservative tidal effects in compact binary systems to next-to-leading post-Minkowskian order, Phys. Rev. D 102 (2020) 124025, http://dx.doi.org/10.1103/PhysRevD.102.124025, arXiv:2008.06047.
- [127] G. Mogull, J. Plefka, J. Steinhoff, Classical black hole scattering from a worldline quantum field theory, J. High Energy Phys. 02 (2021) 048, http://dx.doi.org/10.1007/JHEP02(2021)048, arXiv:2010.02865.
- [128] G.U. Jakobsen, G. Mogull, J. Plefka, J. Steinhoff, Classical gravitational bremsstrahlung from a worldline quantum field theory, Phys. Rev. Lett. 126 (20) (2021) 201103, http://dx.doi.org/10.1103/PhysRevLett.126.201103, arXiv:2101.12688.
- [129] S. Mougiakakos, M.M. Riva, F. Vernizzi, Gravitational bremsstrahlung in the post-Minkowskian effective field theory, Phys. Rev. D 104 (2) (2021) 024041, http://dx.doi.org/10.1103/PhysRevD.104.024041, arXiv:2102.08339.
- [130] Z. Liu, R.A. Porto, Z. Yang, Spin effects in the effective field theory approach to post-Minkowskian conservative dynamics, J. High Energy Phys. 06 (2021) 012, http://dx.doi.org/10.1007/JHEP06(2021)012, arXiv:2102.10059.

- [131] C. Dlapa, G. Kälin, Z. Liu, R.A. Porto, Dynamics of binary systems to fourth post-Minkowskian order from the effective field theory approach, 2021, arXiv:2106.08276.
- [132] G.U. Jakobsen, G. Mogull, J. Plefka, J. Steinhoff, Gravitational bremsstrahlung and hidden supersymmetry of spinning bodies, Phys. Rev. Lett. 128 (1) (2022) 011101, http://dx.doi.org/10.1103/PhysRevLett.128.011101, arXiv:2106.10256.
- [133] G.U. Jakobsen, G. Mogull, J. Plefka, J. Steinhoff, SUSY in the sky with gravitons, J. High Energy Phys. 01 (2022) 027, http://dx.doi.org/10.1007/ [HEP01(2022)027, arXiv:2109.04465.
- [134] M.M. Riva, F. Vernizzi, Radiated momentum in the post-Minkowskian worldline approach via reverse unitarity, J. High Energy Phys. 11 (2021) 228, http://dx.doi.org/10.1007/JHEP11(2021)228, arXiv:2110.10140.
- [135] C. Dlapa, G. Kälin, Z. Liu, R.A. Porto, Conservative dynamics of binary systems at fourth post-Minkowskian order in the large-eccentricity expansion, Phys. Rev. Lett. 128 (16) (2022) 161104, http://dx.doi.org/10.1103/PhysRevLett.128.161104, arXiv:2112.11296.
- [136] G.U. Jakobsen, G. Mogull, Conservative and radiative dynamics of spinning bodies at third post-Minkowskian order using worldline quantum field theory, Phys. Rev. Lett. 128 (14) (2022) 141102, http://dx.doi.org/10.1103/PhysRevLett.128.141102, arXiv:2201.07778.
- [137] G. Kälin, J. Neef, R.A. Porto, Radiation-reaction in the effective field theory approach to post-Minkowskian dynamics, J. High Energy Phys. 01 (2023) 140, http://dx.doi.org/10.1007/JHEP01(2023)140, arXiv:2207.00580.
- [138] C. Dlapa, G. Kälin, Z. Liu, J. Neef, R.A. Porto, Radiation reaction and gravitational waves at fourth post-Minkowskian order, Phys. Rev. Lett. 130 (10) (2023) 101401, http://dx.doi.org/10.1103/PhysRevLett.130.101401, arXiv:2210.05541.
- [139] G.U. Jakobsen, G. Mogull, Linear response, Hamiltonian, and radiative spinning two-body dynamics, Phys. Rev. D 107 (4) (2023) 044033, http://dx.doi.org/10.1103/PhysRevD.107.044033, arXiv:2210.06451.
- [140] C. Dlapa, G. Kälin, Z. Liu, R.A. Porto, Bootstrapping the relativistic two-body problem, 2023, arXiv:2304.01275.
- [141] P.H. Damgaard, E.R. Hansen, L. Planté, P. Vanhove, The relation between KMOC and worldline formalisms for classical gravity, 2023, arXiv:2306.11454.
- [142] M. Ciafaloni, D. Colferai, G. Veneziano, Infrared features of gravitational scattering and radiation in the eikonal approach, Phys. Rev. D99 (6) (2019) 066008, http://dx.doi.org/10.1103/PhysRevD.99.066008, arXiv:1812.08137.
- [143] A. Addazi, M. Bianchi, G. Veneziano, Soft gravitational radiation from ultra-relativistic collisions at sub- and sub-sub-leading order, J. High Energy Phys. 05 (2019) 050, http://dx.doi.org/10.1007/JHEP05(2019)050, arXiv:1901.10986.
- [144] P. Di Vecchia, C. Heissenberg, R. Russo, Angular momentum of zero-frequency gravitons, 2022, arXiv:2203.11915.
- [145] P. Di Vecchia, C. Heissenberg, R. Russo, G. Veneziano, Classical gravitational observables from the eikonal operator, 2022, arXiv:2210.12118.
- [146] A. Brandhuber, G.R. Brown, G. Chen, S. De Angelis, J. Gowdy, G. Travaglini, One-loop gravitational bremsstrahlung and waveforms from a heavy-mass effective field theory, 2023, arXiv:2303.06111.
- [147] A. Herderschee, R. Roiban, F. Teng, The sub-leading scattering waveform from amplitudes, 2023, arXiv:2303.06112.
- [148] A. Georgoudis, C. Heissenberg, I. Vazquez-Holm, Inelastic exponentiation and classical gravitational scattering at one loop, 2023, arXiv: 2303.07006.
- [149] R.C. Brower, J. Polchinski, M.J. Strassler, C.-I. Tan, The Pomeron and gauge/string duality, J. High Energy Phys. 0712 (2007) 005, http: //dx.doi.org/10.1088/1126-6708/2007/12/005, arXiv:hep-th/0603115.
- [150] L. Cornalba, M.S. Costa, J. Penedones, R. Schiappa, Eikonal approximation in AdS/CFT: From shock waves to four-point functions, J. High Energy Phys. 08 (2007) 019, http://dx.doi.org/10.1088/1126-6708/2007/08/019, arXiv:hep-th/0611122.
- [151] L. Cornalba, M.S. Costa, J. Penedones, R. Schiappa, Eikonal approximation in AdS/CFT: Conformal partial waves and finite N four-point functions, Nuclear Phys. B 767 (2007) 327–351, http://dx.doi.org/10.1016/j.nuclphysb.2007.01.007, arXiv:hep-th/0611123.
- [152] L. Cornalba, M.S. Costa, J. Penedones, Eikonal approximation in AdS/CFT: Resumming the gravitational loop expansion, J. High Energy Phys. 09 (2007) 037, http://dx.doi.org/10.1088/1126-6708/2007/09/037, arXiv:0707.0120.
- [153] L. Cornalba, M.S. Costa, J. Penedones, Eikonal methods in AdS/CFT: BFKL Pomeron at weak coupling, J. High Energy Phys. 06 (2008) 048, http://dx.doi.org/10.1088/1126-6708/2008/06/048, arXiv:0801.3002.
- [154] X.O. Camanho, J.D. Edelstein, J. Maldacena, A. Zhiboedov, Causality constraints on corrections to the graviton three-point coupling, J. High Energy Phys. 02 (2016) 020, http://dx.doi.org/10.1007/JHEP02(2016)020, arXiv:1407.5597.
- [155] M. Kulaxizi, A. Parnachev, A. Zhiboedov, Bulk phase shift, CFT Regge limit and Einstein gravity, J. High Energy Phys. 06 (2018) 121, http://dx.doi.org/10.1007/JHEP06(2018)121, arXiv:1705.02934.
- [156] D. Li, D. Meltzer, D. Poland, Conformal bootstrap in the Regge limit, J. High Energy Phys. 12 (2017) 013, http://dx.doi.org/10.1007/JHEP12(2017) 013, arXiv:1705.03453.
- [157] D. Meltzer, AdS/CFT unitarity at higher loops: High-energy string scattering, J. High Energy Phys. 05 (2020) 133, http://dx.doi.org/10.1007/ JHEP05(2020)133, arXiv:1912.05580.
- [158] M. Kulaxizi, G.S. Ng, A. Parnachev, Black holes, heavy states, phase shift and anomalous dimensions, SciPost Phys. 6 (6) (2019) 065, http://dx.doi.org/10.21468/SciPostPhys.6.6.065, arXiv:1812.03120.
- [159] R. Karlsson, M. Kulaxizi, A. Parnachev, P. Tadić, Black holes and conformal Regge bootstrap, J. High Energy Phys. 10 (2019) 046, http: //dx.doi.org/10.1007/JHEP10(2019)046, arXiv:1904.00060.
- [160] Y.-Z. Li, Heavy-light bootstrap from Lorentzian inversion formula, J. High Energy Phys. 07 (2020) 046, http://dx.doi.org/10.1007/JHEP07(2020) 046, arXiv:1910.06357.
- [161] Y.-Z. Li, H.-Y. Zhang, More on heavy-light bootstrap up to double-stress-tensor, J. High Energy Phys. 10 (2020) 055, http://dx.doi.org/10.1007/ JHEP10(2020)055, arXiv:2004.04758.
- [162] S. Giusto, M.R.R. Hughes, R. Russo, The Regge limit of AdS₃ holographic correlators, J. High Energy Phys. 11 (2020) 018, http://dx.doi.org/10. 1007/JHEP11(2020)018, arXiv:2007.12118.
- [163] N. Ceplak, M.R.R. Hughes, The Regge limit of AdS₃ holographic correlators with heavy states: towards the black hole regime, J. High Energy Phys. 07 (2021) 021, http://dx.doi.org/10.1007/JHEP07(2021)021, arXiv:2102.09549.
- [164] A. Antunes, M.S. Costa, T. Hansen, A. Salgarkar, S. Sarkar, The perturbative CFT optical theorem and high-energy string scattering in AdS at one loop, J. High Energy Phys. 04 (2021) 088, http://dx.doi.org/10.1007/JHEP04(2021)088, arXiv:2012.01515.
- [165] M.S. Costa, V. Goncalves, A. Salgarkar, J. Vilas Boas, Conformal multi-Regge theory, 2023, arXiv:2305.10394.
- [166] L.P. de Gioia, A.-M. Raclariu, Eikonal approximation in celestial CFT, J. High Energy Phys. 03 (2023) 030, http://dx.doi.org/10.1007/JHEP03(2023) 030, arXiv:2206.10547.
- [167] J. Parra-Martinez, M.S. Ruf, M. Zeng, Extremal black hole scattering at $O(G^3)$: graviton dominance, eikonal exponentiation, and differential equations, J. High Energy Phys. 11 (2020) 023, http://dx.doi.org/10.1007/JHEP11(2020)023, arXiv:2005.04236.
- [168] S.B. Giddings, R.A. Porto, The gravitational S-matrix, Phys. Rev. D81 (2010) 025002, http://dx.doi.org/10.1103/PhysRevD.81.025002, arXiv: 0908.0004.
- [169] B. Bellazzini, G. Isabella, M.M. Riva, Classical vs quantum eikonal scattering and its causal structure, J. High Energy Phys. 04 (2023) 023, http://dx.doi.org/10.1007/JHEP04(2023)023, arXiv:2211.00085.
- [170] I. Buric, F. Russo, A. Vichi, Spinning partial waves for scattering amplitudes in d dimensions, J. High Energy Phys. 10 (2023) 090, http: //dx.doi.org/10.1007/JHEP10(2023)090, arXiv:2305.18523.

- [171] M. Levy, J. Sucher, Eikonal approximation in quantum field theory, Phys. Rev. 186 (1969) 1656–1670, http://dx.doi.org/10.1103/PhysRev.186. 1656.
- [172] M. Fabbrichesi, R. Pettorino, G. Veneziano, G.A. Vilkovisky, Planckian energy scattering and surface terms in the gravitational action, Nuclear Phys. B 419 (1994) 147–188, http://dx.doi.org/10.1016/0550-3213(94)90361-1, arXiv:hep-th/9309037.
- [173] L. Maiani, M. Testa, Unstable systems in relativistic quantum field theory, Ann. Physics 263 (1998) 353–367, http://dx.doi.org/10.1006/aphy. 1997.5762, arXiv:hep-th/9709110.
- [174] M. Ciafaloni, D. Colferai, Rescattering corrections and self-consistent metric in Planckian scattering, J. High Energy Phys. 10 (2014) 085, http://dx.doi.org/10.1007/JHEP10(2014)085, arXiv:1406.6540.
- [175] B. Bellazzini, G. Isabella, M. Lewandowski, F. Sgarlata, Gravitational causality and the self-stress of photons, J. High Energy Phys. 05 (2022) 154, http://dx.doi.org/10.1007/JHEP05(2022)154, arXiv:2108.05896.
- [176] V. Ferrari, P. Pendenza, G. Veneziano, Beamlike gravitational waves and their geodesics, Gen. Relativity Gravitation 20 (1988) 1185–1191, http://dx.doi.org/10.1007/BF00758938.
- [177] T. Adamo, A. Cristofoli, P. Tourkine, Eikonal amplitudes from curved backgrounds, SciPost Phys. 13 (2) (2022) 032, http://dx.doi.org/10.21468/ SciPostPhys.13.2.032, arXiv:2112.09113.
- [178] P. Aichelburg, R. Sexl, On the gravitational field of a massless particle, Gen. Relativity Gravitation 2 (1971) 303-312, http://dx.doi.org/10.1007/ BF00758149.
- [179] T. Dray, G. 't Hooft, The gravitational shock wave of a massless particle, Nuclear Phys. B 253 (1985) 173–188, http://dx.doi.org/10.1016/0550-3213(85)90525-5.
- [180] G.M. Shore, Memory, Penrose limits and the geometry of gravitational shockwaves and gyratons, J. High Energy Phys. 12 (2018) 133, http://dx.doi.org/10.1007/JHEP12(2018)133, arXiv:1811.08827.
- [181] Y. Mino, M. Sasaki, T. Tanaka, Gravitational radiation reaction to a particle motion, Phys. Rev. D 55 (1997) 3457–3476, http://dx.doi.org/10. 1103/PhysRevD.55.3457, arXiv:gr-qc/9606018.
- [182] T.C. Quinn, R.M. Wald, An axiomatic approach to electromagnetic and gravitational radiation reaction of particles in curved space-time, Phys. Rev. D 56 (1997) 3381–3394, http://dx.doi.org/10.1103/PhysRevD.56.3381, arXiv:gr-qc/9610053.
- [183] E. Poisson, A. Pound, I. Vega, The motion of point particles in curved spacetime, Living Rev. Rel. 14 (2011) 7, http://dx.doi.org/10.12942/lrr-2011-7, arXiv:1102.0529.
- [184] R. Penrose, 1974. (unpublished).
- [185] D.M. Eardley, S.B. Giddings, Classical black hole production in high-energy collisions, Phys. Rev. D66 (2002) 044011, http://dx.doi.org/10.1103/ PhysRevD.66.044011, arXiv:gr-qc/0201034.
- [186] E. Kohlprath, G. Veneziano, Black holes from high-energy beam-beam collisions, J. High Energy Phys. 06 (2002) 057, arXiv:gr-qc/0203093.
- [187] C. Gundlach, Critical phenomena in gravitational collapse, Phys. Rep. 376 (2003) 339–405, http://dx.doi.org/10.1016/S0370-1573(02)00560-4, arXiv:gr-qc/0210101.
- [188] D.N. Page, Can two ultrarelativistic objects lose almost all their energy to gravitational radiation? Phys. Rev. D 107 (6) (2023) 064057, http://dx.doi.org/10.1103/PhysRevD.107.064057, arXiv:2212.03890.
- [189] D. Bini, T. Damour, Gravitational scattering of two black holes at the fourth post-Newtonian approximation, Phys. Rev. D 96 (6) (2017) 064021, http://dx.doi.org/10.1103/PhysRevD.96.064021, arXiv:1706.06877.
- [190] H. Kawai, D.C. Lewellen, S.H.H. Tye, A relation between tree amplitudes of closed and open strings, Nuclear Phys. B 269 (1986) 1–23, http://dx.doi.org/10.1016/0550-3213(86)90362-7.
- [191] F. Cachazo, A. Guevara, Leading singularities and classical gravitational scattering, 2017, arXiv:1705.10262.
- [192] Z. Bern, J.J.M. Carrasco, H. Johansson, New relations for gauge-theory amplitudes, Phys. Rev. D 78 (2008) 085011, http://dx.doi.org/10.1103/ PhysRevD.78.085011, arXiv:0805.3993.
- [193] W.D. Goldberger, S.G. Prabhu, J.O. Thompson, Classical gluon and graviton radiation from the bi-adjoint scalar double copy, Phys. Rev. D 96 (6) (2017) 065009, http://dx.doi.org/10.1103/PhysRevD.96.065009, arXiv:1705.09263.
- [194] W.D. Goldberger, A.K. Ridgway, Bound states and the classical double copy, Phys. Rev. D 97 (8) (2018) 085019, http://dx.doi.org/10.1103/ PhysRevD.97.085019, arXiv:1711.09493.
- [195] W.D. Goldberger, J. Li, S.G. Prabhu, Spinning particles, axion radiation, and the classical double copy, Phys. Rev. D 97 (10) (2018) 105018, http://dx.doi.org/10.1103/PhysRevD.97.105018, arXiv:1712.09250.
- [196] J. Li, S.G. Prabhu, Gravitational radiation from the classical spinning double copy, Phys. Rev. D 97 (10) (2018) 105019, http://dx.doi.org/10. 1103/PhysRevD.97.105019, arXiv:1803.02405.
- [197] C.-H. Shen, Gravitational radiation from color-kinematics duality, J. High Energy Phys. 11 (2018) 162, http://dx.doi.org/10.1007/JHEP11(2018) 162, arXiv:1806.07388.
- [198] W.D. Goldberger, J. Li, Strings, extended objects, and the classical double copy, J. High Energy Phys. 02 (2020) 092, http://dx.doi.org/10.1007/ JHEP02(2020)092, arXiv:1912.01650.
- [199] F. Bastianelli, F. Comberiati, L. de la Cruz, Worldline description of a bi-adjoint scalar and the zeroth copy, J. High Energy Phys. 12 (2021) 023, http://dx.doi.org/10.1007/JHEP12(2021)023, arXiv:2107.10130.
- [200] C. Shi, J. Plefka, Classical double copy of worldline quantum field theory, Phys. Rev. D 105 (2) (2022) 026007, http://dx.doi.org/10.1103/ PhysRevD.105.026007, arXiv:2109.10345.
- [201] Z. Bern, L.J. Dixon, R. Roiban, Is N=8 supergravity ultraviolet finite? Phys. Lett. B 644 (2007) 265–271, http://dx.doi.org/10.1016/j.physletb. 2006.11.030, arXiv:hep-th/0611086.
- [202] Z. Bern, J.J. Carrasco, L.J. Dixon, H. Johansson, D.A. Kosower, R. Roiban, Three-loop superfiniteness of N=8 supergravity, Phys. Rev. Lett. 98 (2007) 161303, http://dx.doi.org/10.1103/PhysRevLett.98.161303, arXiv:hep-th/0702112.
- [203] Z. Bern, J.J. Carrasco, L.J. Dixon, H. Johansson, R. Roiban, The ultraviolet behavior of N=8 supergravity at four loops, Phys. Rev. Lett. 103 (2009) 081301, http://dx.doi.org/10.1103/PhysRevLett.103.081301, arXiv:0905.2326.
- [204] Z. Bern, S. Davies, T. Dennen, Enhanced ultraviolet cancellations in N = 5 supergravity at four loops, Phys. Rev. D 90 (10) (2014) 105011, http://dx.doi.org/10.1103/PhysRevD.90.105011, arXiv:1409.3089.
- [205] Z. Bern, J.J. Carrasco, W.-M. Chen, A. Edison, H. Johansson, J. Parra-Martinez, R. Roiban, M. Zeng, Ultraviolet properties of N = 8 supergravity at five loops, Phys. Rev. D98 (8) (2018) 086021, http://dx.doi.org/10.1103/PhysRevD.98.086021, arXiv:1804.09311.
- [206] M. Green, M. Green, J. Schwarz, E. Witten, Superstring Theory, in: Cambridge Monographs on Mathematical Physics, Cambridge University Press, 1988.
- [207] V.P. Nair, A current algebra for some gauge theory amplitudes, Phys. Lett. B 214 (1988) 215–218, http://dx.doi.org/10.1016/0370-2693(88) 91471-2.
- [208] R.H. Boels, D. O'Connell, Simple superamplitudes in higher dimensions, J. High Energy Phys. 06 (2012) 163, http://dx.doi.org/10.1007/ JHEP06(2012)163, arXiv:1201.2653.
- [209] H. Elvang, Y.-t. Huang, Scattering Amplitudes in Gauge Theory and Gravity, Cambridge University Press, 2015.

- [210] A. Koemans Collado, S. Thomas, Eikonal scattering in Kaluza-Klein gravity, J. High Energy Phys. 04 (2019) 171, http://dx.doi.org/10.1007/ JHEP04(2019)171, arXiv:1901.05869.
- [211] S. Caron-Huot, Z. Zahraee, Integrability of black hole orbits in maximal supergravity, J. High Energy Phys. 07 (2019) 179, http://dx.doi.org/10. 1007/JHEP07(2019)179, arXiv:1810.04694.
- [212] E.T. Newman, A.I. Janis, Note on the Kerr spinning particle metric, J. Math. Phys. 6 (1965) 915-917, http://dx.doi.org/10.1063/1.1704350.
- [213] Z. Zhang, G. Fan, J. Jia, Effect of particle spin on trajectory deflection and gravitational lensing, J. Cosmol. Astropart. Phys. 09 (2022) 061, http://dx.doi.org/10.1088/1475-7516/2022/09/061, arXiv:2207.09194.
- [214] C. Angelantonj, A. Sagnotti, Open strings, Phys. Rep. 371 (2002) 1–150, http://dx.doi.org/10.1016/S0370-1573(02)00273-9, arXiv:hep-th/ 0204089. Erratum; Phys. Rep. 376 (2003) 407.
- [215] I.R. Klebanov, L. Thorlacius, The size of p-branes, Phys. Lett. B371 (1996) 51-56, http://dx.doi.org/10.1016/0370-2693(95)01576-0, arXiv:hep-th/9510200.
- [216] M.R. Garousi, R.C. Myers, Superstring scattering from D-branes, Nuclear Phys. B475 (1996) 193-224, http://dx.doi.org/10.1016/0550-3213(96) 00316-1, arXiv:hep-th/9603194.
- [217] A. Hashimoto, I.R. Klebanov, Scattering of strings from D-branes, Nuclear Phys. Proc. Suppl. 55B (1997) 118-133, arXiv:hep-th/9611214.
- [218] S. Stieberger, Open & closed vs. Pure open string disk amplitudes, 2009, arXiv:0907.2211.
- [219] M.R. Garousi, Superstring scattering from D-branes bound states, J. High Energy Phys. 12 (1998) 008, http://dx.doi.org/10.1088/1126-6708/1998/12/008, arXiv:hep-th/9805078.
- [220] G. D'Appollonio, P. Di Vecchia, R. Russo, G. Veneziano, A microscopic description of absorption in high-energy string-brane collisions, J. High Energy Phys. 03 (2016) 030, http://dx.doi.org/10.1007/JHEP03(2016)030, arXiv:1510.03837.
- [221] E. Cremmer, J. Scherk, Factorization of the Pomeron sector and currents in the dual resonance model, Nuclear Phys. B 50 (1972) 222–252, http://dx.doi.org/10.1016/S0550-3213(72)80016-6.
- [222] L. Clavelli, J.A. Shapiro, Pomeron factorization in general dual models, Nuclear Phys. B 57 (1973) 490-535, http://dx.doi.org/10.1016/0550-3213(73)90113-2.
- [223] E. Del Giudice, P. Di Vecchia, S. Fubini, General properties of the dual resonance model, Ann. Physics 70 (1972) 378–398, http://dx.doi.org/ 10.1016/0003-4916(72)90272-2.
- [224] M. Ademollo, E. Del Giudice, P. Di Vecchia, S. Fubini, Couplings of three excited particles in the dual-resonance model, Nuovo Cim. A19 (1974) 181–203, http://dx.doi.org/10.1007/BF02801846.
- [225] M. Ademollo, A. Bellini, M. Ciafaloni, Superstring Regge amplitudes and emission vertices, Phys. Lett. B223 (1989) 318-324, http://dx.doi.org/ 10.1016/0370-2693(89)91609-2.
- [226] M. Ademollo, A. Bellini, M. Ciafaloni, Superstring Regge amplitudes and graviton radiation at planckian energies, Nuclear Phys. B338 (1990) 114–142, http://dx.doi.org/10.1016/0550-3213(90)90626-0.
- [227] W. Black, C. Monni, High energy string-brane scattering for massive states, Nuclear Phys. B859 (2012) 299–320, http://dx.doi.org/10.1016/j. nuclphysb.2012.02.009, arXiv:1107.4321.
- [228] M. Bianchi, P. Teresi, Scattering higher spins off D-branes, J. High Energy Phys. 1201 (2012) 161, http://dx.doi.org/10.1007/JHEP01(2012)161, arXiv:1108.1071.
- [229] P. Di Vecchia, A. Liccardo, D branes in string theory, i, NATO Sci. Ser. C 556 (2000) 1–60, http://dx.doi.org/10.1007/978-94-011-4303-5_1, arXiv:hep-th/9912161.
- [230] P. Di Vecchia, A. Liccardo, D-branes in string theory. 2, in: YITP Workshop on Developments in Superstring and M Theory, 1999, arXiv:hepth/9912275.
- [231] S.B. Giddings, Locality in quantum gravity and string theory, Phys. Rev. D74 (2006) 106006, http://dx.doi.org/10.1103/PhysRevD.74.106006, arXiv:hep-th/0604072.
- [232] G. D'Appollonio, P. Vecchia, R. Russo, G. Veneziano, Microscopic unitary description of tidal excitations in high-energy string-brane collisions, J. High Energy Phys. 11 (2013) 126, http://dx.doi.org/10.1007/JHEP11(2013)126, arXiv:1310.1254.
- [233] M.B. Green, J.H. Schwarz, Superstring interactions, Nuclear Phys. B218 (1983) 43-88, http://dx.doi.org/10.1016/0550-3213(83)90475-3.
- [234] M.B. Green, J.H. Schwarz, L. Brink, Superfield theory of type II superstrings, Nuclear Phys. B219 (1983) 437–478, http://dx.doi.org/10.1016/0550-3213(83)90651-X.
- [235] R. Brower, K. Friedman, Spectrum generating algebra and no ghost theorem for the Neveu–Schwarz model, Phys. Rev. D7 (1973) 535–539, http://dx.doi.org/10.1103/PhysRevD.7.535.
- [236] J. Schwarz, Physical states and Pomeron poles in the dual pion model, Nuclear Phys. B46 (1972) 61-74, http://dx.doi.org/10.1016/0550-3213(72)90201-5.
- [237] K. Hornfeck, Three Reggeon light cone vertex of the Neveu-Schwarz string, Nuclear Phys. B293 (1987) 189, http://dx.doi.org/10.1016/0550-3213(87)90068-X.
- [238] D.P. Skliros, M.B. Hindmarsh, Covariant vertex operators for cosmic strings, 2009, arXiv:0911.5354.
- [239] M. Hindmarsh, D. Skliros, Covariant closed string coherent states, Phys. Rev. Lett. 106 (2011) 081602, http://dx.doi.org/10.1103/PhysRevLett. 106.081602, arXiv:1006.2559.
- [240] M. Bianchi, M. Firrotta, DDF operators, open string coherent states and their scattering amplitudes, Nuclear Phys. B 952 (2020) 114943, http://dx.doi.org/10.1016/j.nuclphysb.2020.114943, arXiv:1902.07016.
- [241] A. Aldi, M. Firrotta, String coherent vertex operators of Neveu-Schwarz and ramond states, Nuclear Phys. B 955 (2020) 115050, http: //dx.doi.org/10.1016/j.nuclphysb.2020.115050, arXiv:1912.06177.
- [242] M. Firrotta, V. Rosenhaus, Photon emission from an excited string, J. High Energy Phys. 09 (2022) 211, http://dx.doi.org/10.1007/JHEP09(2022) 211, arXiv:2207.01641.
- [243] E. Cremmer, J.-L. Gervais, Infinite component field theory of interacting relativistic strings and dual theory, Nuclear Phys. B90 (1975) 410–460, http://dx.doi.org/10.1016/0550-3213(75)90655-0.
- [244] S.B. Giddings, Black hole information, unitarity, and nonlocality, Phys. Rev. D 74 (2006) 106005, http://dx.doi.org/10.1103/PhysRevD.74.106005, arXiv:hep-th/0605196.
- [245] G. Veneziano, Topics in string theory, in: H.S. Mani, R. Ramachandran (Eds.), Proceedings of DST workshop on Particle Physics: Superstring Theory, Kanpur 1988, World Scientific Publishing Company, 1988, p. 1, URL: https://cds.cern.ch/record/187534/files/198806345.pdf.
- [246] R. Dolen, D. Horn, C. Schmid, Prediction of Regge parameters of rho poles from low-energy pi N data, Phys. Rev. Lett. 19 (1967) 402-407, http://dx.doi.org/10.1103/PhysRevLett.19.402.
- [247] R. Dolen, D. Horn, C. Schmid, Finite energy sum rules and their application to pi N charge exchange, Phys. Rev. 166 (1968) 1768–1781, http://dx.doi.org/10.1103/PhysRev.166.1768.
- [248] J. Polchinski, Dirichlet-branes and ramond-ramond charges, Phys. Rev. Lett. 75 (1995) 4724–4727, http://dx.doi.org/10.1103/PhysRevLett.75. 4724, arXiv:hep-th/9510017.
- [249] C. Bachas, D-brane dynamics, Phys. Lett. B 374 (1996) 37-42, http://dx.doi.org/10.1016/0370-2693(96)00238-9, arXiv:hep-th/9511043.

- [250] M. Billo, P. Di Vecchia, D. Cangemi, Boundary states for moving D-branes, Phys. Lett. B 400 (1997) 63-70, http://dx.doi.org/10.1016/S0370-2693(97)00329-8, arXiv:hep-th/9701190.
- [251] M.R. Douglas, D.N. Kabat, P. Pouliot, S.H. Shenker, D-branes and short distances in string theory, Nuclear Phys. B 485 (1997) 85-127, http://dx.doi.org/10.1016/S0550-3213(96)00619-0, arXiv:hep-th/9608024.
- [252] A.R. Fazio, E.A. Reyes R., A functional approach to the next-to-eikonal approximation of high energy gravitational scattering, Modern Phys. Lett. A 36 (20) (2021) 2150138, http://dx.doi.org/10.1142/S0217732321501388, arXiv:2102.00253.
- [253] M. Beneke, V.A. Smirnov, Asymptotic expansion of feynman integrals near threshold, Nuclear Phys. B 522 (1998) 321–344, http://dx.doi.org/ 10.1016/S0550-3213(98)00138-2, arXiv:hep-ph/9711391.
- [254] V.A. Smirnov, Applied asymptotic expansions in momenta and masses, Springer Tracts Mod. Phys. 177 (2002) 1-262.
- [255] S. Caron-Huot, M. Giroux, H.S. Hannesdottir, S. Mizera, What can be measured asymptotically? J. High Energy Phys. 01 (2024) 139, http://dx.doi.org/10.1007/JHEP01(2024)139, arXiv:2308.02125.
- [256] S. Caron-Huot, M. Giroux, H.S. Hannesdottir, S. Mizera, Crossing beyond scattering amplitudes, 2023, arXiv:2310.12199.
- [257] A. Elkhidir, D. O'Connell, M. Sergola, I.A. Vazquez-Holm, Radiation and reaction at one loop, 2023, arXiv:2303.06211.
- [258] D. Kosmopoulos, Simplifying D-dimensional physical-state sums in gauge theory and gravity, Phys. Rev. D 105 (5) (2022) 056025, http: //dx.doi.org/10.1103/PhysRevD.105.056025, arXiv:2009.00141.
- [259] K. Westpfahl, High-speed scattering of charged and uncharged particles in general relativity, Fortschr. Phys. 33 (8) (1985) 417–493, http://dx.doi.org/10.1002/prop.2190330802.
- [260] T. Damour, Classical and quantum scattering in post-Minkowskian gravity, Phys. Rev. D 102 (2) (2020) 024060, http://dx.doi.org/10.1103/ PhysRevD.102.024060, arXiv:1912.02139.
- [261] D. Bini, T. Damour, A. Geralico, Radiated momentum and radiation reaction in gravitational two-body scattering including time-asymmetric effects, Phys. Rev. D 107 (2) (2023) 024012, http://dx.doi.org/10.1103/PhysRevD.107.024012, arXiv:2210.07165.
- [262] K. Haddad, A. Helset, Tidal effects in quantum field theory, J. High Energy Phys. 12 (2020) 024, http://dx.doi.org/10.1007/JHEP12(2020)024, arXiv:2008.04920.
- [263] A. Pasquinucci, On the scattering of gravitons on two parallel D-branes, Modern Phys. Lett. A12 (1997) 1537–1546, http://dx.doi.org/10.1142/ S0217732397001564, arXiv:hep-th/9703066.
- [264] S. Lee, S.-J. Rey, Absorption and recoil of fundamental string by D string, Nuclear Phys. B508 (1997) 107–121, http://dx.doi.org/10.1016/S0550-3213(97)00610-X, arXiv:hep-th/9706115.
- [265] J. Polchinski, String Theory, Univ. Pr., Cambridge, UK, 1998, (1998).
- [266] M. Duff, R.R. Khuri, J. Lu, String solitons, Phys. Rep. 259 (1995) 213-326, http://dx.doi.org/10.1016/0370-1573(95)00002-X, arXiv:hep-th/9412184.
- [267] M.B. Green, J.H. Schwarz, Supersymmetrical string theories, Phys. Lett. B 109 (1982) 444-448, http://dx.doi.org/10.1016/0370-2693(82)91110-8.
- [268] N. Berkovits, ICTP lectures on covariant quantization of the superstring, ICTP Lect. Not. Ser. 13 (2003) 57-107, arXiv:hep-th/0209059.
- [269] C.R. Mafra, O. Schlotterer, Tree-level amplitudes from the pure spinor superstring, 2022, arXiv:2210.14241.
- [270] M. Blau, J.M. Figueroa-O'Farrill, G. Papadopoulos, Penrose limits, supergravity and brane dynamics, Classical Quantum Gravity 19 (2002) 4753, http://dx.doi.org/10.1088/0264-9381/19/18/310, arXiv:hep-th/0202111.
- [271] G. D'Appollonio, P. Di Vecchia, R. Russo, G. Veneziano, Regge behavior saves string theory from causality violations, J. High Energy Phys. 05 (2015) 144, http://dx.doi.org/10.1007/JHEP05(2015)144, arXiv:1502.01254.
- [272] H.J. de Vega, N.G. Sanchez, Quantum string scattering in the Aichelburg–Sexl geometry, Nuclear Phys. B317 (1989) 706–730, http://dx.doi.org/ 10.1016/0550-3213(89)90540-3.
- [273] G. Veneziano, Strings and gravitation, 1988, In *Perth 1988, Proceedings, Recent developments in theoretical and experimental general relativity, gravitation and relativistic field theories, pt. A* 173- 186.
- [274] G.T. Horowitz, A.R. Steif, Strings in strong gravitational fields, Phys. Rev. D42 (1990) 1950–1959, http://dx.doi.org/10.1103/PhysRevD.42.1950.
- [275] S.B. Giddings, D.J. Gross, A. Maharana, Gravitational effects in ultrahigh-energy string scattering, Phys. Rev. D77 (2008) 046001, http: //dx.doi.org/10.1103/PhysRevD.77.046001, arXiv:0705.1816.
- [276] G. Veneziano, Construction of a crossing symmetric, Regge behaved amplitude for linearly rising trajectories, Nuovo Cim. A57 (1968) 190–197, http://dx.doi.org/10.1007/BF02824451.
- [277] G. Veneziano, A stringy nature needs just two constants, Europhys. Lett. 2 (1986) 199, http://dx.doi.org/10.1209/0295-5075/2/3/006.
- [278] D.J. Gross, Superstrings and unification, in: 24th International Conference on High-Energy Physics, 1988.
- [279] G. Veneziano, An enlarged uncertainty principle from gedanken string collisions? Conf. Proc. C 8903131 (1989) 86-103.
- [280] G. Veneziano, String-theoretic unitary S-matrix at the threshold of black-hole production, J. High Energy Phys. 0411 (2004) 001, http: //dx.doi.org/10.1088/1126-6708/2004/11/001, arXiv:hep-th/0410166.
- [281] G. Veneziano, An unexpected use of the AGK rules in black-hole physics, in: Gribov-75: Memorial Workshop on Quarks, Hadrons, and Strong Interactions, 2005, pp. 496–509.
- [282] A. Addazi, M. Bianchi, G. Veneziano, Glimpses of black hole formation/evaporation in highly inelastic, ultra-planckian string collisions, J. High Energy Phys. 02 (2017) 111, http://dx.doi.org/10.1007/JHEP02(2017)111, arXiv:1611.03643.
- [283] V.A. Abramovsky, V.N. Gribov, O.V. Kancheli, Character of inclusive spectra and fluctuations produced in inelastic processes by multi Pomeron exchange, Yad. Fiz. 18 (1973) 595–616.
- [284] J. Koplik, A.H. Mueller, On the Abramovskii-Kancheli-Gribov Reggeon cutting rules, Phys. Lett. B 58 (1975) 166, http://dx.doi.org/10.1016/0370-2693(75)90629-2.
- [285] G. Veneziano, Quantum hair and the string-black hole correspondence, Classical Quantum Gravity 30 (2013) 092001, http://dx.doi.org/10.1088/ 0264-9381/30/9/092001, arXiv:1212.2606.
- [286] G.T. Horowitz, J. Polchinski, Selfgravitating fundamental strings, Phys. Rev. D 57 (1998) 2557–2563, http://dx.doi.org/10.1103/PhysRevD.57.2557, arXiv:hep-th/9707170.
- [287] T. Damour, G. Veneziano, Selfgravitating fundamental strings and black holes, Nuclear Phys. B 568 (2000) 93–119, http://dx.doi.org/10.1016/ S0550-3213(99)00596-9, arXiv:hep-th/9907030.
- [288] A. Luna, I. Nicholson, D. O'Connell, C.D. White, Inelastic black hole scattering from charged scalar amplitudes, J. High Energy Phys. 03 (2018) 044, http://dx.doi.org/10.1007/JHEP03(2018)044, arXiv:1711.03901.
- [289] T. Damour, Radiative contribution to classical gravitational scattering at the third order in *G*, Phys. Rev. D 102 (12) (2020) 124008, http://dx.doi.org/10.1103/PhysRevD.102.124008, arXiv:2010.01641.
- [290] A.R. Koemans Collado, The Eikonal Approximation and the Gravitational Dynamics of Binary Systems (Ph.D. thesis), Queen Mary, U. of London, 2020, arXiv:2009.11053.
- [291] G. Kälin, R.A. Porto, From boundary data to bound states, J. High Energy Phys. 01 (2020) 072, http://dx.doi.org/10.1007/JHEP01(2020)072, arXiv:1910.03008.
- [292] G. Kälin, R.A. Porto, From boundary data to bound states. Part II. Scattering angle to dynamical invariants (with twist), J. High Energy Phys. 02 (2020) 120, http://dx.doi.org/10.1007/JHEP02(2020)120, arXiv:1911.09130.

- [293] G. Cho, G. Kälin, R.A. Porto, From boundary data to bound states. Part III. Radiative effects, J. High Energy Phys. 04 (2022) 154, http: //dx.doi.org/10.1007/JHEP04(2022)154, arXiv:2112.03976.
- [294] K.S. Thorne, Nonradial pulsation of general-relativistic stellar models.IV. The weakfield limit, Astrophys. J. 158 (1969) 997, http://dx.doi.org/ 10.1086/150259.
- [295] W.L. Burke, Gravitational radiation damping of slowly moving systems calculated using matched asymptotic expansions, J. Math. Phys. 12 (1971) 401–418, http://dx.doi.org/10.1063/1.1665603.
- [296] D. Bini, T. Damour, Gravitational radiation reaction along general orbits in the effective one-body formalism, Phys. Rev. D 86 (2012) 124012, http://dx.doi.org/10.1103/PhysRevD.86.124012, arXiv:1210.2834.
- [297] T. Damour, P. Rettegno, Strong-field scattering of two black holes: Numerical relativity meets post-Minkowskian gravity, Phys. Rev. D 107 (6) (2023) 064051, http://dx.doi.org/10.1103/PhysRevD.107.064051, arXiv:2211.01399.
- [298] N. Bjerrum-Bohr, A. Cristofoli, P.H. Damgaard, Post-Minkowskian scattering angle in Einstein gravity, J. High Energy Phys. 08 (2020) 038, http://dx.doi.org/10.1007/JHEP08(2020)038, arXiv:1910.09366.
- [299] R. Gold, B. Brügmann, Eccentric black hole mergers and zoom-whirl behavior from elliptic inspirals to hyperbolic encounters, Phys. Rev. D 88 (6) (2013) 064051, http://dx.doi.org/10.1103/PhysRevD.88.064051, arXiv:1209.4085.
- [300] T. Damour, F. Guercilena, I. Hinder, S. Hopper, A. Nagar, L. Rezzolla, Strong-field scattering of two black holes: Numerics versus analytics, Phys. Rev. D 89 (8) (2014) 081503, http://dx.doi.org/10.1103/PhysRevD.89.081503, arXiv:1402.7307.
- [301] S. Hopper, A. Nagar, P. Rettegno, Strong-field scattering of two spinning black holes: Numerics versus analytics, 2022, arXiv:2204.10299.
- [302] M. Khalil, A. Buonanno, J. Steinhoff, J. Vines, Energetics and scattering of gravitational two-body systems at fourth post-Minkowskian order, Phys. Rev. D 106 (2) (2022) 024042, http://dx.doi.org/10.1103/PhysRevD.106.024042, arXiv:2204.05047.
- [303] T. Adamo, R. Gonzo, Bethe-Salpeter equation for classical gravitational bound states, 2022, arXiv:2212.13269.
- [304] S. Weinberg, Photons and gravitons in S-matrix theory: Derivation of charge conservation and equality of gravitational and inertial mass, Phys. Rev. 135 (1964) B1049-B1056, http://dx.doi.org/10.1103/PhysRev.135.B1049.
- [305] M. Mirbabayi, M. Porrati, Dressed hard states and black hole soft hair, Phys. Rev. Lett. 117 (21) (2016) 211301, http://dx.doi.org/10.1103/ PhysRevLett.117.211301, arXiv:1607.03120.
- [306] S. Choi, R. Akhoury, BMS supertranslation symmetry implies Faddeev-Kulish amplitudes, J. High Energy Phys. 02 (2018) 171, http://dx.doi. org/10.1007/JHEP02(2018)171, arXiv:1712.04551.
- [307] A. Laddha, A. Sen, Observational signature of the logarithmic terms in the soft graviton theorem, Phys. Rev. D100 (2) (2019) 024009, http://dx.doi.org/10.1103/PhysRevD.100.024009, arXiv:1806.01872.
- [308] B. Sahoo, A. Sen, Classical and quantum results on logarithmic terms in the soft theorem in four dimensions, J. High Energy Phys. 02 (2019) 086, http://dx.doi.org/10.1007/JHEP02(2019)086, arXiv:1808.03288.
- [309] A.P. Saha, B. Sahoo, A. Sen, Proof of the classical soft graviton theorem in D=4, J. High Energy Phys. 06 (2020) 153, http://dx.doi.org/10.1007/ JHEP06(2020)153, arXiv:1912.06413.
- [310] B. Sahoo, A. Sen, Classical soft graviton theorem rewritten, J. High Energy Phys. 01 (2022) 077, http://dx.doi.org/10.1007/JHEP01(2022)077, arXiv:2105.08739.
- [311] F. Bloch, A. Nordsieck, Note on the radiation field of the electron, Phys. Rev. 52 (1937) 54-59, http://dx.doi.org/10.1103/PhysRev.52.54.
- [312] W.E. Thirring, B. Touschek, A covariant formulation of the Bloch–Nordsieck method, Phil. Mag. Ser. 7 42 (1951) 244–249.
- [313] N. Arkani-Hamed, M. Pate, A.-M. Raclariu, A. Strominger, Celestial amplitudes from UV to IR, J. High Energy Phys. 08 (2021) 062, http: //dx.doi.org/10.1007/JHEP08(2021)062, arXiv:2012.04208.
- [314] S. Weinberg, The Quantum Theory of Fields. Vol. 1: Foundations, Cambridge University Press, 2005.
- [315] A. Strominger, A. Zhiboedov, Gravitational memory, BMS supertranslations and soft theorems, J. High Energy Phys. 01 (2016) 086, http: //dx.doi.org/10.1007/JHEP01(2016)086, arXiv:1411.5745.
- [316] A. Strominger, Lectures on the Infrared Structure of Gravity and Gauge Theory, Princeton University Press, 2018, URL: https://arxiv.org/abs/ 1703.05448.
- [317] Y.B. Zel'dovich, A.G. Polnarev, Radiation of gravitational waves by a cluster of superdense stars, Sov. Astron. 18 (1974) 17.
- [318] D. Christodoulou, Nonlinear nature of gravitation and gravitational wave experiments, Phys. Rev. Lett. 67 (1991) 1486–1489, http://dx.doi.org/ 10.1103/PhysRevLett.67.1486.
- [319] A.G. Wiseman, C.M. Will, Christodoulou's nonlinear gravitational wave memory: Evaluation in the quadrupole approximation, Phys. Rev. D 44 (10) (1991) R2945–R2949, http://dx.doi.org/10.1103/PhysRevD.44.R2945.
- [320] K.S. Thorne, Gravitational-wave bursts with memory: The christodoulou effect, Phys. Rev. D 45 (2) (1992) 520–524, http://dx.doi.org/10.1103/ PhysRevD.45.520.
- [321] G. Veneziano, G.A. Vilkovisky, Angular momentum loss in gravitational scattering, radiation reaction, and the Bondi gauge ambiguity, 2022, arXiv:2201.11607.
- [322] S. Weinberg, Gravitation and Cosmology: Principles and Applications of the General Theory of Relativity, John Wiley and Sons, New York, 1972.
- [323] A. Ashtekar, T. De Lorenzo, N. Khera, Compact binary coalescences: The subtle issue of angular momentum, Phys. Rev. D 101 (4) (2020) 044005, http://dx.doi.org/10.1103/PhysRevD.101.044005, arXiv:1910.02907.
- [324] H. Bondi, Gravitational waves in general relativity, Nature 186 (4724) (1960) http://dx.doi.org/10.1038/186535a0, 535-535.
- [325] R.K. Sachs, Gravitational waves in general relativity. 8. Waves in asymptotically flat space-times, Proc. R. Soc. Lond. Ser. A 270 (1962) 103–126, http://dx.doi.org/10.1098/rspa.1962.0206.
- [326] H. Bondi, M.G.J. van der Burg, A.W.K. Metzner, Gravitational waves in general relativity. 7. Waves from axisymmetric isolated systems, Proc. R. Soc. Lond. Ser. A 269 (1962) 21–52, http://dx.doi.org/10.1098/rspa.1962.0161.
- [327] R. Sachs, Asymptotic symmetries in gravitational theory, Phys. Rev. 128 (1962) 2851-2864, http://dx.doi.org/10.1103/PhysRev.128.2851.
- [328] P.-N. Chen, M.-T. Wang, Y.-K. Wang, S.-T. Yau, Supertranslation invariance of angular momentum, Adv. Theor. Math. Phys. 25 (3) (2021) 777-789, http://dx.doi.org/10.4310/ATMP.2021.v25.n3.a4, arXiv:2102.03235.
- [329] R. Javadinezhad, M. Porrati, Supertranslation-invariant formula for the angular momentum flux in gravitational scattering, Phys. Rev. Lett. 130 (1) (2023) 011401, http://dx.doi.org/10.1103/PhysRevLett.130.011401, arXiv:2211.06538.
- [330] P. Mao, J.-B. Wu, X. Wu, Angular momentum and memory effect, Phys. Rev. D 107 (10) (2023) L101501, http://dx.doi.org/10.1103/PhysRevD. 107.L101501, arXiv:2301.08032.
- [331] M.M. Riva, F. Vernizzi, L.K. Wong, Angular momentum balance in gravitational two-body scattering: Flux, memory, and supertranslation invariance, 2023, arXiv:2302.09065.
- [332] A.V. Manohar, A.K. Ridgway, C.-H. Shen, Radiated angular momentum and dissipative effects in classical scattering, Phys. Rev. Lett. 129 (12) (2022) 121601, http://dx.doi.org/10.1103/PhysRevLett.129.121601, arXiv:2203.04283.
- [333] P.D. D'Eath, High speed black hole encounters and gravitational radiation, Phys. Rev. D18 (1978) 990, http://dx.doi.org/10.1103/PhysRevD.18. 990.

- [334] A. Gruzinov, G. Veneziano, Gravitational radiation from massless particle collisions, Classical Quantum Gravity 33 (12) (2016) 125012, http://dx.doi.org/10.1088/0264-9381/33/12/125012, arXiv:1409.4555.
- [335] M. Ciafaloni, D. Colferai, F. Coradeschi, G. Veneziano, Unified limiting form of graviton radiation at extreme energies, Phys. Rev. D93 (4) (2016) 044052, http://dx.doi.org/10.1103/PhysRevD.93.044052, arXiv:1512.00281.
- [336] R. Britto, R. Gonzo, G.R. Jehu, Graviton particle statistics and coherent states from classical scattering amplitudes, 2021, arXiv:2112.07036.
- [337] S.J. Kovacs, K.S. Thorne, The generation of gravitational waves. 3. Derivation of bremsstrahlung formulas, Astrophys. J. 217 (1977) 252–280, http://dx.doi.org/10.1086/155576.
- [338] S.J. Kovacs, K.S. Thorne, The generation of gravitational waves. 4. Bremsstrahlung, Astrophys. J. 224 (1978) 62–85, http://dx.doi.org/10.1086/ 156350.
- [339] L. Smarr, Gravitational radiation from distant encounters and from headon collisions of black holes: The zero frequency limit, Phys. Rev. D 15 (1977) 2069–2077, http://dx.doi.org/10.1103/PhysRevD.15.2069.
- [340] N.E.J. Bjerrum-Bohr, J.F. Donoghue, B.R. Holstein, Quantum gravitational corrections to the nonrelativistic scattering potential of two masses, Phys. Rev. D 67 (2003) 084033, http://dx.doi.org/10.1103/PhysRevD.71.069903, arXiv:hep-th/0211072. Erratum; Phys. Rev. D 71 (2005) 069903.
- [341] B.R. Holstein, J.F. Donoghue, Classical physics and quantum loops, Phys. Rev. Lett. 93 (2004) 201602, http://dx.doi.org/10.1103/PhysRevLett.93. 201602, arXiv:hep-th/0405239.
- [342] V.A. Smirnov, Analytical result for dimensionally regularized massive on-shell planar double box, Phys. Lett. B 524 (2002) 129–136, http://dx.doi.org/10.1016/S0370-2693(01)01382-X, arXiv:hep-ph/0111160.
- [343] C. Anastasiou, K. Melnikov, Higgs boson production at hadron colliders in NNLO QCD, Nuclear Phys. B 646 (2002) 220–256, http://dx.doi.org/ 10.1016/S0550-3213(02)00837-4, arXiv:hep-ph/0207004.
- [344] C. Anastasiou, L.J. Dixon, K. Melnikov, NLO higgs boson rapidity distributions at hadron colliders, Nucl. Phys. B Proc. Suppl. 116 (2003) 193–197, http://dx.doi.org/10.1016/S0920-5632(03)80168-8, arXiv:hep-ph/0211141.
- [345] C. Anastasiou, L.J. Dixon, K. Melnikov, F. Petriello, Dilepton rapidity distribution in the Drell-Yan process at NNLO in QCD, Phys. Rev. Lett. 91 (2003) 182002, http://dx.doi.org/10.1103/PhysRevLett.91.182002, arXiv:hep-ph/0306192.
- [346] C. Anastasiou, C. Duhr, F. Dulat, E. Furlan, F. Herzog, B. Mistlberger, Soft expansion of double-real-virtual corrections to higgs production at N³LO, J. High Energy Phys. 08 (2015) 051, http://dx.doi.org/10.1007/JHEP08(2015)051, arXiv:1505.04110.
- [347] N. Berkovits, E. D'Hoker, M.B. Green, H. Johansson, O. Schlotterer, Snowmass white paper: String perturbation theory, in: Snowmass 2021, 2022, arXiv:2203.09099.
- [348] M.V.S. Saketh, J. Vines, J. Steinhoff, A. Buonanno, Conservative and radiative dynamics in classical relativistic scattering and bound systems, Phys. Rev. Res. 4 (1) (2022) 013127, http://dx.doi.org/10.1103/PhysRevResearch.4.013127, arXiv:2109.05994.
- [349] L. Cangemi, M. Chiodaroli, H. Johansson, A. Ochirov, P. Pichini, E. Skvortsov, Kerr black holes enjoy massive higher-spin gauge symmetry, 2022, arXiv:2212.06120.
- [350] J.-W. Kim, J. Steinhoff, Spin supplementary condition in quantum field theory, part I : covariant SSC and physical state projection, 2023, arXiv:2302.01944.
- [351] F. Alessio, Kerr binary dynamics from minimal coupling and double copy, 2023, arXiv:2303.12784.
- [352] M. Bianchi, C. Gambino, F. Riccioni, A rutherford-like formula for scattering off Kerr-Newman BHs and subleading corrections, 2023, arXiv:2306.08969.
- [353] L. Cangemi, P. Pichini, Classical limit of higher-spin string amplitudes, 2022, arXiv:2207.03947.
- [354] G.U. Jakobsen, G. Mogull, J. Plefka, B. Sauer, Y. Xu, Conservative scattering of spinning black holes at fourth post-Minkowskian order, 2023, arXiv:2306.01714.
- [355] D. Amati, M. Ciafaloni, G. Veneziano, Towards an S-matrix description of gravitational collapse, J. High Energy Phys. 02 (2008) 049, http://dx.doi.org/10.1088/1126-6708/2008/02/049, arXiv:0712.1209.
- [356] L.N. Lipatov, High-energy scattering in QCD and in quantum gravity and two-dimensional field theories, Nuclear Phys. B 365 (1991) 614–632, http://dx.doi.org/10.1016/0550-3213(91)90512-V.
- [357] G. Marchesini, E. Onofri, High energy gravitational scattering: A numerical study, J. High Energy Phys. 0806 (2008) 104, http://dx.doi.org/10. 1088/1126-6708/2008/06/104, arXiv:0803.0250.
- [358] G. Veneziano, J. Wosiek, Exploring an S-matrix for gravitational collapse, J. High Energy Phys. 0809 (2008) 023, http://dx.doi.org/10.1088/1126-6708/2008/09/023, arXiv:0804.3321.
- [359] G. Veneziano, J. Wosiek, Exploring an S-matrix for gravitational collapse. II. A momentum space analysis, J. High Energy Phys. 0809 (2008) 024, http://dx.doi.org/10.1088/1126-6708/2008/09/024, arXiv:0805.2973.
- [360] M. Ciafaloni, D. Colferai, G. Falcioni, Unitarity alternatives in the reduced-action model for gravitational collapse, J. High Energy Phys. 1109 (2011) 044, http://dx.doi.org/10.1007/JHEP09(2011)044, arXiv:1106.5628.
- [361] M. Ciafaloni, D. Colferai, Unitarity restoring graviton radiation in the collapse regime of gravitational scattering, Phys. Rev. D 96 (12) (2017) 126008, http://dx.doi.org/10.1103/PhysRevD.96.126008, arXiv:1709.08405.
- [362] G.L. Almeida, S. Foffa, R. Sturani, Gravitational radiation contributions to the two-body scattering angle, Phys. Rev. D 107 (2) (2023) 024020, http://dx.doi.org/10.1103/PhysRevD.107.024020, arXiv:2209.11594.
- [363] Z. Bern, E. Herrmann, R. Roiban, M.S. Ruf, A.V. Smirnov, V.A. Smirnov, M. Zeng, Conservative binary dynamics at order O(α⁵) in electrodynamics, 2023, arXiv:2305.08981.
- [364] T. Damour, P. Jaranowski, G. Schäfer, Nonlocal-in-time action for the fourth post-Newtonian conservative dynamics of two-body systems, Phys. Rev. D 89 (6) (2014) 064058, http://dx.doi.org/10.1103/PhysRevD.89.064058, arXiv:1401.4548.
- [365] C.R. Galley, A.K. Leibovich, R.A. Porto, A. Ross, Tail effect in gravitational radiation reaction: Time nonlocality and renormalization group evolution, Phys. Rev. D 93 (2016) 124010, http://dx.doi.org/10.1103/PhysRevD.93.124010, arXiv:1511.07379.
- [366] A. Georgoudis, C. Heissenberg, R. Russo, An eikonal-inspired approach to the gravitational scattering waveform, 2023, arXiv:2312.07452.
- [367] D. Bini, T. Damour, A. Geralico, Comparing one-loop gravitational bremsstrahlung amplitudes to the multipolar-post-Minkowskian waveform, Phys. Rev. D 108 (12) (2023) 124052, http://dx.doi.org/10.1103/PhysRevD.108.124052, arXiv:2309.14925.
- [368] R. Aoude, K. Haddad, C. Heissenberg, A. Helset, Leading-order gravitational radiation to all spin orders, Phys. Rev. D 109 (3) (2024) 036007, http://dx.doi.org/10.1103/PhysRevD.109.036007, arXiv:2310.05832.
- [369] A. Georgoudis, C. Heissenberg, R. Russo, Post-Newtonian multipoles from the next-to-leading post-Minkowskian gravitational waveform, 2024, arXiv:2402.06361.
- [370] D. Bini, T. Damour, S. De Angelis, A. Geralico, A. Herderschee, R. Roiban, F. Teng, Gravitational waveform: A tale of two formalisms, 2024, arXiv:2402.06604.
- [371] A. Bissi, A. Sinha, X. Zhou, Selected topics in analytic conformal bootstrap: A guided journey, Phys. Rep. 991 (2022) 1–89, http://dx.doi.org/ 10.1016/j.physrep.2022.09.004, arXiv:2202.08475.
- [372] T. Hartman, D. Mazac, D. Simmons-Duffin, A. Zhiboedov, Snowmass white paper: The analytic conformal bootstrap, in: Snowmass 2021, 2022, arXiv:2202.11012.

- [373] N. Gaddam, N. Groenenboom, G. 't Hooft, Quantum gravity on the black hole horizon, J. High Energy Phys. 01 (2022) 023, http://dx.doi.org/ 10.1007/JHEP01(2022)023, arXiv:2012.02357.
- [374] N. Gaddam, N. Groenenboom, 2 1 2N scattering: Eikonalisation and the page curve, J. High Energy Phys. 01 (2022) 146, http://dx.doi.org/10. 1007/JHEP01(2022)146, arXiv:2110.14673.
- [375] J. Maldacena, S.H. Shenker, D. Stanford, A bound on chaos, J. High Energy Phys. 08 (2016) 106, http://dx.doi.org/10.1007/JHEP08(2016)106, arXiv:1503.01409.
- [376] M.J. Duff, Quantum tree graphs and the Schwarzschild solution, Phys. Rev. D7 (1973) 2317–2326, http://dx.doi.org/10.1103/PhysRevD.7.2317.
- [377] G.U. Jakobsen, Schwarzschild-Tangherlini metric from scattering amplitudes, Phys. Rev. D 102 (10) (2020) 104065, http://dx.doi.org/10.1103/ PhysRevD.102.104065, arXiv:2006.01734.
- [378] S. Mougiakakos, P. Vanhove, Schwarzschild-Tangherlini metric from scattering amplitudes in various dimensions, Phys. Rev. D 103 (2) (2021) 026001, http://dx.doi.org/10.1103/PhysRevD.103.026001, arXiv:2010.08882.